School of Public Health

Predictors of Inappropriate Antibiotic Prescribing in Australian General Practice / Primary Care Settings

AMY ELIZABETH HARRISON

ORCID 0000-0002-7650-8885

This thesis is presented for the Degree of Doctor of Philosophy in the discipline of Public Health Curtin University

September 2023

1

2 DECLARATION

3

To the best of my knowledge and belief this thesis contains no material previously published by any other person except where due acknowledgement has been made. This thesis contains no material which has been accepted for any other degree or diploma in any university.

8

9 Human ethics

- 10 This research study received human research ethics approval from the Curtin University
- 11 Human Research Ethics Committee, Approval number: HRE2016-0433.
- 12 MedicineInsight Data Governance Committee, Approval Number DG 2016-011.
- 13
- . .
- 14
- 15
- 16
- 17
- 18
- 10
- 19
- 20 Amy Elizabeth Harrison
- 21
- 22 12 September 2023

1 ABSTRACT

2

Increasing antibiotic resistance is a threat to human health globally. Unnecessary use of antibiotics needs to be reduced to preserve the effectiveness of current antibiotics. It is the unnecessary prescribing of antibiotics which needs to be researched, to gain insights into what factors drive it so the unnecessary prescribing can be reduced.

7

The aim of this study was to define and identify predictors of inappropriate prescribing of systemic antibiotics for initial presentations of acute upper respiratory tract infection (URTI) and urinary tract infection (UTI). The presenting condition groups of interest were acute, uncomplicated URTI, including acute rhinosinusitis / non-specific URTI, acute pharyngitis and / or tonsillitis, acute otitis media, and influenza / influenza-like illness, as well as UTI limited to the condition of acute cystitis.

14

Large-scale, longitudinal datasets were obtained from general practice in the state of 15 Western Australia (WA). The reference point was the recommendations contained within the 16 Australian national therapeutic guidelines (the guidelines) for each condition. Mixed-effects 17 logistic regression models were used to elucidate patient- and practice-related factors 18 associated with inappropriate prescribing. Potential predictors of interest included patient 19 age, gender, socioeconomic status, comorbid conditions, as well as practice rurality / 20 remoteness. Aggregate trends over time in inappropriate antibiotic prescribing and overall 21 antibiotic prescribing were also examined. 22

23

Significant unnecessary antibiotic prescribing was identified for URTI conditions, and second-line antibiotics featured minimally for URTI. For both URTI and UTI conditions, there was substantial non-first-line antibiotic prescribing for initial presentations of infection. For UTI, culture and sensitivity testing were performed infrequently for children and men despite being recorded in the guidelines as mandatory for children and strongly recommended for men.

30

Young children had the lowest probability of inappropriate prescribing for URTI, but were at notably high probability of receiving non-first-line antibiotics for UTI. By URTI condition, the chance of receiving likely unnecessary prescribing was highest for the URTI condition of rhinosinusitis.

1 For URTI, the outcomes of likely unnecessary prescribing in general practice, the choice of antibiotic prescribed, and receiving prescriptions with repeats were all found to predict each 2 3 other. For URTI models of various levels of inappropriate prescribing, predictors also included URTI condition, patient allergy label status, mental health condition status, 4 5 comorbid conditions status, government concession status, socioeconomic disadvantage status and multiple URTI episode status. Non-patient-related predictors for URTI included 6 weekend consultation status, primary health network, prescribing reason recorded status 7 8 and practice size.

9

For UTI, antibiotic choice and receiving prescribing with repeats were also found to predict each other. Other predictors of likely inappropriate prescribing in UTI models included patient age, gender, comorbid condition status, repeat prescription status and urine dipstick and culture testing, temperature recording status and multiple UTI episodes.

14

Trend analyses identified downward trends in likely unnecessary antibiotic prescribing for
 URTI, however, increasing non-first-line prescribing was identified for both URTI and UTI.

17

The magnitude of likely inappropriate antibiotic prescribing occurring at several different levels and definitions suggests reason for concern. Despite some small improvements in prescribing practices found over time, more action is urgently needed. Among all models developed for URTI and UTI, individual general practitioners (GPs) were responsible for greater residual variation not explained by fixed effects than individual practices, indicating that individual practitioners' prescribing behaviour should be targeted in stewardship efforts.

This research presents multiple, new insights regarding predictors of likely inappropriate prescribing in WA general practice, and identifies several areas for further research. This research was supported with funding provided by the WA Primary Health Alliance. Expert advice was also obtained from practising GPs to guide this project and aid in the interpretation of results.

- 30
- 31
- 32

1 ACKNOWLEDGEMENTS

2

The author would like to acknowledge the Western Australian Primary Health Alliance (WAPHA) for the provision of funding for this research. The author also wishes to acknowledge the support of an Australian Government Research Training Program Scholarship.

7

Additionally, the author wishes to acknowledge MedicineInsight for its provision of access
to its rich, large-scale, de-identified general practice dataset for Western Australia which has
been vital for this research project.

11

I wish to acknowledge my supervisors for their support, encouragement, patience and
 invaluable guidance throughout this project, for which I am most grateful:

- 14 Professor Suzanne Robinson
- 15 Associate Professor Linda Selvey
- 16 Professor Mark Harris
- 17 Professor Rachael Moorin
- 18
- 19 I also wish to thank the Health Economics and Data Analytics team, within Curtin University's
- 20 School of Population Health, for its support, particularly Dr Ninh Ha and Dr David Youens.
- 21

I am sincerely thankful for the expert GP advice provided by Clinical Associate Professor
 Daniel Xu and Dr Damian Zilm.

- 24
- Amanda Ellis of Quenda Communications provided copy-editing services.
- 26
- 27
- 28

ACKNOWLEDGEMENT OF COUNTRY

1 2

We acknowledge that Curtin University works across hundreds of traditional lands and custodial groups in Australia, and with First Nations people around the globe. We wish to pay our deepest respects to their ancestors and members of their communities, past, present, and to their emerging leaders. Our passion and commitment to work with all Australians and peoples from across the world, including our First Nations peoples are at the core of the work we do, reflective of our institutions' values and commitment to our role as leaders in the Reconciliation space in Australia.

11

ΤA	BLE OF	CONTENTS	
DE	CLARATIC	DN	iii
AB	STRACT		iv
AC	KNOWLE	DGEMENTS	vi
AC	KNOWLE	DGEMENT OF COUNTRY	vii
TA	BLE OF C	ONTENTS	ix
TA	BLE OF FI	GURES	xiv
LIS	ST OF TAB	LES	xvii
LIS	ST OF ACR	ONYMS	xix
GL	OSSARY	OF MODEL OUTCOMES AND VARIABLES DEFINITIONS	XX
GL	OSSARY	OF MULTILEVEL MIXED-EFFECTS MODELING TERMINOLOGY	.xxiii
C⊦	IAPTER 1	INTRODUCTION	27
1.1	The probl	em	27
1.2	Hypothes	is	28
1.3	Objective	S	29
1.4	Significan	ice	30
1.5	Thesis str	ucture	31
C⊦	IAPTER 2	BACKGROUND	34
2.1	Introducti	on	34
	2.1.1	The problem: growing, global antibiotic resistance	34
	2.1.2	The problem locally: inappropriate antibiotic prescribing in Australian primary care	34
2.2	Antibiotics antibiotics	s, the development of antibiotic resistance and other side effects from ta	king 35
	2.2.1	Antibiotics and antimicrobials	35
	2.2.2	Antibiotic resistance concepts and side effects	36
2.3	Condition	s commonly suspected of receiving inappropriate antibiotic prescribing.	36
	2.3.1	Upper respiratory tract infection	37
	2.3.1.1	Acute rhinosinusitis / non-specific URTI	38
	2.3.1.2	Acute pharyngitis / tonsillitis	38
	2.3.1.3	Acute otitis media	38
	2.3.1.4	Influenza and influenza-like illness	39
	2.3.2	Urinary tract infection	39
	2.3.2.1	Acute cystitis	39
	2.3.3	Repeats issued on antibiotic prescriptions	40
2.4	Antibiotic	stewardship and the surveillance of antibiotic use	40
	2.4.1	What is antibiotic stewardship?	40
	 I AI DE AB AC AC TA LIS GL GL GL C⊢ 1.1 1.2 1.3 1.4 1.5 C⊢ 2.1 2.2 2.3 	IABLE OF DECLARATIO ABSTRACT ACKNOWLED ACKNOWLED TABLE OF CO TABLE OF FI LIST OF TAB LIST OF ACR GLOSSARY O CHAPTER 1 1.1 The probl 1.2 Hypothes 1.3 Objective 1.4 Significan 2.1 2.2 2.3 Condition	IABLE OF CONTENTS DECLARATION ABSTRACT ACKNOWLEDGEMENT OF COUNTRY TABLE OF CONTENTS TABLE OF CONTENTS TABLE OF FIGURES LIST OF TABLES LIST OF ACRONYMS GLOSSARY OF MODEL OUTCOMES AND VARIABLES DEFINITIONS GLOSSARY OF MULTILEVEL MIXED-EFFECTS MODELING TERMINOLOGY

1		2.4.2	What is surveillance and its role in antibiotic stewardship	41
2		2.4.3	Stewardship and surveillance of antibiotic use in Australian primary car	e42
3		2.4.3.1	Government surveillance, stewardship and other initiatives	42
4		2.4.3.2	Bettering the Evaluation and Care of Health survey	43
5		2.4.3.3	NPS MedicineWise	44
6		2.4.3.4	The Antibiotic Use and Resistance in Australia project	44
7		2.4.3.5	The Atlas of Healthcare Variation	46
8		2.4.4	Surveillance of Antimicrobial Resistance in Australia	46
9	2.5	Access to	general practice prescribing data in Australia	48
10 11	2.6	Literature prescribin	on primary care interventions and qualitative research on inappropriate g	49
12 13		2.6.1	A summary of interventions to reduce inappropriate prescribing in prima care	ary 49
14		2.6.2	A summary of qualitative research on inappropriate prescribing	50
15	2.7	Literature	review of studies using quantitative methods	52
16		2.7.1	Different definitions of inappropriate antibiotic prescribing	53
17		2.7.2	Considerations regarding the literature criteria	54
18		2.7.3	Literature review overview	55
19		2.7.3.1	Patient factors	55
20		2.7.3.2	Clinician and healthcare setting factors	57
21		2.7.3.3	Conclusions from the literature overview	58
22		2.7.3.4	What this thesis contributes	60
23	CH	APTER 3	METHODS	62
24	3.1	Introductio	วท	62
25	3.2	Establishr	nent of steering committee	63
26	3.3	Data sour	ce and sample size	63
27	3.4	Data clea	ning and preparation	64
28		3.4.1	Diagnoses	64
29		3.4.1.1	Upper respiratory tract infection	66
30		3.4.1.2	Urinary tract infection	66
31		3.4.2	Antibiotic prescriptions	67
32	3.5	Data anal	ysis variables	67
33		3.5.1	Response variables	67
34		3.5.1.1	Antibiotic prescribing: standard used for assessment	67
35 36		3.5.1.2	Upper respiratory tract infection: inappropriate decision, unnecessary antibiotic prescribing and ordered choice of antibiotic prescribed	71
37		3.5.1.3	Urinary tract infection: ordinal choice of antibiotic prescribed	75

1 2		3.5.1.4	Additional response variables common to upper respiratory tract infection and urinary tract infection	ion 76
3		3.5.2	Predictor / confounder variables	77
4		3.5.2.1	Patient-related variables	77
5		3.5.2.2	Clinical observations and pathology data	78
6		3.5.2.3	Consultation-related variables	78
7	3.6	Analytical	methods	78
8		3.6.1	Predictors of inappropriate antibiotic prescribing	78
9		3.6.2	Trends in inappropriate antibiotic prescribing	82
10	3.7	Synthesis	e findings	84
11 12	CH	APTER 4	PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR UPPER RESPIRATORY TRACT INFECTION	86
13	4.1	Introductio	on	86
14	4.2	Specific m	ethods	86
15	4.3	Results		90
16		4.3.1	Introduction to modelling - clustering considerations	.100
17 18		4.3.2	Model 0: predictors of inappropriate decisions (unnecessary prescribing versus necessary prescribing and appropriate non-prescribing)	g .100
19		4.3.2.1	Model 0: summary	.103
20 21		4.3.3	Model 1: predictors of unnecessary prescribing (versus necessary prescribing) among all antibiotic prescriptions	.103
22		4.3.3.1	Model 1: summary	.105
23		4.3.4	Model 2: predictors of increasing choice of antibiotic prescribed	.106
24		4.3.4.1	Model 2: summary	.110
25		4.3.5	Comparing mixed effects models with fixed effects models	.111
26		4.3.6	Residual variance unexplained by fixed effects	.112
27	4.4	Summary		.113
28 29	CH	APTER 5	PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR URINARY TRACT INFECTION	.116
30	5.1	Introductio	on	.116
31	5.2	Specific m	ethods	.116
32	5.3	Results		.119
33		5.3.1	Modelling introduction - clustering considerations	.125
34		5.3.2	Model 1: ordinal choice of antibiotic prescribed	.126
35		5.3.3	Model 3: predictors of repeat positive prescribing	.132
36		5.3.4	Modelling considerations	.134
37	5.4	Summary		.135

1 2	CH	IAPTER 6	ANALYSIS OF CHANGES IN ANTIBIOTIC PRESCRIBING FOR RESPIRATORY TRACT INFECTION CONDITIONS OVER TIME	UPPER 138
3 4	6.1	Backgrou	nd to trends analyses for upper respiratory tract infection and urina	ary tract 138
5	6.2	Introductio	on to trends analyses for upper respiratory tract infection	
6	6.3	Results		144
7		6.3.1	All upper respiratory tract infection conditions altogether	145
8		6.3.2	By individual upper respiratory tract infection condition	146
9		6.3.2.1	Acute rhinosinusitis	146
10		6.3.2.2	Acute pharyngitis / tonsillitis	152
11		6.3.2.3	Acute otitis media	154
12	6.3	Summary		
13 14	CH	IAPTER 7	ANALYSIS OF CHANGES IN ANTIBIOTIC PRESCRIBING FOR URINARY TRACT INFECTION OVER TIME	
15	7.1	Introductio	วท	
16	7.2	Results		
17		7.2.1	All patients with initial presentations of urinary tract infection	
18		7.2.2	Trends by patient group	
19		7.2.2.1	Women	
20		7.2.2.2	Men	
21		7.2.2.3	Children	
22	7.3	Summary		171
23	C⊦	IAPTER 8	DISCUSSION	
24	8.1	Overall fin	ndings	173
25	8.2	Upper res	piratory tract infection	174
26		8.2.1	Main descriptive findings for upper respiratory tract infection	174
27		8.2.2	Predictors identified for upper respiratory tract infection models	174
28		8.2.2.1	Patient age	176
29 30		8.2.2.2	Interaction between patient age group and gender in inappropriat decision model	e 176
31		8.2.2.3	Upper respiratory tract infection condition	
32		8.2.2.4	Patient allergy labels for penicillin	
33		8.2.2.5	Practice size	
34		8.2.2.6	Patient comorbid and mental health conditions	178
35		8.2.2.7	Other	178
36		8.2.3	Trends in prescribing for upper respiratory tract infection	179
37				

1	8.3	Urinary tra	act infection	180
2		8.3.1	Main descriptive findings for urinary tract infection	180
3		8.3.2	Predictors identified for urinary tract infection models	180
4		8.3.2.1	Patient age and gender and (un)complicated infection	181
5		8.3.2.2	Repeats issued on prescriptions	181
6		8.3.2.3	Urine dipstick and culture testing	181
7		8.3.2.4	Patient comorbid conditions	182
8		8.3.3	Trends in prescribing for urinary tract infection	182
9 10	8.4	Specific fi infection	ndings common to upper respiratory tract infection and urinary tract	183
11	8.5	Implicatio	ns to policy, practice and research	184
12		8.5.1	Future research directions	189
13	8.6	Limitation	S	190
14		8.6.1	Prescriptions and patient groups	190
15		8.6.2	Definition of inappropriate prescribing according to the guidelines	190
16		8.6.3	Diagnoses	191
17		8.6.4	Other potential covariates	191
18		8.6.5	Statistical limitations	191
19		8.6.6	Complexities of the project	192
20	8.7	Significan	ce	193
21	CH	IAPTER 9	CONCLUSION	194
22	RE	FERENCE	S	197
23	AP	PENDICES	5	227
24	AP	PENDICES	S TABLE OF CONTENTS	229
25				

1 TABLE OF FIGURES

_		
3	Figure 3-1:	Project flow diagram62
4 5	Figure 3-2:	Picture of example list of terms appearing in the diagnosis field relating to ear infection, followed by frequency of their exact occurrence in the dataset
6 7	Figure 3-3:	Flow chart of outcome variables for identification of predictors of inappropriate prescribing for initial presentations of upper respiratory tract infection
8 9	Figure 3-4:	Flow chart of inappropriate decisions and unnecessary prescribing models for initial presentations of upper respiratory tract infection
10 11 12	Figure 3-5:	Flow chart of response variables and models for ordinal choice of antibiotic prescribed, non-first-line prescribing, and repeat positive antibiotic prescribing for upper respiratory tract infection and urinary tract infection analyses
13 14 15 16	Figure 3-6:	Illustration of the form of the generalised linear mixed model, using direct quotation from Stroup WW. Generalized linear mixed models: modern concepts, methods and applications. 1st ed. Boca Raton, FL: CRC Press, an imprint of Taylor and Francis; 2012. Table 1.4, Typology of Linear Models; p. 20. (375) 81
17 18 19	Figure 3-7:	Groups of interest in trends analyses for upper respiratory tract infection, including influenza / influenza-like illness and urinary tract infection, by condition group
20 21	Figure 3-8:	Example simple moving average of outcome rate, with fitted lines, 95% confidence intervals, and Lowess rates
22 23	Figure 4-1:	Flow chart of variables, numerators and denominators used in models for initial presentations of upper respiratory tract infection
24 25 26	Figure 4-2:	Bar graph of prescribing-related decisions for initial presentations of upper respiratory tract infection condition, graphed by upper respiratory tract infection condition
27 28	Figure 4-3:	Bar graph of ordinal choice of antibiotic prescribed for initial diagnoses of upper respiratory tract infection, by condition
29 30 31	Figure 4-4:	Plots of the marginal predicted mean of the outcome of unnecessary antibiotic prescribing occurring, across different upper respiratory tract infection conditions and different patient age groups, by condition
32 33 34	Figure 4-5:	Plot of margins at representative values for the effect on the probability of various levels of ordinal choice of antibiotic prescribed, with change in condition, across different patient age groups, relative to patients with rhinosinusitis
35 36 37 38	Figure 4-6:	Plot of margins at representative values for the effect on the probability of the ordinal choice of antibiotic being prescribed, with change in whether repeats were issued on the prescription from negative to positive, across different patient age groups, relative to prescriptions without repeats issued on them
39 40	Figure 5-1:	Depiction of the main outcome variables utilised in this analysis for initial presentations of urinary tract infection
41 42	Figure 5-2:	Bar graph of ordinal choice / line of antibiotic prescribed for initial presentations of urinary tract infection, by patient group
43 44	Figure 5-3:	Bar graph of counts of ordinal choice of antibiotic prescribed for initial presentations of urinary tract infection, by patient group

1 2 3	Figure 5-4:	Plot of marginal effects at representative values for change in patient gender from female to male, at different patient age groups for model 1 (ordinal line of antibiotic prescribed) for initial presentations of urinary tract infection
4 5 6	Figure 5-5:	Plot of marginal effects at representative values for change in repeat on prescription status from negative to positive by patient age group for model 1 (ordinal line of antibiotic prescribed) for urinary tract infection
7 8 9	Figure 5-6:	Graph of adjusted predictions at the means for model 3 (repeats present on antibiotic prescription) at differing values of patient gender and age group, keeping all other covariates in the model constant at their sample means 133
10 11	Figure 6-1:	Plots for proportions of participating general practices by Western Australian primary health network, January 2012 to June 2017, inclusive, by quarter 139
12 13	Figure 6-2:	Depiction of the main outcome variables utilised in this trends analysis, and previous analyses for initial presentations of upper respiratory tract infection 141
14 15 16	Figure 6-3:	Depiction of the main response variables used in this trends analysis, and previous analyses for, initial presentations of upper respiratory tract infection, continued
17 18	Figure 6-4:	Plot of counts of diagnoses, by upper respiratory tract infection condition, from January 2012 to June 2017, inclusive, by month
19 20	Figure 6-5:	Time series plot for antibiotic prescribing rate among initial presentations of upper respiratory tract infection, January 2012 to June 2017, inclusive, by month 145
21 22 23	Figure 6-6:	Time series plot of prescribing rates for second-line antibiotics for initial presentations of acute rhinosinusitis, January 2012 to June 2017 inclusive, by month
24 25 26	Figure 6-7:	Time series plot of amoxicillin and amoxicillin with clavulanate prescribing rates for initial presentations of acute rhinosinusitis, January 2012 to June 2017, inclusive, by month
27 28 29	Figure 6-8:	Time series plot of antibiotics not recommended in the guidelines for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017 inclusive, by month
30 31 32	Figure 6-9:	Time series plot for phenoxymethylpenicillin and amoxicillin prescribing rates for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017 inclusive, by month
33 34	Figure 6-10:	Time series plot of antibiotic prescribing rate among initial presentations of acute otitis media, January 2012 to June 2017, inclusive, by month
35 36 37	Figure 6-11:	Time series plot for amoxicillin and amoxicillin with clavulanate prescribing for initial presentations of acute otitis media, January 2012 to June 2017, inclusive, by month
38 39	Figure 7-1:	Depiction of the main outcome variables utilised in this trends analysis, and previous analyses for initial presentations of urinary tract infection
40 41 42	Figure 7-2:	Graph of proportions of initial presentations of urinary tract infection, by patient group (women, men, and children under sixteen years) from January 2012 to June 2017, inclusive, by month
43 44 45	Figure 7-3:	Time series plot of second-line antibiotic prescribing for all women sixteen years and over with initial presentations of urinary tract infection, from January 2012 to June 2017, inclusive, by month

1 2	Figure 7-4:	Time series plot of simple moving average prescribing rates for cefalexin and trimethoprim for women, January 2012 to June 2017, inclusive, by month 168
3 4 5	Figure 7-5:	Time series plot of second-line antibiotic prescribing for all men at least sixteen years of age with initial presentations urinary tract infection, from January 2012 to June 2017, inclusive, by month
6 7 8	Figure 7-6:	Time series plot of prescribing for not recommended antibiotic for children under sixteen years of age with initial presentations of urinary tract infection, from January 2012 to June 2017, inclusive, by month
9 10 11	Figure 7-7:	Time series plot of prescribing for antibiotic agents not recommended in the guidelines for children under sixteen years of age with initial presentations of urinary tract infection, from January 2012 to June 2017, inclusive, by month 171
12		
13		

1 LIST OF TABLES

2 3 4	Table 3-1:	Antibiotic classifications for the ordered choice variable for upper respiratory tract infection conditions, based on the order of antibiotics recommended in the Therapeutic Guidelines: Antibiotic (29), by condition
5 6 7	Table 3-2:	Antibiotic classifications for the ordered choice variable for acute cystitis, based on the order of antibiotics recommended in Therapeutic Guidelines: Antibiotic (29), by patient group
8 9 10 11	Table 4-1:	Choice of antibiotic for upper respiratory tract infection conditions, by condition, and by patient allergy label for penicillin, based on the order of recommendations and penicillin hypersensitivity options listed within Therapeutic Guidelines: Antibiotic (29)
12 13	Table 4-2:	Frequency table of patient characteristics for all patients, and patients with initial episodes of care for upper respiratory tract infection (column percentage)95
14 15 16	Table 4-4:	Frequency table of active ingredients prescribed for initial presentations of upper respiratory tract infections (denominator all antibiotics including influenza / influenza-like illness)
17 18 19	Table 4-3:	Frequency table of patient, consultation, and prescription characteristics for antibiotic prescriptions issued at initial episodes of care for upper respiratory tract infection, including by condition (column percentage)
20 21	Table 4-5:	Frequency table of patient age group by necessary / unnecessary antibiotic prescribing for initial presentations of upper respiratory tract infection
22 23 24	Table 4-6:	Frequency table of first-line and non-first-line antibiotic prescriptions for initial presentations of upper respiratory tract infection, by upper respiratory tract infection condition
25 26	Table 4-7:	Frequency table of ordinal choice of antibiotic prescribed, by patient age group, for patients with initial presentations of upper respiratory tract infection
27 28 29	Table 4-8:	Frequency table of unnecessary and necessary antibiotic prescribing for initial presentations of upper respiratory tract infection, by ordinal choice of antibiotic prescribed
30 31 32	Table 4-9:	Frequency table of active ingredients prescribed for initial presentations of URTI with repeats issued, excluding prescriptions for influenza / influenza-like illness.
33 34 35	Table 4-10:	Mixed effects logit regression models of binary inappropriate decisions (Model 0) and binary unnecessary antibiotic prescribing (Model 1) for initial presentations of upper respiratory tract infection
36 37 38	Table 4-11:	Mixed effects logit regression models of choice of antibiotic (Model 2) and binary non-first-line antibiotic prescribing (Model 3) for upper respiratory tract infection
39 40 41	Table 5-1:	Antibiotic classifications for the ordered choice variable for acute cystitis, based on the order of antibiotics recommended in Therapeutic Guidelines: Antibiotic (29), by patient group
42 43	Table 5-4:	Frequency table to all active ingredients for systemic antibiotics prescribed for initial presentations of urinary tract infection
44 45	Table 5-2:	Frequency table of patient characteristics for all patients, and patients with initial episodes of care for urinary tract infection (column percentage)

1 2 3	Table 5-3:	Frequency table of patient, consultations and prescription characteristics for patients prescribed an antibiotic for urinary tract infection (column percentage)
4 5	Table 5-5:	Frequency table of first-line and non-first-line antibiotics prescribed for initial episodes of urinary tract infection, by patient group
6 7	Table 5-6:	Frequency table of antibiotic prescriptions issued with repeats for initial presentations of urinary tract infection, by active ingredient
8 9 10	Table 5-7:	Frequency table of whether repeats were issued on antibiotic prescriptions (negative or positive) for initial presentations of urinary tract infection, by patient age group
11	Table 5-8:	Mixed effects models for initial presentations of urinary tract infection
12 13	Table 6-1:	Choice of antibiotic for upper respiratory tract infection conditions, by condition, and by allergy label, as per Therapeutic Guidelines: Antibiotic (29)
14 15	Table 6-2:	Prescribing outcomes for patients with initial presentations of upper respiratory tract infection, by condition group153
16 17 18	Table 6-3:	Individual antibiotic agents prescribed for upper respiratory tract infection conditions: two antibiotics with the highest magnitude of statistically significant change
19 20	Table 7-1:	Summary of Therapeutic Guideline: Antibiotic (29) recommendations in order of choice for acute cystitis, by patient group
21 22	Table 7-2:	Prescribing outcomes for each patient group with initial presentations of urinary tract infection, by patient group
23 24	Table 7-3:	Individual antibiotic agents prescribed for urinary tract infection conditions: three antibiotics with the highest magnitude of statistically significant change
25		

1 LIST OF ACRONYMS

2

3

- AOM Acute otitis media AOR Adjusted odds ratio
- 5 AMR Antimicrobial resistance
- 6 ARF Acute rheumatic fever
- 7 AME Average marginal effects
- 8 AURA Antimicrobial Use and Resistance Australia
- 9 CALD Culturally and linguistically diverse
- 10 CIs Confidence intervals
- 11 CME Continuing medical education
- 12 GPs General practitioners
- 13 ICC Intra-class correlation
- 14 ID Identification number
- 15 ILI Influenza-like Illness
- 16 MERs Marginal effects at representative values
- 17 NAPS National Antibiotic Prescribing Survey
- 18 NCAS National Centre for Antimicrobial Stewardship
- 19 NAUSP National Antimicrobial Utilisation Surveillance Program
- 20 NPS National Prescribing Service
- 21 PBAC Pharmaceutical Benefits Advisory Committee
- 22 PBS Pharmaceutical Benefits Scheme
- 23 PHN Primary health network
- 24 RHD Rheumatic heart disease
- 25 RPBS Repatriation Pharmaceutical Benefits Scheme
- 26 RTI Respiratory tract infection
- 27 SEIFA Socioeconomic index for area
- 28 SES Socioeconomic status
- 29 UK United Kingdom
- 30 URTI Upper respiratory tract infection
- 31 US United States
- 32 UTI Urinary tract infection
- 33 WA Western Australia, state of Australia
- 34 WAPHA WA Primary Health Alliance
- 35 WHO World Health Organization

GLOSSARY OF MODEL OUTCOMES AND VARIABLES DEFINITIONS

3

4 This glossary provides a brief summary of outcomes and variables used in the statistical 5 models, and is designed only for reference during the analyses chapters after having read 6 the Methods chapter in detail. This glossary does NOT suffice for reading the Methods 7 chapter in full.

8

9 Allergy label – the recorded history of a patient's sensitivity to a medication in the patient 10 record. In the analyses of this research, an allergy label refers to the antibiotic penicillin or 11 other antibiotics in the penicillin group. This is particularly relevant to the classification of 12 first-line antibiotics for UTI.

13

Appropriate decision * – *likely* appropriate decision (URTI only) : contains appropriate
 non-prescribing and necessary prescribing combined together within the denominator of the
 Inappropriate *versus* Appropriate Decision Model.

17

Appropriate non-prescribing * – *likely* appropriate non-prescribing (URTI only) – not
 prescribing an antibiotic for the patient as an antibiotic is not clinically indicated or required
 for the condition.

21

22 **Choice of antibiotic prescribed model** – ordinal, increasing level of recommendations for 23 specific antibiotics to be prescribed to treat an infection: starting at first-line, second-line, 24 third-line (where relevant), last resort, and not recommended antibiotics. Third-line and last 25 resort options are relevant to UTI only, and both options were combined into a single level 26 for modelling purposes. The denominator was all antibiotics prescribed for that condition 27 group (URTI or UTI).

28

First-line antibiotic – the recommended choice of antibiotic listed in the guidelines for the relevant condition as the first option to try for the patient. This should be chosen to prescribe at initial presentations of infection when prescribing an antibiotic. Note that patients with an allergy label for penicillin specifically, or other penicillins, who were prescribed a suitable penicillin sensitivity option for UTI according to the guidelines was also considered first-line prescribing. First-line antibiotics were included in the denominator of the Choice of antibiotic prescribed model and the Non-first-line antibiotic prescribing model for URTI and UTI. 1

2 Inappropriate versus Appropriate Decision Model * (URTI only):

Inappropriate decision * – *likely* inappropriate decision (URTI only) – this includes unnecessary prescribing. The inappropriate decision model examined inappropriate decisions as the outcome versus appropriate decisions as the base / reference. As such, the numerator was inappropriate decisions, which is the same as unnecessary prescriptions in the Unnecessary *versus* Necessary antibiotic prescribing model but a different denominator. The denominator was all diagnoses of URTI including influenza / influenzalike illness for initial presentations.

10

Last resort antibiotic – the last choice in the list of ordered recommendations for what antibiotic to prescribe for a patient (UTI only). Last resort options should ideally be tried at the fourth consultation if the first- to third-line options have been tried in that order and not been effective, and should have culture and sensitivity testing performed prior to prescribing one. Last resort antibiotics were included in the denominator of the Choice of antibiotic prescribed model and the Non-first-line antibiotic prescribing model for UTI.

17

Necessary prescribing * – *likely* necessary prescribing (URTI only) – prescribing an antibiotic in accordance with the recommendations in the guidelines for the particular URTI condition diagnosed. Necessary prescribing was included in the denominator of the Unnecessary *versus* Necessary antibiotic prescribing model for URTI excluding influenza / ILI.

23

Not recommended antibiotic – the prescribing of an antibiotic not listed anywhere in the guidelines of suggested antibiotics to prescribe for that condition. Not recommended antibiotics were included in the denominator of the Choice of antibiotic prescribed model and the Non-first-line antibiotic prescribing model for URTI excluding influenza / ILI as well as for UTI.

29

30 Non-first-line prescribing model:

Non-first-line prescribing – prescribing of an antibiotic other than the first-line recommended choice of antibiotic listed in the guidelines for the relevant condition. The model examined non-first-line antibiotics versus first-line antibiotics within the condition group (URTI excluding influenza / ILI or UTI). The denominator was all antibiotics prescribed for that condition group (URTI excluding influenza / ILI or UTI). **Repeat negative prescribing** – repeat negative antibiotic prescribing – prescriptions issued without repeats on the prescription. Repeat negative prescriptions were included in the denominator of the Repeat prescribing model for URTI excluding influenza / ILI as well as for UTI.

5

6 **Repeat prescribing model:**

Repeat (positive) prescribing – repeat positive antibiotic prescribing – prescriptions issued
 with one or more repeats present on the prescription, prescriptions positive for repeats on
 the prescription. The numerator in this model was repeat positive prescriptions for the
 condition group (URTI excluding influenza / ILI or UTI). The denominator was all antibiotics
 prescribed for that condition group (URTI including excluding influenza / ILI or UTI).

12

Second-line antibiotic – the second choice in the list of ordered recommendations for what antibiotic to prescribe for a patient. This should ideally be tried at the second consultation if the first-line option antibiotic has been tried initially but was not effective. Second-line antibiotics were included in the denominator of the Choice of antibiotic prescribed model and the Non-first-line antibiotic prescribing model for URTI excluding influenza / ILI and UTI.

18

Third-line antibiotic– the third choice in the list of ordered recommendations for what antibiotic to prescribe for a patient (UTI only). Third-line options should ideally be tried at the third consultation if the first- and second-line options have been tried in that order and have not been effective. Third-line antibiotics were included in the denominator of the Choice of antibiotic prescribed model and the Non-first-line antibiotic prescribing model for UTI.

24

25 Unnecessary versus Necessary antibiotic prescribing model * (URTI only):

Unnecessary prescribing * – *likely* unnecessary prescribing (URTI only) – the prescribing of an antibiotic which was unlikely to have been required for the URTI condition. The numerator in this model was unnecessary prescribing for URTI excluding influenza / ILI. Note that unnecessary prescribing constitutes the numerator in both this model and the Inappropriate *versus* Appropriate Decision Model but has a different denominator. The denominator in this Unnecessary *versus* Necessary antibiotic prescribing model was all antibiotics prescribed for initial presentations of URTI excluding influenza / ILI.

33

* Note: this variable / outcome is considered to have "likely" as a prefix, as the guidelines cannot
 cover every situation and there are situations in which it may be appropriate or necessary to
 prescribe an antibiotic which may not be possible to identify from the data available

GLOSSARY OF MULTILEVEL MIXED-EFFECTS MODELING TERMINOLOGY

3

This glossary provides a brief description of statistical terms utilised in the mixed-effects models, and is designed for reference in conjunction with Methods chapter, and the analyses chapters. It does NOT suffice for reading the Methods chapter in full. This glossary provides only the briefest summary of each term, and I refer you to the references for further information. This glossary borrows frequently from STATA's Multilevel Mixed-effects Reference Manual (Release 17) Glossary (1).

10

Average Marginal Effects (AMEs) - a method of interpretation of a parametric model, and 11 the simplest method of estimating effect of each variable in the model. This method is easy 12 to both interpret and summarise in order to convey to others, however, the method can be 13 deceptive as it, by definition, averages out the effect of each variable rather than provide the 14 exact effect for each variable at its full range of values (1-4). It does not allow for the fact 15 that effects of a variable on the outcome can in fact vary by other characteristics of the 16 17 individual member of a level in the model (for example, at the patient, GP / provider or practice level) (1-4). An AME is the average change in the outcome variable for each one 18 19 point change in the values of other independent variables in the model.

20

21 Adjusted Predictions at the Means – a method of interpretation of a parametric model, which calculates the effect of a variable in the model on the predicted value of the outcome 22 when all other variables are held constant at sample means (1-4). This method does not 23 show the full extent of how the effects of a variable in the model may change depending 24 upon other characteristics of the individual observation or other characteristics of the 25 individual member of a level in the model (for example, at the patient, GP / provider or 26 practice level). This is the predicted value of the outcome variable at the average values of 27 the explanatory variables in the model. 28

29

Fixed-effects model – a model which considers all variables to be constant (i.e. fixed) across all individual members within each level of a model: either not changing over time, or changing at a constant rate over time (1,5,6). (Note that many, seemingly constant, variables may in fact change over time, or even potentially change at a constant rate over time.) A fixed-effects model considers any random variables (and random-effects) to be fixed or non-random, and does not control for change over time (1,5,6). While one might potentially consider using dummy variables to compensate for this issue within a fixedeffects model, the addition of numerous dummy variables can also negatively impact upon
a model's accuracy (5,6).

4

Generalised linear mixed-effects model – a extended version of the generalised linear
model allowing for the inclusion of random (and therefore mixed) effects.

7

Marginal effects (margins) – partial (i.e. marginal) derivatives of the regression equation 8 for each variable in the model and each unit in the data (1,2). They measure the impact 9 10 (incremental change) that an instantaneous change, in the unit of one variable in the model, has on the independent / outcome variable, while all other variables are held constant. This 11 can also involve calculating the impact a unit change in one variable in the model has on 12 the independent / outcome variable, as well as how the effect on the outcome variables 13 changes across different values of a second variable in the model (1,2). There are many 14 types of marginal effects, the most relevant of which include: average marginal effects, 15 adjusted predictions at the means, and margins at representative values (1,5,6). Margins is 16 an abbreviation for marginal effects, and is the term commonly used for the function in 17 18 statistical packages which provides the marginal effects for a regression equation.

19

20 Marginal Predicted Mean – a method of interpretation of a parametric model, which calculates the average of the predicted value of the outcome, when other independent 21 variables are held at specified different values or levels (1-4). This method does not show 22 the full extent of how the effects of a variable in the model may change the outcome 23 24 depending upon other characteristics of the individual observation or other characteristics of the individual member of a level in the model (for example, at the patient, GP / provider 25 or practice level). This method predicts the average probability of the outcome occurring at 26 each of the specified values / levels of the explanatory variables in the model. 27

28

Margins at Representative Values (MERs) – a method of interpretation of a parametric model, which demonstrates how the value of the dependent/outcome variable changes in response to changes in the values of the independent variables in the model. It is considered a superior method of interpretation, particularly in complex models with interaction effects or hierarchical models where the coefficients themselves can be difficult to interpret. MERs are interpreted as the difference in the outcome (versus the comparator) at the defined range of values of the explanatory variables.

- 1
- 2 **Mixed-effects model** a model including both fixed and random effects (1,5,6).
- 3

Random coefficient – in a mixed-effects model, a random intercept can be considered to
be equivalent to the slope in a fixed-effects equation (1). It can therefore be regarded as a
randomly varying slope at the relevant level with the multilevel model.

7

Random intercept – in a mixed-effects model, a random intercept can be considered to be
equivalent to the intercept in a fixed-effects equation (1). It can therefore be regarded as a
randomly-varying intercept at the relevant level within the multilevel model.

11

Random-effects model – a model which allows for unobserved heterogeneity, thereby also controlling for variables which are either: non-constant across individuals or members of any level, or which change over time (1,5,6). Random effects are effects which can vary between members of any level within the multilevel model.

16

Residual variance – a term used in this research project to represent the remaining variance unexplained by the fixed-effects in a mixed-effects model. Calculating the intraclass correlation between the second and third levels in a three-level model provides the value of the residual variance for each of the two, higher levels.

21

Unobserved heterogeneity – where there may be unmeasured differences between 22 members in the level of a model, in the form of an unmeasured / unobserved (but relevant) 23 24 variable (1). This can occur when there is an unmeasured/unobserved variable, which is in fact related to the measured / observed variables within the study. For example, an 25 unobserved/unmeasured variable might be correlated with an observed / measured 26 (independent and / or dependent) variable. Statistical inferences may not be valid or correct 27 if unobserved heterogeneity is present but not allowed for in a model. One way of ensuring 28 that statistical inferences are valid and correct, in the presence of unobserved 29 heterogeneity, is to use multilevel models with mixed effects. 30

- 31
- 32

1 CHAPTER 1 INTRODUCTION

2

3 1.1 The problem

4

5 Antimicrobial resistance (AMR) has been described as one of the most concerning 6 threats to humanity (5-7), with global deaths from antibiotic resistant bacteria being 7 estimated at 700,000 annually in 2016, and predicted to rise to 10 million in 2050 (8). 8 Antibiotic use promotes the evolutionary process of bacteria developing antibiotic 9 resistance (9-11).

10

Systemic antibiotic use is high in Australia when compared to other first-world countries 11 (12-14). It is also known to be high in the Australian community setting (12,14-18). In 12 2014, 46% of the Australian population was dispensed one or more systemic 13 antimicrobial in the community, and general practitioners (GPs) were responsible for 14 prescribing 88% of these (12). By 2019, dispensing in the community had dropped to 15 40%, however, of these 50% were issued with at least one repeat on the prescription 16 (18). While the majority of antibiotics are prescribed in community settings, rather than 17 hospitals, antibiotic stewardship and surveillance of antibiotic prescribing and 18 dispensing in Australia still predominantly focuses on hospital settings (19-21). 19

20

Antibiotic prescribing is termed 'inappropriate' when it is not in accordance with local 21 prescribing guidelines (22-27). This is internationally accepted terminology and sets the 22 benchmark against which prescribing is assessed (22-25,28). Guidelines are developed 23 to minimise antibiotic resistance and side effects from antibiotics by recommending 24 effective yet conservative use to preserve their efficacy (29). Australia's national 25 26 guidelines for antibiotic prescribing are titled the Therapeutic Guidelines: Antibiotic and will be referred to as the guidelines (29). These define the circumstances in which it is 27 appropriate to prescribe a specific antibiotic for a particular condition or diagnosis (29). 28 29

Inappropriate antibiotic prescribing likely occurs in all Australian health settings (15,20,29-32). However, there are also notable difficulties with obtaining reliable, largescale electronic patient data from Australian general practice (33), by which to investigate this. It is possible though, to collect data on antibiotic prescriptions yielding data about volumes and trends of antibiotic use.

However, importantly there is currently no substantial body of quantitative research about factors associated with inappropriate antibiotic prescribing for upper respiratory tract infection (URTI) and urinary tract infection (UTI) using patient data from Australian general practice, nor from the state of Western Australia (WA). Research in this field, such as this study, will contribute to the growing knowledge base internationally, and frame any issues for clinicians, system managers and policymakers.

7

8 The current body of research regarding the predictors of inappropriate antibiotic 9 prescribing in general practice is limited, particularly in Australia. This is distinct to 10 research regarding predictors of antibiotic prescribing generally. A deeper 11 understanding of the drivers of the inappropriate prescribing of systemic antibiotics is 12 vital to inform antimicrobial stewardship policy and practice in Australian primary care.

13

14 **1.2** Hypothesis

15

The hypothesis was that there is substantial inappropriate antibiotic prescribing 16 occurring for these conditions within WA general practice. It also included the 17 proposition that there were likely to be patient, consultation, provider and practice-18 related predictors of inappropriate antibiotic prescribing. It is possible that patients with 19 20 comorbid conditions, living in areas with varying measures of rurality or remoteness and accessibility to health care, with different SES, and of different ages, may be 21 22 predisposed to receiving inappropriate antibiotic prescriptions when they technically 23 should not. This might feasibly occur in response to concerns their treating GPs had about these patients' welfare and/or access to healthcare. 24

25

An Australian, national report from 2015 had found that antibiotic dispensing rates in the 26 community were the highest in areas with the lowest SES, and dispensing rates 27 decreased with increasing status (15). Another national report published in 2016 (12) 28 had found that patients in major cities had the highest rates of prescribing of systemic 29 antibiotics compared to those living in other areas (34,35). This publication (12) also 30 found that patients living in the second-most socioeconomically disadvantaged quintile 31 areas had the lowest antibiotic prescribing rates (36,37). One study of patients with sore 32 throat presenting to general practice had found that chronically unwell patients were 33 more likely to receive antibiotics (38), while another study of patients with bronchitis 34 35 found that patient comorbid condition status may be linked to the choice of specific antibiotic patients were prescribed (39). A national survey of prescribing in hospitals had 36

also found that patients with comorbid conditions including chronic obstructive 1 pulmonary disease (COPD) and asthma were more likely to receive antibiotic 2 prescriptions (12,40). Subsequent to the commencement of this research, Bernardo et 3 al. (41) had gueried whether patient SES or comorbid conditions might affect antibiotic 4 5 or antiviral prescribing for influenza-like illness (ILI) in general practice, on the basis of existing literature. 6 7 1.3 **Objectives** 8 9 The aim of this study was to define and identify, predictors of, guideline non-conforming 10 antibiotic prescribing for initial presentations of acute URTI and UTI. 11 12 This research had included the following objectives: 13 14 1. to define and quantify the levels of inappropriate prescribing of systemic 15 16 antibiotics for initial presentations of common infections to WA general practice using large-scale, routinely collected electronic patient data 17 2. to use quantitative methods to ascertain predictors of inappropriate prescribing 18 of systemic antibiotics in WA general practice, including patient-, provider-, 19 consultation- and practice-related factors associated with such prescribing 20 3. to determine trends in inappropriate antibiotic prescribing and in overall antibiotic 21 prescribing by WA GPs over time. 22 23 The presenting condition groups of interest were acute, uncomplicated URTI, including 24 acute rhinosinusitis / non-specific URTI, acute pharyngitis and / or tonsillitis, acute otitis 25 media (AOM), and influenza / ILI, as well as UTI limited to the condition of acute cystitis 26 (29). 27 28 29 Large-scale, longitudinal datasets were obtained from general practice in WA. A list of conditions of interest was developed from the data received. The reference point was 30 the recommendations contained within the guidelines for each condition, the treatment 31 (29), from which algorithms were developed to identify inappropriate prescribing. Data 32 from patient records of initial consultations were utilised to assess the indication for 33 prescribing and the appropriateness of antibiotic choice or selection when an antibiotic 34

- 35 was prescribed.
- 36

The outcomes were then analysed using quantitative methods, specifically, mixedeffects logistic regression to elucidate patient- and practice-related factors associated with inappropriate prescribing. Relating back to the hypothesis, potential predictors of interest included patient age, gender, SES, comorbid conditions, as well as practice rurality / remoteness. Aggregate trends over time in inappropriate antibiotic prescribing and overall antibiotic prescribing were also examined.

7

8 This research was supported with funding provided by the WA Primary Health Alliance 9 (WAPHA), which is the government-funded commissioning body with the responsibility 10 for delivering all three primary health networks (PHNs) across the state (42). Expert 11 advice was also obtained from GPs currently practising in WA to help guide this project 12 and aid in the interpretation of results.

13

14 **1.4 Significance**

15

Antibiotic resistance is a growing global concern of critical importance (16,43,44). There is currently very limited empirical information regarding antibiotic prescribing in Australian general practice, particularly with respect to drivers of inappropriate prescribing. This thesis fills this important gap in the knowledge base. While there is research regarding drivers of prescribing generally, there is very limited research of such for inappropriate prescribing, which must be reduced to extend the effectiveness of current antibiotics (15).

23

This thesis supports antimicrobial stewardship programs in primary care, and adds to 24 the work of the Antimicrobial Use and Resistance Australia (AURA) project, the national 25 body doing work on antibiotic prescribing using large-scale data from community care 26 (12,14,17,18). AURA reports provide quantities of inappropriate prescriptions for various 27 conditions, however, they state that all prescribing for likely viral, respiratory conditions 28 is inappropriate and do not differentiate appropriate from inappropriate prescribing for 29 these conditions (12,14,17,18). Furthermore, these publications do not differentiate 30 initial from non-initial consultations, to ascertain the quality of antibiotic prescribing and 31 clinical management occurring at initial consultations (12,14,17,18). This thesis defines 32 inappropriate and appropriate prescribing using more data than the diagnostic condition 33 alone and limits the diagnoses included to initial consultations only, thereby facilitating 34 35 enhanced analysis.

Research in this field is an urgent necessity borne from increasing antibiotic resistance globally (5-7). This project creates the potential for new knowledge regarding antibiotic prescribing in the community and primary care, with potential impact both nationally and internationally. The findings will inform policy and practice, and lead to further research on antibiotic prescribing in primary care. Its applications include informing the development of evidence-based interventions to reduce inappropriate antibiotic prescribing to stem the emergence of resistance.

8

9 To the best of my knowledge, at the time of writing, this is the first Australian research 10 using quantitative methods and large-scale empirical data to identify predictors of 11 inappropriate prescribing in general practice for UTI and URTI, using more clinical 12 information than the condition diagnosed to differentiate inappropriate from appropriate 13 prescribing. Furthermore, this is believed to be the first such analysis in Australia to 14 allow for unobserved heterogeneity and also to limit analyses to initial presentations of 15 infection.

16

17 **1.5** Thesis structure

18 The overarching structure and focus of the thesis are outlined below.

The next section of this thesis, Chapter 2, outlines the problem of antibiotic resistance 19 and how inappropriate antibiotic prescribing in primary care unnecessarily promotes the 20 development of antibiotic resistance. It explains the resulting need for research to help 21 identify what is driving inappropriate antibiotic prescribing in primary care and outlines 22 23 the project objectives. It covers the development of antibiotic resistance, the use of antibiotics in human health, and the need for the preservation of antibiotic effectiveness 24 by limiting any unnecessary antibiotic usage. The chapter also provides a summary of 25 URTI and UTI conditions, and details some of the complexities regarding antibiotic 26 27 prescribing for these conditions. The guidelines for treatment of URTI and UTI conditions are also discussed (29). Antibiotic stewardship and the surveillance of 28 antibiotic use are briefly summarised for the Australian setting, including what is known 29 about inappropriate antibiotic prescribing for URTI and UTI in Australian general 30 practice. An overview of the published literature on predictors of inappropriate antibiotic 31 prescribing identified using quantitative methods is presented. Brief summaries of 32 gualitative research and interventions designed to improve prescribing practices are 33 also provided. 34

Chapter 3 outlines the data source, methodological approaches, data cleaning and
preparation. It explains how the guidelines were used to define multiple levels, or
aspects, of inappropriate antibiotic prescribing for both URTI and UTI conditions (29).

5

1

6 Chapter 4 contains the results of the analyses of predictors of inappropriate antibiotic prescribing for URTI. Descriptive results for the various definitions of inappropriate 7 antibiotic prescribing for the multiple URTI conditions: acute rhinosinusitis, acute 8 pharyngitis and or tonsillitis, AOM, and influenza / ILI, are presented, followed by the 9 regression results for the mixed-effects logistic models (generalised linear mixed 10 modelling with random effects). The majority of prescriptions were found to be 11 unnecessary, and over 50% of antibiotics prescribed were non-first-line when first-line 12 antibiotics should have been used for these initial presentations of URTI. 13

14

15 Chapter 5 details the results of the analyses of predictors for inappropriate antibiotic 16 prescribing for initial presentations of the condition of acute cystitis / UTI, with a focus 17 on patient groups (women, men, children). The findings include substantial proportions 18 of non-first-line antibiotics being prescribed to patients with initial presentations of UTI, 19 most notably for children.

20

Chapter 6 provides a brief summary of the dataset from the perspective of analysis over 21 22 time. Chapter 7 contains the results of the analyses of trends in prescribing for URTI over time. These were undertaken to identify any significant trends in prescribing over 23 24 time, utilizing the same outcomes and variables defined and used in Chapter 4. There were significant downward trends in unnecessary prescribing over time for rhinosinusitis 25 26 and pharyngitis but not AOM. While there was a significant decreasing trend in non-firstline prescribing over time for pharyngitis, upward trends were found for rhinosinusitis 27 and AOM. 28

29

Chapter 8 presents the results of analyses for trends in prescribing for UTI, using response variables detailed in Chapter 5. This includes the presentation of trends in outcomes over time for all patients with UTI altogether, then for women, men and children separately. The results summarised also include trends in overall antibiotic prescribing over time for each patient group. There were upward trends in non-first-line prescribing identified for adult patient groups but there was a downward trend for
 children.

3

The Discussion chapter follows in Chapter 9, summarising the most pertinent findings 4 5 from the four analysis chapters. The results are further explained, and the findings are compared with those of the broader literature. This provides context regarding what the 6 7 findings mean for WA general practice, as well from the broader perspective. It provides 8 an overview regarding the substantial inappropriate antibiotic prescribing found for initial presentations of URTI and UTI. It summarises the most important predictors of 9 inappropriate prescribing identified, as well as the trends over time in inappropriate 10 prescribing. This is followed by discussion of the context and implications of these 11 results for general practice policy and practice. In the Australian setting, although 12 evidence of some progress is noted, the overall situation appears concerning. Priority 13 areas for antibiotic stewardship and opportunities for improvement are identified, and 14 potential areas for further research are raised. The limitations of the research are also 15 16 discussed. It raises questions regarding the need for more proactive steps on this important topic and opportunities for future research. 17

18

19 Chapter 10 provides a brief overview of the pertinent findings and implications for 20 Australian general practice in the context of inappropriate prescribing for URTI and UTI. 21 It notes the limited progress identified on inappropriate prescribing in Australian general 22 practice appears insufficient to have notable impact on curbing antibiotic resistance. It 23 also highlights the value and significance of this research and what it contributes to the 24 knowledge base both at the national and international levels.

25

26

1

-

2 3

CHAPTER 2 BACKGROUND

- 4 2.1 Introduction
- 5
- 6 7

2.1.1 The problem: growing, global antibiotic resistance

Antibiotic resistance is becoming an increasing problem globally, raising concerns about future capacity to control infections with antibiotics (6,16,44). Antibiotic resistance is encouraged by the frequent use of antibiotics in human health care, veterinary medicine, as well as agriculture, and exposure to antibiotics in the environment can also facilitate resistance (9,11). There is a strong association between antibiotic use and antibiotic resistance in individuals (45,46). An apt description of the situation is a rapidly progressing yet "silent pandemic" (47).

15

16 17

2.1.2 The problem locally: inappropriate antibiotic prescribing in Australian primary care

18

Australia's First National Antimicrobial Resistance Strategy 2015–2019 (6,16) aims to 19 20 develop and implement strategies to prevent and minimise growing AMR, whilst ensuring continuing effectiveness and availability of treatments for infectious diseases 21 22 (6,16). Inappropriate antibiotic prescribing is thought to occur in all Australian health settings (15,20,26-28,30-32,48). It is believed to occur frequently in the community, of 23 which general practice constitutes the majority of the patient interactions (15,20,26-24 28,30-32,48). Despite the existing quantitative research on antibiotic prescribing, 25 resistance and its current surveillance within the Australian health system (14,17), there 26 is limited surveillance or empirical information regarding how much of the prescribing in 27 primary care is inappropriate for various conditions (49-51). Furthermore, there is no 28 substantial body of quantitative research analysing large general practice data sets to 29 identify factors associated with inappropriate antibiotic prescribing. There is qualitative 30 research on inappropriate prescribing, and although not the focus of this research, a 31 summary of qualitative studies is provided in Section 2.6.2 for reference. 32

33

This thesis focuses on the state of WA, which poses unique challenges for health services due to its very large geographical area and small population density by area (52). It is also characterised by a proportionally high number of Aboriginal and Torres Strait Islander peoples living in remote communities (52,53). People living in these rural
and remote communities have high rates of infectious disease incidence, chronic
disease prevalence, and poor health outcomes (52-58), as well as high rates of bacterial
disease that should be treated by antibiotics.

- 5
- 6
- 7

8

2.2 Antibiotics, the development of antibiotic resistance and other side effects from taking antibiotics

9 10

2.2.1 Antibiotics and antimicrobials

Medically important microorganisms include bacteria, fungi, viruses and parasites, 11 which are treated with antimicrobial agents: antibiotic, antifungal, antiviral and 12 antiparasitic medications, respectively (59). Antibiotics typically treat bacterial 13 14 infections, although there is some antibiotic use to treat non-bacterial pathogens (59). Antibiotic agents are ineffective against viruses. This is relevant to diagnosing 15 respiratory tract infections (RTIs) with multiple aetiologies. Viruses are commonly 16 cleared by the human body without specific antiviral treatment. Antibiotics have either 17 systemic or topical mode of delivery. This research addresses systemic antibiotic 18 prescribing. 19

20

Antibiotic use is distinct to prescribing, as many prescriptions written are not dispensed 21 (60-64). However, prescribing is a necessary precursor to dispensing. Quantities of 22 antibiotics dispensed are often reported as a proxy for prescribing. Antibiotic use can 23 be classified into multiple different types. The most common antibiotic use is empirical 24 prescribing, which is immediate treatment of a patient presenting with signs and 25 symptoms of existing infection, if antibiotic use is indicated (65-67). There are also 26 27 prophylactic uses for antibiotics, which prevent infection in the first place (66,67). Prophylaxis is common and recommended for chronic or recurrent infections or for 28 patients with compromised immune systems (66,67). Definitive / directed therapy refers 29 30 to antibiotic treatment following laboratory diagnosis of the causative pathogen, guiding the choice of an effective antibiotic (66,67). Delayed prescribing is a dispensing 31 minimization strategy, where the patient is given a prescription but told to hold off filling 32 33 it, and to wait and only fill the script several days later if the condition does not improve (68,69). 34

This research addresses the clinical decision whether to prescribe at initial 1 presentations and what empirical prescribing occurs in these situations. A distinct 2 weakness of empirical therapy is the absence of laboratory pathology results which 3 guide effective antibiotic choice. The time (typically several days), for a pathology test 4 to be performed and for the result to be available to the clinician, often requires that the 5 clinician consider empirical prescribing initially without pathology results, if an antibiotic 6 is indicated. With the development of rapid testing techniques, guicker access to 7 pathology results is foreseeable in the future (70,71). 8

- 9
- 10

2.2.2 Antibiotic resistance concepts and side effects

11

Antibiotic resistance is encouraged by the frequent use of antibiotics in human health care (72,73). There is increasing risk of bacterial resistance with increasing duration and course number of antibiotics (45). In addition to promoting antibiotic resistance at the community (72,74) and individual level (45,46), taking antibiotics can lead to other serious side effects for individual patients (75-81). Antibiotic stewardship is the promotion of both awareness regarding growing antibiotic resistance and the careful use of antibiotics (82,83).

19

Clinicians must know the local epidemiology of infectious diseases, common pathogens, as well as their susceptibilities to available antibiotics, as well as local resistance patterns to make an informed decision including selecting an appropriate antibiotic (29,67,84). Any decision to prescribe an antibiotic must consider individual patient benefit against the potential harms of prescribing including side effects, antibiotic resistance, and the potentially serious sequelae of not prescribing (29,67).

26

27 2.3 Conditions commonly suspected of receiving inappropriate antibiotic 28 prescribing

29

In Australia, epidemiological data on infection incidence, treatment efficacy and local resistance patterns are used by experts to develop therapeutic guidelines and treatment benchmarks to help guide clinicians. Despite this, antibiotic prescribing for URTI conditions occurs more often than the epidemiological data and benchmarks suggest it is required (12,14,17). There are also large quantities of non-first-line agents prescribed for both URTI and UTI conditions when the majority of patient interactions are believed to be initial presentations (12,14,17).
1

In different healthcare settings, there may be more than one set of guidelines available
to guide clinicians. For example, in the hospital setting, there may be specialist
paediatric guidelines which may address the multitude of factors influencing prescribing
and therefore appropriateness.

6

7 The standard for assessment in this thesis was version 15 of Australia's Therapeutic 8 Guidelines: Antibiotic (the guidelines), published in 2014, to reflect the timeframe of the data analysed (29). Narrow-spectrum antibiotics are recommended as the initial, first-9 line choice for treatment, as they are less predisposed to promoting antibiotic resistance 10 than non-first-line agents (29). Resistance should be confirmed using sensitivity and 11 susceptibility testing prior to using non-first-line antibiotics (29), however, note this takes 12 several days. A brief description including an outline of the guidelines for each condition 13 follows below. Please see the Methods chapter for details of how the guidelines were 14 applied for each condition (29). For a summary of the prescribing guidelines published 15 directly before and after the version used for analysis (85,86), please see Appendix A.1. 16 17

Note that prescribing can be appropriate despite not meeting guideline compliance, as there are many factors influencing antibiotic use, which cannot cover all scenarios. For example, patients living in remote areas may not have regular access to healthcare, such that issuing a prescription at an initial consultation may be reasonable.

22

23 2.3.1 Upper respiratory tract infection

24

URTI is the term given to a group of conditions affecting the upper regions of the 25 respiratory tract, and these are most commonly viral in origin (29,87). This thesis defines 26 URTI to include rhinosinusitis / common colds, influenza and ILI, pharyngitis and / or 27 tonsillitis and AOM. GPs in all countries commonly see patients presenting with URTIs 28 (88-90), which are difficult to diagnose correctly due to the multitude of possible 29 aetiologies, despite most still being viral (91,92). URTI was the reason for 26% of GP 30 presentations by Australian children (29) and the most frequent reason for presentation 31 in infants (93). Most URTIs, regardless of aetiology, are self-limiting and full recovery is 32 usual without treatment (29). 33

1 2.3.1.1 Acute rhinosinusitis / non-specific URTI

Acute rhinosinusitis is marked by inflamed paranasal sinuses and nasal passages 2 (94,95). It is almost always a viral infection, although it can be bacterial (94,96). 3 Approximately 20% of bacterial rhinosinusitis which does not resolve within two weeks 4 5 may require antibiotic treatment (29). The common cold and non-specific URTI have been included in this condition. The guidelines used for analysis in this study suggest 6 symptomatic treatment and recommend against the routine use of antibiotics for 7 rhinosinusitis. Antibiotics can be considered in patients with symptoms over seven days, 8 high fever in excess of three days from symptom onset, or double-sickening - a term 9 describing symptoms which worsen after several days of initial, milder illness (29,97). 10 11

12 2.3.1.2 Acute pharyngitis / tonsillitis

Pharyngitis and / or tonsillitis are infections characterised by sore throat, and the strong 13 majority of presentations are viral (98,99). The only indication for antibiotic treatment of 14 pharyngitis / tonsillitis is Group A Streptococcus bacterial infection (29,57). This is an 15 absolute indication for antibiotic treatment due to potentially fatal complications, 16 including acute rheumatic fever (ARF) (57). Group A Streptococcus causes between 17 18 20-30% of paediatric, and 5-15% of adult, cases of pharyngitis (100-102). Australia has a very high rate of ARF and rheumatic heart disease (RHD) in remote, Indigenous 19 communities in central and northern Australia (56,57,103). Patients 2-25 years of age 20 in communities with high incidence of ARF, and / or any patient with current RHD or 21 22 scarlet fever are high-risk patients requiring antibiotics (29,57).

23

24 2.3.1.3 Acute otitis media

AOM is inflammation of the middle ear, caused by viral or bacterial infection (29). It often 25 presents in young children with ear pain and fever, and complications include tympanic 26 membrane perforation and consequent conductive deafness (104,105). As there is high 27 incidence among Aboriginal and Torres Strait Islander children, these children are 28 29 considered high risk (29,106,107). The guidelines used for analysis in this study state 30 that prescribing is appropriate for children with systemic features including vomiting, lethargy or high fever (29). Symptomatic treatment without antibiotics is recommended 31 initially for children aged six months or more without systemic features. For children 32 under six months without systemic features, antibiotic prescribing may be appropriate, 33 however, symptomatic treatment may be sufficient initially but review of the patient is 34 recommended after 24 hours. 35

1

2 2.3.1.4 Influenza and influenza-like illness

Influenza is infection with one of several influenza viruses which are typically seasonal 3 (41,108,109). The infection is most often moderate in severity but can be fatal in young 4 5 children, the elderly and people with chronic diseases (108,110). Estimated deaths from annual influenza epidemics exceed 500,000 people per year globally (108,111). ILI is a 6 diagnosis based on a set of symptoms including fever, lethargy and cough (109,110). 7 ILI can be caused by influenza viruses, as well as parainfluenza viruses, adenoviruses, 8 respiratory syncytial virus (109). These viruses can also cause lower respiratory tract 9 infection, which is out of scope for this thesis. The guidelines used for analysis 10 recommend symptomatic treatment for influenza / ILI, as these are always viral in 11 aetiology (112), and antibiotics are not recommended unless secondary bacterial 12 complications are noted (108,113). 13

- 14
- 15 16

2.3.2 Urinary tract infection

17 2.3.2.1 Acute cystitis

Acute cystitis is a bacterial infection of the lower urinary tract, and it is a common presentation in primary care (114). Acute cystitis is common among women, particularly women of reproductive age (29,115). It is uncommon in men, usually presenting in older males with functional abnormality (29,116,117). Acute cystitis is a common infection of childhood (29,115). The condition can be painful, and serious side effects include potentially fatal urosepsis, which is uncommon but can happen in neonates (115,118).

24

Patients frequently receive non-first-line antibiotics for presumed, initial presentations of
 UTI, when first-line options are recommended (15,17,29,48,119). These patients also
 receive overly long durations of antibiotic therapy (119-122). Patients diagnosed with
 recurrent / chronic UTI have different implications for treatment (123).

29

Patient age, gender and anatomical differences are used to classify the infection into uncomplicated and complicated (123). The guidelines used for analysis in this study vary by patient age and gender, and include routine empirical prescribing for adult patients with this condition (29). Empirical prescribing is recommended for symptomatic children who are positive for nitrites or leukocyte esterase, or if bacteriuria is identified by microscopy. For patients under one month old, hospitalisation and intravenous antibiotics are required, and this is not covered in the guidelines. 1

2

3

2.3.3 Repeats issued on antibiotic prescriptions

Repeats issued on prescriptions for antibiotics can be considered another form, or level, of inappropriate prescribing. However, there are some medications for which a repeat is required in order for a patient to receive a guideline-concordant course of antibiotics. This relates to the pack size produced by the manufacturer. Some prescriptions for cefalexin fall within this category. However, for the majority of formulations for antibiotics prepeats are not required for a single course of treatment. This is especially likely to be the case for initial presentations of an infection in urban areas.

11

Notable quantities of antibiotics are dispensed from repeat prescriptions in Australia (14,124) The dispensing of repeats often occurs a substantial time (months) after the original prescription (12,14,124), and therefore unlikely to be used for the same episode of infection (12). However, the default option in many medical practice software packages included repeats on the prescription (14), requiring prescribers to amend the prescription to remove. The aligning of manufacturing pack size with local guidelines has been raised as a stewardship opportunity (125,126).

19

A national report published in 2019 by the Australian Commission for Safety and Quality 20 in Health Care (the Commission) found that 50% of antibiotic prescriptions were issued 21 with one or more repeats on the prescription (14). Potential solutions presented included 22 amending either the default options in software or the amount dispensed at the 23 pharmacy level to align with course recommendations for antibiotics with substantial 24 course to pack size discrepancies (14,75). In 2017, Del Mar et al. (49) had raised the 25 26 need for change on unnecessary, repeat antibiotic prescriptions (75). Effective April 2020, the Pharmaceutical Benefits Advisory Committee (PBAC) proceeded to limit 27 prescribing authority for repeats on the following antibiotics: amoxicillin, amoxicillin with 28 clavulanate, cefalexin, doxycycline and roxithromycin (127,128). 29

- 30
- 31

2.4 Antibiotic stewardship and the surveillance of antibiotic use

32

33 **2.4.1** What is antibiotic stewardship?

34

Antibiotic stewardship promotes the prudent use of antibiotics, including both the prescribing by health professionals and the administration by patients (129-131). This involves developing and implementing strategies to prevent and minimise the
development of antibiotic resistance, whilst ensuring continuing effective and available
treatments for patients with infectious diseases (6,16,129,130).

4

5 Antibiotic stewardship includes promoting awareness regarding antibiotic resistance 6 among prescribers and the public (129,130,132). This includes implementing tools to 7 improve antibiotic prescribing practices to prescribe only when required, and 8 encouraging compliance with local prescribing guidelines (129,130,132). Stewardship 9 also includes the monitoring and measurement of antibiotic use, for continuing 10 improvement (129,130,132). Governance is also an important part of stewardship at the 11 organizational level (130).

12

In the out of hospital setting, important aspects of antibiotic stewardship include commitment and policy, strategies for practice, monitoring and reporting of antibiotic use, and education tools and resources (132). Examples of suitable proactive activities include providing prescribing feedback to prescribers, the distribution of educational resources to patients, ongoing prescriber education and promotion of guidelines, and facilitating access to expertise in antibiotic stewardship for prescribers (132).

19

21

20 2.4.2 What is surveillance and its role in antibiotic stewardship

The surveillance of antibiotic use involves the collection and analysis of data on antibiotic prescribing, dispensing or consumption, in order establish patterns and provide population health information, as well as reducing harm from antibiotic misuse (16,133). This is a notable level above the monitoring and reporting typically required of routine antibiotic stewardship, which does not involve analysis, for example. The surveillance of antibiotic use also facilitates detailed evaluation of antibiotic stewardship strategies (16,134).

29

The World Health Organization (WHO) is leading the international collaboration on surveillance of both antibiotic use and resistance and to promote antibiotic stewardship (6,47). Antibiotic stewardship and surveillance go hand in hand and inform each other. While the surveillance of antibiotic use is a focus of this research, the surveillance of antibiotic resistance is outside its scope. Nevertheless, a summary of surveillance programs for antibiotic resistance within Australia in Section 2.4.4 below.

1

2.4.3 Stewardship and surveillance of antibiotic use in Australian primary 3 care

4 5

2.4.3.1 Government surveillance, stewardship and other initiatives

In Australia the government subsidises the costs of many pharmaceuticals including 6 antibiotics under the Pharmaceutical Benefits Scheme (PBS) and the Repatriation 7 Pharmaceutical Benefits Scheme (RPBS) (17). Antibiotic prescriptions dispensed under 8 9 the PBS and RPBS are collated in the Department of Human Services' Medicare pharmacy claims database (17). This includes dispensing claims for antibiotic 10 prescriptions written by GPs, specialists, dentists, nurse practitioners, hospital 11 discharge, public hospital outpatients, and private hospital inpatients (17). Antibiotics 12 prescribed and dispensed in some remote Aboriginal Health Services, (and relevant to 13 WA), fall outside the PBS and affect the accuracy of its reported prescribing rates in 14 these remote communities (12). The PBAC makes decisions regarding which antibiotics 15 and other medicines are included on the PBS / RPBS (135). In the case of antibiotics, 16 the PBAC is currently advised by the Expert Advisory Group on Antimicrobial 17 Resistance (135). 18

19

In the 1990s and early 2000s, following quinolones being registered in Australia, the PBAC restricted their use in human and food production, with specific indications requirements to limit their prescription (135). This was an early but important step (136). Another initiative was the then Australia's Chief Medical Officer sending peercomparison letters to GPs with high rates of antibiotic prescribing in 2017 (137). This initiative's three forms of peer-comparison letters led to an overall nine percent reduction in antibiotic prescriptions from these GPs for a period of twelve months thereafter (138).

Australia's First National Antimicrobial Resistance Strategy 2015–2019 was published in 2015 by the Commission, to address AMR in humans, animals, agriculture, food production and the environment (16). This was followed by an updated strategy, endorsed in 2020, entitled Australia's National Antimicrobial Resistance Strategy – 2020 and Beyond (the 2020 Strategy) (139). While the majority of antibiotics are prescribed in community settings, rather than hospitals, at this time in Australia, stewardship on antibiotic prescribing predominantly focuses on hospital settings (19-21).

Using hospital data, the National Antimicrobial Prescribing Survey (NAPS) is a 1 standardised tool for auditing the quantity and quality of antimicrobial prescribing, is 2 coordinated by the National Centre for Antimicrobial Stewardship (NCAS) (140). NAPS 3 includes surveys, such as, the Surgical NAPS, Hospital NAPS, and Aged Care NAPS, 4 5 however, there is no survey specific to general practice (141). NCAS is developing stewardship in primary care through a current research project (142). In 2018, there 6 were 324 participants from both public and private hospital involved in the Hospital 7 NAPS (143,144). The 2019 Surgical NAPS involved 144 public and private facilities 8 participating across Australia (18,143,144). NAPS also has a collaborative partnership 9 with the Commission and the Department of Health. Furthermore, the National 10 Antimicrobial Utilisation Surveillance Program (NAUSP) monitors antimicrobial 11 consumption in participating private and public hospitals (145). In 2019, there were 219 12 acute-care hospitals participating in NAUSP (144,145), including 170 public and 49 13 private hospitals, with 100% involvement by principal referral hospitals. Both NAPS and 14 NAUSP programs are involved in the AURA project (144), which is detailed in Section 15 16 2.4.3.4 below.

17

18 The Commission has antibiotic stewardship initiatives (130), as does the WA Department of Health (146). The Antimicrobial Stewardship Clinical Care Standard 19 (147,148) was developed to guide optimal practice across all human health sectors. It 20 was published in 2014 (147) and updated in 2020 (148) and provides quality 21 22 requirements for how to document the clinical indication for prescribing antimicrobials and outlines adherence to current guidelines. Having an antimicrobial stewardship 23 program in place, and monitoring both antibiotic use and resistance are also 24 accreditation requirements of hospitals and day procedure services in Australia 25 (130,149). These requirements are contained within the National Safety and Quality 26 27 Health Service Standards (Second Edition) (150,151), as specific requirements of the Preventing and Controlling Infections Standard of 2021 (152), replacing the 2017 28 Preventing and Controlling Healthcare-Associated Infection Standard. Government-29 funded stewardship and surveillance initiatives, past and present, which incorporate 30 Australian primary care, are covered below in Sections 2.4.3.2 - 2.4.3.5. 31

32

33 <u>2.4.3.2</u> Bettering the Evaluation and Care of Health survey

The Bettering the Evaluation and Care of Health (BEACH) survey of general practice was an annual report produced by the Faculty of Medicine and Health at the University

of Sydney (89,153,154). From 1998 until it lost funding in 2016, the voluntary BEACH 1 survey reported the proportion of GP visits for URTIs for which antibiotics were 2 prescribed (89,153,154). Its data from 2015–16 demonstrates that URTI and UTI were 3 in the top 5 most frequently managed new problems (155). Over the decade 2006–07 4 and 2015–16, URTI as a newly managed problem declined from 4.4 to 4.2 per 100 5 encounters, whereas UTI decreased from 1.3 to 1.1 per 100 encounters (156). There 6 was no meaningful change in the URTI management rate between 2006-07 and 2015-7 16 (156). Over the decade 2006–07 and 2015–16, the frequency of prescriptions for 8 amoxicillin reduced from 2.2 to 1.8 per 100 problems managed (156). Roxithromycin 9 prescriptions also declined over the decade, from 0.9 to 0.5 per 100 problems managed 10 (156). There was, however, no significant change in the prescriptions of cefalexin, 11 amoxicillin with clavulanate, or doxycycline over the same period (156). 12

13

14 2.4.3.3 NPS MedicineWise

Evolving from the National Prescribing Service (NPS), NPS MedicineWise was a 15 leading authority on the quality use and prescribing of medicines (157,158). It focused 16 on educating the public and primary health providers regarding responsible antibiotic 17 18 prescribing from 1999 to 2022 but lost funding and ceased to continue (159,160). Its' early campaigns on viral URTI were believed to have decreased antibiotic dispensing 19 (159,161). Activities for GPs typically included clinical auditing, visiting educational 20 activities on appropriate prescribing, and academic detailing with peer-comparison 21 22 prescribing feedback (159,162,163). NPS was known for its antibiotic stewardship campaigns targeting both health professionals and the public (159,164-167). Its 23 'Choosing Wisely' campaign consisting of written, educational publications, and the 24 'Resistance Fighter' was a multi-media consumer awareness campaign to support the 25 fight against antibiotic resistance (159,164-167). NPS MedicineWise undertook 26 surveillance of systemic antibiotic prescribing by participating GPs in Australia 27 (157,158). Its program, called MedicineInsight (162,168,169), collected de-identified 28 data from participating general practices, and was the data source for this research, and 29 was utilised in AURA, as detailed below. 30

31

32 2.4.3.4 The Antibiotic Use and Resistance in Australia project

The Antibiotic Use and Resistance in Australia (AURA) project was established by the Commission, as part of the National Antimicrobial Resistance Strategy 2015–2019 (16). AURA also uses PBS and MedicineInsight data to examine antibiotic prescribing in

general practice. NAUSP provides hospital data to AURA (145), while NAPS provides 1 hospital and surgical data, as well as data from the aged care setting (141,143,144). 2 The first AURA report (12), published in 2016, found that non-first line antibiotics were 3 prescribed in 68% of sinusitis presenting in primary care in 2014. The second AURA 4 5 report found that in 2015 patients presenting primary care in major cities had increased antibiotic prescribing rates than patients in other areas (17). The third AURA report (14) 6 identified patients 65 years and over and children two to four years of age presenting to 7 general practice had the highest antibiotic use in 2017. It reported declining antibiotic 8 use since 2010, however, in 2015 and 2017 women with UTI received first-line 9 antibiotics 45-46% of the time (14). 10

11

The fourth AURA report (18) published in 2021 found overall increasing antibiotic use 12 in the hospital setting but decreasing use in the primary care setting. Specific to the 13 hospital setting in 2019, NAUSP reported the total antibiotic use across its participating 14 hospitals had seen an increase from 848 defined daily doses per 1,000 occupied bed 15 days in 2015, to 883 defined daily doses per 1,000 occupied bed days in 2019 (18). The 16 Surgical NAPS of 2019 is an example of notable hospital-based findings on antibiotic 17 18 use forming part of the fourth AURA report (18,143). The 2019 Surgical NAPS identified several issues in antimicrobial administration in the acute-care setting, relating to 19 procedural surgical prophylaxis and post-procedural surgical prophylaxis, as well as 20 specific problems with topical antibiotic use in both contexts (18,143). For procedural 21 22 surgical prophylaxis, there were issues in documenting both incision time and antimicrobial administration time, whereas for post-procedural surgical prophylaxis, 23 there were errors in the dose, frequency and duration of antibiotic administration 24 (18,143). 25

26

Specific to community care setting, the fourth AURA report (18) utilised PBS / RPBS 27 data to demonstrate a 15% reduction in the age-standardised rate of antibiotics 28 prescribing per 1,000 persons between 2015 and 2019. It was also found that 50% of 29 these prescriptions for antibiotics were issued with repeats. Specific to the GP setting, 30 which is the focus of this research, the fourth AURA report (18) utilised MedicineInsight 31 data and found declining antibiotic use from 2015 to 2019, and a marked reduction due 32 to decreased patient consultation with GPs during the Covid-19 pandemic. In 2019, 33 systemic antimicrobials were prescribed to 31% of patients attending practices 34 participating in MedicineInsight. The fourth AURA report (18) found that despite 35

downward trends in antimicrobial prescribing in the MedicineInsight program since
2010, that antimicrobials still appeared to be prescribed too frequently based on
guideline recommendations. For example, 82% of patients diagnosed with acute
bronchitis and 80% of patients with sinusitis, presenting to MedicineInsight-participating
practices were prescribed antibiotics, despite symptomatic treatment without antibiotics
being the recommendation in the guidelines.

7

8 2.4.3.5 The Atlas of Healthcare Variation

The first and third Atlas of Healthcare Variation (Atlas), published in 2015 (15) and 2018 9 10 (75) respectively, contain relevant information regarding antibiotic prescribing using PBS / RPBS dispensing data. The first Atlas found that antibiotic dispensing rates 11 appeared to increase with decreasing SES, in line with infection rates (15). Antibiotic 12 prescriptions, particularly quinolones, dispensed in WA were notably lower in WA than 13 other states and territories, although Aboriginal Health Services dispensing is not 14 included (15). The third Atlas (75,170,171) found that for 2016-17, children 0-9 years of 15 in inner regional and major city areas with lower SES had decreased antibiotic 16 dispensing compared to children with other SES. However, this was not the case for 17 children 0-9 years in other areas (75,170,171). 18

19

Despite these initiatives, antibiotic stewardship in the general practice setting is limited
and requires sustained investment (126,172).

22

24

23 2.4.4 Surveillance of Antimicrobial Resistance in Australia

There are multiple systems in place for the surveillance of AMR in Australia, in both the hospital and community setting. These include the National Alert System for Critical Antimicrobial Resistance, and the National Notifiable Diseases Surveillance System. Pathogen-specific surveillance systems include the National Neisseria Network, and its component programs. There is also the Australian Group on Antimicrobial Resistance and the Australian Passive AMR Surveillance system.

31

Critical antimicrobial resistances are resistance profiles of microorganisms which are a dangerous threat to last-line antimicrobials. This means microorganisms demonstrating resistance to the last-line antimicrobials available (173). The Commission established the National Alert System for Critical Antimicrobial Resistances, which is abbreviated to the name, CARAlert (173). Its objective is to establish a nationally coordinated system for critical antimicrobial resistance and to monitor and facilitate early response to any
outbreaks of these organisms (173). There is a list of priority organisms for which
participating laboratories regularly provide data (173). CARAlert functions as part of the
AURA surveillance program (173).

5

The National Notifiable Diseases Surveillance System coordinates incidence data for 6 70 diseases which can pose a danger to Australians (174). This includes blood-borne 7 viruses, certain sexually transmitted infections, childhood diseases, arthropod-borne 8 diseases like malaria as well as respiratory infections like influenza and coronavirus-19 9 (174). Although it is not tailored specifically to antimicrobial resistance but infection 10 incidence, several diseases have their own surveillance systems and those specific to 11 resistance. The collation of data enables timely monitoring and surveillance and, where 12 necessary, response (174). 13

14

The Australian Group on Antimicrobial Resistance involves microbiology laboratories 15 from all states and territories nationally (175). This collaboration conducts surveillance 16 and susceptibility testing of specific pathogens of concern (175). It also collates 17 18 treatment, demographic and outcome data for diagnoses such as bacteraemia (175). It also forms part of AURA surveillance (175). Molecular testing is undertaken for some 19 isolates for its blood-stream infection programs, including the Gram-negative Sepsis 20 Outcome Program, the Australian Staphylococcal Sepsis Outcome Program, and the 21 22 Australian Enterococcal Sepsis Outcome Program (175).

23

The National Neisseria Network involves a group of public and private laboratories that contribute to passive surveillance of resistance and susceptibility for the pathogenic *Neisseria* species, *Neisseria gonorrhoeae* and *Neisseria meningitidis* (176). It runs two programs on invasive disease: the Australian Gonococcal Surveillance Programme and the Australian Meningococcal Surveillance Programme (176).

29

The Australian Passive AMR Surveillance collates resistance data from pathology services across Australia, both public and private (177). It was established by the Commission with support from Queensland Health (177). It facilitates analysis and reporting on resistance and its output includes both cumulative antibiograms and resistance profiles during specified time frames (177).

1 **2.5**

2

Access to general practice prescribing data in Australia

Australia is lagging behind many developed countries in general practice data collection and access for primary care research (51). The Australian Government has limited oversight of general practice, as the strong majority is private enterprise (51,58,178). Unlike in England, for example, patients at general practices in Australia can move entirely freely between practices and attend multiple practices within the same time period (179). Due to the private business structure (58), a fear of losing patients has been noted as a reason for inappropriate prescribing in the Australian GP setting (180).

Patient data is routinely collected by administrative systems in daily general practice but 11 12 is not well utilised in Australia, particularly for research, due to barriers in sharing and access (50,178,181,182). Some notable GP datasets include MedicineInsight (168), 13 PenCS and its CAT4 program (183,184), the University of Melbourne's Patron data 14 repository (185), and the Melbourne East Monash General Practice Database, formerly 15 known as MAGNET, operating as POLAR by Outcome Health (186-188). These 16 datasets do not include all GP practices, however, they provide opportunities for 17 research and feedback to general practice (26). Data are typically not linked to PBS / 18 RPBS dispensing data or to either secondary or tertiary care, which limits the research 19 (182). Substantial change is needed to improve access to administrative datasets (189). 20 21 MedicineInsight offered a high quality, large-scale data source for general practice research, however, it ceased operation in 2022 (160). 22

23

The detailed review of individual patient records enables the most accurate 24 classification of inappropriate from appropriate prescribing. As this is not feasible on a 25 large-scale, a common trade-off with the analysis of large-scale prescribing data in 26 Australia has been the use of basic classification of appropriate from inappropriate 27 prescribing, such as, using the diagnostic condition alone to determine appropriateness 28 (12,14,17,18). For example, some publications note that all URTIs are presumably viral 29 and therefore all antibiotics for these conditions are inappropriately prescribed 30 31 (12,14,17,18). However, this thesis presents a middle ground, in which more patient data is incorporated to enhance the differentiation between inappropriate and 32 appropriate prescribing whilst using large-scale data, and thereby improve our 33 understanding of inappropriate prescribing. Albeit, the approach taken here in this thesis 34 35 requires additional resources for further data cleaning, preparation and analysis.

12.6Literature on primary care interventions and qualitative research on2inappropriate prescribing

4 Australia has followed the international trend of quantitative research on interventions aiming to limit antibiotic prescribing (28,137,138,190-193) as well as qualitative 5 research on exploring reasons for inappropriate prescribing (180,194,195). While both 6 qualitative and intervention studies are important in their own right, and help to 7 understand why inappropriate prescribing occurs, and how to reduce antibiotic 8 prescribing, respectively, they do not address the research question and fill the gap in 9 knowledge that this project does. Brief summaries of the literature on interventions and 10 qualitative studies follow in Sections 2.6.1 and 2.6.2, respectively. 11

12

3

13**2.6.1**A summary of interventions to reduce inappropriate prescribing in14primary care

15

There is a significant body of research on interventions designed to reduce inappropriate prescribing. These can be separated into interventions for clinicians, pharmacists, patients, as well as communication between these groups. The majority of community care interventions focus on acute respiratory infections, rather than a single condition, with greater potential for impact (196).

21

In their 2017 systematic review of interventions for reducing URTI prescribing in primary 22 care, Tonkin-Crine et al. (197) found that the strongest evidence was for shared 23 decision-making, C-reactive protein and procalcitonin testing, while they found low 24 evidence for patient- and clinician-targeted education-related interventions. Other 25 26 reviews have found that electronic decision support systems, adult procalcitonin testing, and educational interventions, particularly when multifaceted and addressing several 27 levels, show benefit without notable adverse effects (198-201). Meeker et al. (202) 28 found commitment letters on display within the practice were effective. Peer-29 comparison-based feedback to clinicians appears to be one of the most useful 30 strategies (203-212). However, as Bell et al. (213) note, the likely uptake of any 31 32 interventions as well as their population impact also require consideration.

33

Shared decision-making appears to be a promising strategy to facilitate improved guideline compliance through enhanced communication between clinicians and patients (214,215). Delayed prescribing strategies also appear useful although it is unclear about their uptake (126,216-219). Pharmacist involvement appears promising for improving
guideline concordance (220-222). Studies exploring point-of-care testing to resolve
diagnostic uncertainties offer possible solutions, despite potentially increasing follow-up
consultations (28,70,223-225).

5

6 Advancing information technology and large-scale data offer new opportunities for interventions and stewardship. A Chinese study validated an algorithm created for 7 reading free-text outpatient antibiotic prescribing data (226). Another study explored the 8 potential for social marketing to help improve prescribing by GPs (227). Tsopra et al. 9 (228) used preference learning to find inconsistences in antibiotic prescribing 10 guidelines, and created a decision aid for such situations (229). A computer game was 11 also created for medical students to teach the appropriate prescribing of antibiotics in 12 primary care (230). For a useful summary of information technology applications for 13 interventions and stewardship, please see Chapter 4 of the Commission's report, 14 Antimicrobial Stewardship in Australian Health Care (231). 15

16

17

2.6.2 A summary of qualitative research on inappropriate prescribing

18

It remains unclear why over-prescribing of antibiotics appears to continue in Australia 19 20 and internationally despite the presence of specific national guidelines. It likely involves a complex array of interconnected factors relating to the individual prescriber, the 21 22 practice, the patient, as well as socio-cultural influences. Some of these factors are thought to include patient demand, perceived patient demand, time pressure and / or 23 high patient loads, and complacency by the prescriber (23,180,232-241). Other factors 24 include uncertainty regarding the correct diagnosis on the part of the prescriber, which 25 is commonly termed, 'diagnostic uncertainty', and ignorance regarding antibiotic 26 resistance by both patients and prescribers (23,180,232-241). 27

28

29 Diagnostic uncertainty and low uptake of diagnostic testing methods are potential reasons for unnecessary prescribing (242,243). Fletcher-Lartey et al. (244) summarised 30 what they termed "patient expectations" as the predominant reason why inappropriate 31 prescribing occurs, and this term included time limitations, diagnostic uncertainty, and 32 suboptimal communication. In their study of over-prescribing, Rose et al. (240) grouped 33 their findings into three categories: GP fears, including anxieties and attitudes, patient 34 factors including expectations/demands and limited patient education, as well as 35 external factors, such as financial and time pressure concerns. In a Canadian study of 36

primary care physicians (245), a lack of continuity of care, such as in the case of single, episodic consultations (and the absence of established trust) has been linked to clinicians reporting lower confidence in their skills of avoiding antibiotic prescribing. One study found GPs perceived more pressure to prescribe in the after-hours settings and with fee-paying patients (246). Pharmaceutical company representatives were a source of prescribing pressure in another study from general practice (243).

7

In a study comparing general practices with lower and higher volumes of antibiotic 8 prescribing (247), prescribers in lower prescribing practices detailed helpful tools and 9 mechanisms to support not prescribing an antibiotic during the consultation. These 10 included helpful practice policies, sufficient consultation time, increased communication 11 with colleagues, and consistent prescribing behaviours within the practice (247). 12 Whereas, a qualitative study of Australian GP trainees found that antibiotic prescribing 13 was more likely among patients of Aboriginal or Torres Strait Islander descent, older 14 patient age, and affluent practice location (30,248). 15

16

17 Several studies note GPs' concerns regarding a lack of access to resistance data and guidelines (243,249). Public awareness regarding antibiotics and the accessibility and 18 usability of the guidelines for Australian GPs has also been questioned (250,251). 19 20 Internationally, issues have been raised with both the comprehension and the incorporation of the guidelines (246,252), as well as questions on prescribing metric 21 validity (253). Studies have found that prescribing decisions are frequently made based 22 on individual, past experience instead of guidelines and resistance data (246,252). 23 Tyrstrup et al. (254) highlight the need to consider GPs' preconceptions in guideline 24 development. The potential for using real-world patient data for guideline development 25 has also been raised (255). 26

27

It is notable that several of these potential reasons, for either over-prescribing or 28 29 inappropriate prescribing of antibiotics as listed above, also feature in Lam et al.'s (256) 30 thematic framework of reasons for over-testing by clinicians, albeit not necessarily GPs. While interactions with colleagues were raised as a supportive mechanism against 31 (unnecessary) antibiotic prescribing (247), there were several instances of pressure 32 from colleagues linked to over-testing in Lam et al. (256). It may be possible that there 33 may be a relatively common, risk-adverse behaviour spanning multiple areas of clinical 34 medical care including both inappropriate antibiotic prescribing and over-testing (and 35

1 overdiagnosis), functioning under the guise of (potentially misguided) medical diligence

2 and / or self-protection.

- 3
- 4 5

2.7 Literature review of studies using quantitative methods

As detailed above in Sections 2.6.1 and 2.6.2, there is a substantial body of research on 6 interventions designed to reduce inappropriate antibiotic prescribing and gualitative 7 research to try to explain why it occurs. While these studies can help provide insights 8 regarding the question, neither truly address the crux of the problem - which is, 9 ascertaining what factors drive inappropriate prescribing. Quantitative research 10 analysing patient-level data to identify what drives inappropriate prescribing is limited. 11 Stedman et al. (257) note that influencing prescribing inherently requires knowledge of 12 its drivers locally. One must firstly research inappropriate prescribing in detail in order 13 to truly understand it, prior to developing interventions or qualitative explorations of why 14 15 it occurs (239).

16

Given the focus of the empirical research presented in this thesis, it was considered 17 important to focus on quantitative studies which use real, patient consultation data in 18 primary care to identify what factors are associated with inappropriate prescribing. The 19 definition of the general practice / primary care setting used in this research excludes 20 settings such as the emergency department, hospital inpatients, aged care, dental, and 21 other medical specialist care in the community. However, it is important to note that 22 there are differences in the setting and terminology of primary care provision 23 internationally. For the purposes of direct comparison with Australian general practice 24 at an international level, the review focused on medical professionals in the primary care 25 setting, although it is noted that nurse practitioners and physician assistants can also 26 provide primary care. For example, in the United Stated (US), paediatricians and family 27 medicine physicians provide primary care. However, in contrast, paediatricians do not 28 provide primary but secondary and tertiary care in Australia and the United Kingdom 29 (UK), and GPs are the main primary care providers. Despite variations in the 30 terminology, it was considered important not to exclude research involving health 31 professionals that perform the same function in primary care as GPs. Furthermore, 32 these differences have also led to the inclusion of some studies encompassing general 33 practice or primary care alongside other healthcare settings outside primary care, such 34 35 as outpatient or emergency care. Excluding such research would result in a limited number of relevant studies identified. These differences in primary care providers does
add complexity to making international comparisons with general practice in the
Australian primary care context.

4

5 Please note there are multiple, existing quality indicators which have been developed to measure antibiotic prescribing (258-265). These tend to measure proportions of 6 diagnoses and / or prescriptions meeting specific criteria, and they can be useful for 7 8 comparing prescribing rates across areas / countries or against benchmarks. However, quality indicators do not typically classify inappropriate from appropriate prescribing, 9 which facilitates analysis of predictors of inappropriate prescribing using individual-level 10 data. For a summary of prominent antibiotic prescribing quality indicators, please see 11 12 Appendix A.2.

13

15

14 2.7.1 Different definitions of inappropriate antibiotic prescribing

Due to the multiple levels of advice within prescribing guidelines, this leads to multiple different definitions of inappropriate prescribing (259-262,266-269). This ranges from prescribing for conditions where not indicated, such as for viral URTIs where antibiotics would be ineffective (210,270-274), to the use of non-first-line antibiotics when first-line antibiotics have not yet been tried (22,270,275-280). The evaluation of situations in which an antibiotic was indicated but not prescribed is another possibility (25,281).

22

Some studies consider prescribing by specific antibiotic class, such as fluoroquinolones, 23 to represent inappropriate prescribing (282-285). Many studies focus on the use of non-24 first-line antibiotics (22,270,277-280,286). Antibiotic prescriptions can further be 25 26 evaluated against the recommended dose / frequency and duration recorded in guidelines such as excessive duration (22,285,287-291). An alternate approach taken 27 in one study was to define all prescriptions with missing indication field as inappropriate, 28 as well as examining prescriptions with non-specific indications entered by the GP 29 (292). The definitions used in this thesis are explained in detail in the Methods chapter. 30 31

Due to the multiple definitions of inappropriate prescribing and the various terms for general practice or its equivalent internationally and the many types of antibiotics, the search strategy is not listed here but can be found in **Appendix A.3**.

1

2 3

2.7.2 Considerations regarding the literature criteria

4 When examining the existing literature, it was considered relevant to focus on research investigating what is intrinsically driving inappropriate prescribing, when the objective 5 was to identify predictors from within patient-, consultation-, provider-, and practice-6 related factors. Published literature was limited to studies which clearly differentiate 7 inappropriate from appropriate prescribing. Most primary care research use only the 8 diagnosed to separate inappropriate from appropriate prescribing 9 condition (12,14,17,18,274,293,294). Internationally, there is limited quantitative research from 10 large-scale primary care data which differentiates inappropriate from appropriate 11 prescribing using more than diagnostic condition alone. 12

13

14 Most studies also do not classify initial from non-initial consultations by order of occurrence in time to ascertain the quality of clinical management occurring at initial (or 15 subsequent) consultations for the same episode of infection 16 (12,14,17,18,26,277,279,293,295). This includes AURA using large-scale GP data 17 (12,14,17,18). Although it is typical to exclude diagnoses coded as "chronic" (295), 18 which may represent a different condition or guideline, few studies indeed limit analysis 19 20 to initial consultations for the infection of interest by examining consultations longitudinally and selecting the first occurrence (296). Fossum et al. (297) took a 21 22 different approach, by limiting the analysis to the first antibiotic prescription per patient - although this does not necessarily imply it was the initial consultation. 23

24

There are also quantitative studies which analyse predictors despite having not 25 differentiated inappropriate prescribing from appropriate prescribing, thereby identifying 26 predictors of overall antibiotic prescribing (257,298-303). Although these can provide 27 28 useful information, it is the inappropriate prescribing which requires curbing, while appropriate prescribing may prove dangerous to reduce, hence the need for research 29 which both distinguishes between the two and identifies predictors. Alternatively, there 30 are studies which define (in)appropriate prescribing using predominantly descriptive 31 statistics but do not use analytical methods to identify factors associated with such 32 (in)appropriate prescribing (26,87,288,304-308). 33

34

This overview of the literature includes only published literature based on real patient data rather than hypothetical vignettes or questionnaires, for which clinician awareness

regarding the study and subsequent participation may in itself influence prescribing 1 (309). There are multiple small-scale questionnaires of prescribers which directly ask 2 what influences decision-making, and small, vignette studies which may elicit responses 3 with more ideal prescribing behaviour (123,310-312). For guality control, if the 4 5 prescribing data was obtained via questionnaire / vignette, these were excluded from this overview of the literature, apart from national surveys with strong design which were 6 considered less likely to be biased. Multiple studies which combined real patient-7 prescriber data obtained from the electronic record, with prescriber demographics 8 obtained by survey, were considered suitable for inclusion (313,314). 9

- 10
- 11 12

2.7.3 Literature review overview

Much of quantitative research on inappropriate antibiotic prescribing as per the definition used in this research is from the US. The Netherlands, Canada, Denmark, Sweden, Norway, Iceland, Germany, France, the UK, and Spain also contribute. It may be the accessibility of primary care data for research that facilitates such research. From the available literature, predictors of inappropriate prescribing can be classified as factors relating to the patients, or factors relating to the consultation and practice setting, or factors relating to GPs themselves.

20

21 2.7.3.1 Patient factors

Patient age has been found to be a risk factor for inappropriate antibiotic prescribing in 22 studies. however, with inconsistent direction of effect 23 several (25,39,280,281,283,284,291,294,296,297,315-325). Several paediatric studies found 24 that the youngest patients had higher odds of inappropriate prescribing than older 25 children (297,321-323), while others found the that the odds increased with increasing 26 child age (280,296). Increasing probability of inappropriate prescribing was found with 27 increasing age in several studies including adults or patients of all ages 28 (25,281,283,284,316,324). Antibiotic prescriptions received without an encounter with a 29 clinician in the US was found to be linked to increased patient age (318). However, 30 multiple studies found insufficient evidence of an association between inappropriate 31 prescribing and patient age (270,276,295,315,319,326,327). One study found that 32 increasing patient age was associated with unnecessary prescribing for presumably 33 viral infections, while decreasing patient age was linked to suboptimal choice of 34 antibiotic for bacterial infections (294). Another study including many types of infection 35

found that children were more likely to receive an antibiotic dosing error, while adults
were more likely to receive the wrong duration of antibiotic (320).

3

Whether patient gender is a genuine predictor of inappropriate antibiotic prescribing -4 5 for conditions unrelated to anatomical gender differences - remains uncertain. Several studies found male patients at higher risk of inappropriate prescribing than females 6 (280,291,328), while others found that females were more predisposed (294,321,326). 7 Malo et al. (39) found that the direction of effect differed between definitions of 8 inappropriate prescribing, with males more likely to receive inappropriate decisions but 9 less likely to receive a non-recommended antibiotic selection than females. Several 10 authors, however, found no association between inappropriate prescribing and patient 11 gender (276,281,297,319,329,330). 12

13

SES and whether patients are from culturally and linguistically diverse (CALD) 14 backgrounds may potentially be linked to inappropriate prescribing. Health insurance 15 status, as a proxy for SES, has also been linked to inappropriate prescribing, with no. 16 or cheaper, health insurance being found to be a predictor of lower rates of non-first-17 18 line prescribing (276,295). However, patients with public health insurance had a high risk of unnecessary prescribing in another study (316). Wattles et al. (317) found that 19 insurance and CALD background status had no influence on inappropriate prescribing 20 for AOM, however. Studies in US ambulatory care, in which inappropriate prescribing 21 22 was compared between patient groups with different CALD backgrounds status, have found an association between patients from CALD backgrounds and less inappropriate 23 24 antibiotic prescribing (276,295,326).

25

Patient comorbidity may prove to be another important factor in predicting inappropriate 26 27 antibiotic prescribing. It has been linked to increasing inappropriate antibiotic prescribing studies in studies, direction in several with consistent of effect 28 (39,281,289,294,317,318,326,331). Dekker et al. (281) found that patients with 29 comorbidities had increased likelihood of inappropriate prescribing by 70%, while Singer 30 et al. (294) found that with every increasing comorbid condition, there was an eleven 31 percentage point increase in unnecessary prescription risk. 32

33

The presence of inflammatory signs, such as fever, may also influence inappropriate prescribing (25,281,332). Patients with antibiotic allergy were independently associated

1 with less chance of receiving appropriate choice of antibiotic as well as less chance of a prescription with appropriate duration (289). A study of UTI patients found 2 inappropriate fluoroquinolone prescribing was more likely among patients with a history 3 of resistance to nitrofurantoin and recent nitrofurantoin use (283). Fernandez-Urrusuno 4 5 et al. (333) found that UTI and dental infection were more likely to be treated appropriately, whereas URTI and skin infections were predisposed to inappropriate 6 treatment. Amoxicillin and amoxicillin with clavulanate have specifically been associated 7 with less appropriate treatment (333). In one study, patients being treated with multiple 8 antibiotics simultaneously were more likely to receive appropriate treatment (333). 9 Fischer et al. (318) found that prescriptions issued without a patient encounter occurred 10 more frequently among patients with multiple encounters, and Singer et al. (294) found 11 increasing risk of inappropriate prescribing with increasing number of patient visits. 12

13

14 2.7.3.2 Clinician and healthcare setting factors

Multiple studies compared the appropriateness of antibiotic prescribing between 15 healthcare settings or the type of medical specialist or health professional 16 (237,295,296,321,322,324,329,330). Several studies found these to be significant 17 18 drivers of inappropriate prescribing (237,295,296,324,329,330), while others found no such association (321,322,324,329). For example, Barlam et al. (326) found that family 19 medicine physicians more frequently prescribed inappropriately than general internal 20 medicine physicians, while another study found the opposite (276). Regardless of the 21 22 type of prescriber, the primary care setting does seem prone to inappropriate prescribing (237,280,295,315,328,330). Chen et al. (296) found higher rates of guideline 23 24 non-adherence for UTI prescribing in community clinics as opposed to medical centres, which may suggest a link to patient SES. 25

26

Measures of rurality, such as the health care provided to patients living in urban or rural areas (290,294,317,334) may also affect inappropriate antibiotic prescribing. Several North American studies have found that inappropriate prescribing was linked to patients living in rural areas (294,317,334). Women with UTI in rural areas were found to receive antibiotics with excessive duration compared to women in urban areas (290).

32

Some GP-related factors associated with more non-first-line prescribing by GPs that
 were identified in studies in the Netherlands include: working in solo general practice
 (271); frequency of consultation for RTIs; less frequency of use of national prescribing

guidelines; and willingness to try prescribing new drugs (277). In 2007, Cadieux et al. 1 (270) found inappropriate prescribing was linked to high-volume practices, and GPs with 2 foreign medical degrees, whereas another study found large practices were 3 predisposed to inappropriate prescribing (294). One study in after-hours care found 4 improving antibiotic choice was correlated with decreasing patient to prescriber ratio, 5 however, there was no relationship with practice size (335). Multiple studies found that 6 prescribers with both higher volumes of patient turn-over and higher antibiotic 7 prescribing rates were predisposed to inappropriate prescribing 8 (270,277,284,297,321,329). Lindberg et al. (322) found no association between first-line 9 prescribing and median consultation duration or days between work sessions, however. 10 11

The prescriber's age or their duration in practice was a significant factor in several 12 studies but with inconsistent direction of effect (270,280,294,296,328,336). Several of 13 these suggest increasing age adds to the risk of inappropriate prescribing 14 (270,296,336). Kozyrskyj et al. (328) found that older prescriber age was linked to higher 15 probability of prescribing for viral RTI but to lower probability of non-first-line prescribing 16 for bacterial infections than younger prescribers. Blommaert et al. (280) found that 17 18 prescribers in the 40-44 age group tended to prescribe non-first-line antibiotics more frequently than their older or younger colleagues. Chang et al. (316) found that clinicians 19 who were recently employed or with less professional post-graduate education were 20 predisposed to inappropriate prescribing of antibiotics. Several other studies found no 21 22 association between the prescriber's age and the odds of inappropriate antibiotic prescriptions (291,297,321,329). 23

24

25

5 2.7.3.3 Conclusions from the literature overview

The literature suggests that patient age may influence inappropriate prescribing. It is possible that the impact of patient age, if any, varies by condition, and the youngest and oldest patients may be most risk of inappropriate prescribing for some conditions. For conditions other than UTI, the literature regarding patient gender is less convincing, and if true, it appears to have less influence than patient age.

31

Clinicians may have increased concern for the health of patients with comorbid conditions, such that clinicians may prescribe for these patients when technically not required. Patients with multiple attendances is also a feasible driver of inappropriate prescribing, however, in many publications it is unclear whether multiple consultations include follow-up or multiple initial consultations for separate episodes of infection.
Furthermore, these do not differentiate initial from non-initial consultations. To this end,
the AURA project (12,14,17,18), which uses large-scale data from Australian primary
care but uses only the condition to classify inappropriate from appropriate prescribing.
This highlights the importance of research in identifying initial from non-initial
consultations in Australian primary care.

7

The literature suggests the possibility of a link between inappropriate prescribing and 8 patients with different SES, measures of rurality and CALD background status. It would 9 appear plausible that patients in affluent, urban areas may attend nearby care providers 10 for minor ailments, exert more pressure on clinicians, receive more inappropriate 11 prescribing, and, that infectious diseases are known to occur more frequently in 12 disadvantaged areas which may increase the chances of appropriate prescribing in 13 these areas. Similarly, patients in rural areas may access care less easily, less 14 frequently and for more serious ailments. However, rurality, by virtue of less accessible 15 healthcare, may be a reasonable reason for providing a technically unnecessary 16 prescription. SES is also known to be complex to measure (337,338). 17

18

With regard to clinician factors, the literature suggests that increasing workload may be 19 a highly viable option as a potential driver of inappropriate prescribing. Prescriber age 20 or year of certification, as a potential predictor, may be a complex mix of years of 21 22 experience working in primary care and degree of involvement in continuing medical education (CME) or peer-group activities. It seems feasible that optimal prescribing 23 behaviours may occur after several years of experience in general practice but this may 24 begin to taper off when GPs are closer to retirement, less involved in CME, and less 25 actively engaged in changing policy and guidelines and consulting with peers. 26

27

In the international studies conducted specifically assessing inappropriate prescribing 28 in primary care, several are not limited purely to general practice (25,270,277,281). 29 Partly due to international differences, many studies including general practice / primary 30 care also cover other areas of ambulatory / outpatient care (237,296,323,326,332,339) 31 or emergency departments (295,324,330). Many studies are also limited to either child 32 (271,296,297,315,323) or adult (237,276,281,326,332) populations. The clinical 33 conditions covered by each study also tend to differ, such as AOM, UTI, URTI, or 34 multiple conditions (237,271,276,281,283,294,296,297,315-320,323,326,332). This is 35

in addition to the fact the populations and disease epidemiology from different nations 1 are not homogenous. Furthermore, there are multiple, different definitions of 2 inappropriate prescribing utilised within the identified studies 3 (25,39,280,281,283,284,291,294,296,297,315-325), and this adds to the difficulty of 4 5 drawing meaningful comparison across them.

6

Much of the existing literature does not appear to allow for multiple types or levels of
unobserved heterogeneity, such as, at the patient, GP, and practice levels
(22,48,277,281,286,295,296,315,323,325,326,329,330,332,340-342). Studies which
do not allow for (potentially important) unobserved heterogeneity at each level may lead
to reliance on spurious results.

12

This literature review highlights that inappropriate prescribing requires quantitative studies that go to the trouble of differentiating appropriate from inappropriate prescribing, with clear definitions. This highlights that the research question is very specific, and it needs a data-driven approach, with real-world patient data at large-scale and at an individual patient-provider level, as well as sound methodology in statistical modeling, to accurately address.

19

There is currently limited information on inappropriate antibiotic prescribing in Australian general practice, and primary care internationally, particularly with respect to empirical evidence of drivers of inappropriate prescribing. This review has the overarching finding of the need for more research to identify factors driving inappropriate antibiotic prescribing for specific conditions in certain settings and populations (294).

25

26 2.7.3.4 What this thesis contributes

This thesis will begin to fill this important knowledge gap, by identifying predictors of inappropriate antibiotic prescribing for initial presentations of URTI and UTI from largescale patient data from WA general practice. It will also explore and identify trends in inappropriate prescribing for these conditions over time.

31

This thesis will use quantitative analytical methods and large-scale patient data to identify predictors of inappropriate prescribing in general practice for UTI and URTI, using more clinical information than the condition diagnosed to differentiate inappropriate from appropriate prescribing. It will provide a more detailed definition of inappropriate prescribing and it will also limit analyses to initial presentations of
infection, to examine clinical management occurring at initial consultations.
Furthermore, this thesis will utilise statistical methods which allow for potentially
important unobserved heterogeneity at each level of the data, which should lead to
increased accuracy in the results of analyses.

6

To the best of the author's knowledge, at the time of writing, this is the first Australian research using quantitative methods and large-scale empirical patient data to identify predictors of inappropriate prescribing in general practice for UTI and URTI, using more clinical information than the condition diagnosed to differentiate inappropriate from appropriate prescribing, and to limit analyses to initial presentations of infection. This is also believed to be the first analysis of inappropriate antibiotic prescribing within Australian primary care to allow for unobserved heterogeneity.

14

This thesis falls within the second strategic objective of WHO's Global Action Plan on Antimicrobial Resistance, *"to strengthen knowledge through surveillance and research"* (6). In the Australian context, it falls under the national strategy's objective five to develop a research agenda and promoting investment in new means of preventing and containing AMR (16). While there is research regarding drivers of antibiotic prescribing, there is limited research of such for inappropriate prescribing - that which must be reduced to extend the effectiveness of current antibiotics (15).

22

Research in this field is an urgent necessity borne from increasing antibiotic resistance 23 24 globally (16,43,44). This project creates the potential for new knowledge regarding antibiotic prescribing in the community and primary care, with potential impact both 25 26 nationally and internationally. The findings from this research will be published in peerreviewed journals. The findings will inform policy and practice, with feedback to be 27 provided to general practice, and to support antibiotic stewardship, as well as lead to 28 further research on antibiotic prescribing in primary care. Its applications include 29 informing the development of evidence-based interventions to reduce inappropriate 30 antibiotic prescribing to stem the emergence of resistance. 31

- 32
- 33

CHAPTER 3 METHODS

1 2 3

4 5

3.1 Introduction

6 This research is a quantitative analysis of large-scale data obtained from general 7 practice in WA to identify predictors of inappropriate antibiotic prescribing. The intention 8 was to establish empirical evidence with respect to antibiotic prescribing in this setting 9 for two main condition groups: URTI and UTI. The overall objective was to identify and 10 quantify the inappropriate prescribing of antibiotics for these conditions, and to identify 11 predictors of, and trends in, inappropriate antibiotic prescribing. The methods employed 12 are depicted in **Figure 3-1**.



1 2 3.2

Establishment of steering committee

The steering committee was established to provide expert guidance throughout the 3 4 project. Consultation with experts in the field occurred to help explain the results and guide interpretation. As well as having a member of supervisory panel with experience 5 working as a general practitioner (GP), it was considered particularly important to have 6 7 GPs on the steering committee. In addition to two GP experts and leaders in their field, membership of the steering committee also included an infectious diseases physician, 8 a pharmacist and representatives from WAPHA (343). Consultation included redirecting 9 specific avenues of enquiry, potential explanations to explore and the provision of 10 contextual advice. 11

12

13

3.3 Data source and sample size

14

De-identified, patient-GP consultation data was obtained from the MedicineInsight 15 program for patients attending enrolled general practices in WA, with at least one GP 16 visit between January 2012 and June 2017, inclusive (168,344,345). The 17 MedicineInsight program was run by the NPS and collects patient-GP-practice level 18 consultation data from 500 consenting general practices nationally, including over 2,000 19 GPs and over 2 million patients at the time of data access (157,158,168,344,346). The 20 data were found to be reasonably comparable to the patient demographics of national 21 Medicare Benefits Schedule data for 2016 to 2017 (344). 22

23

At the time of extraction, there were 74 practices enrolled in WA, with 68% practices in major cities, 14% inner regional, and 11% in outer regional areas, 5% in remote and 2% in very remote areas (34,35). Of these, 52 practices in WA had data available for the entirety of the study period.

28

29 These data included the date of consultation, prescriptions issued at each consultation, free-text reason for visit / presentation, one or multiple free-text diagnoses entered at 30 each consultation, pathology and radiology results, and clinical observations made by 31 the GP during the consultation. Key patient characteristics were also collected, including 32 age, Indigenous status, allergy status and information, smoking status and various 33 indicators of socioeconomic status (including government pension status, healthcare 34 35 concession status and veteran status). MedicineInsight also created algorithms for patient comorbid conditions, resulting in a dichotomous indicator of whether the patient 36

ever had a history of various comorbid conditions based on consultation, pathology and
 diagnostic information (168,345).

3

Patient-related datasets were comorbid conditions, allergy information, diagnoses, 4 5 encounters, prescriptions, atomical and non-atomical pathology, observation files. The patient characteristics dataset contained year of birth and age, residential rurality based 6 on Accessibility / Remoteness Index of Australia, Aboriginal and Torres Strait Islander 7 status, smoking status, and indicators of SES including Socio-Economic Indexes for 8 Areas (SEIFA), an Index of Relative Socio-Economic Disadvantage using postcode of 9 residence, pension status, government concession status and veteran status (34-37). 10 The patient comorbid condition dataset consisted of binary dichotomous variables 11 indicating whether the patient has a history of ever having each of the following: 12 cardiovascular disease, heart failure, atrial fibrillation, stroke, transient ischaemic attack, 13 asthma, chronic pulmonary obstructive disease, cancer, chronic liver disease, chronic 14 kidney disease, mental health problems and substance abuse (see Appendix B.4.10 for 15 16 more details). The prescription dataset provided 40 million prescriptions, as well as the dosage, frequency, duration and date of the prescription. 17

18

The separate datasets were linked prior to analysis. Each patient has a unique de-19 20 identified patient ID, and these patient data are linked within but not across practices. As a result of this, an individual patient who happened to attend multiple 21 22 MedicineInsight-participating practices during the period would appear in the data as several different patients. There are also repeated visits within the same practice in the 23 data: the same individual patient seeing the same individual provider multiple times, as 24 well as the same patient seeing different providers at the same practice, and the fact 25 that the same provider consults multiple different patients within a single practice. Each 26 27 GP / provider and each practice also has a unique de-identified ID to allow for linking of records (patient ID, provider ID and practice ID, respectively). 28

- 29
- 30

3.4 Data cleaning and preparation

- 31
- 32 **3.4.1 Diagnoses**
- 33

The presenting conditions of interest were: URTI (consisting of rhinosinusitis, the common cold, influenza / ILI, sore throat, tonsillitis, pharyngitis, AOM) and UTI (consisting of acute cystitis).

The initial diagnoses file received included 1,557,387 observations, with 275,880 1 different diagnoses. Removing impossible/erroneous dates and missing or blank 2 diagnoses resulted in 1,489,540 observations. As detailed below, to explore prescribing 3 for patients with acute URTI and UTI, patients were filtered by the free-text string 4 entered by GP in the diagnosis field. Frequency tables were created for diagnostic terms 5 used, and relevant key words were collated and refined for conditions of interest. These 6 keywords were used as search terms within the original datasets to create datasets 7 containing conditions of interest, which were subsequently matched with antibiotic 8 prescriptions for the same patient, provider and practice occurring on the same date as 9 10 the diagnosis.

11

In the dataset received, patient diagnosis was predominantly entered and coded using 12 the International Classification of Diseases (10th edition) or the International 13 Classification of Primary Care (347,348), which translate across to specific diagnostic 14 conditions listed in national antibiotic prescribing guidelines, or the Therapeutic 15 Guidelines: Antibiotic (the guidelines) (29). However, the diagnosis field in general 16 practice software was typically a free-text entry completed by the GP. As a result, there 17 18 was usually substantially more variation in the diagnoses received from general practice than ideal, requiring significant cleaning and grouping of relevant diagnoses to form 19 useful datasets. Figure 3-2 below is a snapshot of some diagnoses within the original 20 dataset and potentially relevant to acute otitis media (AOM) prior to refinement into 21 22 appropriate, final search terms. Please note that the diagnoses included in this research always refer to acute infections, even if the term acute is not always mentioned. 23

24

Character string functions were used to search and refine datasets containing relevant 25 search terms for each condition. This process included removal of irrelevant terms, 26 allowance for acronyms, spelling mistakes, spelling variations and punctuation. For 27 conditions where empirical prescribing is appropriate, first-line antibiotics should be tried 28 at the initial visit before trying non-first-line options. Whereas if empirical prescribing is 29 not recommended for a condition, it may be appropriate to prescribe for an ongoing 30 infection at a subsequent consultation. Therefore, accurate analysis requires separation 31 of initial from non-initial presentations. Diagnoses were limited to initial presentations for 32 the episode of infection (295,296), including removing diagnoses within fourteen days 33 of a previous consultation for the same condition group (29). Note that the term 34 diagnoses refers to initial presentations throughout. 35

OM 60 EFFUSION EAR 35 SECRETORY OTITIS MEDIA 32 ROM 35 OTITIS EXTERNA 19013 OE 5 OEL 1 OE RIGHT EAR L 1 OE WITH CELLULITIS LEFT EAR L 1 OE/ OM WITH ? PERF L 1 OE/OML 1 BLOCKED EARS 43 OTOMYCOSIS 99 OTITIS MEDIA 31733 OTITIS MEDIA RECURRENT 979 OTITIS EXTERNA RECURRENT 1070 EAR PAIN 1120 OTITIS MEDIA, RECURRENT 606 OTITIS EXTERNA, RECURRENT 692 EARACHE 1133 OTITIS MEDIA, SUPPURATIVE 502 OTITIS MEDIA; ACUTE 503 RIGHT EAR EFFUSION 3 RIGHT EARACHE 3 RIGHT SIDED OTITIS MEDIA 3 SEROUS OTITIS MEDIA 567 OTALGIA 491 BLOCKED EAR 485 GLUE EAR 1180

1

2

3

Figure 3-2: Picture of example list of terms appearing in the diagnosis field relating to ear infection, followed by frequency of their exact occurrence in the dataset

4

5

6 <u>3.4.1.1 Upper respiratory tract infection</u>

The URTI condition group was defined to include uncomplicated, acute URTIs, influenza 7 / ILI, acute rhinosinusitis / common cold, acute pharyngitis and / or tonsillitis, and AOM, 8 in accordance with the guidelines (29). For acute rhinosinusitis, allergic diagnoses were 9 10 excluded (29). Pharyngitis, laryngitis, and tonsillitis diagnoses were collated and included symptomatic diagnoses of sore throat and throat pain (29). Search terms for 11 AOM focused on otitis media and both otitis externa and chronic suppurative otitis media 12 were excluded (29). Note that influenza / ILI diagnose referred to standalone diagnoses 13 of influenza and/or ILI, and this excluded diagnoses mentioning secondary bacterial 14 infection or lower respiratory tract infection. For more detailed information regarding 15 search terms, please see Appendix B.2.1. 16

17

18 <u>3.4.1.2 Urinary tract infection</u>

UTI was defined to include acute cystitis, as per the guidelines, and relevant search terms included UTI, urinary tract infection, acute cystitis, and cystitis (29). Exclusion criteria included diagnoses with any mention of prostatitis, pyelonephritis, complicated,
 catheter-related UTI, UTI prophylaxis, or sepsis / septicaemia. For more information
 regarding search terms, please see Appendix B.2.2.

- 4
- 5

6

3.4.2 Antibiotic prescriptions

For each condition of interest, patients with at least one presentation for this condition 7 were included in the analysis. From this patient group, patients who were prescribed a 8 systemic antibiotic were identified from a list of antibiotic agents compiled using 9 predominantly MIMS Australia (Integrated or Annual edition) (349). All prescriptions 10 using the medicine active ingredient field were string-searched against the sample list 11 of antibiotics which had been collated. Simple spelling / typographical errors obtained 12 from a list of brand names were allowed for. Antibiotic prescriptions for the URTI and 13 UTI diagnoses groups were then collated into two, separate datasets. Note that the term 14 antibiotics refers to systemic antibiotics throughout. 15

16

Anatomical Therapeutic Chemical classification codes were completed and sorted into
classes created (350). The string variables for prescription dose, frequency and duration
variables were cleaned. For more detailed information, please see Appendix B.3.

- 20
- 21 **3.5 Data analysis variables**
- 22 23

3.5.1 Response variables

24

25 <u>3.5.1.1</u> Antibiotic prescribing: standard used for assessment

The individual guideline publication covering the largest proportion of 2012 to mid-2017, the 2014 edition, was therefore selected as the standard for assessment for the duration of the study period (29).¹ The unique patient-GP pairing was defined as the unit of observation. Categorical data analysis was used to analyse dichotomous (antibiotic considered necessary for the condition: yes / no), nominal or ordinal variables with three or more categories (for example, choice of antibiotic prescribed: first-line, second-line, or third-line).

- 33
- 34

¹ For more information regarding differences in published guidelines, please see Appendix A.2.

- 1 Although there are multiple ways to define the appropriateness of antibiotic prescribing
- 2 (259-262,266-268), the following outcomes will be defined to represent different aspects
- 3 4
- 5 **0.** likely inappropriate decision including non-prescribing 'inappropriate decision'
- 7 **1.** likely unnecessary antibiotic prescribing 'unnecessary prescribing'

of inappropriate prescribing (see Figure 3-3)

- 8 2. ordered, increasing level of choice of antibiotic prescribed at initial
 9 consultation: starting from first-line, second-line, third-line or last resort
 10 (where relevant), to not recommended 'antibiotic choice'
- prescribing any other antibiotic other than first-line 'non-first-line
 prescribing'
- repeat positive antibiotic prescribing (prescriptions issued with one or more
 repeats present on the prescription) 'repeat positive prescribing'.
- 15

Note that outcomes 0 and 1 will be referred to as inappropriate decision and unnecessary prescribing, respectively, from now on. However, these two outcomes should always be considered to have "likely" as a prefix, as the guidelines cannot cover every situation. There are also situations in which it may be appropriate and necessary to prescribe an antibiotic but this may not be identifiable from the data (29).

21

Please note that outcomes 0 to 4 are relevant to URTI. As empirical antibiotic
 prescribing is appropriate for UTI, only outcomes 2 to 4 are relevant for this condition.

24

A summary of these definitions and the analyses approaches follows. Patients receiving 25 inappropriate decisions (namely, unnecessary prescriptions) were compared with 26 patients receiving appropriate decisions (either necessary prescriptions or appropriately 27 not receiving a prescription) for URTI, with the denominator of all patients with initial 28 29 presentations of the condition group. Note that there was considered to be insufficient information available to assess patients not receiving antibiotic prescriptions despite 30 truly needing them, which could also be termed 'under-prescribing', and were therefore 31 excluded from the analysis. 32

33

Patients receiving unnecessary antibiotic prescriptions were also compared with patients receiving necessary prescriptions for URTI, with the denominator being all patients prescribed an antibiotic for the same condition group.

Figure 3-3: Flow chart of outcome variables for identification of predictors of inappropriate prescribing for initial presentations of upper respiratory 1 tract infection



Note: For models 0 and 1, all variables should be considered to include "likely" as a prefix. Model 0 is the only model to include influenza-like illness. All models include upper respiratory tract infection, which is defined as: rhinosinusitis, pharyngitis / tonsillitis, and acute otitis media.

2

2 Patients receiving increasing choice of antibiotic line for each condition group (URTI or UTI) 3 were compared to patients receiving first-line antibiotics for that group, with the denominator being all patients receiving any antibiotics for that same condition group (URTI or UTI). 4 5 Patients receiving non-first-line antibiotic were compared to patients receiving first-line 6 antibiotics for a condition, with the denominator being all patients receiving any antibiotic for the same condition group. Patients receiving antibiotic prescriptions with one or more 7 repeats issued on them, when that was technically not required for a guideline-concordant 8 course for the condition, were also compared against patients not receiving a repeat on the 9 10 prescription. Definitions are covered in detail in the next few sections (3.5.1.2 to 3.5.1.4).

11

1

Algorithms were derived from the guidelines for each condition (29), to record the indication / patient criteria (age, gender, demographics and clinical observations obtained during the consultation) required for an antibiotic prescription to have been most likely justified. Although culture and other pathology testing was examined to assess compliance with guideline recommendations, these were not incorporated in the assessment of antibiotic choice, as these would not have been available to the prescriber at the time of the consultation.

19

20 Patient symptoms were not always recorded by the GP, and sometimes not all symptom information was available for analysis. However, there were many conditions for which 21 patient demographics alone provide sufficient information to assess prescribing. There were 22 conditions, such as acute pharyngitis / tonsillitis, for which the only reason to prescribe - to 23 24 prevent serious sequelae, such as, ARF - is uncommon in Australia, except in remote Aboriginal and Torres Strait Islander communities which are not attended by mainstream 25 general practice. Therefore, in affluent / urban situations, there is no valid justification to 26 prescribe an antibiotic for this condition unless the patient has a history of ARF (29,351). 27 Although one cannot assume there were no GPs in areas with cases of ARF, it was 28 considered rare and therefore unlikely (352). While Monaghan et al. (26) noted there were 29 insufficient data to assess the criteria for pharyngitis and therefore assumed criteria were 30 31 met, the approach here is opposite (important data such as temperature recording should be available in the clinical examination field, and that even in the absence of progress notes, 32 on balance, a genuine indication for prescribing for pharyngitis was unlikely). Each condition 33 was assessed separately during algorithm development as to the potential impact of any 34 unrecorded or missing information and is covered in more detail below. 35

<u>3.5.1.2</u> Upper respiratory tract infection: inappropriate decision, unnecessary antibiotic prescribing and ordered choice of antibiotic prescribed

4

1

5 For URTI, among all antibiotic prescriptions issued, a binary 'prescribing status' variable was 6 created for unnecessary prescribing -versus-necessary prescribing. Among situations when prescriptions were not issued, appropriate non-prescribing was also identified, however, 7 8 there was insufficient data available to assess not prescribing when a prescription was in fact necessary. An inappropriate decision variable was created consisting of unnecessary 9 prescribing situations. Meanwhile, an appropriate decision variable was created to include 10 both necessary prescribing and appropriate non-prescribing. A binary 'decision' variable 11 included *inappropriate decisions versus appropriate decisions* among all URTI diagnoses. 12

13

As can be seen in **Figure 3-4**, note that the numerator was the same for *inappropriate decision and unnecessary prescribing.* However, the denominators differ: all diagnoses of URTI for *inappropriate decision*, and all antibiotic prescriptions for URTI for *unnecessary prescribing.*

18

Prescribing for acute rhinosinusitis was coded into: *necessary; possibly unnecessary;* and *unnecessary prescribing* using the definition by Jørgensen et al. (268). *Necessary prescribing* was classified as either fever recorded or symptom duration of at least five days. *Unnecessary prescribing* was for symptoms of less than five days and no fever recorded (268). *Possible unnecessary prescribing* was symptoms for less than five days or no fever (268) and these 236 prescriptions were excluded from further analyses.

25

Prescriptions for acute pharyngitis / tonsillitis were *necessary* in the presence of fever, otherwise *unnecessary prescribing*. As the indicators of Aboriginal and Torres Strait Islander status and RHD were considered unreliable, the 650 diagnoses of pharyngitis / tonsillitis for these patients were excluded from later multivariable analyses. Meanwhile, for AOM, prescribing was considered *necessary* for patients with fever, and for children one year and under (29), with other prescriptions classified as *unnecessary*.

32

Although the guidelines do not recommend an antibiotic for diagnoses of influenza / ILI (29),
 antibiotics may however be required in rare instances of secondary, bacterial superinfection.
 It was therefore of interest for this thesis to establish whether patients with influenza / ILI

diagnoses were receiving antibiotics. These diagnoses were included in the inappropriate 1 decision model but not included in the unnecessary prescribing model, as these diagnoses 2 would invariably be classified as *unnecessary*, and therefore cannot be modelled. The small 3 number of antibiotic prescriptions for influenza / ILI and possible unnecessary prescribing 4 5 were also excluded from ordinal choice of antibiotic prescribed, binary non-first-line prescribing, and repeat positive antibiotic prescribing models for the same reason (see 6 Figure 3-5). A brief summary of definitions is provided in the Glossary of Model Outcomes and 7 8 Variables Definitions for reference when reading later chapters, however, note the glossary does not suffice for reading this Methods chapter in full. 9


Note: All variables including outcomes should be considered to include "likely" as a prefix. The inappropriate decision model includes influenza-like illness, while unnecessary prescribing model does not. Both models include rhinosinusitis, pharyngitis / tonsillitis, and acute otitis media.

Figure 3-4: Flow chart of inappropriate decisions and unnecessary prescribing models for initial presentations of upper respiratory tract infection



Note: Upper respiratory tract infection for ordinal choice of antibiotic, binary non-first-line prescribing and the repeats present on prescription models included rhinosinusitis, pharyngitis / tonsillitis, and acute otitis media, while urinary tract infection included acute cystitis.

Figure 3-5: Flow chart of response variables and models for ordinal choice of antibiotic prescribed, non-first-line prescribing, and repeat positive antibiotic prescribing for upper respiratory tract infection and urinary tract infection analyses

Table 3-1 details how the guidelines were used to classify individual antibiotics into an 1 2 ordered variable termed choice, based on order of appearance in the guidelines for each 3 condition (29), with the denominator of all antibiotics prescribed for URTI. Penicillin hypersensitivities were allowed for, and suitable alternative antibiotics were also classified 4 5 as first-line choices where the patient had an allergy label for penicillin (332). Antibiotics which were prescribed but are not listed in Table 3-1 were classified as not recommended, 6 as was the use of penicillin hypersensitivity only options despite the patient having no record 7 8 of a relevant allergy label. A categorical variable was created for the URTI condition from which the diagnosis came. 9

- 10
- 11

13

12

Table 3-1: Antibiotic classifications for the ordered choice variable for upper respiratory tract infection conditions, based on the order of antibiotics recommended in the Therapeutic Guidelines: Antibiotic (29), by condition

Condition	Line / Choice	No penicillin hypersensitivity	Penicillin non- immediate hypersensitivity	Penicillin immediate hypersensitivity
Acute	First-line	amoxicillin	cefuroxime	Doxycycline
minosinusitis	Second-line	amoxicillin + clavulanate	doxycycline	
Acute	First-line	phenoxymethylpenicillin	cefalexin	Azithromycin
tonsillitis	Second-line	benzathine penicillin		
Acute otitis media	First-line	amoxicillin	cefuroxime	trimethoprim + sulfamethoxazole
	Second-line	amoxicillin + clavulanate	trimethoprim + sulfamethoxazole	

Note: A first-line antibiotic should be prescribed at initial consultations where prescribing is indicated. Where the antibiotic prescribed is not listed as an option for the condition diagnosed, the prescription was classified as 'not recommended'.

14

3.5.1.3 Urinary tract infection: ordinal choice of antibiotic prescribed 15

As empirical prescribing is considered appropriate for UTI (29), the focus of this research is 16 on what was prescribed and how closely this matches the guidelines for the specific patient. 17 The choice / line of agent prescribed was coded as ordinal categorical variable according to 18 the guidelines for each patient group: women, men and children (29), with the denominator 19 of all antibiotics prescribed for UTI. Although the guidelines vary for pregnant women, as 20 pregnancy was not easily detected from the data, all adult women were assumed to be non-21 pregnant. Urine cultures and susceptibility testing are mandatory for children, and recommended 22 for all men and pregnant women but are stated as not mandatory for non-pregnant women with 23 uncomplicated cystitis. A urine culture is required for diagnostic confirmation of the pathogen 24 (115). 25

Table 3-2 details how the guidelines used in the analysis for classifying individual antibiotic 1 agents into an ordinal choice variable, based on the order of their recommendation in the 2 3 guidelines for UTI (29), with the denominator of all antibiotics prescribed for UTI. First-line agents included trimethoprim for adults, and either trimethoprim with sulfamethoxazole or 4 5 trimethoprim for children, and cefalexin as second-line (29). Third-line amoxicillin with clavulanate (or nitrofurantoin for adults), and last resort norfloxacin were combined into a 6 single category when defining the ordinal line of antibiotic agent variable. Quinolones such 7 as norfloxacin are reserved as the last resort and resistance to lower-line options must be 8 confirmed prior to use (29). Antibiotics which were prescribed but are not listed in the 9 guidelines or Table 3-2 were classified as not recommended. 10

11

12 13

14

Choice	Non-pregnant Women	Men	Children >= 1 mont	h
	patient group			
	the order of antibiotics rec	ommended in	Therapeutic Guidelines: Antibiotic	: (29), by
Table 3-2:	Antibiotic classifications for	the ordered	choice variable for acute cystitis, I	based on

		-	
First-line	trimethoprim	trimethoprim	trimethoprim
Second-line	cefalexin	cefalexin	cefalexin
Third-line	amoxicillin + clavulanate	amoxicillin + clavulanate	amoxicillin + clavulanate
Third-line	nitrofurantoin	nitrofurantoin	
Last resort	norfloxacin	Norfloxacin	norfloxacin

Note: A first-line option should be the antibiotic prescribed at initial consultations. Third-line and last resort options were combined into the third ordinal level for analysis. Where the antibiotic prescribed is not listed as an option for the condition diagnosed, the prescription was classified as 'not recommended'.

15

<u>3.5.1.4</u> Additional response variables common to upper respiratory tract infection and <u>urinary tract infection</u>

Following classification of ordinal line of antibiotic agents for both URTI and UTI condition groups, a binary variable for non-first-line prescribing was also created for each (**Figure 3-**5). Non-first-line was defined to mean all antibiotics other than the first-line recommendations for the relevant condition (or patient group in the case of UTI). Culture and other pathology test results were not incorporated in classifying antibiotic choice, as these would not have been available to the prescriber at the time of the consultation.

24

A binary dichotomous variable was created for whether one or more repeats were issued on
 each antibiotic prescription, as a repeat issued without justification can itself be considered

a level of inappropriate prescribing. Descriptive analysis of antibiotics issued with repeats
occurred, and individual cefalexin prescriptions were examined to quantify the proportion
likely requiring repeats on prescriptions. Due to manufacturing pack size, a repeat may be
required for a single guideline-concordant course of cefalexin. Dummy variables for
cefalexin, or repeats issued on cefalexin prescriptions, were included where possible.

6

7

8

3.5.2 Predictor / confounder variables

9 <u>3.5.2.1 Patient-related variables</u>

All patient-related data included an individual patient identifier to enable longitudinal follow-10 11 up. Demographic variables included: date of consultations; antibiotics prescribed; reason for presentation; diagnosis (271,276,353); patient age (25,281,283,284,296,316,324,353); sex 12 (39,280,291,294,296,321,326,328,353); and approximate geographic location (270,276). 13 As CALD background status has been found to be a predictor of inappropriate prescribing 14 in a number of US studies (276,295,326), it was considered particularly relevant to consider 15 whether patient Aboriginal and Torres Strait Islander status might have been linked to more 16 or less appropriate prescribing. However, the variable for self-identification of Aboriginal 17 and Torres Strait Islander peoples was found to be poorly recorded (354,355), with positive 18 status recorded in only 0.01% of patients and information missing in 45% of all patients, and 19 was not utilised. 20

21

Comorbid condition variables (122,276,281,294,326) were combined into a single if-any 22 variable. However, COPD was not included in the comorbid condition for URTI, as different 23 24 guidelines exist for patients with this condition (29,270). Similarly, mental health conditions and drug-and-alcohol problems were also combined into a single mental health condition 25 variable. The resulting variables were coded as positive for any history of a relevant 26 condition, negative for none, and a third category for missing data (356). A binary variable 27 28 was also created for whether the patient had (any history of) an allergy label for penicillin specifically or for other penicillins (289). 29

30

Indicators of SES (270,276,295,316) included the SEIFA Index of Relative Socio-Economic Disadvantage quintiles of disadvantage based on postcode, based on census data up to and including 2011 (36,37) and ARIA for patient remoteness (290,294,317,334) based on postcode was used, also based on census data up to and including 2011 (34,35). Two categorical variables for socioeconomically disadvantaged and remote patients were created as indicators for the top two quintiles (top 40%) of disadvantage and remoteness,
respectively, as well as categories for missing data. Another variable was created to flag
patient government concession status, and a missing data category. The PHN of the
patient's residential address was coded into a variable with a missing data category.

5

6

3.5.2.2 Clinical observations and pathology data

Binary variables were created for whether a patient's temperature was recorded during the consultation and whether the temperature reading was indicative of fever (281,332) of at least 37.5C. Logical Observation Identifiers Names and Codes (LOINC) version 2.66 (357,358) aided with interpretation of pathology testing performed. Binary pathology variables were subsequently created for whether-or-not culture testing, and sensitivity testing were performed. For UTI, binary variables were created for whether any form of urine dipstick testing (blood, nitrites, or leucocyte esterase) was recorded for the patient.

14

15 <u>3.5.2.3</u> Consultation-related variables

16 Consultation-related variables received included the duration of consultation (short, 17 standard and long consultations), whether the consultation was outside of normal business 18 hours, and if the consultation was bulk-billed. However, duration of consultation and time of 19 consultation were found to be unreliable. Day-of-the-week of the consultation was included, 20 as was a variable identifying a weekend (359). The binary variable for whether the reason 21 for prescribing field had been completed by the GP, was also considered as a potential 22 predictor variable.

23

24 <u>3.5.2.4 Practice-related variables</u>

Practice-related variables included PHN (42), SEIFA (36,37) and ARIA (34,35) of the practice. A binary dichotomous variable was also created for small practices *versus* larger practices (an average of 5-14 providers working simultaneously) (271,294).

28

3.6 Analytical methods

29 30

3.6.1 Predictors of inappropriate antibiotic prescribing

31 32

Exploratory data analyses were undertaken, which alongside the relevant previous literature, help identify the most appropriate choice of predictor variables. STATA release 16 was used for analysis (360).

The data structure for modelling purposes was potentially very complicated, with patients 1 within providers within practice clusters. Moreover, there are also repeat visits at the 2 3 individual patient to individual provider level, (as well as repeated individual patients seen by multiple different providers), in addition to repeated visits at the provider-practice level. 4 5 As a result, the assumption of independence across observational units is very likely to be violated (361,362) and therefore must be taken into account in estimation. Indeed, due to 6 this dependence (and three-level hierarchical structure), standard multivariable logistic 7 regression was not deemed appropriate. Instead, a three-level random-effects logistic 8 hierarchical model was considered ideal to allow for unobserved heterogeneity (363) at each 9 level, including for example repeated measures. 10

Although different terminology exists across various disciplines, note that we used the terms: 11 fixed-effects models for when unobserved heterogeneity is treated as fixed (essentially 12 13 including dummy variables for them), while random-effects models treat the unobserved heterogeneity as random effects (364). Mixed-effects refers to the combination of the two. 14 Mixed-effects logistic regression modelling (generalised linear mixed modelling with random 15 effects) was used to systematically identify the variables associated with several different 16 definitions of appropriate antibiotic prescribing.^{2,3} As antibiotic prescribing may be seasonal, 17 and is certainly for URTI (366-368), and as prescribing behaviours may change over time, 18 both seasonal dummy variables and time effects were also allowed for. 19

20

Much of the existing literature does not appear to use, or allow for, multiple types or levels 21 of *unobserved heterogeneity*, such as (simultaneously) at the patient, GP, and practice 22 levels (22,48,276,277,281,286,295,296,315,323,325,326,329,330,332,340-342). This is 23 potentially important, as, on the assumption that there are multiple observations at 24 potentially each level, then if there is indeed unobserved heterogeneity at these levels, and 25 this is ignored in estimation, then spurious results are likely to follow. This is simply a result 26 of the fact that an incorrect, or mis-specified, likelihood is being maximised. For example, 27 28 the independence of observations is an assumption underlying generalised linear models used (361,362); however, many authors do not allow for repeated measures such as 29

² Although the intention for all models was to include both *random intercepts* and *random coefficient* / slope at each level, due to convergence issues, it was only possible to include the former. Where both practice and provider levels could not be accounted for, a random intercept for the unique combination of practice and provider was included (365).

³ While random intercepts for practice-related variables, namely practice size, were intended for inclusion at the practice level, this was not feasible due to effect size so small that CIs were unable to be calculated, so were therefore included at the patient level instead of practice level.

multiple visits by the same patient, consultations by the same GP or at the practice level. 1 Clearly, once these effects have been allowed for, the approach explicitly allows for 2 3 dependence across these factors, such that conditional on these, the observations are now, indeed, independent as the approach explicitly requires. Although highly complex and 4 5 computer-intensive, fortunately recent improvements in computing power have made the allowance for / of such unobserved heterogeneity at multiple levels more feasible. 6 Accordingly, this thesis allows for multiple levels of unobserved heterogeneity, which should 7 significantly reduce the likelihood of any estimation bias resulting for erroneous omission of 8 such. 9

10

Some studies mixed models with effects other have used random 11 (284,294,297,321,326,328,363,369-372), however, it is unclear how many types or levels of 12 unobserved heterogeneity were allowed for. Some authors used random effects to allow for 13 clustering at the individual patient-level only (294,363) or the clinician level only (372). Some 14 studies generalised estimating equations allow for 'clustering' 15 used to (270,274,276,280,373), however, typically did not allow for the practice level. Poisson 16 regression models were also utilised by some authors (271,374). Some studies used fixed 17 effects only but added GPs as clusters in the model (276,322), which does not rectify for 18 ignoring unobserved heterogeneity. The method used in this thesis controlled directly for a 19 wide range of unobserved effects simultaneously, including at the GP and practice levels, 20 and therefore provides a more nuanced approach. 21

22

As described by Stroup (375), generalised linear mixed models are of the form described below in **Figure 3-6**, where **y** | **b** is the distribution of the data or observations, **y**, conditional on the random effects, **b**. Furthermore, **µ** and **R** are the mean and variance of the population, respectively. Despite the added complexity, there was sufficient variation in the data, and enough degrees of freedom, to adequately identify all these random components in a model. Subject to the nature of the dependent model, both ordinal and binary multilevel random effects models were considered.

30

A brief summary of statistical terminology is also provided in the **Glossary of Multilevel Mixed-effects Modeling Terminology**. However, this glossary does not suffice for reading this chapter in full.

Observations	Link	Linear Predictor	Mean Modeled by ^a
y b ~ G(μ, R) G denotes general distribution	$\eta = g(\mu \mid b)$	$X\beta + Zb \ b \sim N(0,G)$	$\hat{\boldsymbol{\mu}} = h(\hat{\boldsymbol{\eta}})$ $= h(\boldsymbol{X}\hat{\boldsymbol{\beta}} + \boldsymbol{Z}\hat{\boldsymbol{b}})$

^a $h(\mathbf{\eta})$ denotes *inverse link* function, for example, $h(\mathbf{\eta}) = e^{\mathbf{\eta}}/(1+e^{\mathbf{\eta}})$ for logit link.

Figure 3-6: Illustration of the form of the generalised linear mixed model, using direct quotation
 from Stroup WW. Generalized linear mixed models: modern concepts, methods and
 applications. 1st ed. Boca Raton, FL: CRC Press, an imprint of Taylor and Francis;
 2012. Table 1.4, Typology of Linear Models; p. 20. (375)

7 The denominator for all models common to URTI / UTI was all systemic antibiotic prescriptions for each respective condition group: URTI or UTI. However, note for all URTI 8 9 models, possibly unnecessary prescriptions were excluded, as were situations where antibiotics were indicated but not prescribed. Antibiotics prescribed for influenza / ILI were 10 included only in the inappropriate decision model for URTI (Figure 3-5). Due to small sample 11 sizes, the denominator of all antibiotics prescribed for URTI (for both necessary and 12 13 unnecessary prescribing) was chosen. A flow chart of models common to URTI and UTI is provided in Figure 3-6⁴. 14

15

1

6

A range of specifications were experimented with, including allowing for interactions between such variables as patient age and gender. Final model specifications followed multiple specification and model selection tests and metrics, including likelihood ratio tests for joint (in)significance and Akaike Information Criterion, and to a greater extent, Bayesian Information Criteria (376,377). Brant testing of the parallel regression assumption was also performed for ordinal models to establish whether antibiotic lines can be considered equidistant (378,379).

23

While the primary aim of the modelling was essentially to most accurately identify / estimate the predictor coefficients, a secondary aim was to interpret the final models to allocate the source of variance unexplained by fixed effects (380,381). This involved calculating the intraclass correlation (ICC) to determine how much variation unexplained by fixed effects, each level in the model was responsible for (380-382). This specifically related to betweenlevel ICC for calculating the variation explained by random effects at each of the provider and practice levels, for comparison. Where the data was unsuited to a three-level model,

⁴ For information regarding the variables included in each base model, please see Appendix B.6.4.

two-level models with patient level and either the practice or the provider level were used to
calculate ICC, and then compared to give an approximate estimation of the difference in
source of variance explained by random effects between each level.

4

5 Due to the nature of the multilevel model, one cannot interpret effects directly from adjusted odds ratios (AORs) appearing in the final model, as one would do in a single-level model, 6 as the effect may vary across members of higher-levels in the model (383). Although there 7 are multiple approaches to the estimation of various effects in multilevel models, average 8 marginal effects (AMEs) were predominantly used for ease of exposition. Marginal effects 9 at representative values (MERs) were calculated between covariates but are only presented 10 in text to illustrate effect modifications identified. Adjusted predictions at the means (with all 11 12 other covariates held constant at sample means) and marginal predicted mean were also used. MERs were compared to AMEs with limited notable difference identified.⁵ For 13 information regarding these values, please see the Glossary of Multilevel Mixed-effects 14 Modeling Terminology. 15

- 16
- 17 18

3.6.2 Trends in inappropriate antibiotic prescribing

Having identified predictors of inappropriate prescribing at an individual level, are there
aggregate trends over time for URTI and UTI (384-389). All prescribing-related outcomes
were analysed, firstly, for all URTI or UTI diagnoses altogether, then for each URTI condition
individually, and for UTI by each patient group (Figure 3-7).

23

For each outcome within each sub-population of interest, rates were calculated as aggregate count data over units of time (weekly, fortnightly, monthly, quarterly, half-yearly). The denominators remained the same as in previous predictor identification analyses, please see **Appendix B.7** for more details. Graphical depictions of outcomes as rates were examined and compared using different units of time. The most appropriate unit of time was then selected, using a trade-off between "noise" and the number of data points, resulting in monthly mean rates for all trends analyses (390).

⁵ Please see the Appendix to each (URTI/UTI) predictors chapter. Full results are also available on request.



1

4

Figure 3-7: Groups of interest in trends analyses for upper respiratory tract infection, including
 influenza / influenza-like illness and urinary tract infection, by condition group

5 For the purposes of basic smoothing and to eliminate any seasonal effects, a simple moving average was utilised (391). Different time windows were experimented with these, and 6 invariably simply graphical depiction was used to select what was deemed the most 7 appropriate. A six-monthly simple moving average was tested initially, being half the twelve-8 9 month time period in use (392). Among several others tried, a five-monthly time period was finally selected as this was not noticeably different from six-monthly but with slightly less 10 noise. This use of predicted values obtained from moving average smoothing reduces the 11 possibility of distortion using of end points only, which may be extreme and not be 12 representative of the entire sample. There were many outcomes analysed for URTI and UTI, 13 as well as for each URTI condition and each UTI patient group independently, as well as 14 individual antibiotics use for each condition. Only the most notable trends identified are 15 presented for each condition, such as major outcomes of non-first-line or second-line 16 prescribing, and interesting trends in the prescribing of individual antibiotics. Additional 17 results are presented in the appendix to each trends chapter. 18

19

The graphical depiction of the final smoothed, moving average outcome rates for each outcome within each sub-population followed. Fitted lines, 95% confidence intervals (CIs), and Lowess rates were added to the smoothed rates (393-396), as seen in **Figure 3-8** below, which is included purely as an example.



Note: Denominator all patients prescribed an Antibiotic for initial presentations of urinary tract infection.

1 2 3

4

Figure 3-8: Example simple moving average of outcome rate, with fitted lines, 95% confidence intervals, and Lowess rates

5 Simple linear regression was then used to test for a significant trend (330,397,398). As 6 URTIs are subject to seasonal variation (366,367), influenza notifications from WA were 7 incorporated into the linear regression models to check for any remaining seasonal effects 8 following the smoothing process (399). An insignificant influenza season result would imply 9 that the smoothing technique was sufficient, and *vice versa*.

10

The mean rate over study period was then calculated for each outcome using the smoothed moving averaged data. The difference between the first and last predicted values over the study period formed the basis of the estimated percentage change.

14

15 **3.7** Synthesise findings

16

The synthesis of findings involved interpretation of results obtained, in discussion with the steering committee, with careful consideration of the complex, multifaceted situations in which antibiotic prescribing occurs. This helps provide context to, and a more comprehensive understanding of, the results.

The methodological development of this research has also aided in the identification of further avenues for study. The evaluation of the relative strengths and weaknesses of various methods employed was also undertaken wherever possible, such as comparison with the results from studies utilising other methods.

5

CHAPTER 4 PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR UPPER RESPIRATORY TRACT INFECTION

3 4

5 6 7

4.1 Introduction

URTI encompasses several common conditions for which inappropriate antibiotic 8 prescribing is known to occur (14,17,18,87), due in part to their frequent viral aetiology. URTI 9 is also responsible for the most antibiotic utilization of any condition group commonly 10 presenting in primary care (75). URTI was therefore a key focus area of this project. The 11 12 national guidelines titled, Therapeutic Guidelines: Antibiotic (the guidelines), were developed to recommend effective antibiotics, minimise antibiotic resistance and limit side 13 14 effects (29). Reducing unnecessary use of antibiotics for URTI is considered crucial to managing the spread of antibiotic resistance (6,13,16,49) and improving quality of care. 15

16

The aim of this chapter was to quantify the prescribing of guideline non-conforming systemic 17 antibiotics in Australian general practice for patients presenting with initial episodes of URTI 18 (29). The aim was also to elucidate patient-, practice- or consultation- related predictors of 19 20 inappropriate antibiotic prescribing for patients presenting with URTI. The focus was on assessing any association of inappropriate prescribing with patient age, patient comorbid 21 conditions, and practice remoteness / accessibility. This inherently involved the identification 22 of inappropriate prescribing for patients presenting for initial episodes of care for URTI. As 23 detailed in the Background chapter, the guideline advice on diagnosis, treatment, indications 24 for prescribing, and recommended antibiotics, differs by URTI condition (29). Assessing 25 compliance with the guidelines can help provide new insights and identify opportunities for 26 interventions and policy. 27

28

30

29 4.2 Specific methods

A multinominal categorical variable was created for prescribing-related decisions including: unnecessary prescriptions, necessary prescriptions, and appropriate non-prescribing. It was considered that there were insufficient data to assess under-prescribing when a prescription may have been necessary. Splitting the above variables by appropriate / necessary and inappropriate / unnecessary, a binary dichotomous variable was created for inappropriate versus appropriate decisions (**Figure 4-1**). A binary dichotomous variable was also created referred to as the prescribing status variable. A discrete categorical variable was created to
 indicate which URTI condition the diagnosis related to.

3

The standard used for assessment of prescribing was version 15 of the guidelines (29). As 4 5 explained in chapter 3, prescribing for acute rhinosinusitis / non-specific URTI was coded into: necessary, possibly unnecessary, and unnecessary prescriptions using definitions by 6 Jørgensen et al. (268). If fever was present or symptoms had lasted at least five days (using 7 diagnosis onset date) prescribing was considered to be likely necessary (268). If symptoms 8 were present and symptoms lasted at least five days and no fever was recorded, prescribing 9 was considered to be unnecessary (268). Possibly unnecessary prescribing was classified 10 as symptom duration of less than five days or no fever (268), and was excluded from 11 multivariable analyses (see Methods chapter). Prescriptions for acute pharyngitis / tonsillitis 12 were considered necessary in the presence of fever, otherwise they were considered 13 unnecessary. As the indicators of Aboriginal and Torres Strait Islander status and RHD were 14 considered unreliable, decisions for acute pharyngitis / tonsillitis provided to Aboriginal and 15 Torres Strait Islander peoples and patients with RHD were excluded from multivariable 16 analyses. It was considered appropriate to prescribe for AOM for children one year or under, 17 or those with fever recorded. 18

19

20 The choice of antibiotic prescribed was coded as an ordinal categorical variable according to the guidelines: including first-line, second-line and not recommended antibiotics (29). 21 First-line agents included amoxicillin, phenoxymethylpenicillin, and amoxicillin, for acute 22 rhinosinusitis, acute pharyngitis / tonsillitis and AOM, respectively (Table 4-1) (29). Penicillin 23 24 sensitivity was allowed for, such that use of a suitable agent in a patient with an allergy label for penicillin was recorded as first-line. Antibiotics prescribed for each increasing choice 25 were compared to first-line prescriptions, with the denominator of all antibiotics prescribed 26 for URTI. A binary variable for non-first-line prescribing was also created, defined as all non-27 first-line agents. 28

29

A binary dichotomous variable was created for whether one or more repeats were issued on the antibiotic prescription, as issuing a repeat without indication can be considered a level of inappropriate prescribing. Individual cefalexin prescriptions were assessed by strength, dosage and duration to identify prescriptions requiring a repeat.

- 34
- 35

1

2 3

4

Table 4-1: Choice of antibiotic for upper respiratory tract infection conditions, by condition, and by patient allergy label for penicillin, based on the order of recommendations and penicillin hypersensitivity options listed within Therapeutic Guidelines: Antibiotic (29)

Condition	Line / Choice	No penicillin hypersensitivity	Penicillin non- immediate hypersensitivity	Penicillin immediate hypersensitivity
Acute	First-line	amoxicillin	cefuroxime	doxycycline
minosinusius	Second-line	amoxicillin + clavulanate	doxycycline	
Acute	First-line	phenoxymethylpenicillin	cefalexin	azithromycin
tonsillitis	Second-line	benzathine penicillin		
Acute otitis	First-line	amoxicillin	cefuroxime	trimethoprim +
inoulu	Second-line	amoxicillin + clavulanate	trimethoprim + sulfamethoxazole	

Note: A first-line antibiotic should be prescribed at initial consultations where prescribing is indicated. Where the antibiotic prescribed is not listed as an option for the condition diagnosed, the prescription was classified as 'not recommended'. For more information, please refer to the Methods chapter.

5

6 Mixed-effects logistic regression modelling (generalised linear mixed modelling with random effects) was utilised to identify variables associated with inappropriate decisions, 7 8 unnecessary antibiotic prescribing, increasing choice / order of antibiotic prescribed, non-9 first-line prescribing, and repeat prescribing (Figure 4-1). The model for inappropriate decisions versus likely appropriate decisions, with the denominator being all diagnoses of 10 URTI including influenza / ILI, is presented as Model 0. Model 1 is unnecessary prescriptions 11 versus necessary prescriptions, with denominator of all systemic antibiotic prescriptions for 12 URTI excluding influenza / ILI. The model for the outcome of ordinal line of antibiotic agents 13 is then presented as Model 2, with denominator of all systemic antibiotic prescriptions for 14 URTI excluding influenza / ILI. Additional modelling was performed including but not limited 15 to mixed effects logistic regression for non-first-line prescribing (Model 3, Appendix C.8), 16 and repeats being issued on antibiotic prescriptions (Model 4, Appendix C.9).⁶ 17

⁶ For more information, please refer to the Methods chapter (Chapter 3).

Figure 4-1: Flow chart of variables, numerators and denominators used in models for initial presentations of upper respiratory tract infection



Flow Chart of Inappropriate Decisions, Unnecessary Prescribing, Ordinal Choice / Line of Antibiotic, Non-first-line Antibiotic Prescribing, and Repeat Positive Prescribing

Note: For models 0 and 1, all variables should be considered to include "likely" as a prefix. Model 0 is the only model to include influenza-like illness. All models include upper respiratory tract infection, which is defined as: rhinosinusitis, pharyngitis / tonsillitis, and acute otitis media.

By encompassing random effects, this allowed for multiple visits by the same patient, and
multiple consultations with or without prescriptions by the same GP / provider. The intention
was to develop three-level hierarchical models with patient, provider and practice levels.
Due to the clustering / non-independence of observations, random intercepts were included,
where possible, for both practice ID and provider ID.

7

1

As the guidelines never recommended prescribing an antibiotic for a standalone diagnosis of influenza / ILI (29), antibiotic prescriptions for influenza / ILI were not included in the models for unnecessary prescribing, ordinal choice, non-first-line prescribing or repeat positive prescribing. Diagnoses for influenza / ILI were included in the model for inappropriate decisions, however. Possibly unnecessary prescriptions and prescriptions for influenza / ILI were excluded from the ordinal line of agent and binary non-first-line prescribing models.

15

16 4.3 Results

17

There were 112,734 diagnoses of URTI, for initial consultations, during the study period 1 18 January 2012 to 30 June 2017, inclusive, with an antibiotic prescribing rate of 46% for these 19 URTI diagnoses (Table 4-2, Table 4-3). By patient age, consultations and antibiotic 20 prescribing peaked at five and six years, respectively. Penicillins with extended spectrum 21 22 represented 28% of all antibiotics prescribed, followed by penicillin combinations (23%), beta-lactamase sensitive penicillins (20%), macrolides (14%), and first-generation 23 24 cephalosporins (11%) (Table 4-4, Appendix C.2). The reason for prescribing field was 25 completed in 12% of antibiotic prescriptions.

26

A temperature reading was recorded in 35% consultations, within which 24% of these were indicative of fever of at least 37.5C. While laboratory pathology results were not incorporated in modelling, as they would not have been available at the time of decisions regarding prescribing, a result for culture testing was available in 0.2% (n=272) consultations, and within these, 85% of cultures were positive for growth. There were no results for sensitivity susceptibility testing available for any initial episodes of care for URTI, despite the availability of results for other diagnoses.

- 34
- 35

Frequency table of patient characteristics for all patients, and patients with initial episodes of care for upper respiratory tract infection (column percentage) Table 4-2:

	All patients	All visits	Initial care episode URTI	Initial care episode acute rhino- sinusitis	Initial care episode acute pharyngitis / tonsillitis	Initial care episode AOM	Initial care episode influenza / ILI
Characteristic	n=791,280	n=1,925,985	n=112,734	n=62,236	n=28,899	n=19,424	, n=2,175
Patient			·		- · ·		
Female gender, %	52.8	59.1	54.6	54.3	58.4	50.6	49.6
	38.1	48.0	25.0	26.8	25.7	17.1	36.2
Mean age, years (s.d.) *	(22.07)	(23.69)	(20.09)	(21.19)	(17.33)	(18.19)	(18.78)
Patient's Primary Health							
	12.0	32.8	18.7	1/ 0	22.0	24.5	21.8
Perth South	33.1	33.5	45.5	48.1	43.4	40.1	50.1
Perth North	34.8	31.8	33.4	34.8	31.4	32.8	25.6
Interstate	2.2	0.5	0.9	0.8	09	0.9	1 1
Missing	1.5	1.3	1.5	1.4	1.5	1.7	1.5
Patient concession status	34.1	35.6	17.1	17.1	16.8	18.0	13.2
positive. %	0.112	00.0			2010	2010	10.12
Comorbid condition positive. %	1.9	10.2	15.4	16.0	14.8	14.1	17.7
Missing	3.4	2.5	3.3	4.2	2.3	2.2	3.7
Mental health condition	12.9	28.7	12.3	12.5	14.5	7.8	18.5
positive, %							
Missing	3.4	2.5	3.3	4.1	2.3	2.2	3.7
Patient remote, % (remote &	6.0	4.4	4.3	3.0	5.8	5.9	5.0
very remote Australia, ARIA)							
Missing	1.1	1.0	0.6	0.5	0.7	0.7	0.9
Patient disadvantaged, % (top 40	11.7	10.3	9.5	8.8	10.6	9.7	11.0
percentiles most disadvantage, SEIFA IRSD)							
Missing	1.5	1.3	12.4	11.6	12.2	15.4	12.9
Penicillin allergy label positive %			5.3	5.4	5.5	4.6	5.6
Multiple URTI episodes, % (same			77.0	77.5	76.9	75.0	83.1
or other URTI condition)							
Consultation							
Temperature recorded, %,			34.5	36.6	34.2	28.5	31.3
(% of which fever positive			(24)	(19)	(32)	(31)	(47)
>=37.5C)							
Culture performed, %,			0.2	0.1	0.5	0.2	0.3
(% of which positive for growth			(34)	(18)	(32)	(85)	(43)
of any pathogen)							
Sensitivity performed, %			0.0	0.0	0.0	0.0	0.0
Weekend consult, %			10.2	10.3	10.3	10.4	4.8
Prescription & classification							
Prescribing rate, %,			46.3	32.2	70.9	58.5	11.6
(prescriptions over diagnoses)							
Multinominal prescribing							
decision, % (denominator of all							
alagnoses per condition/							
purposes only see Outcomes)							
Appropriate a set outcomes)			522	67.8	28.4	3/1 2	88 /
Appropriate non-prescribing			7.7	3.1	9.1	21.2	0.0
Necessary prescribing			20.2	207	60.2	44.0	11.0
Unnecessary prescribing			39.2	28.7	60.2	44.6	11.6
Excluded (insufficient information)			0.6	0.0	2.3	0.0	0.0
Possibly Unnecessary Prescribing (excluded)			0.2	0.4	0.0	0.0	0.0

	All patients	All visits	Initial care episode URTI	Initial care episode acute rhino- sinusitis	Initial care episode acute pharyngitis / tonsillitis	Initial care episode AOM	Initial care episode influenza / ILI
Characteristic	n=791,280	n=1,925,985	n=112,734	n=62,236	n=28,899	n=19,424	n=2,175
<u>Outcome</u>							
Binary Inappropriate Decision % (denominator of all diagnoses per condition/ condition group) #							
Appropriate decision (numerator: appropriate non-prescribing + necessary prescribing)			60.0	70.9	37.5	55.4	88.4
Inappropriate decision (numerator: Unnecessary prescribing)			39.2	28.7	60.2	44.6	11.6
Excluded total			0.7	0.4	2.3	0.0	0.0

Table 4-4: Frequency table of active ingredients prescribed for initial presentations of upper respiratory tract infections (denominator all antibiotics including influenza / influenza-like illness)

* Note: Patient age was missing for n=10 initial episodes of care for URTI and these observations were excluded from multivariable analyses.

Note: insufficient information was available to assess under-prescribing (not prescribing despite indication).

Active ingredient	Frequency	Percent	Cumulative Percent
Amoxicillin	14,722	28.22	28.22
Amoxicillin with clavulanate	11,843	22.7	50.92
Azithromycin	896	1.72	52.64
Benzathine benzylpenicillin	2	0	52.64
Cefaclor	939	1.8	54.44
Cefalexin	5,701	10.93	65.37
Ceftriaxone	4	0.01	65.38
Cefuroxime	247	0.47	65.85
Ciprofloxacin	108	0.21	66.06
Clarithromycin	1,773	3.4	69.45
Clindamycin	43	0.08	69.54
Dicloxacillin	5	0.01	69.55
Doxycycline	481	0.92	70.47
Erythromycin	1,498	2.87	73.34
Flucloxacillin	105	0.2	73.54
Gentamicin	4	0.01	73.55
Minocycline	25	0.05	73.6
Nitrofurantoin	2	0	73.6
Norfloxacin	4	0.01	73.61
Phenoxymethylpenicillin	9,060	17.37	90.97
Procaine benzylpenicillin (procaine penicillin)	1,156	2.22	93.19
Roxithromycin	3,084	5.91	99.1
Tobramycin	2	0	99.1
Trimethoprim	47	0.09	99.19
Trimethoprim with sulfamethoxazole	420	0.81	100
Total	52,171	100	

	Initial care episode URTI	Antibiotic Prescribed for URTI including influenza / ILI	Antibiotic prescribed for rhinosinusitis	Antibiotic prescribed for pharyngitis / tonsillitis	Antibiotic prescribed for AOM	Antibiotic prescribed for influenza / ILI	No antibiotic prescribed for URTI including influenza/ILI
Characteristic	n=112,734	n=52,171	n=20,064	n= 20,500	n=11,354	n=253	n=60,563
Patient							
Female gender, %	54.6	55.5	55.4	58.2	50.6	54.6	53.9
Mean age, years (s.d.) *	25.0 (20.09)	25.8 (19.81)	31.4 (21.74)	24.8 (16.73)	17.6 (18.12)	35.7 (19.71)	24.4 (20.31)
Patient's Primary Health Network, %							
Country WA	18.7	18.5	12.4	23.7	20.0	14.6	18.9
Perth South	45.5	46.0	47.8	44.1	45.9	60.5	45.1
Perth North	33.4	33.1	37.7	29.8	31.4	23.7	33.7
Interstate	0.9	0.9	1.0	1.0	0.9	0.0	0.8
Missing	1.5	1.5	1.2	1.5	1.8	1.2	1.5
Patient concession status positive, %	17.1	17.3	18.8	16.3	16.7	15.0	17.0
Comorbid condition positive, %	15.4	15.4	17.9	13.8	13.6	20.6	15.4
Missing	3.3	3.9	5.9	2.3	3.1	7.9	2.8
Mental health condition positive, %	12.3	12.8	14.2	13.7	8.2	24.5	11.9
Missing	3.3	3.9	5.9	2.31	3.1	7.9	2.8
Patient remote positive, % (remote & very remote Australia, ARIA)	4.3	4.6	2.2	6.3	5.6	6.7	4.0
Missing	0.6	0.6	0.4	0.6	0.6	0.0	0.6
Patient disadvantaged positive, % (top 40% percentiles of most disadvantage, SEIFA IRSD)	9.5	9.0	6.5	11.2	9.4	10.7	9.8
Missing	12.4	11.8	11.2	11.1	14.0	6.3	13.0
Patient penicillin allergy label positive %	5.3	6.4	7.4	5.8	5.8	12.3	4.3
Multiple URTIs positive patient, %	77.0	77.0	78.5	76.9	74.3	77.9	77.0
Consultation							
Temperature recorded, %, (% of which >=37.5C)	34.5 (8)	34.8 (11)	37.5 (10)	36.0 (13)	27.8 (9)	37.2 (25)	34.2 (18)

 Table 4-3:
 Frequency table of patient, consultation, and prescription characteristics for antibiotic prescriptions issued at initial episodes of care for upper respiratory tract infection, including by condition (column percentage)

	Initial care episode URTI	Antibiotic prescribed for all URTI diagnoses including influenza / ILI	Antibiotic prescribed for rhinosinusitis	Antibiotic prescribed for pharyngitis / tonsillitis	Antibiotic prescribed for AOM	Antibiotic prescribed for influenza / ILI	No antibiotic prescribed for URTI including influenza/ILI
Characteristic	n=112,734	n=52,171	n=20,064	n= 20,500	n=11,354	n=253	n=60,563
Consultation							
Culture performed. %. (% of which had positive	0.2 (34)	0.3 (36)	0.1 (14)	0.6 (34)	0.2 (88)	0.8 (50)	0.2 (31)
growth for any pathogen)							
Sensitivity performed, %	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Weekend consult, %	10.2	11.4	12.3	10.8	10.9	5.9	9.2
Prescription							
Prescribing rate, %, (prescriptions over diagnoses)		46.3	32.2	70.9	58.5	11.6	
Reason for prescribing recorded, %		5.4	2.0	10.4	9.8	0.1	
Outcomes (denominators :all antibiotics prescribed for specific condition / condition group)							
Unnecessary/necessary prescribing % †							
Likely unnecessary prescribing, %		84.7	89.1	84.9	76.2		
Likely necessary prescribing, %		14.1	9.7	12.8	23.8		
Excluded, %		1.4	1.2	2.3	0.0		
Choice/line of antibiotic prescribed, % †							
First-line, %		39.3	32.4	42.4	46.8		
Second-line, %		17.0	25.2	0.0	32.8		
Not recommended, %		39.4	42.4	57.6	20.6		
Excluded, % (influenza, n=235)		0.5	0.0	0.0	0.0		
Non-first-line antibiotic prescribed, % †							
First-line, %		39.3	32.4	42.4	46.8		
Non-first-line (non-first-line), %		60.2	67.6	57.6	53.2		
Excluded, % (influenza, n=235)		0.5	0.0	0.0	0.0		
Repeat(s) issued on prescription, % †							
Positive for one or more repeats		32.3	36.0	25.9	37.3		
Negative (no repeats on prescription)		67.7	64.0	74.1	62.7		
Excluded, % (influenza, n=235)		0.5	0.0	0.0	0.0		

Note: Patient age was missing for n=10 initial episodes of care for URTI and these observations were excluded from multivariable analyses. † Note: As antibiotics are not recommended for standalone diagnoses of influenza/ILI, prescriptions for this were excluded from models for these outcomes.

*

2 Inappropriate decisions (i.e. unnecessary prescribing) occurred in 39% situations of the total 112,734 URTI diagnoses, compared to 61% of appropriate decisions (comprising both 3 appropriate non-prescribing and appropriate prescribing situations). Pharyngitis had the 4 5 highest proportion of inappropriate decisions (60%), followed by AOM (45%), rhinosinusitis (29%) and influenza / ILI (12%). Pharyngitis also had the lowest rate of appropriate non-6 prescribing (28%), followed by AOM (34%), rhinosinusitis (68%), and influenza / ILI had the 7 highest (88%), as seen in Figure 4-2. Children 0-8 years had lower proportions of 8 9 inappropriate decisions (29%) than other age groups, which increased by age group, ranging from 39% for 9-21 years to 45% for patients 35 years and over (Appendix C.2: Table 10 11 C-2).



1





16

The strong majority of all 52,171 antibiotic prescriptions were unnecessary (85%). Of the 1 2 20,064 antibiotic prescriptions for acute rhinosinusitis / non-specific URTI, 89% were 3 unnecessary and 10% were necessary (Figure 4-2). Prescribing for acute pharyngitis / tonsillitis was unnecessary in 85% of the total 20,500 prescribing situations, appropriate in 4 5 13%. The lowest proportion of unnecessary prescribing was for AOM, at 76% of the total 11,354 prescriptions for the condition. While insufficient information was available to assess 6 under-prescribing, there was one notable diagnosis of tonsillitis with scarlet fever with no 7 antibiotic prescribed. 8

9

Children aged 0-8 years had the lowest proportion of unnecessary prescriptions among all prescriptions per age group (69%). This was followed by 85% for the 9-21 year-old age group, 91% for patients 22-34 years, to a maximum of 94% for patients 35 years and over receiving unnecessary prescriptions among all antibiotic prescriptions (**Table 4-5**). Unnecessary prescribing increased proportionally with increasing age.

16 17

15

 Table 4-5:
 Frequency table of patient age group by necessary / unnecessary antibiotic prescribing for initial presentations of upper respiratory tract infection

	Necessary Prescribing	Unnecessary Prescribing	Excluded	Total
Patient Age Group				
0-8 yrs	3,990	9,266	99	13,355
	29.88	69.38	0.74	100
9-21 yrs	1,226	7,633	149	9,008
	13.61	84.74	1.65	100
22-34 yrs	1,242	14,094	148	15,484
	8.02	91.02	0.96	100
25± vrs	806	13 206	72	14 084
551 813	500 E 72	13,200	0.51	100
	5.72	93.77	0.51	100
Missing	0	2	238	240
	0	0.83	99.17	100
Total	7,264	44,201	706	52,171
	13.92	84.72	1.35	100



18 19

Among antibiotic prescriptions for these initial presentations of URTI, 60% were non-firstline. Rhinosinusitis had the highest proportion of non-first-line among all antibiotics prescribed per condition (68%), followed by pharyngitis / tonsillitis (58%), and AOM (53%) (**Figure 4-3**, **Table 4-6**). AOM was the only condition with the highest proportion of antibiotics being first-line, and the second-highest proportion being second-line, and so forth, in the order recommended in the guidelines (29).



1 2 3

4 5

6

7

Bar graph of ordinal choice of antibiotic prescribed for initial diagnoses of upper respiratory tract infection, by condition

Table 4-6: Frequency table of first-line and non-first-line antibiotic prescriptions for initial presentations of upper respiratory tract infection, by upper respiratory tract infection condition

				0
	First-line	Non-first-line	Excluded	Total ⁸
URTI Condition				
Rhinosinusitis	6,504	13,560	0	20,064
	32.42	67.58	0	100
Pharyngitis/Tonsillitis	8,683	11,817	0	20,500
	42.36	57.64	0	100
Acute Otitis Media	5,317	6,037	0	11,354
	46.83	53.17	0	100
Influenza/ILI	0	0	253	253
	0	0	100	100
Total	20,504	31,414	253	52,171
	39.3	60.21	0.48	100

9 10 percentage

Of antibiotics not recommended in the guidelines for each respective condition, macrolides, first-generation cephalosporins and penicillins dominated (**Appendix C.2: Table C-3**). Betalactamase inhibitors are not recommended for acute pharyngitis / tonsillitis but were commonly used. There were no prescriptions of higher generation cephalosporins.

6

1

With respect to ordered choice of antibiotic prescribed, first-line prescribing decreased with 7 increasing patient age, meanwhile antibiotics not recommended in the guidelines (29) 8 increased in proportion with increasing patient age (Table 4-7). Non-first-line antibiotic 9 prescribing also increased with increasing age. Within necessary prescribing situations 10 (n=7,264), 46% were first-line, 20% were second-line, and 33% were not recommended 11 (Table 4-8). Within unnecessary prescribing (n=43,946), 43% were first-line, 17% second-12 line and 40% were not recommended (Table 4-8). For a summary of how the antibiotic 13 prescribing for URTI analysed here compares with prominent quality indicators, please see 14 15 Appendix G.1.

16

17	
18	

Table 4-7:Frequency table of ordinal choice of antibiotic prescribed, by patient age group, for
patients with initial presentations of upper respiratory tract infection

	Not				19
	First-line	Second-line	Recommended	Excluded	Total
Patient Age Group					
0-8 yrs	6,443	2,110	4,781	21	13,355
	48.24	15.8	35.8	0.16	100
9-21 yrs	3,942	1,303	3,724	39	9,008
	43.76	14.46	41.34	0.43	100
22-34 yrs	5,917	2,159	7,338	70	15,484
	38.21	13.94	47.39	0.45	100
35+ yrs	4,092	3,142	6,727	123	14,084
	29.05	22.31	47.76	0.87	100
Missing	110	33	97	0	240
	45.83	13.75	40.42	0	100
Total	20,504	8,747	22,667	253	52,171
	39.3	16.77	43.45	0.48	100

20

Of all antibiotic prescriptions for URTI, 32% of these antibiotic prescriptions were issued with one or more repeats issued on them. Proportionally, repeats were most commonly issued on non-first-line prescriptions, to patients with comorbid conditions or mental health conditions, and most frequently to young children (**Appendix C.4.1**). 1 Table 4-8: Frequency table of unnecessary and necessary antibiotic prescribing for initial 2 presentations of upper respiratory tract infection, by ordinal choice of antibiotic 3 prescribed

					1
Unnecessary / Necessary Prescribing	First-line	Second-line	Not Recommended	Total	
Necessary Prescribing	3,368	1,485	2,411	7,264	Кеу
	46.37	20.44	33.19	100	frequency row percentage
Unnecessary Prescribing	18,852	7,311	17,783	43,946	-
	42.9	16.64	40.47	100	
Total	22,220	8,796	20,194	51,210	
	43.39	17.18	39.43	100	

5

Of the antibiotics commonly used for URTI, prescriptions for amoxicillin and amoxicillin with
 clavulanate represented over 60% of prescriptions issued with repeats (Table 4-9). For
 details regarding repeats being issued on antibiotic prescriptions for influenza / ILI, see

- 9 Appendix C.4.1: Table C-11.
- 10

Table 4-9: Frequency table of active ingredients prescribed for initial presentations of URTI with
 repeats issued, excluding prescriptions for influenza / influenza-like illness.

Active Ingredient	Frequency	Percent	Cumulative Percent
Amoxicillin	3,792	22.63	22.63
Amoxicillin with clavulanate	6,397	38.18	60.8
Azithromycin	70	0.42	61.22
Benzathine benzylpenicillin	1	0.01	61.23
Cefaclor	499	2.98	64.21
Cefalexin	2,093	12.49	76.7
Cefuroxime	124	0.74	77.44
Ciprofloxacin	18	0.11	77.54
Clarithromycin	574	3.43	80.97
Clindamycin	13	0.08	81.05
Doxycycline	145	0.87	81.91
Erythromycin	506	3.02	84.93
Flucloxacillin	6	0.04	84.97
Minocycline	20	0.12	85.09
Nitrofurantoin	1	0.01	85.09
Norfloxacin	2	0.01	85.1
Phenoxymethylpenicillin	953	5.69	90.79
Procaine benzylpenicillin (procaine p	73	0.44	91.23
Roxithromycin	1,285	7.67	98.9
Tobramycin	1	0.01	98.9
Trimethoprim	10	0.06	98.96
Trimethoprim with sulfamethoxazole	174	1.04	100
Total	16,757	100	

2 Cefalexin is the recommended antibiotic for penicillin hypersensitive patients with 3 pharyngitis / tonsillitis. However, a repeat can be required to complete a single of cefalexin course due to manufacturer pack sizes. While 897 prescriptions for cefalexin were issued to 4 5 patients with pharyngitis / tonsillitis, only 120 patients had a penicillin allergy label recorded 6 (Appendix C.4.2). All but one of the 897 patients who received a repeat would have required one to complete a guideline-concordant course. Additionally, there was substantial variation 7 in duration of treatment among cefalexin prescriptions, ranging from five to 28 days. Please 8 see Appendix C.4.2 for details. 9

10

1

11 12

4.3.1 Introduction to modelling - clustering considerations

Noting that the same patient may present multiple times to the same GP during the study 13 period, multiple patients may attend the same GP / provider, and multiple GPs may work at 14 the same practice, this violates the assumption of independence underpinning the basis of 15 generalised linear regression with fixed effects (361,362). For URTI diagnoses, there were 16 between 68 to 10,513 patients per practice, between five and 104 unique providers per 17 practice. For antibiotic prescribing for URTI, there were between one and 2007 prescriptions 18 per provider, and between 76 and 17,015 prescriptions per practice. The large variation 19 between cluster sizes reinforces the need to allow, wherever possible, for non-20 21 independence. Model design was driven by variation in the size and numbers of clusters, with random intercepts for practice ID, and within practice level, the provider ID. 22

- 23
- 24 25

26

4.3.2 Model 0: predictors of inappropriate decisions (unnecessary prescribing versus necessary prescribing and appropriate non-prescribing)

Patient age group, gender, penicillin allergy label status, concession status, mental health condition status, socioeconomic disadvantage, weekend consultations, practice size, and patient's PHN were predictors of inappropriate decisions (Table 4-10, Appendix C.5.1: Table C-20). The following variables were insignificant in the multivariable model: other patient comorbid conditions, remoteness and accessibility, URTI condition, number of URTI episodes, and temperature recording status. Culture testing information were too few for inclusion.

- 34
- 35
- 36

 Table 4-10:Mixed effects logit regression models of binary inappropriate decisions (Model 0) and
binary unnecessary antibiotic prescribing (Model 1) for initial presentations of upper
respiratory tract infection

Model	Inappropriate Decision (ref. Appropriate Decision, including		Unnecessary Prescribing Among Prescriptions (ref. Appropriate	
	Appropriate Non-Prescribing)		Prescribing)	
Independent Variable	Adj. Odds Ratio	95% C.I.	Adj. Odds Ratio	95% C.I.
Patient Age Group, (ref. 0-8 yrs)	0.10.1444			
9-21 yrs	2.164***	[2.051,2.283]	3.344***	[3.077,3.634]
22-34 yrs	2.219***	[2.107,2.336]	4.691***	[4.279,5.143]
35+ yrs	2.056***	[1.958,2.160]	5.59/***	[5.106,6.135]
Patient penicillin allergy label, (ref. Negative)				
Positive	1.421***	[1.341,1.506]	0.832**	[0.737,0.938]
Patient Concession (ref. Negative)				
Positive	1.041*	[1.003.1.080]	1.125**	[1.035.1.223]
	1.011	[1.000)1.000]	1.125	[1:000;1:220]
Patient Mental Health Condition (ref. Negative)				
Positive	1.054*	[1.010,1.100]	1.297***	[1.156,1.455]
Missing	1.315	[0.857,2.017]	1.28	[0.687,2.387]
Patient gender, (ref. Female)				
Male	1.051*	[1.003,1.101]		
Ago group # Condor (rof Ecmalo # 0.8 yrs)				
Age group # Gender (rei. Female # 0-8 yrs)	0 012***	[0 752 0 979]		
22-24 yrs # Male	0.848***	[0.752,0.878]		
22-54 yrs # Male	0.048	[0.787,0.914]		
	0.500	[0.047,0.373]		
URTI Condition, (ref. Rhinosinusitis)				
Pharyngitis / Tonsillitis			0.430***	[0.373,0.494]
AOM			0.254***	[0.221,0.292]
Ordinal Line Agent Prescribed, (ref. First-line)				
Second-line			0.612***	[0.522,0.718]
Not Recommended			0.763***	[0.656,0.886]
LIBTI Condition # Ordinal Line of Antibiotic				
Phanyngitis # Not Recommended			1 607***	[1 /13 2 030]
AOM # Second-line			2 6/3***	[2 176 3 210]
AOM # Not Recommended			1.769***	[1.446.2.166]
				[1.1.0)2.200]
Disadvantaged Patient, (ref. No Disadvantage)				
Positive	1.063*	[1.000,1.130]		
Missing disadvantage status	0.922*	[0.861,0.986]		
Repeat on script, (ref. negative)				
Positive			0.903**	[0.838,0.974]
Multiple LIBTL episodes for patient (ref. Negative)				
Positive			0.897**	[0 839 0 959]
			0.037	[0.035,0.555]
Patient's Primary Health Network, (ref. Perth North)				
Perth South	0.954	[0.881,1.033]		
Country WA	1.034	[0.944,1.133]		
Interstate PHN	1.209*	[1.040,1.406]		
Patient's PHN Missing	1.134	[0.996,1.290]		
Practice size, (ref. Medium / Large)				
Small	1.678***	[1.300,2.166]		
Weekend (ref Weekday)				
Positive	1 241***	[1 183 1 202]		
	1.271	[1.103,1.302]		

Model	Inappropriate Decision (ref. Appropriate Decision, including Appropriate Non-Prescribing)		Unnecessary Prescribing Among Prescriptions (ref. Appropriate Prescribing)		
Independent Variable	Adj. Odds Ratio 95% C.I.		Adj. Odds	95% C.I.	
Seasonality allowed for in form of dummy variables for annual influenza seasons					
var(_cons[~c])	2.756***	[2.429,3.127]	4.282***	[3.359,5.458]	
N	111848		51210		
AIC	131446.7		32977.7		
BIC	131706.6		33189.9		
ICC					
Level	ICC (S.E.)	[95% C.l.]	ICC (S.E.)	[95% C.I.]	
unique provider ID & practice ID combination	0.236	[0.214,0.259]	0.307	[0.272,0.343]	
Level	ICC (S.E.)	[95% C.I.]	ICC (S.E.)	[95% C.I.]	
Note: SE in parentheses, *** p<0.01, ** p<0.05, * p<0.1.					
Note #: STATA 16 does not calculate ICC (incl. SE and 95%CI for ordinal models. ICC calculated by the author.					

Note #: STATA 16 does not calculate ICC (incl. SE and 95%CI for ordinal models. ICC calculated by the author. **Note:** Direct application of adjusted odds ratios to estimate the effect of a single variable in the model is not valid without also allowing for multi-level clustering. As such, direct application of adjusted odds ratios is not recommended.

1

There was an interaction identified between patient age group and gender. When 2 considered without patient gender, patients 22-34 years were at highest chance of receiving 3 inappropriate decisions, at fourteen percentage points higher (0.141, p<0.001, 95%CI: 4 0.133, 0.149), while patients 9-21 years and patients 35 and over had thirteen percentage 5 points higher chance of inappropriate decisions than young children (both p<0.001, see 6 7 Appendix C.5.1: Table C-20). Without considering patient age, males were one percentage point less likely to receive inappropriate decisions than females (-0.010, p<0.001, -0.015, -8 0.004). 9

10

However, when, using MERs, patient age and gender are considered together, this is not 11 12 the case. In this situation, relative to similar females, young male children were one percentage point more likely (0.009, p=0.035, 95%CI: 0.001, 0.018) while male patients 9-13 21 years were three percentage points less likely, and males 22-34 years were two 14 percentage points less likely (both p<0.001, see Appendix C.5.2) to receive an inappropriate 15 decision. There was no significant difference in gender for patients 35 years and over (-16 0.010, p=0.074, 95%CI: -0.020, 0.001). Regardless of patient gender, all other age groups 17 were eleven to sixteen percentage points more likely to receive an inappropriate decision 18 than young children 0-8 years (all p<0.001, see Appendix C.5.3). However, the CIs for the 19 AMEs for the three older female age groups partially overlap, as do those for males, 20 21 suggesting there is no substantial difference for patients of the same gender across the three older age groups. With other variables held constant at sample means, young children 22 had a probability of 32-33% of receiving an inappropriate decision (0.328, p<0.001, 95%CI: 23

0.313, 0.344), (0.319, p<0.001, 95%CI: 0.304, 0.334). Under the same conditions, patients
 of either of the two adult female age groups had probabilities of 46-48% of receiving likely
 inappropriate decisions, while adult male groups had probabilities of 44-45% (see Appendix
 C.5.4).

5

6 <u>4.3.2.1 Model 0: summary</u>

Young children had lower probabilities of receiving inappropriate decisions than other age 7 groups, and patients 22-34 years had the highest probability. There was an interaction 8 between patient age and gender, with young male children having an increased likelihood 9 of the outcome than young female children. However, females had a slightly higher chance 10 of inappropriate decision than males for patients 9-21 and 22-34 years. Other than patient 11 age and gender, it was practice size and penicillin allergy label status which had the highest 12 magnitude of effect upon the outcome of inappropriate decisions. Other predictors of 13 inappropriate decisions, which were identified but not presented in detail, included weekend 14 consultation status and patient's PHN, and there were small effects for mental health 15 condition status and government concession status (see Table 4-10, Appendix C.5.1: Table 16 C-20 for more detail). Of the variance not explained by fixed effects, the level consisting of 17 the unique combination of individual provider and individual practice combination accounted 18 for 24% (see Table 4-10). This suggests that the individual provider drives most of this 19 20 variation among the two, as the provider represents the lower level of the two (and with notably higher membership) within the data hierarchy of patient-provider-practice structure.⁷ 21

22

23 24

25

4.3.3 Model 1: predictors of unnecessary prescribing (versus necessary prescribing) among all antibiotic prescriptions

The predictors of unnecessary prescribing identified included patient age group, mental health condition status, patient concession status, patient penicillin allergy label status, URTI condition, number of URTI episodes, ordinal line of antibiotic prescribed and repeat prescription status (**Table 4-10**, **Appendix C.6.1: Table C-21**). The following variables were insignificant in the multivariable model: patient gender, patient-registered PHN, patient comorbid conditions status, patient remoteness and socioeconomic disadvantage indicators, and the dummy variable for cefalexin. Culture testing was too infrequent for

⁷ A model with three, nested levels was not feasible for the data. As such, a two-level model with a level for the unique combination of the provider and practice IDs was used, in addition to the patient level.

1 inclusion and temperature recording status was excluded due to correlation with URTI

- 2 condition.
- 3

The probability of receiving unnecessary prescribing increased with increasing age group, 4 5 regardless of the URTI condition. One may note in Figure 4-4 below, the notable step between the probability for young children and the remaining age groups for all conditions. 6 Relative to children 0-8 years: the probability of unnecessary prescribing for patients 9-21 7 years increased by twelve percentage points (0.117, p<0.001, 95%CI: 0.107, 0.127), for 8 patients 22-34 years the probability increased by thirteen to fourteen percentage points, and 9 for patients 35 years and over, it increased by fourteen percentage points (0.137, p<0.001, 10 95%CI: 0.126, 0.148), and by fifteen percentage points for patients aged 35 years and over 11 (0.146, p<0.001, 95%CI: 0.135, 0.157). 12

- 13
- 14



Note: This method provides the average probability of unnecessary prescribing occurring at each specific combination of upper respiratory tract infection and patient age group.

Figure 4-4: Plots of the marginal predicted mean of the outcome of unnecessary antibiotic
 prescribing occurring, across different upper respiratory tract infection conditions and
 different patient age groups, by condition

19

When URTI condition was considered independently, rhinosinusitis had the highest 1 probability of receiving unnecessary prescriptions, followed by pharyngitis / tonsillitis, then 2 AOM (Table 4-10, Appendix C.6.1: Table C-21). However, there was an interaction identified 3 between URTI condition and choice of antibiotic prescribed. With other covariates held 4 5 constant at sample means (Appendix C.6.4), the probability of receiving unnecessary prescriptions with first-line antibiotics was 94% for rhinosinusitis (0.945, p<0.001, 95%CI: 6 0.937, 0.953), 89% for pharyngitis (0.890, p<0.001, 95%CI: 0.879, 0.901), and 84% for AOM 7 (0.838, p<0.001, 0.823, 0.853). When second-line antibiotics were prescribed, the chance 8 of unnecessary prescribing dropped to 92% for rhinosinusitis (0.917, p<0.001, 95%CI: 9 0.906, 0.927) and 84% for pharyngitis (0.842, p<0.001, 95%CI: 0.820, 0.865) but rose to 10 89% for AOM (0.886, p<0.001, 95%CI: 0.873, 0.899). For not recommended antibiotics, the 11 probability of unnecessary prescriptions was 93% for rhinosinusitis (0.931, p<0.001, 95%CI: 12 0.922, 0.939), rising to a maximum of 91% for pharyngitis (0.910, p<0.001, 95%CI: 0.900, 13 0.920) and 87% for AOM (0.870, p<0.001, 95%CI: 0.854, 0.886). By condition and antibiotic 14 choice, the probability of unnecessary prescribing was highest for patients with: 15 rhinosinusitis receiving first-line, pharyngitis receiving not recommended, and AOM 16 receiving second-line antibiotics. However, the probability of unnecessary prescribing 17 remained higher for rhinosinusitis than any other condition regardless of antibiotic choice. 18

19

Patients with mental health conditions had a probability of receiving likely unnecessary prescriptions of two percentage points higher (0.020, p<0.001, 95%CI: 0.011, 0.028) than patients without mental health conditions. There was no significant difference for patients with missing mental health condition status (0.019, p=0.407, 95%CI: -0.026, 0.063). Patients with penicillin allergy labels had a two percentage point decrease in chance of receiving an unnecessary prescription (-0.015, p=0.004, 95%CI: -0.026, -0.005) than patients without these allergy label. Please see **Appendix C.6** for more details.

27

28 4.3.3.1 Model 1: summary

The probability of unnecessary prescribing increased with increasing age. By condition and choice of antibiotic, the probability of unnecessary prescribing was highest for rhinosinusitis receiving first-line, pharyngitis receiving not recommended, and AOM receiving second-line. However, the likelihood of this outcome remained higher for rhinosinusitis than any other condition regardless of the choice of antibiotic prescribed. Patients with penicillin allergy labels and repeats issued on prescriptions were linked to lower chance of receiving unnecessary prescriptions. Of the variance not explained by fixed effects, the unique provider accounted for 31% of remaining variation (**Table 4-10**). When the same model is calculated but with three, nested levels, the provider accounts for 31% while practice accounts for 5%. In both models, as the provider level is responsible for the higher percentages of variance unexplained by fixed effects than the practice level, this suggests that the individual provider drives the majority of this variation.

- 6
- 7

8

4.3.4 Model 2: predictors of increasing choice of antibiotic prescribed

9 The final model for increasing line of agent includes age group, prescribing reason recorded, 10 URTI condition, multiple URTI episodes status, patient comorbid conditions, patient 11 disadvantaged status, practice size, as well as unnecessary versus necessary prescribing 12 status and repeat prescription status (**Table 4-11, Appendix C.7.1: Table C-22**). The following 13 variables were insignificant in the multivariable model: patient gender, mental health 14 conditions, temperature recording status, patient remoteness and accessibility, concession 15 status, PHN, penicillin allergy label status, and private prescription status.

16

17 There was an effect modification between patient age group and URTI condition. When patient age group is considered with URTI condition using MERS, the interaction is apparent 18 19 for patients with pharyngitis receiving first-line and not recommended antibiotics (Figure 4-5, Appendix C.7.2). Relative to patients with rhinosinusitis: patients with pharyngitis aged 0-20 21 8 years were eight percentage points less likely to receive first-line antibiotics (-0.081, p<0.001, 95%CI: -0.097, -0.066), and seven percentage points more likely to receive not 22 recommended antibiotics (0.073, p<0.001, 95%CI: 0.059, 0.087). Meanwhile, patients with 23 pharyngitis aged 9-21 years and 22-34 years were linked to a two (0.018, p=0.046, 95%CI: 24 0.0002, 0.0351) and a three (0.0274, p=0.001, 95%CI: 0.011, 0.044) percentage point 25 increase, respectively, in the chance of receiving first-line antibiotics, compared to patients 26 with rhinosinusitis. There were no significant differences between patients with pharyngitis 27 aged 35 years and over and those with rhinosinusitis (all p>0.06, Appendix C.7.2). Compared 28 to patients with rhinosinusitis, patients with AOM were eight percentage points more likely 29 to receive first-line antibiotics (all p<0.001, Appendix C.7.2), and seven to eight percentage 30 points less likely to receive not recommended antibiotics (all p<0.001, Appendix C.7.2). 31

- 32
- 33

1 2

 Table 4-11:
 Mixed effects logit regression models of choice of antibiotic (Model 2) and binary nonfirst-line antibiotic prescribing (Model 3) for upper respiratory tract infection

Model	Ordinal Line of Antibiotic Binary Non-first-line			ne Prescribing (ref.
	Prescribed (ref. First-line)		First-line)	
Independent Variable	Adj. Odds Ratio	95% C.I.	Adj. Odds Ratio	95% C.I.
Patient Age Group, ref 0-8 yrs				
9-21 yrs	1.207***	[1.111,1.310]	1.491***	[1.354,1.642]
22-34 yrs	1.330***	[1.227,1.440]	2.041***	[1.853,2.248]
35+ yrs	1.665***	[1.550,1.788]	2.845***	[2.608,3.104]
Prescribing Decision, (ref Appropriate)				
Inappropriate	1.124***	[1.063,1.188]	0.636***	[0.550,0.736]
LIDTI Condition (rof Dhinosinusitis)	1			
Bhanyngitis / Tonsillitic	1 510***	[1 404 1 642]	0 400***	[0 254 0 502]
	0.646***	[1.404,1.043]	0.422	[0.334,0.302]
	0.040	[0.012,0.082]	0.485	[0.405,0.574]
Age Group # Pharynigitis (ref 0-8 yrs# Pharyngitis)				
9-21 yrs # Pharynigitis	0.601***	[0.538,0.671]	0.500***	[0.440,0.568]
22-34 yrs # Pharynigitis	0.570***	[0.511,0.636]	0.383***	[0.337,0.436]
35+ yrs # Pharynigitis	0.711***	[0.640,0.791]	0.387***	[0.341,0.438]
Patient Comorbid Conditions (ref Negative)				
Positive	1.151***	[1.094,1.211]	1.137***	[1.068,1.211]
Missing Status	1.228	[0.738,2.044]	1.031	[0.596,1.783]
Repeat on Prescription (ref Negative)				
Positive	1.140***	[1.061.1.225]	1.431***	[1.312.1.561]
Age Group # Repeat Status (ref U-8yrs # Repeat)	1 401***		1 700***	
9-21 yrs # Repeat Positive	1.481***		2.067***	
22-34 yrs # Repeat Positive	1.941	[1.737,2.170]	2 220***	[2.031,3.548]
	1.545	[1.220,1.484]	2.520	[2.038,2.040]
Disadvantaged Patient, (ref No Disadvantage)				
Positive	0.903*	[0.832,0.981]	0.914	[0.828,1.009]
Missing status	0.872***	[0.805,0.944]	0.845***	[0.770,0.928]
Multiple URTI episodes per Patient, (ref Negative)				
Positive	0.933**	[0.893,0.976]	0.905***	[0.858,0.955]
Practice size (ref medium / large)				
Small	1 783***	[1 319 2 409]	1 791***	[1 294 2 481]
	1.705	[1.515,2.405]	1.751	[1.234,2.401]
Prescribing Reason Recorded (ref Negative)				
Positive	0.681***	[0.591,0.784]	0.686***	[0.587,0.801]
Condition # Decision (ref Rhinosinusitis #				
Appropriate Decision)				
Pharyngitis / Tonsillitis # Inappropriate Decision			2.419***	[2.025,2.890]
AOM # Inappropriate Decision			1.768***	[1.480,2.111]
Patient Gender (ref Female)				
Male			1.092***	[1.045,1.141]
Seasonality allowed for in fo	rm of dummy varia	bles for annual in	fluenza seasons	
cut1	0.370 (0.06765)	[0.237, 0.502]		
cut2	1.252 (0.06788)	[1.119, 1.385]		
var(_cons)	0.323 (0.09242)	[1.153, 1.517]	1.488 (0.10379)	[1.298, 1.706]
Ν	51210	ĺ	51210	
AIC	93253.7		52719.6	
BIC	93510.2		52993.8	
ICC Level	ICC	95% C.I.	ICC	95% C.I.
Unique combination of provider ID & practice ID	0.28673092 #	#	0.311 (0.01496)	[0.283, 0.342]
			/	

Note: SE in parentheses, *** p<0.01, ** p<0.05, * p<0.1.

Note #: STATA 16 does not calculate ICC (incl. SE and 95%CI for ordinal models. ICC calculated by the author.

Note: Direct application of adjusted odds ratios to estimate the effect of a single variable in the model is not valid without also allowing for multi-level clustering. As such, direct application of adjusted odds ratios is not recommended.

Margins at Representative Values for the Effect, on the Ordinal Choice of Antibiotic Prescribed for Upper Respiratory Tract Infection, of Changes in Upper Respiratory Tract Infection Condition and Patient Age Groups, Relative to the Probability for Patients with Acute Rhinosinusitis



Note: This method provides the difference in the outcome of the ordinal choice of antibiotic prescribed (versus the outcome for acute rhinosinusitis) at the defined range of values for upper respiratory tract infection condition and patient age group.

Note: AOM represents the upper respiratory tract infection condition of acute otitis media.

2 3

1

Figure 4-5: Plot of margins at representative values for the effect on the probability of various levels of ordinal choice of antibiotic prescribed, with change in condition, across different patient age groups, relative to patients with rhinosinusitis

4 5

There was also an effect modification between patient age group and repeat prescription 6 7 status. When repeat status was considered with age group using MERs, the increase in probability of receiving not recommended antibiotics mirrored a similar and opposite 8 decrease in the probability of receiving first-line antibiotics (Figure 4-6, Appendix C.7.3). 9 Relative to patients receiving prescriptions with no repeats, patients aged 0-8 years with a 10 repeat on the prescription were three percentage points less likely to receive first-line (-11 0.025, p<0.001, 95%CI: -0.039, -0.011) and two percentage points more likely to receive a 12 not recommended antibiotic (0.022, p<0.001, 95%CI: 0.010, 0.034) (Appendix C.7.3). 13
Relative to patients receiving prescriptions without a repeat, patients 9-21 years receiving a 1 repeat were ten percentage points less likely to receive first-line (-0.101, p<0.001, 95%CI: -2 3 0.119, -0.083) and nine percentage points more likely to receive not recommended antibiotics (0.093, p<0.001, 95%CI: 0.076, 0.110). This effect was maximum for patients 22-4 5 34 years receiving repeats on prescriptions. Compared to prescriptions with no repeats, patients 22-34 years were fifteen percentage points less likely to receive first-line antibiotics 6 (-0.152, p<0.001, 95%CI: -0.170, -0.134), and fifteen percentage points more likely to 7 receive not recommended antibiotics (0.146, p<0.001, 95%CI: 0.128, 0.164). When 8 compared to prescriptions without repeats, patients aged 35 years and over receiving 9 repeats were eight percentage points less likely to receive a first-line antibiotic (-0.081, 10 p<0.001, 95%CI: -0.095, -0.067) and eight percentage points more likely to receive a not 11 recommended antibiotic (0.080, p<0.001, 95%CI: 0.066, 0.094). Figure 4-6 also highlights 12 the fact that second-line antibiotics featured little in the choice of antibiotic prescribed (see 13 Appendix C.7.3). 14

Margins at Representative Values for Change in Repeats Issued on Prescription Status from Negative to Positive, Across Different Patient Age Groups, for the Ordinal Choice of Antibiotic Prescribed for Upper Respiratory Tract Infection



Note: This method provides the difference in the outcome of the ordinal choice of antibiotic prescribed for upper respiratory tract infection for patients receiving prescriptions with repeats (versus the outcome for patients receiving prescriptions with no repeats) across different patient age groups.

15

Figure 4-6: Plot of margins at representative values for the effect on the probability of the ordinal choice of antibiotic being prescribed, with change in whether repeats were issued on the prescription from negative to positive, across different patient age groups, relative to prescriptions without repeats issued on them

Compared to medium / large GP practices, small practice size was linked to an eleven 1 percentage point decrease in the chance of receiving a first-line antibiotic (-0.110, p<0.001, 2 3 95%CI: -0.166, -0.054), and an equivalent eleven percentage point increase in the chance of receiving a not recommended antibiotic (0.105, p<0.001, 95%CI: 0.049, 0.161, Appendix 4 5 C.7.1: Table C-22). The reason for prescribing field being completed resulted in a seven percentage point increase in probability of first-line antibiotics (0.074, p<0.001, 95%CI: 6 0.047, 0.100), and a six percentage point reduction in the chances of receiving not 7 recommended antibiotics (-0.065, p<0.001, 95%CI: -0.088, -0.042), compared to the 8 probability when the reason for prescribing was not completed. Likely unnecessary 9 prescribing reduced the probability of first-line antibiotics by two percentage points (-0.022, 10 p<0.001, 95%CI: -0.033, -0.012), and a corresponding two percentage point increase in the 11 probability of receiving not recommended antibiotics (0.020, p<0.001, 95%Cl: 0.011, 0.030). 12 relative to likely necessary prescriptions (Appendix C.7.1: Table C-22). 13

14

Relative to patients without comorbid conditions, patients with comorbid conditions were 15 three percentage points less likely to receive first-line antibiotics (-0.027, p<0.001, 95%CI: -16 0.037, -0.017) and two percentage points more likely to receive not recommended antibiotics 17 (0.025, p<0.001, 95%CI: 0.016, 0.034). However, there was no significant difference for 18 patients with missing comorbid condition status (all p>0.250, Appendix C.7.1: Table C-22) 19 20 compared to patients without comorbid conditions. Relative to patients without socioeconomic disadvantage, patients with socioeconomic disadvantage were two 21 22 percentage points more likely to receive a first-line antibiotic (0.019, p=0.015, 95%CI: 0.004, 0.035) and two percentage points less likely to receive an antibiotic not recommended in the 23 24 guidelines (-0.018, p=0.014, 95%Cl: -0.032, -0.004), (Appendix C.7.1: Table C-22). Meanwhile, patients with missing socioeconomic disadvantage status were linked to a two 25 percentage point increase in first-line (0.026, p=0.001, 95%CI: 0.011, 0.041) and a two 26 percentage point decrease in not recommended (-0.024, p=0.001, 95%CI: -0.037, 0.010) 27 antibiotics, when compared to patients without socioeconomic disadvantage. Please see 28 Appendix C.7 for more details. 29

30

31 <u>4.3.4.1 Model 2: summary</u>

Patient age and repeats issued on prescription status had substantial effects on the ordinal choice of antibiotic prescribed. Patients aged 22-34 years were least likely to receive firstline antibiotics, and most likely to receive not recommended antibiotics. Second-line antibiotics barely featured in the results, with the main effects being any decrease in first1 line antibiotics was met with similar magnitude in increase in not recommended antibiotics,

2 and vice versa.

3

Prescriptions classified as unnecessary were associated with increased probability of nonfirst-line choice of agent than for necessary prescriptions.

6

Binary equivalent models for non-first-line prescribing versus first-line prescribing were also
developed (Model 3), with the same denominator of all antibiotics prescribed for URTI (Table
4-11, Appendix C.8: Table C-23). The results of this model are similar to model 2, and are
therefore not covered in detail here.

11

Common to both models 2 and 3 was the fact that individual provider was responsible for much more variation than individual practice, of variation otherwise unexplained by fixed effects. Of the variance not explained by fixed effects, the unique provider within unique practice combination accounted for 29% and 31% of variation in Models 2 and 3, respectively (see **Table 4-11**).

- 17
- 18 19

4.3.5 Comparing mixed effects models with fixed effects models

All outcomes were modelled with fixed effects using multivariable logistic regression for 20 comparison with the results of the mixed effects models. Varying results were obtained using 21 models with fixed effects only, and for all outcomes, additional variables became significant 22 (Appendix C.10: Table C-26). This highlights that different, likely misleading, results can arise 23 when important unobserved heterogeneity has been (erroneously) ignored. The model for 24 unecessary prescribing with fixed effects had an additional four variables, including PHN. 25 When the ordinal choice of antibiotic was modelled using fixed effects, the PHN also became 26 significant, as well as two other variables. When inappropriate decisions were modelled 27 using fixed effects, additional variables became significant, and scrutiny was placed on 28 different PHNs and not interstate.⁸ 29

30

When the same modelling process was followed for unnecessary prescribing but without allowing for individual provider within practice, the final model also included patient PHN and patient comorbid condition status. This would put undue scrutiny on PHNs (AOR =0.81 for

⁸ Please see the Appendix to this chapter for more details.

Perth South, and AOR=1.60 for Country WA, versus Perth North PHN) and potentially spurious association with patient comorbid conditions (AOR=1.19 for comorbid condition positive). This highlights a methodological advantage of using mixed-effects models for primary care data, despite the added complexity in both modelling and interpretation. This is a substantial advantage of this research over international studies which do not allow as accurately (or at all) for clustering.

- 7
- 8

9

4.3.6 Residual variance unexplained by fixed effects

10 It was of substantial interest whether individual practices or individual providers were 11 responsible for more of the remaining variance unexplained by fixed effects in each model, 12 which will be termed '*residual variance*' for ease of reference. The intention was to calculate 13 between-level intra-class correlations (ICCs) in three-level models for each outcome in order 14 to gain insight regarding the contribution to residual variance by each of the practice and 15 provider levels (380,382).

16

However, the final *mixed-effects models* developed utilised a two-level model including the 17 patient level within a level comprising a unique combination of individual provider and 18 individual practice IDs (365), this structure was unsuited to calculating between-level ICC 19 for providers and practices. Further hierarchical mixed models were therefore developed to 20 address this question, using the same variables as had been identified in the *mixed-effects* 21 models but with different random-effects structure. This was performed using three-level 22 models where possible, and separate two-level models, using patient individual level and 23 24 either unique provider ID or practice ID as the second level, for comparing ICC.

25

A three-level model of patient within provider within practice levels for unnecessary prescribing was developed, and the practice level is responsible for 5% while the provider level accounts for 31% of residual variance. A three-level model was not feasible for ordinal choice of antibiotic. However, an equivalent two-level model of patient within provider level had an ICC of 29% for the provider level, while a two-level model for patient within practice level demonstrated that 7% of residual variance was attributable to the practice level.

32

Consistent to all models was the fact that unique provider levels were responsible for the largest variance from random effects, and substantially more than the practice levels in each model or comparison model.

2 **4.4** Summary

3

It is apparent, based on the guidelines, that inappropriate prescribing of systemic antibiotics 4 is occurring frequently for initial episodes of care for URTI in WA general practice, and in 5 multiple forms: inappropriate decisions among diagnoses, unnecessary prescriptions 6 among antibiotic prescriptions, non-first-line antibiotics being prescribed for initial 7 consultations, and repeats being issued on prescriptions without justification (29). Thirty-8 nine percent of patients received inappropriate decisions among all initial presentations of 9 URTI. Among all antibiotic prescriptions for URTI, 85% of antibiotic prescriptions were found 10 to be unnecessary, 68% of antibiotic prescriptions were non-first-line, for these initial 11 presentations of URTI. Both figures are concerning. 12

13

Among all antibiotics prescribed for each condition, non-first-line prescribing occurred in 14 68% of occasions for acute rhinosinusitis, notably higher than for other conditions, while 15 non-first-line antibiotics comprised 58% of prescriptions for pharyngitis / tonsillitis, and 53% 16 17 prescribing situations for AOM. All three other patient age groups had notably higher probabilities of receiving inappropriate decisions than patients 0-8 years. Unnecessary 18 19 prescribing increased with increasing patient age, regardless of the URTI condition. Pharyngitis was the condition with proportionally the most inappropriate decisions and the 20 21 lowest appropriate non-prescribing rate. However, rhinosinusitis was the condition with the 22 highest likelihood of unnecessary prescribing among antibiotics received for URTI.

23

There are other notable factors in discord with the guidelines (29). Second-line antibiotics featured little in the results of any model. Instead, the majority of results demonstrate that if first-line antibiotics were not prescribed, not recommended agents were prescribed, and vice versa. Additionally, based on the low numbers of pathology results available, it is clear that few laboratory culture and sensitivity tests are both being requested by GPs and the testing being performed by patients.

30

The fact that the unique provider levels demonstrated the largest variance from random effects, and substantially more than the practice levels in each model, suggests that individual providers have much more influence on inappropriate prescribing than individual practices.

Inappropriate decisions, unnecessary prescribing, the choice of antibiotic when prescribed, non-first-line antibiotic prescribing, and repeat prescribing are five different measures of inappropriate prescribing with different predictors, albeit with commonalities. Patient age group was significant in all models, while patient remoteness was consistently insignificant. Of the three outcomes of unnecessary, ordinal choice and repeat prescribing (not presented), these outcomes were all found to predict each other. Patient age and URTI condition were significant predictors in all three models.

8

9 Second to patient age, it was small practice size and weekend consultations, and the reason 10 for prescribing being recorded that substantially increased the probability of patients 11 receiving an inappropriate decision. Patient age and URTI condition had notable impact on 12 the likelihood of unnecessary prescribing and non-first-line antibiotics being prescribed. This 13 analysis demonstrates that patient factors of age, gender, comorbid conditions, mental 14 health conditions, government concessions and penicillin allergy label status, do influence 15 inappropriate antibiotic prescribing.

16

There were effect modifiers identified between patient age and gender in the likely inappropriate decision model, and between age and URTI condition in both the ordinal choice and the repeat positive prescribing models. Effect modification was also identified between choice of antibiotic and URTI condition in the likely unnecessary prescribing model, as well as between age and repeat prescription status in the ordinal choice model.

22

Repeat positive prescribing was associated with necessary prescribing. Conversely, 23 24 necessary prescribing was more likely to occur with repeats issued on the prescription. It is feasible that GPs may tend to prescribe antibiotics with repeats on the prescription for the 25 most seriously unwell patients in need of treatment, such as, necessary prescribing 26 situations. Receiving repeats on prescriptions was also linked to increased rates of non-first-27 line prescribing in adults but was linked to lower chance of non-first-line choice for children. 28 This may reflect that GPs may be more cautious with choice of antibiotic and whether to 29 prescribe repeats when dealing with children. GPs may also be more comfortable issuing 30 31 repeats for lower-line agents for children.

32

Having a penicillin allergy label was linked to patients receiving inappropriate decisions but
it was also associated with necessary prescribing. This may be explained by the difference
in denominators for both outcomes.

The large proportions of beta-lactamase inhibitors and second-generation cephalosporins issued with repeats supports the recent PBAC decision to amend the requirements for repeat prescriptions (127,128). The newly restricted agents represented 82% of repeats issued for initial presentations of URTI in this dataset. It is hopeful that PBAC may also consider extending the requirements to agents including phenoxymethylpenicillin, clindamycin and erythromycin, which were used frequently for URTI and were commonly issued repeats when likely not required.

CHAPTER 5 PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR URINARY TRACT INFECTION

3 4

5 6

5.1 Introduction

UTI encompasses the condition acute cystitis (29). High rates of non-first-line prescribing for initial presentations of UTI are a common problem internationally, particularly in developed countries (119,122,335,400-403). In Australia, non-first-line prescribing occurs for UTI more frequently than it feasibly should (15,17,48). Antibiotic prescribing by GPs for initial presentations of UTI was therefore a focus of this project. Antibiotics are prescribed for UTIs more frequently than in any other presenting group in Australian primary care, second only to URTI (15).

14

The nationally agreed guidelines, called Therapeutic Guidelines: Antibiotic (the guidelines) recommend that (empirical) prescribing for UTI is appropriate (29). Research indicates that approximately 80% of adult females diagnosed with UTI require antibiotics (258,404). The guidelines for acute cystitis were developed to direct diagnosis and pathology testing to obtain aetiology, recommend effective antibiotics, minimise antibiotic resistance, and limit side effects (29). Reducing unnecessary use of antibiotics for UTI is considered crucial to lessening the spread of antibiotic resistance (6,16) and improving the quality of primary care.

22

The aim of this study was to quantify inappropriate, guideline non-conforming prescribing of 23 systemic antibiotics in Australian general practice. The focus was on patients presenting 24 with initial presentations of UTI, the choice of antibiotic prescribed, and any likely 25 26 unnecessary repeat positive antibiotic prescriptions. The aim was also to elucidate patient-, practice- or consultation-related predictors of various levels of inappropriate antibiotic 27 prescribing for patients presenting with UTI. The point of the study was to determine if there 28 were any associations of inappropriate prescribing with patient factors such as age, 29 comorbid conditions, SES, and remoteness / accessibility to health care. 30

31

32 **5.2 Specific methods**

33

Non-initial and / or chronic consultations were excluded, including consultations occurring within fourteen days of a previous UTI consultation for the same patient, and any mention of chronic, recurrent, resistant infections. Inappropriate prescribing was defined to include non-first-line antibiotic prescribing for these initial presentations of UTI, as well as the issuing
 of repeats on prescriptions without justification. A graphical depiction of the variables and

3 models developed follows in Figure 5-1.

nitrofurantoin

4

As depicted in **Table 5-1**, the guidelines for treatment of initial presentations of UTI were utilised to classify recommended antibiotics into an ordinal variable termed choice based on the order of their recommendation in the guidelines (29), by patient group: women, men and children under 16 years of age.⁹ The antibiotic choice prescribed to each patient group was the numerator and the denominator was all antibiotics prescribed to that patient group.

10

11 12 13	Table 5-1:	Antibiotic classifications for the ordered choice variable for acute cystitis, based on the order of antibiotics recommended in Therapeutic Guidelines: Antibiotic (29), by patient group					
	Choice	Non-pregnant Women	Men	Children >= 1 month			
	First-line	trimethoprim	trimethoprim	trimethoprim			
	Second-line	cefalexin	cefalexin	cefalexin			
	Third-line	amoxicillin + clavulanate	amoxicillin + clavulanate	amoxicillin + clavulanate			

 Last resort
 norfloxacin
 Norfloxacin
 norfloxacin

Note: A first-line option should be the antibiotic prescribed at initial consultations. Third-line and last resort options were combined into the third ordinal level for analysis. Where the antibiotic prescribed is not listed as an option for the condition diagnosed, the prescription was classified as 'not recommended'. For more information, please refer to the Methods chapter.

nitrofurantoin

14

Third-line

While quantitative research using analytic methods to elucidate predictors of inappropriate, non-first-line prescribing for UTI in primary care has been undertaken internationally (122,296,323,353,405-407), Australia lags behind. Despite the existing research on antibiotic prescribing, resistance and its current surveillance within the Australian health system occurring at several levels (14,17), this is to the best of the author's knowledge the first Australian research using quantitative methods to identify predictors of inappropriate prescribing for UTI in primary care.

⁹ For more information, please refer to the Methods chapter (Chapter 3).



Figure 5-1: Depiction of the main outcome variables utilised in this analysis for initial presentations of urinary tract infection

1 **5.3 Results**

3 Initial descriptive analyses found that there were 21,205 initial episodes of care for UTI identified, of which 40% were to female patients aged 16-44 years (Table 5-2, Table 5-4 3). There were 17,973 systemic antibiotics issued, and the strong majority (81%) were 5 prescribed to females at least sixteen years of age (Appendix D.1). The prescribing rate 6 of prescriptions among diagnoses per patient group, was 85% for all patients, 87% for 7 women, and 78% for men and 75% for children, respectively (Appendix D.1). Of patients 8 9 with UTI, 18% presented with multiple, independent occurrences of UTI. The reason for prescribing field was completed in eighteen percent of antibiotic prescriptions. 10

11

First-generation cephalosporins were the most common class of antibiotic prescribed, 12 followed by trimethoprim and derivatives. First-line cefalexin was the most commonly 13 prescribed active ingredient (41%), followed by first-line trimethoprim (38%), and third-14 15 line amoxicillin with clavulanate (8%) (Table 5-4). Among antibiotics not recommended in the guidelines for the patients receiving them were nitrofurantoin (40%), amoxicillin 16 without clavulanate (33%), and trimethoprim with sulfamethoxazole (12%) (Appendix 17 D.1). There were 68 fluoroquinolones and one third-generation cephalosporin 18 prescribed. 19

- 20
- 21 22
- Table 5-4:
 Frequency table to all active ingredients for systemic antibiotics prescribed for initial presentations of urinary tract infection

Active Ingredient	Frequency	Percent	Cumulative Percent
Amoxicillin	510	2.84	2.84
Amoxicillin with clavulanate	1,402	7.8	10.64
Ampicillin	1	0.01	10.64
Azithromycin	11	0.06	10.7
Cefaclor	83	0.46	11.17
Cefalexin	7,374	41.03	52.19
Ceftriaxone	1	0.01	52.2
Cefuroxime	11	0.06	52.26
Ciprofloxacin	68	0.38	52.64
Clarithromycin	10	0.06	52.7
Clindamycin	3	0.02	52.71
Doxycycline	17	0.09	52.81
Erythromycin	16	0.09	52.9
Flucloxacillin	5	0.03	52.92
Nitrofurantoin	702	3.91	56.83
Norfloxacin	430	2.39	59.22
Phenoxymethylpenicillin	8	0.04	59.27
Roxithromycin	9	0.05	59.32
Trimethoprim	6,976	38.81	98.13
Trimethoprim with sulfamethoxazole	336	1.87	100
Total	17,973	100	

1
1

Table 5-2:Frequency table of patient characteristics for all patients, and patients with initial
episodes of care for urinary tract infection (column percentage)

	All natients	All visits	Initial care	Antibiotic Prescribed for UTI	No antibiotic
Characteristic	n=791 280	n=1 925 985	n=21 205	n=17 793	n=3 232
Patient	11-751,200	11-1,525,505	11-21,203	11-17,735	11-5,252
Female gender, %	52.8	59.1	88.4	89.6	81.3
(s.d.)	(22.07)	(23.69)	(23.83)	(23.06)	(27.72)
Patient's Primary Health Network, %					
Country WA	12.9	32.8	27.4	26.2	33.9
Perth South	33.1	33.5	32.2	32.4	31.0
Perth North	34.8	31.8	37.5	38.5	32.2
Interstate	2.2	0.5	1.2	1.3	1.0
Missing	1.5	1.3	1.7	1.7	2.0
Patient concession status positive, %	34.1	35.6	33.9	33.0	39.1
Comorbid condition, %	1.9	10.2	20.6	19.6	26.0
Missing	3.4	2.5	2.8	2.0	2.0
Mental health condition, %	12.9	28.7	25.9	25.9	26.2
Missing	3.4	2.5	2.8	2.9	2.0
Patient remote, % (remote &	6.0	4.4	5.8	5.9	5.5
very remote Australia, ARIA)	1 1	1.0			
Missing	1.1	1.0			
Patient disadvantaged, % (top 40% percentiles of most disadvantage, SEIFA IRSD)	11.7	10.3	1.0	1.0	1.2
Missing	1.5	1.3	10.8	10.5	12.4
Multiple UTIs per patient, %			17.6	16.8	21.9
Consultation					
Temperature recorded, %, (% of which fever positive >=37.5C)			15.9 (12)	16.3 (11)	13.8 (15)
Urine dipstick tested, %, (% of which positive result)			95.5 (93)	95.5 (93)	96.0 (92)
Culture performed, %, (% of which had positive growth for any pathogen)			2.8 (100)	3.1 (99)	1.3 (100)
Sensitivity performed, %			0.0	0.0	0.0
Weekend consult, %			8.9	9.2	7.3
Prescription					
Repeat issued on				20.4	
prescription, %				28.4	
entered, %				17.8	

Table 5-3:	Frequency table of patient, consultations and prescription	characteristics for	
	patients prescribed an antibiotic for urinary tract infection (co	olumn percentage)	

	Any Antibiotic Prescribed for UTI	First-line Agent Prescribed	Second-line Agent Prescribed	Third-line Agent Prescribed	Not Recommended Agent Prescribed
Characteristic	n=17,793 (100%)	n=7,127 (40.0%)	n=7,374 (41.0%)	n=1,906 (10.6%)	n=1,556 (8.7%)
Patient					
Female gender, %	89.6	36.6	36.9	8.2	7.9
Mean age, years	45.3	48.3	42.8	45.6	42.8
(s.d.) Patient's Primary Health Network.	(23.06)	(21.74)	(23.61)	(24.66)	(22.62)
%					
Country WA	26.2	36.5	38.0	41.3	46.2
Perth South	32.4	33.9	30.1	34.2	34.2
Perth North	38.5	26.7	28.8	21.3	17.2
Interstate	1.3	1.3	1.5	1.1	0.4
Missing	1.7	1.6	1.6	2.1	2.0
Patient concession status positive, %	33.0	34.8	31.8	33.7	29.0
Comorbid condition positive, %	19.6	20.1	18.2	23.8	18.7
Missing	2.0	1.6	3.8	2.9	4.3
Mental health condition positive, %	25.9	28.2	23.5	26.1	26.3
Missing	2.9	1.6	3.8	2.9	4.3
Patient remote positive, %	5.9	4.7	7.5	5.0	4.5
(remote & very remote Australia, ARIA)		-	-		
Missing	1.0	1.0	1.0	1.1	0.8
(top 40% percentiles of most disadvantaged positive, %	10.5	10.1	11.6	8.2	9.8
Missing	16.8	18.0	16.6	15.8	14.1
Multiple UTIs per patient, %	17.0	17.7	15.6	19.1	17.7
<u>Consultation</u>					
Temperature recorded, %, (% of	16.3 (11)	16.1 (6)	16.9 (13)	16.6 (23)	14.6 (11)
Which rever positive >=37.5C) Urine dipstick tested. %. (% of	95.5 (93)	5,5 (92)	3.8 (95)	2.8 (91)	5.8 (93)
which positive result)	00.0 (00)	0.0 (02)	0.0 (00)	(0 _)	010 (00)
Culture performed, %, (% of which had positive growth for any pathogen)	3.1 (99)	3.9 (100)	2.6 (100)	2.3 (95)	2.8 (100)
Sensitivity performed, %	0.0	0.0	0.0	0.0	0.0
Weekend consult, %	9.2	9.2	9.6	8.2	8.6
Prescription					
Repeat issued on prescription, %	28.4	22.0	30.7	48.2	22.0
Reason for prescribing entered, %	17.8	21.7	21.8	19.7	15.4

As seen in **Figure 5-2**, there were notable changes in the proportions of antibiotic lines prescribed across patient age groups. Children had much higher ratios of non-first-line to first-line prescriptions than adult age groups, by a factor of 3.6-6.4 (**Appendix D.3**: **Table D-4**). The proportions of second-line and not recommended antibiotic prescriptions increased with decreasing age (**Appendix D.3**: **Table D-5**).

6





8 9 10 Figure 5-2: Bar graph of ordinal choice / line of antibiotic prescribed for initial presentations of urinary tract infection, by patient group

Among all antibiotic prescriptions issued for each patient group, 57% of women, 68% men and 82% children under sixteen received antibiotic prescriptions other than firstline (**Table 5-5**). While the proportions of first-line and second-line agents prescribed for women appeared in order in which they appear in the guidelines (29), both men and children received higher proportions of second-line than first-line antibiotics (**Figure 5-3**). Children under sixteen years received second-line (56%) more than three times as often as first-line antibiotics (18%).

18

19 20 21 Table 5-5:Frequency table of first-line and non-first-line antibiotics prescribed for initial
episodes of urinary tract infection, by patient group

			22
	First-line	Non-first-line	Totaĺ 23
Women	6,290	8,238	14,528
	43.3	56.7	100
Men	504	1,068	1,572
	32.06	68	100
Children under 16yrs	333	1,540	1,873
	18	82	100
Total	7,127	10,846	17,973
	39.65	60.35	100



1 2



Figure 5-3: Bar graph of counts of ordinal choice of antibiotic prescribed for initial presentations of urinary tract infection, by patient group

5 Despite guideline recommendations stating that both children and men should receive both culture and sensitivity testing (29), no children or adult males receiving 6 7 prescriptions for initial UTI received both forms of testing. Among all UTI diagnoses for each patient group, culture testing occurred for 2% children, 3% of women and 3% of 8 men. There were no sensitivity testing results available for patients presenting with initial 9 UTI, although there were sensitivity results present for other conditions, and, despite 10 guideline recommendations, there were no sensitivity test results available for all 430 11 norfloxacin prescriptions issued (29). Urine dipstick testing was recorded in 96% 12 consultations (Table 5-2). For a summary of how the antibiotic prescribing for UTI 13 analysed here compares with prominent quality indicators, please see Appendix G.2. 14 15

A repeat was issued on 28% prescriptions for UTI, and 99% of these were for one repeat. While a repeat can be required to provide a guideline-concordant course for cefalexin, 56% prescriptions with repeats issued were for antibiotic agents other than cefalexin, and for which a repeat is typically not required. Other than cefalexin, repeats were numerically most common for trimethoprim and amoxicillin with clavulanate (**Table 5-6**). Proportionally, however, repeats on prescription were much more common among third-line / last resort antibiotics (48%) than first-line (21%), or second-line cefalexin 1 (31%) (Appendix D.4). The antibiotics restricted by PBAC in 2020 account for 60% of repeats issued for initial UTI in this dataset (127,128). By age group, children 6-15 years 2 3 received the highest proportion of repeats issued, and adults 16-24 years received the lowest (Table 5-7). 4

5

presentations of urinary tract infection, by active ingredient								
Active ingredient	Repeat Negative	Repeat Positive	Total					
Amoxicillin	396	114	510					
Amoxicillin with clavulanate	725	677	1,402					
Ampicillin	1	0	1					
Azithromycin	11	0	11					
Cefaclor	72	11	83					
Cefalexin	5,108	2,266	7,374					
Ceftriaxone	1	0	1					
Cefuroxime	8	3	11					
Ciprofloxacin	60	8	68					
Clarithromycin	8	2	10					
Clindamycin	2	1	3					
Doxycycline	12	5	17					
Erythromycin	10	6	16					
Flucloxacillin	5	0	5					
Nitrofurantoin	572	130	702					
Norfloxacin	214	216	430					
Phenoxymethylpenicillin	8	0	8					
Roxithromycin	5	4	9					
Trimethoprim	5,455	1,521	6,976					
Trimethoprim with sulfamethoxazole	202	134	336					
Total	12,875	5,098	17,973					

6 Frequency table of antibiotic prescriptions issued with repeats for initial Table 5-6: 7

10

11 12 Table 5-7: Frequency table of whether repeats were issued on antibiotic prescriptions (negative or positive) for initial presentations of urinary tract infection, by patient age group

	Repeat	Repeat	Total (count,
Patient Age Group	Negative	Positive	row %) 14
45+ yrs, ref	5,701	2,446	8,147 15
	69.98	30.02	100 16
16-44 yrs	5,947	2,006	7,953
	74.78	25.22	1 70
6-15 yrs	630	364	9 84
	63.38	36.62	10 0
0-5 yrs	597	282	279
	67.92	32.08	100 21
Total	12,875	5,098	17,973
			22

⁸ 9

While cefalexin is typically the recommended antibiotic for patients with penicillin 2 sensitivity (29), it was prescribed in 7,374 situations when only 5% of these patients had 3 any record of a penicillin allergy label (Appendix D.4.2). Of cefalexin prescriptions with 4 repeats, 0% men (n=5), 0% women at least 18 years of age, 53% (n=10) patients 16-5 17yrs, and 99% children (n=294) likely required a repeat for a full course (Appendix 6 D.4.2). While over 99% repeats were likely unnecessary for adults over seventeen, 99% 7 were potentially necessary for children. For patients at least sixteen years, the duration 8 of cefalexin prescription was variable, most commonly five or seven days and up to 9 fourteen days. Over-treatment was common for women receiving cefalexin prescriptions 10 with repeats, however, under- and over-treatment were common for men. For children 11 under sixteen years, the duration varied between three and fourteen days, and most 12 commonly five days. Please see Appendix D.4.2 for details. 13

14

16

1

15

5.3.1 Modelling introduction - clustering considerations

Noting that multiple patients may attend the same provider / GP, and multiple GPs may 17 work at the same practice, this violates the assumption of independence underpinning 18 the basis of generalised linear regression with allowance for unobserved heterogenous 19 effects (362). With respect to antibiotics prescribed for initial UTI, there were for example 20 between 46 to 1,929 patients per practice, and between five and eight unique providers 21 per practice. Model design was driven by these clusters, and it was possible to include 22 a random intercept to allow for unobserved heterogeneity at both the practice level, and 23 24 within practice, the provider level for these data (408).

25

Following experimentation, the final model consisted of patient age group and gender with an interaction term. A three-level hierarchical model of patient level, within provider level, within practice level was developed, including random intercepts for individual practice ID and individual provider ID within their respective levels (409,410).¹⁰

- 30
- 31

¹⁰ Recall that AMEs are used to summarise effects unless otherwise stated.

1 5.3.2 Model 1: ordinal choice of antibiotic prescribed

The ordinal model of choice of antibiotic prescribed (Model 1) and the equivalent, binary model of non-first-line prescribing (Model 2) are similar outcomes. As Model 1 utilises more information, the focus of discussions will be Model 1, and Model 2 will not be discussed in detail.

Patient age, gender, repeat prescription status, and patient comorbid condition status, 8 9 urine dipstick testing and culture testing status were identified as predictors of non-firstline prescribing (Table 5-8, Appendix D.5). When considered independently, male 10 gender, young patient age, prescriptions with repeats, and comorbid conditions were 11 linked to non-first-line prescribing. Female gender, older patient age, prescriptions 12 13 without repeats, urine dipstick testing and culture testing were independently associated with first-line prescribing, and were less likely to receive non-first-line prescribing. The 14 following variables were insignificant in the multivariable model: patient PHN, patient 15 measures of remoteness and socioeconomic disadvantage, concession status, patient 16 17 mental health condition status, temperature recording status, day of the week, and whether a reason for prescribing was recorded. 18

19

2

7

Two effect modifiers were identified in the ordinal model for antibiotic choice: repeat prescription status and patient gender, both of which were found to independently interact with patient age. There first effect modification was between patient age group and gender. Increasing patient age was independently associated with decreasing probability of non-first-line antibiotic. However, when patient gender is allowed for, males had an increased probability of non-first-line prescribing for adults, compared to females, and the magnitude of effect increased with increasing age.

- 27
- 28
- 29

2 Tabl

Table 5-8: Mixed effects models for initial presentations of urinary tract infection

Model	Ordinal Line Prescribed ref. First-line		Binary Non-first-line ref. First-line		Repeat Positive Prescribing ref. Prescriptions without Repeats	
Independent Variable	Odds Ratio	95% C.I.	Odds Ratio	95% C.I.	Odds Ratio	95% C.I.
Patient Age group, (ref 45+yrs)						
16-44 yrs	1.180***	[1.099,1.268]	1.294***	[1.188,1.410]	0.886*	[0.799,0.983]
6-15 yrs	2.191***	[1.904,2.520]	3.724***		1.6/0***	[1.363,2.047]
0-5 yrs	2.839	[2.443,3.300]	7.435	[5.791,9.540]	1.158	[0.923,1.453]
Patient gender, (ref Female) Male	1.791***	[1.581,2.029]	1.825***	[1.559,2.137]	2.114***	[1.775,2.516]
Age group # Gender						
16-44 yrs # Male	0.877	[0.696,1.105]	0.809	[0.600,1.090]	0.967	[0.699,1.339]
6-15 yrs # Male	0.759	[0.528,1.091]	0.838	[0.487,1.442]	0.405***	[0.238,0.690]
0-5 yrs # Male	0.596**	[0.422,0.844]	0.596	[0.336,1.057]	0.343***	[0.200,0.588]
Repeat on script, (ref. negative) Positive	1.627***	[1.501,1.763]	2.088***	[1.880,2.319]		
Comorbid condition positive, (ref. negative)						
Positive	1.215***	[1.117,1.322]	1.262***	[1.143,1.394]		
Missing	1.649	[0.899,3.024]	1.65	[0.779,3.493]		
Culture tested, (ref. negative) Positive	0.740**	[0.609,0.900]	0.749*	[0.599,0.938]	1.472**	[1.116,1.943]
Dipstick tested, (ref. negative) Positive	0.710**	[0.576,0.875]	0.676**	[0.532,0.860]	0.661*	[0.474,0.921]
Ordinal Line Prescribed, (ref First-line)						
Second-line					1.910***	[1.702,2.144]
Third-line / Last resort					4.112***	[3.522,4.801]
Not Recommended					1.103	[0.915,1.329]
Temperature recorded, (ref negative)						
Positive					1.186*	[1.023,1.375]
Multiple UTI episodes for patient, (ref negative) Positive					1.556***	[1.378.1.758]
	1		1			[,,,,]
Potential changes in prescribing behaviours allowed for over time by inclusion of continuous variable for year of consultation in all models.					ar of	

cut1	55.1715	[4.729,105.614]
cut2	57.47025	[7.027,107.913]
cut3	58.53822	[8.095,108.981]

Variable	Ordinal Line Prescribed ref. First-line		Binary Non-first-line ref. First-line		Repeat Positive Prescribing ref. Prescriptions without Repeats	
	Odds Ratio	95% C.I.	Odds Ratio	95% C.I.	Odds Ratio	95% C.I.
var(cons[practice id)	1.064	[0.987,1.147]	1.075	[0.964,1.200]	1.568**	[1.115,2.204]
var(cons[practice id> provider id]	2.923***	[2.460,3.473]	5.803***	[4.311,7.810]	35.00***	[19.17,63.89]
Observations	17973		17973		17973	
Information Criterion						
AIC	38670.8		19842.5		14537.1	
BIC	38811.1		19967.3		14677.4	
ICC Level	ICC	[95% C.I.]	ICC	[95% C.I.]	ICC	[95% C.I.]
Practice	0.014		0.014	[0.003,0.061]	0.062	[0.030,0.124]
Provider within practice	0.257		0.358	[0.321,0.395]	0.549	[0.507,0.590]

Note: Exponentiated coefficients; 95% confidence intervals in brackets: * p<0.05, ** p<0.01, *** p<0.001

As depicted in Figure 5-4, older men are eleven percentage points less likely to receive first-line antibiotics (-0.109, p<0.001, 95%CI: -0.132, -0.086), four percentage points more likely to receive third-line or not recommended antibiotics (both p<0.001, see Appendix D.5.2) than similarly aged women. Relative to young adult females, young adult men were eight percentage points less likely to receive first-line (-0.083, p<0.001, -0.118, -0.047), and three percentage points more likely to receive third-line or not recommended antibiotics (both p<0.001, see Appendix D.5.2). For children, there was no significant difference between genders in the probability of receiving first-line antibiotics (both p>0.064, see Appendix D.5.2), or for non-first-line choices (all p>0.05, see Appendix D.5.2).



Note: This method provides the difference in the outcome of the ordinal choice of antibiotic prescribed for urinary tract infection for male patients (versus the outcome for female patients) across different patient age groups.

Figure 5-4: Plot of marginal effects at representative values for change in patient gender from
 female to male, at different patient age groups for model 1 (ordinal line of
 antibiotic prescribed) for initial presentations of urinary tract infection

1

7 The second effect modification was between patient age group and repeat prescription status. Recall that increasing patient age was independently associated with decreasing 8 9 probability of receiving a non-first-line antibiotic. When repeat prescription status is taken into account, however, a repeat being present decreased the probability of first-10 line antibiotics and increased the probability of non-first-line prescribing for adult age 11 groups. Compared to the patients receiving prescriptions without repeats, adults with a 12 repeat issued on the prescription were eleven percentage points less likely to receive 13 14 first-line antibiotics (both p<0.001, see Appendix D.5.4), and three to four percentage points more likely to receive a non-first-line antibiotic (all p<0.001, see Appendix D.5.4), 15 (Figure 5-5). With a repeat present, there was no significant difference in the antibiotic 16 received for children (all p>0.10, see Appendix D.5.4). There was no three-way effect 17 modification present between patient age, gender and repeat prescription status. 18



Note: This method provides the difference in the outcome of the ordinal choice of antibiotic prescribed for urinary tract infection for patients receiving prescriptions with repeats (versus the outcome for prescriptions with no repeats) across different patient age groups.

Figure 5-5: Plot of marginal effects at representative values for change in repeat on prescription status from negative to positive by patient age group for model 1 (ordinal line of antibiotic prescribed) for urinary tract infection

5 6

1

With all other covariates held constant at the sample means, depending upon patient
gender and repeat status, adults had a 26-47% chance of receiving a first-line antibiotic,
while for older children this was 22-30% and 23-26% for young children (all p<0.001,
see Appendix D.5.5). Young children had the lowest probability of receiving first-line
antibiotics, while adult patient groups had the highest.

12

Again, with all other covariates held constant at the sample means, women 45 years and over receiving a prescription without a repeat had a 47% chance of receiving a firstline antibiotic (0.471, p<0.001, 95%CI: 0.445, 0.498), while with a repeat this dropped to 36% (0.363, p<0.001, 95%CI: 0.334, 0.392). For women 16-44 years, the probability of receiving a first-line antibiotic was 44% without a repeat (0.438, p<0.001, 95%CI:

0.412, 0.465) but dropped to 33% with a repeat (0.331, p<0.001, 95%CI: 0.303, 0.360). 1 Women aged at least 45 years and 16-44 years had probabilities of 39-43%, and 40-2 43%, respectively, of receiving second-line antibiotics, regardless of repeat prescription 3 status (all p<0.001, see Appendix D.5.5). With other covariates held at sample means, 4 for men the highest probability outcome (of 43-44%) was a second-line antibiotic. 5 regardless of their adult age group, or repeat prescription status (all p<0.001, see 6 Appendix D.5.5). Similarly, the most likely outcome for children was second-line 7 antibiotics (at 43-44%), regardless of the age group, gender, or repeat status (all 8 p<0.001, see Appendix D.5.5). 9

10

Relative to patients not receiving dipstick testing, across all age groups, patients who 11 received urine dipstick testing were seven percentage points more likely to receive first-12 line (0.065, p=0.002, 95% CI: 0.024, 0.107), and two percentage points less likely to 13 receive second-line (-0.025, p=0.008, 95% CI: -0.043, -0.007), third-line (-0.019, 14 p=0.001, 95% CI: -0.031, -0.008) or not recommended antibiotics (-0.021, p=0.001, 95% 15 Cl: -0.0332, -0.009), (Appendix D.5.1: Table D-21). Compared to patients not receiving 16 culture testing, patients receiving culture testing were six percentage points more likely 17 18 to receive first-line antibiotics (0.060, p=0.002, 95%CI: 0.021, 0.098), and two percentage points less likely to receive any non-first-line antibiotic option (all p<0.009, 19 see Appendix D.5.1: Table D-21). 20

21

Patients with a comorbid condition were four percentage points less likely to receive 22 first-line (-0.037, p<0.001, 95%CI: -0.053, -0.021), and one percentage point more likely 23 to receive second-line (0.011, p<0.001, 95%CI: 0.006, 0.016), third-line (0.012, 24 p<0.001, 95%CI: 0.007, 0.017) or not recommended (0.014, p<0.001, 95%CI: 0.008, 25 0.021) antibiotics than patients without chronic conditions. Patients with missing 26 27 comorbid condition status were two percentage points more likely to receive secondline antibiotics (0.022, p=0.002, 95% CI: 0.008, 0.036) than patients without comorbid 28 29 conditions (see Appendix D.5.1: Table D-21).

30

As can be seen in **Table 5-8**, the ICC for practice level in Model 1 for ordinal choice was 0.014 while the ICC for provider level was 0.257. ICC can be interpreted as the remaining variance unexplained by observed heterogeneous effects in the model. Therefore, of the variance not explained by observed heterogenous effects in Model 1, 1 the provider level was responsible for 26% of variance compared to only 1% for practice

- 2 level.
- 3
- 4 5 6

5.3.3 Model 3: predictors of repeat positive prescribing

The predictors of a repeat being issued on the antibiotic prescription in Model 3 were 7 identified as patient age, gender, ordinal choice of antibiotic prescribed, whether the 8 patient had multiple UTI episodes, temperature recording, culture testing and urine 9 dipstick testing status (Table 5-7, Appendix D.6). The probability of receiving a repeat on 10 prescription was linked to both third-line prescribing and second-line prescribing. There 11 was an effect modification between patient age group and gender. The following 12 variables were insignificant in the multivariable model: day of the week of the 13 consultation, patient concession status, measures of patient remoteness and 14 socioeconomic disadvantage, mental health conditions status and comorbid condition 15 status. 16

17

Patients receiving third-line prescriptions were nineteen percentage points more likely (0.187, p<0.001, 95%CI: 0.163, 0.211), and patients receiving second-line antibiotics were eight percentage points more likely (0.080, p<0.001, 95%CI: 0.064, 0.095), to receive a repeat than patients receiving first-line antibiotics. There was no significant difference between patients receiving not recommended antibiotics (0.011, p=0.310, 95%CI: -0.010, 0.033) (**Appendix D.6.1: Table D-22**).

24

25 When patient age group and gender are considered together using MERs, there is a significant difference for adults between the effect of gender on the chance of receiving 26 27 a repeat (Appendix D.6.2). Men 45 years and over were ten percentage points (0.100, 28 p<0.001, 95%CI: 0.075, 0.124) and men 16-44 years were nine percentage points (0.093, p<0.001, 95%CI: 0.054, 0.131) more likely to receive a repeat on their 29 prescription than equivalent women. There was no significant difference in the effect of 30 gender on children (-0.021, p=0.543, 95%Cl: -0.089, 0.047), (-0.040, p=0.205, 95%Cl: 31 -0.101, 0.022). 32

33

As one may see in **Figure 5-6**, when all other covariates are kept at sample means, the adjusted probability of receiving a repeat on the prescription is highest for men 45 years and over at 38% (0.377, p<0.001, 95%CI: 0.335, 0.419). For women of equivalent age,

the chance of receiving a repeat was ten percentage points less (0.276, p<0.001, 1 95%CI: 0.243, 0.309). Men aged 16-44 years had a 36% chance of receiving a repeat 2 (0.355, p<0.001, 95%CI: 0.303, 0.407), and again this was ten percentage points lower 3 for similar women (0.261, p<0.001, 95%CI: 0.229, 0.293). Female children 6-15 years 4 had a probability of 34% (0.344, p<0.001, 95%CI: 0.300, 0.387) of receiving a repeat, 5 while this was two percentage points lower (0.322, p<0.001, 95%CI: 0.250, 0.395) for 6 equivalent males, albeit noting the 95% CIs partly overlap. Young female children had 7 a 29% probability of receiving a repeat (0.295, p<0.001, 95%CI: 0.252, 0.337), while 8 similar males had the lowest probability at 25% (0.254, p<0.001, 95%CI: 0.191, 0.318), 9 again noting the partial overlap of CIs. Please see Appendix D.6.3 for more detail. 10

11



Note: This method provides the predicted value of repeats being issued on antibiotic prescriptions for urinary tract infection, at the average values of each level of patient gender and patient age group, with all other covariates being held constant at respective, sample means.

12 13

Figure 5-6:

- 14
- 15 16



Graph of adjusted predictions at the means for model 3 (repeats present on

antibiotic prescription) at differing values of patient gender and age group,

- receive a repeat on the prescription than patients with a single episode (0.057, p<0.001, 18
- 95%CI: 0.040, 0.073). Patients receiving urine dipstick testing were five percentage 19

points less likely to receive a repeat than patients without dipstick testing (-0.049, p=0.011, 95%CI: -0.087, -0.012). Meanwhile, patients receiving culture testing were five percentage points more likely to receive a repeat than patients who did not receive culture testing (0.050, p=0.008, 95%CI: 0.013, 0.087). Regardless of patient age, patients receiving temperature testing were two percentage points more likely to receive a repeat than patients more likely to receive a repeat than patients who did not have their temperature tested (0.022, p=0.026, 95%CI: 0.003, 0.040). (Please see Appendix D.6.1: Table D-22).

8

9 The ICC for the practice level was found to be 0.062 (95% CI: 0.030, 0.124), whereas 10 the ICC for the provider level was 0.549, (95% CI: 0.507, 0.590). As such, the provider 11 level was responsible for 55% of variance compared to only 6% for practice level, of 12 variance not explained by homogenous effects in Model 3 (see **Table 5-8**).

- 13
- 14

15 16

5.3.4 Modelling considerations

For the purposes of comparison, the same modelling process was performed without 17 allowance for unobserved heterogeneity, using logit and ordered logit models, in 18 19 addition to trialling a dummy variable for practice ID number (in lieu of a random intercept for practice). With no allowance for unobserved heterogeneity, the ordinal 20 21 logistic model for choice of antibiotic included patient remoteness status, day of the week and patient's PHN also become significant predictors in addition to those identified 22 23 from the mixed effects models. This represents nine of the total fourteen variables tested. For the model repeat prescribing, with observed heterogeneity only, eleven of 24 the total fourteen variables became significant, including day of the week, reason 25 recording status, and patient PHN. Without allowance for unobserved heterogenous 26 effects, the models included more variables and tended to have lower AORs, for the 27 most part, for the variables in both models (see Appendix D.8). This highlights that 28 different, likely misleading, results are obtained when using fixed models which do not 29 allow for unobserved heterogeneity as achieved by using mixed models. 30

31

Likelihood ratio testing demonstrated that the adult male and child samples were insufficient samples to use in separate models (376). Brant testing indicated that the distances between first-line, second-line and not recommended antibiotics are not the same (378,411).

1 5.4 Summary

2

This research offers new insights regarding the complex nature of antibiotic prescribing 3 for UTI in Australian general practice. It demonstrates that there is substantial, 4 unjustifiably non-first-line antibiotic prescribing occurring for patients with initial 5 presentations of UTI in Australian general practice, and repeats frequently issued on 6 prescription without justification. Sixty percent of patients received non-first-line 7 prescriptions for initial episodes of care for UTI: children under sixteen (82%) and men 8 9 (68%) and women (57%). Culture testing was low, results were present for 2% children, 3% of women and 3% of men, and sensitivity testing results were absent. Children and 10 men were prone to receiving non-first-line antibiotics and frequently lacking both culture 11 and sensitivity testing (29). 12

13

The predictors of increasing line of antibiotic prescribed were identified as patient age 14 15 group, patient gender, comorbid condition status, repeat prescription status, urine dipstick-testing, culture testing. Predictors of repeat prescribing included patient age 16 17 group, ordinal choice of antibiotic, urine dipstick testing, temperature recording, multiple episodes. Day of the week, practice size, patient concession, remoteness, and 18 19 disadvantage status, mental health condition and cefalexin prescription status were insignificant in all models. It is also apparent that non-first-line prescribing and repeat 20 21 positive prescribing are linked, due to each variable's presence in the model for the 22 other.

23

From the model of ordinal choice of antibiotic, repeat negative and female gender are linked to lower probability of higher line prescribing in adults. Young children were least likely to receive first-line antibiotics, followed by older children, then adults. Children and men were most likely to receive second-line antibiotics. Urine dipstick and culture testing being performed were associated with lowered chance of non-first-line prescribing. Patients with comorbid conditions were found to be at increased probability of non-firstline prescribing compared to patients without comorbid conditions.

31

Repeats being issued on prescriptions appears strongly linked to third-line, and to a lesser extent, to second-line antibiotic prescribing. Adult males had a notably higher probability of receiving repeats on antibiotic prescriptions than adult females. Patients with multiple, separate UTI episodes during the study period were six percentage points more likely to receive a repeat than patients with only a single UTI episode. Urine
dipstick testing being performed was linked to lower likelihood of repeat positive
prescribing, while culture and temperature testing being performed increased the
probability of repeat positive prescribing.

5

Patient age and gender are the most pressing drivers of inappropriate prescribing for UTI, and it is clear that there are differences in the patient groups as is expected given different guidelines for different patient groups (29). By virtue of the fact that UTI in men and children may be considered complicated as opposed to uncomplicated infection in women (412,413), it is unsurprising that men and children have increased probabilities for non-first-line prescribing and repeat positive prescribing than adult women.

12

While attempting to clarify whether predominantly patient-specific factors might be driving inappropriate UTI prescribing, several of the predictors identified are also, consultation- or prescription-related. Measures of patient remoteness and socioeconomic disadvantage were hypothesised as potential predictors of inappropriate prescribing but this does not appear to be the case for UTI. Similarly, patient comorbid and mental health conditions were considered potentially relevant to inappropriate prescribing, however, only comorbid condition status emerged as a predictor for UTI.

20

Of the variance not explained by observed heterogenous effects in Model 1 for ordinal choice of antibiotic, the provider level was responsible for 26% of variance compared to only 1% for practice level, as detailed earlier. Of variance not explained by observed heterogenous effects in Model 3 for repeat positive prescribing, the provider level was responsible for 55% of variance compared to only 6% for practice level.

26

27 Conditional upon the homogenous effects, we find that all outcomes of all models for 28 UTI are minimally correlated within the same practice, however, they are moderately 29 correlated within the same provider and practice. This can be interpreted as the provider 30 level being predominantly responsible for the remaining variance unexplained by 31 observed heterogenous effects in all models, rather than the practice level which is 32 responsible for little residual variance. This suggests that the individual provider has 33 substantially more effect on choice of antibiotic than variation by practice.

- A comparison of results obtained from the mixed effects models allowing for unobserved heterogeneity and those without unobserved heterogeneous effects highlights that there are notable differences in the results. For example, the inclusion of PHN as a predictor of non-first-line prescribing in the ordinal logit model for ordinal prescribing would likely place pressure on PHNs, which may be unfounded.
- 6

CHAPTER 6 ANALYSIS OF CHANGES IN ANTIBIOTIC PRESCRIBING FOR UPPER RESPIRATORY TRACT INFECTION CONDITIONS OVER TIME

4

5

8

6 6.1 Background to trends analyses for upper respiratory tract infection 7 and urinary tract infection

9 There were 19,5 million encounters by 791,280 patients, of which 53% were female, 10 and the mean patient age was 38 years. The strong majority (97%) of patients had 11 residential addresses listed as in WA. By PHN for WA-listed patients, 30% patients 12 had residential addresses within the Country WA PHN, 34% in Perth North PHN, 35% 13 in Perth South PHN, and 1% were missing PHN information (42). By accessibility and 14 remoteness index (34,35). these correspond to 66% in major cities of Australia, 5% in 15 remote, 1% in very remote, and 1% were missing PHN information.

16

Over the study period of 1 January 2012 to 30 June 2017, inclusive, there were fluctuations in involvement of individual GPs, as well as to a lesser degree, individual general practices, with a small number of practices opening and closing. **Figure 6-1** depicts the proportion of practices located within each PHN over time, including interstate PHNs.

22

Multiple steps were taken to limit the dataset to initial presentations for both condition 23 groups, URTI and UTI. Some of these steps included the removal of diagnoses of 24 UTI/URTI containing various terms to describe either chronic/recurrent UTI/URTI or 25 26 follow-up consultation from the dataset. Furthermore, any UTI/URTI diagnoses with another presentation for the same condition group occurring within fourteen days prior 27 to the diagnosis of interest were considered the same episode of UTI/URTI and were 28 therefore excluded. Diagnoses containing a definitive pathogen, which would have 29 30 required laboratory pathology to ascertain, were also excluded, as these are unlikely to represent initial consultations. 31







There were 145,889 initial episodes of care for URTI and 21,206 for UTI during the study period. For URTI there was an antibiotic prescribing rate of 36%, and the frequency of consultation and antibiotic prescribing peaked at approximately six years. For UTI, there were 17,974 systemic antibiotics issued, and the strong majority of

9 these (81%) were prescribed to females at least sixteen years of age.

10

11 12

6.2 Introduction to trends analyses for upper respiratory tract infection

The objective of this analysis was to explore trends in antibiotic prescribing for patients 13 with initial presentations with URTI conditions over time. Inappropriate prescribing is 14 frequently reported for URTI presentations to primary care, ranging from prescribing 15 when not indicated (210,270,272-274,332), to using non-first-line antibiotics without 16 17 initially treating with first-line antibiotics (270,271,275-277,280,282,284,295,297,321,322,329,332). As antibiotic prescribing is typically not 18

indicated for initial episodes of care for URTI (29), it was primarily of interest to
establish whether there were notable changes over time in unnecessary antibiotic
prescribing to patients at these initial episodes of care for URTI. Secondary questions
included whether there were substantial changes over time in the proportion of
patients receiving antibiotic prescriptions for initial diagnoses of URTI conditions, and
the proportion of patients receiving non-first-line antibiotics among all prescribed
antibiotics.

8

9 While the local epidemiology of these URTI conditions is not expected to change meaningfully over a relatively short period of time, there may be changes to antibiotic 10 prescribing behaviours in light of growing antibiotic resistance (12,14,18), as well as 11 increasing publicity regarding antibiotic resistance and antibiotic stewardship 12 initiatives as described in the Background chapter. Due to these factors, one would 13 hope that inappropriate antibiotic prescribing may be decreasing. Therefore, it was of 14 interest to establish whether or not antibiotic prescribing was decreasing and, if so, by 15 how much. Favourable changes would include decreasing rates of likely unnecessary 16 prescribing over time, for all URTI conditions, and reducing proportions of non-first-17 line antibiotics among all antibiotics prescribed for these conditions (275). 18

19

Previously defined outcome variables were used in this analysis, as described in the 20 Methods chapter (Chapter 3). For ease of reference, visual depictions of how these 21 22 outcomes were previously defined and obtained are included below (Figure 6-2 and Figure 6-3). These diagrams also contain details regarding the denominator for each 23 24 outcome variable. Recall that non-initial and / or repeat consultations were excluded for URTI, including the exclusion of consultations occurring within fourteen days of a 25 previous URTI consultation for the same patient. URTI was defined to include 26 uncomplicated URTIs, acute rhinosinusitis including the common cold and non-27 28 specific URTI, acute pharyngitis and / or tonsillitis, and AOM, influenza and ILI, in accordance with the Therapeutic Guidelines: Antibiotic (the guidelines) (29). 29



tonsillitis, and acute otitis media but not influenza / influenza-like illness.

1

Figure 6-2: Depiction of the main outcome variables utilised in this trends analysis, and
 previous analyses for initial presentations of upper respiratory tract infection



Note: Models 2, 3 and 4 include acute rhinosinusitis, pharyngitis / tonsillitis, and acute otitis media but not influenza / influenza-like illness.

Figure 6-3: Depiction of the main response variables used in this trends analysis, and previous analyses for, initial presentations of upper respiratory tract infection, continued

Table 6-1 serves as a reminder of the guideline recommendations (29) for antibiotic 1 2 treatment of for initial presentations of URTI conditions as were applied in the analysis, by classifying individual antibiotic agents into an ordered choice based on the order of 3 their recommendation in the guidelines for each condition. Penicillin hypersensitivities 4 were allowed for (332), and suitable alternative antibiotics were also classified as first-5 line choices where the patient had a recorded allergy label for penicillin. Antibiotics 6 7 which were prescribed but are not listed in this table were classified as not recommended, as were situations in which penicillin hypersensitivity options were 8 9 utilised despite the patient having no record of a penicillin allergy label.

10

11 12	Table 6-1:	5-1: Choice of antibiotic for upper respiratory tract infection conditions, by condition, and by allergy label, as per Therapeutic Guidelines: Antibiotic (29)			
	Condition	Line / Choice	No penicillin hypersensitivity	Penicillin non- immediate hypersensitivity	Penicillin immediate hypersensitivity
	Acute rhinosinusitis	First-line	amoxicillin	cefuroxime	doxycycline
		Second-line	amoxicillin + clavulanate	doxycycline	
	Acute pharyngitis / tonsillitis	First-line	phenoxymethylpenicillin	cefalexin	azithromycin
		Second-line	benzathine penicillin		
	Acute otitis media	First-line	amoxicillin	cefuroxime	trimethoprim + sulfamethoxazole
		Second-line	amoxicillin + clavulanate	trimethoprim + sulfamethoxazole	

Note: A first-line antibiotic should be prescribed at initial consultations where prescribing is indicated. Where the antibiotic prescribed is not listed as an option for the condition diagnosed, the prescription was classified as 'not recommended'.

13

By virtue of its definition, any reduction in unnecessary prescribing would coincide with a reduction in overall antibiotic prescribing (within a static epidemiological setting). This would represent a significant increase in compliance with the guidelines (29), as to when to prescribe antibiotics to patients presenting with initial episodes of URTI, as well as secondary improvements in the choice of antibiotic by clinicians in situations when antibiotics are prescribed.

Although there were many outcomes analysed, for all patients with initial presentations of URTI, as well as for each specific URTI condition, only the most pertinent results are presented here. For additional results from this analysis, please see **Appendix E**.

Please note that the linear trend models utilise smoothed, five-month moving average, data for each outcome, which were used to calculate mean monthly rates (414). Note that percentage point difference reported for each outcome uses the difference between the predicted values of these linear models at the commencement of the study in January 2012 and its conclusion in June 2017. For more details, please refer to the Methods chapter (Chapter 3).

7

8

9

6.3 Results

Throughout the study period from January 2012 to June 2017 inclusive, children of eight years and under accounted for 25 to 44% of the URTI patient population, by month. The proportion of patients diagnosed with acute rhinosinusitis / non-specific URTI ranged between 33 and 68% of URTI diagnoses per month, whereas pharyngitis / tonsillitis represented 17 to 40% of URTI diagnoses (**Figure 6-4**). Each month, AOM diagnoses ranged from 10 to 27% of URTI diagnoses, and influenza / ILI (with no mention of bacterial superinfection) represented 0 to 6% of all URTI diagnoses.



17

18 Figure 6-4: P 19 Ja

6-4: Plot of counts of diagnoses, by upper respiratory tract infection condition, from January 2012 to June 2017, inclusive, by month

Note: Influenza / influenza-like illness included purely for the purposes of observing
 seasonality, also noting the different scale used for influenza / influenza-like
 illness diagnoses
1 6.3.1 All upper respiratory tract infection conditions altogether

2

3

4

5

6

7

For all patients with initial presentations of URTI, there was a significant, downward trend in antibiotic prescribing of -0.0023 per month over the study period January 2012 to June 2017 inclusive (-0.0023, p<0.001, 95%CI: -0.0025, -0.0022). Although the raw coefficients from these linear regression models of the monthly aggregate data (**Table 6-2**) appear 'small', it is important to note that these represents unit changes per month. For example, the linear regression coefficient of -0.0023 unit increase per

8 month. For example, the linear regression coefficient of -0.0023 unit increase per 9 month for antibiotic prescribing extrapolates to a three percentage point decrease per 10 year, and a fifteen percentage point reduction in antibiotic prescribing the over the five-11 and-a-half year study period. This was a notable reduction in antibiotic prescribing, 12 from 56 to 41%, (**Figure 6-5**). One will note that the linear trend line in **Figure 6-5** varies 13 from the smoothed prescribing rate data at various points on visual inspection, such 14 as a peak in late 2013, as well as a calculated R-squared of 93%. Despite this, the 15 linear trend line provides a reasonable estimate of the overall downward trend which

16 can be clearly seen in antibiotic prescribing for rhinosinusitis (Figure 6-5).



Note: Denominator is all patients diagnosed with upper respiratory tract infection, initial presentations only

17

Figure 6-5: Time series plot for antibiotic prescribing rate among initial presentations of
 upper respiratory tract infection, January 2012 to June 2017, inclusive, by
 month

There was a significant, downward trend in unnecessary antibiotic prescribing by -1 0.00092 per month (-0.0009, p<0.001, 95%CI: -0.0010, -0.0008) (Table 6-2). This 2 corresponds to a six percentage point reduction, from 89 to 83% of antibiotic 3 prescriptions for initial presentations of URTI. There was also a downward trend in the 4 not recommended prescribing (-0.0014, p<0.001, 95%CI: -0.0015, -0.0012), which 5 reduced by nine percentage points, from 44 to 35% of all antibiotics (Table 6-2). There 6 7 was also a significant, increasing trend in second-line antibiotic prescribing (0.0011, p<0.001, 95%CI: 0.1251, 0.1991), which increased from thirteen to twenty percent of 8 9 all antibiotics prescribed for initial presentations. However, there was no significant trend for non-first-line prescribing (-0.0002, p=0.178, 95%CI: 0.5658, 0.5502) among 10 all antibiotics prescribed for all patients with URTI (Table 6-2). See Appendix E for more 11 detail. 12

13

It is important to note that patients with acute rhinosinusitis comprise the majority of 14 patients with URTI, and as such, the results for all URTI and rhinosinusitis are similar. 15 16

6.3.2 By individual upper respiratory tract infection condition 17

18

After considering overall trends for all URTI conditions, the focus is now on the 19 20 individual component conditions, which is important as they have different treatment guidelines, disease patterns, and risks of serious segualae (29).¹¹ 21

22

6.3.2.1 Acute rhinosinusitis 23

For acute rhinosinusitis (and non-specific URTI), there was a significant, downward 24 trend in antibiotic prescribing (-0.0020, p<0.001, 95%CI: -0.0023, -0.0018), involving 25 a thirteen percentage point reduction from 40% to 27% (Table 6-2). Unnecessary 26 antibiotic prescribing also reduced significantly from 97% to 85% (-0.0019, p<0.001, 27 95%CI: -0.0020, -0.0017) (Table 6-2). However, significant upward trends were found 28 for both second-line prescribing (0.0031, p<0.001, 95%CI: 0.0028, 0.0033) and non-29 first-line prescribing (0.0028, p<0.001, 95%CI: 0.0024, 0.0033). As can be seen in 30 Figure 6-6, second-line prescribing increased by twenty percentage points, from 14 to 31 34% of antibiotic prescriptions for rhinosinusitis. The "fit" of this linear trend model 32 varies, with several peaks and troughs about the linear trend line upon visual 33

¹¹ Due to the small sample size for antibiotics prescribed for initial presentations of influenza/ILI (n=253), trends analyses for this condition are not presented here. Please see the Appendix E for more details.

inspection in Figure 6-6. One also notes a calculated R-squared of 88% (Table 6-2). 1 2 Although a strict linear trend does not appear particularly appropriate here, there is a clear upward trend, and the linear trend model intends to provide an estimate of this 3 trend. There was a significant, eighteen percentage point, increase in non-first-line 4 prescribing from 58 to 76%. There was no significant trend over time for prescribing of 5 antibiotics not recommended in the guidelines (-0.0003, p=0.056, 95%CI: -0.0005, 6 7 6.6E-06), which occurred at a mean monthly rate of 43% of all antibiotic prescriptions for rhinosinusitis (Table 6-2). 8

9



10

11Figure 6-6:Time series plot of prescribing rates for second-line antibiotics for initial12presentations of acute rhinosinusitis, January 2012 to June 2017 inclusive, by13month

examining specific antibiotics prescribed for initial presentations 15 When of rhinosinusitis, there was a significant downward trend for amoxicillin (-0.0028, 16 p<0.001, 95%CI: -0.0033, -0.0024), and a significant upward trend for amoxicillin with 17 clavulanate (0.0030, p<0.001, 95%CI: 0.0027, 0.0033) (Table 6-3). Amoxicillin use 18 decreased from 42 to 23%, while the use of amoxicillin with clavulanate increased from 19 14 to 34% of all antibiotics prescribed for the condition (Figure 6-7). As seen in Figure 20 6-7, the increases in second-line amoxicillin with clavulanate use appear to correspond 21 to similar decreases in the use of first-line amoxicillin. Trends in these two antibiotics 22

- agents are likely to be relevant to the increases in second-line and therefore non-first-
- 2 line prescribing for initial presentations for rhinosinusitis over the study period.



Time series plot of amoxicillin and amoxicillin with clavulanate prescribing rates for initial presentations of acute rhinosinusitis, January 2012 to June 2017,

3

4 5

Figure 6-7:

inclusive, by month

6

<u>URTI</u>	Dependent Variable	a) (Mov	Descriptive ing Average	<u>Statistics</u> Data)	b) Linear Regression Model for Trend									
<u>Condition</u>	Prescribing Outcome Monthly Rate	Mean prop.	January 2012 prop.	June 2017 prop.	Coefficient (unit increase per month)	[95% Conf.	Interval]	p-value	R-squared	January 2012 predicted	June 2017 predicted value	Percentage Point Difference *		
All URTI	Unnecessary Antibiotic Prescribing	0.86	0.90	0.84	-0.00092	-0.001041	-0.000803	0.0000	0.7899	0.885750	0.825816	-6		
	Overall Antibiotic Prescribing +	0.48	0.54	0.39	-0.00232	-0.002478	-0.002165	0.0000	0.9320	0.556847	0.405952	-15		
	Prescribing of Second-line antibiotic	0.16	0.10	0.18	0.001139	0.0008867	0.0013912	0.0000	0.5597	0.125075	0.199103	7		
	Prescribing of antibiotic Not Recommended in guidelines	0.4	0.45	0.36	-0.00138	-0.001527	-0.001231	0.0000	0.8447	0.440770	0.351132	-9		
	Non-first-line (non-first- line) antibiotic prescribing	0.56	0.54	0.54	-0.00024	-0.000592	0.0001121	0.1780	0.0282	0.565844	0.550236	-2		
Acute Rhinosinusitis	Unnecessary Antibiotic Prescribing	0.91	0.98	0.85	-0.00185	-0.001984	-0.00172	0.0000	0.9244	0.969503	0.849124	-12		
	Overall Antibiotic Prescribing +	0.34	0.40	0.26	-0.00200	-0.00225	-0.001753	0.0000	0.8018	0.4013533	0.2712579	-13		
	Prescribing of Second-line antibiotic	0.67	0.55	0.76	0.003066	0.0027845	0.0033484	0.0000	0.8806	0.143700	0.343019	20		
	Prescribing of Not Recommended antibiotic	0.24	0.12	0.34	-0.00025	-0.000513	6.61E-06	0.0559	0.0559	0.437753	0.421293	-2		
	Non-first-line (non-first- line) antibiotic prescribing	0.43	0.43	0.42	0.002813	0.0023695	0.0032569	0.0000	0.7149	0.581453	0.764312	18		

Table 6-2: Prescribing outcomes for patients with initial presentations of upper respiratory tract infection, by condition group

URTI Condition	Prescribing Outcome Monthly Rate	Mean prop.	Jan 2012 prop.	Jun 2017 prop.	Coefficient (unit increase per month)	[95% Conf.	Interval]	p-value	R-squared	Jan 2012 predicted value	Jun 2017 predicted value	Percentage Point Difference *
Acute Pharyngitis /	Unnecessary Antibiotic Prescribing	0.86	0.90	0.84	-0.00118	-0.001368	-0.001001	0.0000	0.7223	0.893786	0.816808	-8
Tonsillitis	Overall Antibiotic Prescribing +	0.71	0.75	0.69	-0.00064	-0.000774	-0.000503	0.0000	0.5795	0.732203	0.690698	-4
	Prescribing of Second-line antibiotic	-	-	-	n/a	n/a	n/a	n/a	n/a	n/a	n/a	n/a
	Prescribing of antibiotic Not Recommended in guidelines	0.48	0.62	0.36	-0.00425	-0.004463	-0.00404	0.0000	0.9618	0.619348	0.343029	-28
	Non-first-line (non-first- line) antibiotic prescribing	0.48	0.62	0.36	-0.00425	-0.004463	-0.00404	0.0000	0.9618	0.619348	0.343029	-28
Acute Otitis Media	Unnecessary Antibiotic Prescribing	0.77	0.81	0.78	0.000168	-6.04E-05	0.0003958	0.1468	0.0326	0.765647	0.776547	1
	Overall Antibiotic Prescribing +	0.59	0.68	0.50	-0.00265	-0.002741	-0.002555	0.0000	0.9805	0.673169	0.501051	-17
	Prescribing of Second-line antibiotic	0.53	0.44	0.53	0.001456	0.0008301	0.0020819	0.0000	0.2523	0.280424	0.375064	9
	Prescribing of antibiotic Not Recommended in guidelines	0.33	0.21	0.31	0.000114	-0.00013	0.0003571	0.3547	0.0134	0.203054	0.210439	1
	Non-first-line (non-first- line) antibiotic prescribing	0.21	0.23	0.22	0.00157	0.0010925	0.0020467	0.0000	0.4029	0.483478	0.585502	10

Note: The rate calculation for each particular prescribing outcome (excluding overall antibiotic prescribing) was calculated with the numerator being all patients with initial presentations of acute otitis media, who were prescribed an antibiotic classified as being part of the particular outcome. The denominator is all patients with initial presentations of acute otitis media who were prescribed an antibiotic.

Note +: The rate calculation for overall antibiotic prescribing was calculated with the numerator being all patients with initial presentations of the specific condition or the condition group, who were prescribed an antibiotic. The denominator is all patients with initial presentations of the specific condition or the condition group.

Note *: The percentage point difference uses predicted values for first and last months of the study period, January 2012 to June 2017 inclusive.

	Prescribing Rates for Individual Antibiotic Agents for Patients with initial presentations of Acute Rhinosinusitis / non-specific URTI													
	<u>Dependent</u> <u>Variable</u>	b) Linear Regression Model for Trend												
URTI condition	Prescribing Outcome Monthly Rate	Mean prop.	January 2012 prop.	June 2017 prop.	Coefficient (unit increase per month)	[95% Conf.	Interval]	p-value	R-squared	January 2012 predicted value	June 2017 predicted value	Percentage Point Difference*		
Acute	amoxicillin	0.33	0.45	0.24	-0.002824	-0.003273	-0.002375	0.0000	0.7117	0.418076	0.234534	-19		
Kninosinusitis	amoxicillin with clavulanate	0.24	0.11	0.33	0.003009	0.002723	0.003295	0.0000	0.8734	0.141548	0.337153	20		
Acute Pharyngitis /	amoxicillin	0.16	0.30	0.09	-0.003214	-0.003517	-0.002911	0.0000	0.8751	0.260872	0.051951	-21		
	phenoxymethyl- penicillin	0.40	0.26	0.50	0.003621	0.003438	0.003803	0.0000	0.9608	0.284396	0.519731	24		
Acute Otitis Media	amoxicillin	0.46	0.56	0.46	-0.001656	-0.002131	-0.001182	0.0000	0.4315	0.515498	0.407845	-11		
	amoxicillin with clavulanate	0.32	0.20	0.30	0.001402	0.000772	0.002032	0.0000	0.2358	0.278309	0.369422	9		
Note: The rate calcula particular antibiotic. T Note *: The percentag	ition for each particular he denominator is all p ge point difference uses	antibiotic a atients with predicted v	gent was cal initial prese values for fire	lculated wit entations of st and last r	h the numerator l the same specific nonths of the stud	being all patient condition who dy period, Janua	ts with initial p were prescrib ary 2012 to Ju	presentation bed any anti ne 2017 inc	ns of the specif biotic agent. lusive.	ic condition w	ho were pres	cribed the		

 Table 6-3:
 Individual antibiotic agents prescribed for upper respiratory tract infection conditions: two antibiotics with the highest magnitude of statistically significant change

1 6.3.2.2 Acute pharyngitis / tonsillitis

2 There were two prescriptions for the second-line antibiotic recommended for pharyngitis, benzathine penicillin, occurring over duration of the study period. Second-3 line prescriptions were therefore considered negligible and too few to model so they 4 were excluded from the analyses. The antibiotic choices for pharyngitis were therefore 5 either first-line or not recommended (comprising non-first-line) agents. For initial 6 7 presentations of pharyngitis, there were significant downward trends in likely unnecessary prescribing (-0.0012, p<0.001, 95%CI: -0.0014, -0.0010) and overall 8 9 antibiotic prescribing (-0.0006, p<0.001, 95%CI: -0.0008, -0.0005) over the study period (Table 6-2). Antibiotic prescribing decreased from 73 to 69% among all 10 pharyngitis diagnoses, while unnecessary prescribing reduced from 89 to 82% of all 11 antibiotic prescriptions for the condition (Table 6-2). 12

13

There was also a significant downward trend in the prescribing of antibiotics not 14 recommended in the guidelines for pharyngitis (-0.0043, p<0.001, 95%CI: -0.0045, -15 0.0040), which is the same as non-first-line antibiotic prescribing for this condition 16 (Table 6-2). As depicted in Figure 6-8), the prescribing of not recommended antibiotics 17 decreased notably from 62 to 34% of all antibiotics prescribed for initial presentations 18 of pharyngitis. On visual inspection of Figure 6-8), one will note that a small peak in 19 mid-2014 and a small trough in early 2016 are notable variations of the linear trend 20 21 model from the smoothed data series for not recommended prescribing for pharyngitis. Also, the R-squared of the linear model was calculated at 96% (Table 6-2). Despite 22 variations, there is an apparent downward trend in not recommended prescribing 23 24 (Figure 6-8), which the linear model approximates.

- 25
- 26



 Figure 6-8: Time series plot of antibiotics not recommended in the guidelines for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017 inclusive, by month

For initial presentations of pharyngitis, there was a significant upward trend in phenoxy-methylpenicillin use (0.0036, p<0.001, 95%CI: 0.0034, 0.0038) and a significant downward trend for amoxicillin without clavulanate (-0.0032, p<0.001, 95%CI: -0.0035, -0.0029) (Table 6-3). First-line phenoxymethylpenicillin prescribing increased from 28 to 52% of all antibiotics prescribed for pharyngitis, while the use of not recommended amoxicillin without clavulanate decreased from 26 to 6% of all antibiotics for the condition (Table 6-3). The graphical depiction of both follows below (Figure 6-9) suggests that there has been a downward shift in the use of amoxicillin in exchange for phenoxymethylpenicillin.



Figure 6-9: Time series plot for phenoxymethylpenicillin and amoxicillin prescribing rates for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017 inclusive, by month

7 6.3.2.3 Acute otitis media

For patients with initial presentations of AOM, there were significant, upward trends in antibiotic prescribing (-0.0027, p<0.001, 95%CI: -0.0027, -0.0026), second-line prescribing (0.0015, p<0.001, 95%CI: 0.0008, 0.0021), and non-first-line prescribing (0.0016, p<0.001, 95%CI: 0.0011, 0.0020) during the study period (Table 6-2). However, there was no notable change in unnecessary antibiotic prescribing for initial presentations of AOM over time, (0.0027, p=0.147, 95%CI: -6.0E-05, 0.0004), which remained at a mean monthly rate of 77% of all antibiotic prescriptions for AOM. Similarly, the prescribing of not recommended antibiotic agents was also insignificant (0.0001, p=0.355, 95%CI: -0.0001, 0.0004), with a mean monthly rate of 21% of all antibiotics for AOM (Table 6-2).

2 The percentage of antibiotic prescriptions among diagnoses of AOM decreased by 3 seventeen percentage points from 67% to 50% (Figure 6-10, Table 6-2) There are several variations between the smoothed data series for antibiotic prescribing for AOM 4 and the linear trend model. A small trough in the second half of 2014 and a small peak 5 in the second half of 2015 are notable variations from the linear trend line clearly visible 6 7 in Figure 6-10, and R-squared for the linear model was 98% (Table 6-2). Despite variations, there is an apparent downward trend in antibiotic prescribing for AOM 8 9 (Figure 6-10). The linear model goes some way to summarizing this trend.



1



11

Figure 6-10: Time series plot of antibiotic prescribing rate among initial presentations of
 acute otitis media, January 2012 to June 2017, inclusive, by month

There was an increase in second-line prescribing for AOM, from 28 to 38% of all antibiotics for initial AOM (**Table 6-2**). Non-first-line prescriptions also increased from 48 to 59% of all antibiotic prescriptions for initial presentations of AOM, however, it is important to note that non-first-line prescribing appears to peak at over 60% in early 2015 then decrease again (**Appendix E.2.1.3**).

When examining individual antibiotic prescribed for initial presentations of AOM, there 1 was a significant downward trend in the prescribing of first-line amoxicillin (-0.0017, 2 p<0.001, 95%CI: -0.0021, -0.0012), and a significant upward trend in the use of 3 second-line amoxicillin with clavulanate (0.0014, p<0.001, 95%CI: 0.0008, 0.0020) 4 (Table 6-3). As seen in Figure 6-11, the use of amoxicillin decreased from 52 to 41% 5 of all antibiotics prescribed for AOM, while amoxicillin with clavulanate use increased 6 7 from 28 to 40% of all antibiotic prescriptions for the condition (Table 6-3). The peak in amoxicillin with clavulanate prescribing also appears to correlate reasonably in time 8 9 with the peak seen in non-first-line prescribing and may be relevant to this change. For additional results from this analysis, please see Appendix E. 10





Note: The denominator each rate is all patients prescribed any antibiotic for initial presentations of acute otitis media.

12

Figure 6-11: Time series plot for amoxicillin and amoxicillin with clavulanate prescribing for
 initial presentations of acute otitis media, January 2012 to June 2017, inclusive,
 by month

- 16
- 17

1 6.3 Summary

2

3 There was a common trend to all URTI conditions: the significant reduction in rates of antibiotic prescribing among initial presentations of URTI. There were significant 4 5 reductions in the rate of unnecessary prescribing for rhinosinusitis and pharyngitis but not AOM. For both rhinosinusitis and AOM, there were increases in second-line 6 7 prescribing and decreases in first-line prescribing. There were notable increases in second-line amoxicillin with clavulanate usage and decreases in first-line agent 8 amoxicillin, for both rhinosinusitis and AOM. For pharyngitis, however, there was an 9 increase in first-line usage and a decrease in the use of antibiotics not recommended 10 in the guidelines. This correlates to an increase in first-line phenoxymethylpenicillin 11 and a decrease in not recommended antibiotic amoxicillin for pharyngitis. 12

13

There were significant decreases in the rates of unnecessary prescribing for both 14 rhinosinusitis and pharyngitis. Pharyngitis was the only condition, however, with 15 decreases in both unnecessary prescribing and non-first-line antibiotic prescribing. 16 17 Despite a decreasing trend for antibiotic prescribing for AOM, there was no significant 18 change in unnecessary prescribing, and there was also an increase in non-first-line antibiotic use. For both rhinosinusitis and AOM, there were increases in second-line 19 20 and non-first-line antibiotic prescribing. There was a dramatic increase in the use of second-line antibiotics for rhinosinusitis. 21

22

While the largest reduction in unnecessary prescribing was for rhinosinusitis, this was 23 24 accompanied by increasing non-first-line prescribing. The mean monthly rate of unnecessary prescribing was 91% among all antibiotic prescriptions for patients with 25 26 rhinosinusitis, which was the highest among all URTI conditions. For pharyngitis, there 27 were reductions in the three most prominent outcomes: overall antibiotic prescribing, unnecessary prescribing, and non-first-line antibiotic prescribing. It is on this basis one 28 might consider there being consistent reductions in prominent outcomes for 29 pharyngitis. 30

31

To this end, for AOM, there was no significant change in unnecessary prescribing (p=0.1468), as well as increasing non-first-line prescribing. The mean monthly rate of unnecessary prescribing was 0.77 among all antibiotic prescriptions for patients with initial AOM. This is noteworthy, given that AOM does not usually require an antibiotic prescription in most circumstances (29). However, an alternate perspective is that
AOM also had the lowest mean rates for both unnecessary prescribing and non-firstline prescribing, and also represents the smallest cohort, and may therefore be in less
need of change.

5

6 For patients with acute rhinosinusitis, the magnitude of change was similar for overall 7 antibiotic prescribing and unnecessary prescribing, which were thirteen percentage points and twelve percentage points, respectively. However, for patients with 8 9 pharyngitis / tonsillitis, the reduction of unnecessary prescribing of eight percentage points was twice that of the decrease in antibiotic prescribing of four percentage points. 10 11 For rhinosinusitis, the magnitude of change was also similar for second-line and nonfirst-line prescribing for rhinosinusitis, with increases of twenty and eighteen 12 percentage points, respectively. There was a 28 percentage point reduction for not 13 recommended / non-first-line antibiotic prescribing for pharyngitis. Meanwhile, for 14 AOM, there were increases in second-line and non-first-line prescribing, of nine and 15 ten percentage points, respectively. With regard to antibiotic prescribing, the 16 seventeen percentage point reduction for AOM outweighs that of rhinosinusitis with a 17 thirteen percentage point decrease and a four percentage point reduction for 18 pharyngitis. However, this notable reduction in antibiotic prescribing for AOM must be 19 considered in balance with no significant reduction in unnecessary prescribing for this 20 21 condition.

22

Regardless of the perspective taken, it is apparent that there are meaningful changes
occurring in prescribing behaviours among all URTI conditions over the study period,
and that many avenues for improvement can be identified from these results.

26

- 28
- 29

CHAPTER 7 ANALYSIS OF CHANGES IN ANTIBIOTIC PRESCRIBING FOR URINARY TRACT INFECTION OVER TIME

3

4 5

7.1 Introduction

6 The objective of this analysis was to explore antibiotic prescribing for patients with initial presentations of UTI / acute cystitis over time. As empirical antibiotic prescribing is 7 8 accepted for initial episodes of UTI (29,123), the main question is not so much whether prescribing occurs but rather what specific antibiotics are being prescribed. Inappropriate 9 10 antibiotic prescribing is commonly reported from primary care internationally for UTIs (15,17,29,48,119-122,278,283,285,296,305,323,336,387,415,416). While the 11 local 12 epidemiology of UTI is not expected to change substantially over a relatively short period of time, it is possible that antibiotic prescribing behaviour may vary. This is particularly 13 14 given growing antibiotic resistance among UTI pathogens nationally (12,14,18,167,417), 15 as well as increasing publicity regarding antibiotic resistance and antibiotic stewardship, as detailed in the Background chapter. In light of these factors, one would hope that 16 decreases in non-first-line prescribing for UTI was occurring over time, which would 17 represent favourable changes in the antibiotic prescribing (275). 18

19

It was therefore of interest to establish whether there were notable reductions in the 20 proportions of non-first-line antibiotics being prescribed to patients with initial 21 presentations for UTI over the study period, and, if so, by how much. It was also a focus 22 to identify any substantial changes in the use of specific antibiotics over this time, and 23 particularly in light of local reports of changing antibiotic use (12,14,18,167). The objective 24 was to explore these factors for all patients with initial presentations of UTI, and also for 25 independent patient groups defined in the Therapeutic Guidelines: Antibiotic (the 26 27 guidelines). These patient groups were men, non-pregnant women, and children (29).

28

Previously defined datasets and outcome variables, as described in the Methods chapter, were used in this analysis (**Figure 7-1**). UTI was defined to include acute cystitis (29). In order to accurately evaluate the antibiotic prescribing at initial presentations for UTI, noninitial and chronic / recurrent consultations were excluded, by removing consultations occurring within fourteen days of a previous UTI consultation for the same patient.



Figure 7-1: Depiction of the main outcome variables utilised in this trends analysis, and previous analyses for initial presentations of urinary tract infection

Table 7-1 serves as a reminder of the guidelines for treatment of initial presentations of UTI, which were used to categorise each specific antibiotic agent into ordinal choice based on their order of recommendation in the guidelines (29). Antibiotics which were prescribed but are not listed in **Table 7-1** were classified as not recommended. For more information, please refer to the **Methods chapter**.¹²

10

1

Improved antibiotic stewardship should result in a reduction in the proportion of prescribed antibiotics that were non-first-line antibiotics over time for all patient groups. By definition, this would also equate to an increase in the proportion of first-line antibiotics prescribed for these initial episodes of UTI, representing improved compliance by clinicians with the guidelines (29).

¹² For more information, please refer to the Methods chapter.

Table 7-1:	Fable 7-1:Summary of Therapeutic Guideline: Antibiotic (29) recommendations in order of choice for acute cystitis, by patient group.												
Choice	Non-pregnant Women	Men	Children >= 1 month										
First-line	trimethoprim	trimethoprim	trimethoprim										
Second-line	cefalexin	cefalexin	cefalexin										
Third-line	amoxicillin + clavulanate	amoxicillin + clavulanate	amoxicillin + clavulanate										
Third-line	nitrofurantoin	nitrofurantoin											
Last Resort	norfloxacin	Norfloxacin	norfloxacin										

Note: A first-line option should be the antibiotic prescribed at initial consultations. Third-line and Last Resort options were combined into the third ordinal level for analysis. Where the antibiotic prescribed is not listed as an option for the condition diagnosed, the prescription was classified as 'not recommended'.

3

1 2

Note that the linear trend models utilise five-month moving average data, which were used to calculate mean monthly rates (414). The percentage point differences reported use the predicted values of these linear models at the start and end of the study period, January 2012 and June 2017, respectively. For more information, please refer to the Methods chapter. There were many outcomes analysed, for all patients with initial presentations of UTI, as well as for each patient group, however, only the most relevant results are presented here. For further results from this analysis, please see Appendix F.

11

12 7.2 Results

13

Each month in the study period between 72 and 83% of UTI diagnoses were women of at least sixteen years of age, while between 8 and 16% were children under sixteen years of age (**Figure 7-2**). Among all patients with initial presentations of UTI, between 81-94% of women, 58-89% of men, and 61-90% of children under sixteen, per month, received antibiotic prescriptions. Across all three patient groups receiving antibiotic prescriptions, the mean monthly percentage of each of these patient groups receiving non-first-line antibiotics remained at over 55% throughout the study period.



Figure 7-2: Graph of proportions of initial presentations of urinary tract infection, by patient group (women, men, and children under sixteen years) from January 2012 to June 2017, inclusive, by month

7.2.1 All patients with initial presentations of urinary tract infection

For all patients with initial presentations of UTI, there was a significant, increasing trend in the proportions of patients receiving second-line antibiotics, increasing by 0.0016 per month (p<0.001, 95%CI: 0.0014, 0.0017) (Table 7-2). Although the raw coefficients from these linear regression models of the monthly aggregate data in Table 7-2 appear 'small', it is important to realise that these represents unit changes per month. For example, when considering second-line prescribing, the linear regression coefficient of 0.0016 unit increase change per month extrapolates to an increase of two percentage points per year. When this is extrapolated to the five-and-a-half-year study period, this corresponds to a ten percentage point increase in second-line prescribing, up to 46% of all antibiotics prescribed at the end of the study.

Non-first-line prescribing increased from 56 to 64% of all antibiotic prescriptions over the study period. The linear regression model indicates an increase of 0.0009 per month (p<0.001, 95% CI: 0.0007, 0.0011), equivalent to an eight percentage point increase overall in non-first-line prescribing for UTI (**Table 7-2**). By definition, this means that there was an equal and opposite decrease in first-line antibiotic prescribing for all patients with initial presentations of UTI.

7

There was also a downward trend in the use of antibiotics not recommended in the guidelines for UTI of -0.0005 (p<0.001, 95%CI: -0.0006, -0.0003) (**Table 7-2**). The use of antibiotic agents not recommended in the guidelines at all decreased from 10 to 7% of all antibiotic prescriptions for all patients. See **Appendix F** for more detail.

12

14

13 7.2.2 Trends by patient group

After considering overall trends for all patients with initial UTI presentations altogether, the focus will now be on the individual patient groups, which is important as they have different causes, guidelines for treatment, and risks. Recall that the predicted values from linear regression models are used to report percentage point change for each outcome.

19

20 <u>7.2.2.1 Women</u>

Women of sixteen years and over with initial presentations of UTI comprise the majority 21 of initial UTI diagnoses so it is expected that the prescribing for women demonstrates 22 similar trends identified to those for all patients. Among these women prescribed any 23 antibiotic, there were significant upward trends in second-line prescribing, increasing by 24 0.0016 per month (p<0.001, 95%CI: 0.0015, 0.0018) and an eleven percentage point 25 increase over the 66 months (Table 7-2). As seen in Figure 7-3, second-line prescribing 26 among all antibiotics prescribed to women with initial presentations of UTI increased from 27 28 34 to 44%. As with all the models presented in this chapter, the 'fit' of the linear trend model in Figure 7-3 below varies from the smoothed series, particularly the peak in mid-29 2015. In this situation, the linear trend model for second-line prescribing varies with an 30 R-squared of 87%, which is high for an error term. However, there is an undeniable 31 32 upward trend overall, which the strictly linear model can reasonably estimate.

There was an increasing trend in non-first-line prescribing of 0.0010 per month (p<0.001, 1 95%CI: 0.0008, 0.0012), implying a seven percentage point increase over the study 2 period, from 53 to 60% of all antibiotics prescribed to women (Table 7-2). The increase in 3 second-line prescribing may be responsible for the increase in non-first-line prescribing. 4 There was also a significant downward trend in the prescribing of antibiotics not 5 recommended in the guidelines for women, decreasing by 0.0004 per month (p<0.001, 6 95%CI: -0.0005, -0.0003), implying a three percentage point decrease overall (Table 7-7 8 2).



9

Figure 7-3: Time series plot of second-line antibiotic prescribing for all women sixteen years and over with initial presentations of urinary tract infection, from January 2012 to June 2017, inclusive, by month

13

By specific antibiotic, there was a significant, upward trend in the use of cefalexin among women, with an increase of 0.0016 per month (p<0.001, 95%CI: 0.0015, 0.0018). This represents an eleven percentage point increase (**Table 7-3**). There was also a seven percentage point reduction in (first-line) trimethoprim use, decreasing by 0.0010 per month (p<0.001, 95%CI: -0.0012, -0.0008). **Figure 7-4** depicts the trimethoprim prescribing rate reduction from 47 to 40%, which appears to mirror the cefalexin increase.

	Prescribing Outcomes for All Patients with initial presentation of UTI													
	Dependent Variable	a) (Mo	Descriptive	<u>Statistics</u> ge Data)	b) Linear Regression for Trend									
Patient Group	Prescribing Outcome Monthly Rate	Mean prop.	January 2012 prop.	June 2017 prop.	Coefficient (unit increase per month)	[95% Conf.	Interval]	p-value	R-squared	January 2012 predicted	June 2017 predicted value	Percentage Point Difference *		
All Patients	Prescribing of Second-line	0.40	0.35	0.46	0.001555	0.0014094	0.001701	0.0000	0.8764	0.345626	0.462771	11		
	Prescribing of antibiotic Not Recommended in guidelines	0.09	0.12	0.09	-0.00045	-0.000595	-0.000306	0.0000	0.378	.1031465	.0738786	-3		
	Non-first-line (non-first-line) antibiotic prescribing	0.60	0.45	0.36	0.000907	0.0007352	0.0010792	0.0000	0.6344	0.561391	0.639921	8		
Women at least 16	Prescribing of Second-line	0.39	0.33	0.45	0.001626	0.001471	0.001782	0.0000	0.8728	0.336198	0.441917	11		
	Prescribing of antibiotic Not Recommended in guidelines	0.09	0.12	0.09	-0.00039	-0.00052	-0.00026	0.0000	0.3666	0.100878	0.075505	-3		
	Non-first-line (non-first-line) antibiotic prescribing	0.56	0.51	0.60	0.001015	0.000833	0.001197	0.0000	0.6604	0.530338	0.596313	7		
Men at	Prescribing of Second-line	0.36	0.35	0.43	0.001421	0.001083	0.00176	0.0000	0.5241	0.312730	0.405115	9		
10 years	Prescribing of antibiotic Not Recommended in guidelines	0.07	0.13	0.10	-9.5E-05	-0.00041	0.000218	0.5463	0.0057	0.076461	0.070279	-1		
	Non-first-line (non-first-line) antibiotic prescribing	0.67	0.67	0.77	0.001969	0.00132	0.002619	0.0000	0.3642	0.606071	0.734079	13		

Table 7-2: Prescribing outcomes for each patient group with initial presentations of urinary tract infection, by patient group

<u>Patient</u> <u>Group</u>	Prescribing Outcome Monthly Rate	Mean prop.	January 2012 prop.	June 2017 prop.	Coefficient (unit increase per month)	[95% Conf.	Interval]	p-value	R-squared	January 2012 predicted	June 2017 predicted value	Percentage Point Difference *
Children 16 years	Prescribing of Second-line	0.56	0.54	0.59	0.000975	0.000737	0.001214	0.0000	0.5101	0.528409	0.591799	6
	Prescribing of antibiotic Not Recommended in guidelines	0.11	0.18	0.10	-0.00126	-0.00156	-0.00095	0.0000	0.5099	0.147108	0.065484	-8
	Non-first-line (non-first-line) antibiotic prescribing	0.83	0.87	0.84	-0.00055	-0.00083	-0.00026	0.0003	0.1871	0.843880	0.808382	-4
Note: The rate	l e calculation for each particular pre	scribing out	come withir	n each patie	nt group was calc	ulated with the	numerator be	eing all pation	ents within pati	ent group wit	h initial prese	ntations of
UTI, who were	e prescribed an antibiotic classified	as being par	rt of the part	ticular prese	cribing outcome.	The denominato	or is all patien	ts within the	e same patient	group with in	itial presentat	ions of UTI
who were pres	who were prescribed any antibiotic. For example, second-line antibiotic prescribing rate for men was calculated using all men at least sixteen years of age with initial presentations of UTI who were											
prescribed an	antibiotic classified as second-line,	over all mei	n at least six	teen years	of age with initial	presentations o	012 to lune 2	re prescribe	ed any antibioti	с.		

	Prescribing Rates for Individual Antibiotic Agents for Women >= 16 years with initial presentations of UTI													
	<u>Dependent</u> <u>Variable</u>	c) (Movi	Descriptive	<u>Statistics</u> Data)	d) Linear Regression Model for Trend									
Patient Group	Prescribing Outcome Monthly Rate	Mean prop.	January 2012 prop.	June 2017 prop.	Coefficient (unit increase per month)	[95% Conf.	Interval]	p-value	R-squared	January 2012 predicted	June 2017 predicted value	Percentage Point Difference*		
Women at least 16 years	amoxicillin cefalexin	0.03	0.07	0.01	-0.00072	-0.00085	-0.00059	0.0000	-0.00085	0.048543	0.001880	-5		
	trimethoprim	0.44	0.49	0.40	-0.001020	-0.0012	-0.00083	0.0000	-0.0012	0.469662	0.441917	-7		
Men at least 16 years	amoxicillin cefalexin trimethoprim	0.03 0.36 0.33	0.08 0.35 0.33	0.02 0.43 0.23	-0.00058 0.001421 -0.00197	-0.00078 0.001083 -0.00262	-0.00038 0.00176 -0.00132	0.0000 0.0000 0.0000	-0.00078 0.001083 -0.00262	0.046490 0.312730 0.393929	0.008796 0.405115 0.265921	-4 9 -13		
Children under 16 years	amoxicillin cefalexin trimethoprim	0.08 0.56 0.09	0.19 0.54 0.04	0.07 0.59 0.09	-0.00152 0.000975 0.000629	-0.00188 0.000737 0.000412	-0.00117 0.001214 0.000846	0.0000 0.0000 0.0000	-0.00188 0.000737 0.000412	0.130712 0.528409 0.073813	0.031651 0.591799 0.114707	-10 6 4		

Table 7-3: Individual antibiotic agents prescribed for urinary tract infection conditions: three antibiotics with the highest magnitude of statistically significant change

Note: The rate calculation for each particular antibiotic agent was calculated with the numerator being all patients within specific patient group with initial presentations of UTI who were prescribed the particular antibiotic. The denominator is all patients within specific patient group with initial presentations of UTI who were prescribed any antibiotic agent. **Note *:** The percentage point difference uses predicted values for first and last months of the study period, January 2012 to June 2017 inclusive.



3 4 5

Figure 7-4: Time series plot of simple moving average prescribing rates for cefalexin and trimethoprim for women, January 2012 to June 2017, inclusive, by month

6 <u>7.2.2.2 Men</u>

For men at least sixteen years of age with initial presentations of UTI, there was a significant, upward trend in second-line antibiotic prescribing of 0.0014 per month (p<0.001, 95%CI: 0.0011, 0.0018), equating to a nine percentage point increase over the study period (**Table 7-2**). Note that in **Figure 7-5** the linear model varies from the smoothed data at multiple points. Although a strict linear trend may not appear very appropriate for this reason (and noting an R-squared of 52%), there is a clear upward trend.

13

There was also an overall thirteen percentage point increase in non-first-line prescribing of 0.0020 per month (p<0.001, 95%CI: 0.0013, 0.0026), among all antibiotics prescribed to men with initial presentations for UTI (**Table 7-2**). There was no statistically significant trend for the prescribing of not recommended antibiotics for men, with a coefficient of -9.5E-05 per month (p=0.546, 95%CI: -0.0004, 0.0002), which were prescribed at a mean monthly rate of 7% (**Table 7-2**).



1

5

Figure 7-5: Time series plot of second-line antibiotic prescribing for all men at least sixteen
 years of age with initial presentations urinary tract infection, from January 2012 to
 June 2017, inclusive, by month

For individual antibiotics prescribed to men, there was a significant downward trend in
the prescribing rate for first-line trimethoprim, reducing by 0.0020 per month (p<0.001,
95%CI: -0.0026, -0.0013), equating to a thirteen percentage point decrease overall (Table
7-3). There was also an upward trend in cefalexin use of 0.0014 per month (p<0.001,
95%CI: 0.0011, 0.0018), a nine percentage point increase from 31 to 41% of all antibiotics
for men (Table 7-3). These changes in trimethoprim and cefalexin prescribing may be
relevant to the ten percentage point increase in non-first-line prescribing for men.

13

14 <u>7.2.2.3 Children</u>

Among children under sixteen years of age with initial presentations of UTI and prescribed any antibiotic, there was a significant, upward trend in second-line prescribing, increasing by 0.0010 per month (p<0.001, 95%CI: 0.0007, 0.0012) (**Table 7-2**). This is equivalent to a six percentage point increase, from 53 to 59% of all antibiotics prescribed to children over the study period.

There was also a significant, downward trend in the use of antibiotics not recommended 1 in the guidelines for children, decreasing by 0.0013 per month (p<0.001, 95%CI: -0.0016 2 -0.0010) (Table 7-2). This equates to an eight percentage point decrease overall from 15 3 to 7% of antibiotics for children. While this strictly linear model in Figure 7-6 below may 4 not appear especially appropriate to the data series, and noting an R-squared of 51%, 5 the point is that there is a clear downward trend which can be estimated by the linear 6 trend line. Non-first-line prescribing to children also decreased by 0.0006 per month 7 (p=0.0003, 95%CI: -0.0008, -0.0003), from 84 to 81% of all antibiotics prescribed to 8 9 children (Table 7-2). This is substantially higher than for other patient groups, and especially noting that these data were restricted to initial presentations of UTI. 10





12

Note: Denominator is all children under sixteen years of age prescribed any antibiotic for initial presentations of urinary tract infection.

13Figure 7-6:Time series plot of prescribing for not recommended antibiotic for children under14sixteen years of age with initial presentations of urinary tract infection, from15January 2012 to June 2017, inclusive, by month

16

By individual antibiotic prescribed to children, there was a significant, upward trend in cefalexin use, increasing by 0.0010 per month (p<0.001, 95%CI: 0.0007, 0.0012), representing a six percentage point increase over the study (**Table 7-3**). There was a downward trend in use of the antibiotic, amoxicillin without clavulanate, which is not recommended, decreasing by 0.0015 per month (p<0.001, 95%Cl: -0.0019, 0.0012), from thirteen to three percent (**Table 7-3**). Trimethoprim use also increased by 0.0006 per month (p<0.001, 95%Cl: 0.0004, 0.0008), equivalent to a four percentage point increase, as seen in **Figure 7-7**. Both the increase in first-line trimethoprim and decrease in not recommended amoxicillin may contribute to the decrease in non-first-line prescribing. For additional results from this analysis, please see **Appendix F**.

7



Note: The denominator for each rate is all antibiotics prescribed to children under sixteen years with initial presentations of urinary tract infection.

8

9 Figure 7-7: Time series plot of prescribing for antibiotic agents not recommended in the 10 guidelines for children under sixteen years of age with initial presentations of 11 urinary tract infection, from January 2012 to June 2017, inclusive, by month

12

13 7.3 Summary

14

For patients with initial presentations of UTI, there are upward trends in the prescribing of second-line antibiotics for all patient groups: women, men and children. There are upward trends in non-first-line antibiotic prescribing over time for adult patient groups, however, a downward trend for children. There were also notable downward trends in the prescribing of not recommended antibiotics for children and women.

There was a consistent increase in second-line prescribing for all patient groups during the study period, which can be attributed to the antibiotic cefalexin, as the only secondline antibiotic. The reduction in the prescribing of antibiotics not recommended in the guidelines for women and children may be in part related to decreasing use of amoxicillin without clavulanate.

6

Using the smoothed data, all patient groups with initial UTI had mean monthly non-firstline antibiotic prescribing rates of the over 55% of all antibiotics within each patient group, and markedly so in the case of children, with a mean of 83% prescriptions being nonfirst-line. Children were the patient group with the highest mean proportion of non-firstline (83%), second-line (56%), and not recommended (11%) prescribing, as a proportion of all antibiotics prescribed to each patient group.

13

Among the three patient groups with initial presentations of UTI, the most substantial 14 progress over the study period appeared to be for children, however, with respect to the 15 use of non-first-line, second-line and not recommended antibiotics. To this end, children 16 were the only patient group with a significant reduction in non-first-line prescribing. The 17 18 highest reduction in the use of not recommended antibiotics was also in children, and they also had the lowest increase in second-line antibiotics. The analysis also identified 19 20 that prescribing differs by UTI patient group, which is expected and will be discussed in 21 more detail.

22

It is clear that there was an increasing reliance of clinicians upon cefalexin for initial presentations of UTI during this period is clear. For adults, there may have been a gradual shift away from trimethoprim towards cefalexin. While cefalexin use also increased for children, there might be an interesting shift away from amoxicillin an increased use of trimethoprim.

28

These results suggest that, although there is evidence of most progress towards appropriate prescribing according to the guidelines in children, this may well be this patient group in most need of further focus in the future.

- 32
- 33

1 CHAPTER 8 DISCUSSION

2

8.1 Overall findings

3 4

5 This thesis demonstrates that substantial inappropriate prescribing of systemic antibiotics 6 occurred in WA general practice, for initial episodes of care for URTI and UTI conditions, 7 contrary to the national guidelines, called Therapeutic Guidelines: Antibiotic (the guidelines) 8 (29). This inappropriate prescribing takes multiple forms: unnecessary prescribing for URTI 9 conditions, non-first-line antibiotics prescribed for initial presentations of URTI and UTI 10 without justification, and unnecessary repeats issued on many prescriptions (29).

11

For initial episodes of care for URTI, the strong majority (85%) of antibiotic prescriptions were found to be unnecessary. Of antibiotic prescriptions issued at these initial presentations, 60% were issued with non-first-line antibiotics. Repeats were issued on prescriptions nearly a third of the time, the majority of which were likely not required. Older patients had higher probability of receiving unnecessary antibiotic prescribing and non-firstline prescribing. Rhinosinusitis was the condition with the highest chance of unnecessary prescribing and non-first-line prescribing.

19

20 For initial presentations of UTI, 60% percent of patients with initial episodes of care for UTI received non-first-line antibiotics. Given that 82% of children under sixteen years received 21 non-first-line antibiotics for initial consultations, children more commonly receive poor 22 prescribing choices than other age groups. Culture testing occurred for only 2% of children, 23 3% of women and 3% of men. There were no children or adult males receiving prescriptions 24 for initial UTI who received both culture and sensitivity testing, which are specified in the 25 guidelines (29) as mandatory for children and strongly recommended for men. Repeats were 26 issued on 28% antibiotic prescriptions for UTI. 27

28

The fact that these results are drawn from multiple levels of inappropriate prescribing further points to significant, overarching problems in general practice with respect to adherence to the guidelines (29). There is therefore substantial room (and need) for improvement in antibiotic prescribing for URTI and UTI in WA general practice. These multiple levels of inappropriate prescribing also help identify various opportunities for antibiotic stewardship and interventions in primary care. Furthermore, it was individual GP providers, rather than

practices, that were found to be source of most variance unexplained by fixed effects. This 1 suggests that individual providers should be the focus for improving prescribing practices. 2

3

It is hopeful that this research will prove informative for general practice. The high 4 proportions of unnecessary and non-first-line antibiotic prescribing for initial presentations 5 6 of URTI and UTI conditions are concerning. While the trends analyses demonstrate some 7 minor improvements over time, the overall situation presents a challenge for policy-makers. 8 GPs need to do more to take responsibility and take proactive steps to improve their prescribing practices but they also need substantial government support and investment to 9 facilitate this. 10

- 11
- 12

13

8.2 Upper respiratory tract infection

Main descriptive findings for upper respiratory tract infection 14 8.2.1

15

This thesis found that for initial presentations of URTI, 85% of antibiotic prescriptions were 16 unnecessary, and 60% of antibiotic prescriptions were non-first-line, when first-line should 17 be the norm. Repeats were also issued on prescriptions 32% of the time. In the Australian 18 setting, it is known that there are high rates of antibiotic prescribing, the strong majority of 19 which are believed to be unnecessary, in addition to low use of first-line antibiotics 20 (12,14,17,18,26,418). For example, the second AURA report (17) for a similar time period 21 as this thesis found that tonsillitis diagnoses received antibiotic prescriptions on 71% of 22 occasions, however, first-line antibiotics were received by only 39% for tonsillitis. In 2022, 23 Monaghan et al. (26) also found acute rhinosinusitis was the condition most likely to receive 24 prescriptions when likely not required. High proportions of antibiotic prescriptions with 25 repeats have also previously been noted (14,18,419). Similarly, in the international setting, 26 high antibiotic prescribing rates and higher than ideal non-first-line prescribing is commonly 27 reported from general practice (268,276,277,322,420). 28

- 29
- 30

8.2.2 Predictors identified for upper respiratory tract infection models

31

The various models for inappropriate prescribing for URTI found that patient age group was 32 significant in all models, and young children 0-8 years were at substantially lower likelihood 33 of the outcome in all models. The three outcomes: unnecessary prescribing; ordinal choice 34 of antibiotic prescribing; and repeat positive prescribing predicted each other in the models, 35

suggesting these are all related. In addition to patient age, URTI condition was a significant
 predictor in these three models.

3

The highest-level model, inappropriate decisions, identified that patients 22-34 years, small 4 practice size and patients with penicillin allergy labels had the highest probability of 5 6 inappropriate decisions. Predictors of unnecessary prescribing included patient age group, 7 mental health condition status, patient concession status, patient penicillin allergy label 8 status, URTI condition, number of URTI episodes, ordinal line of antibiotic prescribed and repeat prescription status. The chance of unnecessary prescribing increased with increasing 9 age. By URTI condition, the probability of unnecessary prescribing was highest for 10 rhinosinusitis. Patients with penicillin allergy labels and patients with repeats issued on 11 12 prescriptions were associated with lower chances of receiving unnecessary prescriptions. Note that despite having the same numerator, the difference in denominators between 13 14 model 0 (inappropriate decisions among diagnoses) and (unnecessary prescribing among prescriptions) may explain some of the seemingly divergent results. 15

16

The predictors for the ordinal choice of antibiotic model included patient age group, 17 prescribing reason recorded, URTI condition, multiple URTI episode status, patient 18 comorbid conditions status, patient socioeconomically disadvantaged status, practice size, 19 as well as unnecessary prescribing status and repeat prescription status. Patients aged 22-20 34 years were least likely to receive first-line and most likely to receive not recommended 21 antibiotics. Unnecessary prescriptions had higher chances of being for non-first-line 22 antibiotics than necessary prescriptions. Second-line featured minimally throughout URTI 23 models. Instead, any decrease in first-line prescribing appeared to be met with a similar 24 25 increase in not recommended antibiotics, and vice versa.

26

27 One interpretation of the increased probability of non-first-line prescribing in necessary prescribing situations compared to unnecessary prescribing situations is that the GPs might 28 resort to prescribing the safest option (first-line) for patients despite knowing that a 29 30 prescription is technically unnecessary. Whereas in situations where prescribing is 31 genuinely necessary, GPs might tend to give non-first-line prescriptions to try to cover more bases susceptibility-wise. This finding bears resemblance to an Irish questionnaire-based 32 33 study (22), in which first-line antibiotics were commonly prescribed when the GP considered a prescription not strictly necessary but were prescribed anyway. 34

Repeats being issued on prescriptions being linked to lower probabilities of unnecessary 1 prescribing is equivalent to necessary prescribing being more likely to occur with repeats on 2 the prescriptions. One feasible explanation is that GPs may tend to prescribe antibiotics with 3 repeats on the prescription for the most seriously unwell patients in need of treatment, such 4 as, necessary prescribing situations. Receiving repeats on prescriptions was linked to an 5 6 increased probability of receiving non-first-line antibiotics for adults but to a decreased probability of non-first-line antibiotics for children. This may reflect that GPs may be more 7 8 cautious with choice of antibiotic and whether to prescribe repeats when dealing with children. GPs may also be more comfortable issuing repeats for first-line agents for children. 9

10

11 8.2.2.1 Patient age

12 The likelihood of inappropriate prescribing increasing with increasing patient age, as found here, has been found in several studies in terms of both unnecessary prescribing and non-13 14 first-line prescribing (25,280,281,284,296,324). With regard to inappropriate decisions and unnecessary prescribing in this research, young children were at substantially lower 15 likelihood than other age groups, and young adults 22-34 years were at slightly higher than 16 older children and older adults. These results are comparable to other findings, in terms of 17 adults being a notably higher probability of inappropriate prescribing than children. For 18 example, a study of respiratory tract infection (RTI) in the Netherlands found that adults 18-19 65 years received proportionally more unnecessary prescriptions than children (281). In their 20 study of Norwegian patients with RTI, Gjelstad et al. (329) also found that younger patient 21 age was linked to lower probability of non-first-line antibiotics, as was the case here. Another 22 study of children with URTI conducted in the UK found increasing prescribing rates with 23 increasing patient age (420). As found in this thesis, an Australian study of patients 24 25 presenting to the emergency department with URTI and other conditions found that increasing patient age was associated with inappropriate prescribing (48). An Italian study 26 27 of paediatric URTI using patient interviews and voluntary prescriber participation also found that inappropriate prescribing increased with increasing age (421). 28

29

8.2.2.2 Interaction between patient age group and gender in inappropriate decision model

In the inappropriate decision model, there was an interaction between patient age group and gender. While for young children, males had higher probabilities than females of receiving inappropriate decisions but the opposite was true for patients aged 9-21 years and 22-34 years, while there was no difference among patients 35 years and over. One study of US adults found that male patients were less likely to receive inappropriate prescriptions than females for RTI presentations (326), while a Belgian study found that females 30-60 years more likely to receive appropriate prescribing than males (280), both of which support these findings. Male children were also more likely subject to non-recommended management than females in a Canadian study (328). A Spanish study of female patients over fourteen years with acute bronchitis were less likely to receive appropriate management than men regarding the prescribing decision, akin to the findings of this research (341).

8

9 <u>8.2.2.3 Upper respiratory tract infection condition</u>

Two Norwegian studies, one of which was for children, found that patients with acute tonsillitis were less likely to receive non-first-line agents than patients with URTI (297,329), as was the case here. Steinman et al. (276) examined non-first-line antibiotic prescribing to adult US outpatients with non-pneumonic acute RTI and they found that other URTIs, namely laryngitis, pharyngitis and tracheitis, were less frequently associated with non-firstline prescribing than non-specific URTI and common cold, which was also the finding here.

16

17 8.2.2.4 Patient allergy labels for penicillin

Having an allergy label for penicillin appeared to predispose patients in this research to 18 receiving more inappropriate decisions and yet it appeared linked to a lower probability of 19 unnecessary prescribing. Appropriate non-prescribing is included in the denominator for 20 inappropriate decisions but not unnecessary prescribing, and it represents the difference 21 between the two response variables. The difference in the direction of effect between the 22 two models may suggest that patient penicillin allergy labels are closely linked to appropriate 23 non-prescribing. Schroek et al. (332) found that US patients with penicillin allergy presenting 24 25 with RTI had increased the likelihood of inappropriate treatment, despite allergy being allowed for in antibiotic choice, as in this thesis. Patients in another study with reported beta-26 27 lactam allergies are likely to receive inappropriate antibiotic choice (422). A Singaporean oncology study found that patients with antibiotic allergy labels or comorbidities were linked 28 29 to inappropriate prescribing, as was found in this research (423).

30

31 8.2.2.5 Practice size

In this research, small practice size was a predictor of non-first-line choice of antibiotics.
 Otters et al. (271) found that GPs working in solo practices in the Netherlands were more
 likely to prescribe a non-first-line antibiotic to children for cough, acute upper airway infection
 or acute bronchitis than group practices, which supports the findings of this research.

Multiple studies did, however, find that prescribers with higher volume patient turn-over and / or higher overall antibiotic prescribing rates were predisposed to inappropriate prescribing (270,277,284,297,321,329). Gjelstad et al. (329) found that GPs with shorter patient lists were less likely to issue non-first-line prescriptions for RTI than those with regular patient lists. Solo *versus* group physician practices has been associated with poorer patient outcomes in a US study of acute myocardial infarction (424).

7

8 8.

8.2.2.6 Patient comorbid and mental health conditions

While this thesis found patient mental health condition status was a predictor of unnecessary 9 prescribing, patient comorbid condition status was found to be a predictor of non-first-line 10 antibiotics prescribed for URTI. A Spanish study of patients with acute bronchitis (39) also 11 12 found that the antibiotic chosen may prove important for patients with comorbid conditions. Several studies have found that patients with comorbid conditions presenting with URTI (and 13 in some cases additional conditions) were more frequently in receipt of inappropriate 14 antibiotic prescriptions than patients without these conditions, particularly poor antibiotic 15 selection (39,281,294,324,326), as was the case here. Bernado et al.'s Australian study (41) 16 also found that patients with comorbid conditions including mental health conditions were 17 linked to higher rates of antibiotic prescribing for ILI, although this utilised the same 18 MedicineInsight dataset but at a national level rather than WA-based. The influence of 19 mental health condition status on likely unnecessary prescribing suggests that clinicians 20 may try to alleviate patient / clinician concerns regarding the health condition and avoid any 21 potential exacerbation of mental health conditions by prescribing despite not it being 22 indicated. It may be a similar misplaced sense of concern for patients with comorbid 23 condition being predisposed to receiving non-first-line antibiotic prescriptions. 24

25

26 <u>8.2.2.7 Other</u>

A small Israeli study of children with AOM had similar findings to those here regarding altered prescribing on the weekends and leading up to the weekend (359). The fact that weekends are associated with lower probability of unnecessary prescribing may reflect less time pressure on weekends during which to explain to patients why they likely do not need antibiotics. High practice activity / high patient loads (and therefore time pressure) have been linked to more frequent inappropriate prescribing for patients with URTI (329,425).

33

It is feasible that prescribers who may diligently complete the reason for prescribing field may also prove more diligent in adhering to the guidelines. There was also infrequent culture and sensitivity / susceptibility testing for these URTI presentations, not unlike Gunnarsson
et al.'s (426) multinational study of the management of patients with sore throat. Tran et al.
(427) note that the guidelines do not provide sufficient advice on diagnostic testing for
bacterial sore throat.

- 5
- 6

7

8.2.3 Trends in prescribing for upper respiratory tract infection

8 In this study, there were reductions in the rate over time of likely unnecessary prescribing for rhinosinusitis and pharyngitis but not AOM. There were increases in second-line 9 10 amoxicillin with clavulanate usage and decreases in first-line agent amoxicillin, and thus increasing non-first-line prescribing for rhinosinusitis and AOM. There were decreases in 11 12 both unnecessary prescribing and non-first-line antibiotic prescribing for pharyngitis. All URTI conditions had downward trends for antibiotic prescribing among initial presentations, 13 similar to findings for these conditions using national PBS dispensing data (14,18,41,428). 14 Despite a decreasing trend for antibiotic prescribing for AOM, there was no significant 15 change in likely unnecessary prescribing, and there was also an increase in non-first-line 16 antibiotic use. 17

18

Increasing non-first-line prescribing for URTI has also been commonly reported in Australia 19 and the US (12,14,295,330). For rhinosinusitis, increasing second-line amoxicillin with 20 21 clavulanate use appear to mirror decreases in first-line amoxicillin over time. As was the case here, a shift in amoxicillin in favour of amoxicillin with clavulanate has been 22 documented nationally using PBS data (12,17), with little change in overall volume over time 23 suggesting direct replacement (12). Another study noted similarly increasing amoxicillin with 24 clavulanate dispensing in Australian dental community care (429). For pharyngitis, first-line 25 phenoxymethylpenicillin increases appeared to mirror decreases in not recommended 26 amoxicillin. The only second-line antibiotic for pharyngitis / tonsillitis, benzathine penicillin, 27 remains essentially unused in community care. For AOM, increasing first-line amoxicillin 28 seemed to correlate with decreasing second-line amoxicillin with clavulanate. 29

30

There were favourable downward trends in antibiotic prescribing for URTI conditions including influenza / ILI. The decreasing trends for two conditions (rhinosinusitis and pharyngitis) for unnecessary prescribing and non-first-line prescribing suggests some progress overall. However, major concerns remain regarding antibiotic prescribing for URTI conditions, including ongoing high levels of inappropriate prescribing, and increases in non first-line prescribing.

3

4

8.3 Urinary tract infection

5 6 7

8.3.1 Main descriptive findings for urinary tract infection

8 The high rates of non-first-line prescribing for all patient groups is concerning, particularly as these are initial consultations for UTI and not secondary or chronic presentations. Non-9 first-line antibiotics were received in sixty percent of prescribing situations for initial episodes 10 of care for UTI, 82% for children under sixteen, 68% men and 57% women, which is very 11 high particularly for children. High non-first-line prescribing rates for UTI have been 12 commonly reported elsewhere in both the Australian setting (14,17,48), and notably for 13 14 children (307). Notable non-first-line prescribing rates for UTI are a frequent finding in international studies among children and women (119,290,296,323). 15

16

Culture testing was recorded for 2% children, 3% of men and 3% of women and there were 17 no sensitivity testing results. It is concerning that culture and sensitivity testing were 18 performed so infrequently for children and men despite being recorded in the guidelines as 19 mandatory for children and strongly recommended for men (29). Repeats were issued on 20 28% of prescriptions for UTI. Of repeats on cefalexin prescriptions, 99% were likely 21 unnecessary for adults eighteen years and over but 99% were likely necessary for children 22 under sixteen years. There was also high variation in antibiotic duration for the cefalexin 23 prescriptions examined, which is consistent with findings for antibiotics commonly 24 prescribed for UTI internationally (120,285,291). 25

26

27 8.3.2 Predictors identified for urinary tract infection models

28

The predictors of increasing line of antibiotic choice included patient age group, gender, comorbid condition status, repeat prescription status and urine dipstick and culture testing. Predictors of repeat positive prescribing included patient age group, ordinal choice of antibiotic, urine dipstick testing, temperature recording status and multiple UTI episodes. Young children had the lowest probability of receiving first-line antibiotics, followed by older children, and then adult age groups. Children and men were most likely to receive secondline antibiotics. Antibiotic choice and repeat positive prescribing were also found to predict
each other. While attempting to clarify whether predominantly patient-specific factors might
be driving various definitions of inappropriate prescribing of UTI, several of the predictors
identified are in fact, consultation- or prescription-related.

4

Prescriptions issued with repeats were linked to third-line and second-line antibiotic 5 6 prescribing for UTI. In the model for choice of antibiotic, male gender increased the probability of non-first-line prescribing for adult patients, and the magnitude of effect 7 increased with increasing age. Having urine dipstick testing or culture testing performed 8 were associated with lower chances of non-first-line prescribing for all patients, while having 9 comorbid conditions was linked to increased chances. Adult males were more likely to 10 receive non-first-line prescriptions as well as prescriptions issued with repeats than women. 11 12 Urine dipstick testing being performed was also linked to lower probability of receiving prescriptions issued with repeats. 13

14

15 8.3.2.1 Patient age and gender and (un)complicated infection

Similar to the findings of this research, studies conducted in the US (323) and Italy (325) found that the youngest patients were most likely to receive inappropriate prescribing for UTI. The US study found that non-first-line prescribing for UTI was more common for children under two years, than children 13-17 years, and for female gender (323), as was found here. As UTI can be serious and potentially complicated in in children and men (123), GPs' concerns for their patients may drive the high proportions of non-first-line prescribing in these groups.

23

24 8.3.2.2 Repeats issued on prescriptions

With respect to repeats being issued on prescriptions being linked to first-line prescribing for children 5-15 years, it is possible that some school-aged children may be issued first-line prescriptions along with repeats on that prescription to limit the notable inconvenience of reattendance for parents of school-aged children presenting with UTI. It is plausible that repeats are driven by the clinician's attempts to reduce the need for re-visitation for parents, and that clinicians may be more comfortable prescribing repeats for first-line agents than non-first-line agents, particularly in children.

32

33 8.3.2.3 Urine dipstick and culture testing

34 It should be noted that this research did not examine pathology requests alone but results,

35 which required both a GP request and patient completion of a pathology test. Nevertheless,

the results are so infrequent that the issue is unlikely to be predominantly due to potential 1 patient non-compliance or difficulties acquiring urine samples from young children (307). 2 Similar to the findings here, Peng et al. (430) also found that there was a low use of urine 3 culture tests for groups at high risk of complicated UTI in the Australian setting, using a 25% 4 sample of national MedicineInsight patients compared to this project's use WA data during 5 6 a similar timeframe. They also found that young children were less likely to receive culture 7 testing than young adults (430). Another Australian study of GPs found that requests for 8 urine culture and microscopy occurred substantially less than recommended for both men and children (307), as found in this research. 9

10

The limited culture and susceptibility testing for UTI is potentially both time and cost-related, 11 12 as well as potential issues with patients adhering to testing requests. It is indeed reasonable for GPs to prescribe antibiotics at the first presentation without culture or susceptibility 13 14 results, due to highly uncomfortable and painful symptoms. However, referrals for such testing at initial presentations should be routine, and GPs should be insisting that patients 15 perform testing prior to commencing antibiotics. There may also be a lag in guideline 16 dissemination, as urine cultures were recommended in previous guidelines of 2010 (431) 17 rather than those current at the time of prescribing (29), in which both susceptibility and 18 culture recommended. Dipstick and culture testing may also be indicators of more diligent. 19 more guideline-aware and antibiotic resistance-aware prescribers. 20

21

22 8.3.2.4 Patient comorbid conditions

The finding that comorbid conditions influence prescribing for UTI is supported by other 23 research from Australian primary care (48). A small US qualitative study not limited to 24 25 primary care found that clinicians are conflicted between adhering to guidelines and tailoring individual patient care based on comorbidities as well as sociodemographic factors (432). 26

27

8.3.3 Trends in prescribing for urinary tract infection

28 29

There was an increase in second-line prescribing for all patient groups during the study 30 31 period, which can be attributed to the antibiotic cefalexin, as the only second-line antibiotic. There were upward trends in non-first-line antibiotic prescribing over time for adult patient 32 groups, however, a downward trend for children, which may be linked to the notable 33 decrease in use of not recommended antibiotics. There was also a downward trend in the 34 prescribing of not recommended antibiotics for women. 35

While there was a modest reduction in non-first-line prescribing for children over the study period, non-first-line prescribing rates for children remain very high and may be the patient group in most need of further, focused improvement. The decreasing norfloxacin use for adults during this period is favourable, although it should be negligible for initial UTI consultations.

7

8 There was notable amoxicillin without clavulanate use for initial UTI presentations when it 9 was not recommended in the guidelines at the time. This suggests that there may have been 10 a potential discord between clinicians and the guidelines, however, it should be noted that it 11 was added as an option in the subsequent guidelines for children and adult non-pregnant 12 women and men for situations of resistance to empirical therapy recommendations (86).

13

16

148.4Specific findings common to upper respiratory tract infection and urinary15tract infection

For initial presentations of either URTI or UTI, non-first line antibiotic prescribing appears far too prominent among these Australian GPs, with limited focus on overall community AMR concerns.

20

Patient age was a highly influential predictor for URTI and UTI, despite differing direction of
effect. Young children had the lowest probability of receiving inappropriate prescribing with
URTI but were at notably high probability of receiving non-first-line prescribing for UTI.
Repeats on antibiotic prescriptions were also influential for both URTI and UTI.

25

resistances 26 There have been important developments in critical antibiotic in carbapenemase-producing enterobacteriaceae, fluoroquinolone non-susceptibility in 27 28 Escherichia coli, and methicillin resistance in Staphylococcus aureus identified among hospitalised patients with bacteriaemia in Australia over the study period (14,18,417). These 29 30 have been found to disproportionally affect patients in Northern Australia and, to some extent, Western Australia, notably Aboriginal and Torres Strait Islander patients (433-436). 31 32 Clinician awareness regarding increasing antibiotic resistance, for example, may also 33 influence decisions to prescribing non-first-line antibiotics.

- 34
- 35

The fact both mental health conditions and other comorbid conditions have emerged as
predictors within differing levels of inappropriate prescribing demonstrates these do
influence clinician decision making.

5

6 The modest influence of number of URTI episodes in unnecessary prescribing, choice of 7 agent among unnecessary prescriptions, and in the model for repeat prescribing suggests 8 that there may be increased clinician concern regarding patients with additional, albeit separate, episodes. This may potentially point to multiple clinical interpretations of what 9 comprises a single episode of infection, and a potential blurring of definitions between 10 separate episodes and re-consultation for the same episode. Patients having multiple 11 12 episodes of UTI was also a predictor for patients receiving prescriptions with repeats present, which may also suggest this link. 13

14

Throughout all modelling processes, of variation not explained by fixed effects, the variation 15 16 explained by individual provider was far greater than the variation explained by individual practice. This suggests that individual provider drives variation far more than the individual 17 practice. The fact that other Australian data (14,17) found high variation in dispensing rates 18 across geographical areas to suggest that individual clinician preference had a strong impact 19 on antibiotic use also supports this. Large variability between prescribers as well as between 20 and within practices has been reported elsewhere (257,425,437,438), although these 21 22 studies focused more on differentiating between patient versus physician variability, rather than comparing prescriber and practice variation as in this thesis. 23

24

8.5 Implications to policy, practice and research

25 26

These results suggest that antibiotic prescribing occurs commonly despite it likely not being in the best interests of the patient, whether it be the decision to prescribe or the choice of antibiotic or the presence of a repeat. While first-line prescribing should, by definition, be the most frequent option for initial presentations of URTI where an antibiotic is prescribed, nonfirst-line agents represented the strong majority throughout the study. Prescribing rates also remained high throughout. Both factors suggest that there is a monumental task ahead in terms of achieving high compliance with the guidelines (29).

- 34
- 35

It is concerning that not recommended antibiotics contribute the highest proportion of any line of antibiotic prescribed for URTI. This suggests that they may be the 'go-to' for many clinicians. It appears that there has been a slight reduction in the prescribing of not recommended antibiotics over the study period, however, this remains far too high and should not be used for initial presentations.

7

1

For URTI, patients other than young children 0-8 years had notable probabilities of receiving inappropriate decisions. Unnecessary prescribing increases with increasing patient age, regardless of the URTI condition, so stewardship for URTI should focus limiting unnecessary prescribing for older patients. Pharyngitis was the condition with proportionally the most inappropriate decisions and the lowest appropriate non-prescribing rate. However, rhinosinusitis was the condition with the highest chance of receiving unnecessary prescribing. These two conditions should be prioritised for stewardship.

15

16 While several publications focus on children receiving the brunt of inappropriate antibiotic prescribing – defined by either prescribing rate for likely viral conditions, or non-first-line 17 prescribing rate (12,17), it is unclear whether consultation / diagnosis rates have been taken 18 into account, and it appears that they have not. When viewed in isolation, children do indeed 19 receive the largest number of unnecessary prescriptions for URTI. However, when the high 20 consultation rates for children with URTI are taken into account, young children are actually 21 receiving proportionally more appropriate treatment than many other age groups. 22 Standalone statements regarding high prescribing or dispensing rates or high non-first-line 23 prescribing rates for children for certain conditions may not be well received by clinicians 24 25 (14,18), as they are undoubtedly aware of the high consultation rates for children for common infections. Such statements, although technically correct, may be perceived as 26 27 naïve and dismissed for this reason. Messages aimed at improving prescribing practices for URTI may prove to be more effective when presented in context with consultation rates. 28

29

The fact that non-first-line prescribing occurred for 68% of acute rhinosinusitis prescriptions, and this notably higher than for other conditions, is concerning. This suggests substantial room for improvement with regard to the choice of agent for acute rhinosinusitis. Given that Country WA was the only PHN in which first-line prescriptions were most common for acute pharyngitis / tonsillitis (**Appendix C.2, Table C-9**), this may suggest that Country WA practitioners may be more familiar with treating pharyngitis, and how to do so appropriately. Aboriginal and Torres Strait Islander peoples are at high-risk for complications of pharyngitis
and prescribing is therefore appropriate. GPs working in Country WA may be acutely aware
of the guidelines for this condition by virtue of the high proportion of Aboriginal and Torres
Strait Islander peoples. Many of these patients may also be treated within Aboriginal Health
Services rather than mainstream general practice, the source of data for this research.

6

For UTI, the non-first-line prescribing dominates for children. Both children and men appear particularly at risk of poor prescribing choices, and the absence of important culture and sensitivity testing for these groups indicates a clear discord with the guidelines (29). There is a need for increased clinician awareness regarding the need for pathology testing for UTI diagnoses. It may also be worth giving consideration for a request to amend GP practice software to automate urine, culture and susceptibility requests for children and men with UTI (114).

14

It is promising that over 50% and over 80% of repeats for all patients with URTI and UTI, 15 respectively, would likely have been prevented by the PBAC decision to limit repeats 16 (127,128). PBAC should consider extending the repeat authority restrictions to include 17 erythromycin and phenoxymethylpenicillin, which were used frequently for URTI and were 18 commonly issued repeats when likely not required. Given the relationship between 19 inappropriate prescribing and repeat prescribing, it may also be worthwhile to ask PBAC to 20 consider shortening prescription validity (12,14,124). It is feasible that shortening 21 prescription validity for many antibiotic prescriptions may result in limiting use of repeat 22 prescriptions. Additionally, PBAC could consider requiring telephone PBS authority in order 23 to prescribe end-of-the-line 'last resort' antibiotics, to further restrict their use and protect 24 25 their efficacy.

26

Patient penicillin sensitivity is an important predictor of inappropriate prescribing for URTI presentations. There are limitations of current antibiotic allergy management, leading to undue use of non-first-line antibiotics (439). Ness et al. (422) note that many reactions likely do not reflect true hypersensitivity and that there is potential for safe rationalisation of patients with self-reported penicillin allergy (422,440). However, there are calls for an Australian allergy register (441), noting antimicrobials caused 48% of drug-related anaphylaxis admissions in a recent study (442).

- 34
- 35

The fact that these data show that for URTI and UTI, the individual provider has substantially more effect on inappropriate prescribing than variation in practice is suggestive an individual behavioural element, rather than practice-based culture. This offers hope in that individual prescribing behaviour at the GP level may be modifiable and that individual GPs should be the targeted in stewardship efforts for improving prescribing.

7

8 We are presented with opportunities to impact future prescribing and improve the quality of care for patients presenting to WA general practice. All of these findings should be 9 incorporated into GP stewardship programs and interventions, such as those outlined in the 10 Background chapter, to improve prescribing practices for GPs. These could be incorporated 11 12 together with promising strategies like delayed prescribing (68,69,126,216,443), social norm feedback (126,444) and shared decision-making (126,214,443). Such stewardship 13 programs should be coordinated at the state and primary health network level. It will be 14 important to consult with and be guided by GPs, and organisations such as WAPHA (42), in 15 16 the development of specific programs / interventions. GPs likely need more incentive for change (445), although careful consideration is required prior to implementing any new 17 policy (446). 18

19

Additionally, the need for culture and susceptibility testing for children and men with UTI, could be incorporated into patient management software in the form of automated reminders, as well as GP stewardship activities. Antibiotic stewardship initiatives for the general public are also required to educate patients regarding the appropriate use of antibiotics and the need to preserve their effectiveness.

25

The accessibility and usability of the guidelines for Australian GPs has also been questioned (26,250,251). Monaghan et al. (26) highlight the need for free access to guidelines, which is not currently the case, and this should be rectified if compliance with them is to be expected.

29

The shortfalls in GP data access and government surveillance go beyond antibiotic prescribing. These shortfalls result in limited prescribing surveillance, as we do not have detailed population-level data from general practice by which to plan stewardship activities. The government could take action to facilitate the collection of more detailed data from general practice than it presently requires for the purposes of Medicare reimbursement of services provided. Additional data collection would simply bring GP data surveillance up to that of a similar level as experienced by all medical practitioners working in the hospital
system. This would likely be met with much discontent from many GPs not wishing to have
their clinical management of patients scrutinised, however, the government could take a
hard-line approach.

5

6 We must somehow reach a situation where the majority of GPs place the future of the 7 community on par in priority with the individual patient needs. In this situation, each potential 8 prescribing situation would be akin to weighing up the balance between prescribing for the individual patient and future impact of resistance on the community. At present it seems that 9 a proportion of GPs do so, however, they are fighting against the tide of majority. Linder's 10 (447) analogy of needing to 'break the cycle' of unnecessary antibiotic prescribing is fitting. 11 12 There is also the possibility that the practices and respective GPs involved in MedicineInsight (168,345), which already promoted AMR stewardship and facilitated 13 prescribing feedback, and may have been more guideline-compliant than most GPs. This 14 raises the possibility that the prescribing by GPs in WA outside this dataset might have the 15 16 potential to be even less guideline-compliant than those studied here, which is a concerning prospect. 17

18

Based on the magnitude of the inappropriate prescribing occurring at different levels, there 19 is reason for concern. Unfortunately, no matter how one looks at it, in times of increasing 20 resistance, the impact of poor prescribing practices is a serious matter. While it may be 21 tempting to cite the difficult contexts GPs face, such as diagnostic uncertainty and the lack 22 of point-of-care testing, no amount of context or explanation can alleviate the severity of the 23 impact of poor prescribing occurring at a health system level. However, the situation is 24 25 simply not good enough and more action is urgently needed. Antibiotic resistance is developing at a far greater rate than any small improvements in prescribing practices found 26 27 here.

28

Antibiotic prescribing is a complex, multifaceted issue and many unknowns remain. Some of these factors are thought to include patient demand, perceived patient demand, time pressure, complacency, and ignorance regarding antibiotic resistance on part of the prescriber (23,180,233-239,448).

- 33
- 34

8.5.1 Future research directions

2

3 More research on drivers of inappropriate prescribing is required and more insights from GPs themselves are essential. The findings from this quantitative research should be 4 explored with detailed qualitative inquiry, to provide complimentary, additional context to 5 these results and gain insight into GPs' perceptions regarding inappropriate antibiotic 6 prescribing. For example, semi-structured interviews could provide insight into GPs' 7 perceptions regarding the circumstances in which inappropriate antibiotic prescribing occurs 8 and perceived driving factors. Future research is also warranted to investigate factors 9 influencing the use of antimicrobials in primary care which may not be addressed well in the 10 guidelines, which is another opportunity to involve GPs. 11

12

Further research is needed to identify factors driving inappropriate prescribing in certain 13 settings and for other specific conditions (294), such as community-acquired pneumonia 14 and cellulitis in Australian general practice. Given the notable inappropriate prescribing 15 identified to date, and the concerning variation found for duration of cefalexin prescribing for 16 UTI, guideline compliance regarding dose and duration of antibiotic prescribing also needs 17 further exploration. Expanding research to include the use of topical antibiotics would also 18 be beneficial. Other options include investigating prescriptions issued without a consultation 19 20 (308,318,449) and long-term antibiotic prescribing such as for acne (450). Second to GPs, other medical specialists are responsible for the most antibiotic prescriptions in the 21 community setting (12), which would also be worthwhile to research. Time pressure and 22 workload in relation to inappropriate prescribing are worthy of further investigation. 23

24

25 There is a need for further research specific to Aboriginal and Torres Strait Islander peoples and pregnant women in relation to inappropriate antibiotic prescribing. Aboriginal and Torres 26 Strait Islander people continue to be under-identified in many Australian health-related data 27 collections (355). They have higher levels of morbidity and mortality than non-Indigenous 28 Australians (451,452) and are at higher risk of serious complications following infection, and 29 30 therefore have a lower threshold for antimicrobial prescribing (453). Detailed study of the appropriateness of antibiotic prescribing provided to Aboriginal and Torres Strait Islander 31 populations is important. Pregnant women are also an important part of the population which 32 should be studied (325). 33

- 34
- 35

- 1 8.6 Limitations
- 2

8.6.1 Prescriptions and patient groups

Limitations of this study include the fact that delayed prescriptions could not be differentiated from those directed for immediate use, as dispensing data was unavailable. This may result in an overestimation of likely unnecessary prescribing. The uptake of delayed prescribing is uncertain due to its limited albeit promising study in Australian clinical practice (218). It is difficult to compare prescribing practices with dispensing practices, as many prescriptions are known not to be filled / dispensed by patients (60-64).

11

Pregnant women could not be easily identified so all women were assumed to be nonpregnant, which may have influenced the results to a small extent. Also, the variable indicating patients identifying as Aboriginal and Torres Strait Islander peoples was considered unreliable such that it was not used in analysis, in addition to the fact that dispensing via Aboriginal Health Services is not captured in these data (12).

- 17
- 18 19

8.6.2 Definition of inappropriate prescribing according to the guidelines

This thesis was limited to identification of 'likely' inappropriate prescribing rather than 20 21 'definitive' inappropriate prescribing according to the national therapeutic guidelines. It is inherently complex to accurately identify inappropriate prescribing from data extracted from 22 23 general practice software, and free-text progress notes made by the GP are typically not available for research, due to difficulties deidentifying these data. For some conditions, the 24 progress notes may include the indication for prescribing according to the guidelines (26), 25 against which recorded criteria are checked to ascertain appropriateness. The possibility of 26 obtaining progress notes was explored but could not be obtained due to confidentiality 27 reasons. Inappropriate non-prescribing (not prescribing despite it being necessary for the 28 patient) was also excluded as it was considered that there was insufficient information 29 available to accurately assess this. 30

31

Misclassification would have been possible in circumstances where the recording of this important information, such as the clinical observations of fever, simply did not occur. This is a reason for using the term 'likely unnecessary' rather than a more definitive judgement. Researchers can only work on the data recorded and accessible, and this concession was necessary to analyse large-scale patient data. A conservative approach
was taken to limit potential overestimation of indications to prescribe using the data at hand
(454).

4

5 8.6.3 Diagnoses

6

Defining an episode of infection for analysis required limiting the definition to encounters with the same diagnosis within a predefined period as a single episode. For URTI, the presence of influenza diagnoses resulting in antibiotic prescriptions may be reflective of poor coding by the GP of bacterial infection secondary to influenza. This highlights the need to be mindful that these data were not created for research purposes but for administrative purposes and this may have implications upon the coding used by GPs.

13

Despite the removal of UTI diagnoses coded as recurrent / chronic, some presentations for chronic / recurrent UTI may have been poorly coded as (initial presentations of) UTI. Note the guidelines used for assessment in these analyses do not define recurrent UTI in terms of a number of episodes within a specified timeframe (29). However, the guidelines for recurrent UTI published in 2019 (86), subsequent to the study period, define recurrent UTI as two or more episodes within six months, or three or more per year, and positive urine cultures (412,430,455,456).

- 21
- 22 23

8.6.4 Other potential covariates

Another limitation of the study was the inability to identify any patients visiting more than one practice within the dataset. There is negligible information in Australia regarding the prevalence and potential impact of this, particularly in the antibiotic prescribing context. Patient movement between practices is thought to be less likely for the young and elderly. It was also not possible to identify pathology tests requested but not subsequently performed by the patient. Practice size was unable to be obtained from MedicineInsight due to confidentiality reasons (168).

31

32 8.6.5 Statistical limitations

33

Although random slopes and random intercepts were the intention at each level, this proved
 too resource-intensive computer-wise. The absence of random slopes can result in standard

errors which may be anti-conservative (457). However, the use of mixed models (allowing
for unobserved heterogeneity) is much preferred over the alternative of fixed effects, which
would have put undue scrutiny on PHNs for likely unnecessary prescribing, for example.

4

5 The trends analyses involved monthly aggregate data for the five-and-a-half-year period, 6 equating to 66 monthly data points, and as such there are limits to what one may infer from 7 these data particularly with regard to any prediction beyond the study period. Nevertheless, 8 this analysis provides useful insights into the overall direction and magnitude of prescribing 9 outcomes and highlight areas of progress as well as areas of concern.

- 10
- 11 12

8.6.6 Complexities of the project

Access to general practice data proved politically sensitive for this research. It took some 13 time to negotiate access to suitable data, involving the refusal for involvement by three other 14 potential data sources. It is only through both WAPHA (42) and MedicineInsight's (168) 15 involvement that this thesis was possible, highlighting the need for collaboration in Australian 16 primary care research. The presence of multiple stakeholders also created minor 17 complications with each having slightly different perspectives. Fears of upsetting member 18 GPs, upon whom both external stakeholders rely, was an issue which arose, however, both 19 stakeholders demonstrated ongoing commitment to this project. There were delays due to 20 technical difficulties with managing large datasets, including obtaining sufficient processing 21 power and data storage. Data cleaning proved more time-consuming than anticipated, 22 23 particularly with regard to free-text string variables including diagnoses and pathology results. 24

25

Difficulties regarding data access relates back to Australian general practice being 26 27 predominantly privately owned practices. For groups of practices in large organisations, who have invested in database management, remuneration is required to obtain access to GP 28 data, and this is not unusual for research data, however, must be considered (182). WAPHA 29 (42) and MedicineInsight (168) had important roles in facilitating and supporting this project, 30 despite the possibility that the findings of this research may raise anxieties of GPs 31 participating in their respective programs. This demonstrates their commitment to research 32 towards limiting antibiotic resistance and improving patient care. 33

8.7 Significance

2

The project has significant potential impact both nationally and internationally. It can inform policy and practice and lead to further research. Its applications include informing the development of evidence-based interventions to reduce inappropriate antibiotic prescribing to stem the emergence of resistance.

7

8 This research goes a notable way to filling the large gap in knowledge regarding 9 inappropriate antibiotic prescribing in Australia. The importance of this project also relates 10 to the fact that limited research has been undertaken using large-scale data from general 11 practice in Australia to date.

12

To the best of our knowledge at the time of writing, this is the first Australian research using 13 quantitative methods and empirical data to identify predictors of inappropriate prescribing in 14 general practice for UTI and URTI. This is also believed to particularly be the case for the 15 state of WA. This is believed to be the first research in Australian general practice, to the 16 best of our knowledge, limits consultations to initial presentations for the condition. This is 17 also believed to be the first study in Australian general practice, to the best of our knowledge, 18 which uses more information than purely the condition diagnosed to differentiate 19 20 inappropriate from appropriate prescribing.

21

In light of the increasing international threat of antibiotic resistance, and the need for action
on this front, it is vital to inform this action by measuring adherence to the guidelines and
identifying any factors affecting adherence (29).

3

2 CHAPTER 9 CONCLUSION

This research identified the substantial inappropriate prescribing of systemic antibiotics 4 occurred within WA general practice between January 2012 and June 2017, inclusive. Over 5 80% of antibiotic prescriptions for URTI presentations were found to be likely unnecessary. 6 Non-first-line prescribing was found in more than half of all antibiotic prescriptions for URTI 7 and UTI. Repeats were issued on over 25% antibiotic prescriptions occurring for URTI and 8 9 UTI, the strong majority of which were unnecessary. There was concerning deviation from the guidelines for UTI for children and men (29), with infrequent culture and susceptibility 10 11 testing for performed.

12

For URTI, the predictors of inappropriate prescribing identified were patient age, URTI condition, practice size, mental health condition status, patient concession status, patient penicillin sensitivity, number of URTI episodes, repeat prescriptions status, weekend consultation status and patients from interstate PHNs. For UTI, the predictors of inappropriate prescribing included patient age group, gender, comorbid condition status, repeat prescription status and urine dipstick and culture testing status, temperature recording status and multiple UTI episodes.

20

All URTI conditions had downward trends for antibiotic prescribing among initial presentations over the study period. There were also reductions in the rate of likely unnecessary prescribing for rhinosinusitis and pharyngitis but not AOM. There were upward trends in non-first-line prescribing for rhinosinusitis and AOM but a downward trend for pharyngitis. For UTI, there were increases in second-line antibiotic prescribing for all patient groups over time. There are upward trends in non-first-line antibiotic prescribing over time for adult patient groups, however, a downward trend for children.

28

The magnitude of inappropriate prescribing occurring at several different levels or definitions of inappropriate prescribing demonstrates that there is reason for concern. Despite some small improvements in prescribing practices found over time, more action is urgently needed.

The findings of this thesis should be incorporated into GP stewardship programs / 1 2 interventions to improve prescribing practices. Antibiotic stewardship for the general public is also in need. PBAC should give consideration to various amendments to GP prescribing 3 authorities for relevant antibiotic prescriptions. Changes regarding culture and susceptibility 4 testing could also be incorporated into patient management software. The government could 5 6 additionally consider requiring more detailed data from general practices to facilitate improved surveillance of antibiotic prescribing and enhanced planning of stewardship 7 8 activities.

9

This thesis has covered the priority condition groups of URTI and UTI. Further areas for 10 research should include inappropriate antibiotic prescribing for community-acquired 11 12 pneumonia and cellulitis in Australian general practice, as these are prevalent conditions with large populations affected. Research should be conducted specific to Aboriginal and 13 14 Torres Strait Islander peoples and pregnant women, as these groups were not clearly identifiable from the data available in this research. Both the dose and duration of antibiotic 15 prescribing also warrant further investigation for URTI and UTI conditions. Future research 16 should also explore the use of topical antibiotics. Time pressure and workload are important 17 to investigate further for any association with inappropriate antibiotic prescribing in the GP 18 setting. It may be worthwhile researching inappropriate antibiotic prescribing by medical 19 specialists other than GPs in community setting. 20

21

REFERENCES

- StataCorp. Glossary. Multilevel Mixed-Effects Reference Manual, Release 18. College Station, TX:
 Stata Press; 2023. ISBN-13: 978-1-59718-389-5. Available from: https://www.stata.com/manuals/meglossary.pdf
- StataCorp. margins Marginal means, predictive margins, and marginal effects. MULTILEVEL
 MIXED-EFFECTS REFERENCE MANUAL RELEASE 17. College Station, TX: StataCorp LLC.;
 2021 [11/06/2023]. Available from: <u>https://www.stata.com/manuals/rmargins.pdf</u>
- 9 3. Williams R. Using the margins command to estimate and interpret adjusted predictions and 10 marginal effects. Stata Journal. 2012;12(2):308-331.
- Bornmann L, Williams R. How to calculate the practical significance of citation impact differences?
 An empirical example from evaluative institutional bibliometrics using adjusted predictions and marginal effects. Journal of Informetrics. 2013 2013/04/01/;7(2):562-574.
 doi:<u>https://doi.org/10.1016/j.joi.2013.02.005</u>.
- World Health Organization. The evolving threat of antimicrobial resistance: options for action.
 [Internet]. Geneva: World Health Organization; 2012 [cited 2015 Jun 27]. Available from: http://www.who.int/patientsafety/implementation/amr/publication/en/
- 18 6. World Health Organization. Global Action Plan on Antimicrobial Resistance [Internet]. Geneva: 19 World Health Organization; 2015 [cited 2017 Nov 7]. Available from: 20 http://www.who.int/drugresistance/global action plan/en/
- 21 7. Organisation for Economic Cooperation and Development. Stemming the Superbug Tide [Internet]. Paris: OECD: 2018 [cited 2021 Nov Available 22 30]. from: https://www.oecd-23 ilibrary.org/content/publication/9789264307599-en doi:doi:https://doi.org/10.1787/9789264307599-en. 24
- 25 The Wellcome Trust. Review on Antimicrobial Resistance. Tackling drug-resistant infections 8. 26 globally: final report and recommendations—review on antimicrobial resistance. [Internet]. London: 27 Wellcome Trust: 2016 [cited 2019 Sep 7]. Available from: The https://amr-28 review.org/sites/default/files/160518_Final%20paper_with%20cover.pdf
- Holmes AH, Moore LSP, Sundsfjord A, Steinbakk M, Regmi S, Karkey A, Guerin PJ, Piddock LJV.
 Understanding the mechanisms and drivers of antimicrobial resistance. Lancet.
 2015;10.1016/S0140-6736(15)00473-0. Epub 2015 Nov 18. doi:<u>http://dx.doi.org/10.1016/S0140-6736(15)00473-0</u>.
- Laxminarayan R, Duse A, Wattal C, Zaidi AKM, Wertheim HFL, Sumpradit N, Vlieghe E, Hara GL,
 Gould IM, Goossens H, Greko C, So AD, Bigdeli M, Tomson G, Woodhouse W, Ombaka E, Peralta
 AQ, Qamar FN, Mir F, Kariuki S, Bhutta ZA, Coates A, Bergstrom R, Wright GD, Brown ED, Cars
 O. Antibiotic resistance—the need for global solutions. Lancet Infect Dis. 2013;13(12):1057-1098.
 doi:http://dx.doi.org/10.1016/S1473-3099(13)70318-9.
- Davies J, Davies D. Origins and Evolution of Antibiotic Resistance. Microbiol Molecul Biol Rev.
 2010;74(3):417.
- Australian Commission on Safety and Quality in Health Care. AURA 2016: First Australian report on antimicrobial use and resistance in human health [Internet]. Sydney (NSW): Australian Commission on Safety and Quality in Health Care; 2016 [cited 2017 Jun 30]. Available from: <u>https://www.safetyandquality.gov.au/publications/aura-2016-first-australian-report-on-</u> antimicroibal-use-and-resistance-in-human-health/
- van Driel M, Merlo G, Baillie E, Dartnell J, Hall L, Heal C. Preserving antibiotics for the future:
 Where Australian general practice sits on the global spectrum. Aust J Gen Pract. 2022 Jan 20;51:10-13.
- 48 14. Australian Commission on Safety and Quality in Health Care. AURA 2019: Third Australian report 49 on antimicrobial use and resistance in human health [Internet]. Sydney (NSW): Australian 50 Commission on Safety and Quality in Health Care; 2019 [cited 2021 Jun 17]. Available from: 51 <u>https://www.safetyandquality.gov.au/publications-and-resources/resource-library/aura-2019-third-52 australian-report-antimicrobial-use-and-resistance-human-health
 </u>
- Australian Commission on Safety and Quality in Health Care and National Health Performance
 Authority. Australian Atlas of Healthcare Variation [Internet]. Sydney (NSW): Australian
 Commission on Safety and Quality in Health Care 2015 [cited 2017 Oct 3]. Available from:
 http://www.safetyandquality.gov.au/atlas/

- 116.Department of Health & Department of Agriculture. National Antimicrobial Resistance Strategy22015–2019 [Internet]. Department of Health, Australian Government; 2015 [cited 2015 Nov 18].3Availablefrom: http://www.health.gov.au/internet/main/publishing.nsf/Content/ohp-4amr.htm#tocstrategy
- Australian Commission on Safety and Quality in Health Care. AURA 2017: Second Australian report on antimicrobial use and resistance in human health [Internet]. Sydney (NSW): Australian Commission on Safety and Quality in Health Care; 2017 [cited 2019 Oct 30]. Available from:
 <u>https://www.safetyandquality.gov.au/publications/second-australian-report-on-antimicrobial-use-and-resistance-in-human-health/</u>
- Australian Commission on Safety and Quality in Health Care. AURA 2021: Fourth Australian report on antimicrobial use and resistance in human health [Internet]. Sydney (NSW): Australian Commission on Safety and Quality in Health Care; 2021 [cited 2017 Nov 20]. Available from: <u>https://www.safetyandquality.gov.au/sites/default/files/2021-09/aura_2021_-_report_-</u>
 <u>final_accessible_pdf_-_for_web_publication.pdf</u>
- Shaban RZ, Cruickshank M, Christiansen K, Antimicrobial, Resistance, Standing, Committee.
 National Surveillance and Reporting of Antimicrobial Resistance and Antibiotic Usage for Human Health in Australia [Internet]. Antimicrobial Resistance Standing Committee AHPPC, editor.
 Canberra (ACT): Australian Commission on Safety and Quality in Health Care; 2013 [cited 2015 Jun 27]. Available from: <u>http://www.safetyandquality.gov.au/publications/national-surveillance-andreporting-of-antimicrobial-resistance-and-antibiotic-usage-for-human-health-in-australia/</u>
- McKenzie D, Rawlins M, Del Mar C. Antimicrobial stewardship: what's it all about? Aust Prescriber.
 2013;36(4):116-120.
- 23 21. Pan Y, Henderson J, Britt H. Antibiotic prescribing in Australian general practice: how has it
 24 changed from 1990-91 to 2002-03? Respiratory medicine. 2006 Nov;100(11):2004-11.
 25 doi:10.1016/j.rmed.2006.02.015. Cited in: Pubmed; PMID 16616483.
- Murphy M, Bradley CP, Byrne S. Antibiotic prescribing in primary care, adherence to guidelines
 and unnecessary prescribing an Irish perspective. BMC Fam Pract. 2012;13(1):1-8.
 doi:10.1186/1471-2296-13-43.
- 29 23. Bagger K, Nielsen ABS, Siersma V, Bjerrum L. Inappropriate antibiotic prescribing and demand for antibiotics in patients with upper respiratory tract infections is hardly different in female versus male 30 patients 31 as seen in primary care. Eur J Gen Pract. 2015;21(2):118-123. 32 doi:10.3109/13814788.2014.1001361.
- Sigurdardottir N, Nielsen ABS, Munck A, Bjerrum L. Appropriateness of antibiotic prescribing for
 upper respiratory tract infections in general practice: Comparison between Denmark and Iceland.
 Scand J Prim Health Care. 2015;33(4):269-274. doi:10.3109/02813432.2015.1114349.
- Akkerman A, Kuyvenhoven M, van der Wouden J, Verheij TM. Determinants of antibiotic
 overprescribing in respiratory tract infections in general practice. J Antimicrob Chemother.
 2005;56(5):930-7453.
- Monaghan T, Biezen R, Buising K, Hallinan C, Cheah R, Manski-Nankervis JA. Clinical insights
 into appropriate choice of antimicrobials for acute respiratory tract infections. Aust J Gen Pract.
 2022 Jan 1;51(1-2):33-37. English. doi:<u>https://dx.doi.org/10.31128/AJGP-07-21-6073</u>. Cited in:
 Pubmed; PMID 637140121.
- 43 27. Biezen R, Ingram R, James R, Manski-Nankervis JA, Thursky K, Buising K. A need for action: 44 Results from the Australian general practice national antimicrobial prescribing survey [Conference 45 Abstract]. Aust J Prim Health. 2019;25(3):viii-ix. English. 46 doi:http://dx.doi.org/10.1071/PYv25n3abs. Cited in: Pubmed; PMID 630751680.
- 47 28. Gunnarsson R, Orda U, Elliott B, Heal C, Gorges H, Glasziou P, Del Mar C. PCR point of care
 48 testing for group a streptococci in patients with uncomplicated acute sore throat improves targeting
 49 of antibiotics [Conference Abstract]. Aust J Prim Health. 2021;27(4):xvii-xviii. English.
 50 doi:http://dx.doi.org/10.1071/PYv27n4abs. Cited in: Pubmed; PMID 635820582.
- 29. Antibiotic Expert Groups. Therapeutic guidelines: antibiotic. Version 15. Melbourne (VIC):
 Therapeutic Guidelines Limited; 2014. ISBN: 9780992527211.
- 30. Dallas A, Magin P, Morgan S, Tapley A, Henderson K, Ball J, Scott J, Spike N, McArthur L, van
 Driel M. Antibiotic prescribing for respiratory infections: a cross-sectional analysis of the ReCEnT
 study exploring the habits of early-career doctors in primary care. Family Practice. 2015 Feb
 1;32(1):49-55. doi:10.1093/fampra/cmu069.

- Gottlieb T, Nimmo GR. Antibiotic resistance is an emerging threat to public health: an urgent call
 to action at the Antimicrobial Resistance Summit 2011. Med J Aust. 2011;194(6):281-3.
- 3 32. McManus P, Hammond M, Whicker S, Primrose J, Mant A, Fairall S. Antibiotic use in the Australian
 4 community, 1990–1995. Med J Aust. 1997;167(3):124-7.
- 33. Gordon J, Miller G, Britt H, Family Medicine Research Centre, University of Sydney. Reality check
 reliable national data from general practice electronic health records emphasises the need to
 urgently address the lack of reliable national primary health data, particularly data from general
 practice. Deeble Institute Issues Brief [Internet]. 2016 Jul 24;(No. 18). Epub 2016 Jul 17. Available
 from: https://ahha.asn.au/system/files/docs/publications/deeble_institue_issues_brief_no_18.pdf
- 1034.Department of Health. Australian Standard Geographical Classification-Remoteness Area (ASGC-11RA) 2016. [Internet]. Canberra (ACT): Department of Health; 2016. [cited 2020 Nov 5]. Available12from: https://www.health.gov.au/topics/rural-health-workforce/classifications/asgc-ra
- Australian Bureau of Statistics. Australian Statistical Geography Standard (ASGS): Volume 5 -13 35. Remoteness Structure, July 2011 (cat. no. 1270.0.55.005). [Internet]. Adelaide: Australian Bureau 14 15 of Statistics: 2013 [cited 2022 Jan 20]. Available from: 16 https://www.abs.gov.au/ausstats/abs@.nsf/mf/1270.0.55.005
- Australian Bureau of Statistics. Socioeconomic Index for Area (SEIFA) [Internet]. Canberra (ACT):
 Australian Bureau of Statistics; 2016 4 Jun. Available from:
 <u>http://www.abs.gov.au/websitedbs/censushome.nsf/home/seifa</u>
- 37. Australian Bureau of Statistics. 2033.0.55.001 Census of Population and Housing: Socio-Economic Indexes for Areas (SEIFA), Australia, 2016 [Internet]. Australian Bureau of Statistics AG, editor.: Australian Bureau of Statistics; 2016 [cited 2018 12 Dec]. Available from: <u>https://www.abs.gov.au/ausstats/abs@.nsf/Lookup/by%20Subject/2033.0.55.001~2016~Main%2</u>
 OFeatures~SOCIO-ECONOMIC%20INDEXES%20FOR%20AREAS%20(SEIFA)%202016~1
- Mehta N, Schilder A, Fragaszy E, H ERE, Dukes O, Manikam L, Little P, Smith SC, Hayward A.
 Antibiotic prescribing in patients with self-reported sore throat. J Antimicrob Chemother. 2017 Mar
 1;72(3):914-922. English. doi:10.1093/jac/dkw497. Cited in: Pubmed; PMID 27999063.
- Malo S, Poblador-Plou B, Prados-Torres A, Lallana MJ, Laguna-Berna C, Rabanaque MJ. Poor
 congruence with guidelines in the use of antibiotics for acute bronchitis: a descriptive study based
 on electronic health records. Fam Prac. 2016 May 24. doi:10.1093/fampra/cmw037.
- Australian Commission on Safety and Quality in Health Care. Antimicrobial prescribing practice in 31 40. 32 Australian hospitals: results of the 2014 National Antimicrobial Prescribing Survey. Sydney (NSW): 33 ACSQHC: 2015 06/05/2020. Available from: https://www.safetvandguality.gov.au/ourwork/antimicrobial-resistance/antimicrobial-use-and-resistance-australia-surveillance-system-34 aura/hospital-antimicrobial-use/appropriateness-antimicrobial-use-australian-hospitals-naps 35
- Bernardo CDO, Gonzalez-Chica D, Stocks N. Influenza-like illness and antimicrobial prescribing in
 Australian general practice from 2015 to 2017: a national longitudinal study using the
 MedicineInsight dataset. BMJ Open. 2019;9(4):e026396. doi:10.1136/bmjopen-2018-026396.
- WA Primary Health Alliance. PHNs [Internet]. Rivervale, WA: WA Primary Health Alliance; 2015.
 [updated 2016; cited 2016 Feb 2]. Available from: <u>http://www.wapha.org.au/primary-health-</u>
 <u>networks/</u>
- 4243.Chan M. World Health Day 2011 Combat drug resistance: no action today means no cure
tomorrow. 2011 [cited 2017 Dec 6]:Statement by WHO Director-General, Dr Margaret Chan. Epub
2011 Apr 6. Available from:
http://www.who.int/mediacentre/news/statements/2011/whd_20110407/en/
- 46 44. National Action Plan For Combating Antibiotic-Resistant Bacteria, (2015).
- 47 45. Costelloe C, Metcalfe C, Lovering A, Mant D, Hay AD. Effect of antibiotic prescribing in primary
 48 care on antimicrobial resistance in individual patients: systematic review and meta-analysis. BMJ.
 49 2010;340:c2096. doi:10.1136/bmj.c2096. Cited in: Pubmed; PMID 20483949.
- 46. Butler CC, Simpson SA, Dunstan F, Rollnick S, Cohen D, Gillespie D, Evans MR, Alam MF,
 Bekkers M-J, Evans J, Moore L, Howe R, Hayes J, Hare M, Hood K. Effectiveness of multifaceted
 educational programme to reduce antibiotic dispensing in primary care: practice based randomised
 controlled trial. BMJ. 2012;344:d8173. doi:10.1136/bmj.d8173.
- 47. Cars O, Chandy SJ, Mpundu M, Peralta AQ, Zorzet A, So AD. Resetting the agenda for antibiotic
 resistance through a health systems perspective. Lancet Glob Health. 2021;9(7):e1022-e1027.
 doi:10.1016/S2214-109X(21)00163-7.

- 48. O'Keefe C, Thompson A, McKenzie D, Lee K. Concordance with antibiotic guidelines in Australian primary care: A retrospective study of prior-to-hospital therapy. Int J Infect Control. 2019 Dec 1;73(12) (no pagination). English. doi:<u>http://dx.doi.org/10.1111/ijcp.13427</u>. Cited in: Pubmed; PMID 2003441612.
- 49. Del Mar C, Scott A, Glasziou P, Hoffmann T, van Driel M. Reducing antibiotic prescribing in
 Australian general practice: time for a national strategy. Med J Aust. 2017;207(9). Epub 2017 Oct
 23.
- Tran B, Straka P, Falster MO, Douglas KA, Britz T, Jorm LR. Overcoming the data drought:
 exploring general practice in Australia by network analysis of big data. Med J Aust. 2018;209(2):6873. doi:10.5694/mja17.01236.
- Sturgiss E, van Boven K. Datasets collected in general practice: an international comparison using
 the example of obesity. Aust Health Rev. 2018;42(5):563-567.
 doi:<u>https://doi.org/10.1071/AH17157</u>.
- 52. WA Country Health Service. The WA Country Health Service Annual Report 2018-19. Perth (WA):
 WA Country Health Service GoWA; 2019.
- Australian Institute of Health and Welfare. Rural and remote health [Internet]. Australian Institute of
 Health and Welfare; 2022. [updated 2022 Jul; cited 2022 Aug 23]. Available from:
 https://www.aihw.gov.au/reports/rural-remote-australians/rural-and-remote-health
- Bowen AC, Daveson K, Anderson L, Tong SY. An urgent need for antimicrobial stewardship in Indigenous rural and remote primary health care. Med J Aust. 2019;211(1):9-11.e1. doi:<u>https://doi.org/10.5694/mja2.50216</u>.
- 55. Fremantle E, Zurynski YA, Mahajan D, D'Antoine H, Elliott EJ. Indigenous child health: urgent need
 for improved data to underpin better health outcomes [Paper in: Indigenous Health]. Med J Aust.
 2008;188(10):588-591. doi:10.5694/j.1326-5377.2008.tb01797.x.
- S6. Cuningham W, McVernon J, Lydeamore MJ, Andrews RM, Carapetis J, Kearns T, Clucas D,
 Dhurrkay RG, Tong SYC, Campbell PT. High burden of infectious disease and antibiotic use in
 early life in Australian Aboriginal communities. Aust N Z J Public Health. 2019;43(2):149-155.
 doi:<u>https://doi.org/10.1111/1753-6405.12876</u>.
- 57. Parnaby MG, Carapetis JR. Rheumatic fever in indigenous Australian children. Journal of
 paediatrics and child health. 2010;46(9):527-533.
- 31 Healy J, Sharman, E., Lokuge, B. . Australia: health system review European Observatory on 58. 32 Health Care Systems. [Internet]. Copenhagen: The European Observatory on Health Systems and 33 2006 [cited 2022 Nov Available Policies: 5]. from: data/assets/pdf_file/0007/96433/E89731.pdf 34 https://www.euro.who.int/
- Sumbo T. General Principles of Antimicrobial Therapy. In: Brunton LL, Hilal-Dandan R, Knollmann
 BC, editors. Goodman & Gilman's: The Pharmacological Basis of Therapeutics, 13e. New York,
 NY: McGraw-Hill Education; 2017.
- Kemp A, Preen DB, Glover J, Semmens J, Roughead EE. How much do we spend on prescription
 medicines? Out-of-pocket costs for patients in Australia and other OECD countries. Aust Health
 Rev. 2011;35(3):341-349. doi:10.1071/AH10906.
- 41 61. Hynd A, Roughead EE, Preen DB, Glover J, Bulsara M, Semmens J. The impact of co-payment
 42 increases on dispensings of government-subsidised medicines in Australia. Pharmacoepidemiol
 43 Drug Saf. 2008;17(11):1091-1099. doi:10.1002/pds.1670.
- Balram P. Patterns of medication use in Queensland Australia and the impact of a large copayment
 increase: An exploratory study. St Lucia, QLD: The University of Queensland, School of Nursing,
 Midwifery and Social Work; 2015. Doctor of Philosophy.
- 47 63. Kennedy J, Tuleu I, Mackay K. Unfilled prescriptions of medicare beneficiaries: prevalence,
 48 reasons, and types of medicines prescribed. J Manag Care Pharm. 2008;14(6):553.
- 64. Gardner TL, Dovey SM, Tilyard MW, Gurr E. Differences between prescribed and dispensed
 medications. N Z Med J. 1996 Mar 8;109(1017):69-72. English. Cited in: Pubmed; PMID 8606821.
- 65. Claridge JA, Pang P, Leukhardt WH, Golob JF, Carter JW, Fadlalla AM. Critical analysis of empiric
 antibiotic utilization: establishing benchmarks. Surg Infect (Larchmt). 2010 Apr;11(2):125-31.
 English. doi:10.1089/sur.2009.047. Cited in: Pubmed; PMID 20210653.
- Ferguson J. Antibiotic prescribing: How can emergence of antibiotic resistance be delayed? Aust
 Prescriber. 2004;27(2):39-51. doi:10.18773/austprescr.2004.037.

- Leekha S, Terrell CL, Edson RS. General principles of antimicrobial therapy. Mayo Clin Proc. 2011
 Feb;86(2):156-67. English. doi:10.4065/mcp.2010.0639. Cited in: Pubmed; PMID 21282489.
- 3 68. Spurling GK, Del Mar CB, Dooley L, Foxlee R, Farley R. Delayed antibiotics for respiratory infections. Cochrane Database Syst Rev. 2013;(4):1-65. Epub 2013 Apr 4 30. 5 doi:10.1002/14651858.CD004417.pub4.
- 6 69. Arroll B, Kenealy T, Kerse N. Do delayed prescriptions reduce antibiotic use in respiratory tract infections? A systematic review. Br J Gen Pract. 2003;53(496):871-877.
- Martinez-Gonzalez NA, Keizer E, Plate A, Coenen S, Valeri F, Verbakel JYJ, Rosemann T, Neuner-8 70. 9 Jehle S, Senn O. Point-of-care c-reactive protein testing to reduce antibiotic prescribing for respiratory tract infections in primary care: Systematic review and meta-analysis of randomised 10 11 controlled trials [Review]. Antibiotics. 2020 Sep;9(9):1-31. English. 12 doi:http://dx.doi.org/10.3390/antibiotics9090610. Cited in: Pubmed; PMID 2005075174.
- Cooke J, Butler C, Hopstaken R, Dryden MS, McNulty C, Hurding S, Moore M, Livermore DM. 13 71. Narrative review of primary care pointof- care testing (POCT) and antibacterial use in respiratory 14 BMJ 15 tract infection (RTI). Open Respir Res. 2015;2(1):1-10. English. 16 doi: http://dx.doi.org/10.1136/bmjresp-2015-000086. Cited in: Pubmed; PMID 607496249.
- 72. Bell BG, Schellevis F, Stobberingh E, Goossens H, Pringle M. A systematic review and metaanalysis of the effects of antibiotic consumption on antibiotic resistance. BMC Infect Dis. 2014;14(1):1-25.
- Zo 73. Low M, Neuberger A, Hooton TM, Green MS, Raz R, Balicer RD, Almog R. Association between
 urinary community-acquired fluoroquinolone-resistant Escherichia coli and neighbourhood
 antibiotic consumption: a population-based case-control study. Lancet Infect Dis. 2019;19(4):419 428.
- 74. Brown KA, Khanafer N, Daneman N, Fisman DN. Meta-analysis of antibiotics and the risk of
 community-associated Clostridium difficile infection. Antimicrob Agents Chemother. 2013
 May;57(5):2326-32. English. Epub 2013 Mar 11. doi:10.1128/aac.02176-12. Cited in: Pubmed;
 PMID 23478961.
- Australian Commission on Safety and Quality in Health Care and Australian Institute of Health and
 Welfare. The Third Australian Atlas of Healthcare Variation [Internet]. Sydney: Australian
 Commission on Safety and Quality in Health Care and Australian Institute of Health and Welfare, ;
 2018 [cited 2021 Oct 30]. Available from: https://www.safetyandquality.gov.au/our-
- 76. Ahmadizar F, Vijverberg SJH, Arets HGM, de Boer A, Turner S, Devereux G, Arabkhazaeli A,
 Soares P, Mukhopadhyay S, Garssen J, Palmer CNA, de Jongste JC, Jaddoe VWV, Duijts L, van
 Meel ER, Kraneveld AD, Maitland-van der Zee AH. Early life antibiotic use and the risk of asthma
 and asthma exacerbations in children. Pediatr Allergy Immunol. 2017 Aug;28(5):430-437. English.
 Epub 2017 Jun 8. doi:10.1111/pai.12725. Cited in: Pubmed; PMID 28423467.
- Vingaro R, Bernstein CN, Gearry R, Hviid A, Kolho KL, Kronman MP, Shaw S, Van Kruiningen H,
 Colombel JF, Atreja A. Antibiotics associated with increased risk of new-onset Crohn's disease but
 not ulcerative colitis: a meta-analysis. Am J Gastroenterol. 2014 Nov;109(11):1728-38. English.
 Epub 20140916. doi:10.1038/ajg.2014.246. Cited in: Pubmed; PMID 25223575.
- Rasmussen SH, Shrestha S, Bjerregaard LG, Ängquist LH, Baker JL, Jess T, Allin KH. Antibiotic
 exposure in early life and childhood overweight and obesity: A systematic review and metaanalysis. Diabetes Obes Metab. 2018 Jun;20(6):1508-1514. English. Epub 20180225.
 doi:10.1111/dom.13230. Cited in: Pubmed; PMID 29359849.
- Penders J, Kummeling I, Thijs C. Infant antibiotic use and wheeze and asthma risk: a systematic
 review and meta-analysis. Eur Respir J. 2011 Aug;38(2):295-302. English. Epub 20110113.
 doi:10.1183/09031936.00105010. Cited in: Pubmed; PMID 21233272.
- 80. Shao X, Ding X, Wang B, Li L, An X, Yao Q, Song R, Zhang JA. Antibiotic Exposure in Early Life
 Increases Risk of Childhood Obesity: A Systematic Review and Meta-Analysis. Front Endocrinol
 (Lausanne). 2017;8:170. English. Epub 2017 Jul 20. doi:10.3389/fendo.2017.00170. Cited in:
 Pubmed; PMID 28775712.
- 53 81. Theochari NA, Stefanopoulos A, Mylonas KS, Economopoulos KP. Antibiotics exposure and risk
 54 of inflammatory bowel disease: a systematic review. Scand J Gastroenterol. 2018 Jan;53(1):1-7.
 55 English. Epub 20171012. doi:10.1080/00365521.2017.1386711. Cited in: Pubmed; PMID
 56 29022402.

- Barlam TF. The state of antibiotic stewardship programs in 2021: The perspective of an
 experienced steward. Antimicrob Stewardship Health Epidemiol. 2021;1(1):e20. Epub 2021 Aug 5.
 doi:10.1017/ash.2021.180.
- 4 83. MacDougall C, Polk RE. Antimicrobial Stewardship Programs in Health Care Systems. Clin
 5 Microbiol Rev. 2005;18(4):638-656. doi:doi:10.1128/CMR.18.4.638-656.2005.
- 6 84. Gunduz S, Uludağ Altun H. Antibiotic resistance patterns of urinary tract pathogens in Turkish
 7 children. Glob Health Res Policy. 2018 Mar 16;3(1):10. doi:10.1186/s41256-018-0063-1.
- 8 85. Antibiotic Expert Groups. Therapeutic guidelines: respiratory. Version 4. West Melbourne (VIC):
 9 Therapeutic Guidelines Limited.; 2009. <u>http://www.tg.org.au/</u>
- 86. Antibiotic Expert Groups. Therapeutic guidelines: antibiotic. Version 16. Melbourne (VIC):
 Therapeutic Guidelines Limited; 2019.
- McCullough A, Pollack A, Hansen M, Bjerrum L, Glasziou P, Looke D, Britt H, Del Mar C. Antibiotics
 for acute respiratory infections in general practice: comparison of prescribing rates with guideline
 recommendations. Med J Aust. 2017;207(2):65-69. doi:10.5694/mja16.01042.
- 15 Gjelstad S, Hoye S, Straand J, Brekke M, Dalen I, Lindbaek M. Improving antibiotic prescribing in 88. acute respiratory tract infections: cluster randomised trial from Norwegian general practice 16 17 (prescription peer academic detailing (Rx-PAD) studv). BMJ. 2013:347:f4403. 18 doi:10.1136/bmj.f4403. Cited in: Pubmed; PMID 23894178.
- Britt H, Miller GC, Henderson J, Bayram C, Harrison C, Valenti L, Wong C, Gordon J, Pollack AJ,
 Pan Y, Charles J. General practice activity in Australia 2013–14 [Internet]. Sydney (NSW): Sydney
 University Press; 2014 [cited 2015 Jun 27]. (General Practice series no. 36). Available from:
 http://www.purl.library.usyd.edu.au/sup/9781743324219
- 23 90. Cooke G, Valenti L, Glasziou P, Britt H. Common general practice presentations and publication
 24 frequency. Aust Fam Physician. 2013 Jan 1;42:65-68.
- Baker I, Barton E. URTIs: recommended diagnosis and treatment in general practice. Prescriber.
 2013;24(19):16-28. doi:10.1002/psb.1111.
- Skolnik NS. Essential Infectious Disease Topics for Primary Care / edited by Neil S. Skolnik, Ross
 H. Albert. Totowa, NJ: Totowa, NJ : Humana Press; 2008. (Albert RH, SpringerLink, editors.).
- 83. Rowland R, Sass Z, Ponsonby A-L, Pezic A, Tang ML, Vuillermin P, Gray L, Burgner D, Group
 BISI. Burden of infection in Australian infants. J Paediatr Child Health. 2021;57(2):204-211.
 doi:<u>https://doi.org/10.1111/jpc.15174</u>.
- Bell MH, McKay B, Dale A, Guilbault R, Ermias Y. Accuracy of Signs and Symptoms for the
 Diagnosis of Acute Rhinosinusitis and Acute Bacterial Rhinosinusitis. Ann Fam Med.
 2019;17(2):164-172. doi:10.1370/afm.2354.
- J5 95. Lemiengre MB, van Driel ML, Merenstein D, Liira H, Mäkelä M, De Sutter AI. Antibiotics for acute
 rhinosinusitis in adults. Cochrane Database Syst Rev. 2018;9(9):CD006089-CD006089.
 doi:10.1002/14651858.CD006089.pub5.
- 38 96. Aring AMMD, Chan MMP. Acute Rhinosinusitis in Adults. Am Fam Physician. 2011;83(9):1057 39 1063.
- 40 97. Morcom S, Phillips N, Pastuszek A, Timperley D. Sinusitis. Aust J Gen Pract. 2016 May 41 17;45(6):374-377.
- Barnett ML, Linder JA. Antibiotic prescribing to adults with sore throat in the United States, 1997JAMA Intern Med. 2014 Jan;174(1):138-40. English. Conflicts of Interest: Dr. Linder has no
 conflicts of interest related to this manuscript. doi:10.1001/jamainternmed.2013.11673. Cited in:
 Pubmed; PMID 24091806.
- 46 99. Harris AM, Hicks LA, Qaseem A, for the High Value Care Task Force of the American College of
 47 P, for the Centers for Disease C, Prevention. Appropriate antibiotic use for acute respiratory tract
 48 infection in adults: Advice for high-value care from the american college of physicians and the
 49 centers for disease control and prevention. Ann Intern Med. 2016;164(6):425-434.
 50 doi:10.7326/M15-1840.
- Shulman ST, Bisno AL, Clegg HW, Gerber MA, Kaplan EL, Lee G, Martin JM, Van Beneden C.
 Clinical practice guideline for the diagnosis and management of group A streptococcal pharyngitis:
 2012 update by the Infectious Diseases Society of America. Clin Infect Dis. 2012 Nov
 15;55(10):1279-82. English. doi:10.1093/cid/cis847. Cited in: Pubmed; PMID 23091044.

- Tsevat J, Kotagal UR. Management of sore throats in children: a cost-effectiveness analysis. Arch
 Pediatr Adolesc Med. 1999 Jul;153(7):681-8. English. doi:10.1001/archpedi.153.7.681. Cited in:
 Pubmed; PMID 10401800.
- Pichichero ME. Group A streptococcal tonsillopharyngitis: cost-effective diagnosis and treatment.
 Ann Emerg Med. 1995 Mar;25(3):390-403. English. doi:10.1016/s0196-0644(95)70300-4. Cited in:
 Pubmed; PMID 7864482.
- 103. Spinks A, Glasziou P, Del Mar C. Antibiotics for Sore Throat. Cochrane Database Syst Rev.
 2013;(11):1-76. Epub 2013 Nov 5. doi:10.1002/14651858.CD000023.pub4.
- 9 104. Venekamp R, Sanders S, Glasziou P, Del Mar C, Rovers M. Antibiotics for acute otitis media in
 10 children. Cochrane Database Syst Rev. 2015;(6):1-86. Epub 2015 Jun 23.
 11 doi:10.1002/14651858.CD000219.pub4.
- 105. Kenna MA. Acute Otitis Media The Long and the Short of It. N Engl J Med. 2016;375(25):2492 2493. doi:10.1056/NEJMe1614712. Cited in: Pubmed; PMID 28002710.
- Jervis-Bardy J, Sanchez L, Carney AS. Otitis media in Indigenous Australian children: review of
 epidemiology and risk factors. J Laryngol Otol. 2014;128(S1):S16-S27.
 doi:10.1017/S0022215113003083.
- Morris P, Leach A, Shah P, Nelson S, Anand A, Allnutt R, Bainbridge D, Edwards K, Patel H.
 Recommendations for Clinical Care Guidelines on the Management of Otitis Media: In Aboriginal
 and Torres Strait Islander Populations 2010. Menzies School of Health; 2011. ISBN: 174241477X.
- 108. World Health Organization. Ask the expert: Influenza Q&A [Internet]. Geneva: World Health
 Organization; 2018. Influenza (Seasonal) [updated 2018 Nov 6; cited 2022 Jan 3]. Available from:
 https://www.who.int/en/news-room/fact-sheets/detail/influenza-(seasonal))
- Bollaerts K, Antoine J, Van Casteren V, Ducoffre G, Hens N, Quoilin S. Contribution of respiratory
 pathogens to influenza-like illness consultations. Epidemiol Infect. 2013;141(10):2196-2204. Epub
 2012 Dec 6. doi:10.1017/S0950268812002506.
- Bernardo CDO, González-Chica DA, Chilver M, Stocks N. Influenza-like illness in Australia: A
 comparison of general practice surveillance system with electronic medical records. Influenza
 Other Respir Viruses. 2020;14(6):605-609. doi:<u>https://doi.org/10.1111/irv.12774</u>.
- 111. Parrella A, Dalton CB, Pearce R, Litt JCB, Stocks N. ASPREN Surveillance System for Influenzalike Illness: A Comparison with Flutracking and the National Notifiable Diseases Surveillance
 System. Aust Fam Physician. 2009;38(11):932-936. doi:10.3316/informit.299978872089917.
- 32 112. Uyeki T. Influenza. Ann Int Med. 2021;174(11):ITC1. doi:10.7326/AITC202111160.
- World Health Organization. Guidelines for the clinical management of severe illness from influenza
 virus infections. Geneva: World Health Organization; 2022. p. viii, 78 p. en. ISBN: 9789240040816
 (electronic version) 9789240040823 (print version). https://apps.who.int/iris/handle/10665/352453
- Goebel MC, Trautner BW, Grigoryan L. The Five Ds of Outpatient Antibiotic Stewardship for Urinary
 Tract Infections. Clin Microbiol Rev. 2021 Dec 15;34(4):e0000320. English. Epub 20210825.
 doi:10.1128/cmr.00003-20. Cited in: Pubmed; PMID 34431702.
- Kaufman J, Temple-Smith M, Sanci L. Urinary tract infections in children: an overview of diagnosis
 and management. BMJ Paediatr Open. 2019;3(1):e000487. English. Competing interests: None
 declared. Epub 20190924. doi:10.1136/bmjpo-2019-000487. Cited in: Pubmed; PMID 31646191.
- 42 116. Etienne M, Galperine T, Caron F. Urinary tract infections in older men. NEJM. 2016;374(22):2191 43 2192.
- 117. Schaeffer AJ, Nicolle LE. Urinary tract infections in older men. NEJM. 2016;374(6):562-571.
- 118. Newman DH, Shreves AE, Runde DP. Pediatric Urinary Tract Infection: Does the Evidence Support
 Aggressively Pursuing the Diagnosis? Ann Emerg Med. 2013 2013 May 1;61(5):559-565. Epub
 2013 Jan 11. doi:https://doi.org/10.1016/j.annemergmed.2012.10.034.
- Mullakary J, Visintainer S, Tucci M, Dionne B, Conley MP, Reid DJ, Matta TM, Bartucca MA,
 Bouwmeester CJ. A multicenter retrospective review of antibiotic prescribing patterns for treatment
 of urinary tract infections in the primary care setting. J Am Coll Clin Pharm. 2021 Jun;4(6):689-696.
 English. doi:<u>http://dx.doi.org/10.1002/jac5.1430</u>. Cited in: Pubmed; PMID 2011071060.
- 120. Lee RA, Centor RM, Humphrey LL, Jokela JA, Andrews R, Qaseem A. Appropriate use of short-52 53 course antibiotics in common infections: Best practice advice from the American College of 54 Physicians [Review]. Ann Intern Med. 2021 Jun 1;174(6):822-827. English. doi:https://dx.doi.org/10.7326/M20-7355. Cited in: Pubmed; PMID 2015412545. 55

- 121. Durkin MJ, Keller M, Butler AM, Kwon JH, Dubberke ER, Miller AC, Polgreen PM, Olsen MA. An
 Assessment of Inappropriate Antibiotic Use and Guideline Adherence for Uncomplicated Urinary
 Tract Infections. Open Forum Infect Dis. 2018;5(9). doi:10.1093/ofid/ofy198.
- 4 122. Grigoryan L, Zoorob R, Wang H, Trautner BW. Low Concordance With Guidelines for Treatment
 5 of Acute Cystitis in Primary Care. Open Forum Infect Dis. 2015 Dec;2(4):ofv159. English.
 6 doi:<u>https://dx.doi.org/10.1093/ofid/ofv159</u>. Cited in: Pubmed; PMID 26753168.
- Timm B, Farag M, Bolton D, Lawrentschuk N. Adherence to Guidelines Acute urinary tract
 infection treatment by General Practitioners [Conference Abstract]. BJU Int. 2020
 Mar;125(Supplement 1):82. English. Cited in: Pubmed; PMID 631581611.
- 124. Fredericks I, Hollingworth S, Pudmenzky A, Rossato L, Kairuz T. 'Repeat' prescriptions and antibiotic resistance: findings from Australian community pharmacy. Int J Pharm Pract. 2017;25(1):50-58. doi:10.1111/ijpp.12273.
- 125. Furi J, Widmer A, Bornand D, Berger C, Huttner B, Bielicki JA. The potential negative impact of 13 14 antibiotic pack on antibiotic stewardship in primary care in Switzerland: A modelling study. Control. 2020 May 8:9(1) 15 Antimicrob Resist Infect (no pagination). Enalish. 16 doi:http://dx.doi.org/10.1186/s13756-020-00724-7. Cited in: Pubmed; PMID 631766781.
- Glasziou P, Dartnell J, Biezen R, Morgan M, Manski-Nankervis J. Antibiotic stewardship: A review of successful, evidence-based primary care strategies. Aust J Gen Pract. 2022 Jan 20;51:15-20.
- 19 127. Public Summary Document: Antibiotic repeats on the Pharmaceutical Benefits Scheme (August 20 2019 PBAC intracycle meeting)., (2019).
- 21 128. Pharmaceutical Benefits Scheme. Revised PBS listings for Antibiotic use from 1 April 2020 22 [Internet]. Canberra (ACT): Australian Government Department of Health; 2020. [updated 2020 20: cited 23 Mar 2020 Nov 20]. Available from: https://www.pbs.gov.au/info/news/2020/03/revised_pbs_listings_for_antibiotic_use_from_1_april_ 24 25 2020
- 129. US Centers for Disease Control and Prevention. Core Elements of Antibiotic Stewardship [Internet].
 Centers for Disease Control and Prevention, National Center for Emerging and Zoonotic Infectious
 Diseases, Division of Healthcare Quality Promotion; 2021. [updated 2021 Apr 7; cited 2022 Jan
 Available from: <u>https://www.cdc.gov/antibiotic-use/core-elements/index.html</u>
- Australian Commission on Safety and Quality in Health Care. Antimicrobial Stewardship in
 Australian Health Care [Internet]. Australian Commission on Safety and Quality in Health Care;
 2018 [cited 2021 Oct 17]. Available from: https://www.safetyandquality.gov.au/our-
 work/antimicrobial-stewardship/antimicrobial-stewardship-australian-health-care-ams-book
- Austalian Government. Antimicrobial stewardship. Canberra (ACT): Department of Health,; 2022.
 [2022 Nov 9]. Available from: <u>https://www.amr.gov.au/what-you-can-do/hospitals/antimicrobial-stewardship</u>
- 132. US Centers for Disease Control and Prevention. Core Elements of Outpatient Antibiotic
 Stewardship [Internet]. Centers for Disease Control and Prevention, National Center for Emerging
 and Zoonotic Infectious Diseases, Division of Healthcare Quality Promotion; 2021. [updated 2021
 Sep 8; cited 2023 Jun 16]. Available from: https://www.cdc.gov/antibiotic-use/core-
- 133. Commonwealth of Australia. Surveillance of antimicrobial use and resistance in human health
 [Internet]. Sydney (NSW): Department of Health & Department of Agriculture, Australian
 Government,; 2023 Jun 2. [2 June 2023; cited 2023 Jun 2]. Available from:
 <u>https://www.amr.gov.au/australias-response/objective-5-integrated-surveillance-and-response-</u>
 resistance-and-usage/surveillance-antimicrobial-use-and-resistance-human-health
- 47 134. US Centers for Disease Control and Prevention. Current Report: Antibiotic Use in the United States,
 48 2022 Update: Progress and Opportunities [Internet]. Centers for Disease Control and Prevention,
 49 National Center for Emerging and Zoonotic Infectious Diseases, Division of Healthcare Quality
 50 Promotion; 2022 Jun 1. [2022 Oct 6]. Available from: https://www.cdc.gov/antibiotic-
 51 use/stewardship-
 52 report/current.html#:~:text=The%20Antimicrobial%20Use%20%28AU%29%20Option%20of%20t
- 52 <u>report/current.ntm#.~.text=me%20Antimicrobial%200se%20%28A0%29%200ption%200f%201%201</u>
 53 <u>he%20CDC,hospitals%20and%20before%2C%20during%2C%20and%20after%20stewardship%</u>
 54 <u>20interventions</u>.
- 135. Cheng AC, Turnidge J, Collignon P, Looke D, Barton M, Gottlieb T. Control of fluoroquinolone
 resistance through successful regulation, Australia. Emerg Infect Dis. 2012;18(9):1453.
 doi:10.3201/eid1809.111515.

- Hashmi H, Sasoli NA, Sadiq A, Raziq A, Batool F, Raza S, Iqbal Q, Haider S, Umer Jan S, Mengal MA, Tareen AM, Khalid A, Saleem F. Prescribing Patterns for Upper Respiratory Tract Infections: A Prescription-Review of Primary Care Practice in Quetta, Pakistan and the Implications. Front Public Health. 2021;9:787933. English. doi:<u>https://dx.doi.org/10.3389/fpubh.2021.787933</u>. Cited in: Pubmed; PMID 636855770.
- 137. Commonwealth of Australia. Nudge vs Superbugs: A behavioural economics trial to reduce the overprescribing of antibiotics. [Internet]. Canberra (ACT): Department of Health, Australian Government.; 2018 [cited 2021 Aug 20]. Available from: https://www.health.gov.au/sites/default/files/documents/2021/05/nudge-vs-superbugs-report.pdf
- 138. Commonwealth of Australia. Nudge vs Superbugs: 12 Months On [Internet]. Canberra (ACT):
 Department of Health, Australian Government.; 2020 [cited 2021 Nov 6]. Available from:
 <u>https://www.health.gov.au/resources/publications/nudge-vs-superbugs-12-months-on</u>
- 139. Commonwealth of Australia. Australia's National Antimicrobial Resistance Strategy 2020 and
 Beyond [Internet]. Canberra (ACT): Department of Health, Department of Agriculture, Water and
 Environment, Australian Government; 2019 [2020 Mar 13; cited 2023 Jun 1]. Available from:
 https://www.amr.gov.au/node/1409/
- National Centre for Antimicrobial Stewardship. Our History [Internet]. Melbourne (VIC): Melbourne
 Health; 2020. [cited 2022 Jun 28]. Available from: <u>https://www.ncas-australia.org/history</u>
- 141. National Antibiotic Prescribing Survey. Welcome [Internet]. Melbourne (VIC): Melbourne Health;
 2023. [cited 2023 June 10]. Available from: <u>https://www.naps.org.au/Default.aspx</u>
- 142. National Centre for Antimicrobial Stewardship. Current Research [Internet]. Melbourne (VIC):
 Melbourne Health; 2020. [cited 2023 Jun 1]. Available from: https://www.ncas-australia.org/current-research
- 24 143. Australian Commission on Safety and Quality in Health Care. Antimicrobial prescribing practice in 25 Australian hospitals: results of the 2019 National Antimicrobial Prescribing Survey. . Sydney 26 ACSQHC: 2021 01/06/2023. Available (NSW): from: https://www.safetyandguality.gov.au/publications-and-resources/resource-library/antimicrobial-27 prescribing-practice-australian-hospitals-results-2019-hospital-national-antimicrobial-prescribing-28 29 survey
- Australian Commission on Safety and Quality in Health Care. About the AURA Surveillance System
 [Internet]. Sydney (NSW): Australian Commission on Safety and Quality in Health Care; 2023.
 [cited 2023 Jun 1]. Available from: <u>https://www.safetyandquality.gov.au/our-work/antimicrobial-</u>
 <u>resistance/antimicrobial-use-and-resistance-australia-surveillance-system/about-aura-</u>
 <u>surveillance-system</u>
- 35 145. SA Health. About the National Antimicrobial Utilisation Surveillance Program (NAUSP). SA Health, 36 Government South May Available of Australia,; 2023 11. from: 37 https://www.sahealth.sa.gov.au/wps/wcm/connect/public+content/sa+health+internet/clinical+reso urces/clinical+programs+and+practice+guidelines/infection+and+injury+management/antimicrobi 38 al+stewardship/national+antimicrobial+utilisation+surveillance+program+nausp/about+us/about+t 39 40 he+national+antimicrobial+utilisation+surveillance+program+nausp
- 41 146. WA Department of Health. Antimicrobial Stewardship [Internet]. Perth (WA): Department of Health,
 42 Government of Western Australia; 2022. [cited 2022 Jun 8]. Available from:
 43 <u>https://ww2.health.wa.gov.au/Articles/A_E/Antimicrobial-stewardship</u>
- 147. The Australian Commission on Safety and Quality in Health Care. Antimicrobial Stewardship Clinical Care Standard, [Internet]. Sydney: ACSQHC,; 2014. [cited 2021 Sep 10]. Available from: <u>https://www.safetyandquality.gov.au/sites/default/files/migrated/Antimicrobial-Stewardship-</u> <u>Clinical-Care-Standard-</u>
 web.pdf#:~:text=The%20Antimicrobial%20Stewardship%20Clinical%20Care%20Standard%20ai
- 48 web.pdf#:~:text=The%20Antimicrobial%20Stewardship%20Clinical%20Care%20Standard%20ai
 49 ms%20to,time%20and%20for%20the%20right%20duration%20based%20on
- Australian Commission on Safety and Quality in Health Care. Antimicrobial Stewardship Clinical Care Standard, [Internet]. Sydney: ACSQHC,; 2020 [cited 2022 Jun 28]. Available from: <u>https://www.safetyandquality.gov.au/our-work/clinical-care-standards/antimicrobial-stewardship-</u> <u>clinical-care-standard</u>
- National Centre for Antimicrobial Stewardship. Policy drivers for antimicrobial stewardship
 [Internet]. Melbourne (VIC): Melbourne Health; 2020. [cited 2023 Jun 1]. Available from: https://www.ncas-australia.org/policy-drivers-for-ams

- 150. Australian Commission on Safety and Quality in Health Care. National Safety and Quality Health
 Service Standards (Second Edition) [Internet]. Sydney: ACSQHC,; 2017 [cited 2022 Jun 28].
 Available from: <u>https://www.safetyandquality.gov.au/sites/default/files/2019-04/National-Safety-</u>
 and-Quality-Health-Service-Standards-second-edition.pdf
- 151. Australian Commission on Safety and Quality in Health Care. National Safety and Quality Health
 Service Standards (Second Edition 2021) [Internet]. Sydney: ACSQHC,; 2021 [cited 2023 Jun 1].
 Available from: <u>https://www.safetyandquality.gov.au/sites/default/files/2021-</u>
 05/national_safety_and_quality_health_service_nsqhs_standards_second_edition_ updated_may_2021.pdf
- 152. Australian Commission on Safety and Quality in Health Care. Preventing and Controlling Infections
 Standard. Sydney (NSW): Australian Commission on Safety and Quality in Health Care,; 2021 Jun
 1. [2023]. Available from: <u>https://www.safetyandquality.gov.au/standards/nsqhs-</u>
 standards/preventing-and-controlling-infections-standard
- 14 153. Australian Government Productivity Commission. Primary and Community Health. Report on
 Government Services, 2015. Canberra: Australian Government Productivity Commission; 2015.
 10. 445. 978-1-74037-530-6.
- 154. Britt H, Miller GC, Henderson J, Bayram C, Harrison C, Pan Y, Charles J, Chambers T, Gordon J, 17 18 Pollack AJ. A decade of Australian general practice activity 2004-05 to 2013-14. Sydney, NSW: 19 University Press; 2014 2015 Jun 27. ISBN: 9781743324233. Sydney 20 http://www.purl.library.usyd.edu.au/sup/9781743324233
- 21 155. Britt H, Miller GC, Henderson J, Bayram C, Harrison C, Valenti L, et al. General practice activity in Australia 2015-16. [Internet]. Sydney (NSW): Sydney University Press; 2016 [cited 2023 May 11]. 22 23 (General Practice series 40). Available from: no. 24 https://www.sydney.edu.au/content/dam/corporate/documents/faculty-of-medicine-and-25 health/research/research-collaborations-networks-and-groups/40general-practice-activity-in-26 australia-2015-16.pdf
- Britt H, Miller GC, Bayram C, Henderson J, Valenti L, Harrison C, et al. A decade of Australian
 general practice activity
- 2006-07 to 2015-16. [Internet]. Sydney (NSW): Sydney University Press; 2016 [cited 2023 May 11].
 (General Practice series no. 41). Available from: <u>https://www.sydney.edu.au/content/dam/corporate/documents/faculty-of-medicine-and-</u> <u>health/research/research-collaborations-networks-and-groups/41-2006-07-to-2015-16.pdf</u>
- 157. National Prescribing Service. What We Do [Internet]. Strawberry Hills (NSW): National Prescribing
 Service Limited; 2015. [updated 2015; cited 2015 Oct 29]. Available from:
 http://www.nps.org.au/about-us/what-we-do
- 158. National Prescribing Service. About Us [Internet]. Strawberry Hills (NSW): National Prescribing 36 Service 37 Limited: 2015. [updated 2015; cited 2015 Oct 25]. Available from: http://www.nps.org.au/about-us/ 38
- Wu J, Taylor D, Ovchinikova L, Heaney A, Morgan T, Dartnell J, Holbrook R, Humphreys L, Weekes
 L, Blogg S. Relationship between antimicrobial-resistance programs and antibiotic dispensing for
 upper respiratory tract infection: An analysis of Australian data between 2004 and 2015. J Int Med
 Res. 2018 Apr;46(4):1326-1338. English. Epub 20180114. doi:10.1177/0300060517740813. Cited
 in: Pubmed; PMID 29332434.
- MedicineWise N. NPS MedicineWise to cease operations after 24 years [Internet]. Sydney: NPS
 MedicineWise; 2022. [updated 2022 Sep 14; cited 2022 Nov 1]. Available from: https://www.nps.org.au/media/nps-medicine-wise-to-cease-operations-after-24-years
- Mandryk JA, Mackson JM, Horn FE, Wutzke SE, Badcock CA, Hyndman RJ, Weekes LM.
 Measuring change in prescription drug utilization in Australia. Pharmacoepidemiol Drug Saf.
 2006;15(7):477-484.
- 162. NPS Program Evaluation. Reducing Antibiotic Resistance 2012-2017: Evaluation report; [Internet].
 Strawberry Hills (NSW): National Prescribing Service Limited; 2018 [cited 2022 Jun 20]. Available
 from: <u>https://www.nps.org.au/assets/NPS/pdf/NPS-MedicineWise-Economic-evaluation-report-</u>
 Reducing-Antibiotic-Resistance-2012-17.pdf
- Patterson CA, Weekes LM, Mackson JM. Antibiotic prescribing for upper respiratory-tract infections
 in primary care. Commun Dis Intell Q Rep. 2003;27 Suppl(2003):S39-41.
 doi:10.3316/ielapa.510419649480027.

- 164. The Royal Australian College of General Practitioners. 9. Don't treat otitis media (middle ear infection) with antibiotics, in non-Indigenous children aged 2-12 years, where reassessment is a reasonable option. [Internet]. Sydney (NSW): Choosing Wisely Australia.; 2016.
 Recommendations: The Royal Australian College of General Practitioners. [updated 2016 Mar 1; cited 2022 Jan 18]. Available from: <u>https://www.choosingwisely.org.au/recommendations/racgp9</u>
- 165. The Royal Australian College of General Practitioners. 1. Do not routinely prescribe oral antibiotic
 to children with fever without an identified bacterial infection. [Internet]. Sydney (NSW): Choosing
 Wisely Australia.; 2017. Recommendations: RACP Paediatrics & Child Health Division. [cited 2022
 Jan 18]. Available from: https://www.choosingwisely.org.au/recommendations/racp1
- 166. The Royal Australian College of General Practitioners. 3. Do not give routine prophylactic antibiotics to a child after the first urinary tract infection if at low risk of recurrent urinary tract infections [Internet]. Sydney (NSW): Choosing Wisely Australia.; 2021. Recommendations: The Australian and New Zealand Society of Nephrology. [updated 2021 May 7; cited 2022 Jan 18].
 Available from: <u>https://www.choosingwisely.org.au/recommendations/anzsn3</u>
- 167. Weekes LM. Antibiotic resistance changing management of urinary tract infections in aged care.
 Med J Aust. 2015;203(9):352-352. doi:<u>https://doi.org/10.5694/mja15.01005</u>.
- 168. National Prescribing Service. What is MedicineInsight? [Internet]. Strawberry Hills (NSW): National
 Prescribing Service Limited; 2015. [updated 2015 May 13; cited 2016 May 20]. Available from:
 http://www.nps.org.au/health-professionals/medicineinsight
- 169. Health and Biomedical Informatics Centre, University of Melbourne. Background to the
 GRHANITE™ Health Informatics Unit [Internet]. Carlton (VIC): University of Melbourne. [cited 2015
 Oct 25]. Available from: <u>http://www.grhanite.com/about/background.html</u>
- Australian Bureau of Statistics. THE AUSTRALIAN STATISTICAL GEOGRAPHY STANDARD
 (ASGS) REMOTENESS STRUCTURE. Canberra (ACT): Australian Bureau of Statistics; 2022.
 [cited 2022 Aug 4]. Available from: https://www.abs.gov.au/websitedbs/D3310114.nsf/home/remoteness+structure
- Australian Bureau of Statistics. 1270.0.55.005 Volume 5 Remoteness Structure, July 2016
 [Internet]. Canberra (ACT): Australian Bureau of Statistics; 2016. Australian Statistical Geography
 Standard (ASGS) [updated 2018 Mar 16; cited 2022 Nov 19]. Available from:
 https://www.abs.gov.au/ausstats/abs@.nsf/mf/1270.0.55.005
- In State St
- Australian Commission on Safety and Quality in Health Care. CARAlert frequently asked questions.
 Sydney, NSW: Australian Commission on Safety and Quality in Health Care; 2022. Available from: https://www.safetyandquality.gov.au/our-work/antimicrobial-resistance/antimicrobial-use-andresistance-australia-surveillance-system/national-alert-system-critical-antimicrobial-resistancescaralert/caralert-frequently-asked-questions#how-does-caralert-work?
- 174. Department of Health and Aged Care CoA. National Notifiable Diseases Surveillance System
 (NNDSS). Canberra: Department of Health and Aged Care, Commonwealth of Australia; 2022 Sep
 26. [2022 Jul 17]. Available from: <u>https://www.health.gov.au/initiatives-and-programs/nndss</u>
- The Australian Group for Antimicrobial Resistance (AGAR). WHAT IS AGAR? The Australian Group for Antimicrobial Resistance (AGAR); 2022. Available from: <u>https://agargroup.org.au/about-agar/</u>
- Lahra M, Enriquez, RP for the National Neisseria Network, Australian Gonococcal Surveillance
 Programme annual report, 2015. 2015. Available from:
 https://www1.health.gov.au/internet/main/publishing.nsf/Content/cda-cdi4101i.htm
- Australian Commission on Safety and Quality in Health Care. Australian Passive AMR Surveillance (APAS). Sydney, NSW: Australian Commission on Safety and Quality in Health Care; 2022.
 Available from: <u>https://www.safetyandquality.gov.au/our-work/antimicrobial-</u>
 resistance/antimicrobial-use-and-resistance-australia-surveillance-system-aura/communityantimicrobial-resistance/australian-passive-amr-surveillance-apas
- 178. Canaway R, Boyle D, Manski-Nankervis J-A, Gray K. Identifying primary care datasets and perspectives on their secondary use: a survey of Australian data users and custodians. BMC Medical Informatics and Decision Making. 2022 Apr 6;22(1):94. doi:10.1186/s12911-022-01830-9.
- Wright M, Hall J, van Gool K, Haas M. How common is multiple general practice attendance in
 Australia? Aust Journal Gen Pract. 2018 May;47(5):289-296.

- Biezen R, Brijnath B, Grando D, Mazza D. Management of respiratory tract infections in young children - A qualitative study of primary care providers' perspectives. NPJ Prim Care Respir Med.
 2017 Dec 1;27(1) (no pagination). English. doi:<u>http://dx.doi.org/10.1038/s41533-017-0018-x</u>. Cited in: Pubmed; PMID 616374286.
- 181. Canaway R, Boyle DIR, Manski-Nankervis JAE, Bell J, Hocking JS, Clarke K, Clark M, Gunn JM,
 Emery JD. Gathering data for decisions: best practice use of primary care electronic records for
 research. Med J Aust. 2019;210(S6):S12-S16. doi:10.5694/mja2.50026.
- Youens D, Moorin R, Harrison A, Varhol R, Robinson S, Brooks C, Boyd J. Using general practice clinical information system data for research: the case in Australia. Int J Popul Data Sci. 2020 Jan 27;5(1):1099. English. Statement on Conflicts of Interest: Two of the authors, JB and SR, are currently working on a research project with NPS MedicineWise. RM sits on NPS MedicineWise's General Practice Insights advisory group. Epub 20200127. doi:10.23889/ijpds.v5i1.1099. Cited in: Pubmed; PMID 34164582.
- 14 183. PenCS. PAT CAT. Melbourne, VIC: PenCS; 2015 Feb 2. [2015; cited 2016 Feb 2]. Available from: http://www.pencs.com.au/products/pat-cat/
- 16 184. Pen CS. General Practice and AMS [Internet]. Melbourne (VIC): Pen CS; 2022. [cited 2022 Nov 2].
 17 Available from: <u>https://www.pencs.com.au/general-practice/</u>
- 18
 185. Department of General Practice UoM. Data for Decisions and the Patron program of research.

 19
 [Internet]. Melbourne (VIC): Department of General Practice, University of Melbourne; 2018. [cited

 20
 2022
 Aug
 16]. Available
 from: https://medicine.unimelb.edu.au/school-structure/generalpractice/engagement/data-for-decisions
- Mazza D, Pearce C, Turner LR, de Leon-Santiago M, McLeod A, Ferriggi J, Shearer M. The
 Melbourne East Monash General Practice Database: Using data from computerised medical
 records for research. BMJ Health Care Inform. 2016;23(2):523-528. doi:10.14236/jhi.v23i2.181.
- 187. Outcome Health. ORGANISATIONAL PRIVACY POLICY: POLAR [Internet]. Blackburn, VIC:
 Outcome Health; 2022. [cited 2022 Nov 15]. Available from:
 <u>https://www.outcomehealth.org.au/privacy-policy/</u>
- 188. Hawes L, Turner L, Buising K, Mazza D. Use of electronic medical records to describe general practitioner antibiotic prescribing patterns. Aust J Gen Pract. 2018 Nov;47(11):796-800. English. doi:10.31128/ajgp-05-18-4570. Cited in: Pubmed; PMID 31207679.
- 189. Productivity Commission. Data Availability and Use [Internet]. Canberra (ACT): Productivity
 Commission, Australian Government; 2017 [cited 2022 Aug 23]. Available from:
 https://www.pc.gov.au/inquiries/completed/data-access/report
- 190. Dollman WB, LeBlanc VT, Stevens L, O'Connor PJ, Turnidge JD. A community-based intervention
 to reduce antibiotic use for upper respiratory tract infections in regional South Australia. Med J Aust.
 2005 20 Jun;182(12):617-620. English. doi:<u>https://dx.doi.org/10.5694/j.1326-5377.2005.tb06847.x</u>. Cited in: Pubmed; PMID 40932330.
- Biezen R, Buising K, Monaghan T, Bal R, Thursky K, Cheah R, Clark M, Manski-Nankervis JA.
 Evaluating the implementation of a pilot quality improvement program to support appropriate antimicrobial prescribing in general practice. Antibiotics. 2021 Jul;10(7) (no pagination). English.
 doi:<u>http://dx.doi.org/10.3390/antibiotics10070867</u>. Cited in: Pubmed; PMID 2013048697.
- Neels AJ, Bloch AE, Gwini SM, Athan E. The effectiveness of a simple antimicrobial stewardship intervention in general practice in Australia: A pilot study. BMC Infect Dis. 2020 Aug 7;20(1) (no pagination). English. doi:<u>http://dx.doi.org/10.1186/s12879-020-05309-8</u>. Cited in: Pubmed; PMID 632556755.
- 46 193. McCullough JM, Zimmerman FJ, Rodriguez HP. Impact of clinical decision support on receipt of 47 antibiotic prescriptions for acute bronchitis and upper respiratory tract infection [Research Support, 48 Non-U.S. Gov't]. J Am Med Inform Assoc. 2014 Nov;21(6):1091-7. English. doi:http://dx.doi.org/10.1136/amiajnl-2014-002648. Cited in: Pubmed; PMID 25002458. 49
- Laka M, Milazzo A, Merlin T. Inappropriate antibiotic prescribing: understanding clinicians'
 perceptions to enable changes in prescribing practices. Aust Health Rev. 2022 Feb 1;46(1):21-27.
 English. doi:<u>https://dx.doi.org/10.1071/AH21197</u>. Cited in: Pubmed; PMID 636303707.
- Dallas A, Van Driel M, Van De Mortel T, Magin P. Antibiotic prescribing for the future: Exploring the attitudes of trainees in general practice. Br J Gen Pract. 2014 Sep 1;64(626):e561-e567. English. doi:http://dx.doi.org/10.3399/bjgp14X681373. Cited in: Pubmed; PMID 600030043.

- 196. Ranji RS, Steinman AM, Shojania GK, Gonzales GR. Interventions to Reduce Unnecessary
 Antibiotic Prescribing: A Systematic Review and Quantitative Analysis. Med Care. 2008;46(8):847 862. doi:10.1097/MLR.0b013e318178eabd.
- 197. Tonkin-Crine SKG, Tan PS, van Hecke O, Wang K, Roberts NW, McCullough A, Hansen MP, Butler
 CC, Del Mar CB. Clinician-targeted interventions to influence antibiotic prescribing behaviour for
 acute respiratory infections in primary care: An overview of systematic reviews. Cochrane
 Database Syst Rev. 2017;2017(9):CD012252-CD012252.
 doi:10.1002/14651858.CD012252.pub2.
- 198. Carracedo-Martinez E, Gonzalez-Gonzalez C, Teixeira-Rodrigues A, Prego-Dominguez J,
 Takkouche B, Herdeiro MT, Figueiras A. Computerized Clinical Decision Support Systems and
 Antibiotic Prescribing: A Systematic Review and Meta-analysis [Review]. Clin Ther. 2019
 Mar;41(3):552-581. English. doi:<u>http://dx.doi.org/10.1016/j.clinthera.2019.01.018</u>. Cited in:
 Pubmed; PMID 2001632557.
- McDonagh MS, Peterson K, Winthrop K, Cantor A, Lazur BH, Buckley DI. Interventions to reduce inappropriate prescribing of antibiotics for acute respiratory tract infections: summary and update of a systematic review. J Int Med Res. 2018;46(8):3337-3357. doi:10.1177/0300060518782519.
- Arnold SR, Straus SE. Interventions to improve antibiotic prescribing practices in ambulatory care.
 Cochrane Database Syst Rev. 2005;(4):CD003539.
- Drekonja DM, Filice G, Greer N, Olson A, Macdonald R, Rutks I, Wilt T. Antimicrobial Stewardship
 in Outpatient Settings: A Systematic Review. Infect Control Hosp Epidemiol2015. p. 142-152.
- Meeker D, Knight TK, Friedberg MW, Linder JA, Goldstein NJ, Fox CR, Rothfeld A, Diaz G, Doctor
 JN. Nudging guideline-concordant antibiotic prescribing: a randomized clinical trial. JAMA Intern
 Med. 2014 Mar;174(3):425-31. English. doi:10.1001/jamainternmed.2013.14191. Cited in:
 Pubmed; PMID 24474434.
- 25 203. Funaro J, Moehring RW, Yang S, Lee HJ, Sarubbi C, Anderson DJ, Wrenn R. Impact of education 26 and data feedback interventions on outpatient prescribing for urinary tract infections [Conference 2018 Nov;5(Supplement 27 Abstract1. Open Forum Infect Dis. 1):S89. English. 28 doi:http://dx.doi.org/10.1093/ofid/ofy210.217. Cited in: Pubmed; PMID 629442584.
- 29 204. Grigoryan L, Zoorob R, Germanos G, Sidani M, Horsfield M, Khan F, Zare M, Goebel M, Atmar R,
 30 Trautner B. Case-based audit and feedback around a decision aid improved antibiotic choice and
 31 duration for uncomplicated cystitis in primary care clinics. Fam Med Community Health. 2021;9(3).
 32 English. doi:<u>https://dx.doi.org/10.1136/fmch-2020-000834</u>. Cited in: Pubmed; PMID 635577250.
- Shively NR, Buehrle DJ, Clancy CJ, Decker BK. Reduction of overall and inappropriate antibiotic
 prescribing within a veterans affairs primary care system through peer comparison of overall
 antibiotic prescribing rates [Conference Abstract]. Open Forum Infect Dis. 2017 Sep;4(Supplement
 1):S275. English. doi:<u>http://dx.doi.org/10.1093/ofid/ofx163.614</u>. Cited in: Pubmed; PMID
 628003610.
- Shively NR, Buehrle DJ, Wagener MM, Clancy CJ, Decker BK. Improved antibiotic prescribing
 within a veterans affairs primary care system through a multifaceted intervention centered on peer
 comparison of overall antibiotic prescribing rates. Antimicrob Agents Chemother. 2020;64(1) (no
 pagination). English. doi:<u>http://dx.doi.org/10.1128/AAC.00928-19</u>. Cited in: Pubmed; PMID
 2004514290.
- 43 207. Gulliford MC, Prevost AT, Charlton J, Juszczyk D, Soames J, McDermott L, Sultana K, Wright M,
 44 Fox R, Hay AD, Little P, Moore MV, Yardley L, Ashworth M. Effectiveness and safety of
 45 electronically delivered prescribing feedback and decision support on antibiotic use for respiratory
 46 illness in primary care: REDUCE cluster randomised trial. BMJ. 2019;364:I236-I236.
 47 doi:10.1136/bmj.I236.
- Katz SE, Spencer P, Cates J, Harnack L, Xu M, Banerjee R. Improvements in appropriate
 ambulatory antibiotic prescribing using a bundled antibiotic stewardship intervention in general
 pediatrics practices. Infect Control Hosp Epidemiol. 2022:1-7. Epub 2022 Jan 31.
 doi:10.1017/ice.2021.534.
- Wasylyshyn AI, Kaye KS, Chen J, Haddad H, Nagel J, Petrie JG, Gandhi TN, Petty LA. Improving
 antibiotic use for sinusitis and upper respiratory tract infections: A virtual-visit antibiotic stewardship
 initiative. Infect Control Hosp Epidemiol. 2022:1-4. Epub 2022 Jan 3101. doi:10.1017/ice.2022.19.
- Meeker D, Linder JA, Fox CR, et al. Effect of behavioral interventions on inappropriate antibiotic
 prescribing among primary care practices: A randomized clinical trial. JAMA. 2016;315(6):562-570.
 doi:10.1001/jama.2016.0275.

- Richards AR, Linder JA. Behavioral Economics and Ambulatory Antibiotic Stewardship: A Narrative
 Review. Clin Ther. 2021 Oct;43(10):1654-1667. English. Epub 20211023.
 doi:10.1016/j.clinthera.2021.08.004. Cited in: Pubmed; PMID 34702589.
- 4 212. Ostini R, Hegney D, Jackson C, Williamson M, Mackson JM, Gurman K, Hall W, Tett SE.
 5 Systematic Review of Interventions to Improve Prescribing. Ann Pharmacother. 2009;43(3):5026 513. doi:10.1345/aph.1L488. Cited in: Pubmed; PMID 19261953.
- 7 213. Bell KJL, McCullough A, Del Mar C, Glasziou P. What's the uptake? Pragmatic RCTs may be used to estimate uptake, and thereby population impact of interventions, but better reporting of trial 8 9 recruitment processes is needed. BMC Medical Research Methodology. 2017 10 2017/12/22;17(1):174. doi:10.1186/s12874-017-0443-0.
- Coxeter P, Del Mar CB, McGregor L, Beller EM, Hoffmann TC. Interventions to facilitate shared decision making to address antibiotic use for acute respiratory infections in primary care. Cochrane Database Syst Rev. 2015;11:CD010907. doi:10.1002/14651858.CD010907.pub2.
- Van Esch T, Brabers A, Hek K, Van Dijk L, Verheij R, De Jong J. Does shared decision-making
 reduce antibiotic prescribing in primary care? J Antimicrob Chemother. 2018 Nov 1;73(11):3199 3205. English. doi:<u>https://dx.doi.org/10.1093/jac/dky321</u>. Cited in: Pubmed; PMID 626271512.
- Sargent L, McCullough A, Del Mar C, Lowe J. Is Australia ready to implement delayed prescribing
 in primary care? A review of the evidence. Aust J Gen Practitioners. 2016 Sep;45:688-690.
- Little P, Moore M, Kelly J, Williamson I, Leydon G, McDermott L, Mullee M, Stuart B. Delayed antibiotic prescribing strategies for respiratory tract infections in primary care: pragmatic, factorial, randomised controlled trial. BMJ. 2014 Apr 6;348. doi:10.1136/bmj.g1606.
- 218. Davey A, Tapley A, Mulquiney K, van Driel M, Fielding A, Holliday E, Davis J, Glasziou P, Dallas
 A, Ball J, Spike N, FitzGerald K, Magin P. Immediate and delayed antibiotic prescribing strategies
 used by Australian early-career general practitioners. Br J Gen Prac. 2021 Jun
 4;71:BJGP.2021.0026. doi:10.3399/BJGP.2021.0026.
- 26 219. Grossman Z, Silverman BG, Porter B, Miron D. Implementing the Delayed Antibiotic Therapy
 27 Approach Significantly Reduced Antibiotics Consumption in Israeli Children With First Documented
 28 Acute Otitis Media. J Pediatr Infect Dis. 2010;29(7):595-599. doi:10.1097/INF.0b013e3181d7625e.
- 220. Choi PW, Benzer JA, Coon J, Egwuatu NE, Dumkow LE. Impact of pharmacist-led selective audit and feedback on outpatient antibiotic prescribing for UTIs and SSTIs. Am J Health Syst Pharm.
 2021 01 Jun;78:S62-S69. English. doi:<u>http://dx.doi.org/10.1093/ajhp/zxab110</u>. Cited in: Pubmed; PMID 2015132315.
- Beahm NP, Smyth DJ, Tsuyuki RT. Antimicrobial utilization and stewardship in patients with
 uncomplicated urinary tract infections managed by pharmacists in the community: A sub-study of
 the RxOUTMAP trial. J Assoc Med Microbiol Infect Dis Can. 2021 Sep;6(3):205-212. English.
 doi:<u>http://dx.doi.org/10.3138/JAMMI-2020-0047</u>. Cited in: Pubmed; PMID 2014466949.
- Saeed S, Patel B, Thompson J. Evaluating clinical pharmacist management of acute sore throats
 [Conference Abstract]. Pharmacoepidemiol Drug Saf. 2020 Mar;29(Supplement 2):14. English.
 doi:<u>http://dx.doi.org/10.1002/pds.4977</u>. Cited in: Pubmed; PMID 632156081.
- 223. Delsors E, Monso F, Lopez-Roman FJ, Menarguez-Puche JF, Gonzalez-Barbera M, Hukelova H,
 Martinez-Ros MT, Lopez-Santiago A. Changes in antibiotic prescription following an education
 strategy for acute respiratory infections. NPJ Prim Care Respir Med. 2021 Dec;31(1) (no
 pagination). English. doi:<u>http://dx.doi.org/10.1038/s41533-021-00247-7</u>. Cited in: Pubmed; PMID
 2012311256.
- Althaus T, Greer RC, Swe MMM, Cohen J, Tun NN, Heaton J, Nedsuwan S, Intralawan D,
 Sumpradit N, Dittrich S, Doran Z, Waithira N, Thu HM, Win H, Thaipadungpanit J, Srilohasin P,
 Mukaka M, Smit PW, Charoenboon EN, Haenssgen MJ, Wangrangsimakul T, Blacksell S,
 Limmathurotsakul D, Day N, Smithuis F, Lubell Y. Effect of point-of-care C-reactive protein testing
 on antibiotic prescription in febrile patients attending primary care in Thailand and Myanmar: an
 open-label, randomised, controlled trial. Lancet Glob Health. 2019 Jan;7(1):e119-e131. eng.
 doi:10.1016/s2214-109x(18)30444-3. Cited in: Pubmed; PMID 30554748.
- Althaus T, Greer RC, Mukaka M, Lubell Y. Point-of-care C-reactive protein testing and antibiotic
 prescribing. Lancet Glob Health. 2021 Jan;9(1):e16. English. doi:10.1016/s2214-109x(20)30451 4. Cited in: Pubmed; PMID 33338451.
- 55 56

- Zhao H, Bian J, Wei L, Li L, Ying Y, Zhang Z, Yao X, Zhuo L, Cao B, Zhang M, Zhan S. Validation
 of an algorithm to evaluate the appropriateness of outpatient antibiotic prescribing using big data
 of Chinese diagnosis text. BMJ Open. 2020 Mar 19;10(3):e031191. English. Epub 20200319.
 doi:10.1136/bmjopen-2019-031191. Cited in: Pubmed; PMID 32198296.
- Machowska A, Marrone G, Saliba-Gustafsson P, Borg MA, Saliba-Gustafsson EA, Lundborg CS.
 Impact of a social marketing intervention on general practitioners' antibiotic prescribing practices
 for acute respiratory tract complaints in malta. Antibiotics. 2021 Apr;10(4) (no pagination). English.
 doi:http://dx.doi.org/10.3390/antibiotics10040371. Cited in: Pubmed; PMID 2006948778.
- Sedki K. Using preference learning for detecting inconsistencies in clinical practice guidelines: Methods and application to antibiotherapy. Artif Intell Med. 2018 Jul;89:24-33.
 English. doi:<u>http://dx.doi.org/10.1016/j.artmed.2018.04.013</u>. Cited in: Pubmed; PMID 2000773819.
- 12 229. Tsopra R, Sedki K, Courtine M, Falcoff H, De Beco A, Madar R, Mechai F, Lamy JB. Helping GPS to extrapolate guideline recommendations to patients for whom there are no explicit 13 the 14 recommendations, through visualization of drug properties. the example of AntibioHelp in bacterial diseases. J Am Med Inform Assoc. 2019 Jun 24;26(10):1010-15 16 1019. English. doi:http://dx.doi.org/10.1093/jamia/ocz057. Cited in: Pubmed; PMID 629687404.
- Tsopra R, Courtine M, Sedki K, Eap D, Cabal M, Cohen S, Bouchaud O, Mechai F, Lamy JB.
 AntibioGame: A serious game for teaching medical students about antibiotic use. Int J Med Inform.
 2020 Apr;136 (no pagination). English. doi:<u>http://dx.doi.org/10.1016/j.ijmedinf.2020.104074</u>. Cited
 in: Pubmed; PMID 2004547718.
- 231. Australian Commission on Safety and Quality in Health Care. Information technology to support antimicrobial stewardship [Internet]. Antimicrobial Stewardship in Australian Health Care. Sydney (NSW): Australian Commission on Safety and Quality in Health Care; 2022. 4 [2023 Apr 4]. 978-1-922563-99-6. Available from: <u>https://www.safetyandquality.gov.au/sites/default/files/2021-</u> 11/chapter_4_-_information_technology_to_support_antimicrobial_stewardship.pdf
- 26 232. King LMLM, Fleming-Dutra KEKE, Hicks LALA. Advances in optimizing antibiotic prescribing in
 27 outpatient settings. BMJ. 2018;363:k3047-k3047. doi:10.1136/bmj.k3047.
- 28 233. McCullough A, Rathbone J, Parekh S, Hoffmann T, Del Mar C. Not in my backyard: a systematic
 29 review of clinicians' knowledge and beliefs about antibiotic resistance. J Antimicrob Chemother.
 30 2015;70(9):2465.
- 234. Coenen S, Michiels B, Renard D, Denekens J, Van Royen P. Antibiotic prescribing for acute cough:
 the effect of perceived patient demand. Br J Gen Pract. 2006 Mar 1;56(524):183-190.
- Altiner A, Knauf A, Moebes J, Sielk M, Wilm S. Acute cough: A qualitative analysis of how GPs
 manage the consultation when patients explicitly or implicitly expect antibiotic prescriptions. Fam
 Pract. 2004;21(5):500-506. doi:10.1093/fampra/cmh505.
- Mohan S, Dharamraj K, Dindial R, Mathur D, Parmasad V, Ramdhanie J, Matthew J, Pinto Pereira
 LM. Physician behaviour for antimicrobial prescribing for paediatric upper respiratory tract
 infections: a survey in general practice in Trinidad, West Indies. Ann Clin Microbial Antimicrob.
 2004;3:11.
- 237. Gidengil CA, Mehrotra A, Beach S, Setodji C, Hunter G, Linder JA. What Drives Variation in
 Antibiotic Prescribing for Acute Respiratory Infections? Journal of general internal medicine. 2016
 Apr 11. English. doi:10.1007/s11606-016-3643-0. Cited in: Pubmed; PMID 27067351.
- Teixeira Rodrigues A, Roque F, Falcão A, Figueiras A, Herdeiro MT. Understanding physician
 antibiotic prescribing behaviour: a systematic review of qualitative studies. Int J Antimicrob Agents.
 2013;41(3):203. doi:10.1016/j.ijantimicag.2012.09.003.
- Lopez-Vazquez P, Vazquez-Lago JM, Figueiras A. Misprescription of antibiotics in primary care: a
 critical systematic review of its determinants. J Eval Clin Pract. 2012;18(2):473-484.
 doi:10.1111/j.1365-2753.2010.01610.x.
- 240. Rose J, Crosbie M, Stewart A. A qualitative literature review exploring the drivers influencing antibiotic over-prescribing by GPs in primary care and recommendations to reduce unnecessary prescribing [Review]. Perspect Public Health. 2021 January;141(1):19-27. English.
 doi:https://dx.doi.org/10.1177/1757913919879183. Cited in: Pubmed; PMID 2003508769.
- Substrate State
 Substrate State
 Substrate State
 Substrate
 Substrate

- 242. Xia R, Willcox M, Moore M, Liu J, Hu XY, Fei Y. Chinese doctors' perspectives in managing 1 2 antibiotic prescribing: a qualitative study [Conference Abstract]. Eur J Integr Med. 2021 3 Dec:Conference: ECIM 2022. Malaga Spain. 48 (no pagination). English. doi:https://dx.doi.org/10.1016/j.eujim.2021.101893. Cited in: Pubmed; PMID 2015980753. 4
- Saliba-Gustafsson EA, Nyberg A, Borg MA, Rosales-Klintz S, Lundborg CS. Barriers and
 facilitators to prudent antibiotic prescribing for acute respiratory tract infections: A qualitative study
 with general practitioners in Malta. PLoS ONE. 2021 Feb;16(2 Febuary) (no pagination). English.
 doi:http://dx.doi.org/10.1371/journal.pone.0246782. Cited in: Pubmed; PMID 2011046347.
- 9 244. Fletcher-Lartey S, Yee M, Gaarslev C, Khan R. Why do general practitioners prescribe antibiotics
 10 for upper respiratory tract infections to meet patient expectations: a mixed methods study. BMJ
 11 Open. 2016 Oct 24;6(10):e012244. English. Conflicts of Interest: None declared. Epub 20161024.
 12 doi:10.1136/bmjopen-2016-012244. Cited in: Pubmed; PMID 27798010.
- Simeoni M, Saragosa M, Laur C, Desveaux L, Schwartz K, Ivers N. Coping with 'the grey area' of
 antibiotic prescribing: a theory-informed qualitative study exploring family physician perspectives
 on antibiotic prescribing. BMC Primary Care. 2022 2022/07/28;23(1):188. doi:10.1186/s12875-022 01806-8.
- O'Doherty J, Leader LFW, O'Regan A, Dunne C, Puthoopparambil SJ, O'Connor R. Over
 prescribing of antibiotics for acute respiratory tract infections; a qualitative study to explore Irish
 general practitioners' perspectives. BMC Fam Pract. 2019 Feb 14;20(1):27. English. Epub
 20190214. doi:10.1186/s12875-019-0917-8. Cited in: Pubmed; PMID 30764777.
- 247. van der Zande MM, Dembinsky M, Aresi G, van Staa TP. General practitioners' accounts of
 negotiating antibiotic prescribing decisions with patients: a qualitative study on what influences
 antibiotic prescribing in low, medium and high prescribing practices. BMC Family Practice. 2019
 2019/12/10;20(1):172. doi:10.1186/s12875-019-1065-x.
- 248. Morgan S, Magin PJ, Henderson KM, Goode SM, Scott J, Bowe SJ, Regan CM, Sweeney KP,
 Jackel J, van Driel ML. Study protocol: The registrar clinical encounters in training (ReCEnT) study.
 BMC Family Practice. 2012;13:50-50. doi:10.1186/1471-2296-13-50.
- 249. Spernovasilis N, lerodiakonou D, Milioni A, Markaki L, Kofteridis DP, Tsioutis C. Assessing the knowledge, attitudes and perceptions of junior doctors on antimicrobial use and antimicrobial resistance in Greece. J Glob Antimicrob Resist. 2020 Jun;21:296-302. English. doi:<u>http://dx.doi.org/10.1016/j.jgar.2019.11.004</u>. Cited in: Pubmed; PMID 2005874331.
- Thompson A, Peterson G, O'Sullivan P, Banham E. Community antimicrobial prescribing the
 case for tailored guidelines with universal free access. In: 106. National Medicines Symposium;
 2016 May 19; Canberra (ACT). Surry Hills, NSW: NPS Medicine Wise 2016 [cited 2016 May 27].
 Internet. Available from: <u>http://www.nps.org.au/______data/assets/pdf__file/0006/318759/NMS-2016-</u>
 <u>Oral-Abstracts.pdf</u>.
- National Prescribing Service. Quick reference guide improves antibiotic prescribing [Internet].
 Strawberry Hills (NSW): National Prescribing Service Limited; 2016. [updated 2016 May 19; cited
 2021 Oct 18]. Available from: <u>http://www.nps.org.au/media-centre/media-</u>
 releases/repository/quick-reference-guide-improves-antibiotic-prescribing
- 252. Pinkerton M, Bongu J, James A, Lowder J, Durkin M. A gualitative analysis of diagnostic testing, 41 42 antibiotic selection, and quality improvement interventions for uncomplicated urinary tract September) 43 infections. PLoS ONE. 2020 Sep;15(9 pagination). (no English. doi:http://dx.doi.org/10.1371/journal.pone.0238453. Cited in: Pubmed; PMID 2007739412. 44
- Zetts RM, Stoesz A, Garcia AM, Doctor JN, Gerber JS, Linder JA, Hyun DY. Primary care
 physicians' attitudes and perceptions towards antibiotic resistance and outpatient antibiotic
 stewardship in the USA: a qualitative study. BMJ Open. 2020 Jul 14;10(7):e034983. English. Epub
 20200714. doi:10.1136/bmjopen-2019-034983. Cited in: Pubmed; PMID 32665343.
- Tyrstrup M, Andre M, Brorsson A, Grondal H, Strandberg EL, Hedin K. A study of guidelines for respiratory tract infections and their references from Swedish GPs: a qualitative analysis. Scand J
 Prim Health Care. 2020 Mar 1;38(1):83-91. English.
 doi:http://dx.doi.org/10.1080/02813432.2020.1717073. Cited in: Pubmed; PMID 630862256.
- Steels S, van Staa TP. The role of real-world data in the development of treatment guidelines: a
 case study on guideline developers' opinions about using observational data on antibiotic
 prescribing in primary care. BMC Health Serv Res. 2019 Dec 5;19(1):942. English.
 doi:<u>http://dx.doi.org/10.1186/s12913-019-4787-5</u>. Cited in: Pubmed; PMID 630098353.

- Lam JH, Pickles K, Stanaway FF, Bell KJL. Why clinicians overtest: development of a thematic framework. BMC Health Services Research. 2020 2020/11/04;20(1):1011. doi:10.1186/s12913-020-05844-9.
- 257. Stedman M, Lunt M, Davies M, Fulton-McAlister E, Hussain A, van Staa T, Anderson SG, Heald
 AH. Controlling antibiotic usage-A national analysis of General Practitioner/Family Doctor practices
 links overall antibiotic levels to demography, geography, comorbidity factors with local discretionary
 prescribing choices. Int J Clin Pract. 2020 Aug 1;74(8) (no pagination). English.
 doi:http://dx.doi.org/10.1111/ijcp.13515. Cited in: Pubmed; PMID 2004814035.
- 258. Coenen S, Ferech M, Haaijer-Ruskamp FM, Butler CC, Vander Stichele RH, Verheij TJM, Monnet
 DL, Little P, Goossens H, the ESAC Project Group. European Surveillance of Antimicrobial
 Consumption (ESAC): quality indicators for outpatient antibiotic use in Europe. Qual Safety Health
 Care. 2007;16(6):440-445. doi:10.1136/qshc.2006.021121.
- Leung V, Langford BJ, Ha R, Schwartz KL. Metrics for evaluating antibiotic use and prescribing in outpatient settings. J Antimicrob Chemother -Antimicrobial Resistance. 2021;3(3). dlab098.
 doi:10.1093/jacamr/dlab098.
- Marechal ML, Tebano G, Monnier AA, Aiaenssens N, Gyssens ICJ, Huttner B, Milanic R, Schouten
 JA, Benic MS, Versporten A, Vlahovic-Palcevski V, Zanichelli V, Wertheim HFL, Hulscher ME,
 Pulcini C. Quality indicators assessing antibiotic use in the outpatient setting: a systematic review
 followed by an international multidisciplinary consensus procedure. J Antimicrob Chemother.
 2018;73(suppl_6):vi40-vi49. doi:10.1093/jac/dky117.
- 21 261. Thilly N, Pereira O, Schouten J, Hulscher ME, Pulcini C. Proxy indicators to estimate appropriateness of antibiotic prescriptions by general practitioners: a proof-of-concept cross-22 23 sectional study based on reimbursement data, north-eastern France 2017. Euro Surveill. 2020 24 Jul;25(27). English. Conflict of interest: None declared. doi:10.2807/1560-25 7917.Es.2020.25.27.1900468. Cited in: Pubmed; PMID 32672150.
- Versporten A, Gyssens ICJ, Pulcini C, Monnier AA, Schouten JA, Milanic R, Benic MS, Tebano G,
 Marechal ML, Zanichelli V, Huttner B, Vlahovic-Palcevski V, Goossens H, Wertheim HFL, Hulscher
 ME, Aiaenssens N. Metrics to assess the quantity of antibiotic use in the outpatient setting: a
 systematic review followed by an international multidisciplinary consensus procedure. J Antimicrob
 Chemother. 2018;73(suppl_6):vi59-vi66. doi:10.1093/jac/dky119.
- World Health Organization. 2019 WHO AWaRe Classification Database of Antibiotics for evaluation
 and monitoring of use [Internet]. Geneva: World Health Organization; 2019. [updated 2019 Nov 21;
 cited 2020 Mar 20]. Available from: https://www.who.int/publications/i/item/WHOEMPIAU2019.11
- 264. World Health Organization. 2021 AWaRe classification: WHO access, watch, reserve, classification 34 35 of antibiotics for evaluation and monitoring of use [Internet]. Geneva: World Health Organization; 36 2021. [updated 2021 30; Available Sep cited 2022 Jan 201. from: 37 https://www.who.int/publications/i/item/2021-aware-classification
- Hansen M, Bjerrum, L, , Gahrn-Hansen B, Jarbol, DE, . Quality indicators for diagnosis and
 treatment of respiratory tract infections in general practice: A modified Delphi study. Scand J Prim
 Health Care. 2010;28(1):4-11. doi:10.3109/02813431003602724.
- 266. Rowe TA, Linder JA. Novel approaches to decrease inappropriate ambulatory antibiotic use. Expert
 Rev Anti Infect Ther. 2019 Jul;17(7):511-521. English. Epub 20190705.
 doi:10.1080/14787210.2019.1635455. Cited in: Pubmed; PMID 31232615.
- 267. Spivak ES, Cosgrove SE, Srinivasan A. Measuring Appropriate Antimicrobial Use: Attempts at
 Opening the Black Box. Clin Infect Dis. 2016 Dec 15;63(12):1639-1644. English. The authors have
 no reported conflicts of interest. Epub 2016 Sep 28. doi:10.1093/cid/ciw658. Cited in: Pubmed;
 PMID 27682070.
- Jorgensen LC, Friis Christensen S, Cordoba Currea G, Llor C, Bjerrum L. Antibiotic prescribing in
 patients with acute rhinosinusitis is not in agreement with European recommendations. Scand J
 Prim Health Care. 2013 Jun;31(2):101-5. English. doi:10.3109/02813432.2013.788270. Cited in:
 Pubmed; PMID 23659709.
- Wu J-C, Langford, B, Ha, R, Garber, G, Daneman, N, Johnstone, J,. Defining appropriate antibiotic
 prescribing in primary care: A modified Delphi panel approach. Official Journal of the Association
 of Medical Microbiology and Infectious Disease Canada,. 2020; 5(2):e20190023.
 doi:10.3138/jammi.2019-0023.
- 56

- Cadieux G, Tamblyn R, Dauphinee D, Libman M. Predictors of inappropriate antibiotic prescribing among primary care physicians. CMAJ : Canadian Medical Association journal = journal de l'Association medicale canadienne. 2007 Oct 9;177(8):877-83. doi:10.1503/cmaj.070151. Cited in: Pubmed; PMID 17923655.
- 5 271. Otters HBM, van der Wouden JC, Schellevis FG, van Suijlekom-Smit LWA, Koes BW. Trends in
 6 prescribing antibiotics for children in Dutch general practice. J Antimicrob Chemother.
 7 2004;53(2):361-366. doi:10.1093/jac/dkh062.
- 8 272. Samore MH, Bateman K, Alder SC, Hannah E, Donnelly S, Stoddard GJ, Haddadin B, Rubin MA,
 9 Williamson J, Stults B, Rupper R, Stevenson K. Clinical decision support and appropriateness of
 10 antimicrobial prescribing: a randomized trial. JAMA. 2005;294(18):2305.
- Wang EEL, Einarson TR, Kellner JD, Conly JM. Antibiotic Prescribing for Canadian Preschool
 Children: Evidence of Overprescribing for Viral Respiratory Infections. Clin Infect Dis.
 1999;29(1):155-160. doi:10.1086/520145.
- 14 274. Cadieux G, Abrahamowicz M, Dauphinee D, Tamblyn R. Are physicians with better clinical skills 15 on licensing examinations less likely to prescribe antibiotics for viral respiratory infections in ambulatory care settings?.[Erratum appears in Med Care. 2011 May;49(5):527-8] [Research 16 Med 17 Support, Non-U.S. Gov't]. Care. 2011 Feb;49(2):156-65. Enalish. 18 doi:https://dx.doi.org/10.1097/MLR.0b013e3182028c1a. Cited in: Pubmed; PMID 21206293.
- Alzahrani MS, Maneno MK, Daftary MN, Wingate Lm, Ettienne EB. Factors Associated with
 Prescribing Broad-Spectrum Antibiotics for Children with Upper Respiratory Tract Infections in
 Ambulatory Care Settings. Clin Med Insights Pediatr. 2018;12. doi:10.1177/1179556518784300.
- 22 276. Steinman M, Landefeld C, Gonzales R. Predictors of broad-spectrum antibiotic prescribing for
 acute respiratory tract infections in adult primary care. JAMA. 2003;289(6):719-25.
- 24 277. van Duijn HJ, Kuyvenhoven MM, Schellevis FG, Verheij TJM. Determinants of prescribing of
 25 second-choice antibiotics for upper and lower respiratory tract episodes in Dutch general practice.
 26 J Antimicrob Chemother. 2005;56(2):420-422. doi:10.1093/jac/dki214.
- 278. Islam S, Mannix MK, Breuer RK, Hassinger AB. Guideline Adherence and Antibiotic Utilization by
 Community Pediatricians, Private Urgent Care Centers, and a Pediatric Emergency Department.
 Clin Pediatr. 2020 01 Jan;59(1):21-30. English. doi:<u>http://dx.doi.org/10.1177/0009922819879462</u>.
 Cited in: Pubmed; PMID 2003502598.
- Granlund D, Zykova YV. Can Private Provision of Primary Care Contribute to the Spread of
 Antibiotic Resistance? A Study of Antibiotic Prescription in Sweden. Pharmacoecon Open. 2021
 Jun;5(2):187-195. English. doi:<u>http://dx.doi.org/10.1007/s41669-020-00234-7</u>. Cited in: Pubmed;
 PMID 2007048803.
- Blommaert A, Coenen S, Gielen B, Goossens H, Hens N, Beutels P. Patient and prescriber
 determinants for the choice between amoxicillin and broader-spectrum antibiotics: a nationwide
 prescription-level analysis. J Antimicrob Chemother. 2013 Oct 1;68(10):2383-2392.
 doi:10.1093/jac/dkt170.
- 281. Dekker ARJ, Verheij TJM, van der Velden AW. Inappropriate antibiotic prescription for respiratory
 tract indications: Most prominent in adult patients. Fam Pract. 2015 Aug 1;32(4):401-407. English.
 doi:<u>http://dx.doi.org/10.1093/fampra/cmv019</u>. Cited in: Pubmed; PMID 605695575.
- Linder JA, Huang ES, Steinman MA, Gonzales R, Stafford RS. Fluoroquinolone prescribing in the
 United States: 1995 to 2002. Am J Med. 2005 Mar;118(3):259-68. English.
 doi:10.1016/j.amjmed.2004.09.015. Cited in: Pubmed; PMID 15745724.
- 283. Robinson TF, Barsoumian AE, Aden JK, Giancola SE. Evaluation of the trends and
 appropriateness of fluoroquinolone use in the outpatient treatment of acute uncomplicated cystitis
 at five family practice clinics. J Clin Pharm Ther. 2020;45(3):513-519. doi:10.1111/jcpt.13099.
- Altiner A, Wilm S, Wegscheider K, Sielk M, Brockmann S, Fuchs A, Abholz HH, In der Schmitten
 J. Fluoroquinolones to treat uncomplicated acute cough in primary care: predictors for unjustified
 prescribing of antibiotics. J Antimicrob Chemother. 2010 Jul;65(7):1521-5. English. Epub 2010 May
 25. doi:10.1093/jac/dkq151. Cited in: Pubmed; PMID 20494927.
- Brueggen K, Revolinski S, Hart M, Wrzesinski M, Daniels AR. Empiric Fluoroquinolone Prescribing
 Trends for Cystitis in Primary Care [Conference Abstract]. Open Forum Infect Dis. 2021
 Nov;8(SUPPL 1):S794. English. doi:<u>https://dx.doi.org/10.1093/ofid/ofab466.1613</u>. Cited in:
 Pubmed; PMID 637442760.
- 56

- 286. Hamilton KW, Degnan KO, Cluzet V, Cressman L, Adu-Gyamf AB, Tolomeo P, David MZ.
 Development and validation of novel ambulatory antibiotic stewardship metrics [Conference
 Abstract]. Open Forum Infect Dis. 2018 Nov;5(Supplement 1):S514-S515. English.
 doi:<u>http://dx.doi.org/10.1093/ofid/ofy210.1469</u>. Cited in: Pubmed; PMID 629387109.
- 5 287. Brower KI, Hecke A, Mangino JE, Gerlach AT, Goff DA. Duration of Antibiotic Therapy for General
 6 Medicine and General Surgery Patients Throughout Transitions of Care: An Antibiotic Stewardship
 7 Opportunity for Noninfectious Disease Pharmacists. Hosp Pharm. 2021 Oct;56(5):532-536.
 8 English. doi:<u>https://dx.doi.org/10.1177/0018578720928265</u>. Cited in: Pubmed; PMID 2005242794.
- Polischuk E, Kathryn Mannix M, Islam S. Accuracy of outpatient antibiotic prescriptions for urinary tract infection in pediatric ambulatory care [Conference Abstract]. Open Forum Infect Dis. 2020
 Oct;7(SUPPL 1):S679. English. doi:<u>https://dx.doi.org/10.1093/ofid/ofaa439.1517</u>. Cited in:
 Pubmed; PMID 634731818.
- 13 289. Kiel A, Catalano A, Clark CM, Wattengel BA, Mason J, Sellick J, Mergenhagen KA. Antibiotic
 14 prescribing in the emergency department versus primary care: Implications for stewardship. J Am
 15 Pharm Assoc. 2020 Nov 1;60(6):789-795.e2. English.
 16 doi:<u>http://dx.doi.org/10.1016/j.japh.2020.03.016</u>. Cited in: Pubmed; PMID 2005655037.
- Clark AW, Durkin MJ, Olsen MA, Keller M, Ma Y, O'Neil CA, Butler AM. Rural–urban differences in antibiotic prescribing for uncomplicated urinary tract infection. Infect Control Hosp Epidemiol. 2021;42(12):1437-1444. Epub 2021 Feb 24. doi:10.1017/ice.2021.21.
- 291. Agdestein B, Lindbæk M, Gjelstad S. Do general practitioners follow the national guidelines for treating urinary tract infections with antibiotics? Tidsskrift for den Norske lægeforening : tidsskrift for praktisk medicin = Journal of the Norwegian Medical Association. 2011;131(17):1641.
 Norwegian, English. doi:10.4045/tidsskr.10.0396.
- 24 292. Aabenhus R, Hansen MP, Siersma V, Bjerrum L. Clinical indications for antibiotic use in Danish general practice: results from a nationwide electronic prescription database. Scand J Prim Health Care. 2017 Jun 1;35(2):162-169. English. doi:<u>http://dx.doi.org/10.1080/02813432.2017.1333321</u>.
 27 Cited in: Pubmed; PMID 621388540.
- Daga A. Nguyen OT, Moothedan E, Czyz DM, Faldu A, Ham T, Goyal A, Motwani K, Feller DB. 28 293. Antibiotic prescribing patterns for acute respiratory infections in a free clinic network: a pooled 29 Perspect. 30 cross-sectional study. Drugs Ther 2022 Jan;38(1):51-55. English. doi:https://dx.doi.org/10.1007/s40267-021-00883-6. Cited in: Pubmed; PMID 2014405693. 31
- Singer A, Fanella S, Kosowan L, Falk J, Dufault B, Hamilton K, Walus A. Informing antimicrobial stewardship: Factors associated with inappropriate antimicrobial prescribing in primary care. Fam Pract. 2018 Jul 23;35(4):455-460. English. doi:<u>http://dx.doi.org/10.1093/fampra/cmx118</u>. Cited in:
 Pubmed; PMID 624428844.
- 295. Coco AS, Horst MA, Gambler AS. Trends in broad-spectrum antibiotic prescribing for children with
 acute otitis media in the United States, 1998–2004. BMC Pediatr. 2009;9:41-41. doi:10.1186/1471 2431-9-41.
- 296. Chen CC, Wu LC, Li CY, Liu CK, Woung LC, Ko MC. Non-adherence to antibiotic prescription guidelines in treating urinary tract infection of children: a population-based study in Taiwan. J Eval
 Clin Pract. 2011;17(6):1030-1035. English. doi:<u>http://dx.doi.org/10.1111/j.1365-</u>
 2753.2010.01469.x. Cited in: Pubmed; PMID 911927607; 20738469.
- Fossum GH, Lindbæk M, Gjelstad S, Dalen I, Kværner KJ. Are children carrying the burden of
 broad-spectrum antibiotics in general practice? Prescription pattern for paediatric outpatients with
 respiratory tract infections in Norway. BMJ Open. 2013 Jan 1;3(1). doi:10.1136/bmjopen-2012002285.
- 47 298. Galimberti F, Olmastroni E, Casizzi F, Casula M, Catapano AL, Tragni E. Patterns of antibiotic
 48 prescription among adult outpatients in Italy [Conference Abstract]. Pharmacoepidemiol Drug Saf.
 49 2021 Aug;30(SUPPL 1):102. English. doi:<u>http://dx.doi.org/10.1002/pds.5305</u>. Cited in: Pubmed;
 50 PMID 636200037.
- Olsen JK, Lykkegaard J, Hansen MP, Waldorff FB, Lous J, Andersen MK. Prescription of antibiotics
 to children with acute otitis media in Danish general practice. BMC Fam Pract. 2020 Aug
 27;21(1):177. English. doi:<u>http://dx.doi.org/10.1186/s12875-020-01248-0</u>. Cited in: Pubmed; PMID
 632719013.
- 55 56

- 300. Dallas A, van Driel M, Morgan S, Tapley A, Henderson K, Oldmeadow C, Ball J, Davey A, Mulquiney K, Davis J, Spike N, McArthur L, Stewart R, Magin P. Antibiotic prescribing for acute otitis media and acute sinusitis: A cross-sectional analysis of the ReCEnT study exploring the habits of early career doctors in family practice. Fam Pract. 2017 Apr 1;34(2):180-187. English. doi:<u>http://dx.doi.org/10.1093/fampra/cmw144</u>. Cited in: Pubmed; PMID 615891710.
- 301. Wong CKM, Kung K, Au-Doung PLW, Ip M, Lee N, Fung A, Wong SYS. Antibiotic resistance rates and physician antibiotic prescription patterns of uncomplicated urinary tract infections in southern Chinese primary care. PLoS ONE. 2017 May;12(5) (no pagination). English.
 doi:http://dx.doi.org/10.1371/journal.pone.0177266. Cited in: Pubmed; PMID 615987522.
- 302. Lakkis NA, Alameddine R, Issa HG, Mahmassani D, Osman MH. Prescribing antibiotics in adults
 with respiratory tract infections in Lebanon. Int J Clin Pract. 2021 Oct;75(10) (no pagination).
 English. doi:<u>http://dx.doi.org/10.1111/ijcp.14514</u>. Cited in: Pubmed; PMID 2012909335.
- 303. Pynnonen MA, Lynn S, Kern HE, Novis SJ, Akkina SR, Keshavarzi NR, Davis MM. Diagnosis and treatment of acute sinusitis in the primary care setting: A retrospective cohort. Laryngoscope.
 2015;125(10):2266-2272. doi:10.1002/lary.25363.
- 304. Debets VEC, Verheij TJM, Van Der Velden AW. Antibiotic prescribing during office hours and outof-hours: A comparison of quality and quantity in primary care in the Netherlands. Br J Gen Pract.
 2017 Mar;67(656):e178-e186. English. doi:<u>http://dx.doi.org/10.3399/bjgp17X689641</u>. Cited in: Pubmed; PMID 614997669.
- So5. Lindback H, Lindback J, Melhus A. Inadequate adherence to Swedish guidelines for uncomplicated
 lower urinary tract infections among adults in general practice. Apmis. 2017 Sep;125(9):816-821.
 English. doi:<u>http://dx.doi.org/10.1111/apm.12718</u>. Cited in: Pubmed; PMID 616657516.
- 306. Truitt KN, Brown T, Lee JY, Linder JA. Appropriateness of Antibiotic Prescribing for Acute Sinusitis
 in Primary Care: A Cross-sectional Study. Clin Infect Dis. 2021 Jan 27;72(2):311-314. English.
 doi:10.1093/cid/ciaa736. Cited in: Pubmed; PMID 33501972.
- 307. Davey A, Tapley A, Mulquiney K, van Driel M, Fielding A, Holliday E, Ball J, Spike N, FitzGerald K,
 Magin P. Management of urinary tract infection by early-career general practitioners in Australia. J
 Eval Clin Pract. 2020 Dec;26(6):1703-1710. English. doi:<u>http://dx.doi.org/10.1111/jep.13340</u>. Cited
 in: Pubmed; PMID 2003934607.
- 308. Chua K-P, Fischer MA, Linder JA. Appropriateness of outpatient antibiotic prescribing among
 privately insured US patients: ICD-10-CM based cross sectional study. BMJ. 2019;364:k5092.
 doi:10.1136/bmj.k5092.
- 309. Thursky KA, Buising KL, Bak N, Macgregor L, Street AC, Macintyre CR, Presneill JJ, Cade JF,
 Brown GV. Reduction of broad-spectrum antibiotic use with computerized decision support in an
 intensive care unit. Int J Qual Health Care. 2006 Jun;18(3):224-31. English. Epub 20060113.
 doi:10.1093/intqhc/mzi095. Cited in: Pubmed; PMID 16415039.
- 37 310. Reinholdt KB, Rusan M, Hansen PR, Klug TE. Management of sore throat in Danish general practices. BMC Fam Pract. 2019 Jun 1;20(1):75. English. doi:<u>http://dx.doi.org/10.1186/s12875-</u>
 39 019-0970-3. Cited in: Pubmed; PMID 628075461.
- 40 311. Hummers-Pradier E, Denig P, Oke T, Lagerlov P, Wahlstrom R, Haaijer-Ruskamp FM. GPs'
 41 treatment of uncomplicated urinary tract infections A clinical judgement analysis in four European
 42 countries. Fam Pract. 1999;16(6):605-607. English. doi:<u>http://dx.doi.org/10.1093/fampra/16.6.605</u>.
 43 Cited in: Pubmed; PMID 30000177.
- Wigton RS, Darr CA, Corbett KK, Nickol DR, Gonzales R. How do community practitioners decide
 whether to prescribe antibiotics for acute respiratory tract infections? J Gen Intern Med. 2008
 Oct;23(10):1615-1620. English. doi:<u>http://dx.doi.org/10.1007/s11606-008-0707-9</u>. Cited in:
 Pubmed; PMID 50206998.
- 48 313. Keohavong B, Vonglokham M, Phoummalaysith B, Louangpradith V, Inthaphatha S, Kariya T, Saw YM, Yamamoto E, Hamajima N. Antibiotic prescription for under-fives with common cold or upper respiratory tract infection in Savannakhet Province, Lao PDR. Trop Med Health. 2019 Feb 28;47(1)
 51 (no pagination). English. doi:<u>http://dx.doi.org/10.1186/s41182-019-0143-z</u>. Cited in: Pubmed;
 52 PMID 626550638.
- 53 314. Teixeira Rodrigues A, Ferreira M, Pineiro-Lamas M, Falcao A, Figueiras A, Herdeiro MT. Determinants of physician antibiotic prescribing behavior: a 3 year cohort study in Portugal. Current 54 55 2016 Mar 4:1-9. English. Epub medical research and opinion. 2016/02/16. 56 doi:10.1185/03007995.2016.1154520. Cited in: Pubmed; PMID 26878083.
- Nash D, Harman J, Wald E, Kelleher K. Antibiotic prescribing by primary care physicians for children with upper respiratory tract infections. Arch Pediatr Adolescent Med. 2002;156(11):1114-9.
- 4 316. Chang Y, Chusri S, Sangthong R, McNeil E, Hu J, Du W, Li D, Fan X, Zhou H, Chongsuvivatwong 5 V, Tang L. Clinical pattern of antibiotic overuse and misuse in primary healthcare hospitals in the southwest of China. PLoS ONE. 2018;14(6) (no pagination). English. 6 doi:http://dx.doi.org/10.1371/journal.pone.0214779. Cited in: Pubmed; PMID 2002358297. 7
- 8 317. Wattles BA, Jawad KS, Feygin Y, Kong M, Vidwan NK, Stevenson MD, Smith MJ. Inappropriate outpatient antibiotic use in children insured by Kentucky Medicaid. Infect Control Hosp Epidemiol. 2022;43(5):582-588. Epub 2021 May 12. doi:10.1017/ice.2021.177.
- 318. Fischer MA, Mahesri M, Lii J, Linder JA. Non-Visit-Based and Non-Infection-Related Antibiotic Use
 in the US: A Cohort Study of Privately Insured Patients During 2016-2018. Open Forum Infect Dis.
 2021 Sep;8(9):ofab412. English. Epub 20210801. doi:10.1093/ofid/ofab412. Cited in: Pubmed;
 PMID 34580643.
- 319. Wickemeyer JL, Schmit M, Weinreich H. Following guidelines: prescribing practices for acute otitis
 media in children [Conference Abstract]. J Otolaryngol Head Neck Surg. 2021 Sep;165(1
 SUPPL):P296-P297. English. doi:<u>http://dx.doi.org/10.1177/01945998211030910g</u>. Cited in:
 Pubmed; PMID 636200398.
- Gasson J, Blockman M, Willems B. Antibiotic prescribing practice and adherence to guidelines in primary care in the cape town Metro district, South Africa. S Afr Med J. 2018 Apr;108(4):304-310.
 English. doi:<u>http://dx.doi.org/10.7196/SAMJ.2018.v108i4.12564</u>. Cited in: Pubmed; PMID 621511023.
- 321. Gjelstad S, Straand J, Dalen I, Fetveit A, Strom H, Lindbaek M. Do general practitioners'
 consultation rates influence their prescribing patterns of antibiotics for acute respiratory tract
 infections? [Research Support, Non-U.S. Gov't]. J Antimicrob Chemother. 2011 Oct;66(10):242533. English. doi:http://dx.doi.org/10.1093/jac/dkr295. Cited in: Pubmed; PMID 21784782.
- Lindberg BH, Gjelstad S, Foshaug M, Høye S. Antibiotic prescribing for acute respiratory tract
 infections in Norwegian primary care out-of-hours service. Scand J Prim Health Care. 2017 Apr
 3;35(2):178-185. doi:10.1080/02813432.2017.1333301.
- 30 323. Copp HL, Shapiro DJ, Hersh AL. National ambulatory antibiotic prescribing patterns for pediatric
 31 urinary tract infection, 1998-2007. Pediatrics. 2011;127(6):1027. doi:10.1542/peds.2010-3465.
- 32 324. Shapiro DJ, Hicks LA, Pavia AT, Hersh AL. Antibiotic prescribing for adults in ambulatory care in
 the USA, 2007–09. J Antimicrob Chemother. 2014;69(1):234-240. doi:10.1093/jac/dkt301.
- 325. Valent F, Gongolo F, Deroma L, Zanier L. Prescription of systemic antibiotics during pregnancy in
 primary care in Friuli Venezia Giulia, Northeastern Italy. J Matern Fetal Neonatal Med. 2015
 Jan;28(2):210-5. doi:10.3109/14767058.2014.906572. Cited in: Pubmed; PMID 24766037.
- 326. Barlam TF, Morgan JR, Wetzler LM, Christiansen CL, Drainoni M-L. Antibiotics for respiratory tract infections: a comparison of prescribing in an outpatient setting. Infect Control Hosp Epidemiol. 2015 Feb;36(2):153-159. English. doi:<u>http://dx.doi.org/10.1017/ice.2014.21</u>. Cited in: Pubmed; PMID 1652443016; 25632997.
- 41 327. Hersh AL, Fleming-Dutra KE, Shapiro DJ, Hyun DY, Hicks LA. Frequency of First-line Antibiotic
 42 Selection Among US Ambulatory Care Visits for Otitis Media, Sinusitis, and Pharyngitis. JAMA
 43 Intern Med. 2016;176(12):1870-1872. doi:10.1001/jamainternmed.2016.6625.
- Kozyrskyj AL, Dahl ME, Chateau DG, Mazowita GB, Klassen TP, Law BJ. Evidence-based
 prescribing of antibiotics for children: role of socioeconomic status and physician characteristics.
 CMAJ : Canadian Medical Association journal = journal de l'Association medicale canadienne.
 2004;171(2):139-145. doi:10.1503/cmaj.1031629. Cited in: Pubmed; PMID PMC450362.
- 48 329. Gjelstad S, Dalen I, Lindbaek M. GPs' antibiotic prescription patterns for respiratory tract infections49 -still room for improvement. Scand J Prim Health Care. 2009;27(4):208-15. English.
 50 doi:<u>http://dx.doi.org/10.3109/02813430903438718</u>. Cited in: Pubmed; PMID 19929185.
- S1 330. Linder JA, Bates DW, Lee GM, Finkelstein JA. Antibiotic treatment of children with sore throat.
 JAMA. 2005;294(18):2315-2322. doi:10.1001/jama.294.18.2315.
- 53 331. Fortin E, Deceuninck G, Sirois C, Quach C, Simard M, Jean S, Irace-Cima A, Magali-Ufitinema N.
 54 Presence of Chronic Diseases and Compliance with Quebec Provincial Guidelines for Outpatient
 55 Antibiotic Prescription, Quebec, Canada, 2010-2017 [Conference Abstract]. Open Forum Infect
 56 Dis. 2021 Nov;8(SUPPL 1):S164. English. doi:<u>https://dx.doi.org/10.1093/ofid/ofab466.301</u>. Cited
 57 in: Pubmed; PMID 637443114.

- 332. Schroeck JL, Ruh CA, Sellick JA, Ott MC, Mattappallil A, Mergenhagen KA. Factors associated
 with antibiotic misuse in outpatient treatment for upper respiratory tract infections. Antimicrob
 Agents Chemother. 2015;59(7):3848. doi:10.1128/AAC.00652-15.
- 333. Fernandez-Urrusuno R, Flores-Dorado M, Vilches-Arenas A, Montero-Balosa MC. Predictors
 associated with inappropriate antibiotic prescribing in primary care. Int J Clin Pharm.
 2013;35(6):1273-1273.
- 334. Dantuluri KL, Bruce J, Edwards KM, Banerjee R, Griffith H, Howard LM, Grijalva CG. Rurality of Residence and Inappropriate Antibiotic Use for Acute Respiratory Infections Among Young Tennessee Children. Open Forum Infect Dis. 2021 Jan;8(1):ofaa587. English. Epub 2020 Dec 15. doi:10.1093/ofid/ofaa587. Cited in: Pubmed; PMID 33511228.
- 335. De Man J, Remmen R, Philips H. Differences in antibiotic prescribing quality in Belgian out-of-hours primary care services. Acta Clin Belg. 2022. English.
 doi:<u>https://dx.doi.org/10.1080/17843286.2022.2081772</u>. Cited in: Pubmed; PMID 2017458990.
- 14 336. Plate A, Kronenberg A, Risch M, Mueller Y, Di Gangi S, Rosemann T, Senn O. Treatment of urinary
 tract infections in Swiss primary care: quality and determinants of antibiotic prescribing. BMC Fam
 Pract. 2020 Jul 1;21(1):125. English. doi:<u>http://dx.doi.org/10.1186/s12875-020-01201-1</u>. Cited in:
 Pubmed; PMID 632250855.
- 337. Oakes JM, Kaufman JS. Methods in social epidemiology. 1st ed. San Francisco, CA: Jossey-Bass;
 2006. ISBN: 9780787985943.
- 338. Williams DR. Race/ethnicity and socioeconomic status: measurement and methodological issues.
 Int J Health Serv. 1996;26(3):483-505. English. doi:10.2190/u9qt-7b7y-hq15-jt14. Cited in:
 Pubmed; PMID 8840198.
- 339. Fleming-Dutra KE, Hersh AL, Shapiro DJ, Bartoces M, Enns EA, File TM, Jr., Finkelstein JA, Gerber
 JS, Hyun DY, Linder JA, Lynfield R, Margolis DJ, May LS, Merenstein D, Metlay JP, Newland JG,
 Piccirillo JF, Roberts RM, Sanchez GV, Suda KJ, Thomas A, Woo TM, Zetts RM, Hicks LA.
 Prevalence of Inappropriate Antibiotic Prescriptions Among US Ambulatory Care Visits, 2010-2011.
 JAMA. 2016 May 3;315(17):1864-73. English. doi:10.1001/jama.2016.4151. Cited in: Pubmed;
 PMID 27139059.
- 340. Hersh AL, Shapiro DJ, Pavia AT, Shah SS. Antibiotic Prescribing in Ambulatory Pediatrics in the
 United States. Pediatrics. 2011;128(6):1053-1061. doi:10.1542/peds.2011-1337.
- 341. Malo S, Bjerrum L, Feja C, Lallana M-J, Moliner J, Rabanaque M-J. Compliance with 31 32 Recommendations on Outpatient Antibiotic Prescribing for Respiratory Tract Infections: The Case 33 Spain. Basic Clin Pharmacol Toxicol. 2015 Apr:116(4):337-342. Enalish. of doi:http://dx.doi.org/10.1111/bcpt.12316. Cited in: Pubmed; PMID 1662573921. 34
- 342. Van Roosmalen MS, Braspenning JCC, De Smet PAGM, Grol RPTM. Antibiotic prescribing in primary care: First choice and restrictive prescribing are two different traits. Qual Safety Health Care. 2007 Apr;16(2):105-109. English. doi:<u>http://dx.doi.org/10.1136/qshc.2006.018580</u>.
- 343. WA Primary Health Alliance. About Us [Internet]. Rivervale, WA: WA Primary Health Alliance; 2022.
 [updated 2022; cited 2022 Sep 5]. Available from: <u>https://www.wapha.org.au/about-us/</u>
- 344. Busingye D, Gianacas C, Pollack A, Chidwick K, Merrifield A, Norman S, Mullin B, Hayhurst R,
 Blogg S, Havard A, Stocks N. Data Resource Profile: MedicineInsight, an Australian national
 primary health care database. Int J Epidemiol. 2019;48(6):1741-1741h. doi:10.1093/ije/dyz147.
- 43 345. MedicineInsight. MedicineInsight Data Book Version 1.2. Sydney: NPS MedicineWise; 2016.
- 44 346. Anonymous. NPS MedicineWise: reducing antibiotic resistance. J Antimicrob Chemother 45 Antimicrobial Resistance. 2019;1(1). dlz020. doi:10.1093/jacamr/dlz020.
- 46 347. World Health Organization. International Classification of Diseases (ICD) [Internet]. Geneva: World
 47 Health Organization; 2016. [cited 2016 May 1]. Available from: 48 <u>http://www.who.int/classifications/icd/en/</u>
- 348. Consortium and the World Organization of Family Doctors' (WONCA) International Classification
 Committee (WICC). International Classification of Primary Care, 2nd Edition (ICPC-2) [Internet].
 Geneva.: World Health Organization; 2003. [cited 2018 Aug 20]. Available from:
 https://www.who.int/standards/classifications/other-classifications/international-classification-of-primary-care
- MIMS Australia. Product Overview [Internet]. St Leonards (NSW): MIMS Australia; 2015. [cited
 2016 Jun 8]. Available from: <u>http://www.mims.com.au/index.php/products/product-overview</u>

- 350. World Health Organization. Anatomical Therapeutic Chemical classification system: Structure and 1 2 principles [Internet]. Oslo, Norway: WHO Collaborating Centre for Drug Statistics Methodology; 2019. [updated 2018 Feb 15: cited 2019 Nov 1]. Available 3 from: 4 https://www.whocc.no/atc/structure and principles/
- 351. Herath VCK, Carapetis J. Sore throat: Is it such a big deal anymore? J Infect. 2015;71:S101-S105.
 doi:10.1016/j.jinf.2015.04.010.
- Australian Institute of Health and Welfare. Acute rheumatic fever and rheumatic heart disease—in brief. [Internet]. Canberra (ACT): Australian Institute of Health and Welfare.; 2019. Cat. no. CVD 87. [cited 2022 Jun 5]. Available from: <u>https://www.aihw.gov.au/getmedia/a352bfd2-0af8-4157-9f45-9b066360a7d4/aihw-cvd-87.pdf.aspx?inline=true</u>
- Stuck AK, Täuber MG, Schabel M, Lehmann T, Suter H, Mühlemann K. Determinants of Quinolone
 versus Trimethoprim-Sulfamethoxazole Use for Outpatient Urinary Tract Infection. Antimicrob
 Agents Chemother. 2012;56(3):1359-1363.
- 354. Steering Committee for the Review of Government Service Provision. Report on government
 services 2015: Indigenous compendium. . Canberra (ACT): Productivity Commission, Australian
 Government; 2015.
- Australian Institute of Health and Welfare. National best practice guidelines for collecting
 Indigenous status in health data sets [Internet]. Canberra (ACT): Australian Institute of Health and
 Welfare; 2010 [cited 2022 Nov 1]. Available from: https://www.aihw.gov.au/reports/indigenous-australians/national-guidelines-collecting-health-data-sets/notes
- 356. Little R. Statistical analysis with missing data / Roderick J. A. Little, Donald B. Rubin. 2nd ed.:
 Hoboken, New Jersey : John Wiley & Sons, Inc.; 2002. (Rubin DBa, editor.).
- 357. Regenstrief Institute I. What LOINC is [Internet]. Indianapolis, IN: Regenstrief Institute, Inc; 2022.
 [cited 2022 Jun 15]. Available from: <u>https://loinc.org/get-started/what-loinc-is/</u>
- 358. Regenstrief Institute I. LOINC version 2.66 [Internet]. Indianapolis, IN: Regenstrief Institute, Inc;
 2019. [cited 2019 Aug 20]. Available from: <u>https://loinc.org/download/loinc-2-66-table/</u>
- 359. Grossman Z, Silverman BG, Miron D. Physician specialty is associated with adherence to treatment
 guidelines for acute otitis media in children. Acta Paediatr. 2013 Jan;102(1):e29-e33. English.
 doi:<u>http://dx.doi.org/10.1111/apa.12051</u>. Cited in: Pubmed; PMID 52295530.
- 30 360. StataCorp. Stata Statistical Software: Release 16. . College Station, TX: StataCorp LLC.; 2019.
- 361. Howell DC, Everitt BS, editors. Encyclopedia of statistics in behavioral science. New York: New
 York : John Wiley & Sons; 2007.
- 362. Pituch KAa. Applied multivariate statistics for the social sciences. Sixth edition / Keenan A. Pituch,
 James P. Stevens.. ed.: London : Routledge; 2016. (Stevens Ja, Stevens J, editors.).
- 363. Byrne MK, Miellet S, McGlinn A, Fish J, Meedya S, Reynolds N, van Oijen AM. The drivers of
 antibiotic use and misuse: the development and investigation of a theory driven community
 measure. BMC Public Health. 2019 Oct 30;19(1):1425. doi:10.1186/s12889-019-7796-8.
- 364. Greene W, Zhang Q. Chapter 3 Nonlinear and Related Panel Data Models. In: Tsionas M, editor.
 39 Panel Data Economet. Academic Press; 2019. p. 45-96.
- 365. Matyas L. The econometrics of multi-dimensional panels. Advanced studies in theoretical and
 applied econometrics Berlin: Springer. 2017.
- 42 366. Martinez ME. The calendar of epidemics: Seasonal cycles of infectious diseases. PLoS Pathog.
 43 2018 Nov;14(11):e1007327. English. The authors have declared that no competing interests exist.
 44 Epub 2018 Nov 8. doi:10.1371/journal.ppat.1007327. Cited in: Pubmed; PMID 30408114.
- 45 367. Moriyama M, Hugentobler WJ, Iwasaki A. Seasonality of Respiratory Viral Infections. Annu Rev
 46 Virol. 2020;7(1):83-101. doi:10.1146/annurev-virology-012420-022445. Cited in: Pubmed; PMID
 47 32196426.
- 368. Durkin MJ, Jafarzadeh SR, Hsueh K, Sallah YH, Munshi KD, Henderson RR, Fraser VJ. Outpatient
 Antibiotic Prescription Trends in the United States: A National Cohort Study. Infect Control Hosp
 Epidemiol. 2018;39(5):584-589. Epub 2018 Mar 8. doi:10.1017/ice.2018.26.
- Senting State A, Kronenberg A, Risch M, Mueller Y, Di Gangi S, Rosemann T, Senn O. Treatment of urinary
 tract infections in Swiss primary care: quality and determinants of antibiotic prescribing. BMC Fam
 Pract. 2020 Jul 1;21(1):125. eng. The authors declare that they have no competing interests. Epub
 20200701. doi:10.1186/s12875-020-01201-1. Cited in: Pubmed; PMID 32611320.
- 55

- 370. Gerber JS, Prasad PA, Fiks AG, Localio AR, Grundmeier RW, Bell LM, Wasserman RC, Keren R,
 Zaoutis TE. Effect of an outpatient antimicrobial stewardship intervention on broad-spectrum
 antibiotic prescribing by primary care pediatricians: a randomized trial. JAMA. 2013 Jun
 12;309(22):2345-52. English. doi:10.1001/jama.2013.6287. Cited in: Pubmed; PMID 23757082.
- 371. Gerber JS, Prasad PA, Localio AR, Fiks AG, Grundmeier RW, Bell LM, Wasserman RC, Rubin
 DM, Keren R, Zaoutis TE. Racial differences in antibiotic prescribing by primary care pediatricians.
 Pediatrics. 2013 Apr;131(4):677-684. English. doi:<u>http://dx.doi.org/10.1542/peds.2012-2500</u>. Cited
 in: Pubmed; PMID 368692957.
- 372. Van Staa T, Li Y, Gold N, Chadborn T, Welfare W, Palin V, Ashcroft DM, Bircher J. Comparing
 antibiotic prescribing between clinicians in UK primary care: an analysis in a cohort study of eight
 different measures of antibiotic prescribing. BMJ Qual Saf. 2022:bmjqs-2020-012108.
 doi:10.1136/bmjqs-2020-012108.
- 373. Cadieux G. Predictors of antibiotic prescribing among primary care physicians. Montreal: McGill,
 Department of Epidemiology & Biostatistics; 2004. Master of Science.
- Petersen I, Gilbert R, Evans S, Ridolfi A, Nazareth I. Oral antibiotic prescribing during pregnancy
 in primary care: UK population-based study [Research Support, Non-U.S. Gov't]. J Antimicrob
 Chemother. 2010 Oct;65(10):2238-46. English. doi:<u>http://dx.doi.org/10.1093/jac/dkq307</u>. Cited in:
 Pubmed; PMID 20716554.
- Stroup WW. Generalized linear mixed models : modern concepts, methods and applications. 1st
 ed. Boca Raton, FL: CRC Press, an imprint of Taylor and Francis; 2012. Table 1.4, Typology of
 Linear Models; p. 20.
- 376. StataCorp. Irtest Likelihood-ratio test after estimation [Internet]. College Station, TX: StataCorp
 LLC.; 2021 [cited 2022 Jul 5]. Available from: <u>https://www.stata.com/manuals/rlrtest.pdf</u>
- 377. Scott MAe, Simonoff JSe, Marx BDe. The SAGE handbook of multilevel modeling / edited by Marc
 A. Scott, Jeffrey S. Simonoff and Brian D. Marx. Los Angeles : SAGE; 2013. (Handbook of multilevel
 modeling).
- 378. Brant R. Assessing Proportionality in the Proportional Odds Model for Ordinal Logistic Regression.
 Biometrics. 1990;46(4):1171-1178. Full publication date: Dec., 1990. doi:10.2307/2532457.
- 379. StataCorp. Base Reference Manual [Internet]. College Station, TX: StataCorp LLC.; 2021 [cited
 2022 Jun 3]. Available from: <u>https://www.stata.com/manuals/rologit.pdf</u>
- 380. StataCorp. MULTILEVEL MIXED-EFFECTS REFERENCE MANUAL RELEASE 17 [Internet].
 College Station, TX: StataCorp LLC.; 2021 [cited 2022 Jul 6]. Available from: https://www.stata.com/manuals/meestaticc.pdf#meestaticc
- 34 381. StataCorp. Base Reference Manual. College Station, TX: StataCorp LLC.; 2021. Available from: https://www.stata.com/manuals/ricc.pdf
- 36 382. Liljequist D, Elfving B, Skavberg Roaldsen K. Intraclass correlation A discussion and
 37 demonstration of basic features. PLOS ONE. 2019;14(7):e0219854.
 38 doi:10.1371/journal.pone.0219854.
- 383. Austin P, Goel V, van Walraven C. An Introduction to Multilevel Regression Models. Can J Pub
 Health. 2001 Mar 1;92:150-4. doi:10.1007/BF03404950.
- 41 Ahmed H. Farewell D. Jones HM. Francis NA. Paraniothy S. Butler CC. Incidence and antibiotic 384. prescribing for clinically diagnosed urinary tract infection in older adults in UK primary care, 2004-42 43 2014. PLoS ONE. 2018 Jan:13(1) (no pagination). English. doi:http://dx.doi.org/10.1371/journal.pone.0190521. Cited in: Pubmed; PMID 620083241. 44
- 385. Gong Y, Li H, Yang H, Tan K, Liu W, Li X, Wu J, Zhang G, Yin X. Evaluation of the Quality of
 Antibiotic Prescribing in Primary Care: A Multicenter Longitudinal Study From Shenzhen, China.
 Front Pharmacol. 2020 Feb 19;11 (no pagination). English.
 doi:http://dx.doi.org/10.3389/fphar.2020.617260. Cited in: Pubmed; PMID 634383882.
- 386. Andrews A, Budd EL, Hendrick A, Ashiru-Oredope D, Beech E, Hopkins S, Gerver S, MullerPebody B. Surveillance of antibacterial usage during the COVID-19 pandemic in England, 2020.
 Antibiotics. 2021 Jul;10(7) (no pagination). English.
 doi:http://dx.doi.org/10.3390/antibiotics10070841. Cited in: Pubmed; PMID 2007864100.
- 387. Gunnlaugsdottir MR, Linnet K, Jonsson JS, Blondal AB. Encouraging rational antibiotic prescribing
 behaviour in primary care prescribing practice among children aged 0-4 years 2016-2018: an
 observational study. Scand J Prim Health Care. 2021 Sep 1;39(3):373-381. English.
 doi:http://dx.doi.org/10.1080/02813432.2021.1958506. Cited in: Pubmed; PMID 635812798.

- 388. Mulder M, Baan E, Verbon A, Stricker B, Verhamme K. Trends of prescribing antimicrobial drugs for urinary tract infections in primary care in the netherlands: A population-based cohort study. BMJ
 Open. 2019 May 1;9(5) (no pagination). English. doi:<u>http://dx.doi.org/10.1136/bmjopen-2018-027221</u>. Cited in: Pubmed; PMID 627736217.
- 389. Piraux A, Faure S, Naber KG, Alidjanov JF, Ramond-Roquin A. Changes in the management of
 urinary tract infections in women: impact of the new recommendations on antibiotic prescribing
 behavior in France, between 2014 and 2019. BMC Health Serv Res. 2021 Jun 28;21(1):612.
 English. doi:<u>http://dx.doi.org/10.1186/s12913-021-06653-4</u>. Cited in: Pubmed; PMID 635479369.
- Signorphilic Structure
 Signorphilic Structure
 Kimura Y, Fukuda H, Hayakawa K, Ide S, Ota M, Saito S, Ishikane M, Kusama Y, Matsunaga N, Ohmagari N. Longitudinal trends of and factors associated with inappropriate antibiotic prescribing for non-bacterial acute respiratory tract infection in Japan: A retrospective claims database study, 2012-2017. PloS one. 2019;14(10):e0223835-e0223835. doi:10.1371/journal.pone.0223835.
- 391. Bee Dagum E, Bianconcini S. Linear Filters Seasonal Adjustment Methods: Census Method II and
 Its Variants. Cham: Cham: Springer International Publishing; 2016. p. 79-114.
- 392. Svetunkov I, Petropoulos F. Old dog, new tricks: a modelling view of simple moving averages. Int
 J Product Res. 2018;56(18):6034-6047. doi:10.1080/00207543.2017.1380326.
- 17 393. Cox NJ. Speaking Stata: Smoothing in Various Directions. Stata J. 2005;5(4):574-593.
 18 doi:10.1177/1536867X0500500408.
- 394. Royston P, Cox NJ. A Multivariable Scatterplot Smoother. Stata J. 2005;5(3):405-412.
 doi:10.1177/1536867X0500500309.
- 395. StataCorp. lowess Lowess smoothing [Internet]. College Station, TX: StataCorp LLC.; 2021
 [cited 2022 Aug 3]. Available from: <u>https://www.stata.com/manuals/rlowess.pdf</u>
- 396. Bianchi M, Boyle M, Hollingsworth D. A comparison of methods for trend estimation. Appl Econ
 Lett. 1999;6(2):103-109. Applied Economics Letters. doi:10.1080/135048599353726.
- 397. Barnes EA, Barnes RJ. Estimating Linear Trends: Simple Linear Regression versus Epoch
 Differences. J Clim. 2015;28(24):9969-9976. doi:10.1175/JCLI-D-15-0032.1.
- 398. StataCorp. regress Linear regression [Internet]. College Station, TX: StataCorp LLC.; 2021 [cited
 2022 Jun 20]. Available from: <u>https://www.stata.com/manuals/rregress.pdf</u>
- 399. Department of Health. Public Datasets: Influenza (laboratory confirmed) Canberra (ACT):
 Department of Health, Australian Government; 2021. [cited 2021 Feb 19]. Available from: https://www1.health.gov.au/internet/main/publishing.nsf/Content/ohp-pub-datasets.htm
- Kahan NR, Friedman NL, Lomnicky Y, Hemo B, Heymann AD, Shapiro M, Kokia E. Physician
 speciality and adherence to guidelines for the treatment of unsubstantiated uncomplicated urinary
 tract infection among women [Comparative Study]. Pharmacoepidem Drug Safety. 2005
 May;14(5):357-61. English. Cited in: Pubmed; PMID 15517543.
- 401. Meyer HE, Lund BC, Heintz BH, Alexander B, Egge JA, Livorsi DJ. Identifying Opportunities to
 Improve Guideline-Concordant Antibiotic Prescribing in Veterans with Acute Respiratory Infections
 or Cystitis. Infect Control Hosp Epidemiol. 2017 Jun 1;38(6):724-728. English.
 doi:<u>http://dx.doi.org/10.1017/ice.2017.59</u>. Cited in: Pubmed; PMID 616862011.
- 402. Ong DS, Kuyvenhoven MM, van Dijk L, Verheij TJ. Antibiotics for respiratory, ear and urinary tract disorders and consistency among GPs [Research Support, Non-U.S. Gov't]. J Antimicrob Chemother. 2008 Sep;62(3):587-92. English. doi:<u>http://dx.doi.org/10.1093/jac/dkn230</u>. Cited in:
 43 Pubmed; PMID 18544602.
- 403. Tyrstrup M, van der Velden A, Engstrom S, Goderis G, Molstad S, Verheij T, Coenen S,
 Adriaenssens N. Antibiotic prescribing in relation to diagnoses and consultation rates in Belgium,
 the Netherlands and Sweden: use of European quality indicators. Scand J Prim Health Care. 2017
 Mar;35(1):10-18. English. Epub 20170303. doi:10.1080/02813432.2017.1288680. Cited in:
 Pubmed; PMID 28277045.
- 49 404. Adriaenssens N, Coenen S, Tonkin-Crine S, Verheij TJ, Little P, Goossens H, on behalf of the 50 EPG. European Surveillance of Antimicrobial Consumption (ESAC): disease-specific quality 51 indicators for outpatient antibiotic prescribing. BMJ Qual Saf. 2011 Mar 21. doi:10.1136/bmjqs.2010.049049. Cited in: Pubmed; PMID 21441602. 52
- Lin YC, Lin HC, Lin HC. Doctor characteristics and prescribing antibiotics for urinary tract infections:
 the experience of an Asian country. J Eval Clin Prac. 2010;16(6):1221-1226. doi:10.1111/j.1365 2753.2009.01299.x.

- 406. Huang ES, Stafford RS. National Patterns in the Treatment of Urinary Tract Infections in Women by Ambulatory Care Physicians. Arch Int Med. 2002;162(1):41-47. doi:10.1001/archinte.162.1.41.
- 407. Veninga CCM, Lundborg CS, Lagerløv P, Hummers-Pradier E, Denig P, Haaijer-Ruskamp FM,
 Additional Members of the Drug Education Project Group FM. Treatment of Uncomplicated Urinary
 Tract Infections: Exploring Differences in Adherence to Guidelines Between Three European
 Countries. Ann Pharmacother. 2000;34:19-26. doi:10.1345/aph.19068.
- 408. StataCorp. me Introduction to multilevel mixed-effects models. STATA MULTILEVEL MIXED-EFFECTS REFERENCE MANUAL RELEASE 17. College Station, TX: StataCorp LLC.; 2021 [5 Jun 2022]. ISBN-10: 1-59718-339-3 ISBN-13: 978-1-59718-339-0. Available from: https://www.stata.com/manuals/me.pdf
- StataCorp. melogit Multilevel mixed-effects logistic regression. MULTILEVEL MIXED-EFFECTS
 REFERENCE MANUAL RELEASE 17. College Station, TX: StataCorp LLC.; 2021 [4 Jul 202].
 ISBN-10: 1-59718-339-3 ISBN-13: 978-1-59718-339-0. Available from: https://www.stata.com/manuals/memelogit.pdf
- 410. StataCorp. meologit Multilevel mixed-effects ordered logistic regression. MULTILEVEL MIXED EFFECTS REFERENCE MANUAL RELEASE 17. College Station, TX: StataCorp LLC.; 2021 [4
 Jul 2022]. ISBN-10: 1-59718-339-3 ISBN-13: 978-1-59718-339-0. Available from:
 https://www.stata.com/manuals/memeologit.pdf
- 411. StataCorp. ologit Ordered logistic regression [Internet]. College Station, TX: StataCorp LLC.;
 2021 [cited 2022 Jul 5]. Available from: <u>https://www.stata.com/manuals/rologit.pdf</u>
- 412. McKertich K, Hanegbi U. Recurrent UTIs and cystitis symptoms in women. Aust J Gen Practition.
 2021 Apr;50:199-205. Epub 2021 Mar 24.
- 413. Naber KG. Treatment options for acute uncomplicated cystitis in adults. J Antimicrob Chemother.
 2000 Sep;46 Suppl 1:23-7; discussion 63-5. English. Cited in: Pubmed; PMID 11051620.
- 414. Kotwani A, Wattal C, Katewa S, Joshi PC, Holloway K. Factors influencing primary care physicians
 to prescribe antibiotics in Delhi India. Fam Pract. 2010;27(6):684-690. doi:10.1093/fampra/cmq059.
- 415. Llor C, Rabanaque G, Lopez A, Cots JM. The adherence of GPs to guidelines for the diagnosis
 and treatment of lower urinary tract infections in women is poor. Fam Pract. 2011 Jun 1;28(3):294299. English. doi:<u>http://dx.doi.org/10.1093/fampra/cmq107</u>. Cited in: Pubmed; PMID 361822651.
- 416. Essafi S, Letaief AO, Phillips E, Vardanega V. Antimicrobial stewardship and economic evaluation
 of urinary tract infection management in primary health care in tunisia. Fam Med Primary Care Rev.
 2021;23(3):295-300. English. doi:<u>http://dx.doi.org/10.5114/fmpcr.2021.108193</u>. Cited in: Pubmed;
 PMID 2014114910.
- 34 417. Australian Commission on Safety and Quality in Health Care. CARAlert First Annual Report: March 2016-March 2017 [Internet]. Sydney (NSW): Australian Commission on Safety and Quality in 35 Oct 36 Health Care: 2017 [cited 2022 10]. Available from: https://www.safetyandguality.gov.au/sites/default/files/migrated/CARAlert-Report-March-2016-to-37 March-2017.pdf 38
- 418. Clay-Williams R, Stephens JH, Williams H, Hallahan A, Dalton C, Hibbert P, Ting HP, Arnolda G,
 Wiles L, Braithwaite J. Assessing the appropriateness of the management of otitis media in
 Australia: A population-based sample survey. J Paediatr Child Health. 2020 Feb 1;56(2):215-223.
 English. doi:<u>http://dx.doi.org/10.1111/jpc.14560</u>. Cited in: Pubmed; PMID 628803673.
- 419. Gadzhanova S, Roughead E. Prescribed antibiotic use in Australian children aged 0-12 years. Aust
 Fam Physician. 2016;45(3):134-138.
- 420. Williams MR, Greene G, Naik G, Hughes K, Butler CC, Hay AD. Antibiotic prescribing quality for
 children in primary care: an observational study. Br J Gen Pract. 2018;68(667):e90-e96.
 doi:10.3399/bjgp18X694409.
- 48 421. Bianco A, Licata F, Nobile CGA, Napolitano F, Pavia M. Pattern and appropriateness of antibiotic
 49 prescriptions for upper respiratory tract infections in primary care paediatric patients. Int J
 50 Antimicrob Agents. 2022;59(1):106469-106469. doi:10.1016/j.ijantimicag.2021.106469.
- 422. Ness RA, Bennett JG, Elliott WV, Gillion AR, Pattanaik DN. Impact of beta-Lactam Allergies on
 Antimicrobial Selection in an Outpatient Setting. South Med J. 2019 Nov 1;112(11):591-597.
 English. doi:<u>http://dx.doi.org/10.14423/SMJ.00000000001037</u>. Cited in: Pubmed; PMID
 630486356.
- 423. Chew C, Shih V, Han Z. Evaluation of antibiotic appropriateness at an outpatient oncology centre.
 J Oncol Pharm Pract. 2022;0(0):10781552221087604. doi:10.1177/10781552221087604. Cited in:
 Pubmed; PMID 35306916.

- Ketcham JD, Baker LC, MacIsaac D. Physician practice size and variations in treatments and outcomes: evidence from Medicare patients with AMI. Health Aff (Millwood). 2007 Jan;26(1):195-205. English. doi:10.1377/hlthaff.26.1.195. Cited in: Pubmed; PMID 17211029.
- 4 425. Mousques J, Renaud T, Scemama O. Is the "practice style" hypothesis relevant for general practitioners? An analysis of antibiotics prescription for acute rhinopharyngitis. Soc Sci Med. 2010
 6 Apr;70(8):1176-84. English. doi:<u>https://dx.doi.org/10.1016/j.socscimed.2009.12.016</u>. Cited in:
 7 Pubmed; PMID 20137844.
- 426. Gunnarsson R, Orda U, Elliott B, Heal C, Del Mar C. What is the optimal strategy for managing primary care patients with an uncomplicated acute sore throat? Comparing the consequences of nine different strategies using a compilation of previous studies. BMJ Open. 2022 Apr 29;12(4) (no pagination). English. doi:<u>https://dx.doi.org/10.1136/bmjopen-2021-059069</u>. Cited in: Pubmed; PMID 638040680.
- 427. Tran J, Danchin M, A CS, Pirotta M. Management of sore throat in primary care. Aust J Gen Pract.
 2018 Jul;47(7):485-489. English. doi:10.31128/ajgp-11-17-4393. Cited in: Pubmed; PMID 30114874.
- 428. Australian Commission on Safety and Quality in Health Care. Antimicrobial medicines dispensing
 from 2013–14 to 2017–18 [Internet]. Sydney (NSW): Australian Commission on Safety and Quality
 in Health Care; 2019 [cited 2022 May 5]. Available from: https://www.safetyandquality.gov.au/our-
 work/healthcare-variation/antimicrobial-medicines-dispensing-2013-14-2017-18
- 429. Teoh L, Stewart K, Marino RJ, McCullough MJ. Part 1. Current prescribing trends of antibiotics by
 dentists in Australia from 2013 to 2016. Aust Dent J. 2018 May 13. English. Epub 20180513.
 doi:10.1111/adj.12622. Cited in: Pubmed; PMID 29754452.
- 430. Peng Z, Hayen A, Hall J, Liu B. Microbiology testing and antibiotic treatment for urinary tract infections in general practice: a nationwide observational study. Infection. 2021 April;49(2):249-255. English. doi:<u>http://dx.doi.org/10.1007/s15010-020-01512-6</u>. Cited in: Pubmed; PMID 2006010964.
- 431. Antibiotic Expert Groups. Therapeutic guidelines: antibiotic. Version 14. Melbourne (VIC):
 Therapeutic Guidelines Limited.; 2010. <u>http://www.tg.org.au/</u>
- 432. Spicer JO, Dukes AW, O'Neill KE, Herrera R, Hicks LA. Healthcare professionals' knowledge, attitudes, and beliefs regarding factors tat contribute to inappropriate antibiotic use [Conference Abstract]. Open Forum Infect Dis. 2018 Nov;5(Supplement 1):S541. English. doi:<u>http://dx.doi.org/10.1093/ofid/ofy210.1546</u>. Cited in: Pubmed; PMID 629386658.
- 433. Coombs G, Daley D, Lee Y, Pang S. Australian Group on Antimicrobial Resistance (AGAR)
 Australian Enterococcal Sepsis Outcome Programme (AESOP) Annual Report 2017. Commun Dis
 Intell (2018). 2019 Sep 16;43. English. Epub 20190916. doi:10.33321/cdi.2019.43.42. Cited in:
 Pubmed; PMID 31522664.
- 434. White R. *Escherichia coli*: placing resistance to third-generation cephalosporins and
 fluoroquinolones in Australia and New Zealand into perspective. Microbiology Australia.
 2021;42(3):104-110. doi:<u>https://doi.org/10.1071/MA21031</u>.
- 40 435. Coombs G, Daley D, Nimmo G CP, JM B, and fA, K D, for Australian Commission on Safety and 41 Quality in Health Care. MRSA in Australia: MRSA bacteraemia – 2013 to 2018. [Internet]. Sydney: 42 ACSQHC; 2020 [cited 2022 Nov Available from: 2]. https://www.safetyandguality.gov.au/sites/default/files/2020-09/methicillin-43 resistant staphylococcus aureus in australia mrsa bacteraemia 2013 to 2018.pdf 44
- 436. Bell JM, Gottlieb T, Daley DA, Coombs GW. Australian Group on Antimicrobial Resistance (AGAR)
 Australian Gram-negative Sepsis Outcome Programme (GNSOP) Annual Report 2016. Commun
 Dis Intell (2018). 2018;42. English. Epub 2018 Dec 17. Cited in: Pubmed; PMID 30626303.
- 437. Palin V, Molter A, Belmonte M, Ashcroft DM, White A, Welfare W, van Staa T. Antibiotic prescribing
 for common infections in UK general practice: variability and drivers. J Antimicrob Chemother. 2019
 Aug 1;74(8):2440-2450. English. doi:<u>https://dx.doi.org/10.1093/jac/dkz163</u>. Cited in: Pubmed;
 PMID 2018241892.
- 52 Swe MMM, Ashley EA, Althaus T, Lubell Y, Smithuis F, McLean ARD. Inter-prescriber variability in 438. 53 the decision to prescribe antibiotics to febrile patients attending primary care in Myanmar. JAC-(no 54 Antimicrobial Resistance. 2021 Mar 1:3(1)pagination). English. doi:https://dx.doi.org/10.1093/jacamr/dlaa118. Cited in: Pubmed; PMID 2010976744. 55

- 439. Lucas M, Loh RK, Smith WB. Improving drug allergy management in Australia: education,
 communication and accurate information. Med J Aust. 2019;210(2):62-64.
 doi:<u>https://doi.org/10.5694/mja18.00467</u>.
- 440. Blumenthal KG, Shenoy ES, Varughese CA, Hurwitz S, Hooper DC, Banerji A. Impact of a clinical guideline for prescribing antibiotics to inpatients reporting penicillin or cephalosporin allergy. Ann Allerg Asthma Immunol. 2015;115(4):294-300. e2.
- 441. Lucas M, Vale, S. Drug-induced anaphylaxis in Australia: we need a national drug allergy registry.
 Med J Aust. 2022. doi:10.5694/mja2.51527.
- 9 442. Drewett GP, Encena J, Gregory J, Franklin L, Trubiano JA. Anaphylaxis in Victoria: presentations
 10 to emergency departments, with a focus on drug- and antimicrobial-related cases. Med J Aust.
 11 2022. doi:10.5694/mja2.51459.
- 443. Del Mar C, Hoffmann T, Bakhit M. How can general practitioners reduce antibiotic prescribing in
 collaboration with their patients? Aust J Gen Practitioners. 2022 Jan 19;51:25-30.
- 444. Zeng Y, Shi L, Liu C, Li W, Li J, Yang S, Yang X, Huang Q, Yang L. Effects of social norm feedback
 on antibiotic prescribing and its characteristics in behaviour change techniques: a mixed-methods
 systematic review. Lancet Infect Dis. doi:10.1016/S1473-3099(22)00720-4.
- Slekovec C, Leroy J, Vernaz-Hegi N, Faller J-P, Sekri D, Hoen B, Talon D, Bertrand X. Impact of a
 region wide antimicrobial stewardship guideline on urinary tract infection prescription patterns. Int
 J Clin Pharm Pharmaceut Care. 2012;34(2):325-329. doi:10.1007/s11096-012-9606-6.
- 446. Beilby J, Marley J, Walker D, Chamberlain N, Burke M. Effect of Changes in Antibiotic Prescribing
 on Patient Outcomes in a Community Setting: A Natural Experiment in Australia. Clin Infect Dis.
 2001;34(1):55-64. doi:10.1086/338232.
- 447. Linder JA. Breaking the Ambulatory Antibiotic Prescribing Cycle with All-Antibiotic Stewardship,
 Patient Stewardship, and Visit Stewardship. Clin Infect Dis. 2021 Oct 5;73(7):e1680-e1683.
 English. doi:10.1093/cid/ciaa1170. Cited in: Pubmed; PMID 32776131.
- 448. Thursky KA, Hardefeldt LY, Rajkhowa A, Ierano C, Bishop J, Hawes L, Biezen R, Saha SK, Dowson L, Bailey KE, Scarborough R, Little SB, Gotterson F, Hur B, Khanina A, Urbancic K, Crabb HK, Richards S, Sri A, James R, Kong DCM, Marshall C, Mazza D, Peel T, Stuart RL, Manski-Nankervis JA, Friedman ND, Bennett N, Schulz T, Billman-Jacobe H, Buono E, Worth L, Bull A, Richards M, Ayton D, Gilkerson JR, Browning GF, Buising KL. Antimicrobial stewardship in Australia: the role of qualitative research in programme development. JAC Antimicrob Resist. 2021 Dec;3(4):dlab166.
 English. Epub 20211118. doi:10.1093/jacamr/dlab166. Cited in: Pubmed; PMID 34806005.
- 449. Mundkur ML, Franklin J, Huybrechts KF, Fischer MA, Kesselheim AS, Linder JA, Landon J, Patorno
 E. Changes in Outpatient Use of Antibiotics by Adults in the United States, 2006-2015. Drug Saf.
 2018 Dec;41(12):1333-1342. English. doi:10.1007/s40264-018-0697-4. Cited in: Pubmed; PMID
 29987757.
- 450. Lown M, McKeown S, Stuart B, Francis N, Santer M, Lewith G, Su F, Moore M, Little P. Prescribing
 of long-term antibiotics to adolescents in primary care: a retrospective cohort study. Br J Gen Pract.
 2021 Dec;71(713):e887-e894. English. Epub 20211125. doi:10.3399/bjgp.2021.0332. Cited in:
 Pubmed; PMID 34607798.
- 451. Australian Institute of Health and Welfare. Closing the Gap targets: 2017 analysis of progress and key drivers of change. [Internet]. Canberra (ACT): Australian Institute of Health and Welfare.; 2018
 43 [cited 2022 Nov 6]. Available from: <u>https://www.aihw.gov.au/getmedia/a352bfd2-0af8-4157-9f45-</u>
 44 <u>9b066360a7d4/aihw-cvd-87.pdf.aspx?inline=true</u>
- 45 452. Australian Institute of Health and Welfare. Closing the Gap Report 2019 [Internet]. Canberra (ACT):
 46 Australian Institute of Health and Welfare.; 2019 [cited 2022 Nov 5]. Available from:
 47 <u>https://www.aihw.gov.au/getmedia/a352bfd2-0af8-4157-9f45-9b066360a7d4/aihw-cvd-</u>
 48 87.pdf.aspx?inline=true
- 49 453. Murray R. Prescribing issues for Aboriginal people. Aust Prescriber. 2003;26(5):106-109.
 50 doi:10.18773/austprescr.2003.080.
- 454. Willis Z, Walters E. Specificity of diagnosis codes and adequacy of supportive documentation for
 common acute pediatric infections: Implications for ambulatory stewardship [Conference Abstract].
 Open Forum Infect Dis. 2019 Oct;6(Supplement 2):S399. English.
 doi:http://dx.doi.org/10.1093/ofid/ofz360.987.
- 455. National Institute for Health and Care Excellence. Urinary Tract Infection (Recurrent): Antimicrobial
 Prescribing. [Internet]. London NICE; 2018 [cited 2022 Aug 10]. Available from: https://www.nice.org.uk/guidance/ng112.

- 456. Anger J, Lee U, Ackerman AL, Chou R, Chughtai B, Clemens JQ, Hickling D, Kapoor A, Kenton
 KS, Kaufman MR, Rondanina MA, Stapleton A, Stothers L, Chai TC. Recurrent Uncomplicated
 Urinary Tract Infections in Women: AUA/CUA/SUFU Guideline. J Urol. 2019;202(2):282-289.
 doi:doi:10.1097/JU.0000000000296.
- 457. Bell A, Fairbrother M, Jones K. Fixed and random effects models: making an informed choice. Qual
 Quant. 2019 Mar 1;53(2):1051-1074. doi:10.1007/s11135-018-0802-x.

- č

1	
2	
3	
4	
5	
6 7	APPENDICES
8	ТО
9	THESIS
10	
11	
12	Predictors of Inappropriate Antibiotic Prescribing in Australian
13	General Practice / Primary Care Settings
14	
15	
16	
17	
18	
19	AMY ELIZABETH HARRISON
20	
21	ORCID 0000-0002-7650-8885
22	
23	

APPENDICES TABLE OF CONTENTS

AP	APPENDICES			
AP	APPENDICES TABLE OF CONTENTS			
TABLE OF FIGURES FOR THE APPENDICES2				
LIST OF TABLES FOR THE APPENDICES				
AP	PENDIX A	– APPENDICES FOR THE BACKGROUND CHAPTER	246	
A.1	Previous	and subsequent guidelines for relevant conditions	246	
	A.1.1	Upper respiratory tract infection	246	
	A.1.2	Urinary tract infection	249	
A.2	Prominen	t antibiotic prescribing quality indicators	250	
	A.2.1	The European Surveillance Of Antimicrobial Consumption Network	250	
	A.2.2	Quality indicators by Hansen et al. (266)	253	
	A.2.3	Quality indicators by Le Maréchal et al. (466)	255	
	A.2.4	The World Health Organization's Access, Watch and Reserve progr 2019	am, 257	
	A.2.5	Quality indicators by Thilly et al. (262)	257	
A.3	Overview	of the literature	259	
	A.3.1	Literature considerations	259	
	A.3.2	Revised literature search strategy	261	
AP	PENDIX B	– APPENDICES TO THE METHODS CHAPTER	263	
B.1	Summary	y of the data received	263	
B.2	Diagnose	9S	263	
	B.2.1	Upper respiratory tract infection	265	
	B.2.2	Urinary tract infection	272	
B.3	Antibiotic	prescriptions	275	
B.4	Patient va	ariables	285	
	B.4.1	Patient age groups	285	
	B.4.2	Patient gender	286	
	B.4.3	Patient allergy labels	286	
	B.4.4	Acute rheumatic fever	286	
	B.4.5	Primary health network	287	
	B.4.6	Patient indigenous status	287	
	B.4.7	Patient smoking status	287	
	B.4.8	Patient socio-economic disadvantage	288	
	B.4.9	Patient rurality	288	

	B.4.10	Cardiovascular disease	289
	B.4.11	Mental health condition	292
B.5	Consultati	ion-related variables	292
	B.5.1	Patients with multiple, independent episodes of urinary tract infection	292
	B.5.2	Temperature-related variables	293
	B.5.3	Urine dipsticks for urinary tract infection testing	293
B.6	Predictor	s analyses	294
	B.6.1	Explanatory data analysis	294
	B.6.2	Linking records	295
	B.6.3	Standard of assessment	295
	B.6.4	Base model inclusion	296
	B.6.5	Modeling considerations	298
B.7	Trends a	nalyses	299
AP	PENDIX C	- APPENDICES TO THE PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR UPPER RESPIRATORY TRACT INFECTION CHAPTER (CHAPTER 4)	302
C.1	Clustering infection	of patients and providers for initial presentations of upper respiratory t 302	tract
C.2	Antibiotic	prescriptions	303
C.3	Antibiotic	prescriptions for influenza / influenza-like illness	307
C.4	Repeats is	ssued on antibiotic prescription status	308
	C.4.1	Repeats issued on all antibiotic prescriptions for upper respiratory tra infection	ct 308
	C.4.2	Repeats issued on cefalexin prescriptions for treatment of acute pharyngitis / tonsillitis	310
C.5	Marginal e	effects for the inappropriate decision model for initial presentations of u y tract infection	pper 316
	C.5.1	Average marginal effects	316
	C.5.2	Margins at representative values for effect on the probability of the inappropriate decision with change in patient gender, across different patient age groups	317
	C.5.3	Margins at representative values for the effect on inappropriate decis with change in patient age groups, across different values of patient gender	ion 318
	C.5.4	Adjusted predictions for the effect on the inappropriate decision mode specific values of patient age group and patient gender	el at 319
C.6	Marginal e	effects for the unnecessary antibiotic prescribing model	323
	C.6.1	Average marginal effects	323

	C.6.2	Margins at representative values for effect on unnecessary prescribing with change in upper respiratory tract infection condition, across different levels of ordinal choice of antibiotic prescribed
	C.6.3	Margins at representative values for the effect on unnecessary prescribing with change in ordinal choice of antibiotic prescribed, across different upper respiratory tract infection conditions
	C.6.4	Predictive margins for the effect on unnecessary prescribing, at specific values of patient age group and upper respiratory tract infection conditions
	C.6.5	Adjusted predictions for the effect on unnecessary prescribing, at specific values of upper respiratory trat infection condition and ordinal choice of antibiotic prescribed
C.7	Marginal e	effects for the ordinal choice of antibiotic prescribing model for upper 331
	C.7.1	Average marginal effects
	C.7.2	Margins at representative values for effect on ordinal choice of antibiotic, with change in upper respiratory tract infection condition, across different values of patient age group
	C.7.3	Margins at representative values for the effect on ordinal choice of antibiotic prescribed, with change in repeat prescription status, across different values of patient age group
	C.7.4	Adjusted predictions for the effect on ordinal choice of antibiotic, at specific values of patient age group and upper respiratory tract infection condition
	C.7.5	Predictive margins for the effect on ordinal choice, at specific values of whether repeats were issued on prescription and patient age group340
	C.7.6	Predictive margins for the effect on ordinal choice of antibiotic prescribed, at specific values of upper respiratory tract infection condition and patient age group, graphed by upper respiratory tract infection condition
C.8	Non-first-l	ine antibiotic prescribing model for upper respiratory tract infection344
	C.8.1	Average marginal effects
C.9	Repeats b	being issued on prescriptions model for upper respiratory tract infection345
	C.9.1	Model
	C.9.2	Model explanation347
	C.9.2.1	Summary
	C.9.3	Marginal effects for the repeats being issued on prescriptions model for upper respiratory tract infection
	C.9.3.1	Average marginal effects
	C.9.3.2	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in upper respiratory tract infection condition, across different values of patient age group

	C.9.3.2	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of upper respiratory tract infection condition,351
	C.9.3.2	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of ordinal choice of antibiotic prescribed
	C.9.3.2	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in ordinal choice of antibiotic prescribed, across patient age groups
C.10	Comparis	son with fixed-effects models355
AP	PENDIX D	– APPENDICES TO THE PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR URINARY TRACT INFECTION CHAPTER (CHAPTER 5)
D.1	Clustering	of patients and providers for initial presentations of urinary tract infection
D.2	Antibiotic	prescriptions
D.3	Ordinal ch	noice of antibiotic prescribed for urinary tract infection
D.4	Whether	repeats were issued on antibiotic prescriptions
	D.4.1	Repeats issued on all antibiotic prescriptions for urinary tract infection.362
	D.4.2	Analysis of cefalexin prescriptions issued with repeats, for initial presentations of urinary tract infection
D.5	Marginal e	effects for the ordinal choice of antibiotic prescribing model for urinary tract
	D.5.1	Average marginal effects
	D.5.2	Margins at representative values for the effect on ordinal choice of antibiotic, with change in patient age group, across different values of whether repeats were issued on the prescription
	D.5.3	Margins at representative values for the effect on ordinal choice of antibiotic, with change in patient age group, across different values of patient gender
	D.5.4	Margins at representative values for then effect on ordinal choice of antibiotic, with change in whether repeats were issued on the prescriptions, across different values of patient age group
	D.5.5	Adjusted predictions for the effect on ordinal choice of antibiotic, at specific values of patient gender, patient age group and whether repeats were issued on the prescription
	D.5.6	Predictive margins for the effect on ordinal choice of antibiotic outcomes, across different values of patient age group and patient gender
	D.5.7	Predictive margins for effect on the ordinal choice of antibiotic, across different values of patient age group and whether repeats were issued on the prescription

D.6	Marginal e	effects for the model for repeats being issued on antibiotic prescriptions for sentations of urinary tract infection
	D.6.1	Average marginal effects
	D.6.2	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient gender, across different values of patient age group
	D.6.3	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of patient gender
	D.6.4	Predictive margins for the probability of repeats being issued on antibiotic prescriptions, across different values of patient age group and patient gender
	D.6.5	Adjusted predictions for the effect on the probability of repeats being issued on antibiotic prescribing, at specific values of patient gender and age group
D.7	Marginal e	effects for the binary model of non-first-line antibiotic prescribing for upper y tract infection
	D.7.1	Average marginal effects
	D.7.2	Margins at representative values for the effect on non-first-line antibiotic prescribing with change in patient gender, across different values of patient age group
	D.7.3	Margins at representative values for the effect on non-first-line antibiotic prescribing occurring with change in patient age group, across different values of patient gender
	D.7.4	Predictive margins for the effect on non-first-line antibiotic prescribing occurring at different values of patient age group and patient gender396
D.8	Comparis	on of final mixed-effects models with fixed-effects models
AP	PENDIX E	– APPENDICES TO TRENDS IN PRESCRIBING FOR UPPER RESPIRATORY TRACT INFECTION CHAPTER (CHAPTER 6)400
E.1	Tables	
	E.1.1	Upper respiratory tract infection conditions excluding influenza / influenza- like illness
	E.1.2	Influenza / influenza-like illness405
	E.1.3	Mean prescribing rates for all upper respiratory tract infection conditions including influenza / influenza-like illness
E.2	Additional	time series plots408
	E.2.1	All upper respiratory tract infection conditions excluding influenza / influenza-like illness
	E.2.1	By specific condition410
	E.2.1.1	Acute rhinosinusitis410
	E.2.1.2	Acute pharyngitis / tonsillitis411

	E.2.1.3	Acute otitis media4	13
AP	PENDIX F	– APPENDICES TO TRENDS IN PRESCRIBING FOR URINARY TRAC INFECTION CHAPTER (CHAPTER 7)4	T 14
F.1	Tables	414	
F.2	Additional	time series plots4	20
	F.2.1	All patients with initial presentations of urinary tract infection4	20
	F.2.2	Women sixteen years and over with initial presentations of urinary tract infection4	121
	F.2.3	Men sixteen years and over with initial presentations of urinary tract infection4	122
	F.2.4	Children under sixteen years of age with initial presentations of urinary tract infection4	124
AP	PENDIX G	– APPENDIX TO THE DISCUSSION (CHAPTER 8): COMPARISON WI QUALITY INDICATORS	TH 125
G.1	Comparis	on with quality indicators for upper respiratory tract infection4	25
G.2	Comparis	ons with quality indicators for urinary tract infection4	27
RE	FERENCE	S TO APPENDICES4	30

1 TABLE OF FIGURES FOR THE APPENDICES

2			
3 4	Figure A-1:	Copy of European Surveillance of Antimicrobial Consumption prescribing qualit indicators presented by Coenen et al. (259)2	y 51
5 6	Figure A-2:	Copy of European Surveillance of Antimicrobial Consumption quality indicators published by Adriaenssens <i>et al.</i> (464)	52
7	Figure A-3:	Copy of Hansen et al.'s quality indicators (266)2	54
8	Figure A-4:	Copy of quality indicators by Hansen et al (266)2	54
9	Figure A-6:	Prescribing Quality Indicators Presented by Thilly et al. (262)2	58
10	Table A-2:	Summary of literature search terms performed2	60
11	Table A-3:	Medline Mesh index terms search {including related terms}2	61
12	Table A-4:	Medline keyword search2	61
13	Table A-5:	Embase Em index terms search {including related terms}2	61
14	Table A-6:	Embase keyword search2	61
15	Table A-7:	Combining all four search results2	62
16	Table A-8:	Examples of removal of articles not meeting criteria2	62
17 18	Figure C-1:	Bubble plot of individual patients and providers at each practice for initial presentations of upper respiratory tract infection3	02
19 20 21	Figure C-2:	Margins at representative values for effect on the probability of the inappropriate decision with change in patient gender from female to male, across different patient age groups, relative to the probability for female patients	e 17
22 23 24	Figure C-3:	Margins at representative values for the effect on the probability of inappropriate decision occurring, with change in patient age group, across different values of patient gender, relative to the probability for patients 0-8 years of age	е 18
25 26 27	Figure C-4:	Adjusted predictions for effect on the probability of the inappropriate decision model at specific values of patient age group and patient gender, with all other covariates kept constant at sample means	22
28 29 30 31	Figure C-5:	Margins at representative values for effect on unnecessary prescribing occurrin with change in upper respiratory tract infection condition, across different levels ordinal choice of antibiotic prescribed, relative to the effect for unnecessary prescribing for acute rhinosinusitis	g of 24
32 33 34 35	Figure C-6:	Margins at representative values for the effect on the probability of unnecessary prescribing occurring, with change in ordinal choice of antibiotic prescribed, across different upper respiratory tract infection conditions, relative to the effect on the probability of unnecessary prescribing for first-line prescriptions	/ 25
36 37 38	Figure C-7:	Predictive margins for the effect on the probability of unnecessary prescribing with specific values of patient age group and upper respiratory tract infection condition, graphed by upper respiratory tract infection condition	27
39 40 41 42	Figure C-8:	Margins at representative values for the effect on the probability of each of the three outcomes of ordinal choice of antibiotic occurring, with change upper respiratory tract infection condition, across different values of patient age group relative to the effect on the probability for acute rhinosinusitis	, 34

1 2 3 4	Figure C-9:	Margins at representative values for the effect on the probability of each of the three outcomes of ordinal choice occurring, with change in repeat prescription status, across different values of patient age group, relative to the effect on the probability for prescriptions issued without repeats
5 6 7 8	Figure C-10:	Predictive margins for the effect on the probability of each of the three outcomes of ordinal choice of antibiotic occurring, at specific values of whether repeats were issued on prescription and patient age group, graphed by repeat on prescription status
9 10 11 12	Figure C-11:	Predictive margins for the effect on the probability of each of the three outcomes of ordinal choice of antibiotic occurring, at specific values of upper respiratory tract infection condition and patient age group, graphed by upper respiratory tract infection condition
13 14 15 16	Figure C-12:	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in upper respiratory tract infection condition, across different values of patient age group, relative to the probability of repeats being issued on prescriptions for acute rhinosinusitis
17 18 19 20	Figure C-13:	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of upper respiratory tract infection condition, relative to the effect of repeats being issued on prescriptions for patients aged 0-8 years
21 22 23 24	Figure C-14:	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of ordinal choice of antibiotic prescribed, relative to the probability of repeats being issued on prescriptions for patients aged 0-8 years
25 26 27 28	Figure C-15:	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in ordinal choice of antibiotic prescribed, across patient age groups, relative to the probability of repeats being issued on first-line prescriptions
29 30	Figure D-1:	Bubble plot of clustering by inidividual patients, providers and practices for initial presentations of urinary tract infection
31 32	Figure D-2:	Bar graph of ordinal line of antibiotic prescribed for initial presentations of urinary tract infection, graphed by patient's primary health network
33 34	Figure D-3:	Bar graph of ordinal choice of antibiotic prescribed for initial presentations of urinary tract infection, by (proxy for) practice size
35 36 37	Figure D-4:	Bar graph of antibiotic prescriptions with repeats issued on them, for initial presentations of urinary tract infection, by ordinal choice of antibiotic prescribed
38 39 40	Figure D-5:	Margins at representative values for the effect on the probability of each of the four ordinal choice of antibiotic outcomes occurring, with change in patient age group, across different values of patient gender
41 42 43	Figure D-6:	Margins at representative values for the effect on the probability of the four ordinal choice of antibiotic outcomes occurring, with change in whether repeat were issued on the prescription, across different values of patient age group378
44 45 46	Figure D-7:	Predictive margins for the effect on the probability of each of the four ordinal choice of antibiotic outcomes occurring, across different values of patient age group and patient gender, graphed by patient gender

1 2 3 4	Figure D-8:	Predictive margins for the effect on the probability of each of the four ordinal choice of antibiotic outcomes occurring, across different values of patient age group and whether repeats were issued on the prescription, graphed by repeat on prescription status
5 6 7	Figure D-9:	Margins at representative values for the effect on the probability of repeats being issued on prescriptions with change in patient gender, across different values of patient age group, relative to the effect on the probability for female patients 386
8 9 10 11	Figure D-10:	Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions with change in patient age group, across different values of patient gender, relative to the effect on the probability for patients aged 45 years and over
12 13 14	Figure D-11:	Predictive margins for the probability of repeats being issued on antibiotic prescriptions. across different values of patient age group and patient gender, graphed by patient gender
15 16 17	Figure D-12:	Predictive margins for the probability of repeats being issued on antibiotic prescriptions, across different values of patient age group and patient gender, graphed by patient age group
18 19 20	Figure D-13:	Adjusted predictions for the effect on the probability of repeats being issued on antibiotic prescriptions, at specific values of patient gender and age group, with all other covariates kept constant at sample means
21 22 23	Figure D-14:	Margins at representative values for the effect on the probability of non-first-line antibiotic prescribing occurring with change in patient gender, across different values of patient age group, relative to the effect for female patients
24 25 26 27	Figure D-15:	Margins at representative values for the effect on the probability of non-first-line antibiotic prescribing occurring with change in patient age group, across different values of patient gender, relative to the effect on the probability for patients aged 45 years and over
28 29 30	Figure D-16:	Predictive margins for the effect on the probability of non-first-line antibiotic prescribing occurring at different values of patient age group and patient gender, graphed by patient age group
31 32	Figure E-1:	Time Series Plot of Unnecessary Antibiotic Prescribing rates for all URTI diagnoses, Jan 2012-Jun 2017 inclusive, by month
33 34 35	Figure E-2:	Time series plot of antibiotic prescribing rates for antibiotics for all initial presentations of upper respiratory tract infection, Jan 2012-Jun 2017 inclusive, by month
36 37 38	Figure E-3:	Time series plot of second-line antibiotic prescribing rates for all initial presentations of upper respiratory tract infection, Jan 2012-Jun 2017 inclusive, by month
39 40 41	Figure E-4:	Time series plot of antibiotic prescribing rates for antibiotics not recommended in the guidelines for all initial presentations of upper respiratory tract infection, Jan 2012-Jun 2017 inclusive, by month
42 43	Figure E-5:	Time series plot of antibiotic prescribing rates for initial presentations of acute rhinosinusitis, January 2012 to June 2017, inclusive, by month
44 45 46	Figure E-6:	Time series plot of non-first-line antibiotic prescribing rates for initial presentations of acute rhinosinusitis, January 2012 to June 2017, inclusive, by month

1 2	Figure E-7:	Time series plot of second-line antibiotic prescribing rates for initial presentations of acute rhinosinusitis, January 2012 to June 2017, inclusive, by month
3 4 5	Figure E-8:	Time series plot of unnecessary antibiotic prescribing rates for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017, inclusive, by month
6 7	Figure E-9:	Time series plot of antibiotic prescribing rates for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017, inclusive, by month
8 9 10	Figure E-10:	Time series plot of amoxicillin and phenoxymethylpenicillin prescribing rates for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017, inclusive, by month
11 12 13	Figure E-11:	Time series plot of non-first-line antibiotic prescribing rates for initial presentations of acute otitis media, January 2012 to June 2017, inclusive, by month
14 15	Figure E-12:	Time series plot of second-line antibiotic prescribing rates for initial presentations of acute otitis media, January 2012 to June 2017, inclusive, by month
16 17	Figure F-1:	Time series plot of second-line antibiotic prescribing for all patients with initial presentations of urinary tract infection over time, by month
18 19	Figure F-2:	Time series plot of antibiotic prescribing for all patients with initial presentations of urinary tract infection over time, by month
20 21	Figure F-3:	Time series plot of second-line antibiotic prescribing for women of sixteen years and over with initial presentation of urinary tract infection over time, by month 421
22 23 24	Figure F-4:	Time series plot of second-line antibiotic prescribing for women of sixteen years and over with initial presentations of urinary tract infection over time, by month
25 26 27	Figure F-5:	Time series plot of prescribing for antibiotic agents not recommended in the guidelines for women of sixteen years and over with initial presentations of urinary tract infection over time, by month
28 29	Figure F-6:	Time series plot of non-first-line antibiotic prescribing for all men of sixteen years and over with initial presentation of urinary tract infection over time, by month 422
30 31 32	Figure F-7:	Time series plot of second-line antibiotic prescribing for all men of sixteen years and over with initial presentations of urinary tract infection over time, by month
33 34 35	Figure F-8:	Time series plot of third-line (and last resort) prescribing for all men of sixteen years and over with initial presentations of urinary tract infection over time, by month
36 37	Figure F-9:	Time series plot of non-first-line antibiotic prescribing for children under sixteen years over time, by month
38 39	Figure F-10:	Time series plot of second-line antibiotic prescribing for children under sixteen years of age over time, by month
40 41 42	Figure G-1:	Comparison of the antibiotic prescribing for upper respiratory tract infection conditions in this dataset with the World Health Organization's 2019 Access, Watch and Reserve classifications (467)
43 44 45	Figure G-2:	Comparison of the antibiotic prescribing for urinary tract infection in this dataset with the World Health Organization's 2019 Access, Watch and Reserve classifications (467)

1 LIST OF TABLES FOR THE APPENDICES

2 3	Table A-1:	Summary of Therapeutic Guideline: Antibiotic recommendations in numeric order for empirical prescribing for initial presentations of acute cystitis (86)
4	Table B-1:	Summary of digital files received
5	Table B-2:	List of final medicine active ingredients282
6	Table B-3:	List of antibiotic classes
7	Table B-4:	Base model inclusion for each model for upper respiratory tract infection 296
8	Table B-5:	Base model inclusion for each model for urinary tract infection
9 10	Table B-6:	Details of numerators and denominators for outcome rates for trends analysis for upper respiratory tract infection
11 12	Table B-7:	Details of numerators and denominators for outcome rates for trends analysis for urinary tract infection
13 14 15	Table C-1:	Frequency table of antibiotic class by Anatomical Therapeutic Chemical classification (352), for systemic antibiotics prescribed for initial presentations of upper respiratory tract infection
16 17	Table C-2:	Frequency table of patient age group by likely appropriate / likely inappropriate decision status for initial presentations of upper respiratory tract infection 304
18 19 20	Table C-3:	Frequency table of medicine active ingredients for systemic antibiotics prescribed but not recommended in the guidelines for the condition it was prescribed for, for initial presentations of upper respiratory tract infection
21 22 23	Table C-4:	Frequency table of upper respiratory tract infection condition by likely necessary / likely unnecessary prescribing status, for initial presentations of upper respiratory tract infection
24 25 26	Table C-5:	Frequency table of ordinal choice of antibiotic prescribed by upper respiratory tract infection condition, for initial presentations of upper respiratory tract infection
27 28 29	Table C-6:	Frequency table of patient primary health network by ordinal choice of antibiotic prescribed, for patients with initial presentations of acute pharyngitis / tonsillitis
30 31	Table C-7:	Frequency table of antibiotic active ingredients prescribed for initial presentations of influenza / influenza-like illness
32 33 34	Table C-8:	Frequency table of whether repeats were issued on prescription by likely unnecessary / necessary prescribing status, for initial presentations of upper respiratory tract infection
35 36 37	Table C-9:	Frequency table of whether repeats were issued on prescription by ordinal choice of antibiotic prescribed, for initial presentations of upper respiratory tract infection
38 39	Table C-10:	Frequency table of patient age group and whether repeats were issued on prescription, for initial presentations of upper respiratory tract infection
40 41	Table C-11:	Frequency table of prescriptions issued with repeats for initial presentations of influenza / influenza-like illness, by active ingredient

1 2 3	Table C-12:	Frequency table of medicine quantity and medicine strength, of cefalexin prescriptions with repeats present, for adults with initial presentations of pharyngitis
4 5 6 7	Table C-13:	Frequency table of medicine quantity and medicine strength, of cefalexin prescriptions with repeats present, for patients under eighteen years receiving cefalexin prescriptions with repeats for initial presentations of pharyngitis / tonsillitis
8 9 10	Table C-14:	Frequency table of medicine quantity and medicine strength, of cefalexin prescriptions with repeats present, for children under eight years of age for initial presentations of pharyngitis / tonsillitis
11 12 13	Table C-15:	Frequency table of medicine quantity and medicine strength, of cefalexin prescriptions with repeats present, for children aged 9-16 years, for initial presentations of pharyngitis / tonsillitis
14 15 16	Table C-16:	Frequency table of medicine instructions, on cefalexin prescriptions of 500mg strength with repeats present, prescribed to adults with initial presentations of pharyngitis / tonsillitis
17 18 19	Table C-17:	Frequency table of medicine instructions, on cefalexin prescriptions of 500mg strength with repeats present, prescribed to children for initial presentations of pharyngitis / tonsillitis
20 21 22	Table C-18:	Frequency table of medicine instructions, on cefalexin prescriptions of 250mg strength with repeats present, prescribed to children with initial presentations of pharyngitis / tonsillitis
23 24 25	Table C-19:	Frequency table of medicine instructions on prescriptions for liquid formulations of cefalexin with repeats present, prescribed children with initial presentations of pharyngitis / tonsillitis
26 27 28 29	Table C-20:	Average marginal effects for Model 0 – Inappropriate decisions for initial presentations of upper respiratory tract infection (unnecessary prescriptions versus reference of appropriate prescriptions together with appropriate non-prescribing situations)
30 31	Table C-21:	Average marginal effects for Model 1 – Unnecessary antibiotic prescribing for initial presentations of upper respiratory tract infection
32 33	Table C-22:	Average marginal effects for Model 2 – Ordinal choice of antibiotic prescribed for initial presentations of upper respiratory tract infection
34 35	Table C-23:	Average marginal effects for the binary model for non-first-line antibiotic prescribing for upper respiratory tract infection
36 37	Table C-24:	Mixed effects logit model for repeats being issued on antibiotic prescriptions for initial presentations of upper respiratory tract infection (Model 4)
38 39	Table C-25:	Average marginal effects for Model 4 – Repeats being issued on antibiotic prescriptions
40	Table C-26:	Three fixed-effects only comparisons for the inappropriate decision model 355
41 42	Table D-1:	Frequency table of antibiotic prescriptions for initial presentations of urinary tract infection, by Anatomical Therapeutic Chemical classification
43 44	Table D-2:	Frequency table of antibiotic active ingredients prescribed but not recommended in the guidelines for initial presentations of urinary tract infection

1 2 3	Table D-3:	Frequency table of ordinal choice of antibiotic prescribed, and primary health network, for antibiotic prescriptions for initial presentations of urinary tract infection
4 5 6	Table D-4:	Frequency table of first-line and non-first-line antibiotic prescriptions, for initial presentations of urinary tract infection, by patient age group, including a ratio of non-first-line to first-line prescriptions
7 8	Table D-5:	Frequency table of ordinal choice of antibiotic prescribed, for initial presentations of urinary tract infection, by patient age group
9 10	Table D-6:	Frequency table of patient group by ordinal choice of antibiotic prescribed, for initial presentations of urinary tract infection
11 12	Table D-7:	Frequency table of ordinal choice of antibiotic prescribed by patient primary health network, for patients with initial presentations of urinary tract infection 360
13 14	Table D-8:	Frequency table of ordinal choice of antibiotic prescribed by patient primary health network, for patients with initial presentations of urinary tract infection 360
15 16	Table D-9:	Frequency table for ordinal choice of antibiotic prescribed by patient's primary health network, for patients with initial presentations of urinary tract infection 362
17 18 19	Table D-10:	Frequency table of whether repeats were issued on antibiotic prescriptions for initial presentations of urinary tract infection, by Anatomical Therapeutic Chemical class
20 21	Table D-11:	Frequency table of whether repeats were issued on prescriptions, by ordinal choice of antibiotic prescribed, for initial presentations of urinary tract infection363
22 23	Table D-12:	Frequency table of patient age group by whether repeats were issued on the prescription, for initial presentations of urinary tract infection
24 25	Table D-13:	Frequency table of medicine quantity and medicine strength, for male adults prescribed repeats for cephalexin with urinary tract infection
26 27	Table D-14:	Frequency table of medicine quantity and medicine strength, for female adults prescribed repeats for cephalexin with urinary tract infection
28 29 30	Table D-15:	Frequency table of medicine quantity and medicine strength, for children under eighteen years issued repeats on cephalexin prescriptions for initial presentations of urinary tract infection
31 32 33	Table D-16:	Frequency table of medicine quantity and medicine strength, for 16-17 year olds issued repeats on cefalexin prescriptions for initial presentations of urinary tract infection
34 35 36	Table D-17:	Frequency table of medicine quantity and medicine strength, for children under sixteen years issued repeats for cefalexin for initial presentations of urinary tract infection
37 38 39	Table D-18:	Frequency table of medicine instructions for adult females receiving cefalexin prescriptions with repeats issued on them for initial presentations of urinary tract infection
40 41 42	Table D-19:	Frequency table of medicine instructions for adult males receiving cefalexin prescriptions with repeats issued on them, for initial presentations of urinary tract infection
43 44 45	Table D-20:	Frequency table of medicine instructions for children under sixteen years of age receiving cefalexin prescriptions with repeats issued on them, for initial presentations of urinary tract infection

1 2	Table D-21:	Average marginal effects for ordinal choice of antibiotic prescribed for initial presentations of urinary tract infection (Model 1)
3 4	Table D-22:	Average marginal effects for repeat positive antibiotic prescribing for initial presentations of urinary tract infection (Model 3)
5 6	Table D-23:	Average marginal effects for non-first-line antibiotic prescribing for initial presentations of urinary tract infection (Model 2)
7 8	Table D-24:	Summary of fixed-effects models compared against mixed model for choice of antibiotic prescribed for initial presentations of urinary tract infection
9 10	Table E-1:	Linear chi-squared tests for trend using linear regression for upper respiratory tract infection conditions
11 12	Table E-2:	Linear chi-squared tests for trend using linear regression for acute rhinosinusitis
13 14	Table E-3:	Linear chi-squared tests for trend using linear regression for acute pharyngitis / tonsillitis
15 16	Table E-4:	Linear chi-squared tests for trend using linear regression for acute otitis media
17 18	Table E-5:	Antibiotic prescribing rate and prescribing rates for individual antibiotic agents prescribed for influenza / influenza-like illness
19 20	Table E-6:	Linear chi-squared tests for trend using linear regression for influenza / influenza- like illness
21 22	Table E-7:	Linear chi-squared tests for trend using linear regression for influenza / influenza- like illness
23 24	Table E-8:	Mean prescribing rates of individual antibiotics, by upper respiratory tract infection condition, January 2012 to June 2017 inclusive
25 26	Table F-1:	Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for patients with urinary tract infection
27 28	Table F-2:	Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for women with initial presentations of urinary tract infection
29 30	Table F-3:	Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for men with initial presentations of urinary tract infection
31 32	Table F-4:	Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for children with initial presentations of urinary tract infection
33 34 35	Table F-5:	Frequency table of mean prescribing rates for individual antibiotics prescribed for initial presentations of urinary tract infection, January 2012 to June 2017 inclusive, by patient group
36 37	Table F-6:	Frequency table of counts of antibiotic prescriptions for initial presentations of urinary tract infection per half-year, by patient group
38 39	Table F-7:	Frequency table of antibiotic prescribing rates per half-yearly period, by patient group
40 41 42	Table G-1:	Compilation of indicators including the European Surveillance of Antibiotic Consumption indicators by Adriaenssens et al. (464) and relating to respiratory tract infection (464)

1	Table G-2:	Compilation of indicators including the European Surveillance of Antibiotic
2		Consumption indicators published by Adriaenssens et al. (464) relating to acute
3		cystitis

1	APPENDIX A – APPENDICES FOR THE BACKGROUND
2	CHAPTER
3	
4	
5	A.1 Previous and subsequent guidelines for relevant conditions
6	
7	A.1.1 Upper respiratory tract infection
8	
9	A.1.1.1 Acute rhinosinusitis / non-specific upper respiratory tract infection
10	A.1.1.1.1 Previous guidelines version 4 (2009) Therapeutic Guidelines Respiratory (1)
11	- relative to the guidelines used in analysis – Therapeutic Guidelines Antibiotic
12	version 15 (2014) (2)
13	
14	Prior to 2014, acute rhinosinusitis was listed within the respiratory guidelines published in
15	2009 (1). Within these, rhinosinusitis and rhinitis were listed together (1). For Acute
16	rhinosinusitis, symptomatic treatment is listed as the target of treatment (1). It notes that
17	viral infection constitutes the strong majority, and that bacterial complications eventuate in
18	an estimated 0.5% to 2.0% of acute viral rhinosinusitis presentations (1). In cases of severe
19	symptoms lasting in excess of five to seven days plus high fever or severe headache or
20	worsening symptoms or unilateral maxillary tenderness of the sinus (1). Amoxicillin is listed
21	as the first-line recommendation (1).
22	A 1 1 1 2 Subactuant quidalines varsian 16 (2010) Therapoutis Cuidalines Antibiotis (2)
23	- relative to the quidelines used in analysis - version 15 (2014) (2)
24 25	-1 etalive to the guidelines used in analysis - version 15 (2014) (2)
25	This version of the guidelines notes that acute viral and bacterial rhinosinusitis are unable
20	to be distinguished in the first three or four days of symptoms but that bacterial presentation
28	is rare (3). It notes that acute bacterial illness is most often self-limiting and that antibiotics
29	have little impact (3). For patients believed to likely have acute bacterial rhinosinusitis, these
30	patients should still receive only symptomatic therapy but with follow-up if symptoms
31	continue or worsen (3). The first-line antibiotic of choice in situations when antibiotics are
32	indicated is amoxicillin, followed by amoxicillin with clavulanate (3). These guidelines also
33	mentions shared decision making as an option (3).

A.1.1.2 Acute pharyngitis / tonsillitis

A.1.1.2.1 Previous guidelines version 14 (2010) Therapeutic Guidelines Antibiotic (4)
 - relative to the guidelines used in analysis - version 15 (2014) (2)

5

6 Difficulties between differentiating viral and bacterial sore throat are acknowledged in this 2010 version of the guidelines (4). Justifications for prescribing antibiotics are listed as 7 8 preventing non-suppurative complications of patients at high risk of complications, preventing suppurative complications or to lessen illness duration (4). These 2010 9 guidelines note that high-risk patients include 2-25 years in communities with high rates of 10 acute rheumatic fever incidence, including remote Aboriginal and Torres Strait Islander 11 12 communities (4). Any patient with scarlet fever or current rheumatic heart disease are also considered high-risk (4). These guidelines also note that symptomatic relief is useful for 13 14 treating suppurative complications (4). Phenoxymethylpenicillin is the first-line recommendation when antibiotics are indeed indicated, followed by benzathine penicillin IM 15 as the second-line option for non-compliant patients or for those unable to tolerate oral 16 antibiotics (4). 17

18

19

A.1.1.2.1

20

Subsequent version 16 (2019) of Therapeutic Guidelines Antibiotic (3) - relative to the guidelines used in analysis - version 15 (2014) (2)

21

In this version of the guidelines, pharyngitis/tonsillitis has been amended from its own 22 condition to within the condition of sore throat (3). It notes that viral pharyngitis/tonsillitis are 23 the most common cause and that antibiotics should not be prescribed for these (3). 24 25 Management is separated into patients at high risk of acute rheumatic fever and otherwise (3). Furthermore, it notes that streptococcal pharyngitis/tonsillitis are most often self-limiting 26 (3). These guidelines note that the strong majority of patients not at high risk of acute 27 rheumatic fever do not required antibiotics (3). Recommendations for when antibiotics are 28 indicated are separated into streptococcal and Arcanobacterium haemolyticum (3). For 29 streptococcal infections, phenoxymethylpenicillin and also amoxicillin are the antibiotics 30 31 recommended (3). For Arcanobacterium haemlyticum, azithromycin or erythromycin are the antibiotics of choice (3). 32

- 33
- 34

A.1.1.3 Acute otitis media

2

A.1.1.3.1 Previous guidelines version 14 (2010) Therapeutic Guidelines Antibiotic (4)
 - relative to the guidelines used in analysis - version 15 (2014) (2)

5

This 2010 version of the guidelines notes that either bacterial or viral AOM most often resolves by itself (4). It notes that antibiotics provide only modest benefit, and that suppurative complications are rare (4). For Aboriginal and Torres Strait Islander peoples, however, such complications are noted as being common so antibiotics are usually recommended for these patients (4). While in this 2010 version, patients hypersensitive to penicillin are recommended cefuroxime or cefaclor (4), however, cefuroxime or trimethoprim with sulfamethoxazole are the recommendations in the 2014 version used for analysis (2).

13

For children of two years of age or over, the 2010 guidelines notes that symptomatic treatment is recommended initially, with follow-up and re-evaluation in case of persisting symptoms for two days (4). It also recommends that children under six months of age be treated with antibiotics (4), however, the 2014 guidelines state that symptomatic treatment may be suitable, although that it is still appropriate to prescribe (4).

19

A.1.1.3.2 Subsequent version 16 (2019) Therapeutic Guidelines Antibiotic (3)
 - relative to the guidelines used in analysis - version 15 (2014) (2)

22

23 This version of the Guidelines for AOM highlight the fact that pain is not a sufficient reason to diagnose AOM (3). Ensuring pain is appropriately managed is important for symptomatic 24 25 treatment for most patients (3). Antibiotic treatment is appropriate for children under six months of age, systematically unwell children, bilateral infection in children under two years, 26 immunocompromised high risk children, those with otorrhea (3). There are separate 27 recommendations for Aboriginal and Torres Strait Islander children (3). Delayed prescribing 28 and shared decision making are strategies mentioned (3). Otitis media with effusion is 29 mentioned as not requiring antibiotics (3). When antibiotics are indicated, amoxicillin is the 30 first-line option and amoxicillin with clavulanate is the second-line options recommended (3). 31

- 32
- 33

1 **A**.

A.1.2 Urinary tract infection

2 3

A.1.2.1 Acute cystitis

4

- A.1.2.1.1 Previous guidelines version 14 (2010) Therapeutic Guidelines Antibiotic (4)
 relative to the guidelines used in analysis version 15 (2014) (2)
- 7

This 2010 version of the guidelines states that urine cultures should be performed before 8 9 antibiotics are prescribed for pregnant women, all men and all children, and all patients who have used antibiotics recently, in the event of treatment failure or infection recurrence (4). 10 11 However, it does not mention susceptibility/sensitivity testing. For men, this version includes treatment duration of fourteen days (4). These guidelines for children are divided into mild 12 and severe infection (4). Ciprofloxacin is also listed as drug of last resort for children with 13 mild infection as well as norfloxacin. Severe infection for children in this version includes 14 gentamicin IV plus amoxicillin/ampicillin (or cefotaxime or ceftriaxone for hypersensitivity) 15 (4). 16

- 17
- 18 19

A.1.2.1.2 Subsequent version 16 (2019) of the Therapeutic Guidelines Antibiotic (3)
 - relative to the guidelines used in analysis - version 15 (2014) (2)

20

This 2019 version of the guidelines clearly states that culture and susceptibility testing should be performed before prescribing antibiotics for pregnant women, men, patients in aged care, as well as any patients having received antibiotics recently, with recurrent cystitis or those at risk of multidrug-resistant bacteria (3). It notes that the broad-spectrum activity of amoxicillin with clavulanate is too wide for empirical treatment (3).

26

It notes that symptomatic treatment is sufficient for most women under 65 years of age (3). 27 28 It warns about nitrofurantoin use in men as therapeutic concentrations not reached in the prostate (3). Several additional options are provided before last resort options for adults of 29 30 fosfomycin, ciprofloxacin or norfloxacin are recommended, under the proviso of resistance to all of the prior options having been confirmed by culture and susceptibility testing before 31 32 use (3). This version contains more detail on diagnostic testing for children and the differentiation between this condition and acute pyelonephritis. **Table A-1** below contains 33 purely the first three recommendations for each patient group (3). 34

Table A-1:Summary of Therapeutic Guideline Antibiotic recommendations in numeric order
for empirical prescribing for initial presentations of acute cystitis (3).

Line/	Non-pregnant Women	Men	Children >= 1 month
Choice			
1	trimethoprim	trimethoprim	trimethoprim + sulfamethoxazole
1			trimethoprim
2	nitrofurantoin	nitrofurantoin	cefalexin
3	cefalexin	cefalexin	

3 4

A.2 Prominent antibiotic prescribing quality indicators

5 6

7 There are multiple, existing quality indicators designed to measure antibiotic prescribing 8 across many settings. A summary is provided below of some prominent quality indicators 9 focussing purely on antibiotic prescribing but is by no means exhaustive. Several of these 10 were developed and collated by expert consensus. Note these quality indicators do not 11 typically assess whether prescribing was in fact indicated at an individual level.

12

13

A.2.1 The European Surveillance Of Antimicrobial Consumption Network

14

The European Surveillance Antimicrobial Consumption Network, formerly European Surveillance of Antimicrobial Consumption (ESAC), developed a series of prescribing quality indicators (5-7). They give recommended ranges of percentages for which antibiotic prescribing should ideally remain within for presentations of specific conditions. They are good rule-of-thumb by which to assess prescribing, with the objective of trying to limit quantities of antibiotics prescribed for each condition.

21

22 A.2.1.1 Quality indicators by Coenen et al. (8)

23

On behalf of ESAC, in 2007 Coenen *et al.* (8) published a set of quality indicators for antibiotic prescribing in the European outpatient setting. These included quantities of antibiotic consumption for specific classes of antibiotics, percentages of these over the total of all antibiotics consumed, and a ratio of broad- to narrow-spectrum antibiotics, in addition to measures of seasonal variation in antibiotic use (8). Table 1 below provides a copy of these prescribing indicators (8).

		Resistance	e		Patient h	ealth I	benefit	Cost effect	tivenes	5	Public h	ealth p	oolicy
Label	Description	Median	N	Consensus	Median	Ν	Consensus	Median	Ν	Consensus	Median	Ν	Consensus
(1) J01_DID+	Consumption of antibacterials for systemic use (J01) expressed in DID‡	8	22	+	6.5	22	+	7	21	+	8	22	+
(2) J01A_DID (3) J01C_DID† (4) J01D_DID†	Consumption of tetracyclines (J01A) expressed in DID‡ Consumption of penicillins (J01C) expressed in DID‡ Consumption of cephalosporins (J01D) expressed in DID‡	6 7 7	22 22 22	+ + +	5 6 6	22 22 22	+ + +	5 6 6	21 21 21	+ + +	5 7 6.5	22 22 22	+ + +
(5) J01E_DID (6) J01F_DID† (7) J01M_DID†	Consumption of sulfonamides and trimethoprim (J01E) expressed in DID‡ Consumption of macrolides, lincosamides and streptogramins (J01F) expressed in DID‡ Consumption of quinolones (J01M) expressed in DID‡	6.5 7.5 8	22 22 22	+ + +	5 6	22 22 22	+ + +	6 6 7	21 21 21	+ + +	5.5 7 7.5	22 22 22	+ + +
(8) J01A_% (9) J01C_% (10) J01D_% (11) J01E_% (12) J01F_%	Consumption of tetracycline (J01A) expressed as percentages Consumption of penicillins (J01C) expressed as percentages Consumption of cephalosporins (J01D) expressed as percentages Consumption of sulfonamides and trimethoprim (J01E) expressed as percentages Consumption of macrolides, lincosamides and streptogramins (J01F)	5.5 5.5 6 5 7	22 22 22 22 22 22	+ + + +	5 5.5 5.5 5 6	22 22 22 22 22 22	+ + + +	5 5 6 5 6	21 21 21 21 21 21	+ + + +	6 6.5 6.5 6	22 22 22 22 22 22	+ + + + +
(13) J01M_%†	expressed as percentages Consumption of quinolones (J01M) expressed as percentages	7	22	+	6.5	22	+	7	21	+	7	22	+
(14) J01CE_%† (15) J01CR_%† (16) J01DD+DE_%†	Consumption of β-lactamase sensitive penicillins (J01CE) expressed as percentage§ Consumption of combination of penicillins, including β-lactamase inhibitor (J01CR) expressed as percentage§ Consumption of third and fourth generation of cephalosporins (J01(DD+DE))	8 7 7	22 22 22	+ + +	7 7 7	22 22 22	+ + +	8 7 8	21 21 21	+ + +	8 7 7.5	22 22 22	+ + +
(17) J01MA_%†	expressed as percentages Consumption of fluoroquinolones (J01MA) expressed as percentages	7	22	+	7	22	+	7	21	+	7.5	22	+
(18) JO1_B/N†	Ratio of the consumption of broad (J01(CR+DC+DD+(F-FA01))) to the consumption of narrow spectrum penicillins, cephalosporins and macrolides (J01(CE+DB+FA01))	7	22	+	7	22	+	7	21	+	7	22	+
(19) J01_SV† (20) J01M_SV† (21) J01M_SVDID†	Seasonal variation¶ of the total antibiotic consumption (J01) Seasonal variation¶ of quinolone consumption (J01M) Seasonal variation¶ of quinolone consumption (J01M) multiplied by their use in DID‡	7 7 6.5	22 21 22	+ + +	7 7 6	22 21 22	+ + +	7 7 7	21 20 21	+ + +	7.5 7 7	22 21 22	+ + +
(22) J01_TT	Index of longitudinal trends of antibiotic consumption	6	21	+	6	21	+	7	20	+	7	20	+

	Figure A-1:	Copy of Europe	an Surveillance	of Antimicrobial	Consumption	prescribing qua	litv indicators	presented by	/ Coenen et al. ((8)
--	-------------	----------------	-----------------	------------------	-------------	-----------------	-----------------	--------------	-------------------	-----

*A scale ranging from 1 (= completely disagree) through 5 (= uncertain) to 9 (= completely agree) was used.

+Proposed indicators were judged relevant and potentially valid if the median score for relevance was not within the 1-6 interval and if there was consensus—that is, if the number of scores within the 1-3 interval was fewer than one third of the panel. (The information provided by indicators 13 (J01M_%) and 20 (J01M_SV) overlapped with that provided by indicators 17 (J01MA_%) and 21 (J01M_SVDID), respectively.) #Defined daily doses (DDD) per 1000 inhabitants per day.

SPercentage of the total consumption of antibacterials for systemic use (J01) in DID‡.

"Overuse in the winter quarters (October-December and January-March) compared with the summer quarters (July-September an April-June) of a 1-year period starting in July and ending the next calendar year in June, expressed as percentage: [DDD (winter quarters)/DDD (summer quarters) -1] ×100.

1 A.2.1.2 Adriaenssens et al. (6)

2

On behalf of ESAC, Adriaenssens *et al.* (6) published a subsequent set of quality indicators for antibiotic prescribing. This included acceptable ranges/standards for measuring prescribing against. A number of indicators are condition-specific (6). The use of 'recommended' antibiotics is also measured, in addition to the use of quinolones, for specific conditions (6). A copy of these prescribing indicators is provided below in Figure A-2 (6).

- 8
- 9

No	Title	Label	Acceptable range (%)
1a.	Percentage of patients aged between 18 and 75 years with acute bronchitis/bronchiolitis (ICPC-2-R: R78) prescribed antibacterials for systemic use (ATC: J01)	(R78_J01_%)	0-30
1b.	=1a receiving the recommended antibacterials (ATC: J01CA or J01AA)	(R78_RECOM_%)	80-100
1c. 2a.	=1a receiving quinolones (ATC: J01M) Percentage of patients older than 1 year with acute upper respiratory infection (ICPC-2-R: R74) prescribed antibacterials for systemic use (ATC: J01)	(R78_J01M_%) (R74_J01_%)	0-5 0-20
2b.	=2a receiving the recommended antibacterials (ATC: J01CE)	(R74_RECOM_%)	80-100
2c.	=2a receiving quinolones (ATC: J01M)	(R74_J01M_%)	0—5
3a.	Percentage of female patients older than 18 years with cystitis/ other urinary infection (ICPC-2-R: U71) prescribed antibacterials for systemic use (ATC: J01)	(U71_J01_%)	80–100
3b.	=3a. receiving the recommended antibacterials (ATC: J01XE or J01EA or J01XX)	(U71_RECOM_%)	80—100
3c.	=3a receiving quinolones (ATC: J01M)	(U71_J01M_%)	0—5
4a.	Percentage of patients older than 1 year with acute tonsillitis (ICPC-2-R: R76) prescribed antibacterials for systemic use (ATC: J01)	(R76_J01_%)	0–20
4b.	=4a receiving the recommended antibacterials (ATC: J01CE)	(R76_RECOM_%)	80-100
4c.	=4a receiving quinolones (ATC: J01M)	(R76_J01M_%)	0-5
5a.	Percentage of patients older than 18 years with acute/chronic sinusitis (ICPC-2-R: R75) prescribed antibacterials for systemic use (ATC: J01)	(R75_J01_%)	0–20
5b.	=5a receiving the recommended antibacterials (ATC: J01CA or J01CE)	(R75_RECOM_%)	80-100
5c.	=5a receiving quinolones (ATC: J01M)	(R75_J01M_%)	0-5
6a.	Percentage of patients older than 2 years with acute otitis media/myringitis (ICPC-2-R: H71) prescribed antibacterials for systemic use (ATC: J01)	(H71_J01_%)	0—20
6b.	=6a receiving the recommended antibacterials (ATC: J01CA or J01CE)	(H71_RECOM_%)	80-100
6c.	=6a receiving guinolones (ATC: J01M)	(H71 J01M %)	0-5
7a.	Percentage of patients aged between 18 and 65 years with pneumonia (ICPC-2-R: R81) prescribed antibacterials for systemic use (ATC: J01)	(R81_J01_%)	90-100
7b.	=7a receiving the recommended antibacterials (ATC: J01CA or J01AA)	(R81_RECOM_%)	80-100
7c.	=7a receiving quinolones (ATC: J01M)	(R81_J01M_%)	0-5

antibacterials; J01XE, nitrofuran derivatives; J01XX, other antibacterials.

10 Figure A-2: Copy of European Surveillance of Antimicrobial Consumption quality indicators

- 11 published by Adriaenssens *et al.* (6)
- 12
- 13
- .
- 1
- 2 3

A.2.2 Quality indicators by Hansen et al. (9)

In 2010, Hansen et al. (9) published an expert consensus review on quality indicators for 5 6 respiratory tract infections. They evaluated the relevance of each indicator for antibiotic 7 resistance in addition to a separate evaluation for the relevance to patient benefit (9). They 8 included relevant diagnostic testing being performed, such as, rapid streptococcus A antigen and CRP testing being performed for relevant conditions, by which to improve the 9 10 appropriateness of prescribing (9). These do not facilitate measurement on a scale but instead detail expert opinion on which indicators are the most pressing factors to consider 11 for each condition. The ratings for relevance for patient benefit were substantially less than 12 that for antibiotic resistance for the strong majority of indicators (9), as seen in Figure A-3 13 below. Hansen et al. (9) also evaluated the importance of the type of antibiotic prescribed 14 for each condition, per Figure A-4. 15

Quality indicators	Relevance for antimicrobial resistance	Relevance for patient health benefit
Patients with acute sinusitis:		
Number of patients treated with antibiotics	92* (7)	35 (4)
Number of patients treated with antibiotics without a diagnostic test	38 (4)	15 (4)
Number of patients treated with antibiotics with a CRP test < 10 mg/l	73 (6)	50 (4.5)
Patients with acute otitis media (AOM):		
Number of patients treated with antibiotics	92* (7)	50 (4.5)
Number of patients < 2 years treated with antibiotics	85* (7)	69 (5.5)
Number of patients > 2 years with less than 3 days of symptoms of AOM treated with antibiotics	96* (7)	46 (6)
Number of patients with discharging ear treated with antibiotics	73 (6)	85* (6)
Patients with acute tonsillitis/pharyngitis:		
Number of patients treated with antibiotics	88* (7)	65 (5)
Number of patients treated with antibiotics without a StrepA test	62 (6)	31 (4)
Number of patients treated with antibiotics with a positive StrepA test	77* (6.5)	50 (4.5)
Number of patients treated with antibiotics with a negative StrepA test	69 (6.5)	27 (4)
Patients with acute bronchitis:		
Number of patients treated with antibiotics	96* (7)	35 (4)
Patients with pneumonia:		
Number of patients treated with antibiotics	62 (5)	58 (6)
Patients with acute exacerbation of chronic obstructive pulmonary disease:		
Number of patients treated with antibiotics	88* (6)	50 (4.5)
Number of patients not fulfilling all the Anthonisen criteria ² treated with antibiotics	88* (7)	62 (5)
Patients with acute lower respiratory tract infections:		
Number of patients treated with antibiotics	85* (7)	50 (4.5)
Number of patients treated with antibiotics without a preceding CRP test or X-ray of thorax	31 (4)	15 (4)
Number of patients treated with antibiotics with a CRP test < 20 mg/l	81* (6.5)	42 (4)
Patients with acute respiratory tract infections:		
Number of patients treated with antibiotics	85* (7)	50 (4.5)
Number of patients with no history of penicillin allergy treated with macrolides	92* (7)	42 (4)

Notes: The values represent agreement rates¹ in% (median on a Likert scale, range 1–7). CRP test = C-reactive protein rapid test. Strep A test = rapid Streptococcus A antigen detection test. *Consensus (agreement rate \geq 75%). ¹Percentage of experts who scored the dimension \geq 5 in the second Delphi round (n = 26) on a Likert scale, range 1–7. ²Increased dyspnoea, increasing expectorate, and increasing purulence of expectorate.

Figure A-3: Copy of Hansen et al.'s quality indicators (9)

Quality indicators	Patients with acute sinusitis	Patients with acute otitis media	Patients with acute tonsillitis/ pharyngitis	Patients with pneumonia	Patients with acute exacerbation of COPD	Patients with acute LRTI
Number of patients treated with narrow-spectrum penicillin	85* (7)	92* (7)	96* (7)	92* (7)	62 (5)	88* (7)
Number of patients treated with broad-spectrum penicillin +/- clavulanic acid	92* (7)	92* (7)	92* (7)	100° (7)	92* (6)	92* (7)
Number of patients treated with macrolides	88* (7)	85* (7)	85* (7)	88* (6)	77* (6)	88* (6.5)
Number of patients treated with cephalosporins	81* (7)	81* (7)	88* (7)	81* (6)	73 (6)	81* (7)
Number of patients treated with quipolones	81* (7)	81* (7)	65 (6)	81* (6.5)	85* (6)	81* (7)

Notes: The values represent agreement rates¹ in% (median on a Likert scale, range 0–7). COPD = chronic obstructive pulmonary disease. LRTI = lower respiratory tract infection. *Consensus (agreement rate \geq 75%). ¹Percentage of experts who scored the dimension \geq 5 in the second Delphi round (n = 26) on a Likert scale, range 1–7.

Figure A-4: Copy of quality indicators by Hansen et al (9)

5 6

4

1

1 A.2.3 Quality indicators by Le Maréchal et al. (10)

2

Le Maréchal et al. (10) performed a systematic review of quality indicators tailored for the outpatient care setting. They included administrative aspects such as provision of access to local guidelines, documentation, provision of information to patients (10). They included multiple indicators relating to parenteral/injection antibiotics in outpatients (10). As seen in Figures A-5a and A-5b, the indicators collated by Le Maréchal et al. (10) provide sensible advice, however, do not facilitate measurement of degree of compliance/quality.

Outpatient quality indicator (OQI)	Type of indicator	References	Study design ^a
OQI-1 Antibiotics should be prescribed for (most) bacterial infections (e.g. acute pneumonia, urinary tract infections)	process	26-30	А
OQI-2 Antibiotics should not be prescribed for viral infections or (most) self-limit- ing bacterial infections (e.g. acute bronchitis, influenza, acute otitis media in children >2 years old)	process	26,28-42	A, B, C
OQI-3 Outpatients should receive antibiotic therapy compliant with guidelines; this includes, but is not limited to, indication, choice of the antibiotic, duration, dose and timina	process	26-31,34,36,39,40, 43,44,46-51	A, B, C
OQI-4 Some antibiotics should be rarely prescribed	process	59	В
OQI-5 Acute upper respiratory infections and bronchitis should not be treated with antibiotics within the first 3 days, unless there is documented indication for treatment	process	29,30,60,61	Α, C
OQI-6 Outpatients with acute tonsillitis/pharyngitis should undergo a group A streptococcal diagnostic test to decide whether or not they should receive antibiotics	process	32	C
OQI-7 Outpatients with an acute tonsillitis/pharyngitis and positive group A strep- tococcal diagnostic test should be treated with antibiotics	process	29,30	A
OQI-8 Antibiotics for an acute tonsillitis/pharyngitis should be withheld, discon- tinued or not prescribed if an outpatient presents a diagnostic test (rapid anti- gen test or throat culture) negative for group A streptococci	process	40	В
OQI-9 Prescribed antibiotics should be chosen from an essential list/formulary	process	47,48,50-52,62-73	С
OQI-10 Possible contraindications should be taken into account when antibiotics are prescribed	process	79,80	A, C
OQI-11 Antibiotics from the list of essential antibiotics should be available in health facilities that dispense antibiotics	structure	47,48,50,51,62,67,70,81	A, C
OQI-12 Key antibiotics should not be out of stock in health facilities that dispense antibiotics	structure	50,51	С
OQI-13 Antibiotics in stock should not be beyond the expiry date	structure	50,81	A, C
OQI-14 Antibiotics that are dispensed to outpatients should be adequately labelled (patient name, antibiotics name, when antibiotics should be taken)	structure	47,48,50,51,65,67,70,82	С
OQI-15 Antibiotics should be adequately conserved and handled in health facilities	structure	50,51	С
OQI-16 Health facilities should keep adequate records of dispensed key antibiotics	structure	50	С
OQI-17 A copy of the essential antibiotics list should be available in health facilities	structure	47,48,50,62,67,70	С
OQI-18 Standard antibiotic treatment guidelines should be available in health facilities	structure	50	С
OQI-19 Health facilities should have access to the Summary of Product Characteristics of prescribed antibiotics, written in a local language	structure	47	C
OQI-20 Antibiotics should not be sold without prescription	structure	50	С
OQI-21 Outpatients and OPAT patients with an antibiotic prescription should be educated on how to take it, on the dosage, on expected side effects, and on the natural history of the disease	process	47,48,50,51,60,62,65,67, 70,75,76,82–86	A, B, C
OQI-22 The treatment plan should be agreed between the OPAT team and the referring clinician before start of treatment	process	84,85	В
OQI-23 All OPAT treatment plans should include dose, frequency of administra- tion and duration of therapy	process	84	В
OQI-24 OPAT antibiotics should be correctly stored, prepared, reconstituted, dis- pensed and administered	structure	84,86	В
OQI-25 Administered doses of OPAT should be documented on a medication card	process	84	В
OQI-26 The first dose of a new antibiotic in an OPAT should be administered in a supervised setting	process	84	В

Figure A-5a: Prescribing quality indicators presented by Le Maréchal et al. (10)

Outpatient quality indicator (OQI)	Type of indicator	References	Study design ^a
OQI-27 OPAT antibiotics should be regularly reviewed to optimize speed of intra- venous-to-oral switch	process	84	В
OQI-28 Each OPAT centre should monitor quality indicators on OPAT antibiotics	structure	84,86	В
OQI-29 An expert in OPAT (physician, nurse, pharmacist) should work in each OPAT centre	structure	86	В
OQI-30 The OPAT plan should be communicated to the general practitioner at discharge	structure	85	В
OQI-31 The OPAT programme should be accredited or certified	structure	86	В
OQI-32 In an OPAT programme, clinical and/or microbiological outcomes, includ- ing treatment failure and adverse events (including <i>Clostridium difficile</i> infec- tions), should be recorded	outcome	84,86	В

^aCategory of the study design: A, consensus-based indicators; B, review-based indicators; C, guideline-based indicators.

Figure A-5b: Prescribing quality indicators presented by Le Maréchal et al. (10)

3

4

1 2

A.2.4 The World Health Organization's Access, Watch and Reserve program, 2019

5

In 2017, the World Health Organization (WHO) published a list of Selection and Use of 6 7 Essential Medicines, classifying medicines into categories of Access, Watch and Reserve (AWaRe) (11,12). These classifications were proved to a useful antibiotic stewardship tool 8 9 (11,13). In 2019 WHO published its AWaRe classification, containing targets for at least 60% systemic antibacterials to be from their Access list (11,12). For example, the penicillin 10 class antibiotic, amoxicillin, is on the Access list, the macrolide azithromycin is on the Watch 11 12 list, and the monobactam aztreonam is on the Reserve list within the 2019 classification (11). A second AWaRe list was subsequently published in 2021 (14). 13

14

15 A.2.5 Quality indicators by Thilly et al. (15)

16

Thilly et al. (15) published a useful collection of ten quality indicators designed for general 17 practice. It includes indicators for patients prescribed quinolones at least twice within six 18 19 months, as well as indicators on seasonal variation by assessing prescribing during the winter/flu season compared to the summer/hot season as well as one for quinolone 20 21 prescribing at these times of year (15). Another indicator is presented to assess prescriptions with the course duration being over eight days for a list of specific antibiotics 22 (15). Co-prescription of antibiotics with systemic non-steroidal anti-inflammatory drugs, as 23 well as antibiotics with systemic corticosteroids, is also a focus (15). 24

25

1 Figure A-6: Prescribing Quality Indicators Presented by Thilly et al. (15)

Proxy indicator	Numerator description	Denominator description	Target value	Target patients
Pl 1 Antibiotic prescriptions against UTI in men (ratio)	Number of prescriptions of: nitrofurantoin (Jo1XEo1)+certain (fluoro) quinolonesª (Jo1MB+Jo1MAo6+Jo1MAo4+Jo1MAo7)+fosfomycin- trometamol (Jo1XXo1)	100 active [®] male patients≥16 years old	Optimal target: o Acceptable target:<0.5	Men≥16 years old
Pl 2 Antibiotic prescriptions against UTI in women (ratio)	Number of prescriptions of: nitrofurantoin (Jo1XEo1)+pivmecillinam (Jo1CAo8)+fosfomycin-trometamol (Jo1XXo1)	Number of prescriptions of quinolones (Jo1M)	Target:>1	Women≥16 years old
PI 3 Repeated prescription of quinolones (%)	Number of prescriptions of quinolones (Jo1M) among patients having been prescribed a quinolone (Jo1M) in the preceding 6 months	Total number of prescriptions of quinolones (Jo1M)	Optimal target: o Acceptable target: <10%	Men and women≥ 16 years old
PI 4 Seasonal variation of total antibiotic prescriptions (%)	(Number of prescriptions of antibiotics (Jo1) during the cold-weather and October–December) / Number of prescriptions of antibiotics (Jo1 season (April–September) – 1) x 100	season (January–March) during the hot-weather	Target:<20%	All patients
PI 5 Seasonal variation of quinolone prescriptions (%)	(Number of prescriptions of quinolones (Jo1M) during the cold-wea March and October–December) / Number of prescriptions of quinol hot-weather season (April–September) – 1) x 10	ther season (January– ones (Jo1M) during the oo	Optimal target:<5% Acceptable target:<10%	All patients
PI 6 Amoxicillin / second-line antibiotics prescriptions (ratio)	Number of prescriptions of amoxicillin (Jo1CAo4)	Number of prescriptions of: amoxicillin- clavulanic acid (Jo1CRo2) + quinolones (Jo1M) + cephalosporins (Jo1D) + MLSK ^c (Jo1F)	Target: > 1	All patients
Pl 7 Prescriptions of not indicated antibiotics (%)	Number of prescriptions of: lomefloxacin (Jo1MA07), moxifloxacin (Jo1MA14), certain (fluoro) quinolones ^a (Jo1MB+Jo1MA06+Jo1MA04+Jo1MA07), telithromycin (Jo1FA15), spiramycin-metronidazole (Jo1RA04) and cefaclor (Jo1DC04)	Total number of antibiotic prescriptions	Optimal target: o Acceptable target:<0.5%	All patients
PI 8 Estimated duration of antibiotic prescriptions > 8 days (%)	Number of prescriptions>8 days for the following antibiotics: amoxicillin (Jo1CAo4), co-amoxiclav (Jo1CRo2), cefuroxime, cefpodoxime, roxithromycin, clarithromycin, pristinamycin and nitrofurantoin (Jo1FGo)	Total number of antibiotic prescriptions for these eight antibiotics (calculation of this metric is explained in detail in supplementary Table S2)	Optimal target:<5% Acceptable target:<10%	All patients
Pl 9 Co-prescription of antibiotic and systemic non-steroidal anti- inflammatory drugs (%)	Number of antibiotic(s) (Jo1) + systemic NSAID(s) (Mo1A) co-prescribed on the same day	Total number of antibiotic prescriptions	Optimal target: o Acceptable target:<5%	All patients
Pl 10 Co-prescription of antibiotic and systemic corticosteroids (%)	Number of antibiotic(s) (Jo1)+systemic corticosteroid(s) (Ho2AB) co-prescribed on the same day	Total number of antibiotic prescriptions	Optimal target: o Acceptable target:<5%	All patients

UTI: urinary tract infections.

^a Jo1MB (rosoxacin, nalidixic acid, piromidic acid, pipemidic acid, oxolinic acid, cinoxacin, flumequine, nemonoxacin), Jo1MAo6 (norfloxacin)+Jo1MAo4 (enoxacin)+Jo1MAo7 (lomefloxacin).

 $^{\rm b}$ An active patient is a patient seen at least once by the general practitioner during the year 2017.

° MLSK: macrolides, lincosamides, streptogramins and ketolides.

4

A.3 Overview of the literature

3

A.3.1 Literature considerations

5 Conducting a literature review was complicated by the fact that many relevant articles 6 utilise different keywords/subject areas and indexing, making the search significantly 7 larger if all are included. It took some time to clarify, however, it was found that many 8 highly relevant articles were found to use vastly different keywords and indexing. Below 9 are two examples of relevant research with their subject search terms listed below.

11	For example,	Cadieux et al. (21) of 2007 includes the following:
12	•	Anti-Bacterial Agents - therapeutic use,
13	•	Bacterial Infections - drug therapy ,
14	•	Bacterial Infections - epidemiology,
15	•	Drug Utilization Review ,
16	•	Foreign Medical Graduates - statistics & numerical data ,
17	•	Patients - statistics & numerical data,
18	•	Physicians (General practice),
19	•	Practice guidelines (Medicine),
20	•	Practice Patterns, Physicians' - statistics & numerical data,
21	•	Primary Health Care - statistics & numerical data,
22	•	Virus Diseases - drug therapy .
23	•	
24	However, Ba	rlam et al. (22) from 2015 includes the following:
25	•	Ambulatory Care - statistics & numerical data,
26	•	Anti-Bacterial Agents - therapeutic use ,
27	•	Family Practice - statistics & numerical data ,
28	•	Inappropriate Prescribing - statistics & numerical data,
29	•	Internal Medicine - statistics & numerical data,
30	•	Lung Diseases - complications,
31	•	Practice Patterns, Physicians' - statistics & numerical data,
32	•	Respiratory Tract Infections – complications.
33	•	

This process was also complicated by the number of ways of referring to
 Guideline-compliance or non-compliance, such as broad-spectrum or quinolone
 antibiotic prescribing use and insufficient prescription duration to indicate
 inappropriate prescribing.

As a result of these considerations, multiple search strategies were used
(described below in the next section, Appendix A.3.2) as well as 'snowballing'. An
overview of the literature was provided instead of a traditional literature review or
systematic review, which would have required too many resources and time. For
example, please see an Ovid advanced search performed below in Table A-2,
demonstrating the large numbers of articles resulting with the addition of relevant
search terms.

```
15
```

16 Table A-2: Summary of literature search terms performed

<u>#</u>	Searches	Results
1	(General Practice OR Pediatrics OR Community Medicine) AND Drug Therapy AND Antibiotic Agent AND Prescriptions AND Guideline Adherence {Including Related Terms}	10003
2	limit 1 to (english language and yr="2007 -Current")	5069
3	(General Practice OR Pediatrics OR Community Medicine) AND Drug Therapy AND Antibiotic Agent AND Prescriptions AND Guideline Adherence AND Australia {Including Related Terms}	10033
4	limit 3 to yr="2007 -Current"	5901
5	(General Practice OR Community Medicine) AND Drug Therapy AND Antibiotic Agent AND Prescriptions AND Guideline Adherence AND Australia {No Related Terms}	327
6	limit 5 to (english language and humans)	291

2

3

A.3.2 Revised literature search strategy

A revised search strategy was implemented using multiple search terms of MESH index terms and keyword searches, in Tables A-3 to Table A-7. The combination of the results of these as mentioned in Table A-7, were intended to limit the search to a more manageable number of articles. Snowballing was also used as a strategy likely to overcome some of the issues raised above. Despite these attempts to limit the blowing out of articles to review, several thousand articles were reviewed in full.

10 11

12 Table A-3: Medline Mesh index terms search {including related terms}

Search #	MeSH search terms {including related terms}
1	*Anti-Bacterial Agents/tu [Therapeutic Use]
2	*Primary Health Care/ OR *Ambulatory Care/ OR *Family Practice/ OR *General Practice/ OR *Physicians, Family/
3	*Practice Patterns, Physicians'/
4	1 AND 2 AND 3

13

14 Table A-4: Medline keyword search

Search #	Medline search terms
5	(antibiotic OR antimicrobial OR anti-microbial).mp
6	(associat* OR driv* OR influenc* OR relat* OR predict*).mp
7	(general practice OR GP OR general practitioner OR community care OR primary care OR family physician OR family medicine OR general internal medicine).mp
8	(poor OR over OR inappropriate* OR adher* OR congruen* OR excess*).mp
9	(prescr* AND guideline*).mp
10	5 AND 6 AND 7 AND 8 AND 9

15 where [mp=ti, ab, hw, tn, ot, dm, mf, dv, kw, fx, nm, kf, px, rx, an, ui, sy]

16

17 Table A-5: Embase Em index terms search {including related terms}

Search #	Em search terms {including related terms}
12	*antibiotic agent/dt [Drug Therapy]
13	*prescription/
14	Ambulatory care/ OR primary medical care/ OR child health care/ OR general practice/ OR primary health care/ OR General practitioner/ OR pediatrician/
15	12 AND 13 AND 14

18

19 Table A-6: Embase keyword search

Search #	Embase search terms
16	(antibiotic OR antimicrobial OR anti-microbial).mp
17	(associat* OR driv* OR influenc* OR relat* OR predict*).mp
18	(general practice OR GP OR general practitioner OR community care OR primary
	care OR family physician OR family medicine OR general internal medicine).mp
19	(poor OR over OR inappropriate* OR adher* OR congruen* OR excess*).mp
20	(prescr* AND guideline*).mp
21	16 AND 17 AND 18 AND 19 AND 20

20 where [mp=ti, ab, hw, tn, ot, dm, mf, dv, kw, fx, nm, kf, px, rx, an, ui, sy]

2 Tab

 Table A-7:
 Combining all four search results

Search #	search terms
22	4 OR 10
23	Automatic removing of duplicates (4 OR 10)
24	15 OR 21
25	Automatic removal of duplicates in Ovid (15 OR 21)
26	Exported to EndNote (23 OR 25)

Table A-8 below also includes examples of exclusion criteria followed thereafter.

Table A-8: Examples of removal of articles not meeting criteria

	1 5
Action #	Action description
27	remove duplicate references
28	non-English full text
29	non-original research: editorials, perspectives, news, anonymous
30	non-original research: (Systematic) Reviews/Meta-analyses, Study Protocols
31	(practice)guidelines, case reports, conference abstracts/papers
32	main focus of study not on investigating inappropriate/appropriate prescribing of antibiotics
33	qualitative design only
34	research not conducted within present definition of primary care, and investigating general population not specific disease groups
35	interventions/Before&After studies/ Evaluations Guidelines or intervention effectiveness
36	no identification/differentation of inappropriate from appropriate prescribing
37	rates of prescribing only measure used to define/differentiate appropriate from inappropraite prescribing
38	hypothetical Case Vignettes, Questionnaires/Surveys where no recording of real patient consultations & prescribing
39	methods do not include investigation of relationship between patient-, prescriber-, practice-, or consultation-factors and inappropriate/appropriate prescribing
40	statistical methods do not involve multivariable investigation of relationships between patient-, prescriber-, practice-, or consultation-factors and inappropriate/appropriate prescribing
41	Addition of relevant studies identified by snowballing

- APPENDIX B APPENDICES TO THE METHODS CHAPTER 1
- 2

4

B.1

Summary of the data received

5 The five-and-a-half years of longitudinal data was received in SAS database datasets. MedicineInsight data was partially cleaned at the time of receipt. These were all 6 7 predominantly character string files apart from the comorbid conditions file. This 8 administrative data uses real zeros rather than censoring.

9

File	Rows	Variables	Unique	Unique
			Providers	Patients
Patient	791280	32	n/a	791280
Practice	52	13	n/a	n/a
Provider	11976		7893	n/a
Diagnosis	1,557,387	11	2388	658577
Prescription	12,941,223	29	26,230	614,843
Patient Comorbid Conditions	764,751	35	n/a	764751
Patient Allergy	168,931	7	1351	119614
Encounter	19,481,775	8	5586	791280
Service	9,908,909	8	954	667649
Atomic Pathology	40,121,296	13	n/a	363703
Prescription History	5668253	32	1277	617188
Observation	8538890	10	2019	492454
Non-Atomic Pathology	6837651	6	1608	445069

Table B-1: Summary of digital files received 10

11

12 To improve data quality, further cleaning included checking variables for incorrect, 13 impossible and missing entries, to standardise units of measurement, and group diagnoses. Programs were written in STATA Release 16 (16) to allow for semi-14 automated corrections to these data, including allocation of a missing category where 15 appropriate (17,18). Data preparation included the creation of new variables from, and 16 transformation of, existing variables to facilitate intended analyses. 17

18

B.2 19 Diagnoses

20

The initial diagnoses file received included 1,557,387 rows, with 275,880 different 21 22 diagnoses entered. Removing impossible/erroneous dates and missing/blank diagnoses resulted in 1,489,540 rows. Diagnoses were initially explored to see what 23 GPs were entering in the free text diagnosis field, and to collate relevant key words to 24 25 use in search terms for conditions of interest. Datasets of diagnosis information for

relevant conditions were then created by searching for relevant diagnoses of interest
using character string functions. This process was profoundly time-consuming.

3

This process varied in complexity depending upon the specific condition and the types of diagnoses commonly entered by GPs in this dataset. The number of symptoms entered as clinical diagnoses was surprising, e.g. sniffles. This was particularly the case for URTI, most notably acute rhinosinusitis. The process of refining the search terms for each condition of interest took a substantial amount of time.

9

While higher-line antibiotics may be appropriate to prescribe at a subsequent 10 consultation when lower-line options have already been tried, to accurately evaluate the 11 prescribing occurring at initial consultations, non-initial consultations need to be 12 excluded. Similarly, it may potentially be appropriate to prescribe an antibiotic for an 13 ongoing infection at a subsequent consultation when prescribing did not occur initially. 14 Both points highlight the need to separate initial from non-initial consultations, to 15 accurately examine the prescribing behaviour occurring at either. Therefore, in order to 16 limit URTI and UTI diagnoses to initial presentations for the episode of infection (19), 17 any diagnoses with coding suggesting non-initial or follow-up consultations, or chronic, 18 recurrent and/or resistant infections were excluded (20), as well as removing diagnoses 19 occurring within fourteen days of a previous consultation for the same condition group 20 for the same patient. This time period was chosen for both URTI and UTI, as the longest 21 typical treatment duration is up to fourteen days for UTI (2). The removal of diagnoses 22 containing pathology and/or species-specific information followed, as these would not 23 have been available at an initial consultation, for example, "E.coli UTI" and are therefore 24 unlikely to represent initial consultations. Note all diagnoses used for analysis were 25 restricted to initial presentations for the relevant condition. 26

27

Each diagnostic condition dataset were then merged with additional patient information and practice information info by matching the date, patientid, providerid and practiceid in each file.

- 31
- 32

1 B.2.1 Upper respiratory tract infection

2

Search terms for AOM focused on otitis media and diagnoses of ear infection and
symptoms like serous, effusion, discharge, purulent, suppurative, and mucous.
Diagnoses such as otitis externa and chronic suppurative otitis media were excluded
from AOM (2). The quantity of symptoms entered as clinical diagnoses for URTI was
surprisingly high, for example, "sniffles", notably for acute rhinosinusitis.

8

9 The reason for prescribing field was string-searched to remove any diagnoses for which 10 the reason recorded was unrelated to the above diagnoses.

11

12 B.2.1.1 Pharyngitis / tonsillitis

Search terms for pharyngitis / tonsillitis included the following diagnoses, in addition to
variations on these:

- 15 PHARYNGITIS
- 16 PHARINGITIS
- 17 TONSILLITIS
- 18 TONSILITIS
- 19 TONSILLI
- 20 TONSILLAR
- TONASILLITIS
- TONCILITIS
- TONSILS
- SORE THROAT
 - SORE THOAT DAYS
- 26

25

27 <u>B.2.1.2</u> Acute otitis media

AOM is difficult to distinguish clinically from other types of otitis media, which have different guidelines. Due to this fact, it was sometimes unclear whether one should classify a diagnosis as strictly AOM or more likely relating to other types of OM and therefore excluded. Persistent Otitis Media with Effusion (OME), also known as 'glue ear', was most difficult to distinguish from AOM, as it is essentially a chronic version of AOM lasting three months or more, and a natural progression. Ear pain and redness of the tympanic membrane are indicative of AOM, effusion commonly occurs in both AOM and CSOM. Based on expert GP and ENT advice, terms more likely to describe other
types of OM were also removed, including 'otitis media effusion', 'mucus', 'mucous', and
'purulent' OM. However, 'middle ear effusion' was considered AOM if it occurred without
any mention of OM in the diagnosis, and without any suggestion of it being chronic in
nature.

6

Diagnoses including any reference to chronic, resistant, complicated, persistent 7 8 infection were excluded. Chronic suppurative otitis media (CSOM) is a condition which 9 can result in more serious physiological changes. While perforation of the tympanic membrane can also occur in AOM, diagnoses detailing 'perforation' were removed as 10 they are more likely to indicate CSOM, as were instances of 'suppurative' and 11 'adhesive'. Bullous OM were also excluded, as were any diagnoses relating to otitis 12 externa (OE), commonly known as 'swimmer's ear'. Search terms included the following 13 diagnoses, in addition to variations: 14

- 15 AOM
- 16 ACUTE OTITIS MEDIA
- OTITIS MEDIA
- 18 MIDDLE EAR EFFUSION
- 19

20 The following diagnoses were excluded:

- PERSISTENT OTITIS MEDIA WITH EFFUSION
- GLUE EAR
- OTITIS MEDIA EFFUSION
- OM MUCUS
- OM MUCOUS
- PURULENT OM
- 27 CSOM
- CHRONIC SUPPURATIVE OTITIS MEDIA
- 29 PERFORATION
- 30 SUPPURATIVE
- 31 ADHESIVE
- BULLOUS OM
- 33 OTITIS EXTERNA

1	• OE
2	SWIMMERS' EAR
3	
4	B.2.1.3 Influenza / influenza-like illness
5	Search terms for influenza / influenza-like illness included the following diagnoses:
6	• FLU
7	FLU-LIKE ILLNESS
8	FLU LIKE ILLNESS
9	• FLI
10	INFLUENZA
11	INFLUENZA-LIKE ILLNESS
12	INFLUENZA LIKE ILLNESS
13	• ILI
14	
15	B.2.1.4 Acute rhinosinusitis
16	Search terms included the following diagnoses, in addition to variations:
17	• URTI
18	UPPER RESP TRACT INFECTION
19	• URT
20	 INFECTION; UPPER RESP TRACT
21	 VIRAL UPPER RESPIRATORY TRACT INFECTION
22	VIRAL URTI
23	INFECTIONUPPER RESP TRACT
24	VIRAL UPPER RESP TRACT INFECTION
25	V URTI
26	UPPER RESPIRATORY TRACT INFECTION VIRUS
27	PRODROMAL OF COMMON COLD
28	 PRODROMAL OF VIRAL RESP INFECTION
29	PRODROMAL OF VIRAL URTI
30	PRODROMAL VIRAL INFECTION
31	PRODROMAL VIRAL URTI
32	LIKELY VIRAL URTI
33	• COLD

1	COMMON COLD
2	RHINOSINUSITIS
3	RHINO SINUSITIS
4	RHINOISINUSITIS
5	RHINORSINUSITIS
6	RHINOSINISUTUS
7	RHINOSINOSITIS
8	RHINOSINUSIITIS
9	RHINOSINUSITISVIRAL
10	RHINOSINUSITS
11	RHINOSINUTISITS
12	RHINSOSINUSITIS
13	
14	B.2.1.5 Exclusions
15	A large number of diagnoses picked up by the search terms but irrelevant, such as
16	"URTICARIA", required removal. Conditions with separate prescribing guidelines, or
17	indicative of surgery were also excluded. Examples of diagnoses excluded are listed
18	below:
19	LOWER RESPIRATORY TRACT INFECTION
20	• LRTI
21	RESPIRATORY TRACT INFECTIONLOWER
22	PNEUMONIA
23	BRONCHITIS
24	BRONCIOLITIS
25	CHEST INFECTION
26	CHEST INF
27	• ASTHMA
28	ALLERGIC RHINITIS
29	PERENNIAL RHINITIS
30	• CROUP
31	URTICARIA
32	STOMACH FLU
33	COLD SORE

1	COLD AGGLUTINATION
2	COLD AGGLUNTININS
3	HOT & COLD
4	COLD POLYP
5	• FLU AT <age></age>
6	• ILIAC
7	POST SURGICAL
8	TONSILS AGE <age></age>
9	TONSILS AND ADENOIDS
10	TONSILS GROMETS
11	TONSILS REMOVED
12	TONSILS, ADENOIDS
13	TONSILS, GROMETS
14	TONSILLAR ABSCESS
15	
16	Any uncertain URTI diagnoses were removed by following strings relating to a relevant
17	search term above were removed. (Where there was no uncertainty regarding URTI
18	diagnosis but another concurrent diagnosis within the diagnosis string, these were
19	included. For example, "UPPER RESPIRATORY TRACT INFECTION, DEPRESSION,
20	?UTI" was kept. Diagnoses including the following terms were excluded:
21	• QUERY
22	• ?
23	PROBABLE
24	POSSIBLE
25	• POSS
26	• PROB
27	
28	Diagnoses of situations which requiring immediate action, i.e. hospital admission, were
29	removed, as these are not suitable for treatment in the community and therefore outside
30	the scope of the research question. For example, "URTI/FEBRILE CONVULSIONS"
31	was removed. Diagnoses including the following terms were excluded:
32	SEPTICAEMIA
33	SEPTIC

1	SEPSIS
2	CONVULSIONS
3	• TO ED
4	ADMIT
5	ADMITTED
6	ADMISSION
7	• ICU
8	• HOSPITAL
9	
10	Diagnoses of certain viruses or bacteria which can only be confirmed by the lab were
11	also removed, as these do not represent the initial consultation but either subsequent
12	consults when laboratory results were available or the doctor upon receipt of laboratory
13	results. Diagnoses with the following terms were excluded:
14	INFLUENZA A
15	INFLUENZA B
16	ADENOVIRUS
17	• RSV
18	• H1N1
19	SWINE FLU
20	PSEUDOMONAS
21	 <pathogen> CARRIER</pathogen>
22	
23	Diagnoses detailing immunisations and / or vaccinations, especially influenza, were
24	common, and were excluded using the following search terms:
25	IMMUNISATION
26	IMMUNIZATION
27	VACCINATION
28	VACCINE
29	• VACC
30	• VAX
31	• SHOT
32	• JAB
33	NEEDLE

1	• BOOSTER
2	BOOSTRIX
3	INJECTION
4	• INJ
5	
6	For URTI diagnoses, any diagnoses indicative of non-initial consultations, such as
7	review consultations or chronic, resistant infections were removed, as follows:
8	RECURRENT
9	COMPLICATED
10	PROLONGED
11	• RECENT
12	FOLLOW-UP
13	FOLLOW UP
14	FOLLOWUP
15	• FUP
16	• F/U
17	• F/UP
18	• FUP
19	• FF UP
20	CHRONIC
21	RESISTENT
22	RESISTANT
23	REVIEW
24	• RV
25	• R/V
26	• R/O
27	• RX
28	RECURRING
29	ONGOING
30	• RECHECK
31	CHECK UP
32	PERSIST
33	IMPROVED

1	IMPROVING		
2	RESOLVING		
3	RESOLVED		
4	BETTER NOW		
5			
6	B.2.2 Urinary tract infection		
7			
8	B.2.2.1 Acute cystitis		
9	In order to explore prescribing practices for patients presenting with initial episodes of		
10	care for uncomplicated UTI, patients were filtered by consultations considered relevant		
11	to acute cystitis in the Guidelines. This involved developing algorithms to search through		
12	character strings of consultations relating to acute cystitis.		
13			
14	Initial search terms were collated from the description of the condition within the		
15	Guidelines. Final search terms included the following diagnoses plus variations on		
16	these:		
17	• UTI		
18	URINARY TRACT INFECTION		
19	ACUTE CYSTITIS		
20	CYSTITIS		
21			
22	B.2.2.2 Exclusions		
23	Removal of any diagnoses of prostatitis or pyelonephritis, relevant to different but		
24	related conditions for which there are separate guidelines, then followed.		
25			
26	Diagnoses suggestive of chronic or complicated UTI or non-initial consultation were		
27	then removed. This included removal of diagnoses with any mention of UTI which was		
28	chronic, complicated, resistant, catheter-related, post-surgical, resolved infections, or		
29	review consults. This also applied to issuing scripts as prophylaxis in chronic cases.		
30	Diagnoses meeting the criteria for UTI diagnosis in addition with the strings including		
31	but not limited to the following were excluded:		
32	COMPLICATED		
33	COMPLICARTED		
34	CHRONIC		

1	•	MULTIPLE
2	•	BACK UP SCRIPT
3	•	BREAKTHROUGH
4	•	FREQUENT UTI
5	•	RECURANTLY
6	•	CHANGE ABS
7	•	CHECK
8	•	FOLLOW
9	•	RESULT
10	•	RESOLVED
11	•	PESISTANT
12	•	SEVERAL
13	•	RETURNING
14	•	RECURRENCE
15	•	RESIDUAL
16	•	RESISTANCE
17	•	RESISTENCE
18	•	RESISTENT
19	•	RESISTENT TO TRIMETHOPRIM
20	•	REVIEW
21	•	R/O
22	•	RV
23	•	R/V
24	•	RX
25	•	PREV
26	•	PRIOR
27	•	RECENT
28	•	REC
29	•	REPEAT
30	•	PRONE
31	•	PAST
32	•	PERSISTANT
33	•	PERSISTENT

1	PERSISTING
2	PHONE
3	• PHX
4	PROPHYLAXIS
5	PROPYLAXIS
6	2nd EPISODE
7	• UTI x 3
8	
9	For UTI diagnoses, diagnoses entered regarding specific pathogenic species, such as
10	"E. coli UTI" were also excluded from the analysis, as these diagnoses can only be
11	made in light of pathology results and are therefore likely to represent subsequent
12	consultations or the input of test results. Diagnoses including the following strings were
13	excluded:
14	• E.COLI
15	• E COLI
16	• E. COLI
17	• ECOLI
18	PROTEUS
19	KLEBSIELLA
20	• KLEB
21	PSEUDOMONAS
22	ENTEROCOCCUS
23	CITEROBACTER
24	ACINETOBACTER
25	CONFIRMED ON URINE MC
26	STAPHYLOCOCCAL
27	RESULTS
28	
29	Diagnoses including a high degree of uncertainty about the diagnosis were also
30	removed, e.g. POSS, QUERY, ?UTI. Diagnoses including mention of immediate
31	hospital admission were also excluded, as these are not representative of
32	uncomplicated UTI for which prescribing can be assessed. For example, when the GP

33 cuts a consult short and sends a patient straight to the emergency department,

1	antibiotics may well be warranted however they will unlikely be prescribed by the GP.			
2	As such, these situations are not considered suitable for the research question at hand.			
3	Diagnoses containing UTI in addition to the following strings were excluded:			
4	TO ED			
5	ADMISSION			
6	• ADMIT			
7	ADMITTED			
8	DELERIUM			
9	HOSPITAL			
10	SEPTICAEMIA			
11	SEPSIS			
12	INPATIENT			
13				
14	Duplicates of identical diagnosis, with matching date, patient id, provider id and practice			
15	id variables were also removed.			
16				
17	B.3 Antibiotic prescriptions			
18				
19	B.3.1 Antibiotic active ingredients			
20	The prescription dataset was filtered/subset to contain only systemic antibiotics. This			
21	included searching for both active ingredients and brand names, as well as antibiotic			
22	class. The reason for including all three was to cover cases of coding errors in which			
23	class variable field may have been entered where ingredient is intended. The search			
24	was performed on the string variable entries for medicine name in the prescribing			
25	dataset received. A full list of search terms is follows, noting that many were searched			
26	with a 'string starting with (search term)' function. For example, any string starting with			
27	"Tobra" was a line of code implemented.			
28				
29	Search terms included:			
30	Aminoglycoside			
31	Amikacin			
32	Gentamicin			
22	Tobramycin			

• Tobramycin PF

1	•	Tobra-Day
2	•	TOBI
3	•	Carbapenem
4	•	Ertapenem
5	•	Invanz
6	•	Imipenem
7	•	Primaxin
8	•	Meropenem
9	•	Cephalosporin
10	•	Cefaclor
11	•	Ceclor CD
12	•	Karlor CD
13	•	Keflor CD
14	•	Ozcef
15	•	Aclor
16	•	Ceclor
17	•	Keflor
18	•	Cefalexin
19	•	Cephalex
20	•	Cephalexin
21	•	Cilex
22	•	lalex
23	•	Ibilex
24	•	Keflex
25	•	Rancef
26	•	Cefalotin
27	•	Cephalothin
28	•	Cefazolin
29	•	Cefepime
30	•	Cefotaxime
31	•	Cefoxitin
32	•	Ceftaroline
33	•	Zinforo

1 •	Ceftazidime
2 •	Fortum
3 •	Ceftolozane with tazobactam
4 •	Zerbaxa
5 •	Ceftriaxone
6 •	Cefuroxime
7•	Zinnat
8 •	Glycopeptide
9 •	Teicoplanin
10 •	Targocid
11 •	Vancomycin
12 •	Vancocin
13 •	Vancocin CP
14 •	Vycin
15 •	Lincosamide
16 •	Clindamycin
17 •	Cleocin
18 •	Clindamyk
19 •	Dalacin C
20 •	Dalacin C Phosphate
21 •	Lincomycin
22 •	Lincocin
23 •	Macrolide
24 •	Azithromycin
25 •	Zithromax
26 •	Zedd
27 •	Azith
28 •	Clarithromycin
29 •	Clarac
30 •	Clarithro
31 •	Kalixocin
32 •	Klacid
33 •	Erythromycin

1	• EES
2	Erythromycin Ethylsuccinate
3	• Eryc
4	E-Mycin
5	Erythrocin IV
6	Roxithromycin
7	Rulide D
8	 Biaxsig
9	Roxar
10	Roximycin
11	Rulide
12	Penicillin
13	Amoxicillin
14	Yomax
15	Alphamox
16	Amoxil Forte
17	Cilamox
18	Ranmoxy
19	Maxamox
20	Amoxil
21	Ibiamox
22	Fisamox
23	Amoxicillin with clavulanic acid
24	Augmentin Duo
25	Moxiclav Duo
26	 AlphaClav Duo
27	Curam Duo
28	 Moxiclav Duo Forte
29	 AlphaClav Duo Forte
30	Augmentin Duo Forte
31	Clavam
32	Curam Duo Forte
33	Augmentin

1	Curam
2	Amoxiclav
3	Co-amoxiclav
4	 coamoxiclav
5	Ampicillin
6	 Austrapen
7	 Ibimicyn
8	Ampicyn
9	 Benzathine benzylpenicillin
10	Bicillin
11	Bicillin L-A
12	Benzylpenicillin
13	BenPen
14	Dicloxacillin
15	 Distaph
16	 Flucloxacillin
17	 Flopen
18	 Staphylex
19	Flucil
20	Flubiclox
21	Phenoxymethylpenicillin
22	Aspecillin VK
23	Cilicaine VK
24	• LPV
25	Cilicaine V
26	Abbocillin V
27	Phenoxymethylpenicillin
28	 Piperacillin with tazobactam
29	Pipercaillin
30	 Tazopip
31	 PiperTaz
32	Tazocin EF
33	 Tazopip

1	•	Piptaz
2	•	Procaine benzylpenicillin
3	•	Procaine penicillin
4	•	Cilicaine
5	•	Quinolone
6	•	Ciprofloxacin
7	•	Ciprol
8	•	C-Flox
9	•	Ciproxin
10	•	Cifran
11	•	Loxip
12	•	Ciproxin IV
13	•	Ciprofloxacin
14	•	Moxifloxacin
15	•	Avelox
16	•	Norfloxacin
17	•	Nufloxib
18	•	Roxin
19	•	Tetracycline
20	•	Doxycycline
21	•	Doxylin
22	•	Frakas
23	•	Doxsig
24	•	Doryx
25	•	Minocycline
26	•	Akamin
27	•	Minomycin
28	•	Tetracycline
29	•	Other antibacterial
30	•	Aztreonam
31	•	Monobactam
32	•	Azactam
33	•	Colistin

1	•	Colistin
2	•	Colistimethate sodium
3	•	Tadim
4	•	Daptomycin
5	•	Cubicin
6	•	Fidaxomicin
7	•	Dificid
8	•	Linezolid
9	•	Oxazolidinone
10	•	Linevox
11	•	Zyvox
12	•	Methenamine hippurate
13	•	Hexamine hippurate
14	•	Hiprex
15	•	Nitrofurantoin
16	•	Macrodantin
17	•	Sodium fusidate
18	•	Fucidin
19	•	Sulfadiazine
20	•	Sulfonamide
21	•	Tigecycline
22	•	Glycylcycline
23	•	Tygacil
24	•	Trimethoprim
25	•	Alprim
26	•	Triprim
27	•	Trimethoprim with sulfamethoxazole
28	•	Sulfamethoxazole and trimethoprim
29	•	Co-trimoxazole
30	•	Cotrimoxazole
31	•	Bactrim
32	•	Resprim
33	•	Bactrim DS

- 1 Resprim Forte
- 2 Septrin Forte
 - Septrin

3

- 5 By active ingredient, the resulting list was as follows:
- 6 Table B-2: List of final medicine active ingredients

Active Ingredient		
AMOXICILLIN		
AMOXICILLIN WITH CLAVULANIC ACID		
AMPICILLIN		
AZITHROMYCIN		
BENZATHINE BENZYLPENICILLIN		
CEFACLOR		
CEFALEXIN		
CEFTRIAXONE		
CEFUROXIME		
CIPROFLOXACIN		
CLARITHROMYCIN		
CLINDAMYCIN		
DICLOXACILLIN		
DOXYCYCLINE		
ERYTHROMYCIN		
FLUCLOXACILLIN		
GENTAMICIN		
MINOCYCLINE		
NITROFURANTOIN		
NORFLOXACIN		
PHENOXYMETHYLPENICILLIN		
PROCAINE BENZYLPENICILLIN (PROCAINE PENICILLIN)		
ROXITHROMYCIN		
TOBRAMYCIN		
TRIMETHOPRIM		
TRIMETHOPRIM WITH SULFAMETHOXAZOLE		

Topical preparations of antibiotics were picked up in the search and then deleted from
the file, such as drops, ointments and gel preparations of the above, as well as
ingredients including but not limited to SILVER SULFADIAZINE, CHLORHEXIDINE
GLUCONATE, BENZOYL PEROXIDE.

6

7 B.3.2 Antibiotic class

Another variable for antibiotic class was then created, sorting active ingredients into the
following classes:

- 10
- 11 Table B-3: List of antibiotic classes
- 12 AMINOGLYCOSIDES
- 13 CARBAPENEMS
- 14 CEPHALOSPORINS
- 15 GLYCOPEPTIDES
- 16 LINCOSAMIDES
- 17 MACROLIDES
- 18 PENICILLINS
- 19 QUINOLONES
- 20 TETRACYCLINES
- OTHER ANTIBACTERIALS
- 22

The prescription dataset was then merged with each condition file by matching on the date, patient id, provider id and practice id variables. Where more than one prescription for a different antibiotic was prescribed to the same patient by the same provider at the same practice on the same date, this patient and date was flagged and later rechecked to select the most appropriate antibiotic was selected for analysis once pathology data had been incorporated.

29

Essentially when relying on matching one diagnosis with one prescription for analysis, one must be assumed to be the intended prescription, although it must be noted that there were situations where there were more than one diagnosis made for the same patient at an encounter and more than one systemic antibiotic prescribed at that encounter. Assumptions had to be made regarding which was intended for what diagnosis, however, the most appropriate choice was selected according to the
guidelines to give the benefit of the doubt.

3

In order to decrease the introduction of bias, flags were created for consultations when 4 multiple scripts for systemic antibiotics were written for same patient on same date by 5 6 the same provider at the same practice. Flags for instances when more than one of the 7 same active ingredient were then removed, leaving only occasions when different 8 systemic antibiotics were prescribed at the same consultation. Pathology results were 9 the incorporated, which in some cases indicate the most appropriate choice of antibiotic, for example, in the presence of a positive culture which is susceptible to the chosen 10 antibiotic whereas the first-line choice in the guidelines would have been ineffective. In 11 this manner, after taking into account the pathology results, the most appropriate 12 antibiotic was then selected for analysis, making the assumption that the most 13 appropriate was the intended antibiotic to treat the diagnosis of interest, thereby giving 14 providers the benefit of the doubt. 15

- 16
- 17

18 B.3.3 Whether antibiotic was prescribed

19

A binary discrete variable was created to indicate whether or not the patient received a prescription for a systemic antibiotic following the consultation. This was performed following the matching of the systemic antibiotic prescription file with the relevant condition of interest file, and matching on all four variables: patient id, practice id, provider id and date.

25

26 B.3.4 Whether repeats were issued on the antibiotic prescription status

27

A repeat prescription is a numerical field, typically 0 to 5, on a pharmaceutical prescription, which is entered by the GP at the time of prescribing. This allows the patient to fill said additional numbers of the same prescription. While prescriptions should not be issued without an indication, repeats on prescriptions should also not be issued without good reason. Unfortunately, many patient management systems in Australia automatically generate one repeat on prescriptions which takes practitioners time to amend to zero, repeats are often believed to be issued without good reason. An 1 acceptable reason for issuing repeats on prescriptions is patient inability to access care

2 for a period during which a prescription may be required.

3

Whether or not a GP issues any repeats on a prescription is feasibly considered to be
a potential indicator of better or worse prescribing behaviour. A binary discrete variable
was created for whether the antibiotic prescription issued was done so with or without
any repeats.

- 8
- 9 B.4 Patient variables
- 10

11 B.4.1 Patient age groups

12

A continuous age variable was received in the patient dataset, in addition to five-year 13 and ten-year age groups. Two, new ordinal categorical variables were created for 14 patient age for URTI and UTI based on the distribution for each dataset. For stability, 15 the base category was created for a relatively large proportion of patients compared to 16 other age groups. Patients were thereby splitting by age group to create a stable base 17 for analysis. Patient age was missing for n=632 of all individual patients in the dataset 18 and n=10 observations for initial episodes of care for URTI but n=0 observations for UTI. 19 A category for missing age was created for URTI for descriptive analyses but these 20 21 observations were excluded from multivariable analyses.

22

For URTI, with children having the highest incidence, the base reference category was
eventually selected as 0-8 years, as follows:

- 25 1. 0-8 years
- 26 2. 9-21 years
- 27 3. 22-34 years
- 28 4. >=35 years
- 29 5. Missing.
- 30

For UTI, however, incidence was highest as expected in adult women. Therefore, patient age groups were selected in decreasing age for the categorical patient age variable for UTI, as follows:

34 1. >= 45 years

- 1 2. 16-44 years
- 2 3. 6-15 years
- 3 4. 0-5 years
- 4

For UTI, 'adults' were defined to be sixteen years and over. It was considered relevant
that, particularly in light of female anatomy, sexual activity is relevant to UTI and it was
therefore considered relevant to include sixteen years and over rather than eighteen
years and over, which is the usual definition.

9

10

B.4.2

Patient gender

11

For UTI, as the strong majority of patients were women. For stability, female gender
was selected as the base/reference category in the assignment of binary patient gender.
The same patient gender variable was used for URTI.

- 15
- 16 **B.4.3**

.3 Patient allergy labels

17

Allergies to antibiotics was considered highly relevant to which antibiotic a patient 18 receives, particularly in the context of penicillins, which are most relevant to the 19 antibiotics recommended for URTIs. A higher-line antibiotic may be appropriate to 20 21 prescribe for a patient who is allergic to the first-line option. This was taken into account during analysis. Allergy information contained in character strings was searched through 22 23 for different penicillin-class antibiotics including brand names This was used to create binary discrete indicator of whether the patient ever had a history of penicillin sensitivity, 24 25 and otherwise negative if missing.

26

27 **B.4.4** Acute rheumatic fever

28

Acute rheumatic fever is a serious side effect from Group A streptococcus infection, and is a possible complication of acute pharyngitis, and the main justification for any antibiotic prescription to a patient presenting with pharyngitis, should the patient have risk factors predisposing them to Group A streptococcus. Prescribing of an antibiotic for pharyngitis is also perfectly justified for patients with acute rheumatic fever, a history of rheumatic heart disease, or current scarlet fever. As such, it was considered useful to know which patients had these conditions. A binary discrete variable was there created
as an indicator of this by string searching the relevant diagnoses.

3

B.4.5 Primary health network

4 5

A categorical variable was created for which PHN the patient's registered/provided
address fell within. These were coded into a discrete categorical variable with five levels,
as follows:

- 9 Perth North PHN
- 10 Perth South PHN
- Country WA PHN
- Any Interstate PHN
- Missing
- 14

15 **B.4.6**

16

There were concerns regarding the binary variable for self-identification of Aboriginal and Torres Strait Islander peoples, for which a substantial proportion of patients were missing/unknown status. The variable was poorly recorded, and this is known to be a common occurrence (21,22). In this situation, positive status recorded in only 0.01% of patients and information missing in 45% of all patients. This variable was not used in the analysis.

Patient indigenous status

23

24 **B.4.7 Patient smoking status**

25

There were also concerns regarding the smoking status (categorical) variables, for which a substantial proportion of patients were missing/unknown status. This variable was not used in the analysis.

29

1 **B.4.8 Pat**

Patient socio-economic disadvantage

2

Ordinal categorical data were created from SEIFA IRSAD Disadvantage deciles,
recoded into quintiles with the base/reference category being most disadvantaged,
which has higher numbers than least disadvantaged.

6

SEIFA IRSD Deciles for Disadvantage received were recoded into quintiles, to ensure that there were no problems encountered with numbers too small to use for modelling purposes in some categories. Additionally, the reference/base quintile was amended to the least disadvantaged category, with occurs more frequently, i.e. larger numbers and therefore more stable for analysis than the alternative of using most disadvantaged which is typically rare corresponding to small numbers.

- 13
- quintile 1 is the least disadvantaged (reference/base)
- 14
- quintile 5 is the most disadvantaged
- 15

Also look at your category patient rurality - again make sure that capital city (or whatever
the metropolitan category is) is the baseline - not very remote.

18

A binary, discrete variable was then created as an indicator of disadvantage, including
the top two most disadvantaged quintiles (i.e. top 40% most disadvantaged) as positive
for practice disadvantage indicator.

22

Although provided with patient SEIFA and practice SEIFA and rurality for both, patient
 SEIFA and practice rurality, were utilised instead of patient residential address-based
 variables. The reason for this was that there are so few practice variables available.

26

27 **B.4.9** Patient rurality

28

Ordinal categorical data were created from ARIA by postcode deciles, recoded into quintiles with the base/reference category being most remote, which has higher numbers than least remote. ARIA Deciles received were recoded into quintiles, to ensure that there were no problems encountered with numbers too small to use for modelling purposes in some categories. Additionally, the reference/base quintile was amended to the least disadvantaged category, with occurs more frequently, i.e. larger
numbers and therefore more stable for analysis than the alternative of using most 1 2 disadvantaged which is typically rare corresponding to small numbers. 3 • quintile 1 is the least remote (reference/base) • quintile 5 is the most remote 4 5 A binary, discrete variable was then created as an indicator of remoteness, including 6 the top two most remote quintiles (i.e. top 40% most remote) as positive for practice 7 8 disadvantage indicator. 9 Although provided with patient SEIFA and practice SEIFA and rurality for both, patient 10 11 SEIFA and practice rurality, were utilised instead of patient residential address based variables. The reason for this was that there are so few practice variables available. 12 13 **B.4.10** Cardiovascular disease 14 15 16 A binary, dichotomous variable was created to include patients with MedicineInsightprovided flags for patients with any history of cardiovascular disease, including coronary 17 heart disease (see below), peripheral vascular disease, carotid stenosis, renal artery 18 stenosis, heart failure, various cardiovascular disease procedures (see below), as well 19 as any coronary heart disease related activity, such as a heart disease-related care 20 plan, review, script or rehabilitation. 21 22 Coronary heart disease included: 23 ACUTE MYOCARDIAL INFARCTION 24 AMI 25 AMI (ACUTE MYOCARDIAL INFARCTION) 26 ٠ ANGINA 27 ANGINA PECTORIS 28 ANGINA PECTORIS – PRINZMETAL 29 30 ANGINA PECTORIS – UNSTABLE ANTERIOR MYOCARDIAL INFARCT 31 • ANTEROLATERAL MYOCARDIAL INFARCT 32 ATHEROSCLEROTIC HEART DISEASE 33 •

1	BLOCKAGE CORONARY ARTERY
2	CORONARY ARTERY DISEASE
3	CORONARY ARTERY DISEASE
4	CORONARY ARTERY SPASM
5	CORONARY HEART DISEASE
6	CORONARY INSUFFICIENCY
7	CORONARY OCCLUSION
8	HEART ATTACK
9	• IHD
10	IHD (ISCHAEMIC HEART DISEASE)
11	INFERIOR MYOCARDIAL INFARCTION
12	ISCHAEMIC HEART DISEASE
13	• MI
14	MYOCARDIAL DAMAGE
15	MYOCARDIAL INFARCTION
16	MYOCARDIAL INFARCTION – ANTERIOR
17	MYOCARDIAL INFARCTION – ANTEROLATERAL
18	MYOCARDIAL INFARCTION – INFERIOR
19	MYOCARDIAL INFARCTION – POSTERIOR
20	MYOCARDIAL INFARCTION – SUBENDOCARDIAL
21	MYOCARDIAL INFARCTION – SUPERIOR
22	MYOCARDIAL INFARCTION – SUPERIOR
23	MYOCARDIAL INFARCTION - WITH ST ELEVATION
24	MYOCARDIAL INFARCTION - WITHOUT ST ELEVATION
25	MYOCARDIAL INFARCTION, ANTERIOR
26	MYOCARDIAL INFARCTION, ANTEROLATERAL
27	MYOCARDIAL INFARCTION, INFERIOR
28	MYOCARDIAL INFARCTION, POSTERIOR
29	MYOCARDIAL INFARCTION, SUBENDOCARDIAL
30	MYOCARDIAL INFARCTION, SUPERIOR
31	MYOCARDIAL INSUFFICIENCY
32	OCCLUSION - CORONARY ARTERY
33	 OCCLUSION, CORONARY ARTERY

1	POSTERIOR MYOCARDIAL INFARCT
2	PREINFARCTION SYNDROME
3	PRINZMETAL ANGINA
4	STEMI (ST-ELEVATION MYOCARDIAL INFARCTION)
5	SUBENDOCARDIAL INFARCT
6	SUBENDOCARDIAL MYOCARDIAL INFARCT
7	SUPERIOR MYOCARDIAL INFARCT
8	UNSTABLE ANGINA
9	UNSTABLE ANGINA - HIGH RISK
10	UNSTABLE ANGINA - LOW RISK
11	UNSTABLE ANGINA - MODERATE RISK
12	VARIANT ANGINA
13	
14	Specific cardiovascular disease procedure terms including:
15	Angioplasty - coronary
16	Coronary artery surgery
17	 Arterial stent - Coronary artery, not drug-eluting
18	Coronary artery endarterectomy
19	 Arterial stent - Coronary artery, drug-eluting
20	 Angioplasty - coronary (with stent)
21	Bypass - coronary
22	• CABG
23	 Coronary angioplasty, bare metal stent
24	Coronary artery endarterectomy
25	 Coronary angioplasty, drug eluting stent
26	 Coronary angioplasty with stent
27	Angioplasty, coronary
28	
29	

1 **E**

2

B.4.11 Mental health condition

Flags for patients with mental health conditions, as well as drug and alcohol addiction
were combined into a single, if any, binary, dichotomous variable. Mental health
conditions included bipolar affective disorder, anxiety, depression, schizophrenia.

- 6
- 7 B.5 Consultation-related variables
- 8

9 B.5.1 Patients with multiple, independent episodes of urinary tract 10 infection

11

A binary discrete variable was created for whether the individual patient appeared with more than one episode of UTI in the dataset, as multiple episodes of infection in a single patient may feasibly predispose a treating GP to prescribing a non-first-line antibiotic (ref). A count variable was also created for the number of UTI episodes a patient presented with during the study period.

17

There were 25,334 initial episodes of care for uncomplicated UTI, corresponding to 23,173 individual patients seen by 958 different providers during the period. The number of initial episodes ranged from one and fourteen, with 84% having only one consultation, 7% two, 7% three, 1% four, and <1% in excess of four consultations.

22

Of the 16% of patients with multiple consultations, for the strong majority, there were similar proportions of antibiotic choices prescribed regardless of number of UTI consults patients had. Prescribing rates were also similar in this context. For all patients, the mean line of choice prescribed was between 1.88 to 1.92 for patients with 1 to 4 consults, representing 99% of consults.

28

As there appeared to be no remarkable differences in choice of antibiotic prescribing (and also whether an antibiotic prescribed in the first place) based on the specific number of UTI consultations, for the strong majority of episodes, the decision was made to create a binary discrete indicator of patients with more than one initial episode of care for UTI in the period of interest.

B.5.2 Temperature-related variables 1 2 Observation variables within the observation dataset appeared manually entered by 3 GPs as free text, resulting in many different types of measurements being entered. 4 5 6 B.5.2.1 Whether temperature testing occurred 7 A binary discrete variable was also created for whether or not a temperature reading 8 was recorded during the consultation. This therefore included any reasonable/non-9 10 erroneous temperature recording. 11 B.5.2.2 Fever indicator 12 13 Temperature readings to indicate fever and higher likelihood of bacterial infection 14 wherein antibiotics are more likely to be warranted. A binary discrete variable was 15 created for fever with temperature readings of at least 37.5 degrees Celsius. This was 16 used in situations where the classification of likely/unlikely prescribing relied in part upon 17 18 fever being recorded. Please note rhinosinusitis used high fever (see below). 19 20 B.5.2.3 High fever indicator 21 22 However, the guidelines indicate 39 degrees Celsius is suggestive of more serious acute rhinosinusitis, so a further binary discrete variable was coded to allow for this. 23 24 This variable was used in classifying likely necessary from likely unnecessary antibiotic prescribing for rhinosinusitis. 25 26 27 **B.5.3** Urine dipsticks for urinary tract infection testing 28 Observation variables within the observation dataset appeared manually entered by 29 30 GPs as free text, resulting in many different types of measurements being entered. 31 Blood B.5.3.1 32 33 34 28% of the total 20,012 relevant dipsticks were positive for blood.

1						
2	<u>B.5.3.2</u>	Leucocyte esterase				
3						
4	24% of the 1	7,852 relevant dipsticks were positive for leucocyte esterase.				
5						
6	<u>B.5.3.3</u>	Nitrites				
7						
8	6% of the 16	5,473 total relevant dipsticks were positive for nitrites.				
9						
10	<u>B.5.3.4</u>	Dipstick tested status				
11						
12	A binary dis	crete variable was created for whether the patient underwent any form of				
13	urine dipstic	k testing during the consultation. This included blood, leucocyte esterase,				
14	or nitrite urin	ne dipsticks.				
15						
16	<u>B.5.3.5</u>	Any positive dipstick result				
17						
18	A positive re	esult for any of the three urine dipsticks (blood, leucocyte esterase and				
19	nitrites) was	coded as positive result, and otherwise negative.				
20						
21	B.6	Predictors analyses				
22						
23	B.6.1	Explanatory data analysis				
24						
25	Following the	e completion of descriptive statistics, chi-squared tests were performed, and				
26	correlation b	between variables was calculated. The latter was used to inform choices				
27	regarding w	hich explanatory variables might not be suitable to model together. The				
28	examination of clustering also occurred. This included clarification of the size of clusters					
29	of numbers of patients seen by single providers, and providers nested within and across					
30	practices.					
31						
32	The objectiv	ve was to identify variables associated with likely inappropriate decisions				
33	within all l	JRTI diagnoses, and for likely unnecessary prescribing among all				

34 prescriptions for URTI. For URTI and UTI, the objectives included identifying variables

associated with increasing choice of antibiotic prescribed, non-first-line prescribing, and
 repeat prescribing for initial presentations of the condition.

3

B.6.2 Linking records

4 5

6 The MedicineInsight data received were de-identified at the patient, provider / GP and 7 practice levels but is equipped with unique IDs at each level to allow for the linking of records. These variables were referred to as practice ID, provider ID, and practice ID. 8 9 The datasets for each diagnostic condition were then linked with the antibiotic prescription dataset. Following the linkage of diagnosis and prescription data, antibiotic 10 11 prescriptions linked with a single diagnosis containing more than one infectious condition, such as, "OM / cellulitis leg", or prescriptions linked to two, separate infectious 12 diagnoses, were excluded as one could not be certain which diagnosis the antibiotic 13 was intended for (23,24). The pre-prepared datasets of presentation, encounters, 14 patient comorbid conditions, clinical observations, pathology, and provider and practice 15 information data were then merged by matching the consultation date, patient ID, 16 provider ID and practice ID in each row. 17

18

19 **B.6.3** Standard of assessment

20

The standard used for assessment of prescribing was version 15 of the guidelines published in 2014 (2). During the study period 2012 to mid-2017, there were two different versions of these guidelines available to clinicians, as a new version was published during this time (2,4). It was considered difficult to ascertain when clinician use of the previous version published in 2010 ceased and the time taken for the promulgation of the 2014 guidelines (2,4). For this reason, it was not considered feasible to include multiple versions of the guidelines within the analysis.

28

A dummy variable was created for cefalexin prescriptions, for which repeats may be required to complete a single guideline-recommended course of medication.

31

Although the guidelines use an age cut-off of six months, patient age data was only available in years (2).

B.6.4 Base model inclusion

3 B.6.4.1 Base model inclusion for upper respiratory tract infection

Table B-4: Base model inclusion for each model for upper respiratory tract infection

Variable Included in Base Model	Likely Inappropriate Decision	Likely Unnecessary Prescribing	Ordinal Line of Antibiotic Prescribed	Binary Non- first-line Prescribing	Binary Repeat Positive Antibiotic Prescribing
Patient age group	Y	Y	Y	Y	Y
Patient gender	Y	Υ	Y	Y	Υ
Patient comorbid condition status	Y	Y	Y	Y	Y
Patient mental health condition status	Y	Y	Y	Y	Y
Patient socioeconomic disadvantage status	Y	Y	Y	Y	Y
Patient government concession status	Y	Y	Y	Y	Y
PHN for patient's address	Y	Υ	Y	Y	Y
Measure of patient remoteness	Y	Υ	Y	Y	Y
Patient penicillin allergy label	Υ	Υ	Y	Υ	Υ
Day of the week or weekend status	Υ	Υ	Y	Υ	Y
Temperature testing status	Υ	Υ	Y	Υ	Υ
Whether reason for prescribing was recorded	N	Y	Y	Y	Y
Repeat prescription status	Ν	Y	Y	Y	Y
Dummy variable for cephalexin prescriptions	Ν	Y	N	Z	Y
Dummy variable for annual influenza season ¹³	Y	Y	Y	Y	Y
URTI condition ¹⁴	Ν	Ν	Y	Υ	Y
Likely unnecessary prescribing variable	Ν	n/a	Y	Y	Y
Choice of antibiotic prescribing variable	Ν	Y	n/a	n/a	Y
Practice size	Υ	Y	Y	Υ	Υ

¹³ All models included dummy variables for annual influenza seasons to allow seasonal effects (25-27).

¹⁴ URTI condition was sometimes too closely linked to the outcome to permit inclusion.

1 <u>B.6.4.2</u>

Base model inclusion for urinary tract infection

2 3

Table B-5: Base model inclusion for each model for urinary tract infection

Variable Included in Base Model	Ordinal Line	Binary Non-	Binary Repeat
	of Antibiotic	first-line	Positive Antibiotic
	Prescribed	Prescribing	Prescribing
Patient age group	Y	Υ	γ
Patient gender	Y	Υ	γ
Patient comorbid condition status	Y	Y	Y
Patient mental health condition status	Y	Υ	γ
patient socioeconomic disadvantage status	Y	Υ	γ
Patient government concession status,	Y	Y	Y
PHN for patient's address	Y	Υ	γ
Measure of patient remoteness	Y	Y	Y
Patient penicillin sensitivity status	Y	Y	Y
Day of the week or weekend status	Y	Υ	Y
Urine dipstick testing status	Y	Υ	γ
Culture testing status	Y	Y	Y
Temperature testing status	Y	Υ	Y
Whether reason for prescribing was recorded	Y	Υ	γ
Practice size	Y	Y	Y
Repeat prescription status	Y	Υ	n/a
Dummy variable for cephalexin prescriptions	Y	Y	Y
Dummy variable for annual influenza season ¹⁵	Υ	Υ	Υ
Ordinal choice of antibiotic prescribed variable	n/a	n/a	Υ

4

5

¹⁵ All models included dummy variables for annual influenza seasons to allow seasonal effects (25-27).

1 **B.6.5 Modeling considerations**

2

For each model, the intention was to include both random intercept and random slope 3 at each level. However, it was found that due to limitations of processing power with this 4 large dataset, random slopes was not included in the model, only random intercepts. 5 6 Random intercepts were included, where possible, for both individual practice and individual provider. When both levels could not be accounted for, a random intercept for 7 8 the unique combination of practice and provider was used. While in theory, unique 9 patient within unique provider effect would be preferable to provider to practice effect, the clusters were too small such that unique provider within unique practice models 10 were settled upon. The process involved weighing up the size of effect of including three 11 levels and whether there is a notable difference with only two. 12

13

Furthermore, the intention was to include random intercepts for practice-related variables within the practice level. However, resulting models were not ideal and the decision was made to not include random intercepts for variables at higher levels than the individual patient. For example, practice size is a variable relating to the practice level and should ideally be included at that level. However, the inclusion of a random intercept for practice size resulted in an effect so small that CIs were unable to be calculated.

21

Likelihood ratio tests were used to aid selection of a suitable structure for the random 22 effects model. Starting from the base model, modelling involved manually removing 23 variables one at a time to examine the effect of that variable on the dependent/outcome 24 variable of inappropriate prescribing. Care was to be taken regarding exclusion of 25 potential explanatory variables, not purely based on statistical significance but also upon 26 clinical and policy significance. The magnitude and precision of effect also needed 27 consideration, using percentage change of coefficients and standard error, and 28 29 confidence intervals for precision. Reduction in Bayesian Information Criterion was also used as an indicator of improvement in model selection. whilst also checking for feasible 30 size of effect, standard error, and statistical significance. Interactions were tested for 31 effect modification such as patient age and gender, where subgroup analysis indicates 32 sufficient power and there is a plausible reason for considering potential interaction. 33 Routine heteroskedasticity tests could not be performed for these non-linear data. 34

Base models for each condition group and outcome included all potential explanatory
variables of interest, from which models for the random effects structure was derived
(28).

5

6 Due to the small numbers of URTI prescriptions identified as likely necessary, it was 7 unfeasible to model the response variables within the denominator of likely appropriate 8 decisions, (and likely unnecessary prescriptions within likely inappropriate decision). As 9 a result, the denominator of all antibiotics prescribed for URTI, for both likely necessary 10 and likely unnecessary prescriptions, was selected

11

12 To compare the results of this research with models without allowance for 13 heterogeneity, equivalent, multivariable logistic regression models without random 14 effects were developed for all outcomes.

15

16 **B.7** Trends analyses

17

Additionally, there was also a focus on trends in antibiotic prescribing for, as well as trends in the individual antibiotic prescribed (29-34). For UTI only, outcomes included whether urine dipstick testing was recorded. The objective was to identify outcomes demonstrating both statistically significant and clinically meaningful change. **Table B-6** and **Table B-7** provide additional detail regarding the numerators and denominators for trends analyses for URTI and UTI, respectively.

24

Table B-6:Details of numerators and denominators for outcome rates for trends analysis for
upper respiratory tract infection.

Prescribing Outcome Rate	Description of Numerator and Denominator			
Likely Unnecessary antibiotic Prescribing rate	The sum of likely unnecessary antibiotic prescriptions for the particular condition or condition group, over sum of all antibiotic prescriptions for initial presentations of the same condition / condition group.			
Overall Antibiotic Prescribing rate	The sum of all antibiotic prescriptions for patients with initial presentations of the particular condition / condition group, over the sum of all patients with initial diagnoses of the same condition/condition group.			
Second-line antibiotic prescribing rate	The sum of second-line antibiotic prescriptions for patients with initial presentations of the particular condition / condition group, over the sum of all antibiotics prescribed to patients with initial presentations of the same condition / condition group.			
Prescribing of antibiotic Not Recommended in Guidelines for the condition/condition group.	The sum of antibiotic prescriptions not recommended in the Guidelines for the condition given to patients with initial presentations of the particular condition / condition group, over the sum of all antibiotics prescribed to patients with initial presentations of the same condition/condition group. This is also referred to as the not recommended prescribing rate.			
Non-first-line antibiotic prescribing	The sum of non-first-line antibiotic prescriptions for patients with initial presentations of the particular condition / condition group, over the sum of all antibiotics prescribed to patients with initial presentations of the same condition/condition group.			
Repeat(s) issued on antibiotic prescription	The sum of antibiotic prescriptions with one or more repeats given to patients with initial presentations of the particular condition / condition group, over the sum of all antibiotics prescribed to patients with initial presentations of the same condition/condition group.			

Table B-7:Details of numerators and denominators for outcome rates for trends analysis
for urinary tract infection.

Prescribing Outcome Rate	Description of Numerator and Denominator			
Overall antibiotic prescribing rate	The sum of all antibiotic prescriptions for patients (within patient group, where relevant) with initial presentations of UTI, over the sum of all patients (within same patient group, where relevant) with initial diagnoses of UTI. For example, the antibiotic prescribing rate for women used the sum of antibiotic prescriptions provided for women with initial presentations of UTI, over the sum of all women with initial presentations of UTI, over the sum of all women with initial presentations of UTI.			
Prescribing of second-line antibiotic agent	The sum of second-line antibiotic prescriptions for patients (within patient group, where relevant) with initial presentations of UTI, over the sum of all antibiotics prescribed to patients (within same patient group, where relevant) with initial presentations of UTI.			
Prescribing of third-line antibiotic agent	The sum of third-line antibiotic prescriptions for patients (within patient group, where relevant) with initial presentations of UTI, over the sum of all antibiotics prescribed to patients (within same patient group, where relevant) with initial presentations of UTI.			
Prescribing of antibiotic not recommended in guidelines	The sum of antibiotic prescriptions not recommended in the Guidelines given to patients (within patient group, where relevant) with initial presentations of UTI, over the sum of all antibiotics prescribed to patients (within same patient group) with initial presentations of UTI. This is also referred to as the not recommended prescribing rate.			
Non-first-line antibiotic prescribing	The sum of non-first-line antibiotic prescriptions for patients (within patient group) with initial presentations of UTI, over the sum of all antibiotics prescribed to patients (within same patient group) with initial presentations of UTI.			
Repeat(s) issued on antibiotic prescription	The sum of antibiotic prescriptions positive for one or more repeats given to patients (within patient group) with initial presentations of UTI, over the sum of all antibiotics prescribed to patients (within same patient group) with initial presentations of UTI.			
Urine dipstick testing occurring	The sum of patients (within patient group, where relevant) with initial presentations of UTI receiving urine dipstick testing during consultation, over the sum of all patients (within same patient group, where relevant) with initial presentations of UTI.			

APPENDIX C - APPENDICES TO THE PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR UPPER RESPIRATORY TRACT INFECTION CHAPTER (CHAPTER 4)



C.1 Clustering of patients and providers for initial presentations of upper respiratory tract infection



C.2

Antibiotic prescriptions

3 4 5

Table C-1: Frequency table of antibiotic class by Anatomical Therapeutic Chemical classification (35), for systemic antibiotics prescribed for initial presentations of upper respiratory tract infection

Anatomical Therapeutic Chemical Class of Antibiotic	Frequency	Percent	Cumulative Percent
Beta-lactamase resistant penicillins	110	0.21	0.21
Beta-lactamase sensitive penicillins	10,218	19.59	19.8
Combinations of penicillins, incl. beta-lactams	11,843	22.7	42.5
Combinations of sulfonamides and trimethoprim	420	0.81	43.3
First-generation cephalosporins	5,701	10.93	54.23
Fluoroquinolones	112	0.21	54.44
Lincosamides	43	0.08	54.53
Macrolides	7,251	13.9	68.42
Nitrofuran derivatives	2	0	68.43
Other Aminoglycosides	6	0.01	68.44
Penicillins with extended spectrum	14,722	28.22	96.66
Second-generation cephalosporins	1,186	2.27	98.93
Tetracyclines	506	0.97	99.9
Third-generation cephalosporins	4	0.01	99.91
Trimethoprim and derivatives	47	0.09	100
Total	52,171	100	

6

Table C-2:Frequency table of patient age group by likely appropriate / likely inappropriate
decision status for initial presentations of upper respiratory tract infection

	Likely Appropriate Decision	Likely Inappropriate Decision	Excluded	Total
Patient Age Group				
0-8 yrs, ref	22,252	9,266	125	31,643
	70.32	29.28	0.4	100
9-21 yrs	11,550	7,633	195	19,378
	59.6	39.39	1.01	100
22-34 yrs	17,527	14,094	210	31,831
	55.06	44.28	0.66	100
35+ yrs	16,314	13,206	118	29,638
	55.04	44.56	0.4	100
Missing	4	2	240	246
	1.63	0.81	97.56	100
Total	67.647	44,201	888	112,736
	60	39.21	0.79	100

Кеу
frequency
row percentage

2 Table C-3:

 Frequency table of medicine active ingredients for systemic antibiotics prescribed but not recommended in the guidelines for the condition it was prescribed for, for initial presentations of upper respiratory tract infection

Medicine active ingredient	Frequency	Percent	Cumulative Percent
	,		
AMOXICILLIN	3002	14.66	14.66
AMOXICILLIN WITH CLAVULANIC ACID	3081	15.04	29.70
AZITHROMYCIN	648	3.16	32.86
CEFACLOR	937	4.57	37.44
CEFALEXIN	3484	17.01	54.45
CEFTRIAXONE	4	0.02	54.46
CEFUROXIME	183	0.89	55.36
CIPROFLOXACIN	107	0.52	55.88
CLARITHROMYCIN	1726	8.43	64.31
CLINDAMYCIN	43	0.21	64.52
DICLOXACILLIN	4	0.02	64.54
DOXYCYCLINE	389	1.90	66.44
ERYTHROMYCIN	1488	7.26	73.70
FLUCLOXACILLIN	102	0.50	74.20
GENTAMICIN	4	0.02	74.22
MINOCYCLINE	25	0.12	74.34
NITROFURANTOIN	2	0.01	74.35
NORFLOXACIN	4	0.02	74.37
PHENOXYMETHYLPENICILLIN	605	2.95	77.32
PROCAINE BENZYLPENICILLIN (PROCAINE PENICILLIN)	1155	5.64	82.96
ROXITHROMYCIN	3068	14.98	97.94
TOBRAMYCIN	2	0.01	97.95
TRIMETHOPRIM	47	0.23	98.18
TRIMETHOPRIM WITH SULFAMETHOXAZOLE	373	1.82	100.00
Total	20,483	100	

2

Table C-4:Frequency table of upper respiratory tract infection condition by likely necessary
/ likely unnecessary prescribing status, for initial presentations of upper
respiratory tract infection

3 4

	Likely Appropriate Decision	Likely Inappropriate Decision	Excluded	Total
URTI Condition				
Rhinosinusitis	1,944	17,884	236	20,064
	9.69	89.13	1.18	100
Pharyngitis/Tonsillitis	2,622	17,408	470	20,500
	12.79	84.92	2.29	100
Acute Otitis Media	2,698	8,656	0	11,354
	23.76	76.24	0	100
Influenza/ILI	0	253	0	253
	0	100	0	100
Total	7,264	44,201	706	52,171
	13.92	84.72	1.35	100

Кеу
frequency
row percentage

Note: 253 cases of influenza/ILI receiving antibiotic prescriptions were excluded from model of likely

6 unnecessary prescribing as the outcome is invariable likely unnecessary prescribing.

7

5

8 9
 Table C-5:
 Frequency table of ordinal choice of antibiotic prescribed by upper respiratory tract infection condition, for initial presentations of upper respiratory tract infection

	First-line	Second-line	Recommended	Excluded	Total	
URTI Condition						
Rhinosinusitis	6,504	5,049	8,511	0	20, 00 4	
	32.42	25.16	42.42	0	100 13	
Pharyngitis/Tonsillitis	8,683	0	11,817	0	20, <u></u> 50	
	42.36	0	57.64	0	100 15	
Acute Otitis Media	5,317	3,698	2,339	0	11,354	
	46.83	32.57	20.6	0	100 17	
Influenza/ILI	0	0	0	253	2 5 3	
	0	0	0	100	100	
Total	20,504	8,747	22,667	253	52,171	
	39.3	16.77	43.45	0.48	100	

Key frequency row percentage 1 Table C-6:

C-6: Frequency table of patient primary health network by ordinal choice of antibiotic prescribed, for patients with initial presentations of acute pharyngitis / tonsillitis

1	, 1	I	
	First-line	Not Recommended	Total
Patient Primary Health Network			
Perth North	3,427	5,619	9,046
	37.88	62.12	100
Parth South	2 332	3 770	6 102
Fertin South	2,552	3,770	0,102
	38.22	61.78	100
Country WA	2,702	2,149	4,851
	55.7	44.3	100
	01	114	105
nterstate	81	114	195
	41.54	58.46	100
Vissing	141	165	306
	46.08	53.92	100
T - 4 - 1	0.000	44.047	20 500
Iotal	8,683	11,817	20,500
	42.36	57.64	100

C.3

Antibiotic prescriptions for influenza / influenza-like illness

 Table C-7:
 Frequency table of antibiotic active ingredients prescribed for initial presentations of influenza / influenza-like illness

Medicine active ingredient	Frequency	Percent	Cumulative Percent
Amoxicillin	53	20.95	20.95
Amoxicillin with clavulanic acid	60	23.72	44.66
Azithromycin	13	5.14	49.8
Cefaclor	2	0.79	50.59
Cefalexin	24	9.49	60.08
Cefuroxime	5	1.98	62.06
Ciprofloxacin	1	0.4	62.45
Clarithromycin	47	18.58	81.03
Dicloxacillin	1	0.4	81.42
Doxycycline	9	3.56	84.98
Erythromycin	10	3.95	88.93
Flucloxacillin	2	0.79	89.72
Phenoxymethylpenicillin	9	3.56	93.28
Procaine benzylpenicillin (procaine penicillin)	1	0.4	93.68
Roxithromycin	15	5.93	99.6
Trimethoprim with sulfamethoxazole	1	0.4	100
Total	253	100	

C.4 Repeats issued on antibiotic prescription status

4 C.4.1 Repeats issued on all antibiotic prescriptions for upper respiratory 5 tract infection

7 Table C-8:

Frequency table of whether repeats were issued on prescription by likely unnecessary / necessary prescribing status, for initial presentations of upper respiratory tract infection

	Likely Necessary	Likely Unnecessary		11
	Prescribing	Prescribing	Excluded	Total
Repeat issued on prescription				
Negative	4,539	30,142	480	35,161
	12.91	85.73	1.37	100
Positive	2,725	13,806	226	16,757
	16.26	82.39	1.35	100
Excluded	0	253	0	253
	0	100	0	100
Total	7,264	44,201	706	52,171
	13.92	84.72	1.35	100

Кеу	
frequency	
row percentage	

 Table C-9:

 Frequency table of whether repeats were issued on prescription by ordinal choice of antibiotic prescribed, for initial presentations of upper respiratory tract infection

	Not								
	First-line	Second-line	Recommended	Excluded	Total				
Repeat issued on prescription									
Negative	16,480	4,124	14,557	0	35,161				
	46.87	11.73	41.4	0	100				
Positive	4,079	4,622	8,056	0	16,757				
	24.34	27.58	48.08	0	100				
Excluded	0	0	0	253	253				
	0	0	0	100	100				
Total	20,559	8,746	22,613	253	52,171				
	39.41	16.76	43.34	0.48	100				

Key frequency row percentage

Table C-10:

Frequency table of patient age group and whether repeats were issued on prescription, for initial presentations of upper respiratory tract infection

	Repeat iss	ued on Presc	ription	
	Negative	Positive	Excluded	Total
Patient Age Group				
0-8 yrs,	8,912	4,422	21	13,355
	66.73	33.11	0.16	100
0.21 yrs	5 710	2 250	20	0 009
9-21 915	5,719	5,250	59	9,008
	63.49	36.08	0.43	100
22-34 yrs	11,349	4,065	70	15,484
	73.3	26.25	0.45	100
25	0.020	4.021	122	14.004
35+ yrs	9,030	4,931	123	14,084
	64.12	35.01	0.87	100
Missing	151	89	0	240
	62.92	37.08	0	100
	a = <i>i ai</i>		0.50	
Iotal	35,161	16,757	253	52,171
	67.4	32.12	0.48	100

 Table C-11:Frequency table of prescriptions issued with repeats for initial presentations
of influenza / influenza-like illness, by active ingredient

Medicine active ingredient	Frequency	Percent	Cumulative Percent
Amoxicillin	10	11.49	11.49
Amoxicillin with clavulanic acid	27	31.03	42.53
Azithromycin	1	1.15	43.68
Cefaclor	1	1.15	44.83
Cefalexin	10	11.49	56.32
Cefuroxime	2	2.3	58.62
Clarithromycin	19	21.84	80.46
Doxycycline	3	3.45	83.91
Erythromycin	4	4.6	88.51
Roxithromycin	10	11.49	100
Total	87	100	

1C.4.2Repeats issued on cefalexin prescriptions for treatment of acute2pharyngitis / tonsillitis

Of scripts for cephalexin with repeats issued for URTI, with the denominator of antibiotics within the ordinal choice of antibiotic prescribed model, 897 were for acute pharyngitis/tonsillitis (43%). Cefalexin is an option listed in the Guidelines for acute pharyngitis/tonsillitis only (for penicillin immediate hypersensitivity patients), for which the recommended course is 1g, 12hrly for 10 days, totalling a 20g course. Children are recommended 25mg/kg up to 1g.

10

3

All 29 patients receiving 250mg strength and all 436 adults receiving 500mg did require repeats (**Table C-12**). Of liquid formulation, 10 patients required several repeats, 1 patient receiving 4 bottles did not require a repeat. Therefore, all but one adult of the 476 total did require repeats for cephalexin scripts for acute pharyngitis/tonsillitis.

- 15
- 16Table C-12:Frequency table of medicine quantity and medicine strength, of cefalexin17prescriptions with repeats present, for adults with initial presentations of18pharyngitis

	Medicine quantity						
Medicine strength	1	100mL	20	4	40	6	Total
250mg	0	0	28	0	1	0	29
250mg/5mL	5	5	0	1	0	0	11
500mg	0	0	435	0	0	1	436
Total	5	5	463	1	1	1	476

19

20 There were 421 children under 18 years receiving cephalexin scripts with repeats for

21 pharyngitis, as follows in **Table C-13**:

22

23 24

25

26

Table C-13: Frequency table of medicine quantity and medicine strength, of cefalexin prescriptions with repeats present, for patients under eighteen years receiving cefalexin prescriptions with repeats for initial presentations of pharyngitis / tonsillitis

	Medicine quantity						
medicine_strength	1	100mL	100mL*3	20	Total		
125mg/5mL	84	11	0	0	95		
250mg	0	0	0	29	29		
250mg/5mL	235	38	1	0	274		

500mg	0	0	0	23	23	
Total	319	49	1	52	421	

2 For children under eight years of age, the cefalexin medicine strength and quantity are

displayed in **Table C-14** below. All young children under 8 are likely to have required
repeats.

Frequency table of medicine quantity and medicine strength, of cefalexin

5

-

6 7 8 Table C-14:

prescriptions with repeats present, for children under eight years of age for initial presentations of pharyngitis / tonsillitis

Medicine strength	1	Medicine qua 100mL	ntity 20	Total
125mg/5mL 250mg 250mg/5mL	66 0 103	11 0 19	0 1 0	77 1 122
Total	169	30	1	200

9

10 For children aged 9-16 years, as depicted in **Table C-15** below, it is also possible that

all these children required repeats, particularly for liquid formulation, up to 4 100mL

12 bottles of 25mg/5mL for maximum 20g course.

13

14 15

16

Table C-15:Frequency table of medicine quantity and medicine strength, of cefalexin
prescriptions with repeats present, for children aged 9-16 years, for initial
presentations of pharyngitis / tonsillitis

Medicine quantity						
Medicine strength	1	100mL	100mL*3	20	Total	
125mg/5mL	14	0	0	0	14	
250mg	0	0	0	27	27	
250mg/5mL	108	16	1	0	125	
500mg	0	0	0	22	22	
Total	122	16	1	49	188	

17

18

19 It is possible that all but 1 patient did require repeats for cephalexin prescriptions for

20 treatment of pharyngitis.

For adults prescribed cephalexin with repeats, the instructions for 500mg varied substantially (**Table C-16**). Although ten days is recommended (and was the most common instruction), durations appearing included five days commonly, as well as seven days and even three days.

- 5
- 6 7

8

Table C-16:Frequency table of medicine instructions, on cefalexin prescriptions of 500mg
strength with repeats present, prescribed to adults with initial presentations of
pharyngitis / tonsillitis

			Cumulative
Medicine instructions	Frequency	Percent	Percent
1 tab four times a day for 5 days	1	0.92	0.92
1 tab tds	2	1.83	2.75
10 days	3	2.75	5.5
1g BD for 10 days	3	2.75	8.26
2 cap 2 times per day for 10 days	1	0.92	9.17
2 cap 2 times per day for 10 days	2	1.83	11.01
2 cap 2 times per day for 10 days	1	0.92	11.93
2 caps bd until finished	1	0.92	12.84
2 capsules bd for 10 days	1	0.92	13.76
2 stat	3	2.75	16.51
2 tablets BD for 10 days	1	0.92	17.43
2 tabs bd for 10 days	2	1.83	19.27
5-7days	1	0.92	20.18
5d	1	0.92	21.1
For 10 days	1	0.92	22.02
For 5 days	4	3.67	25.69
For 5 days, after food	2	1.83	27.52
for 10 days	19	17.43	44.95
for 10/7	1	0.92	45.87
for 3 days	1	0.92	46.79
for 5 days	5	4.59	51.38
for 5 days for acute infections	1	0.92	52.29
for 5 days for throat infection	1	0.92	53.21
for 5 to 7 days	1	0.92	54.13
for 5-10 days	5	4.59	58.72
for 5-7 days	1	0.92	59.63
for 5-7days	1	0.92	60.55
for 7-10 days	1	0.92	61.47
for five days	1	0.92	62.39
for ten days	2	1.83	64.22
gluten and peanut free please	1	0.92	65.14
ii 12 hourly or i 6 hourly 1- days	1	0.92	66.06
ii 12 hourly or i 6 hourly 10 days	1	0.92	66.97
ii bd 10 days	2	1.83	68.81
m.d.u.	15	13.76	82.57
one BD for 7 days	1	0.92	83.49
one bd for 5 days and then one daily	1	0.92	84.4
one aid	8	7.34	91.74
p.c.	1	0.92	92.66
take two to start and then one 6 hourly	1	0.92	93.58

take two to start and then one capsul	2	1.83	95.41
take two to start and then one four t	3	2.75	98.17
twice a day for 10 days	1	0.92	99.08
until all taken.	1	0.92	100
Total	109	100	

- 2 For children receiving 500mg doses, there was notable variation from 5 to 10 days, as
- 3 follows in **Table C-17**.

Table C-17:Frequency table of medicine instructions, on cefalexin prescriptions of 500mg
strength with repeats present, prescribed to children for initial presentations of
pharyngitis / tonsillitis

Medicine instructions	Frequency	Percent	Cumulative Percent
1 tab bd	1	14.29	14.29
10 days	1	14.29	28.57
2.5mLs TDS for 7 days	1	14.29	42.86
For 10days Start with 2 twice a day	1	14.29	57.14
for 5 days	1	14.29	71.43
for 5-7days	2	28.57	100
Total	7	100	

- 8 Similarly for children receiving 250mg doses, there was notable variation from 5 to 10
- 9 days, as seen in **Table C-18**.

 Table C-18:Frequency table of medicine instructions, on cefalexin prescriptions of 250mg
strength with repeats present, prescribed to children with initial presentations
of pharvngitis / tonsillitis

Medicine instructions	Frequency	Percent	Cumulative Percent
2 bd for 10 days	1	14.29	14.29
2 stat	1	14.29	28.57
for 10 days	1	14.29	42.86
For 5 days for acute infections	1	14.29	57.14
for 5-7days	1	14.29	71.43
for 6 days	1	14.29	85.71
until all taken.	1	14.29	100
Total	7	100	

- 1 The variation in instructions for liquid formulations of 250mg or 125mg / 5mL was
- 2 notable with 5-7 days also being common (**Table C-19**).
- 3 4

Ē

Table C-19:Frequency table of medicine instructions on prescriptions for liquid
formulations of cefalexin with repeats present, prescribed children with initial
presentations of pharyngitis / tonsillitis

Medicine instructions	Frequency	Percent	Cumulative Percent
(27 mg/kg/day)	1	0.81	0.81
10ml stat	1	0.81	1.63
10ml to begin	1	0.81	2.44
10mls twice daily	1	0.81	3.25
11.5ml twice daily for 10 days	1	0.81	4.07
15ml twice daily for 10 days	1	0.81	4.88
16 mls bd for 10 days	1	0.81	5.69
200 mg three times per day for one week	1	0.81	6.5
250mg [5mls] six hourly for seven days	1	0.81	7.32
250mg twice daily	1	0.81	8.13
3.5mLs TDS for 7 days	1	0.81	8.94
3ml tds	1	0.81	9.76
4 mls tds	1	0.81	10.57
40 mg/kg/day 11 kilos	1	0.81	11.38
40 mg/kg/day 14 kilos	1	0.81	12.2
4ml tds	1	0.81	13.01
5 days	2	1.63	14.63
5 mL three times daily for 10 days.	1	0.81	15.45
5 mL three times daily for 7 days	-	0.81	16.26
5 5ml four times daily 7 days	-	0.81	17.07
5.85ml twice daily for 10 days	1	0.81	17.89
5-lul	1	0.81	18.7
500mg twice daily for 7 days	1	0.01	19.5
6mls four times a day	1	0.01	20.33
7 days	2	2 44	20.35
7 mls tds	1	0.81	22.70
7 5 mL three times daily for 5 days	1	0.81	23.38
7.5 me times tany for 5 days	1	0.01	24.33
8 ml twice daily for 10 days	1	0.01	25.2
8 5 ml RD for 7 days	1	0.01	20.02
9.7ml 12 hourly for 10 days	1	0.01	20.83
9.7111 12-houry for 10 days	1	0.01	27.04
For E days	0	0.01	20.40
For 5 days	0	0.5	25 77
For 7 days.	1	0.01	35.77
	1	0.01	30.39
V 10 days	1	0.01	57.4 29.21
X 10 days	1	0.81	38.21
for 10 days	1	12.2	39.02 51.22
for 10 days	15	12.2	51.22
for 10 days (21kg)	1	0.81	52.03
for 10 days (OK 10 mis twice a day)	1	0.81	52.85
for 10 days - 1mi for first time	Ţ	0.81	53.00
for 10 days.	1	0.81	54.47
TOT 5 days	10	8.13	62.6
Tor 5 days for acute infections	T	0.81	63.41
TOF 5 DAYS.	2	1.63	65.04

for 5- 7 days	1	0.81	65.85	
for 5-7 days	6	4.88	70.73	
for 5-7days	5	4.07	74.8	
for 6 days	2	1.63	76.42	
for 7 days	15	12.2	88.62	
for 7 days (33 mg/kg/day)	1	0.81	89.43	
for 7 days.	3	2.44	91.87	
for 7-10 day	1	0.81	92.68	
for 7days	1	0.81	93.5	
for five days.	1	0.81	94.31	
m.d.u.	5	4.07	98.37	
or 10 mls twice a day for 10 days	1	0.81	99.19	
until finished for throat infection	1	0.81	100	
Total	123	100		

1 C.5 Marginal effects for the inappropriate decision model for initial

2 presentations of upper respiratory tract infection

3 C.5.1 Average marginal effects

Table C-20: Average marginal effects for Model 0 – Inappropriate decisions for initial presentations of upper respiratory tract infection (unnecessary prescriptions versus reference of appropriate prescriptions together with appropriate non-prescribing situations)

Average Marginal Effects									
	Inappropriate Decisions for Initial Presentations of URTI								
Independent Variable	dy/dx	Std. Err.	Z	P>z	[95% Conf.	Interval]			
Patient Age Group (ref. 0-8 years)									
9-21 yrs	0.131857	0.004153	31.75	0.000	0.1237167	0.139997			
22-34 yrs	0.140908	0.004149	33.97	0.000	0.132777	0.149039			
35+ yrs	0.131635	0.003858	34.12	0.000	0.1240744	0.139196			
Patient Gender (ref. Female)									
Male	-0 0097/	0 002698	-3 61	0.000	-0 015032	-0 00446			
Wate	-0.00374	0.002098	-5.01	0.000	-0.013032	-0.00440			
Patient Penicillin Sensitivity Stat	us (ref. Negative	:)							
Positive	0.069712	0.005978	11.66	0.000	0.0579966	0.081428			
Patient Concession Status (ref. N	legative)								
Positive	0.007845	0.003689	2.13	0.033	0.0006142	0.015075			
Patient Mental Health Condition	Status (ref. Neg	gative)							
Positive	0.010328	0.004258	2.43	0.015	0.0019818	0.018673			
Missing	0.054179	0.043696	1.24	0.215	-0.031463	0.139821			
Weekend Consultation (ref. Wee	ekday)								
Positive	0.042556	0.004889	8.71	0.000	0.0329743	0.052137			
Practice Size (ref. Medium / Larg	;e)								
Small	0.103297	0.026201	3.94	0.000	0.0519446	0.154649			
Patient Primary Health Network	(ref. Perth Nort	h)							
Perth South	-0.0092	0.007907	-1.16	0.245	-0.024694	0.0063			
Country WA	0.00659	0.009084	0.73	0.468	-0.011215	0.024395			
Interstate	0.03749	0.015302	2.45	0.014	0.0074983	0.067481			
Missing	0.024683	0.013058	1.89	0.059	-0.000911	0.050276			
Patient Disadvantaged (ref. Neg	ative)								
Positive	0 012041	0 006146	1 96	0.050	-5 26F-06	0 024087			
Missing	-0 01588	0.006716	-2.36	0.018	-0 029045	-0 00270			
i i i i i i i i i i i i i i i i i i i	0.01000	0.000710	2.30	0.010	0.020040	0.00272			

C.5.2 Margins at representative values for effect on the probability of the inappropriate decision with change in patient gender, across different patient age groups

margins, dydx(pat_sex) at (agegrp_urti_new= (1 2 3 4)) Average marginal effects Number of obs 111,848 Model VCE : OIM Expression : Marginal predicted mean, predict() dy/dx w.r.t. : 1.pat_sex 1 (0-8 yrs) 1._at : agegrp_urti_new = 2 (9-21 yrs) 2._at : agegrp_urti_new = 3 (22-34 yrs) 3._at : agegrp_urti_new = 4._at : agegrp_urti_new = 4 (35+ yrs) Delta-method dy/dx Std. Err. z P>|z| [95% Conf. Interval] Male. _at 2.11 .0090993 .0006354 1 2 .0043184 0.035 .0175633 0064397 -.0319461 0.000 -.0445676 -.0193246 3 0.000 -.0233131 .0060761 -3.84 -.035222 -.0114041 4 -.0095587 .0053559 -1.780.074 -.0200562 .0009388 Note: dy/dx for factor levels is the discrete change from the base level.

Male young children had probability 0.01 higher probability of receiving an inappropriate decision than females, whereas males 9-21 years had 0.03 less probability, and males 22-34 years had 0.02 less probability.



1

2 3

4

5678901234567890123456789012345678

40 Figure C-2: Margins at representative values for effect on the probability of the

41 inappropriate decision with change in patient gender from female to male, across different

42 patient age groups, relative to the probability for female patients

C.5.3 Margins at representative values for the effect on inappropriate decision with change in patient age groups, across different values of patient gender

margins, dydx(agegrp_urti_new) at (pat_sex= (1 0)) Average marginal effects Model VCE : OIM Number of obs 111,848 Expression : Marginal predicted mean, predict()
dy/dx w.r.t. : 2.agegrp_urti_new 3.agegrp_urti_new 4.agegrp_urti_new (Male) 1._at : pat_sex 1 0 (Female) 2._at : pat_sex = Delta-method dy/dx Std. Err. z P>|z| [95% Conf. Interval] 9-21 years _at .0058061 1094557 $\begin{array}{c} 0.000 \\ 0.000 \end{array}$.0980759 .1208354 18.85 27.09 12 .1505011 .1396129 .1613894 22-34 years _at 0.000 .1232184 .0058257 .1118003 .1346365 1 21.15 2 .1556308 .0053407 29.14 0.000 .1451631 .1660985 35+ years at .1214512 .0053016 22.91 0.000 .1110602 .1318421 2 .1401092 .0050342 27.83 0.000 .1302424 .1499761 Note: dy/dx for factor levels is the discrete change from the base level.

Inappropriate Prescribing Model for URTI Margins At Representative Values for Effect on the Probability of the Inappropriate Decision with Change in Patient Age Groups, Across Different Values of Patient Gender

38 39

Figure C-3: Margins at representative values for the effect on the probability of

40 inappropriate decision occurring, with change in patient age group, across different values of

patient gender, relative to the probability for patients 0-8 years of age

Note: Effects on probability relative to the effect for patients 0-8 years.

C.5.4 Adjusted predictions for the effect on the inappropriate decision model at specific values of patient age group and patient gender

margins, at(pa	at_sex=(1 0) ageg	rp_u	urti_new=(1	2 3 4))	atmeans vsqu	ish po	st
Adjusted predi Model VCE :	ctions OIM			Numb	per of obs	=	111,848
Expression : 1at :	Marginal predic agegrp_urti_new	ted = =	mean, predi 1 1	ict()	(0-8 years) (Male)		
	0.pen_alle~y 1.pen_alle~y	=	.9471962 .0528038	(mean) (mean)	(
	0.pat_conc~n 1.pat_conc~n	=	.8303769 .1696231	(mean) (mean)			
	1.pat_ment~A 2.pat_ment~A	=	.8437701 .1228274	(mean) (mean)			
	3.pat_ment~A 0.weekend	= =	.0334025 .8976557	(mean) (mean)			
	1.weekend 0.practice~w	=	.1023443 .8695283	(mean) (mean)			
	1.practice~w 1.patient ~2	=	.1304717	(mean) (mean)			
	2.patient_~2 3.patient ~2	=	.3342483	(mean) (mean)			
	4.patient_~2 5.patient ~2	=	.0084847	(mean)			
	1.patient_~d 2 patient ~d	=	.7830627	(mean)			
	3.patient_~d	=	.1232834	(mean)			
	$1.flu_s \sim 2012$	=	.0694603	(mean)			
	1.flu_s~2013	=	.0603408	(mean)			
	$1.flu_s \sim 2014$	=	.0993491	(mean)			
	1.flu_s~2015	=	.1205744	(mean)			
	1.flu_s~2016	=	.1331182	(mean)			
2 .+	1.flu_s~2017	=	.0360132	(mean)	$(0, 8, y_{0,2}, r_{0,2})$		
2dt :	pat_sex	=		(2000)	(Female)		
	1.pen_alle~y	=	.9471962	(mean) (mean)			
	0.pat_conc~n 1.pat_conc~n	=	.8303769	(mean) (mean)			
	1.pat_ment~A 2.pat_ment~A	=	.8437701	(mean) (mean)			
	3.pat_ment~A 0.weekend	=	.0334025	(mean) (mean)			
	1.weekend 0.practice~w	=	.1023443 .8695283	(mean) (mean)			
	1.practice~w 1.patient_~2	=	.1304717 .4570489	(mean) (mean)			
	2.patient_~2 3.patient_~2	=	.3342483 .1855554	(mean) (mean)			
	4.patient_~2 5.patient_~2	= =	.0084847 .0146628	(mean) (mean)			
	1.patient_~d 2.patient_~d	= =	.7830627 .0936539	(mean) (mean)			
	3.patient_~d 0.flu_s~2012	=	.1232834 .9305397	(mean) (mean)			
	1.flu_s~2012 0.flu_s~2013	=	.0694603 .9396592	(mean) (mean)			
	1.flu_s~2013 0.flu_s~2014	=	.0603408	(mean) (mean)			
	1.flu_s~2014 0 flu_s~2015	=	.0993491 8794256	(mean)			
	1.flu_s~2015	=	.1205744	(mean)			
	1.flu_s~2016	=	.1331182	(mean)			
3 a+ •	1.flu_s~2017	=	.0360132	(mean)	(9-21 vrs)		
Jat .	pat_sex	=	2 1 0/71062	(maan)	(Male)		
	1.pen_alle~y	=	.0528038	(mean)			
	1.pat_conc~n	=	.1696231	(mean)			
	1.pat_ment~A 2.pat_ment~A	=	.843//01	(mean) (mean)			
	3.pat_ment~A 0.weekend	=	.0334025 .8976557	(mean) (mean)			

	1 weekend		1022442 (m	
		=	.1023443 (mean	
	1 practice~w	=	1204717 (moon	
	1 pationt 2	_	.1304/1/ (moon	
	2 patient 2	_	22/2/82 (moon)	
	3 nationt ~2	_	1855554 (mean)	
	4 nationt ~2	_	0084847 (mean)	
	5 natient ~2	-	0146628 (mean)	
	1 nationt ~d	_	7830627 (mean)	
	2 natient ~d	-	0936539 (mean)	
	3.patient ~d	=	.1232834 (mean)	
	$0.flu s \sim 2012$	=	.9305397 (mean)	
	$1.flu s \sim 2012$	=	.0694603 (mean)	
	0.flu_s~2013	=	.9396592 (mean)	
	1.flu_s~2013	=	.0603408 (mean)	
	0.flu_s~2014	=	.9006509 (mean)	
	1.flu_s~2014	=	.0993491 (mean)	
	0.flu_s~2015	=	.8794256 (mean)	
	1.f]u_s~2015	=	.1205744 (mean)	
	0.flu_s~2016	=	.8668818 (mean)	
	1.tlu_s~2016	=	.1331182 (mean	
	0.TIU_S~2017	=	.9639868 (mean	
4+	1.TIU_S~2017	=	.0360132 (mean)	(0, 21, ymc)
4dl	: agegrp_urt1_new	=	2		(9-21 yrs)
	0 pop allow	-	0471062 (moon)	(Fellia Te)
	1 pen allewy	_	.9471902 (moan)	
	0 pat conc~n	_	8303769 (mean)	
	1 pat conc~n	_	1696231 (mean)	
	1 nat ment~A	=	8437701 (mean)	
	2.pat_ment~A	=	.1228274 (mean)	
	3.pat_ment~A	=	.0334025 (mean)	
	0.weekend	=	.8976557 (mean)	
	1.weekend	=	.1023443 (mean)	
	0.practice~w	=	.8695283 (mean)	
	1.practice~w	=	.1304717 (mean)	
	1.patient_~2	=	.4570489 (mean)	
	2.patient_~2	=	.3342483 (mean)	
	3.patient_~2	=	.1855554 (mean)	
	4.patient_~2	=	.0084847 (mean)	
	5 patient_~2	=	.0146628 (mean)	
	1.patient_~d	=	.7830627 (mean	
	2.patient_~d	=	.0936539 (mean	
	3.patient_~0	=	.1232834 (mean	
	1 flu c.2012	=	.9303397 (moan	
	0 flu s 2012	_	0306502 (moon	
	1 flu s ~ 2013	_	0603408 (mean)	
	0 flu s ~ 2014	-	9006509 (mean)	
	1 flu s ~ 2014	=	0993491 (mean)	
	0.flu_s~2015	=	.8794256 (mean)	
	1.flu_s~2015	=	.1205744 (mean)	
	0.flu_s~2016	=	.8668818 (mean)	
	1.flu_s~2016	=	.1331182 (mean)	
	0.flu_s~2017	=	.9639868 (mean)	
	1.flu_s~2017	=	.0360132 (mean)	
5at	: agegrp_urti_new	=	3		(22-34 yrs)
	pat_sex	=	1		(Male)
	0.pen_alle~y	=	.94/1962 (mean)	
	1.pen_alle~y	=	.0528038 (mean	
	0.pat_conc~n	=	.8303/69 (mean	
	1 pat_conc~n	=	.1090231 (moan	
	2 nat monter	_	1228274 (maan)	
	3 nat ment~A	_	0334025 (mean	
	0 weekend	-	8976557 (mean)	
	1 weekend	=	1023443 (mean)	
	0.practice~w	=	.8695283 (mean)	
	1.practice~w	=	.1304717 (mean)	
	1.patient ~2	=	.4570489 (mean)	
	2.patient_~2	=	.3342483 (mean)	
	3.patient_~2	=	.1855554 (mean)	
	4.patient_~2	=	.0084847 (mean)	
	5.patient_~2	=	.0146628 (mean)	
	1.patient_~d	=	.7830627 (mean)	
	2 patient_~d	=	.0936539 (mean)	
	3.patient_~d	=	1232834 (mean)	
	0.TIU_S~2012	=	.930539/ (mean)	
	1.TIU_S~2012	=	.0694603 (mean	
	0.114_S~2013 1 flu c2012	=	. 3230237 (mean	
	0 f = 0.2013	_	0005400 (mean	
	$1 f_{11} c_{2014}$	_		mean	
	$0.f]_{11} s \sim 2015$	=	.8794256 (mean	
	1.flu_s~2015	=	.1205744	mean	
	0.flu_s~2016	=	8668818	mean	
	1.flu_s~2016	=	.1331182 (mean)	
	0.flu_s~2017	=	.9639868 (mean)	

1		1.flu_s~2017	=	.0360132 (me	ean)	
2	6at	: agegrp_urti_new	=	3		(22-34 yrs)
3		pat_sex	=	0471962 (ma) 190	(Female)
5		1.pen alle~v	_	.0528038 (me	an)	
6		0.pat_conc~n	=	.8303769 (me	an)	
7		1.pat_conc~n	=	.1696231 (me	ean)	
ă		1.pat_ment~A	=	.843//UL (Me 1228274 (me	an)	
10		3.pat_ment~A	=	.0334025 (me	an)	
11		0.weekend	=	.8976557 (me	an)	
12		1.weekend	=	.1023443 (me	an)	
14		0.practice~W 1.practice~W	=	.8695283 (Me 1304717 (me	an)	
15		1.patient ~2	=	.4570489 (me	ean)	
<u>16</u>		2.patient_~2	=	.3342483 (me	an)	
1/		3.patient_~2	=	.1855554 (me	an)	
19		4.patient_~2 5 natient ~2	=	.0084847 (Me 0146628 (me	an)	
ŽŎ		1.patient_~d	=	.7830627 (me	ean)	
21		2.patient_~d	=	.0936539 (me	an)	
22		3.patient_~d	=	.1232834 (me	an)	
23 74		0. Flu_S~2012 1 flu_s~2012	=	.9305397 (Me 0694603 (me	an)	
25		0.flu_s~2013	=	.9396592 (me	ean)	
26		1.f]u_s~2013	=	.0603408 (me	an)	
2/		0.flu_s~2014	=	.9006509 (me	an)	
29		0 flu s~2014	=	.0995491 (ME 8794256 (me	an)	
ĴΟ		1.flu_s~2015	=	.1205744 (me	ean)	
31		0.flu_s~2016	=	.8668818 (me	an)	
32		1.flu_s~2016	=	.1331182 (me	an)	
33		0.TIU_S~2017 1 flu s~2017	=	.9639868 (Me	an)	
35	7. at	: agegrp urti new	=	.0500152 (inc)	(35+ vrs)
36		pat_sex_	=	1		(Male)
37		0.pen_alle~y	=	.9471962 (me	an)	
20		1.pen_alle~y 0.pat_conc~n	=	.0528038 (Me 8303769 (me	an)	
40		1.pat_conc~n	=	.1696231 (me	ean)	
41		1.pat_ment~A	=	.8437701 (me	ean)	
42		2.pat_ment~A	=	.1228274 (me	an)	
43 <i>11</i>		3.pat_ment~A	=	.0334025 (Me 8976557 (me	an)	
45		1.weekend	_	.1023443 (me	an)	
46		0.practice~w	=	.8695283 (me	an)	
47		1.practice~w	=	.1304717 (me	an)	
48 49		1.patient_~2 2 natient ~2	=	.4570489 (Me 3342483 (me	an)	
50		3.patient ~2	=	.1855554 (me	ean)	
51		4.patient_~2	=	.0084847 (me	an)	
52		5.patient_~2	=	.0146628 (me	an)	
53		1.patient_~d 2 natient ~d	=	./83062/ (Me 0936539 (me	an)	
55		3.patient_~d	=	.1232834 (me	ean)	
<u>56</u>		0.flu_s~2012	=	.9305397 (me	an)	
52		1.†lu_s~2012	=	.0694603 (me	an)	
59		0.11u_S~2013 1 flu s~2013	-	.9596592 (IIIE 0603408 (me	an)	
ĞŎ		0.flu_s~2014	=	.9006509 (me	ean)	
<u>61</u>		1.f]u_s~2014	=	.0993491 (me	ean)	
62		0.tlu_s~2015 1.flu_s~2015	=	.8/94256 (Me 1205744 (me	an)	
64		0.flu s~2015	_	.8668818 (me	an)	
65		1.flu_s~2016	=	.1331182 (me	ean)	
66		0.flu_s~2017	=	.9639868 (me	an)	
62	8 a+	1.tlu_s~2017	=	.0360132 (me	ean)	(35) (35)
69	οαι	pat sex	_	4		(Female)
<u>70</u>		0.pen_alle~y	=	.9471962 (me	ean)	(
<u>71</u>		1.pen_alle~y	=	.0528038 (me	ean)	
/2		0.pat_conc~n	=	.8303769 (Me	an)	
73 74		1.pat_conc~n 1.pat_ment~A	=	.8437701 (me	an)	
75		2.pat_ment~A	=	.1228274 (me	ean)	
<u>76</u>		3.pat_ment~A	=	.0334025 (me	ean)	
// 78		0.weekend	=	.89/655/ (Me	an)	
ź9		0.practice~w	_	.8695283 (me	an)	
80		1.practice~w	=	.1304717 (me	an)	
81		1.patient_~2	=	.4570489 (me	an)	
84		2.patient_~2	=	.3342483 (Me 1855554 (me	an)	
8 4		4.patient_~2	=	.0084847 (me	ean)	
85		5.patient_~2	=	.0146628 (me	an)	
ğþ		1.patient_~d	=	.7830627 (me	an)	
88		2.pacient_~u 3 natient ~d	=	.1232834 (Me	an)	
		sipacione_ a				



123456789012345678901234567

Note: All other variables held constant at their means.

29 Figure C-4: Adjusted predictions for effect on the probability of the inappropriate decision

30 model at specific values of patient age group and patient gender, with all other covariates kept

31 constant at sample means

32

C.6 Marginal effects for the unnecessary antibiotic prescribing model

C.6.1 Average marginal effects

 Table C-21:
 Average marginal effects for Model 1 – Unnecessary antibiotic prescribing for initial presentations of upper respiratory tract infection

Average Marginal Effects						
Unnecessary Antibiotic Prescribing for Initial Presentations of URTI						
Independent Variable	dy/dx	Std. Err.	Z	P>z	[95% Conf.	Interval]
Patient Age Group (ref. 0-8 years)						
9-21 yrs	0.116792	0.005048	23.14	0.000	0.106898	0.126686
22-34 yrs	0.136988	0.005517	24.83	0.000	0.126176	0.1478
35+ yrs	0.145759	0.005694	25.60	0.000	0.134599	0.156919
URTI Condition (ref. Rhinosinusitis)						
Pharyngitis / Tonsillitis	-0.04823	0.00398	-12.12	0.000	-0.05603	-0.04043
Acute Otitis Media	-0.08054	0.004419	-18.22	0.000	-0.0892	-0.07188
Ordinal Line of Antibiotic Prescribed (ref. First-line)						
Second-line	-0.01569	0.005635	-2.78	0.005	-0.02674	-0.00465
Not Recommended	0.010901	0.003105	3.51	0.000	0.004816	0.016986
Repeat Prescription Status (ref. Negative)						
Positive	-0.00818	0.003122	-2.62	0.009	-0.01429	-0.00206
Penicillin Sensitivity (ref. Negative)						
Positive	-0.01533	0.005356	-2.86	0.004	-0.02583	-0.00484
Patient Concession Status (ref. Negative)						
Positive	0.009164	0.003262	2.81	0.005	0.00277	0.015558
Patient Mental Health Condition Status (ref. Negative)						
Positive	0.01968	0.00423	4.65	0.000	0.011389	0.02797
Missing	0.01878	0.022631	0.83	0.407	-0.02558	0.063136
Patient With Multiple URTI Episodes (ref. Negative)						
Positive	-0.00852	0.002633	-3.23	0.001	-0.01368	-0.00336

C.6.2 Margins at representative values for effect on unnecessary prescribing with change in upper respiratory tract infection condition, across different levels of ordinal choice of antibiotic prescribed



Note: dy/dx for factor levels is the discrete change from the base level.





38 Note: Effect on probability relative to that for rhinosinusitis.

Figure C-5: Margins at representative values for effect on unnecessary prescribing occurring with change in upper respiratory tract infection condition, across different levels of ordinal choice of antibiotic prescribed, relative to the effect for unnecessary prescribing for acute rhinosinusitis
C.6.3 Margins at representative values for the effect on unnecessary prescribing with change in ordinal choice of antibiotic prescribed, across different upper respiratory tract infection conditions

. margins, dydx(choice_urti_new) at (condition_urti = (1 2 3)) Average marginal effects Number of obs 51,210 Model VCE : OIM Expression : Marginal predicted mean, predict()
dy/dx w.r.t. : 2.choice_urti_new 3.choice_urti_new 1._at condition_urti (Rhinosinusitis) 1 = condition_urti condition_urti (Pharyngitis) (AOM) 2._at 3._at 23 = Delta-method dy/dx Std. Err. P> | z | [95% Conf. Interval] First-line (base outcome) Second-line at -.0305784 .0051739 0.000 -.0407191 -.0204378 1 2 3 5.91 -5.31 7.48 -.0489054 .0488027 $0.000 \\ 0.000$ -.0669733 .0360217 .0092185 -.0308376 .006521 .0615837 Not Recommended at -3.57 4.76 4.16 .0157176 0044062 0.000 .0243537 .0070815 .0211716 .0317079 .0298854 2 .0044459 0.000 .0124579 3 .0076216 0.000 .0167699 .0466459 Note: dy/dx for factor levels is the discrete change from the base level.

Unnecessary Prescribing Model Margins At Representative Values for Effect on the Probability of the Unnecessary Prescribing Occurring with Change in Ordinal Choice of Antibiotic Prescribed, Across Different URTI Conditions 05 Effects on Marginal Predicted Mean 0 05 Rhinosinusitis Pharyngitis AOM **URTI** Condition Second-line Not Recommended Note: Effect relative to the probability for First-line prescriptions.

38

40 Figure C-6: Margins at representative values for the effect on the probability of unnecessary 41 prescribing occurring, with change in ordinal choice of antibiotic prescribed, across different upper respiratory tract infection conditions, relative to the effect on the probability of unnecessary 42

43 prescribing for first-line prescriptions

- 44 45
- 46

C.6.4 Predictive margins for the effect on unnecessary prescribing, at specific values of patient age group and upper respiratory tract infection conditions

. margins,	at(condition_urt	i==(1 2 3)	agegrp_ur [.]	ti_new==(1 2 3 4)))	
Predictive Model VCE	maro :	gins OIM			Number o	f obs	=	51,210
Expression	:	Marginal pre	dicted mean	, predict	0			
1at	:	agegrp_urti_ condition_ur	new = ti =	1 1	(0-8 (Rhin	yrs) osinusit	is)	
2at	:	agegrp_urti_ condition_ur	new = ti =	1 2	(0-8 (Phar	yrs) yngitis)		
3at	:	agegrp_urti_ condition_ur	new = ti =	1 3	(0-8 (AOM)	yrs)		
4at	:	agegrp_urti_ condition_ur	new = ti =	2 1	(9-21 (Rhin	yrs) osinusit	is)	
5at	:	agegrp_urti_ condition_ur	new = ti =	2 2	(9-21 (Phar	yrs) yngitis)		
6at	:	agegrp_urti_ condition_ur	new = ti =	2 3	(9-21 (AOM_	yrs)		
7at	:	agegrp_urti_ condition_ur	new = ti =	3 1	(22-3 (Rhin	4 yrs) osinusit	is)	
8at	:	agegrp_urti_ condition_ur	new = ti =	3 2	(22-3 (Phar	4 yrs) yngitis)		
9at	:	agegrp_urti_ condition_ur	new = ti =	3 3	(22-3 (AOM)	4 yrs)		
10at	:	agegrp_urti_ condition_ur	new = ti =	4 1	(35+ (Rhin	yrs) osinusit	is)	
11at	:	agegrp_urti_ condition_ur	new = ti =	4 2	(35+ (Phar	yrs) yngitis)		
12at	:	agegrp_urti_ condition_ur	new = ti =	4 3	(35+ (AOM)	yrs)		
	 	D	elta-method					
	 +	Margin	Std. Err.	Z	P> z	[95% Co	onf.	Interval]
	at 1 2 3 4 5 6 7 8 9 1 2	.8539329 .7752567 .7253208 .94353 .9040771 .8766895 .9578739 .9268393 .9048698 .963977 .9367782 .9173445	$\begin{array}{c} .0070318\\ .0090823\\ .0097771\\ .0036783\\ .0053785\\ .0066338\\ .0029056\\ .0044543\\ .0056731\\ .0024899\\ .0039846\\ .0050343\\ \end{array}$	121.44 85.36 74.19 256.51 168.72 132.15 329.67 208.08 159.50 387.15 235.10 182.22	$\begin{array}{c} 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ \end{array}$.8401 75745 7061 936320 .89357 .86368 .95217 .91810 .893750 .959090 .928960 .90747	51 58 56 66 74 91 06 68 85 74	.8677149 .7930576 .7444836 .9507394 .8896916 .9635688 .9355695 .9159889 .9688572 .944588 .9272116



Figure C-7: Predictive margins for the effect on the probability of unnecessary prescribing with specific values of patient age group and upper respiratory tract infection condition, graphed by upper respiratory tract infection condition

C.6.5 Adjusted predictions for the effect on unnecessary prescribing, at specific values of upper respiratory trat infection condition and ordinal choice of antibiotic prescribed

The results below provide adjusted predictions for the effect on the probability of unnecessary prescribing occurring, at specific values of upper respiratory trat infection condition and ordinal choice of antibiotic prescribed, with all other covariates kept constant at sample means

margins, at(c	ondition_urti=(1 2	? 3) choice_urt	i_new=(1 2 3)) a	atmeans vsquish post
Adjusted pred Model VCE	ictions : OIM		Number of obs	5 = 51,210
Expression 1at	<pre>: Marginal predict : 1.agegrp_u~w 2.agegrp_u~w 3.agegrp_u~w 4.agegrp_u~w condition_urti choice_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 1.pen_alle~y 0.pat_conc~n 1.pat_conc~n 1.pat_ment~A 2.pat_ment~A 3.pat_ment~A 0.multip~RTI 1.multip~RTI 0.flu_s~2012 1.flu_s~2013</pre>	ted mean, predi = .3303456 = .190041 = .2083968 = .2712166 = 1 = 1 = .6772115 = .3227885 = .9357743 = .0642257 = .8294669 = .1268893 = .1268893 = .7713337 = .923472 = .076528 = .9305019 = .0694981	ct() (mean)	Rhinosinusitis) First-line)

	0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017	= = = = =	.9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694	(mean) (mean) (mean) (mean) (mean) (mean) (mean)
2at	: 1.agegrp_u~w 2.agegrp_u~w 3.agegrp_u~w 4.agegrp_u~w condition_urti	= = = = =	.3303456 .190041 .2083968 .2712166 1	(mean) (mean) (mean) (mean)
	0.repeat_s~t 1.repeat_s~t 0.pen_alle~y 1.pen_alle~y 0.pat_conc~n	= = = =	.6772115 .3227885 .9357743 .0642257 .8294669	(mean) (mean) (mean) (mean) (mean)
	1.pat_conc~n 1.pat_ment~A 2.pat_ment~A 3.pat_ment~A 0.multip~RTI	= = = =	.1705331 .8341925 .1268893 .0389182 .2286663	(mean) (mean) (mean) (mean) (mean)
	1.multip~RI1 0.flu_s~2012 1.flu_s~2012 0.flu_s~2013 1.flu_s~2013 0.flu_s~2014	= = = =	.923472 .076528 .9305019 .0694981	(mean) (mean) (mean) (mean) (mean)
	1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2015 0.flu_s~2016 1.flu_s~2016	= = = =	.1 .8825425 .1174575 .8798477 .1201523	(mean) (mean) (mean) (mean) (mean)
3at	0.TIU_S~2017 1.flu_s~2017 : 1.agegrp_u~w 2.agegrp_u~w 3.agegrp_u~w 4.agegrp_u~w	= = = =	.973306 .026694 .3303456 .190041 .2083968 .2712166	(mean) (mean) (mean) (mean) (mean)
	condition_urti choice_urti_new 0.repeat_s~t 1.repeat_s~t 0.pen_alle~y	= = = =	1 3 .6772115 .3227885 .9357743	(mean) (mean) (mean)
	0.pat_conc~n 1.pat_conc~n 1.pat_ment~A 2.pat_ment~A 3.pat_ment~A	- - - - -	.8294669 .1705331 .8341925 .1268893 .0389182	(mean) (mean) (mean) (mean) (mean)
	0.multip~RTI 1.multip~RTI 0.flu_s~2012 1.flu_s~2012 0.flu_s~2013 1.flu_s~2013	= = = =	.2286663 .7713337 .923472 .076528 .9305019 .0694981	(mean) (mean) (mean) (mean) (mean)
	0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2015 0.flu_s~2016	= = = =	.9 .1 .8825425 .1174575 .8798477	(mean) (mean) (mean) (mean) (mean)
4at	1.TIU_S~2016 0.flu_s~2017 1.flu_s~2017 : 1.agegrp_u~w 2.agegrp_u~w 3.agegrp_u~w	= = = =	.1201523 .973306 .026694 .3303456 .190041 .2083968	(mean) (mean) (mean) (mean) (mean)
	4.agegrp_u~w condition_urti choice_urti_new 0.repeat_s~t 1.repeat_s~t	= = = =	.2712166 2 1 .6772115 .3227885	(mean) (mean) (mean)
	1.pen_alle~y 0.pat_conc~n 1.pat_conc~n 1.pat_ment~A 2.pat_ment~A	- - - - -	.0642257 .8294669 .1705331 .8341925 .1268893	(mean) (mean) (mean) (mean) (mean)
	3.pat_ment~A 0.multip~RTI 1.multip~RTI 0.flu_s~2012 1.flu_s~2012 0.flu_s~2013	= = = =	.0389182 .2286663 .7713337 .923472 .076528 .9305019	(mean) (mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 1.flu_s~2015	= = = =	.0694981 .9 .1 .8825425 .1174575	(mean) (mean) (mean) (mean) (mean)
5at	0.FTU_S~2016 1.fTU_S~2016 0.fTU_S~2017 1.fTU_S~2017 : 1.agegrp_u~w 2.agegrp_u~w	= = = =	.0798477 .1201523 .973306 .026694 .3303456 .190041	(mean) (mean) (mean) (mean) (mean)
	3.agegrp_u~w 4.agegrp_u~w	=	.2083968 .2712166	(mean) (mean)

(Rhinosinusitis) (Second-line)

(Rhinosinusitis) (Not Recommended)

(Pharyngitis) (First-line)

	condition urti	=	2	
	choice_urti_new	=	2	
	0.repeat_s~t	=	.6772115	(mean)
	1.repeat_s~t	=	.3227885	(mean)
	0.pen_alle~y	=	.935//43	(mean)
	0.pat_conc~n	_	.8294669	(mean)
	1.pat_conc~n	=	.1705331	(mean)
	1.pat_ment~A	=	.8341925	(mean)
	2.pat_ment~A	=	.1268893	(mean)
	3.pat_ment~A	=	.0389182	(mean)
	1 multip~RII	_	.2200003	(mean)
	$0.flu s \sim 2012$	_	.923472	(mean)
	1.flu_s~2012	=	.076528	(mean)
	0.flu_s~2013	=	.9305019	(mean)
	1.flu_s~2013	=	.0694981	(mean)
	0.TIU_S~2014	=	.9	(mean)
	$0 flu s \sim 2015$	_	8825425	(mean)
	1.flu_s~2015	=	.1174575	(mean)
	0.flu_s~2016	=	.8798477	(mean)
	1.f]u_s~2016	=	.1201523	(mean)
	0.flu_s~2017	=	.9/3306	(mean)
6 at .	$1.110_S \sim 2017$	_	3303456	(mean)
0ac	2.agegrp_u~w	_	.190041	(mean)
	3.agegrp_u~w	=	.2083968	(mean)
	4.agegrp_u~w	=	.2712166	(mean)
	condition_urti	=	2	
	cnoice_urti_new	=	5	(maan)
	1.repeat_s~t	-	.3227885	(mean)
	0.pen_alle~v	=	.9357743	(mean)
	1.pen_alle~y	=	.0642257	(mean)
	0.pat_conc~n	=	.8294669	(mean)
	1.pat_conc~n	=	.1/05331	(mean)
	2 nat ment~A	_	1268893	(mean)
	3.pat_ment~A	-	.0389182	(mean)
	0.multip~RTI	=	.2286663	(mean)
	1.multip~RTI	=	.7713337	(mean)
	0.flu_s~2012	=	.923472	(mean)
	1.TIU_S~2012 0.flu_s~2013	=	.076528	(mean)
	1.flu_s~2013	_	.0694981	(mean)
	0.flu_s~2014	=	.9	(mean)
	1.flu_s~2014	=	.1	(mean)
	0.flu_s~2015	=	.8825425	(mean)
	1.TIU_S~2015	=	.11/45/5	(mean)
	$1 flu s \sim 2016$	=	1201523	(mean)
	0.flu_s~2017	=	.973306	(mean)
	1.flu_s~2017	=	.026694	(mean)
7at :	: 1.agegrp_u~w	=	.3303456	(mean)
	2.agegrp_u~w	=	.190041	(mean)
	3.agegrp_u~w	=	2712166	(mean)
	condition urti	_	.2712100	(mean)
	choice_urti_new	=	1	
	0.repeat_s~t	=	.6772115	(mean)
	1.repeat_s~t	=	.3227885	(mean)
	1 pen_alle~y	_	.9357745	(mean)
	0.pat_conc~n	=	.8294669	(mean)
	1.pat_conc~n	=	.1705331	(mean)
	1.pat_ment~A	=	.8341925	(mean)
	2.pat_ment~A	=	.1268893	(mean)
	0 multin.PTT	_	2286663	(mean)
	1.multip~RTI	=	.7713337	(mean)
	0.flu_s~2012	=	.923472	(mean)
	1.flu_s~2012	=	.076528	(mean)
	U.TIU_S~2013	=	.9302019	(mean)
	$1.11u_s \sim 2013$ 0 flu s ~ 2014	_	.0094981	(mean)
	1.flu_s~2014	=	.1	(mean)
	0.flu_s~2015	=	.8825425	(mean)
	1.flu_s~2015	=	.1174575	(mean)
	0.tlu_s~2016	=	.8798477	(mean)
	$1.710_{S} \sim 2010$	=	.1201523	(mean)
	1.flu s~2017	=	.026694	(mean)
8at :	: 1.agegrp_u~w	=	.3303456	(mean)
	2.agegrp_u~w	=	.190041	(mean)
	3.agegrp_u~w	=	.2083968	(mean)
	4.ayeyrp_u~w	=	.2/12100 2	(mean)
	choice_urti new	=	2	
	0.repeat_s~t	=	.6772115	(mean)
	1.repeat_s~t	=	.3227885	(mean)
	U.pen_alle~y	=	.9357743	(mean)
	1.pen_arre~y	-	8294660	(mean)
	1.pat_conc~n	=	.1705331	(mean)
	1.pat_ment~A	=	.8341925	(mean)
	2.pat_ment~A	=	.1268893	(mean)
	3.pat_ment~A	=	.0389182	(mean)
	1.multin~RTT	-	.2200003	(mean)
	0.flu_s~2012	=	.923472	(mean)
	1.flu_s~2012	=	.076528	(mean)

(Pharyngitis) (Second-line)

(Pharyngitis) (Not Raecommended)

(AOM) (First-line)

(AOM) (Second-line)

9at	:	0.flu_s~2013 1.flu_s~2013 0.flu_s~2014 1.flu_s~2015 1.flu_s~2015 1.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.agegrp_u~w 2.agegrp_u~w 2.agegrp_u~w 4.agegrp_u~w 4.agegrp_u~w 4.agegrp_u~w 0.repeat_s~t 1.repeat_s~t 0.repeat_s~t 1.repeat_s~t 0.pen_alle~y 0.pat_conc~n 1.pat_conc~n 1.pat_conc~n 1.pat_conc~n 1.pat_conc~n 1.pat_ment~A 3.pat_ment~A 3.pat_ment~A 3.pat_ment~A 3.pat_ment~A 3.pat_ment~A 3.pat_ment~A 3.pat_ment~A 3.pat_ment~A 3.flu_s~2012 1.flu_s~2013 1.flu_s~2013 1.flu_s~2014 0.flu_s~2015 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017	= .9 = .0 = .0 = .0 = .0 = .0 = .1 = .1 = .1 = .2 = .2	3305019 1694981 .9 .1 174575 1798477 .201523 973306 190041 1083968 712166 3 3 3 3772115 1227885 1357743 1642257 1284669 .705331 1341925 .2889182 .288663 1713337 923472 076528 .30778 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .3077888 .30778888 .3077888 .3077888 .30778888 .30778888 .30778888 .30778888 .3077888888888 .30778888 .3077888888888 .30778888	(mean) (m	(AOM) (Not Rec	ommended)
		De Margin	lta-method Std. Err.	l z	P> z	[95% Conf.	Interval]
	_at 1 2 3 4 5 6 7 8 9	.9447867 .9168631 .9305799 .8900707 .842106 .9101885 .8383023 .886312 .8697051	.0040706 .0053206 .0043667 .0055216 .0115252 .0049888 .0076615 .0066989 .0080933	232.10 172.32 213.11 161.20 73.07 182.45 109.42 132.31 107.46	$\begin{array}{c} 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ \end{array}$.9368084 .9064348 .9220213 .8792485 .8195171 .9004105 .823286 .8731824 .8538425	.9527651 .9272913 .9391385 .9008929 .864695 .9199664 .8533185 .8994415 .8855676

1 C.7 Marginal effects for the ordinal choice of antibiotic prescribing model for

2 upper respiratory tract infection

3 C.7.1 Average marginal effects

4 5

 Table C-22:
 Average marginal effects for Model 2
 Ordinal choice of antibiotic prescribed for initial presentations of upper respiratory tract infection

		Average Ma	arginal Effects				
	Ordinal Choice of	of Antibiotic Prescr	ibed for Initial	Presenta	tions of URT	1	
Independent V	/ariable	dy/dx	Std. Err.	z	P>z	[95% Conf.	Interval]
Patient Age Gr	oup						
0-8 years		(base outcome)					
9-21 years (pre	edict outcome:)						
	First-line	-0.02193	0.0053762	-4.08	0.000	-0.03247	-0.01139
	Second-line	0.002669	0.0006873	3.88	0.000	0.001321	0.004016
	Not Recommended	0.01926	0.004797	4.01	0.000	0.009858	0.028662
22-34 years (pi	redict outcome:)						
	First-line	-0.05306	0.0053765	-9.87	0.000	-0.0636	-0.04253
	Second-line	0.003753	0.0008168	4.60	0.000	0.002153	0.005354
	Not Recommended	0.049309	0.0049305	10.00	0.000	0.039646	0.058973
35 and over ye	ars (predict outcome:)						
	First-line	-0.09071	0.0052159	-17.39	0.000	-0.10093	-0.08049
	Second-line	0.007518	0.0010344	7.27	0.000	0.005491	0.009546
	Not Recommended	0.083193	0.0048919	17.01	0.000	0.073605	0.092781
Repeat on Pres	scription Status						
Negative		(base outcome)					
Positive (predi	ct outcome:)						
	First-line	-0.08096	0.0045827	-17.67	0.000	-0.08994	-0.07198
	Second-line	0.003572	0.000923	3.87	0.000	0.001763	0.005381
	Not Recommended	0.07739	0.0043861	17.64	0.000	0.068793	0.085987
URTI Condition	1						
Rhinosinusitis		(base outcome)					
Pharyngitis (pr	edict outcome:)						
	First-line	-0.02184	0.004738	-4.61	0.000	-0.03113	-0.01255
	Second-line	0.002264	0.0004291	5.28	0.000	0.001423	0.003105
	Not Recommended	0.019576	0.004438	4.41	0.000	0.010878	0.028274
AOM (predict o	outcome:)						
	First-line	0.083715	0.0053492	15.65	0.000	0.073231	0.094199
	Second-line	-0.00988	0.0010621	-9.30	0.000	-0.01196	-0.0078
	Not Recommended	-0.07384	0.0047432	-15.57	0.000	-0.08313	-0.06454
Unnecessary /	Necessary Prescription Statu	IS					
Necessary		(base outcome)					
Unnecessary (p	predict outcome:)						
	First-line	-0.02229	0.0054337	-4.10	0.000	-0.03294	-0.01164
	Second-line	0.002119	0.0006078	3.49	0.000	0.000928	0.00331
	Not Recommended	0.020171	0.0048703	4.14	0.000	0.010626	0.029717

		Average M	argina	al Effects conti	nued			
	Ordinal Choice	of Antibiotic	Presci	ribed for Initial	Presenta	tions of URT		
Independent V	/ariable	dy/dx	Std	. Err. z	2 P	>z [9	5% Conf. I	nterval]
Patient with M	ultiple URTI Episodes							
Negative		(base outc	ome)					
Positive (predic	ct outcome:)							
	First-line	0.03	1317	0.004314	3.05	0.002	0.004715	0.021625
	Second-line	-0.0	0109	0.0003615	-3.01	0.003	-0.0018	-0.00038
	Not Recommended	-0.0	1208	0.0039783	-3.04	0.002	-0.01988	-0.00428
Practice Size								
Medium / Larg	e	(base outc	ome)					
Small (predict o	outcome:)							
	First-line	-0.10974		0.028583	-3.84	0.000	-0.16576	-0.05372
	Second-line	0.004687		0.0011485	4.08	0.000	0.002436	0.006938
	Not Recommended	0.105054		0.0287724	3.65	0.000	0.048661	0.161446
Reason for Pre	scribing Recorded							
Negative		(base outc	ome)					
Positive (predic	ct outcome:)							
	First-line	0.073287		0.0136648	5.36	0.000	0.046504	0.10007
	Second-line	-0.0085		0.0019784	-4.30	0.000	-0.01238	-0.00462
	Not Recommended	-0.06479		0.0118295	-5.48	0.000	-0.08797	-0.0416
Patient Disadva	antage Status							
Negative		(base outc	ome)					
Positive (predic	ct outcome:)							
	First-line	0.019428		0.008003	2.43	0.015	0.003742	0.035113
	Second-line	-0.00178		0.0008149	-2.19	0.029	-0.00338	-0.00019
	Not Recommended	-0.01764		0.007213	-2.45	0.014	-0.03178	-0.00351
Missing (predic	ct outcome:)							
	First-line	0.026192		0.0077608	3.37	0.001	0.010982	0.041403
	Second-line	-0.0025		0.0008479	-2.94	0.003	-0.00416	-0.00083
	Not Recommended	-0.0237		0.0069567	-3.41	0.001	-0.03733	-0.01006
Patient Comorl	bid Condition Status							
Negative		(base outc	ome)					
Positive (predic	ct outcome:)							
	First-line	-0.02684		0.0049631	-5.41	0.000	-0.03657	-0.01712
	Second-line	0.002104		0.0004388	4.79	0.000	0.001244	0.002964
	Not Recommended	0.02474		0.0046253	5.35	0.000	0.015674	0.033805
Missing (predic	ct outcome:)							
	First-line	-0.03914		0.0494152	-0.79	0.428	-0.13599	0.05771
	Second-line	0.002809		0.0026093	1.08	0.282	-0.00231	0.007923
	Not Recommended	0.036333		0.046837	0.78	0.438	-0.05547	0.128132

C.7.2 Margins at representative values for effect on ordinal choice of antibiotic, with change in upper respiratory tract infection condition, across different values of patient age group

margins, dydx(condition_urti) at (agegrp_urti_new = (1 2 3 4))

Average marginal et Model VCE : OIM	ffects		Numbe	er of obs	= 5	1,210
dy/dx w.r.t. : 2.co 1predict : Marg 2predict : Marg 3predict : Marg	ondition_urti 3 ginal predicted ginal predicted ginal predicted	3.condition d mean (1.cl d mean (2.cl d mean (3.cl	_urti noice_urti noice_urti noice_urti	i_new), pr i_new), pr i_new), pr	redict(pr ou redict(pr ou redict(pr ou	tcome(1)) tcome(2)) tcome(3))
1at : ageg 2at : ageg 3at : ageg 4at : ageg	<pre>Jrp_urti_new = Jrp_urti_new = Jrp_urti_new = Jrp_urti_new = Jrp_urti_new =</pre>		1 (1 2 (1 3 (1 4 (1)	0-8 yrs) 9-21 yrs) 22-34 yrs 35+ yrs))	
	De	elta-method			[05% carf	
	ay/ax	Sta. Err.	Z	P> Z	[95% CONT.	Intervalj
Rhinosinusitis	(base outcor	ne) 				
Pharyngitis _predict#_at 1 1 1 2 1 3 1 4 2 1 2 2 2 3 2 4 3 1 3 2 3 3 3 4	$\begin{array}{c}0814757\\.017714\\.0273729\\014731\\.008146\\0015345\\0014387\\.0000692\\.0733297\\0161795\\0259342\\.0146618\end{array}$.0078084 .0088954 .0083436 .0079023 .0011457 .000794 .0005393 .0001728 .0081277 .0079004 .0078884	-10.43 1.99 3.28 -1.86 7.11 -1.93 -2.67 0.40 10.29 -1.99 -3.28 1.86	0.000 0.046 0.001 0.062 0.000 0.053 0.008 0.689 0.000 0.689 0.001 0.047 0.001 0.063	0967798 .0002793 .0110198 0302192 .0059005 0030907 0024957 0002696 .0593634 0321095 0414187 0007992	0661716 .0351487 .0437261 .0007572 .0103915 .0000217 0003817 .000408 .087296 0002495 0104497 .0301228
AOM _predict#_at 1 1 1 2 1 3 1 4 2 1 2 2 2 3 2 4 3 1 3 2 3 3 3 4	.0830478 .0844255 .0837281 .0845293 -0151681 -010221 -006916 -0678797 -0742045 -0768121 -0793086	.0053406 .0053775 .0053308 .005402 .0012548 .0011647 .0010819 .0010477 .004497 .004497 .0047971 .0048766 .0050251	15.55 15.70 15.71 15.65 -12.09 -8.78 -6.39 -4.98 -15.09 -15.47 -15.75 -15.78	0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000	.0725804 .0738858 .0732798 .0739415 .0176276 .0125037 .0090365 .0072741 .0766937 .0836066 .0863701 .0891576	.0935152 .0949651 .0941763 .095117 -0127087 -0079382 -0047955 -0031672 -0590656 -0648024 -0672541 -0694597
Note: dy/dx for fac	ctor levels is	the discret	te change	from the	base level.	

Variables that uniquely identify margins: agegrp_urti_new _deriv.



Margins At Representative Values For The Effect of Change in URTI Condition Across Patient Age Groups Model for Ordinal Choice of Antibiotic Prescribed for Initial Presentations of URTI

Note: Effect relative to that of Acute Rhinosinusitis.

Margins at representative values for the effect on the probability of each of the Figure C-8: three outcomes of ordinal choice of antibiotic occurring, with change upper respiratory tract infection condition, across different values of patient age group, relative to the effect on the probability for acute rhinosinusitis

C.7.3 Margins at representative values for the effect on ordinal choice of antibiotic prescribed, with change in repeat prescription status, across different values of patient age group



Figure C-9: Margins at representative values for the effect on the probability of each of the three outcomes of ordinal choice occurring, with change in repeat prescription status, across different values of patient age group, relative to the effect on the probability for prescriptions issued without repeats

C.7.4 Adjusted predictions for the effect on ordinal choice of antibiotic, at specific values of patient age group and upper respiratory tract infection condition

The results below provide the adjusted predictions for the effect on the probability of each of the three outcomes of ordinal choice of antibiotic occurring, at specific values of patient age group and upper respiratory tract infection condition, with all other covariates held constant at sample means.

margins, at	(condition_urti=(1 2	3) agegrp_u	urti_new=(1 2 3	8 4)) atmeans vsquish post
Adjusted pro Model VCE	edictions : OIM		Number of	obs = 51,210
1predict	: Marginal predict	ed mean (1.0	choice_urti_new	<pre>/), predict(pr outcome(1)) /) predict(pr outcome(2))</pre>
3predict	: Marginal predict	ed mean (3.0	choice_urti_new	i), predict(pr outcome(2))
1at	: agegrp_urti_new :	=	1	(0-8 yrs)
	1.repeat_s~t	= .87721.	35 (mean)	
	condition_urti	=	1	(Rhinosinusitis)
	0.de~n_presc	= .141847	73 (mean) 27 (mean)	
	0.multip~RTI :	228660	53 (mean)	
	1.multip~RTI	771333	37 (mean)	
	0.practice~w	867090	53 (mean) 37 (mean)	
	0.reason_r~d	883518	38 (mean)	
	1.reason_r~d	11648	12 (mean)	
	1.patient_~d =	= .794823 - 088499	52 (mean) 33 (mean)	
	3.patient_~d	116670	54 (mean)	
	0.pat_co~rti	808162	25 (mean)	
	2 pat co~rti	= .152919 = 038918	94 (mean) 82 (mean)	
	0.flu_s~2012	= .92342	72 (mean)	
	1.flu_s~2012	= .07652	28 (mean)	
	0.flu_s~2013 = 1.flu_s~2013 =	= .93050. = .069498	19 (mean) 81 (mean)	
	0.flu_s~2014	=	.9 (mean)	
	1.flu_s~2014 :	-	.1 (mean)	
	1.flu s~2015	= .002344	25 (mean) 75 (mean)	
	0.flu_s~2016	87984	77 (mean)	
	1.flu_s~2016		23 (mean)	
	1.flu_s~2017	= .97350	94 (mean)	
2at	: agegrp_urti_new :	=	1	(0-8 yrs)
	0.repeat_s~t	= .67721. - 32278	15 (mean) 85 (mean)	
	condition_urti	= .522760	2	(Pharyngitis)
	0.de~n_presc	14184	73 (mean)	
	1.de~n_presc	= .858157 - 22866	27 (mean) 53 (mean)	
	1.multip~RTI	771333	37 (mean)	
	0.practice~w	86709	53 (mean)	
	1.practice~w = 0 reason r~d =	= .13290: = 883518	37 (mean) 38 (mean)	
	1.reason_r~d	11648	12 (mean)	
	1.patient_~d	79482	52 (mean)	
	2.patient_~d	= .088498 = .116676	53 (mean) 54 (mean)	
	0.pat_co~rti	808162	25 (mean)	
	1.pat_co~rti	152919	94 (mean)	
	0.flu s~2012	= .038910 = .92342	72 (mean)	
	1.flu_s~2012	07652	28 (mean)	
	0.tlu_s~2013	93050	19 (mean) 81 (moan)	
	0.flu_s~2014	= .009498	.9 (mean)	
	1.f]u_s~2014 :	=	1 (mean)	
	0.tlu_s~2015 =	= .882544 - 11745	25 (mean) 75 (mean)	
	0.flu_s~2016	879842	77 (mean)	
	1.f]u_s~2016 :	120152	23 (mean)	
	0.tlu_s~2017 = 1 flu_s~2017 =	= .97330 = 02669	J6 (mean) 94 (mean)	
3at	: agegrp_urti_new	= 10200	1	(0-8 yrs)
	0.repeat_s~t	67721	15 (mean)	
	condition urti	= .322788	3 (mean)	(AOM)
	0.de~n_presc	14184	73 (mean)	
	1 de~n_presc	858152	27 (mean)	
	1.multin~RTT	228060 77133	37 (mean)	
	0.practice~w	86709	63 (mean)	
	1.practice~w		37 (mean)	
	1.reason r~d	= .003310 = .116482	12 (mean)	
	1.patient_~d	79482	52 (mean)	
	2.patient_~d		83 (mean)	
	0.pat_co~rti	= .808162	25 (mean)	

2 3 4	
5 6 7	
7 800-12-01-1-1-1-1-12-00-12-0-12-0-12-0-1	
16780012m456780012	
145678001VM456780	
3012345678901234 36666666666677777	
4567800120045676	
88 90 91 92	

1234
5678
9 10 11 12
134567
18 19 20
22 23 24 25
26 27 28 29
35678
39 40 41 42
445 46 47
48 49 50 51
523455
5589 60
612 663 64
6667 667
59 70 71 72 73
74 75 76 77
78 79 80 81
2000000
99934 994
95 997 92
100 102

	1.pat_co~rti		1 5 3 6 1 6 4	
	a	=	.1529194	(mean)
	2.pat_co~rti	=	.0389182	(mean)
	0.flu_s~2012	=	.923472	(mean)
	1.flu s~2012	=	.076528	(mean)
	$0 f 1 u s \sim 2013$	=	9305019	(mean)
	1 flu s 2013	_	060/081	(moan)
	0 flu c.2014	2	.000,400.	(moan)
	1 flu s 2014	-	. 5	(mean)
	1.11u_S~2014	=	0005405	(mean)
	0.TIU_S~2015	=	.8825425	(mean)
	1.†lu_s~2015	=	.1174575	(mean)
	0.flu_s~2016	=	.8798477	(mean)
	1.flu_s~2016	=	.1201523	(mean)
	0.flu_s~2017	=	973306	(mean)
	1 flu s 2017	_	026694	(mean)
4	1.11u_5~2017	-	.020094	(mean)
4at .	. agegrp_urtr_new	=	6770115	(
	0.repeat_s~t	=	.6//2115	(mean)
	1.repeat_s~t	=	.322/885	(mean)
	condition_urti	=	1	
	0.de~n_presc	=	.1418473	(mean)
	1.de~n presc	=	.8581527	(mean)
	0.multip~RTT	=	2286663	(mean)
	1 multin~RTT	_	7713337	(mean)
	0 practico.w	2	8670963	(moan)
	1 practice~w	_	1220027	(mean)
	1.practice~w	=	.1329037	(mean)
	0.reason_r~d	=	.8835188	(mean)
	1.reason_r~d	=	.1164812	(mean)
	1.patient_~d	=	.7948252	(mean)
	2.patient_~d	=	.0884983	(mean)
	3.patient ~d	=	.1166764	(mean)
	0.pat co~rti	=	.8081625	(mean)
	1 nat co~rti	=	1529194	(mean)
	2 nat counti	-	0300100	(moon)
	2.pat_co~rt1	=	.0309102	(mean)
	0.11U_S~2012	=	.923472	(mean)
	1.†Iu_s~2012	=	.076528	(mean)
	0.flu_s~2013	=	.9305019	(mean)
	1.flu_s~2013	=	.0694981	(mean)
	0.flu s~2014	=	.9	(mean)
	1.flu_s~2014	=	.1	(mean)
	$0.f1u_s \sim 2015$	=	.8825425	(mean)
	1 flu s~2015	=	1174575	(mean)
	0 flu s ~ 2016	_	8798477	(mean)
	1 flu s 2016	Ξ	1201523	(mean)
	0 flu c 2017	_	072206	(mean)
	1 flu c 2017	-	. 97 3 3 0 0	(mean)
F at .	1.11u_S~2017	=	.020094	(mean)
5dl :	agegrp_urti_new	=	6770115	(
	0.repeat_s~t	=	.6//2115	(mean)
	1.repeat_s~t	=	.3227885	(mean)
	condition_urti	=	2	
	0.de~n_presc	=	.1418473	(mean)
	1.de~n_presc	=	.8581527	(mean)
	0.multip~RTI	=	.2286663	(mean)
	1.multip~RTI	=	.7713337	(mean)
	0.practice~w	=	.8670963	(mean)
	1 practice~w	=	1329037	(mean)
	0 reason r~d	_	8835188	(mean)
	1 reason rad	_	116/812	(mean)
	1 nationt d	_	7040252	(mean)
	1.patient_~u	=	./940232	(mean)
	2.patient_~d	=	.0884983	(mean)
	3.patient_~d	=	.1166/64	(mean)
	0.pat_co~rti	=	.8081625	(mean)
	1.pat_co~rti	=	.1529194	(mean)
	2.pat_co~rti	=	.0389182	(mean)
	0.flu_s~2012	=	.923472	(mean)
	1.flu_s~2012	=	.076528	ไต้การทั่ง
	$0.f1u_s \sim 2013$	=	0205010	(mean)
			. 3202013	(mean)
	1.flu_s~2013	=	.0694981	(mean) (mean)
	1.flu_s~2013 0 flu_s~2014	=	.0694981	(mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014	=	.0694981	(mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015	= = =	.9305019 .0694981 .9 .1	(mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015	= = = =	.9303019 .0694981 .9 .1 .8825425	(mean) (mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015	= = = =	.9303019 .0694981 .9 .1 .8825425 .1174575	(mean) (mean) (mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2015 0.flu_s~2016	= = = =	.9303019 .0694981 .9 .1 .8825425 .1174575 .8798477	(mean) (mean) (mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 1.flu_s~2016	= = = = =	.9303019 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523	(mean) (mean) (mean) (mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2016 0.flu_s~2017	= = = = = =	.9305019 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017	= = = = = = = =	.9303019 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at ::	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 : agegrp_urti_new	= = = = = = = = =	.9305019 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .2	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t	- - - - - - - - - - - - - - - - - - -	.9305019 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 2 .6772115	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 1.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 2 .6772115 .3227885	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition urti</pre>		.0505019 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 2 .6772115 .3227885	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 0.dev npresc</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .2 .6772115 .3227885 .3227885 .3227885	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 1.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 2 .6772115 .3227885 .3227885 .3227885 .321527	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 0.multin~BTT</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .2 .6772115 .3227885 .3227885 .1418473 .8581527 .2286663	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 0.de~n_presc 1.de~n_presc 0.multip~FT</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .3227885 .3227885 .3227885 .3227885 .2286663 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286653 .2286553 .228555 .2286553 .2285553 .2285553 .2285553 .2285553 .2285553 .2285553 .2285553 .2285555 .22855555 .2285555555 .2285555555555	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 0.multip~RTI 0.multip~RTI 0.retire</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 2 .6772115 .3227885 .3277885 .3227885 .327787785 .327778778 .3277877778 .3277877777877777777	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 1.de~n_presc 1.de~n_presc 0.multip~RTI 1.multip~RTI 0.practice~w 1 practice~w</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .2 .6772115 .3227885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327885 .327785 .3278577777777777777777777777777777777777	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 1.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 1.de~n_presc 1.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 0.repeat_s~t		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .026694 .3227885 .322785 .3227885 .3227885 .322785 .3227885 .3227855 .3227855 .3227855 .3227855 .3227855 .3227855 .3227855 .3227855 .3227855 .3277855 .3227855 .3277855 .3277855 .3277855 .3277855 .3277855 .3277855 .3277855 .3277855 .3277855 .3277855 .32778555 .32777777855 .32778555 .32778555 .327785555 .327785555 .327785555 .327785555 .3277855555 .3277855555 .3277855555 .3277855555555555555555555555555555555555	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 1.de~n_presc 0.multip~RTI 0.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.reason_r~d</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 2 .6772115 .3227885 .3227885 .3227885 .3227885 .3227885 .3227885 .3227885 .3227885 .3227885 .3226663 .7713337 .8581527 .2286663 .7713337 .8570963 .1329037 .8835188	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.gegrp_urti_new 0.repeat_s~t 1.repeat_s~t 0.redisp_rti 0.de~n_presc 0.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.reason_r~d</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .2 .6772115 .3227885 .3 .1418473 .8581527 .2286663 .1418473 .8581527 .2286663 .1329037 .8835188 .1164812	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 : agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 0.reason_r~d 1.reason_r~d 1.patient_~d</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .026694 .3227885 .327885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .3377885 .33778855 .3377885 .337785 .3377855 .3377855 .3377855 .3377855 .3377855 .33778555 .337785555555555555555555555555555555555	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc 0.multip~RTI 1.multip~RTI 1.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.patient_~d 2.patient_~d</pre>		.0694981 .0694981 .9 .11 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .2 .6772115 .3227885 .3227885 .3227885 .3227885 .3227885 .3226663 .7713337 .8670963 .1329037 .8835188 .1164812 .7948252 .0884983	(mean) (mean)
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 0.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.gegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 0.de~n_presc 0.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 0.reason_r~d 1.reason_r~d 1.reason_r~d 1.patient_~d 3.patient_~d</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .2 .6772115 .3227885 .3 .1418473 .8581527 .2286663 .1418473 .8581527 .2286663 .1329037 .8835188 .1164812 .7948252 .0884983 .1166764	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 1.flu_s~2016 1.flu_s~2016 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 1.repeat_s~t 0.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 1.practice~w 0.reason_r~d 1.reason_r~d 1.patient_~d 2.patient_~d 3.patient_~d</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .026694 .3227885 .3277825 .3227885 .3227885 .3277825 .3227885 .3277825 .327785 .3277825 .327785 .327785 .327785 .327785 .327785 .327785 .327785 .327785 .33777777778 .3377787777777787777777777	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc 1.de~n_presc 1.de~n_presc 0.multip~RTI 1.multip~RTI 1.multip~RTI 1.practice~w 0.reason_r~d 1.patient_~d 2.patient_~d 3.patient_~d 3.pat_co~rti</pre>		$\begin{array}{c} .9303019\\ .0694981\\ .9\\ .1\\ .8825425\\ .1174575\\ .8798477\\ .1201523\\ .973306\\ .026694\\ .026694\\ .2\\ .6772115\\ .3227885\\ .3227865\\ .3227865\\ .3227865\\ .3227865\\ .3227865\\ .3227865\\ $	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 .agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc 1.de~n_presc 0.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.reason_r~d 1.reason_r~d 1.reason_r~d 1.patient_~d 2.patient_~d 3.patient_~d 3.patient_~d 3.patient_~d 3.pat_co~rti</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .2 .6772115 .3227885 .3227885 .3 .1418473 .8581527 .2286663 .1329037 .8835188 .1164812 .7948252 .0884983 .1166764 .8081625 .1529194 .0389182	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 1.flu_s~2016 1.flu_s~2016 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 0.reason_r~d 1.patient_~d 2.patient_~d 2.patient_~d 3.patient_~d 3.pat_co~rti 1.pat_co~rti 2.pat_co~rti 0.flu_s~2012</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .027885 .3278855 .3377855 .3377855 .3377855 .3377855 .33778555 .33778555 .33778555 .33778555 .33778555 .33778555 .33778555 .337785555 .337785555 .337785555 .337785555 .3377855555 .33778555555 .337785555555555555555555555555555555555	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc 1.de~n_presc 0.multip~RTI 1.multip~RTI 1.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.patient_~d 2.patient_~d 3.patient_~d 0.pat_co~rti 1.pat_co~rti 2.pat_co~rti 2.pat_corti 0.flu_s~2012</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .2 .6772115 .3227885 .3227885 .3 .1418473 .8581527 .2286663 .7713337 .8670963 .1329037 .8835188 .1166764 .8081625 .1529194 .0389182 .923472 .076528	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 0.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.de~n_presc 0.de~n_presc 0.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 0.reason_r~d 1.reason_r~d 1.reason_r~d 1.reason_r~d 1.patient_~d 2.patient_~d 3.patient_~d 3.patient_~d 0.pat_co~rti 1.pat_co~rti 2.pat_co~rti 0.flu_s~2012 0.flu_s~2013</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .2 .6772115 .3227885 .322785 .322785 .3227885785 .327787777777777777777777777777777777777	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 1.repeat_s~t 0.de~n_presc 0.multip~RTI 0.practice~w 0.reason_r~d 1.patient_~d 2.patient_~d 2.patient_~d 2.patient_~d 2.pat_co~rti 1.pat_co~rti 0.flu_s~2012 1.flu_s~2013</pre>		.0694981 .0694981 .9 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .3227885 .3227885 .3227885 .3227885 .3227885 .141473 .8581527 .2286663 .1329037 .8855188 .1164812 .7948252 .0884983 .1166764 .8081625 .1529194 .0389182 .923472 .076528 .9305019 .0694981	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc 1.de~n_presc 0.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.patient_~d 2.patient_~d 3.patien</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .2 .6772115 .3227885 .3 .1418473 .8581527 .2286663 .7713337 .8670963 .1329037 .8835188 .1164764 .8835188 .1166764 .8835188 .1166764 .0884983 .1166764 .0884983 .1529194 .0389182 .923472 .076528 .9305019 .0694981	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 1.flu_s~2015 0.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.gegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 0.de~n_presc 0.de~n_presc 0.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 0.reason_r~d 1.reason_r~d 1.reason_r~d 1.reason_r~d 1.patient_~d 2.patient_~d 2.patient_~d 3.patient</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .0389182 .923472 .076528 .9305019 .0694981 .9305019 .0694981 .9305019 .0694981 .0	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t 1.repeat_s~t 1.de~n_presc 0.multip~RTI 0.practice~w 1.practice~w 0.reason_r~d 1.patient_~d 2.patient_~d 2.patient_~d 2.patient_~d 2.pat_co~rti 1.pat_co~rti 1.pat_corti 0.flu_s~2012 1.flu_s~2013 0.flu_s~2014 0.flu_s~2014 0.flu_s~2014 0.flu_s~2014</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .2 .6772115 .3227885 .322785 .327855 .327855 .327855 .327855 .327855 .327855 .3278555 .3278555555555555555555555555555555555555	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 0.flu_s~2015 1.flu_s~2015 0.flu_s~2016 0.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 agegrp_urti_new 0.repeat_s~t 1.repeat_s~t condition_urti 0.de~n_presc 1.de~n_presc 0.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.patient_~d 0.patient_~d 3.flu_s~2012 0.flu_s~2015</pre>		.0694981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .2 .6772115 .3227885 .3 .1418473 .8581527 .2286663 .7713337 .8670963 .1329037 .8835188 .1166764 .889182 .1529194 .0389182 .923472 .076528 .9305019 .0694981 .0694981 .9 .88254255 .9 .0694981 .1 .88254255 .1	(mean) (m
6at :	<pre>1.flu_s~2013 0.flu_s~2014 1.flu_s~2014 1.flu_s~2015 0.flu_s~2015 0.flu_s~2016 1.flu_s~2016 0.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 1.flu_s~2017 0.maltip~RTI 1.multip~RTI 0.multip~RTI 1.multip~RTI 0.multip~RTI 1.multip~RTI 0.practice~w 0.reason_r~d 1.reason_r~d 1.reason_r~d 1.patient_~d 2.patient_~d 3.patient_~d</pre>		.90001981 .0694981 .9 .1 .8825425 .1174575 .8798477 .1201523 .973306 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .026694 .0389182 .0884983 .1166764 .0389182 .923472 .076528 .9305019 .0694981 .9305019 .0694981 .1 .8825425 .1174575 .8798477	(mean) (m

(9-21 yrs)

(9-21 yrs)

(9-21 yrs)

(AOM)

(Pharyngitis)

(Rhinosinusitis)

12004
5670
8 10 11
134
16 17 18 19
20 21 22 23
24 25 26
28 29 30
2004 2004
767 777 770
39 40 41 42
434 445 46
47 489 50
51 52 53 54
55 56 57 58
59 60 62
645 665
67 689 690
71 72 73
75 76 72
79 80 81
823 800 800 800 800 800 800 800 800 800 80
86 87 889
90 91 92
9996 9996
98 100
102

	0.flu_s~2017	=	.973306	(mean)
7at	: agegrp_urti_new	=	.020094	(mean)
	0.repeat_s~t 1.repeat_s~t	=	.6772115	(mean) (mean)
	condition_urti 0.de~n_presc	=	1 .1418473	(mean)
	1.de~n_presc	=	.8581527	(mean)
	1.multip~RTI	=	.7713337	(mean)
	1.practice~w	=	.1329037	(mean)
	0.reason_r~d 1.reason_r~d	=	.8835188	(mean) (mean)
	1.patient_~d 2.patient ~d	=	.7948252	(mean) (mean)
	3.patient_~d	=	.1166764	(mean)
	1.pat_co~rti	=	.1529194	(mean)
	0.flu_s~2012	=	.923472	(mean)
	1.flu_s~2012 0.flu_s~2013	=	.076528	(mean) (mean)
	1.flu_s~2013 0.flu_s~2014	=	.0694981	(mean) (mean)
	1.flu_s~2014	=	.1	(mean)
	1.flu_s~2015	=	.1174575	(mean)
	0.flu_s~2016 1.flu_s~2016	=	.8/984// .1201523	(mean) (mean)
	0.flu_s~2017 1.flu_s~2017	=	.973306	(mean) (mean)
8at	: agegrp_urti_new	=	3 6772115	(mean)
	1.repeat_s~t	=	.3227885	(mean)
	0.de~n_presc	=	.1418473	(mean)
	1.de~n_presc 0.multip~RTI	=	.8581527	(mean) (mean)
	1.multip~RTI 0.practice~w	=	.7713337	(mean)
	1.practice~w	=	.1329037	(mean)
	1.reason_r~d	=	.1164812	(mean)
	1.patient_~d 2.patient_~d	=	.7948252	(mean) (mean)
	3.patient_~d 0.pat co~rti	=	.1166764	(mean) (mean)
	1.pat_co~rti	=	.1529194	(mean)
	0.flu_s~2012	=	.923472	(mean)
	1.flu_s~2012 0.flu_s~2013	=	.9305019	(mean) (mean)
	1.flu_s~2013 0.flu_s~2014	=	.0694981	(mean) (mean)
	1.flu_s~2014	=	.1	(mean)
	1.flu_s~2015	=	.1174575	(mean)
	1.flu_s~2016	=	.1201523	(mean)
	0.flu_s~2017 1.flu_s~2017	=	.973306	(mean) (mean)
9at	: agegrp_urti_new 0.repeat s~t	=	3 .6772115	(mean)
	1.repeat_s~t	=	.3227885	(mean)
	0.de~n_presc	=	.1418473	(mean)
	0.multip~RTI	=	.2286663	(mean)
	1.multıp~RTI 0.practice~w	=	.7713337 .8670963	(mean) (mean)
	1.practice~w 0.reason r~d	=	.1329037	(mean) (mean)
	1.reason_r~d 1 natient ~d	=	.1164812	(mean)
	2.patient_~d	=	.0884983	(mean)
	0.pat_co~rti	=	.8081625	(mean)
	1.pat_co~rti 2.pat_co~rti	=	.1529194 .0389182	(mean) (mean)
	0.flu_s~2012 1.flu_s~2012	=	.923472	(mean) (mean)
	0.flu_s~2013	=	.9305019	(mean)
	0.flu_s~2014	=	.0054501	(mean)
	0.flu_s~2014	=	.8825425	(mean)
	1.flu_s~2015 0.flu_s~2016	=	.1174575 .8798477	(mean) (mean)
	1.flu_s~2016 0.flu_s~2017	=	.1201523	(mean)
10 a+	1.flu_s~2017	=	.026694	(mean)
10al	0.repeat_s~t	=	.6772115	(mean)
	<pre></pre>	=	. 5227885	(mean)
	0.de~n_presc 1.de~n_presc	=	.1418473 .8581527	(mean) (mean)
	0.multip~RTI 1.multip~RTI	=	.2286663	(mean) (mean)
	0.practice~w	=	.8670963	(mean)
	I.practice~w	-	. 102 20 27	(incall)

(21-34 yrs)

(21-34 yrs)

(Pharyngitis)

(21-34 yrs)

(AOM)

(Rhinosinusitis)

(35+ yrs)

(Rhinosinusitis)

1 2
n452
78
10 11
13 14 15
16 18
19 20 21
26
29 30
32 33 34
35
38
43 43 44
45 46 47
48 50
52
55 56 57
58 59 60
62 63
65
68 69 Z0
/1 72 73
75 76
28 79 80
81

	0.reason_r~d	=	.8835188 (mean)
	1.reason_r~d	=	.1164812 (mean)
	2 patient_~d	=	.7948252 (mean)
	3 nationt ad	_	1166764 (mean)
	0.pat co~rti	-	.8081625 (mean)
	1.pat co~rti	=	.1529194 (mean)
	2.pat_co~rti	=	.0389182 (mean)
	0.flu_s~2012	=	.923472 (mean)
	1.flu_s~2012	=	.076528 (mean)
	0.flu_s~2013	=	.9305019 (mean)
	0 flu s 2013	_	.0094961 (mean)
	$1.flu s \sim 2014$	_	.1 (mean)
	0.flu_s~2015	=	.8825425 (mean)
	1.flu_s~2015	=	.1174575 (mean)
	0.f]u_s~2016	=	.8798477 (mean)
	$1.TIU_S \sim 2016$	=	.1201523 (mean)
	1 flu s 2017	_	.973300 (mean)
11. at	: agegrp urti new	_	4
11uc	0.repeat_s~t	=	.6772115 (mean)
	1.repeat_s~t	=	.3227885 (mean)
	condition_urti	=	2
	0.de~n_presc	=	.1418473 (mean)
	1.de~n_presc	=	.8581527 (mean)
	1 multipeRT	_	.2200003 (mean)
	0 practice~w	_	8670963 (mean)
	1.practice~w	=	.1329037 (mean)
	0.reason_r~d	=	.8835188 (mean)
	1.reason_r~d	=	.1164812 (mean)
	1.patient_~d	=	.7948252 (mean)
	2.patient_~d	=	.0884983 (mean)
	0 pat co~rti	=	8081625 (mean)
	1.pat_co~rti	-	.1529194 (mean)
	2.pat_co~rti	=	.0389182 (mean)
	0.flu_s~2012	=	.923472 (mean)
	1.flu_s~2012	=	.076528 (mean)
	0.flu_s~2013	=	.9305019 (mean)
	$1.11u_s \sim 2013$ 0 flu s ~ 2014	=	.0094961 (mean) 9 (mean)
	1.flu_s~2014	=	.1 (mean)
	0.flu_s~2015	=	.8825425 (mean)
	1.flu_s~2015	=	.1174575 (mean)
	0.flu_s~2016	=	.8798477 (mean)
	1.710_{s}^{2016}	=	.1201523 (mean) 973306 (mean)
	$1.flu s \sim 2017$	_	.026694 (mean)
12at	: agegrp_urti_new	=	4
_	0.repeat_s~t	=	.6772115 (mean)
	1.repeat_s~t	=	.3227885 (mean)
	condition_urti	=	3
	0.de~n_presc	=	.1418473 (mean) 8581527 (mean)
	0 multip~RTT	_	2286663 (mean)
	1.multip~RTI	=	.7713337 (mean)
	0.practice~w	=	.8670963 (mean)
	1.practice~w	=	.1329037 (mean)
	0.reason_r~d	=	.8835188 (mean)
	1.reason_r~d	=	.1164812 (mean)
	2 patient ad	_	.7946252 (mean)
	3.patient ~d	_	.1166764 (mean)
	0.pat_co~rti	=	.8081625 (mean)
	1.pat_co~rti	=	.1529194 (mean)
	2.pat_co~rti	=	.0389182 (mean)
	0.TIU_S~2012	=	.923472 (mean)
	0.flu_s~2012	-	.9305019 (mean)
	1.flu_s~2013	=	.0694981 (mean)
	0.f]u_s~2014	=	.9 (mean)
	1.flu_s~2014	=	.1 (mean)
	0.tlu_s~2015	=	.8825425 (mean)
	1.11u_s~2015 0 flu_s~2016	_	.11/43/3 (IIIedii) 8798477 (mean)
	1.flu_s~2016	=	.1201523 (mean)
	0.flu_s~2017	=	.973306 (mean)
	1.flu_s~2017	=	.026694 (mean)

(35+ yrs)

(Pharyngitis)

(35+ yrs)

(AOM)

Where 1= First-line, 2= Second-line and 3= Not Recommended.

C.7.5 Predictive margins for the effect on ordinal choice, at specific values of whether repeats were issued on prescription and patient age group

. margins, a	t(repeat_script==(1 0) agegrp_urti_new==(1 2 3 4))
Predictive m Model VCE	argins Number of obs = 51,210 : OIM
1predict 2predict 3predict	: Marginal predicted mean (1.choice_urti_new), predict(pr outcome(1)) : Marginal predicted mean (2.choice_urti_new), predict(pr outcome(2)) : Marginal predicted mean (3.choice_urti_new), predict(pr outcome(3))
1at	: agegrp_urti_new = 1 (0-8 yrs) repeat_script = 1 (Positive)
2at	: agegrp_urti_new = 1 (0-8 yrs) repeat_script = 0 (Negative)
3at	: agegrp_urti_new = 2 (9-21 yrs) repeat_script = 1 (Positive)
4at	: agegrp_urti_new = 2 (9-21 yrs) repeat_script = 0 (Negative)
5at	: agegrp_urti_new = 3 (22-34 yrs) repeat_script = 1 (Positive)
6at	: agegrp_urti_new = 3 (22-34 yrs) repeat_script = 0 (Negative)
7at	: agegrp_urti_new = 4 (35+ yrs) repeat_script = 1 (Positive)
8at	: agegrp_urti_new = 4 (35+ yrs) repeat_script = 0 (Negative)

	[Margin	Delta-method Std. Err.	z	P> z	[95% Conf.	Interval]
_predict#_at 1 1 1 2 1 3 1 4 1 5 1 6 1 7 1 8 2 1 2 2 2 3 2 4 2 5 2 6 2 7 2 8 3 1 3 2 3 3 3 4	.5124754 .5376436 .4389829 .5403377 .3736665 .5255476 .3839309 .4650984 .1631334 .1601126 .169565 .160967 .168535 .1628917 .162817 .1628171 .3243912 .3022438 .3914521 .2986953	.0107828 .0099114 .0116898 .0104434 .0114366 .0103313 .0105993 .0101089 .0026331 .0026161 .0025937 .0027073 .0027073 .0027073 .0025603 .0025603 .0025888 .0097768 .0088161 .0113274 .0092197	$\begin{array}{c} 47.53\\54.24\\37.55\\51.74\\32.67\\50.87\\36.22\\46.01\\61.95\\61.20\\65.37\\59.46\\65.53\\60.76\\66.05\\64.74\\33.18\\34.28\\34.28\\34.56\\32.40\end{array}$	$\begin{array}{c} 0.000\\ 0.$.4913416 .5182176 .4160713 .519869 .3512511 .5052985 .3631567 .4452853 .1579725 .1549852 .1644814 .1556608 .163803 .1576371 .164096 .1631535 .3052289 .2849646 .3692507 .2806249	5336093 5570696 4618945 5608063 3960819 5457966 4047051 4849115 1682942 1652401 1746486 1662732 1739041 1746486 1662732 1739041 3435535 319523 3436535 3167657
35 36 37 38	.45748 .3115608 .4469549 .3666545	.0118457 .009246 .0108476 .0095796	38.62 33.70 41.20 38.27	0.000 0.000 0.000 0.000	.4342629 .2934389 .4256941 .3478789	.4806971 .3296827 .4682158 .3854302

Where 1= First-line, 2= Second-line and 3= Not Recommended.

Model for Ordinal Line of Antibiotic Agent prescribed for initial URTI Diagnoses



35

Figure C-10: Predictive margins for the effect on the probability of each of the three outcomes of ordinal choice of antibiotic occurring, at specific values of whether repeats were issued on prescription and patient age group, graphed by repeat on prescription status

40

41

C.7.6 Predictive margins for the effect on ordinal choice of antibiotic prescribed, at specific values of upper respiratory tract infection condition and patient age group, graphed by upper respiratory tract infection condition

margins, at(co	ndition_urti==(1	2 3) ag	egrp_urti	_new==(1 2	3 4))	
Predictive mar Model VCE :	gins OIM			Number of	obs =	51,210
1predict : 2predict : 3predict :	Marginal predict Marginal predict Marginal predict	ed mean ed mean ed mean	(1.choic (2.choic (3.choic	e_urti_new) e_urti_new) e_urti_new)	<pre>, predict(, predict(, predict())</pre>	<pre>(pr outcome(1)) (pr outcome(2)) (pr outcome(3))</pre>
1at :	agegrp_urti_new condition_urti	=	1 1	(0-8 yr (Rhinos	rs) sinusitis)	
2at :	agegrp_urti_new condition_urti	=	1 2	(0-8 yı (Pharyr	rs) ngitis)	
3at :	agegrp_urti_new condition_urti	=	1 3	(0-8 yr (AOM)	rs)	
4at :	agegrp_urti_new condition_urti	=	2 1	(9-21) (Rhinos	/rs) sinusitis)	
5at :	agegrp_urti_new condition_urti	=	2 2	(9-21 y (Pharyr	/rs) ngitis)	
6at :	agegrp_urti_new condition_urti	=	2 3	(9-21) (AOM_	rs)	
7at :	agegrp_urti_new condition_urti	=	3 1	(22-34 (Rhinos	yrs) sinusitis)	
8at :	agegrp_urti_new condition_urti	=	3 2	(22-34 (Pharyr	yrs) ngitis)	
9at :	agegrp_urti_new condition_urti	=	3 3	(22-34 (AOM)	yrs)	
10at :	agegrp_urti_new condition_urti	=	4 1	(35+ yr (Rhinos	rs) sinusitis)	
11at :	agegrp_urti_new condition_urti	=	4 2	(35+ yı (Pharyr	rs) ngitis)	
12at :	agegrp_urti_new condition_urti	=	4 3	(35+ yr (AOM)	rs)	
	 Delta	-method				
+	Margin Std	. Err.	Z	P> z	[95% Conf.	. Interval]
_predict#_at 1 1 1 1 2 1 3 1 4 1 5 1 6 1 7 1 8 1 9 1 10 1 11 1 12 2 2 2 3 2 4 2 3 2 4 2 3 3 3 3 4 3 3 3 10 3 12 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	$\begin{array}{ccccc} 5431247 & .01\\ .4616489 & .01\\ .6261724 & .00\\ .4819812 & .01\\ .4996952 & .01\\ .5664066 & .01\\ .471569 & .01\\ .4745298 & .01\\ .5308849 & .01\\ .4745298 & .01\\ .4745298 & .01\\ .4745298 & .01\\ .411107 & .01\\ .5103709 & .01\\ .613109 & .00\\ .1694569 & .00\\ .1669502 & .00\\ .1669502 & .00\\ .1663502 & .00\\ .1663502 & .00\\ .1663733 & .00\\ .1639462 & .00\\ .1639462 & .00\\ .169362 & .00\\ .1639462 & .00\\ .1693642 & .00\\ .1693642 & .00\\ .1693642 & .00\\ .1693642 & .00\\ .1693642 & .00\\ .1639462 & .00\\ .1693642 & .00\\ .1693641 & .01\\ .2276848 & .00\\ .3510686 & .01\\ .3348891 & .01\\ .2768641 & .00\\ .3595969 & .01\\ .308719 & .00\\ .4042122 & .00\\ .418874 & .01\\ .3249035 & .00\\ \end{array}$	04344 09319 97101 10951 00674 09891 00603 01544 07265 27472 26319 28054 26319 28054 26394 28157 25306 26394 28157 25306 26329 91483 003807 78896 02884 00682 02186 02402 02339 02186	$\begin{array}{c} 52.05\\ 42.23\\ 64.49\\ 43.56\\ 51.54\\ 41.95\\ 43.76\\ 47.99\\ 41.94\\ 37.30\\ 47.58\\ 58.72\\ 64.39\\ 52.09\\ 63.28\\ 65.12\\ 65.15\\ 59.79\\ 66.07\\ 66.12\\ 65.15\\ 59.79\\ 66.07\\ 66.12\\ 59.79\\ 66.07\\ 66.12\\ 59.79\\ 66.07\\ 66.12\\ 59.79\\ 66.07\\ 61.40\\ 32.31\\ 35.86\\ 34.12\\ 33.54\\ 28.86\\ 34.12\\ 33.45\\ 37.67\\ 35.19\\ 31.47\\ 40.69\\ 38.03\\ 33.52\\ \end{array}$	0.000 0	.5226736 .4402229 .6071409 .4602352 .4781995 .5448685 .426263 .433278 .5092051 .4059393 .389509 .4893474 .1559266 .1642986 .165155 .165352 .2776341 .3485483 .2122215 .3309038 .3151559 .2584366 .3654731 .3395687 .2894933 .3874853 .3059076	.5635757 .483075 .645204 .5211909 .5879448 .4680507 .4957816 .5525647 .445744 .4327123 .5313945 .1666952 .1746153 .1516412 .1705887 .162248 .1722719 .170887 .162248 .1722719 .1708636 .1656543 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749873 .1749874 .33999 .243148 .38712334 .3712334 .3712344 .3712344 .3712344 .37252916 .4055891 .379448 .4236807 .4404627 .440894

Where 1= First-line, 2= Second-line and 3= Not Recommended.



Figure C-11: Predictive margins for the effect on the probability of each of the three outcomes of ordinal choice of antibiotic occurring, at specific values of upper respiratory tract infection condition and patient age group, graphed by upper respiratory tract infection condition

C.8 Non-first-line antibiotic prescribing model for upper respiratory tract infection

C.8.1 Average marginal effects

Table C-23:Average marginal effects for the binary model for non-first-line antibiotic
prescribing for upper respiratory tract infection

	dy/dx	Std. Err.	Z	P>z	[95% Conf.	Interval]				
Patient Age Group (ref. 0-8 years)										
9-21 yrs	0.059784	0.00619	9.66	0.000	0.047652	0.071916				
22-34 yrs	0.125014	0.006257	19.98	0.000	0.112752	0.137277				
35+ yrs	0.169652	0.006081	27.90	0.000	0.157734	0.181571				
Repeat Prescription Status (ref. Negative)										
Positive	0.163342	0.005593	29.21	0.000	0.152381	0.174304				
URTI Condition (ref. Rhinosinusitis)										
Pharyngitis/Tonsillitis	-0.12464	0.005462	-22.82	0.000	-0.13534	-0.11393				
Acute Otitis Media	-0.04221	0.006147	-6.87	0.000	-0.05426	-0.03016				
Unnecessary/Necessary Prescription	Status (ref. Ne	ecessary)								
Unnecessary	0.005801	0.006556	0.88	0.376	-0.00705	0.018649				
Patient Gender (ref. Female)										
Male	0.01559	0.003963	3.93	0.000	0.007822	0.023357				
Patient with Multiple URTI Episodes ((ref. Negative))								
Positive	-0.01761	0.004851	-3.63	0.000	-0.02712	-0.00811				
Practice Size (ref. Medium/Large)										
Small	0.103144	0.028858	3.57	0.000	0.046584	0.159703				
Reason for Prescribing Recorded (ref	. Negative)									
Positive	-0.06702	0.014037	-4.77	0.000	-0.09453	-0.03951				
Patient Disadvantage Status (ref. Neg	gative)									
Positive	-0.016	0.008947	-1.79	0.074	-0.03353	0.001541				
Missing	-0.02988	0.008463	-3.53	0.000	-0.04647	-0.0133				
Patient Comorbid Condition Status (r	ef. Negative)									
Positive	0.022818	0.005685	4.01	0.000	0.011677	0.033959				
Missing	0.005486	0.049646	0.11	0.912	-0.09182	0.10279				

C.9 Repeats being issued on prescriptions model for upper respiratory tract

2 infection

C.9.1 Model

Table C-24:

: Mixed effects logit model for repeats being issued on antibiotic prescriptions for initial presentations of upper respiratory tract infection (Model 4)

Model	Repeat Positive Antibiotic Prescribing (ref. Repeat Negative)			
Independent Variable	Odds Ratio	95% C.I.		
Patient Age group, (ref 0-8 yrs)				
9-21 years	0.676***	[0.571, 0.801]		
22-34 years	0.425***	[0.358, 0.505]		
35+ years	0.790**	[0.680, 0.917]		
Patient penicillin sensitivity recorded, (ref Negative)				
Positive	1.228***	[1.116, 1.351]		
Unnecessary Prescribing status (ref Necessary)				
Unnecessary	0.827***	[0.765, 0.895]		
Patient Comorbid Condition (ref Negative)				
Positive	1.164***	[1.089, 1.245]		
Missing	0.964	[0.473, 1.965]		
URTI Condition, (ref rhinosinusitis)				
Pharyngitis/Tonsillitits	1.654***	[1.476, 1.853]		
AOM	1.265***	[1.133, 1.411]		
Ordinal Line Agent Prescribed, (ref First-line)	0.00-444	[0.000.0.070]		
Second-line	2.627***	[2.323, 2.972]		
Not Recommended	1.035	[0.938, 1.141]		
Detient Ann Creans II Online Line Described				
Patient Age Group # Ordinal Line Prescribed	2.042***	[4 640 2 522]		
9-21 yrs#Second-line	2.043***	[1.648, 2.533]		
9-21 yrs#Not Recommended	2.2/2***	[1.942, 2.658]		
22-34 yrs#Second-line	3.015***			
22-34 yrs#Not Recommended	4.315			
35+ yrs#Second-line	2.222			
35+ yrs#Not Recommended	2.934	[2.524, 3.411]		
Patient Age Group # LIPTI Condition				
9-21 vrs#Pharvngitis/Tonsillitis	0.630***	[0 528 0 752]		
9-21 vrs#Acute Otitis Media	1 628***	[1 3/0 1 979]		
22-34 vrs#Pharvngitis/Tonsillitis	0.528***	[0 445 0 627]		
22-34 yrs#Acute Otitis Media	1 685***			
35+ vrs#Pharvngitis/Tonsillitis	0.429***	[0.367, 0.502]		
35+ yrs#Acute Otitis Media	1.242*	[1.046, 1.476]		
		[
Patient's Primary Health Network, (ref Perth North)				
Perth South	1.188*	[1.026, 1.376]		
Country WA	1.071	[0.915, 1.253]		
Interstate PHN	1.132	[0.865, 1.480]		
Patient's PHN Missing	1.045	[0.847, 1.288]		
Seasonality allowed for in form of dummy vari	ables for annual influen	iza seasons		
var(_cons)	2.574 (0.18501)	[2.236, 2.963]		
Ν	51210			
AIC	44371.6			
BIC	44690			
ICC				
Level: Unique combination of provider ID & practice ID	0.439 (0.01770)	[0.405, 0.474]		
Note: SE in parentneses, **** p<0.01, *** p<0.05, * p<0.1				

- 1
- 2

3 Patients with a history of comorbid conditions were two percentage points more likely to 4 receive a repeat than patients without comorbid conditions (0.021, p<0.001, 95%CI: 0.012, 5 0.030). Meanwhile patients with missing comorbid condition status were not significantly different from patients without comorbid conditions (-0.00488, p=0.920, 95%CI: -0.100, 6 7 0.091). Patients with a recorded sensitivity to penicillin were three percentage points more likely to receive a repeat on prescription than patients with no such history (0.029, p<0.001, 8 95%CI: 0.015, 0.042). Having temperature recording during the consultation was associated 9 with a one percentage point increase in likelihood of receiving a repeat on the prescription 10 11 (0.0136, p=0.011, 95%CI: 0.003, 0.024), compared with patients who did not receive temperature testing. Compared to patients residing within the Perth North PHN, patients 12 13 residing in Perth South PHN were two percentage points more likely to receive a repeat (0.024, p=0.022, 0.003, 0.044). There were no significant differences for patients residing 14 within the Country WA PHN (0.00925, p=0.392, 95%CI: -0.012, 0.030), an interstate PHN 15 (0.0168, p=0.372, 95%CI: -0.020, 0.054), or with missing PHN information (0.00586, 16 *p*=0.684, 95%*CI*: -0.022, 0.034). 17

C.9.2 Model explanation 1

2

Patient age, URTI condition, comorbid condition status, temperature recording status, 3 4 patient penicillin sensitivity status, and patient's PHN were found to affect the risk of repeat prescribing, in addition to ordinal line of choice of agent prescribed and 5 unnecessary/necessary prescribing status (Table C-24, Table C-25). The line of antibiotic 6 prescribed was found to have the most notable effect on the probability of repeats being 7 prescribed, more so than unnecessary/necessary prescribing status. The following variables 8 were insignificant: patient gender, disadvantage, concession status, mental health 9 conditions, number of URTI episodes, remoteness and accessibility, practice size. 10

11

The probability of receiving a repeat was three percentage points less for patients receiving 12 13 unnecessary prescriptions (-0.026, p<0.001, 95%CI: -0.037, -0.015) than patients receiving necessary prescriptions. In this context, unnecessary prescribing may potentially be 14 considered to predispose patients to receiving prescriptions without repeats, and vice versa. 15

16

There was an effect modification between patient age group and the ordinal choice of 17 antibiotic prescribed. When first-line antibiotics were prescribed, the probability of receiving 18 19 a repeat was 25% for patients aged 0-8 years (0.252, p<0.001, 95%CI: 0.233, 0.272), 19% for patients aged 9-21 years (0.194, p<0.001, 95%CI: 0.175, 0.214), 14% for patients aged 20 22-34 years (0.140, p<0.001, 95%CI: 0.123, 0.157) and 19% for patients 35 years and over 21 (0.188, p<0.001, 95%CI: 0.169, 0.207). Where second-line antibiotics were prescribed, the 22 chance of receiving a repeat was 40% for patients 0-8 years (0.398, p<0.001, 95%CI: 0.371, 23 0.426), 44% for patients 9-21 years (0.440, p<0.001, 95%CI: 0.407, 0.474), and 42% for 24 patients 22-34 years (0.418, p<0.001, 95%CI: 0.387, 0.448) and 45% for patients 35 years 25 and over (0.445, p<0.001, 95%CI: 0.418, 0.473). Not recommended antibiotics being 26 prescribed had a probability of 26% of a repeat being issued for patients aged 0-8 years 27 (0.257, p<0.001, 95%CI: 0.237, 0.277), 31% for patients 9-21 years (0.308, p<0.001, 95%CI: 28 0.284, 0.331), 33% for patients 22-34 years (0.325, p<0.001, 95%CI: 0.302, 0.349) and 34% 29 for patients 35 years and over (0.338, p<0.001, 95%Cl: 0.316, 0.361). 30

31

32 There was an interaction identified between URTI condition and patient age group. Patients with rhinosinusitis were most likely to receive a repeat when aged 35 years and over (0.288, 33 p<0.001, 95%CI: 0.267, 0.310) and least likely to receive a repeat at 22-34 years of age 34 (0.232, p<0.001, 95%CI: 0.210, 0.253). Patients with pharyngitis aged 0-8 years were most 35

- likely to receive a repeat on prescription (0.314, p<0.001, 95%Cl: 0.292, 0.337) and least
 likely aged 22-34 years (0.215, p<0.001, 95%Cl: 0.195, 0.234). For AOM, the risk of
 receiving a repeat was highest for patients aged 9-21 (0.357, p<0.001, 95%Cl: 0.327,
 0.3863) or 35 years and over (0.357, p<0.001, 95%Cl: 0.328, 0.385) and lowest for patients
 0-8 years (0.275, p<0.001, 95%Cl: 0.254, 0.296).
- 6



9 **C.9.2.1** Summary

10

Repeats on prescriptions were most likely for patients with AOM while patients with 11 rhinosinusitis were least likely. For pharyngitis, the risk of repeat prescribing decreased with 12 increasing patient age. For rhino, highest for patients 35 years and over and lowest for 13 patients 22-34 years. For AOM, highest for 9-21 and 35+ and lowest for children 0-8years. 14 The line of antibiotic prescribed was found to have the most notable effect on the probability 15 16 of repeats being prescribed, more so than unnecessary/necessary prescribing status. Unnecessary prescribing appeared linked to receiving prescriptions without repeats on 17 them. Of the variance not explained by fixed effects, the unique provider and practice 18 combination was responsible for 44% of this variance. 19

C.9.3 Marginal effects for the repeats being issued on prescriptions model for upper respiratory tract infection

C.9.3.1 Average marginal effects

 Table C-25:
 Average marginal effects for Model 4 – Repeats being issued on antibiotic prescriptions

Average Marginal Effects									
Repeat Positive Antibiotic Prescribing for Initial Presentations of URTI									
Independent Variable	dy/dx	Std. Err.	Z	P>z	[95% Conf.	Interval]			
Patient Age Group (ref. 0-8 years)								
9-21 yrs	0.003801	0.005181	0.73	0.463	-0.00635	0.013954			
22-34 yrs	-0.01572	0.005234	-3.00	0.003	-0.02598	-0.00546			
35+ yrs	0.014866	0.004921	3.02	0.003	0.005222	0.024511			
URTI Condition (ref. Rhinosinusiti	is)								
Pharyngitis/Tonsillitis	0.007798	0.004622	1.69	0.092	-0.00126	0.016857			
Acute Otitis Media	0.069954	0.005762	12.14	0.000	0.058662	0.081247			
Unnecessary/Necessary Prescript	tion Status (ref.	Necessary)							
Unnecessary	-0.02619	0.005622	-4.66	0.000	-0.03721	-0.01517			
Ordinal Line of Antibiotic Prescrib	oed (ref. First-li	ne)							
Second-line	0.219129	0.007865	27.86	0.000	0.203715	0.234544			
Not Recommended	0.100171	0.004776	20.98	0.000	0.090811	0.109531			
Patient Comorbid Condition Statu	us (ref. Negative	e)							
Positive	0.020951	0.004794	4.37	0.000	0.011555	0.030347			
Missing	-0.00488	0.048728	-0.10	0.920	-0.10038	0.09063			
Penicillin Sensitivity (ref. Negative	e)								
Positive	0.028505	0.006935	4.11	0.000	0.014914	0.042097			
Temperature Recording Status (r	ef. Negative)								
Positive	0.01364	0.005379	2.54	0.011	0.003098	0.024182			
Patient Primary Health Network	(ref. Perth Nort	h)							
Perth South	0.023514	0.010252	2.29	0.022	0.003422	0.043607			
Country WA	0.009245	0.010809	0.86	0.392	-0.01194	0.03043			
Interstate	0.01678	0.018807	0.89	0.372	-0.02008	0.053641			
Missing	0.005863	0.014407	0.41	0.684	-0.02237	0.034099			

C.9.3.2 Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in upper respiratory tract infection condition, across different values of patient age group

margins, dydx(condition_urti) at (agegrp_urti_new= (1 2 3 4)) Average marginal effects Number of obs 51.210 Model VCE : OIM Expression : Marginal predicted mean, predict() : 2.condition_urti 3.condition_urti dy/dx w.r.t. 1._at agegrp_urti_new = 1 (0-8 yrs) 2._at 3._at agegrp_urti_new agegrp_urti_new (9-21 yrs) (22-34 yrs) 2 3 = 4 4._at agegrp_urti_new = (35+ yrs) Delta-method dy/dx Std. Err. z P>|z| [95% Conf. Interval] Rhinosinusitis (base outcome) Pharyngitis at .0703965 .0054493 .0164958 .0081609 .0095991 .008412 8.63 0.57 -1.96 $\begin{array}{c} 0.000 \\ 0.570 \\ 0.050 \end{array}$ 0544014 .0863917 123 -.0133646 .0242631 -8.59e-06 4 .0460829 .0079403 -5.80 0.000 .0616455 .0305203 AOM _at 1 2 .0075594 .0125338 .0135503 .0106691 .0317151 4.20 8.25 7.64 6.21 0.000 .0168991 .0465312 34 1035001 0.000 .076942 .1300583 .0871931 .066282 Note: dy/dx for factor levels is the discrete change from the base level.

Variables that uniquely identify margins: agegrp_urti_new _deriv



44

1

2

Note: Effect on probability relative to that for rhinosinusitis.

Figure C-12: Margins at representative values for the effect on the probability of repeats being
issued on antibiotic prescriptions, with change in upper respiratory tract infection condition, across
different values of patient age group, relative to the probability of repeats being issued on

- 48 prescriptions for acute rhinosinusitis
- 49

C.9.3.2 Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of upper respiratory tract infection condition,

margins, dydx(agegrp_urti_new) at (condition_urti= (1 2 3))

Average marginal effects Model VCE : OIM Number of obs = 51,210 Expression : Marginal predicted mean, predict() dy/dx w.r.t. : 2.agegrp_urti_new 3.agegrp_urti_new 4.agegrp_urti_new (Rhinosinusitis) (Pharyngitis) (AOM) 1._at 2._at : condition_urti : condition_urti 12 3._at : condition_urti = 3 _____ ____ _____ Delta-method dy/dx [95% Conf. Interval] Std. Err. z P>|z| 0-8 years (base outcome) 9-21 years _at .0123803 -.0525669 .0086986 -.0046685 -.0687747 1.42 -6.36 7.27 $0.155 \\ 0.000 \\ 0.000$ 1 2 3 0294292 -.0363591 .0840365 .0115561 .0613869 .106686 22-34 years _at .007724 .0082588 .0135696 0.14 -10.39 5.37 $0.887 \\ 0.000 \\ 0.000$.0010992 1 2 3 -.0140396 .0162379 -.0857931 .0728842 -.1019801 .0462882 -.0696061 .0994802 35+ years _at .0509757 .0072739 .0082535 .0111936 $\begin{array}{c} 0.000\\ 0.000\\ 0.000\\ 0.000 \end{array}$ 7.01 -7.94 7.64 .0652324 -.0493272 .1074817 .0367191 1 2 3 -.0816802 .0855426 .0636036 Note: dy/dx for factor levels is the discrete change from the base level.

Variables that uniquely identify margins: condition_urti _deriv



Note: Effects on probability relative to the effect for patients 0-8 years.

Figure C-13: Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of upper respiratory tract infection condition, relative to the effect of repeats being issued on prescriptions for patients aged 0-8 years

C.9.3.2 Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of ordinal choice of antibiotic prescribed

Average margin Model VCE	nal ef : OIM	fects		Numb	er of obs	= 51	,210
Expression dy/dx w.r.t.	: Marg : 2.ag	inal predicted egrp_urti_new	l mean, pred [:] 3.agegrp_ur [:]	ict() ti_new 4	.agegrp_u	rti_new	
1at 2at 3at	: choi : choi : choi	ce_urti_new = ce_urti_new = ce_urti_new =	1 2 3		First-lin Second-lin Not Recom	e) ne) mended)	
		dy/dx	elta-method Std. Err.	 z	P> z	[95% Conf.	Interval]
0-8 years		(base outco	ome)				
9-21 years	at 1 2 3	 0569828 .0418153 .0512712	.0073293 .0150563 .0083028	-7.77 2.78 6.18	0.000 0.005 0.000	071348 .0123055 .0349981	0426176 .0713251 .0675444
22-34 years	at 1 2 3	1105408 .0193778 .0691983	.0076258 .0140068 .0083532	-14.50 1.38 8.28	0.000 0.167 0.000	1254871 008075 .0528262	0955944 .0468306 .0855703
35+ years	at _1	+ 0631149	.0074088	-8.52	0.000	0776358	048594

margins, dydx(agegrp_urti_new) at (choice_urti_new= (1 2 3))



Figure C-14: Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of ordinal choice of antibiotic prescribed, relative to the probability of repeats being issued on prescriptions for patients aged 0-8 years.

Second-line

35+ years

Ordinal Choice of Antibiotic Prescribed

9-21 years

Note: Effects on probability relative to the effect for patients 0-8 years.

22-34 years

C.9.3.2 Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in ordinal choice of antibiotic prescribed, across patient age groups

margins, dydx(c	hoice_urti_new) a	t (agegrp_urt	i_new =	(1 2 3 4))		
Average margina Model VCE : (l effects OIM		Numbe	er of obs	= 51	,210
Expression : dy/dx w.r.t. : 2	Marginal predicte 2.choice_urti_new	d mean, predi 3.choice_urt	ct() i_new			
1at :: 2at :: 3at :: 4at ::	agegrp_urti_new = agegrp_urti_new = agegrp_urti_new = agegrp_urti_new =	1 2 3 4		0-8 yrs) 0-21 yrs) 22-34 yrs) 35+ yrs)		
	dy/dx	Delta-method Std. Err.	 z	P> z	[95% Conf.	Interval]
First-line	(base outc	ome)				
Second-line	_at 1 .1454059 2 .244204 3 .2753245 4 .2552736	.0103101 .0153667 .0143587 .0117135	14.10 15.89 19.17 21.79	0.000 0.000 0.000 0.000	.1251985 .2140858 .247182 .2323156	.1656134 .2743223 .303467 .2782317
Not Recommended						

Not Recommended

First-line

12m45670000



$\frac{11}{12}$

13 Figure C-15: Margins at representative values for the effect on the probability of repeats being

issued on antibiotic prescriptions, with change in ordinal choice of antibiotic prescribed, across patient age groups, relative to the probability of repeats being issued on first-line prescriptions

- 16
- 17
- 18

1 C.10 Comparison with fixed-effects models

3

Table C-26: Three fixed-effects only comparisons for the inappropriate decision model

Inappropriate Decision Model	Mixed, two-le for patient combination pra	evel model (levels and for unique of provider ID & ctice ID)	s Fixed model with dummies for practice		nodel with Fixed model with s for practice practice	
	·					t-
	Exp. coeff.	t-statistic	Exp. coeff.	t-statistic	Exp. coeff.	statistic
Patient age group (ref. 0-8 yrs)						
	0.772***	-28.18	0.705***	-27.37	0.662***	-26.38
	0.797***	-30.31	0.689***	-27.83	0.590***	-24.64
	0.721	-28.77	0.001	-28.03	0.000	-20.50
Patient gender (ref. female)						
Male	0.0500*	-2.11	0.0452*	-2.02	0.0479*	-2.19
Patient age group ## Gender (ref. # femal	e)					
# Male	-0.207***	(-5.24)	-0.185***	(-4.96)	-0.181***	(-4.97)
# Male	-0.164***	(-4.32)	-0.179***	(-4.99)	-0.193***	(-5.50)
# Male	-0.0970**	(-2.74)	-0.0804*	(-2.42)	-0.0888**	(-2.73)
Patient penicillin sensitivity (ref_negative)						
Positive	0.351***	-11.85	0.330***	-11.85	0.327***	-12
Patient concessions status (ref. negative)						
Positive	0.0401*	-2.13	0.0483**	-2.74	0.0539**	-3.19
Patient mental health condition (ref. nega	tive)	2.42	0.0616**	2.02	0.0474*	2 41
Missing	0.0527	-2.43	0.0010	-3.03	0.0474	-2.41
IVIISSIIIg	0.274	-1.25	0.747	-5.10	0.499	-13.04
Weekend consultation (ref. negative)						
Positive	0.216***	-8.81	0.300***	-14.08	0.238***	-11.66
Practice Size (ref. Medium/Large)						
Small	0.518***	-3.97	0.271*	-2.41	0.0761***	-3.92
Patient PHN (ref. Perth North)						
Perth South	-0.0472	(-1.16)	-0.407***	(-10.71)	-0.143***	(-9.58)
Country WA	0.0336	-0.73	-0.107*	(-2.24)	0.0272	-1.38
Interstate	0.190*	-2.47	0.0045	-0.06	0.156*	-2.32
Missing	0.125	-1.9	-0.0888	(-1.63)	0.139*	-1.99
Patient disadvantage status (ref. negative)	0.0614*	1.06			0 174***	(7.27)
Missing	0.0014 ·	-1.90			-0.174***	(-7.37) (-5.83)
Wissing	-0.0817	(-2.30)			-0.125	(-5.85)
Patient with multiple URTI episodes (ref. n	egative)					
Positive			-0.0505**	(-3.18)		
Patient in remote area (ref. negative)						-
Positive					0.215***	-6
wissing					-0.121	(-1.16)
cons	-0.992***	(-19.50)	-0.763***	(-10.13)	-0.800***	(-42.34)
	0.002	(15.50)	0.700	(10.10)	0.000	(12.54)
var(_cons[~c	1.014***	-15.72				
N	111848		111848		111848	
Note: exponentiated coefficients, signific	ance: * p<0.05,	** p<0.01, ****p<0.	.001			

APPENDIX D – APPENDICES TO THE PREDICTORS OF INAPPROPRIATE PRESCRIBING FOR URINARY TRACT INFECTION CHAPTER (CHAPTER 5)

D.1 Clustering of patients and providers for initial presentations of urinary tract infection

Figure D-1: Bubble plot of clustering by inidividual patients, providers and practices for initial presentations of urinary tract infection



1 D.2 Antibiotic prescriptions

 Table D-1:
 Frequency table of antibiotic prescriptions for initial presentations of urinary tract infection, by Anatomical Therapeutic Chemical classification

Anatomical Therapeutic Classification class	Frequency	Percent	Cumulative Percent
Beta-lactamase resistant penicillins	5	0.03	0.03
Beta-lactamase sensitive penicillins	8	0.04	0.07
Combinations of penicillins, incl. be	1,402	7.8	7.87
Combinations of sulfonamides and trim	336	1.87	9.74
First-generation cephalosporins	7,374	41.03	50.77
Fluoroquinolones	498	2.77	53.54
Lincosamides	3	0.02	53.56
Macrolides	46	0.26	53.81
Nitrofuran derivatives	702	3.91	57.72
Penicillins with extended spectrum	511	2.84	60.56
Second-generation cephalosporins	94	0.52	61.09
Tetracyclines	17	0.09	61.18
Third-generation cephalosporins	1	0.01	61.19
Trimethoprim and derivatives	6,976	38.81	100
Total	17,973	100	

 Table D-2:Frequency table of antibiotic active ingredients prescribed but not recommended in
the guidelines for initial presentations of urinary tract infection

Active ingredient	Frequency	Percent ⁹
Amoxicillin	510	32.57
Ampicillin	1	0].0] 6
Azithromycin	11	0.7
Cefaclor	83	12 5.3
Ceftriaxone	1	0 1.9 6
Cefuroxime	11	0.7
Ciprofloxacin	68	4.34
Clarithromycin	10	0 1.6 4
Clindamycin	3	0.19
Doxycycline	17	1.09
Erythromycin	16	1 <u>1</u> , 0 2
Flucloxacillin	5	0.32
Nitrofurantoin	628	18 40.1
Phenoxymethylpenicillin	8	0 <u>1.9</u> 1
Roxithromycin	9	0.57
Trimethoprim with sulfamethoxazole	185	11.81
Total	1,566	100 21

Table D-3:

Frequency table of ordinal choice of antibiotic prescribed, and primary health network, for antibiotic prescriptions for initial presentations of urinary tract infection

		·	•	Not	3	
	First-line	Second-line	Third-line	Recommended	Total	
Primary Health						
Network						Key
Perth North	2,603	2,804	787	724	6,918	frequency
	37.63	40.53	11.38	10.47	100	row percentage
Perth South	2.413	2.219	652	535	5.819	
	41.47	38.13	11.2	9.19	100	
C 1 1 1 1	4.004	2 4 2 5	100	260		
Country WA	1,904	2,125	406	269	4,704	
	40.48	45.17	8.63	5.72	100	
Interstate	93	108	21	6	228	
	40.79	47.37	9.21	2.63	100	
Missing	114	118	40	32	304	
1411351116	37.5	38.82	13.16	10.53	100	
Total	7,127	7,374	1,906	1,566	17,973	
1	39.65	41.03	10.6	8.71	100	

Figure D-2: Bar graph of ordinal line of antibiotic prescribed for initial presentations of urinary tract infection, graphed by patient's primary health network



D.3 Ordinal choice of antibiotic prescribed for urinary tract infection

Table D-4:Frequency table of first-line and non-first-line antibiotic prescriptions, for initial
presentations of urinary tract infection, by patient age group, including a ratio of non-
first-line to first-line prescriptions

		Ratio of non-first-line					
	First-line	Non-first-line	to first-line	Total			
Patient age group							
45+ yrs, ref	3,565	4,582	1.3	8,147			
16-44 yrs	3,229	4,724	1.5	7,953			
6-15 yrs	215	779	3.6	994			
0-5 yrs	118	761	6.4	879			
Total	7,127	10,846	1.5	17,973			

 Table D-5:
 Frequency table of ordinal choice of antibiotic prescribed, for initial presentations of urinary tract infection, by patient age group

	2		Third-line/	0 0 1		
	First-line	Second-line	Last Resort	Not Recommended	Total	
Patient Age						Кеу
Group						frequency
45+ yrs	3,565	3,034	916	632	8,147	row percentag
	43.76	37.24	11.24	7.76	100	
16-44 yrs	3,229	3,288	697	739	7,953	
	40.6	41.34	8.76	9.29	100	
6-15 yrs	215	529	155	95	994	
	21.63	53.22	15.59	9.56	100	
0-5 yrs	118	523	138	100	879	
	13.42	59.5	15.7	11.38	100	
Total	7,127	7,374	1,906	1,566	17,973	
	39.65	41.03	10.6	8.71	100	

Table D-6:Frequency table of patient group by ordinal choice of antibiotic prescribed, for initial
presentations of urinary tract infection

			,		
	First-line	Second- line	Third-line	Not Recommended	12 Total
Patient Group					
Women	6,290	5,746	1,234	1,258	14,528
	43.3	39.55	8.49	8.66	100
Men	504	576	379	113	1,572
	32.06	36.64	24.11	7.19	100
Children <					
16yrs	333	1,052	293	195	1,873
	17.78	56.17	15.64	10.41	100
Total	7,127	7,374	1,906	1,566	17,973
	39.65	41.03	10.6	8.71	100

2 Table D-7: 3

D-7: Frequency table of ordinal choice of antibiotic prescribed by patient primary health network, for patients with initial presentations of urinary tract infection

	Perth	Perth	Country			
	North	South	WA	Interstate	Missing	Total
Choice of Antibiotic Prescribed						
First-line	2,603	2,413	1,904	93	114	7,127
	36.52	33.86	26.72	1.3	1.6	100
Second-line	2,804	2,219	2,125	108	118	7,374
	38.03	30.09	28.82	1.46	1.6	100
Third-line/Last						
Resort	787	652	406	21	40	1,906
	41.29	34.21	21.3	1.1	2.1	100
	724	535	269	6	32	1,566
Not Recommended	46.23	34.16	17.18	0.38	2.04	100
	6,918	5,819	4,704	228	304	17,973
Total	38.49	32.38	26.17	1.27	1.69	100

Кеу
frequency
row percentage

Table D-8:

-8: Frequency table of ordinal choice of antibiotic prescribed by patient primary health network, for patients with initial presentations of urinary tract infection

	Small	Medium/Large		
	Practice	Practice	Total	
Ordinal Choice of				
Antibiotic Prescribed				Key
First line	6 652	474	7 1 2 7	frequency
riist-iine	0,055	474	7,127	row perce
	93.35	6.65	100	
Second line	6 567	007	7 274	
Second-line	0,007	807	7,374	
	89.06	10.94	100	
Third line/Last resor	1 750	147	1 006	
Third-line/Last Tesor	1,759	147	1,900	
	92.29	7.71	100	
Not Recommended	1 432	134	1 566	
not neconinclued	1,752	154	1,500	
	91.44	8.56	100	
Total	16,411	1,562	17,973	
	91.31	8.69	100	
Figure D-3: Bar graph of ordinal choice of antibiotic prescribed for initial presentations of urinary tract infection, by (proxy for) practice size



. D.4 Whether repeats were issued on antibiotic prescriptions

D.4.1 Repeats issued on all antibiotic prescriptions for urinary tract infection

Table D-9: Frequency table for ordinal choice of antibiotic prescribed by patient's primary
 health network, for patients with initial presentations of urinary tract infection

	Repeat Negative	Repeat Rositive	Total		
Patient	ivegative	rusitive	TULAT		
Age Group				Кеу	
45+ yrs	5,701	2,446	8,147	freque	ncy
	69.98	30.02	100	row pe	rcentage
				·	
16-44 yrs	5,947	2,006	7,953		
	74.78	25.22	100		
6-15 yrs	630	364	994		
	63.38	36.62	100		
0-5 yrs	597	282	879		
	67.92	32.08	100		
Total	12,875	5,098	17,973		
	71.64	28.36	100		

 Table D-10:
 Frequency table of whether repeats were issued on antibiotic prescriptions for initial presentations of urinary tract infection, by Anatomical Therapeutic Chemical class

· · · · · · · · · · · · · · · · · · ·	Repeat	Repeat	-
ATC class	Negative	Positive	Total
Beta-lactamase inhibitors	732	729	1,461
First-generation cephalosporins	4,934	2,083	7,017
Fluoroquinolones	280	235	515
Glycopeptides	1	0	1
Lincosamides	4	1	5
Macrolides	70	18	88
Nitrofurantoin	565	129	694
Penicillins	331	110	441
Second-generation cephalosporins	82	15	97
Tetracyclines	15	14	29
Third-generation cephalosporins	1	0	1
Trimethoprim	5,145	1,395	6,540
Trimethoprim – sulphonamide combinations	181	132	313
Total	12,341	4,861	17,202

Table D-11:Frequency table of whether repeats were issued on prescriptions, by ordinal choice
of antibiotic prescribed, for initial presentations of urinary tract infection

		•	Not	3	-
Repeat	First-line	Second-line	Recommended	Total	
Prescription Status					
Negative	17,701	4,171	13,138	35,010	Key frequency
	78.65	47.42	64.47	67.75	row percentage
Positive	4,804	4,625	7,239	16,668	
	21.35	52.58	35.53	32.25	
Total	22,505	8,796	20,377	51,678	
	100	100	100	100	





Figure D-4: Bar graph of antibiotic prescriptions with repeats issued on them, for initial presentations of urinary tract infection, by ordinal choice of antibiotic prescribed

Table D-12: Frequency table of patient age group by whether repeats were issued on the prescription, for initial presentations of urinary tract infection

	Repeat Presc	ription Status		
	Negative	Positive	Total	Key
Patient Age Group				frequency
45+ yrs, ref	5,701	2,44	6 8,147	row percentage
	69.98	30.0	2 100	
16-44 yrs	5,947	2,00	6 7,953	
	74.78	25.2	2 100	
6-15 yrs	630	36	4 994	
	63.38	36.6	2 100	
0-5 yrs	597	28	2 879	
	67.92	32.0	8 100	
Total	12,875	5,09	8 17,973	
	71.64	28.3	6 100	

1 **D.4.2** Analysis of cefalexin prescriptions issued with repeats, for initial presentations 2 of urinary tract infection

3

The recommendation for penicillin-sensitive patients with UTI is cephalexin. For adult females, the course recommended is five days' duration and 500mg strength, twelve-hourly, which amounts to a total of 5,000mg. Whereas adult males are recommended seven days' worth, i.e. 7,000mg. Children over one month in age required 12.5mg/kg up to 500mg orally, six-hourly (as opposed to twelve-hourly in adults), for five days, totalling a maximum of 10,000mg.

10

11 The liquid formulation five men received, would have required a repeat to take them from

12 5g course to 7g course. All five men receiving liquid formulation needed repeats but they

- 13 were not needed for capsules (Table D-13).
- 14

Table D-13: Frequency table of medicine quantity and medicine strength, for male adults
 prescribed repeats for cephalexin with urinary tract infection

-					
			Medicine o	quantity	
Medicine strength	1	100mL	20	40	Total
250mg	0	0	11	1	12
250mg/5mL	4	1	0	0	5
500mg	0	0	219	0	219
Total	4	1	230	1	236

17

18 The cefalexin prescriptions with repeats present provided to women at least sixteen years

are listed in Table D-14. None of these women required repeats for cephalexin.

20

Table D-14: Frequency table of medicine quantity and medicine strength, for female adults prescribed repeats for cephalexin with urinary tract infection

modicing quantity								
			meuic	ine quant	ity			
Medicine strength	1	10	100mL	15	2*20	20	40	Total
250mg	0	0	0	0	1	152	4	157
250mg/5mL	12	0	6	0	0	0	0	18
500mg	0	4	0	1	0	1,291	4	1,300
Total	12	4	6	1	1	1,443	8	1,475

23

Table D-15 provides a list of cephalexin prescriptions with repeats provided to children under

eighteen years of age, for good measure, as GPs might work on this cutoff for children,

26 unlike sixteen years used for the modelling in this thesis.

- 1
- 2 3 4
- Table D-15: Frequency table of medicine quantity and medicine strength, for children under eighteen years issued repeats on cephalexin prescriptions for initial presentations of urinary tract infection

	Medicine quantity						
Medicine strength	1	100mL	20	Total			
125mg/5ml	42	14	0	56			
250mg	0	0	13	13			
250mg/5mL	168	65	0	233			
250mg/5ml	1	0	0	1			
500mg	0	0	13	13			
Total	211	79	26	316			

- 6 For 16-17 year olds, both the 250mg and 250mg/5mL could have required repeats if treated
- 7 as a child, or 500mg if treated as adult (Table D-16). This is the reason for examination of
- 8 16-17 year olds separately. However, the 500mg (n=10) did not require repeats as already
- 9 a 10,000mg course.
- 10
- 11 Table D-16: Frequency table of medicine quantity and medicine strength, for 16-17 year olds 12 issued repeats on cefalexin prescriptions for initial presentations of urinary tract 13 infection

	Medicine quantity					
Medicine strength	1	100mL	20	Total		
250mg	0	0	1		1	
250mg/5mL	6	2	0		8	
500mg	0	0	10		10	
Total	6	2	11		19	

- 14
- 15 For children under 16, those receiving 500mg x20 received a full course so did not require
- a repeat, as seen in Table D-17. Those receiving liquid formulations likely required repeats,
- as did the 250mg x20 course.
- 18
- 19Table D-17:Frequency table of medicine quantity and medicine strength, for children under20sixteen years issued repeats for cefalexin for initial presentations of urinary tract21infection

	Medicine quantity					
Medicine strength	1	100mL	20	Total		
125mg/5mL	42	14	0	56		
250mg	0	0	12	12		
250mg/5mL	162	63	0	225		
250mg/5ml	1	0	0	1		
500mg	0	0	3	3		
Total	205	77	15	297		

Therefore for UTI, of cephalexin scripts with repeats, 5/236 men, 0/2475 women, 10/19 16-17yrs, 294/297 children may have required repeats. Totalling 0.2% adults receiving repeats for cephalexin required them. 53% children 16-17 potentially required repeats, and 99% children under sixteen potentially required repeats. The strong majority of repeats were unnecessary for adults and the strong majority were potentially necessary for children.

6

For women 16 or over, the duration was variable, between 5 and 14 days as seen in Table
D-18, and even "after sexual intercourse". Overtreatment was common.

9

10 11

12

Table D-18: Frequency table of medicine instructions for adult females receiving cefalexin prescriptions with repeats issued on them for initial presentations of urinary tract infection

Medicine instructions	Frequency	Percent	Cumulative Percent
1 TDS	1	0.21	0.21
1 bd for 5 days	1	0.21	0.42
1 cap tds	11	2.33	2.75
1 capsule 6 hrly p/o	1	0.21	2.96
1 four times a day for 7 days	1	0.21	3.17
1 po BD for 5 days	1	0.21	3.38
1 qid for 10 days	1	0.21	3.59
1 tab bd	6	1.27	4.86
1 tab tds	37	7.82	12.68
1 tab tds for 5 days	2	0.42	13.11
1 tablet 6 hourly for 5 days	1	0.21	13.32
1 tds for 5 days	1	0.21	13.53
1/2 hr before food	1	0.21	13.74
10 days	2	0.42	14.16
10 mL three times daily for 7 days	1	0.21	14.38
10 ml four times a day	1	0.21	14.59
10-	Jul 2	0.42	15.01
10mls 6hrly, for 5 days	1	0.21	15.22
10mls Twice a day	1	0.21	15.43
14 days	2	0.42	15.86
2 capsule stat then 1 cap twice daily	1	0.21	16.07
2 stat	30	6.34	22.41
2 stat then 1 tds	1	0.21	22.62
2 stat then 2 tonight then one three	1	0.21	22.83
2 stat, then 1 qid for 5 days	1	0.21	23.04
2 to start	2	0.42	23.47
2 weeks	1	0.21	23.68
250mg 6 hrly p/o	1	0.21	23.89
250mg six hourly for five days	1	0.21	24.1
5 days	6	1.27	25.37
5 days but can extend to 10 days	1	0.21	25.58
5-10 days	1	0.21	25.79
5-7 days	1	0.21	26
5-	Jul 1	0.21	26.22
500mg BD for 5 days	2	0.42	26.64
500mg six hourly for five days	2	0.42	27.06
500mg six hourly for seven days	1	0.21	27.27
500mg*2 (1gm) 12 hrly p/o for 5 days	1	0.21	27.48
7 days	2	0.42	27.91
7 to 10 days	1	0.21	28.12

	7-Jul	1	0.21	28.33
7mls tds		1	0.21	28.54
After sexual intercourse as directed		1	0.21	28.75
Double dose first 4 doses		1	0.21	28.96
FOR 5 DAYS		9	1.9	30.87
FOR 6 DAYS		1	0.21	31.08
For 10 days		1	0.21	31.29
For 5 days		8	1.69	32.98
For 5 days after food		1	0.21	33 19
For 7 days		2	0.42	33.62
For five days		1	0.21	33.83
For five days		2	0.42	34 25
May be taken with or without food Ad		1	0.42	34.25
One cansule two times daily for 5 days		1	0.21	34.67
One capsule two times daily for 7 days		1	0.21	34.88
One tablet twice a day for E days		1	0.21	2E 1
One tablet twice a day for sover days		1	0.21	55.1 25.21
One tablet twice doily for E down and		1	0.21	35.31
Disease take 2 take for the 1 st 1 2, th		1	0.21	35.52
Please take 2 tos for the 1st 1-2, th		1	0.21	35.73
THEN 1 Q6 HRS		1	0.21	35.94
Take 2 initially and then 1 tds		1	0.21	36.15
Three times daily		1	0.21	36.36
To start if needed		1	0.21	36.58
a.c.		19	4.02	40.59
and 1 6 hourly		1	0.21	40.8
as directed		1	0.21	41.01
as required if gets a UTI on trip - 3		1	0.21	41.23
complete course		1	0.21	41.44
delayed script		1	0.21	41.65
finish all		1	0.21	41.86
fo 10 days		1	0.21	42.07
for 3 to 5 days		1	0.21	42.28
for 1 week		1	0.21	42.49
for 10 days		26	5.5	47.99
for 10/7		1	0.21	48.2
for 14 days		1	0.21	48.41
for 1st script then one tab at night		1	0.21	48.63
for 2 weeks		1	0.21	48.84
for 24 hours then one bd for 4 days		1	0.21	49.05
for 3-5 days		1	0.21	49.26
for 5 days		76	16.07	65.33
for 5 days ONLY		1	0.21	65.54
for 5 days and review		1	0.21	65.75
for 5 days for acute infections		-	0.85	66.6
for 5 days for bladder infection		1	0.03	66.81
for 5 days for bladder infection		1	0.21	67.02
for 5 days for urine infection		1	0.21	67.02
for 5 days for unite intection		1	0.21	67.44
for 5 days with of without rood.		1	0.21	67.65
for 5 days.		1	0.21	67.05
for 5 10 days		1	0.21	07.00
for 5-7		T	0.21	68.08
for 5-7		1	0.21	68.29
for 5-7 days		2	0.42	68.71
for 5/7		2	0.42	69.13
for 5d		1	0.21	69.34
for 6 days		1	0.21	69.56
for 7 days		21	4.44	74
for 7 days then clearance testing 7		1	0.21	74.21
for 7 days.		1	0.21	74.42
for 7- 10 days		1	0.21	74.63
for 7-10 days		2	0.42	75.05
for 7days		1	0.21	75.26
for UTI		1	0.21	75.48
for about six days for UTI		1	0.21	75.69

for five days	6	1.27	76.96
for five days.	2	0.42	77.38
for one day then one bd total 5 days	1	0.21	77.59
for one week	1	0.21	77.8
for ten days	1	0.21	78.01
i tds	4	0.85	78.86
ii bd 7 days	1	0.21	79.07
ii stat then i tds	1	0.21	79.28
increase to 4 times daily if needed	1	0.21	79.49
m.d.u.	65	13.74	93.23
one bd for 7 days	1	0.21	93.45
one qid	5	1.06	94.5
one tab three times per day for one w	1	0.21	94.71
one tds	1	0.21	94.93
one tds for 5 days	1	0.21	95.14
p.c.	2	0.42	95.56
p.r.n.	4	0.85	96.41
take one tablet every 8 hrs	2	0.42	96.83
take two to start and then one capsul	1	0.21	97.04
two twice a day until finished	1	0.21	97.25
until all taken	1	0.21	97.46
until all taken.	2	0.42	97.89
until course complete	1	0.21	98.1
until finished	1	0.21	98.31
with lots of water	1	0.21	98.52
x 5 days	5	1.06	99.58
x 7 days	2	0.42	100
Total	473	100	

For men at least sixteen years of age, notable variation in duration also, there was under

and over-treatment, ranging from between 5 to 14 days (Table D-19).

Table D-19:Frequency table of medicine instructions for adult males receiving cefalexin
prescriptions with repeats issued on them, for initial presentations of urinary tract
infection

Medicine instructions	Frequency	Porcont	Cumulative Percent
	Frequency	Fercent	culturative Percent
1 BD	1	1	1
1 cap tds	- 3	- 3	- 4
1 tab bd	1	1	5
1 tab tds	17	17	22
10mls tds	1	1	23
for 14/7	1	1	24
2 stat	8	8	32
2 stat the 1 6hrly	1	1	33
2 stat then 1 tds until finished.	1	1	34
250mg six hourly for ten days	1	1	35
for 5/7	1	1	36
7mls tds	1	1	37
FOR 14 DAYS	1	1	38
Ffirst two doses 6 hours apart take t	1	1	39
For 14 days	2	2	41
For 2 weeks	1	1	42
For fourteen days	1	1	43
UTI management	1	1	44
a.c.	2	2	46
for 10 days	8	8	54
for 10 days. Save repeat to use if ha	1	1	55
for 14 days	7	7	62

for 2 weeks	1	1	63
for 2/7 ,1 td	1	1	64
for 5 days	7	7	71
for 7 days	3	3	74
for seven days	1	1	75
ii qid for infection for 5/7	1	1	76
m.d.u.	16	16	92
one table four times daily for three	1	1	93
one tds	1	1	94
take two capsule four times daily 24	1	1	95
take two now and two tonight then one	1	1	96
then 1 tds for 1 week for bladder inf	1	1	97
two tabs stat then two tonight then o	1	1	98
until finished.	1	1	99
x 5 days	1	1	100
Total	100	100	

2 For children under 16, the duration was also varied, ranging between 3 and 14 days, and

3 most commonly 5 days, as per Table D-20 below:

Table D-20:Frequency table of medicine instructions for children under sixteen years of age
receiving cefalexin prescriptions with repeats issued on them, for initial presentations
of urinary tract infection

Medicine instructions	Frequency	Percent	Cumulative Percent
1 cap tds	1	0.62	0.62
1.2 mls	1	0.62	1.23
1/2 hr before food for 10 days	1	0.62	1.85
10 mls twice a day	1	0.62	2.47
10 days	1	0.62	3.09
10mL four times a day for 5 days	1	0.62	3.7
10ml stat	1	0.62	4.32
10mls, 6hrly for 5 days	1	0.62	4.94
12.5mg/kg 6hrly for 5 days orally	1	0.62	5.56
15kg	1	0.62	6.17
15ml (500mg) bd	1	0.62	6.79
250mg six hourly for five days	1	0.62	7.41
250mg/5mL QID for 5 days	1	0.62	8.02
3 mls 6 hourly for 1 week	1	0.62	8.64
3.75mL qid	1	0.62	9.26
300mg 6hourly 5 days	1	0.62	9.88
4 ml x 6 hrly	1	0.62	10.49
4.5ml six hourly for 5 days	1	0.62	11.11
4ml four times daily	1	0.62	11.73
5 days	5	3.09	14.81
5 ml every 6 hours for 5 days	1	0.62	15.43
5 ml four times a day, for 5-7 days	1	0.62	16.05
5 ml orally three times a day for 7 d	1	0.62	16.67
5-Ju	ıl 1	0.62	17.28
500 mg every 6 hours for 7 days.	1	0.62	17.9
5mls four times daily for five days	1	0.62	18.52
5mls tds	4	2.47	20.99
5mls, 6hrly for 5 days	1	0.62	21.6
6 mL two times daily for 7 days	1	0.62	22.22
6 ml (300mg) po qid	1	0.62	22.84
6.5 mls qid for 7 days	1	0.62	23.46
6mL qid	1	0.62	24.07
6ml 4 times a day for 5 days	1	0.62	24.69
7.5 ml tid for 5 days	1	0.62	25.31

	4	0.62	25.02
/ml (1/Smg) PO TDS for 5d	1	0.62	25.93
9 days	1	0.62	26.54
9.5ml four times a day, 5 days	1	0.62	27.16
9ml twice a day for 5-7 days	1	0.62	27.78
FOR 5 DAYS	2	1.23	29.01
FOR 6 DAYS	2	1.23	30.25
For 1 week	1	0.62	30.86
For 3days	1	0.62	31.48
For 5 days	3	1.85	33.33
Give 2.7mls every 6 hours for 5 days	1	0.62	33.95
May be taken with or without food	1	0.62	34.57
Please dissolve one capsule in 5 mls	1	0.62	35.19
a.c.	2	1.23	36.42
by metric measure for 7 days	1	0.62	37.04
c.c.	2	1.23	38.27
for 10 days	6	3.7	41.98
for 10 days.	1	0.62	42.59
for 14 days	2	1.23	43.83
for 3 days	1	0.62	44.44
for 5 days	30	18.52	62.96
for 5 days only	1	0.62	63.58
for 5 days.	1	0.62	64.2
for 5- 7 days	1	0.62	64.81
for 5-7days	1	0.62	65.43
for 6 days	1	0.62	66.05
for 7 days	33	20.37	86.42
for 7 days.	1	0.62	87.04
for 7-10 day	1	0.62	87.65
fro 10 days	1	0.62	88.27
m.d.u.	18	11.11	99.38
x 5 days	1	0.62	100
,			
Total	162	100	

D.5 Marginal effects for the ordinal choice of antibiotic prescribing model for urinary tract infection

D.5.1 Average marginal effects

Table D-21:Average marginal effects for ordinal choice of antibiotic prescribed for initial
presentations of urinary tract infection (Model 1)

	Average N	Aarginal Effec	ts (Mode	1)		8
Ordinal Choic	e of Antibiotic Pr	escribed for li	nitial Pres	sentation	s of UTI Model	6
Variable	dy/dx	Std. Err.	Z	P>z	[95% Conf.	Interval]
Patient Age Group						
45 years and over	(base outcome)				
16-44 years (predict Outcome:)						
First-line	-0.0299301	0.006937	-4.31	0.000	-0.0435263	-0.0163338
Second-line	0.0113018	0.0027619	4.09	0.000	0.0058885	0.0167151
Third-line	0.0089341	0.0021224	4.21	0.000	0.0047743	0.0130939
Not Recommended	0.0096942	0.0024878	3.9	0.000	0.0048183	0.0145702
6-15 years (predict Outcome:)						
First-line	-0.14676	0.0120251	-12.2	0.000	-0.1703287	-0.1231912
Second-line	0.0359399	0.0051579	6.97	0.000	0.0258306	0.0460492
Third-line	0.0488681	0.0045421	10.76	0.000	0.0399658	0.0577704
Not Recommended	0.0619519	0.0071379	8.68	0.000	0.047962	0.0759419
0-5 years (predict Outcome:)						
First-line	-0.1849305	0.0118474	-	0.000	-0.208151	-0.1617101
Second-line	0.0342046	0.0066219	5.17	0.000	0.0212259	0.0471833
Third-line	0.0637304	0.0047629	13.38	0.000	0.0543953	0.0730654
Not Recommended	0.0869956	0.0086858	10.02	0.000	0.0699717	0.1040194
Patient Gender						
Female	(base outcome)				
Male (predict Outcome:)						
First-line	-0.0892231	0.0099851	-8.94	0.000	-0.1087935	-0.0696527
Second-line	0.0234147	0.0033992	6.89	0.000	0.0167523	0.0300771
Third-line	0.0293018	0.0035705	8.21	0.000	0.0223037	0.0363
Not Recommended	0.0365066	0.0050462	7.23	0.000	0.0266162	0.0463969
Repeat on Prescription Status						
Negative	(base outcome)				
Positive (predict Outcome:)						
First-line	-0.0962027	0.0077177	-	0.000	-0.111329	-0.0810763
Second-line	0.0296811	0.0035305	8.41	0.000	0.0227615	0.0366006
Third-line	0.0306878	0.0026988	11.37	0.000	0.0253982	0.0359774
Not Recommended	0.0358338	0.0037132	9.65	0.000	0.0285561	0.0431115
Comorbid Condition Status						
Negative	(base outcome)				
Positive (predict Outcome:)						
First-line	-0.0369297	0.0080796	-4.57	0.000	-0.0527654	-0.0210939
Second-line	0.011216	0.0025104	4.47	0.000	0.0062957	0.0161364
Third-line	0.0116508	0.0026136	4.46	0.000	0.0065282	0.0167735
Not Recommended	0.0140628	0.0033004	4.26	0.000	0.0075942	0.0205314
Missing (predict Outcome:)						
First-line	-0.0928739	0.0543549	-1.71	0.088	-0.1994077	0.0136598
Second-line	0.0217922	0.0072047	3.02	0.002	0.0076713	0.0359131
Third-line	0.0307306	0.0192275	1.6	0.110	-0.0069545	0.0684158
Not Recommended	0.0403511	0.028641	1.41	0.159	-0.0157842	0.0964864
Culture Testing Status						
Negative	(base outcome)				
Positive (predict Outcome:)						
First-line	0.0598695	0.0195999	3.05	0.002	0.0214544	0.0982845
Second-line	-0.0225814	0.0085514	-2.64	0.008	-0.0393418	-0.0058209
Third-line	-0.0176727	0.0055061	-3.21	0.001	-0.0284645	-0.0068809
Not Recommended	-0.0196154	0.0057917	-3.39	0.001	-0.0309669	-0.008264

							1	
Average Marginal Effects (Model 1) continued								
Ordinal Choic	e of Antibiotic Pro	escribed for Ir	nitial Pres	entatior	s of UTI Model		2	
Variable	dy/dx	Std. Err.	z	P>z	[95% Conf.	Interval]	•	
Urine Dipstick Testing Status							3	
Negative	(base outcome)					л	
Positive (predict Outcome:)							4	
First-line	0.0653249	0.0210125	3.11	0.002	0.0241412	0.10650	0 8 6	
Second-line	-0.0248589	0.0093127	-2.67	0.008	-0.0431115	-0.00660	062	
Third-line	-0.0192277	0.0058638	-3.28	0.001	-0.0307205	-0.00773	3 6 9	
Not Recommended	-0.0212383	0.0061098	-3.48	0.001	-0.0332133	-0.00926	633	
							7	

9 Average Marginal Effects

The risk of higher-line prescribing decreased with increasing age. Children under six years were the least likely to receive first-line antibiotics (-0.15, p<0.001, 95%CI: -0.208, -0.162) and more likely to receive higher-line antibiotics than patients of the reference age, 45 years and over. Young children were six and nine percentage points respectively more likely to receive third-line (0.064, p<0.001, 0.054, 0.073) and not recommended (0.087, p<0.001, 95%CI: 0.070, 0.104) antibiotics than patients of reference age.

16

Children aged 6-15 years were less likely to receive first-line antibiotics than reference-age 17 patients by fifteen percentage points (-0.147, p<0.001, 95%CI: -0.170, -0.123). They were 18 also five percentage points more likely to receive third line (0.049, p<0.001, 95%Cl: 0.040, 19 0.058) and six percentage points more likely to receive not recommended antibiotics (0.062, 20 21 p<0.001, 95%CI: 0.048, 0.076). Young adults sixteen to 44 years were three percentage 22 points less likely to receive first-line antibiotics (-0.029, p<0.001, 95%Cl: -0.044, -0.016). They were also one percentage point more likely to receive second-line (0.011, p<0.001, 23 24 95%CI: 0.006, 0.017), third-line (0.009, p<0.001, 95%CI: 0.0048, 0.013), or not recommended (0.010, p<0.001, 95%CI: 0.0048, 0.015) antibiotics than patients 45 years 25 26 and over.

- 27
- 28

D.5.2 Margins at representative values for the effect on ordinal choice of antibiotic, with change in patient age group, across different values of whether repeats were issued on the prescription

margins, d	ydx(a	gegrp_uti_ne	w) at (repe	at_script=	(1 0))		
Average margin Model VCE :	al ef OIM	fects		Numbe	er of ob	s = 1	.7,973
dy/dx w.r.t. : 1predict : 2predict : 3predict : 4predict :	2.ag Marg Marg Marg Marg	egrp_uti_new inal predict inal predict inal predict inal predict	3.agegrp_u ed mean (1. ed mean (2. ed mean (3. ed mean (4.	ti_new 4.ac choice_uti_ choice_uti_ choice_uti_ choice_uti_	gegrp_ut _new), p _new), p _new), p _new), p	i_new redict(pr out redict(pr out redict(pr out redict(pr out	come(1)) come(2)) come(3)) come(4))
1at :	repe	at_script	=	1	(Repeat	Positive)	
2at :	repe	at_script	=	0	(Repeat	Negative)	
		dy/dx	Delta-metho Std. Err.	d z	P> z	[95% Conf.	Interval]
45+ years	ļ	(base outc	ome)				
16-44 years _predict# 1 2 2 3 3 3 4 4	_at 1 2 1 2 1 2 1 2 2	0289152 0303206 .0059848 .0133158 .0100159 .0085362 .0129144 .0084686	.0112113 .0082224 .0024227 .0035409 .0039311 .0023855 .0052841 .0025025	-2.58 -3.69 2.47 3.76 2.55 3.58 2.44 3.38	0.010 0.000 0.013 0.000 0.011 0.000 0.015 0.001	0508889 0464362 .0012365 .0063758 .0023112 .0038607 .0025577 .0035637	0069415 0142049 .0107332 .0202558 .0177207 .0132118 .0232711 .0133735
6-15 years _predict# 1 2 2 3 3 4 4	_at 1 2 1 2 1 2 1 2 1 2	0932033 1677995 .0086763 .0461854 .0342076 .0547683 .0503194 .0668458	.0176908 .0148589 .0042216 .0059019 .0068273 .005543 .0115896 .0084949	-5.27 -11.29 2.06 7.83 5.01 9.88 4.34 7.87	$\begin{array}{c} 0.000\\ 0.000\\ 0.040\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ 0.000\\ \end{array}$	1278766 1969225 .0004021 .0346179 .0208263 .0439043 .0276041 .0501961	05853 1386765 .0169505 .0577529 .047589 .0656324 .0730346 .0834955
0-5 years _predict# 1 2 2 3 3 4 4 	_at 1 2 2 2 2 2 2 2 2	0978473 2191124 .0087943 .04347 .0360062 .0747738 .0530467 .1008686	.0195587 .0140971 .0045555 .0078991 .0076336 .0056461 .0132888 .0104645	-5.00 -15.54 1.93 5.50 4.72 13.24 3.99 9.64	0.000 0.000 0.054 0.000 0.000 0.000 0.000 0.000	1361816 2467422 0001343 .027988 .0210447 .0637076 .0270011 .0803586	0595129 1914827 .0177229 .058952 .0509678 .08584 .0790924 .1213786
,,							

where 1=First=-line, 2=Second-line, 3=Third-line/Last Resort and 4=Not Recommended.

D.5.3 Margins at representative values for the effect on ordinal choice of antibiotic, with change in patient age group, across different values of patient gender

margins, dydx(agegrp_uti_new) at ((pat_sex)= (1 0)) Average marginal effects Number of obs = 17,973 Model VCE : OIM dy/dx w.r.t. : 2.agegrp_uti_new 3.agegrp_uti_new 4.agegrp_uti_new : Marginal predicted mean (1.choice_uti_new), predict(pr outcome(1)) : Marginal predicted mean (2.choice_uti_new), predict(pr outcome(2)) : Marginal predicted mean (3.choice_uti_new), predict(pr outcome(3)) : Marginal predicted mean (4.choice_uti_new), predict(pr outcome(4)) 1._predict 2._predict 3._predict 4._predict (Male) 1._at : pat_sex 1 = 2._at : pat_sex = 0 (Female) _____ _____ Delta-method dy/dx Std. Err. z P>|z| [95% Conf. Interval] | (base outcome) 45+ years _____ 16-44 years _predict#_at -.0059583 -.0326776 .0207157 .0071719 -0.29 -4.56 0.26 -.0465604 -.0467343 .0346437 0.774 1 1 1 2 0.000 -.018621 2 1 2 2 3 1 .0008494 .003271 0.795 -.0055616 .0072604 4.17 0.29 4.53 0.29 .0029695 .0123805 0.000 .0065604 .0182006 .0021195 .0097537 .0029895 .0105435 0.773 -.0122618 .0165008 32 .0021539 .0055321 .0139752 4 1 4 2 .0102435 -.0170874.0230663 0.770 4.40 0.000 .0058444 .0152425 6-15 years _predict#_at 1 1 1 2 2 1 0.001 -.0933382 .0272528 -3.42 -.1467528 -.0399237 -11.92 0.50 7.37 3.26 .0128196 -.1779585 -.1528324 .0031006 -.1277064 .0151377 0.000 $0.614 \\ 0.000$ 2 1 2 2 .0393095 .0053352 .0288527 .0497662 3 .0350447 .0107605 0.001 .0139545 .0561349 ŝ .0505613 .0048239 10.48 0.000 .0411067 .060016 4 1 2.71 .055193 .0203603 0.007 .0152875 .0950984 4 2 .0629617 .007448 8.45 0.000 .0483638 .0775595 _____ 0-5 years /ears _predict#_at 1 1 1 2 .0253827 .0127023 -.0951786 0.000 -.1449278 -.0454294 -3.75 -.1951013 -15.36 0.000 -.2199973 -.1702052 0.53 5.32 3.57 13.15 1 2 2 2 .0031816 .0060462 0.599 -.0086688 .015032 .0371488 .0357668 .0670387 .0508396 .0554239 .0770327 0.000 .0069853 .0234579 3 1 3 2 0.000 .0100293 .0161096 .0050991 .0570447 4 1 .0562302 .0189876 2.96 0.003 .0190153 .0934452 42 .0909138 9.78 .0092933 0.000 .0726993 .1091283 Note: dy/dx for factor levels is the discrete change from the base level.

where 1=First=-line, 2=Second-line, 3=Third-line/Last Resort and 4=Not Recommended.



Figure D-5: Margins at representative values for the effect on the probability of each of the four ordinal choice of antibiotic outcomes occurring, with change in patient age group, across different values of patient gender

D.5.4 Margins at representative values for then effect on ordinal choice of antibiotic, with change in whether repeats were issued on the prescriptions, across different values of patient age group

Margins, dydx(repeat_script) at (agegrp_uti_new)=(1 2 3 4))

Average margi Model VCE	ina :	l effects DIM			Numb	per of ob	os =	17,973
dy/dx w.r.t. 1predict 2predict 3predict 4predict		1.repeat_script Marginal predic Marginal predic Marginal predic Marginal predic	ted mea ted mea ted mea ted mea	an (1.c an (2.c an (3.c an (4.c	choice_uti choice_uti choice_uti choice_uti	i_new), p i_new), p i_new), p i_new), p	predict(pr ou predict(pr ou predict(pr ou predict(pr ou	<pre>itcome(1)) itcome(2)) itcome(3)) itcome(4))</pre>
1at	: :	agegrp_uti_new	=		1	(45+ ye	ars)	
2at	: :	agegrp_uti_new	=		2	(16-44	years)	
3at	: :	agegrp_uti_new	=		3	(6-15 y	ears)	
4at	: :	agegrp_uti_new	=		4	(0-5 ye	ars)	
	 	dy/dx	Delta-r Std.	nethod Err.	Z	P> z	[95% Conf.	Interval]
Repeat Negat	ive	(base outo	come)					
Repeat Posit _predict#	ive all 1222341222341222341423334412444	t 1068389 054335 0322426 .0144263 .0367358 .0294049 0007733 .0020601 .0330083 .034488 .0124476 0057593 .0370947 .0415405 .0205683 0107271	.0104 .0106 .0209 .0042 .0042 .0042 .0032 .0032 .0034 .0033 .0034 .0035 .0043 .0044 .0056 .0134 .0158	1231 5745 5348 5547 5039 2412 9606 1517 7255 7996 1585 3699 9586 1679 3046	-10.25 -9.88 -1.56 0.67 8.16 6.93 -0.39 0.70 9.56 9.26 1.56 -0.67 8.49 8.21 1.53 -0.68	$\begin{array}{c} 0.000\\ 0.000\\ 0.118\\ 0.503\\ 0.000\\ 0.694\\ 0.487\\ 0.000\\ 0.000\\ 0.120\\ 0.502\\ 0.000\\ 0.120\\ 0.000\\ 0.127\\ 0.497 \end{array}$	$\begin{array}{c}1272678\\1263552\\072686\\0278202\\ .0279083\\ .0210922\\0046235\\0037426\\ .026243\\ .0271861\\0032243\\0225855\\ .0285298\\ .031626\\0058282\\00417036\end{array}$	08641 0845118 .0082007 .0566728 .0455634 .0377175 .0030769 .0078628 .0397736 .04179 .0281195 .011067 .0456596 .0514551 .0469648 .0202494
Note: dy/dx f	For	factor levels	is the	discre	ete change	e from th	ne base level	

where 1=First=-line, 2=Second-line, 3=Third-line/Last Resort and 4=Not Recommended.

With a repeat, adults 11% less likely to receive first-line, and 3-4% more likely to receive each of second-line, third-line, or not recommended agents. There was no significant difference in line received for child age groups.



3 4 5 6 7 8 9012345678901234567890123345678901234456

1

2

Figure D-6: Margins at representative values for the effect on the probability of the four ordinal choice of antibiotic outcomes occurring, with change in whether repeat were issued on the prescription, across different values of patient age group

D.5.5 Adjusted predictions for the effect on ordinal choice of antibiotic, at specific values of patient gender, patient age group and whether repeats were issued on the prescription

. margins, at(pat_sex=(0 1) ag	egrp_uti_new=(1 2 3 4)	<pre>repeat_script=(1</pre>	0)) atmeans	vsquish post
Adjusted predictions Model VCE : OIM	Numl	per of obs =	17,973	
<pre>1predict : Marginal predic 2predict : Marginal predic 3predict : Marginal predic 4predict : Marginal predic 1at : agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 0.dipstick~d 1.dipstick~d</pre>	ted mean (1.choice_ut ted mean (2.choice_ut ted mean (3.choice_ut = 1 = 0 = 1 = .7753297 (mean) = .1957381 (mean) = .0289323 (mean) = .0310466 (mean) = .9545429 (mean) = .041602 (mean)	_new), predict(pr _new), predict(pr _new), predict(pr _new), predict(pr (45+ years) (Female) (Repeat Positive)	<pre>outcome(1)) outcome(2)) outcome(3)) outcome(4))</pre>	
2at ; agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d	$\begin{array}{rcrr} = & 2014.002 \ (mean) \\ = & 1 \\ = & 0 \\ = & 0 \\ = & .7753297 \ (mean) \\ = & .0289323 \ (mean) \\ = & .0689534 \ (mean) \\ = & .0310466 \ (mean) \\ = & .9545429 \ (mean) \\ = & .0454571 \ (mean) \\ = & .041602 \ (mean) \\ \end{array}$	(45+ years) (Female) (Repeat Negative)		
3at : agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti	$\begin{array}{cccc} - & 2014.002 \text{ (mean)} \\ = & 1 \\ = & 1 \\ = & .7753297 \text{ (mean)} \\ = & .1957381 \text{ (mean)} \\ = & .0289323 \text{ (mean)} \end{array}$	(45+ years) (Male) (Repeat Positive)		

4. at	0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d year : agegrp uti new	= .9689534 (mean) = .0310466 (mean) = .9545429 (mean) = .0454571 (mean) = 2014.602 (mean) = 1	(45+ vears)
	pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d	= 1 = 0 = .7753297 (mean) = .1957381 (mean) = .0289323 (mean) = .0310466 (mean) = .9545429 (mean) = .0454571 (mean)	(Male) (Repeat Negative)
5at	year : agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0 cult tes~d	= 2014.602 (mean) = 2 = 0 = 1 = .7753297 (mean) = .0289323 (mean) = .9689534 (mean)	(16-44 years) (Female) (Repeat Positive)
6 at	1.cult_tes~d 0.dipstick~d 1.dipstick~d year	$\begin{array}{rcl} - & .9689134 \ (\text{ineal}) \\ = & .0310466 \ (\text{mean}) \\ = & .9545429 \ (\text{mean}) \\ = & .0454571 \ (\text{mean}) \\ = & 2014.602 \ (\text{mean}) \\ = & 2 \end{array}$	(16-44 years)
	pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d year	$\begin{array}{cccc} & & & & & & \\ = & & & & & \\ 0 & = & & & & \\ 0 & = & & & & \\ 1957381 & (mean) & \\ = & & & & & & \\ 0289323 & (mean) & \\ = & & & & & & \\ 0310466 & (mean) & \\ = & & & & & & \\ 0310466 & (mean) & \\ = & & & & & \\ 0454571 & (mean) & \\ = & & & & & \\ 2014.602 & (mean) \end{array}$	(Female) (Repeat Negative)
7at	: agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d vear	= 2 = 1 = .7753297 (mean) = .1957381 (mean) = .0289323 (mean) = .0310466 (mean) = .0454571 (mean) = .014.602 (mean)	(16-44 years) (Male) (Repeat Positive)
8at	: agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d year	= 2 = 1 = 0 = .7753297 (mean) = .1957381 (mean) = .0289323 (mean) = .9689534 (mean) = .0310466 (mean) = .9545429 (mean) = .0454571 (mean) = 2014.602 (mean)	(16-44 years) (Male) (Repeat Negative)
9at	: agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d vear	= 3 = 0 = 1 = .7753297 (mean) = .1957381 (mean) = .0289323 (mean) = .0310466 (mean) = .9545429 (mean) = .0454571 (mean) = 2014.602 (mean)	(6-15 years) (Female) (Repeat Positive)
10at	: agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d	= 3= 0= .7753297 (mean)= .1957381 (mean)= .0289323 (mean)= .0310466 (mean)= .0310466 (mean)= .0454571 (mean)= .0454571 (mean)	(6-15 years) (Female) (Repeat Negative)
11at	: agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d	$\begin{array}{cccc} & & & & & & \\ & & & & & & \\ & & & & & $	(6-15 years) (Male) (Repeat Positive)

14at	repeat_script 0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 1.dipstick~d 1.dipstick~d year agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 1.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~	$\begin{array}{c} = & .77 \\ = & .19 \\ = & .02 \\ = & .96 \\ = & .03 \\ = & .04 \\ = & .04 \\ = & .04 \\ = & .02 \\ = & .04 \\$	1 753297 753297 757381 89323 89534 10466 454529 154571 4.602 4.602 4.602 753297 757381 89534 89534 10466 45429 154571 4.602 4.602 4.602 4.602 10466 104529 10466 104529 10466 104529 1045529 1045529 1045529 10455529 10455529 10455	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)	(Repeat Positive) (0-5 years) (Female) (Repeat Negative) (0-5 years) (Male) (Repeat Positive)	
16at	0.pat_co~uti 1.pat_co~uti 2.pat_co~uti 0.cult_tes~d 1.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d year agegrp_uti_new pat_sex repeat_script 0.pat_co~uti 1.pat_co~uti 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d year	$\begin{array}{c} = & .77\\ = & .19\\ = & .02\\ = & .96\\ = & .03\\ = & .95\\ = & .04\\ = & .04\\ = & .04\\ = & .04\\ = & .04\\ = & .04\\ = & .04\\ = & .05\\ = & .04\\ = & .$	753297 757381 189323 189534 110466 145429 154571 14.602 154571 189323 189323 189534 110466 145429 154571 14.602	(mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean) (mean)	(O-5 years) (Male) (Repeat Negative)	
	Delt Margin St	a-method	 z	 P> z	 [95% Conf. I	
 _predict#_at 1 1 1 2 1 3 1 4 1 5 1 6 1 7 1 8 1 9 1 10 1 11 1 12 1 13 1 14 1 15 1 16 2 1 2 2 2 3 2 4 2 5 2 6 2 7 2 8 2 9 2 10 2 11	$\begin{array}{c} .3630471 & .0\\ .4714561 & .0\\ .2618627 & .0\\ .3585672 & .0\\ .3314096 & .0\\ .2559369 & .0\\ .2559369 & .0\\ .2559369 & .0\\ .264095 & .0\\ .264095 & .0\\ .2459498 & .0\\ .4266171 & .0\\ .4370537 & .0\\ .4277331 & .0\\ .4291452 & .0\\ .4372177 & .0\\ .4373354 & .0\\ .4286922 & .0\\ \end{array}$	0145766 0136237 0148832 0167187 0143658 0134411 0193328 0219831 0193907 0177547 0274272 0281826 0210617 016234 0292038 0252929 0064959 00758348 0066901 0071601 0060319 0074012 0058739 0058273 0098878	24.91 34.61 17.59 21.45 23.07 32.60 13.24 16.04 13.62 16.71 7.90 8.73 12.13 14.85 8.42 9.17 65.68 50.94 74.96 63.22 71.16 56.02 72.37 57.98 74.43 75.05 43.36		$\begin{array}{cccccccccccccccccccccccccccccccccccc$. 3916167 .498158 .2910333 .3595659 .46456 .2938285 .3956603 .3021001 .3315551 .2705236 .3011867 .2968571 .272916 .3032502 .2814396 .4393487 .4011884 .4493487 .4011884 .4484817 .440941 .4452899 .4151313 .4483354 .4487568 .4487568 .4487568

D.5.6 Predictive margins for the effect on ordinal choice of antibiotic outcomes, across different values of patient age group and patient gender

margins, at(agegrp_uti_new=(1 2 3 4) pat_sex=	(1 0))
Predictive m Model VCE	argins : OIM	Number of obs = 17,973
1predict 2predict 3predict 4predict	: Marginal predicted mean (1.cho : Marginal predicted mean (2.cho : Marginal predicted mean (3.cho : Marginal predicted mean (4.cho	<pre>ice_uti_new), predict(pr outcome(1)) ice_uti_new), predict(pr outcome(2)) ice_uti_new), predict(pr outcome(3)) ice_uti_new), predict(pr outcome(4))</pre>
1at	: agegrp_uti_new = 1 pat_sex = 1	(45+ years) (Male)
2at	: agegrp_uti_new = 1 pat_sex = 0	(45+ years) (Female)
3at	: agegrp_uti_new = 2 pat_sex = 1	(16-44 years) (Male)
4at	: agegrp_uti_new = 2 pat_sex = 0	(16-44 years) (Female)
5at	: agegrp_uti_new = 3 pat_sex = 1	(6-15 years) (Male)
6at	: agegrp_uti_new = 3 pat_sex = 0	(6-15 years) (Female)
7at	: agegrp_uti_new = 4 pat_sex = 1	(0-5 years) (Male)
8at	: agegrp_uti_new = 4 pat_sex = 0	(0-5 years) (Female)

	•	Delta-method				
	Margin	Std. Err.	Z	P> Z	95% Cont.	Interval_
predict#_at						
11	.3316927	.0156054	21.26	0.000	.3011068	.3622787
12	.4409312	.0130266	33.85	0.000	.4153997	.4664628
13	.3257344	.0207109	15.73	0.000	.2851418	.3663271
14	.4082536	.0127826	31.94	0.000	.3832002	.4333071
15	.2383545	.0266377	8.95	0.000	.1861456	.2905634
16	.2880988	.0155892	18.48	0.000	.2575445	.3186531
17	.2365142	.0246598	9.59	0.000	.1881818	.2848465
18	.24583	.0148047	16.60	0.000	.2168134	.2748466
21	.4291003	.0061144	70.18	0.000	.4171162	.4410844
22	.3967679	.0069648	56.97	0.000	.3831171	.4104186
23	.4299497	.0063217	68.01	0.000	.4175594	.4423401
24	.4091483	.0065648	62.32	0.000	.3962816	.4220151
25	.4322009	.0074873	57.72	0.000	.4175261	.4468756
26	.4360773	.0057658	75.63	0.000	.4247766	.447378
27	.4322819	.0073669	58.68	0.000	.4178431	.4467207
28	.4339166	.0060522	71.70	0.000	.4220545	.4457787
31	.1271292	.0059369	21.41	0.000	.1154932	.1387653
32	.0921171	.0042119	21.87	0.000	.0838619	.1003724
33	.1292487	.0077191	16.74	0.000	.1141194	.1443779
34	.1018708	.0044097	23.10	0.000	.093228	.1105136
35	.1621739	.0110977	14.61	0.000	.1404229	.183925
36	.1426785	.0064201	22.22	0.000	.1300953	.1552616
37	.162896	.0103635	15.72	0.000	.1425838	.1832082
38	.1591558	.0065775	24.20	0.000	.1462642	.1720474
4 1	.1120777	.0079551	14.09	0.000	.096486	.1276694
4 2	.0701838	.0044889	15.63	0.000	.0613857	.0789819
43	.1150672	.0106672	10.79	0.000	.0941599	.1359745
4 4	.0807272	.0050159	16.09	0.000	.0708962	.0905583
4 5	.1672707	.0210985	7.93	0.000	.1259183	.208623
4 6	.1331454	.0096206	13.84	0.000	.1142893	.1520016
47	.1683079	.0197431	8.52	0.000	.1296121	.2070038
48	.1610976	.0114358	14.09	0.000	.1386838	.1835114





Figure D-7: Predictive margins for the effect on the probability of each of the four ordinal choice
of antibiotic outcomes occurring, across different values of patient age group and patient gender,
graphed by patient gender

D.5.7 Predictive margins for effect on the ordinal choice of antibiotic, across different values of patient age group and whether repeats were issued on the prescription

17,973

=

margins, at(agegrp_uti_new=(1 2 3 4) repeat_script=(1 0))
Predictive margins
Number of obs
Model VCE : OIM

Model VCE	:	OIM							
1predict 2predict 3predict 4predict		Marginal Marginal Marginal Marginal	predic predic predic predic	ted mean ted mean ted mean ted mean	(1.choic (2.choic (3.choic (4.choic	e_uti e_uti e_uti e_uti	_new), _new), _new), _new),	predict(pr predict(pr predict(pr predict(pr	<pre>outcome(1)) outcome(2)) outcome(3)) outcome(4))</pre>
1at	:	agegrp_ut repeat_sc	i_new ript	= =	1 1		(45+ у (Male)	ears)	
2at	:	agegrp_ut repeat_sc	i_new ript	= =	1 0		(45+ y (Femal	ears) e)	
3at	:	agegrp_ut repeat_sc	i_new ript	= =	2 1		(16-44 (Male)	years)	
4at	:	agegrp_ut repeat_sc	i_new ript	= =	2 0		(16-44 (Femal	years) e)	
5at	:	agegrp_ut repeat_sc	i_new ript	= =	3 1		(6-15 (Male)	years)	
6at	:	agegrp_ut repeat_sc	i_new ript	= =	3 0		(6-15 (Femal	years) e)	
7at	:	agegrp_ut repeat_sc	i_new ript	= =	4 1		(0-5 y (Male)	ears)	
8at	:	agegrp_ut repeat_sc	i_new ript	= =	4 0		(0-5 y (Femal	ears) e)	
		Margi	Delt n St	a-method d. Err.	z	 P> z		[95% Conf. 1	Interval]
_predict#_at 1 1 1 2 1 3 1 4 1 5 1 6 1 7 1 8 2 1 2 2 2 3 2 4 2 5 2 6 2 7 2 8 3 1 3 2 3 3 3 4 3 5 3 6 3 7 3 8 4 1 4 2 4 3 4 4 4 5 4 6 4 7 4 8		.353073 .459912 .324158 .429591 .259870 .292112 .255226 .240799 .426375 .3896375 .435351 .435824 .435051 .435824 .435051 .435824 .435051 .435824 .435051 .435824 .03493 .153611 .141164 .1554100 .161169 .101147 .064052 .114061 .072521 .1514666 .130898 .154194 .164921	42 . C 42 . C 44 . C 44 . C 45 . C	142027 134321 .01408 133363 018797 171227 205799 156506 0062708 0059538 0059538 0059538 0059521 0059538 0069695 1059321 0057866 0060915 0062033 0041611 0055513 0043986 0078764 1069235 1086091 0068827 0066827 0066827 0066827 0066827 004233 007523 0047316 1133165 1103419 1148905 1123457	$\begin{array}{c} 24.86\\ 34.24\\ 23.02\\ 32.21\\ 13.83\\ 17.06\\ 12.40\\ 15.39\\ 67.99\\ 53.09\\ 72.62\\ 57.82\\ 73.34\\ 75.32\\ 71.44\\ 69.83\\ 20.76\\ 23.31\\ 21.58\\ 19.50\\ 20.39\\ 18.05\\ 23.26\\ 15.14\\ 15.13\\ 15.16\\ 15.33\\ 11.37\\ 12.66\\ 10.36\\ 13.36\end{array}$	0.00 0.00	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	.3252366 .4335858 .2965619 .4034529 .2230287 .2585529 .2148903 .2101252 .4140846 .3752557 .4206908 .3892952 .4234247 .4244844 .4232303 .4209512 .1090176 .0782402 .1185397 .0863109 .1381742 .1275942 .1385368 .1475867 .0880495 .0557561 .099317 .0632475 .125367 .1250092 .1407241	.3809101 .4862387 .3517545 .4557304 .2967115 .3256726 .2955619 .2714744 .4386657 .4040229 .4460291 .416615 .4466781 .4471086 .4452675 .1297906 .0945513 .1403003 .1035531 .1690492 .154734 .1722838 .1747525 .1142453 .0723493 .1288066 .081795 .1511682 .183379 .1891185
Where 1=First	t=	 -line. 2=S	econd-	line. 3=	 Third-lin	e/Las	t Reso	rt and 4=No	t Recommended.



Predictive margins for the effect on the probability of each of the four ordinal choice

of antibiotic outcomes occurring, across different values of patient age group and whether repeats

were issued on the prescription, graphed by repeat on prescription status

Figure D-8:

D.6 Marginal effects for the model for repeats being issued on antibiotic prescriptions for initial presentations of urinary tract infection

2 3 4 5

6 7

1

D.6.1 Average marginal effects

Table D-22:	Average	marginal	effects	for	repeat	positive	antibiotic	prescribing	for	initial
	presentat	tions of uri								

	Average Marg	ginal Effects (M	odel 3)								
Repeat Positive Ant	ibiotic Prescri	bing for Initial F	resentatio	ons of UTI N	/lodel						
Variable	dy/dx	Std. Err.	z	P>z	[95% Conf.	Interval]					
atient Age Group, ref. 45 years and over											
16-44 yrs	-0.0154515	0.0063311	-2.44	0.015	-0.0278602	-0.0030429					
6-15 yrs	0.0541286	0.0132175	4.1	0.000	0.0282227	0.0800346					
0-5 yrs	0.0035729	0.013635	0.26	0.793	-0.0231513	0.0302971					
Patient Gender, ref. Female											
Male	0.0827534	0.0109402	7.56	0.000	0.061311	0.1041959					
Ordinal Choice of Antibiotic Prescribed, ref. First-line											
Second-line	0.0795068	0.0076602	10.38	0.000	0.064493	0.0945206					
Third-line	0.187035	0.0121971	15.33	0.000	0.1631291	0.210941					
Not Recommended	0.0112022	0.0110261	1.02	0.310	-0.0104085	0.032813					
Culture Testing Status, ref. Negative											
Positive	0.0499152	0.0189255	2.64	0.008	0.012822	0.0870084					
Urine Dipstick Testing Status, ref. Negative	•										
Positive	-0.0493777	0.0193114	-2.56	0.011	-0.0872273	-0.011528					
Temperature Testing Status, ref. Negative											
Positive	0.0215037	0.009648	2.23	0.026	0.0025941	0.0404134					
Multiple IITI Episodes, ref. Negative											
Positive	0.056751	0.0083593	6.79	0.000	0.0403671	0.073135					

8

9

10

11 Average Marginal Effects

12

When patient age is considered solely, patients aged 16-44 were 2% less likely (-0.015, p=0.015, 95%CI: -0.028, -0.003), while older children were 5-6% more likely (0.054, p<0.001, 95%CI: 0.028, 0.080), to receive a repeat positive prescription that patients of 45 years and over. There was no significant difference in probability for young children than that of patients 45 years and over (0.004, p=0.793, 95%CI: -0.023, 0.030). When considered alone, male gender is associated with 8 percentage points higher risk of repeat positive prescribing (0.083, p<0.001, 95%CI: 0.061, 0.104).

20

D.6.2 Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient gender, across different values of patient age group

margins, dydx(pat_sex) at (agegrp_uti_new = (1 2 3 4)) Expression : Marginal predicted mean, predict() dy/dx w.r.t. : 1.pat_sex (45+ years) (16-44 years) agegrp_uti_new 12 1._at 2._at agegrp_uti_new = 3._at agegrp_uti_new = 3 4 6 -15 years) (0-5)years) 4._at agegrp_uti_new Delta-method dy/dx Std. Err. z P>|z| [95% Conf. Interval] Female (base outcome) Male at .0996745 .0126512 .0196783 .0344754 7.88 4.72 0.000 .0748785 .1244704 1 2 .1314365 .0465815 .0542989 3 .020989 0.61 0.543 .0885595 .1012511 4 .0397436 .031382 -1.270.205 .021764

Note: dy/dx for factor levels is the discrete change from the base level.



31 Note: Effect relative to the probability for Female Patients.

32 Figure D-9: Margins at representative values for the effect on the probability of repeats being

issued on prescriptions with change in patient gender, across different values of patient age group,
 relative to the effect on the probability for female patients

37

D.6.3 Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions, with change in patient age group, across different values of patient gender



Variables that uniquely identify margins: pat_sex _deriv



41 Note: Effect relative to that of patients 45 years and over.

Figure D-10: Margins at representative values for the effect on the probability of repeats being issued on antibiotic prescriptions with change in patient age group, across different values of patient gender, relative to the effect on the probability for patients aged 45 years and over

D.6.4 Predictive margins for the probability of repeats being issued on antibiotic prescriptions, across different values of patient age group and patient gender

margins, at(agegrp_uti_new=(1 2 3 4) pat_sex=(1 0))

Predictive ma Model VCE	irg :	gins OIM				Numb	er of obs	=	17,973
Expression	:	Marginal pr	edic	ted mean,	predict	0			
1at	:	agegrp_uti_ pat_sex	new	=	1 1		(45+ years) (Male)		
2at	:	agegrp_uti_ pat_sex	new	= =	1 0		(45+ years) (Female)		
3at	:	agegrp_uti_ pat_sex	new	= =	2 1		(16-44 years (Male))	
4at	:	agegrp_uti_ pat_sex	new	= =	2 0		(16-44 years (Female))	
5at	:	agegrp_uti_ pat_sex	new	= =	3 1		(6-15 years) (Male)		
6at	:	agegrp_uti_ pat_sex	new	= =	3 0		(6-15 years) (Female)		
7at	:	agegrp_uti_ pat_sex	new	= =	4 1		(0-5 years) (Male)		
8at	:	agegrp_uti_ pat_sex	new	= =	4 0		(0-5 years) (Female)		
	. <u>-</u> -		 Dol+						
		Margin	St	d. Err.	z	P> Z	[95% Co	onf.	Interval]
_at 1 2 3 4 5 6 7 8		.3791592 .2794847 .3575923 .2647246 .3254474 .3464364 .2581725 .2979161	.0 .0 .0 .0 .0 .0 .0	212501 165451 258736 161805 362973 217938 321624 214567	17.84 16.89 13.82 16.36 8.97 15.90 8.03 13.88	0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0	0 .337509 0 .24700 0 .30688 0 .23301 0 .254300 0 .30372 0 .19513 0 .25586	97 57 11 13 51 14 54 18	.4208087 .3119125 .4083036 .2964379 .3965888 .3891515 .3212097 .3399704

. marginsplot, bydim(pat_sex) byopt(rows(1))

Variables that uniquely identify margins: agegrp_uti_new pat_sex



1 2

Figure D-11: Predictive margins for the probability of repeats being issued on antibiotic prescriptions. across different values of patient age group and patient gender, graphed by patient

3 prescrip 4 gender



Figure D-12: Predictive margins for the probability of repeats being issued on antibiotic prescriptions, across different values of patient age group and patient gender, graphed by patient age group

D.6.5 Adjusted predictions for the effect on the probability of repeats being issued on antibiotic prescribing, at specific values of patient gender and age group

. .

margins, at(bat_sex=(0 1) ageg	jrp_uti_new=(1 2	3 4)) atmeans vsqui	sn pos	t
Adjusted pred Model VCE	dictions : OIM		Number of obs	=	17,973
Expression 1at 2at	<pre>: Marginal predic : agegrp_uti_new pat_sex 1.choice_u~w 2.choice_u~w 3.choice_u~w 4.choice_u~w 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d 0.temp_tes~d 1.temp_tes~d 1.temp_tes~d 0.multip~UTI 1.multip~UTI year : agegrp_uti_new</pre>	$\begin{array}{cccc} \text{:ted mean, predic} \\ = & 1 \\ = & 0 \\ = & .3965393 (i \\ = & .4102821 (i \\ = & .106048 (i \\ = & .0871307 (i \\ = & .0871307 (i \\ = & .09545429 (i \\ = & .0454571 (i \\ = & .8366995 (i \\ = & .1633005 (i \\ = & .8304123 (i \\ = & .1695877 (i \\ = & .2014.602 (i \\ = & 1 \\ \end{array}$	t() (45+ years) (Female) mean) mean) mean) mean) mean) mean) mean) mean) mean) mean) mean) mean) mean) (45+ years)		
	pat_sex 1.choice_u~w 2.choice_u~w 3.choice_u~w 4.choice_u~w 0.cult_tes~d 1.cult_tes~d 0.dipstick~d 1.dipstick~d 0.temp_tes~d	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(Male) mean) mean) mean) mean) mean) mean) mean) mean) mean)		

-- ->

	1.temp_tes~d 0.multip~UTI 1.multip~UTI vear	= = =	.1633005 (mean) .8304123 (mean) .1695877 (mean) 2014 602 (mean)	
3at	: agegrp_uti_new pat_sex	= =	2014.002 (mean) 2 0	(16-44 years) (Female)
	1.choice_u~w 2.choice_u~w	=	.3965393 (mean) 4102821 (mean)	
	3.choice_u~w	=	.106048 (mean)	
	4.choice_u~w	=	.0871307 (mean)	
	1.cult_tes~d	=	.0310466 (mean)	
	0.dipstick~d	=	.9545429 (mean)	
	1.dipstick~d	=	.0454571 (mean)	
	1.temp_tes~d	=	.1633005 (mean)	
	0 multip~UTI	=	.8304123 (mean)	
	vear	=	2014.602 (mean)	
4at	: agegrp_uti_new	=	2	(16-44 years)
	pat_sex 1 choice u~w	=	1 3965393 (mean)	(Male)
	2.choice_u~w	=	.4102821 (mean)	
	3.choice_u~w	=	.106048 (mean)	
	4.cnoice_u~w 0.cult tes~d	=	.9689534 (mean)	
	1.cult_tes~d	=	.0310466 (mean)	
	0.dipstick~d 1.dipstick~d	=	.9545429 (mean)	
	0.temp_tes~d	=	.8366995 (mean)	
	1.temp_tes~d	=	.1633005 (mean)	
	0.multip~Ull 1.multip~UTT	=	.8304123 (mean) .1695877 (mean)	
_	year	=	2014.602 (mean)	
5at	: agegrp_uti_new	=	3	(6-15 years)
	1.choice_u~w	=	.3965393 (mean)	(Felila Te)
	2.choice_u~w	=	.4102821 (mean)	
	4.choice_u~w	=	.106048 (mean)	
	0.cult_tes~d	=	.9689534 (mean)	
	1.cult_tes~d 0 dinstick~d	=	.0310466 (mean) 9545429 (mean)	
	1.dipstick~d	=	.0454571 (mean)	
	0.temp_tes~d	=	.8366995 (mean)	
	1.temp_tes~a 0.multip~UTI	=	.1633005 (mean)	
	1.multip~UTI	=	.1695877 (mean)	
6 at	year : agegrn uti new	=	2014.602 (mean)	(6-15 years)
0uc	pat_sex	=	1 1	(Male)
	1.choice_u~w	=	.3965393 (mean)	
	3.choice_u~w	=	.106048 (mean)	
	4.choice_u~w	=	.0871307 (mean)	
	0.cult_tes~d 1.cult_tes~d	=	.9689534 (mean) .0310466 (mean)	
	0.dipstick~d	=	.9545429 (mean)	
	1.dipstick~d	=	.0454571 (mean)	
	1.temp_tes~d	=	.1633005 (mean)	
	0.multip~UTI	=	.8304123 (mean)	
	vear	=	2014.602 (mean)	
7at	: agegrp_uti_new	=	4	(0-5 years)
	pat_sex 1.choice u~w	=	.3965393 (mean)	(Female)
	2.choice_u~w	=	.4102821 (mean)	
	3.choice_u~w	=	.106048 (mean)	
	0.cult_tes~d	=	.9689534 (mean)	
	1.cult_tes~d	=	.0310466 (mean)	
	1.dipstick~d	=	.0454571 (mean)	
	0.temp_tes~d	=	.8366995 (mean)	
	1.temp_tes~a 0.multip~UTI	=	.1633005 (mean)	
	1.multip~UTI	=	.1695877 (mean)	
8. at	year : agegrn uti new	=	2014.602 (mean) 4	(0-5 vears)
0uc	pat_sex	=	1	(Male)
	1.choice_u~w	=	.3965393 (mean)	
	3.choice_u~w	=	.106048 (mean)	
	4.choice_u~w	=	.0871307 (mean)	
	1.cult_tes~d	=	.0310466 (mean)	
	0.dipstick~d	=	.9545429 (mean)	
	1.01pst1CK~d 0.temn tes~d	=	.U4545/1 (mean) .8366995 (mean)	
	1.temp_tes~d	=	.1633005 (mean)	

		0.multip~UT 1.multip~UT year	I = .83 I = .10 = 203	304123 († 595877 († L4.602 (†	mean) mean) mean)		
		۱ Margin	Delta-method Std. Err.	z	P> z	[95% Conf.	Interval]
Men Men Men	_at 1 2 3 4 5 6 7 8	.2758586 .3770028 .2609397 .3550592 .3437207 .322412 .294509 2543218	.0167564 .0216522 .0263203 .022166 .0368394 .0217623	16.46 17.41 15.94 13.49 15.51 8.75 13.75 13.75	$\begin{array}{c} 0.000\\ 0.$.2430167 .3345653 .2288449 .3034723 .3002761 .2502081 .2518556 1906071	.3087004 .4194403 .2930345 .406646 .3871654 .3946159 .3371623 .3180364

Adjusted Predictions for Specific Values of Patient Gender and Patient Age Group with 95% Cls, for Repeat Positive Antibiotic Prescribing Model



Note: All Other Covariates Kept Constant At Respective Means.

Figure D-13: Adjusted predictions for the effect on the probability of repeats being issued on antibiotic prescriptions, at specific values of patient gender and age group, with all other covariates

- kept constant at sample means
- 28

D.7 Marginal effects for the binary model of non-first-line antibiotic prescribing for upper respiratory tract infection

D.7.1 Average marginal effects

6 Table D-23: Average marginal effects for non-first-line antibiotic prescribing for initial 7 presentations of urinary tract infection (Model 2)

	Average Marginal Effects (Model 2)											
Non-first-line Ai	ntibiotic Prescr	ibing for Initial Pr	esentation	s of UTI M	odel							
Variable d	y/dx	Std. Err.	Z	P>z	[95% Conf.	Interval						
Patient Age Group, ref. 45 years and over												
16-44 yrs	-0.0154515	0.0063311	-2.44	0.015	-0.0278602	-0.0030429						
6-15 yrs	0.0541286	0.0132175	4.1	0.000	0.0282227	0.0800346						
0-5 yrs	0.0035729	0.013635	0.26	0.793	-0.0231513	0.0302971						
Patient Gender, ref. Female	0 0007504	0.0100.100	7 5 6	0.000	0.000000	0 4044050						
Male	0.0827534	0.0109402	7.56	0.000	0.061311	0.1041959						
Ordinal Choice of Antibiotic Prescribed, ref.	First-line											
Second-line	0.0795068	0.0076602	10.38	0.000	0.064493	0.0945206						
Third-line	0.187035	0.0121971	15.33	0.000	0.1631291	0.210941						
Not Recommended	0.0112022	0.0110261	1.02	0.310	-0.0104085	0.032813						
Culture Testing Status, ref. Negative												
Positive	0.0499152	0.0189255	2.64	0.008	0.012822	0.0870084						
Urine Dipstick Testing Status, ref. Negative Positive	-0.0493777	0.0193114	-2.56	0.011	-0.0872273	-0.011528						
Temperature Testing Status, ref. Negative Positive	0.0215037	0.009648	2.23	0.026	0.0025941	0.0404134						
Multiple UTI Episodes, ref. Negative Positive	0.056751	0.0083593	6.79	0.000	0.0403671	0.073135						

D.7.2 Margins at representative values for the effect on non-first-line antibiotic prescribing with change in patient gender, across different values of patient age group

margins, dydx(pat_sex) at (agegrp_uti_new = (1 2 3 4))

Average marginal effects Model VCE : OIM Number of obs 17,973 Expression : Marginal predicted mean, predict() dy/dx w.r.t. : 1.pat_sex (45+ years) (16-44 years) (6-15 years) 1._ 1 2 at 2 agegrp_uti_new = 2._at = agegrp_uti_new 3._at agegrp_uti_new = 3 4._at 4 (0-5 years) agegrp_uti_new _____ ____ Delta-method dy/dx P > |z|[95% Conf. Interval] Std. Err. z Female (base outcome) Male _at .1051625 .0136892 0.000 0.002 0.088 .0783322 .1319928 7.68 1 2 3.08 .1091146 3 .0539348 -.0081238 .0316631 .1159933 0.763 4 .0084262 .0279886 0.30 -.0464304 .0632828 Note: dy/dx for factor levels is the discrete change from the base level.

Variables that uniquely identify margins: agegrp_uti_new (note: file nonfirst_uti_pat_sex_age.gph not found) (file nonfirst_uti_pat_sex_age.gph saved)

Margins At Representative Values for Effect on the Probability of Non-first-line Prescribing Occurring with Change in Patient Gender, Across Different Values of Patient Age Group



Figure D-14: Margins at representative values for the effect on the probability of non-first-line antibiotic prescribing occurring with change in patient gender, across different values of patient age group, relative to the effect for female patients

41 42 43

Non-first-line Antibiotic Prescribing Model for UTI

D.7.3 Margins at representative values for the effect on non-first-line antibiotic prescribing occurring with change in patient age group, across different values of patient gender

nargins, dydx(agegrp_uti_new) at (pat_sex= (1 0))										
Average marg [.] Model VCE	inal e : OIM	effects 4		١	Number of ob	os = 1	7,973			
Expression dy/dx w.r.t.	: Mar : 2.a	ginal predicted mean, predict() gegrp_uti_new 3.agegrp_uti_new 4.agegrp_uti_new								
1at 2at	: pat : pat	t_sex t_sex	= =	1 0	(Male) (Female))				
		 dy/d>	Delta-meth Std.Err	iod '. 2	z P> z	[95% Conf.	Interval]			
45+ years		(base out	come)							
16-44 years	_at 1 2	.0076189 .0460826) .0247123 5 .0078067	0.3	31 0.758 90 0.000	0408163 .0307818	.0560541			
6-15 years	at 1 2	.1630435 .2142713	5.0319727 3.0146911	5.1 . 14.5	LO 0.000 59 0.000	.1003782 .1854773	.2257089			
0-5 years	at 1 2	2006749 .2974113) .0287321 3 .0151943	6.9 19.5	98 0.000 57 0.000	.1443611 .267631	.2569887 .3271916			
Note: dy/dx t	for fa	actor levels	is the disc	rete cha	ange from th	e base level.				



Figure D-15: Margins at representative values for the effect on the probability of non-first-line antibiotic prescribing occurring with change in patient age group, across different values of patient gender, relative to the effect on the probability for patients aged 45 years and over

D.7.4 Predictive margins for the effect on non-first-line antibiotic prescribing occurring at different values of patient age group and patient gender

. margins, at(agegrp_uti_new==(4 3 2 1) pat_sex==(1 0))											
	Predictive ma Model VCE	ar :	gins OIM			Numbei	r of obs	=	17,973		
	Expression	:	Marginal predic	ted mean,	predict()					
	1at	:	agegrp_uti_new pat_sex	= =	4 1	(0-5 years) Male)				
	2at	:	agegrp_uti_new pat_sex	= =	4 0	(0-5 years) Female)				
	3at	:	agegrp_uti_new pat_sex	= =	3 1	(6-15 years) Male)				
	4at	:	agegrp_uti_new pat_sex	= =	3 0	(6-15 years) Female)				
	5at	:	agegrp_uti_new pat_sex	= =	2 1	(16-44 years) Male)				
	6at	:	agegrp_uti_new pat_sex	= =	2 0	(16-44 years) Female)				
	7at	:	agegrp_uti_new pat_sex	= =	1 1	(45+ years) Male)				
	8at	:	agegrp_uti_new pat_sex	= =	1 0	(45+ years) Female)				




Ø.

S

Female

Male Female

17 Figure D-16: Predictive margins for the effect on the probability of non-first-line antibiotic

Male

Female

Patient Gender

Female

Male

Male

- 18 prescribing occurring at different values of patient age group and patient gender, graphed by 19 patient age group
- 20
- 21
- 22
- 23

D.8 Comparison of final mixed-effects models with fixed-effects models

Table D-24:Summary of fixed-effects models compared against mixed model for choice of
antibiotic prescribed for initial presentations of urinary tract infection

Ordinal Choice of Antibiotic Prescribed Model	Mixed, three-l (patient, provi levels)	evel model der, practice	Fixed model with dummies for practice		Fixed model practice	with no
	Exp. coeff.	t-statistic	Exp. coeff.	t-statistic	Exp. coeff.	t-statistic
Patient age group (ref. 45+ yrs)						
16-44 yrs	0.165***	-3.98	0.172***	-4.36	0.185***	-4.8
6-15 yrs	0.916***	-10.49	0.862***	-10.45	0.862***	-10.63
0-5 yrs	1.257***	-13.91	1.173***	-13.84	1.177***	-14.15
Patient gender (ref. female)						
Male	0.574***	-8.99	0.550***	-9.14	0.540***	-9.06
Patient age group ## Gender (ref. 45	+ vrs # female)					
16-44 yrs # Male	-0 133	(-1 13)	-0.0862	(-0.76)	-0 0824	(-0.73)
6-15 vrs # Male	-0.265	(-1.43)	-0.333	(-1.88)	-0.309	(-1.78)
0.5 yrs # Male	-0 514**	(-2.91)	-0 383*	(-2.29)	-0 474**	(-2.90)
	0.314	(2.51)	0.505	(2.23)	0.474	(2.50)
Repeat Prescription Stats (ref. negat	ive)					
Positive	0.551***	-10.11	0.524***	-10.99	0.469***	-10.28
Patient age group ## Repeat prescri	ntion status (ref	45+ vrs 1 # neg	rative)			
16-44 vrs # Positive	0.00502	-0.07	0.00526	-0.08	-0.00855	(-0.13)
6-15 vrs # Positive	-0.356**	(-2.66)	-0.359**	(-2.84)	-0.284*	(-2.28)
0-5 yrs # Positive	-0.643***	(-4.45)	-0.627***	(-4.63)	-0.447***	(-3.37)
Patient comorbid condition (ref. neg	ative)					
Positive	0.195***	-4.53	0.165***	-4.03	0.146***	-3.62
Missing	0.504	-1.63	-0.212	(-0.86)	0.591***	-7.01
Culture testing status (ref.						
negative)						
Positive	-0.309**	(-3.09)	-0.336***	(-3.60)	-0.457***	(-5.43)
Directick testing status (ref						
negative)						
Positive	-0 336**	(-3 15)	-0 166	(-1 77)	-0 184*	(-2 57)
	0.550	(3.13)	0.100	(1.77)	0.104	(2.57)
Year (unit increase)	0.0274*	-2.15	0.0217*	-2.24	0.0199*	-2.23
Proscribing roacon recorded (ref. po						
Positive	gative)		0 428***	-5.05	-0 0942*	(-2 33)
			0.420	5.05	0.0342	(2.55)
Weekend consultation (ref. negative)					
Positive			0.117*	-2.33		
	、					
Patient in remote area (ref. negative)				0 222***	F 02
Positive					0.332	-5.03
IVIISSIIIg					-0.488	(-2.23)
Patient PHN (ref. Perth North)						
Perth South					-0.241***	(-6.86)
Country WA					-0.309***	(-7.75)
Interstate					-0.247*	(-1.98)
Missing					0.361*	-2.13
Practice Size (ret. Medium/large)					0 4 5 0 * *	
Small					0.150**	-3.02

	Mixed, three (patient, prov levels)	Mixed, three-level model (patient, provider, practice levels)		with practice	Fixed model with no practice		
	Exp. coeff.	t-statistic	Exp. coeff.	t-statistic	Exp. coeff.	t-statistic	
/		2.4.4	40 74*	2.24	20.00*	2.22	
cut1	55.1/*	-2.14	43.71*	-2.24	39.98*	-2.22	
cut2	57.47*	-2.23	45.73*	-2.35	41.91*	-2.33	
cut3	58.54*	-2.27	46.70*	-2.4	42.85*	-2.38	
var(_cons[~)	0.0622	-1.63					
var(_cons[~]	1.075***	-12.19					
Ν	17973		17973		17973		
Note: exponentiated coe	fficients, significance: * p	<0.05, ** p<0.01	l, *** p<0.001				

Г

APPENDIX E – APPENDICES TO TRENDS IN PRESCRIBING FOR UPPER RESPIRATORY TRACT INFECTION CHAPTER (CHAPTER 6)

E.1 Tables

1

E.1.1 Upper respiratory tract infection conditions excluding influenza / influenza-like illness

Table E-1: Linear chi-squared tests for trend using linear regression for upper respiratory tract infection conditions

	All L	JRTI excl. Infl	uenza/ILI	<u>Rhinosinusitis</u>			<u>Pharyngitis</u>			AOM		
Linear Regression Model for:												
	Coef.	[95% Con	f. Interval]	Coef.	[95% Con	f. Interval]	Coef.	[95% Conf	. Interval]	Coef.	[95% Con	f. Interval]
Likely Unnecessary Antibiotic												
Prescribing rate												
Month	-0.00092	-0.001041	-0.000803	-0.00185	-0.001984	-0.00172	-0.00118	-0.001368	-0.001001	0.000168	-6.04E-05	0.0003958
Constant	1.461116	1.383122	1.539111	2.125135	2.038259	2.212011	1.632775	1.512335	1.753215	0.661007	0.5112001	0.810813
Antibiotic Prescribing rate												
Month	-0.00232	-0.002478	-0.002165	-0.02	-0.00225	-0.001753	-0.00064	-0.000774	-0.000503	-0.00265	-0.002741	-0.002555
Constant	2.005439	1.902623	2.108255	1.65027	1.487045	1.813494	1.130649	1.041434	1.219863	2.325499	2.264192	2.386807
Prescribing of Second-line												
Antibiotic agent	0.001120	0.0008867	0.0012012	0.002066	0 0027945	0 0022484	n/2	n/n	2/2	0.001456	0.0009201	0.0020810
Month	0.001139	0.0008867	0.0013912	0.003066	0.0027845	0.0033484	n/a	n/a	n/a	0.001456	0.0008301	0.0020819
Constant	-0.5856	-0.751273	-0.419931	-1./69/5	-1.954933	-1.584575	n/a	n/a	n/a	-0.62812	-1.03921	-0.217021
Proscribing of antibiotic Not												
Recommended in Guidelines												
Month	-0.00138	-0.001527	-0 001231	-0.00025	-0.000513	6 61E-06	-0.00425	-0.004463	-0 00404	0.000114	-0 00013	0 0003571
Constant	1 301287	1 204303	1 398271	0 595765	0.000313	0.011-00	3 272015	3 133117	3 410913	0.132159	-0.00013	0.0003371
constant	1.501207	1.204303	1.550271	0.555705	0.4231143	0.7004134	5.272015	5.155117	5.410515	0.132135	0.027742	0.2520554
Non-first-line antibiotic												
prescribing												
Month	-0.00024	-0 000592	0 0001121	0.002813	0 0023695	0 0032569	-0.00425	-0 004463	-0 00404	0.00157	0 0010925	0 0020467
Constant	0.715685	0.4843257	0.9470449	-1.17399	-1.465379	-0.882599	3.272015	3.133117	3.410913	-0.49596	-0.809316	-0.182597
	0.7 20000	01.0.0207	0.0.701.0	1.17000	1100070	01002000	0.172020	0.10011/	01120020	01.0000	01000020	0.101007
Repeat(s) issued on												
antibiotic prescription												
 Month	0.000181	-0.000237	0.0005978	0.002034	0.0014403	0.0026271	-0.00105	-0.001406	-0.000685	6.06E-05	-0.000298	0.0004195
Constant	0.201558	-0.072546	0.4756627	-0.98178	-1.371488	-0.592079	0.948381	0.7116284	1.185133	0.334132	0.0984428	0.5698209

Antibiotic prescription as Private prescription month constant	-0.00052 0.372086	-0.000662 0.2808433	-0.000384 0.463328	0.000257 -0.06276	-0.000375 -0.477817	0.0008892 0.3523042	-0.00129 0.887227	-0.001706 0.6147981	-0.000877 1.159655	-0.00062 0.440756	-0.000929 0.2397699	-0.000317 0.6417414
Temperature recording												
during consultation	0 002020	0 0024422	0 0044142	0.006440	0 0045496	0 0092402	0.016020	0 0115159	0 0205422	0 020649	0 0240729	0.0252214
constant	-2.25469	-2.573873	-1.935503	-3.59128	-4.839355	-2.343196	-9.27212	-12.23628	-6.307953	-17.8002	-21.46092	-14.13946

 Table E-2:
 Linear chi-squared tests for trend using linear regression for acute rhinosinusitis

	Acute Rhinosinusitis					
Linear Regression Model for Antibiotic agent Prescribing Rate:	Coef.	[95% Conf.	Interval]			
Amoxicillin						
Month	-0.002824	-0.003273	-0.002375	***		
Constant	2.180083	1.885295	2.474871			
Amoxicillin with clavulanate						
Month	0.003009	0.002723	0.003295	***		
Constant	-1.736256	-1.924167	-1.548345			
Cofelovia						
Ceralexin	0.000228	0.000076	0.000400	**		
Month Constant	0.000238	0.000076	0.000400			
Constant	-0.032350	-0.139006	0.074307			
Cefuroxime						
Month	0.0000569	0.000000	0.0000411	***		
Constant	-0.033401	-0.0437467	-0.0230554			
Clarithromycin						
Month	-0 000573	-0.000636	-0 000509	***		
Constant	0.437955	0.396501	0.479409			
constant	0.107000	0.00001	0.175105			
Doxycycline						
Month	0.0003996	0.000321	0.0004781	***		
Constant	2432993	-0.29488	-0.1917186			
Roxithromycin						
Month	-0.000678	-0.000881	-0.000474	***		
Constant	0.547367	0.413765	0.680970			

1
2

Table E-3: Linear chi-squared tests for trend using linear regression for acute pharyngitis / tonsillitis

	<u>Ac</u>	ute Pharyngitis /	<u>Tonsillitis</u>	
Linear Regression Model for Antibiotic				
agent Prescribing Rate:	Coef.	[95% Conf.	Interval]	
Amoxicillin				
Month	-0.003214	-0.003517	-0.002911	***
Constant	2.266512	2.067375	2.465650	
Amoxicillin with clavulanate				
Month	0.000669	0.000193	0.001144	**
Constant	-0.293364	-0.605802	0.019074	
Benzathine benzylpenicillin				
Month	4.58e-06	3.52e-06	5.64e-06	***
Constant	-0.0029229	-0.0036208	-0.0022249	
Cefalexin				
Month	0.000837	0.000707	0.000966	***
Constant	-0.444656	-0.529916	-0.359396	
Month	0.001120	0.001224	0,000026	***
Month Constant	-0.001130	-0.001324	-0.000936	
Constant	0.799339	0.072098	0.920581	
Phenoxymethlypenicillin				
Month	0.003621	0.003438	0.003803	***
Constant	-1.974824	-2.094704	-1.854944	
		-		
Trimethoprim with sufamethoxazole				
Month	0000599	0.000	0000771	***
Constant	.0435457	.0322898	.0548015	

Table E-4: Linear chi-squared tests for trend using linear regression for acute otitis media

	Acute Otitis Media					
Linear Regression Model for Antibiotic agent Prescribing Rate:	Coef.	[95% Conf.	Interval]			
Amoxicillin						
Month	-0.001656	-0.002131	-0.001182	***		
Constant	1.548973	1.237216	1.860730			
Amoxicillin with clavulanate						
Month	0.001402	0.000772	0.002032	***		
Constant	-0.596382	-1.010298	-0.182466			
Cefalcor						
Month	-0.000505	-0.000597	-0.000412	***		
Constant	0.358851	0.297827	0.419876			
Cefalexin						
Month	0.000652	0.000494	0.000810	***		
Constant	-0.336478	-0.440348	-0.232608			
Cefuroxime						
Month	0.0001984	0.0001742	0.0002226	***		
Constant	-0.1212103	-0.137125	-0.1052957			
Erythromycin						
Month	-0.000074	-0.000128	-0.000020	**		
Constant	0.070903	0.035490	0.106317			
Trimethoprim with sulfamethoxazole						
Month	0.0001058	0.0000567	0.000155	***		
Constant	-0.0519367	-0.0842254	-0.019648			

2 E.1.2 Influenza / influenza-like illness

3

7

- There were 253 antibiotic prescriptions among 2,175 initial presentations for influenza/ILI, a prescribing rate of 12%. This ranged from 0 to 26 prescriptions per
 month.
- 6 Table E-5: Antibiotic prescribing rate and prescribing rates for individual antibiotic agents prescribed for influenza / influenza-like illness.

	Prescribing Outcomes for All Patients with initial presentations of Influenza/ILI											
URTI	Dependent Variable	a) (Mov	Descriptive	<u>Statistics</u> Data)			b) Linear Regression Model for Trend					
Condition	Prescribing Outcome Monthly Rate	Mean prop.	Jan 2012 prop.	Jun 2017 prop.	Coefficient (unit increase per month)	[95% Conf.	Interval]	p-value	R-squared	Jan 2012 predicted value	Jun 2017 predicted value	Percentage Point Difference *
Influenza	Overall Antibiotic Prescribing Rate +	0.11	0.09	0.11	-0.00045	-0.00075	-0.00015	0.004	0.1243	0.097008	0.013366	1
	amoxicillin	0.19	0.18	0.56	0.005098	0.003431	0.006765	0.000	0.3684	0.564242	0.497576	50
	amoxicillin with clavulanate	0.23	0.01	0.02	-0.0008	-0.003	0.001413	0.474	0.008	0.018182	0.018182	2
	cefalexin	0.11	0.00	0.20	0.002353	0.001764	0.002943	0.000	0.4986	0.071515	0.071515	7
	clarithromycin	0.22	0.38	0.06	-0.004	-0.00516	-0.00283	0.000	0.4231	0.112727	-0.52061	-52

Note +: The rate calculation for overall antibiotic prescribing was calculated with the numerator being all patients with initial presentations of influenza/ILI, who were prescribed an antibiotic. The denominator is all patients with initial presentations of influenza/ILI.

Note *: The percentage point difference uses predicted values for first and last months of the study period, January 2012 to June 2017 inclusive.

Note: The rate calculation for each individual antibiotic agent was calculated with the numerator being all patients with initial presentations of influenza/ILI who were prescribed the particular antibiotic. The denominator is all patients with initial presentations of the influenza/ILI who were prescribed any antibiotic agent.

 Table E-6:
 Linear chi-squared tests for trend using linear regression for influenza / influenza-like illness

	Influenza/ILI						
Linear Regression Model for:							
	Coef.	[95% Conf.	Interval]				
Overall Antibiotic Prescribing rate							
month	-0.00045	-0.000750	-0.000152	**			
constant	0.416904	0.220684	0.613125				
Repeat(s) issued on antibiotic prescription							
month	-0.00282	-0.003577	-0.002065	***			
constant	2.203313	1.706657	2.699969				
Antibiotic prescription written as Private prescription							
month	-0.00359	-0.005208	-0.001972	***			
constant	2.492115	1.429445	3.554786				
Temperature recording during consultation							
month	0.003194	0.002156	0.004231	***			
constant	-1.76863	-2.449972	-1.087284				

 Table E-7:
 Linear chi-squared tests for trend using linear regression for influenza / influenza-like illness

		Influenza	/111		
ear Regression Model for:	Coef.	[95% Conf.	Intervall		
oxicillin prescribing rate	coen	[55/6 colin.	intervalj		
nth	0.005098	0.003431	0.006765	***	
nstant	-3.15986	-4.254730	-2.064993		
ovicillin with clavulanate prescribing rate					
nth	-0.0008	-0.003004	0.001413		
nstant	0.74856	-0.701690	2.198811		
thromycin prescribing rate					
nth	0.000063	0.000859	0.875000		
nstant	0.021824	0.544591	0.934000		
alexin prescribing rate					
nth	0.002353	0.001764	0.002943	***	
nstant	-1.43917	-1.826245	-1.052097		
rithromycin prescribing rate					
nth	-0.004	-0.005160	-0.002830	***	
nstant	2.843512	2.078337	3.608688		
cloxacillin prescribing rate					
nth	-0.0010289	-0.0013899	-0.0006678	***	
nstant	0.6882292	0.451111	0.9253474		
kithryomycin prescribing rate					
nth	-0.0006743	-0.0010638	-0.0002847	***	
nstant	0.4781669	0.2223176	0.7340161		
nth instant thromycin prescribing rate inth instant alexin prescribing rate inth instant rithromycin prescribing rate inth instant cloxacillin prescribing rate inth instant cithryomycin prescribing rate inth instant	-0.0008 0.74856 0.000063 0.021824 0.002353 -1.43917 -0.004 2.843512 -0.0010289 0.6882292 -0.0006743 0.4781669	-0.003004 -0.701690 0.000859 0.544591 0.001764 -1.826245 -0.005160 2.078337 -0.0013899 0.451111 -0.0010638 0.2223176	0.001413 2.198811 0.875000 0.934000 0.002943 -1.052097 -0.002830 3.608688 -0.0006678 0.9253474 -0.0002847 0.7340161	*** *** ***	

E.1.3 Mean prescribing rates for all upper respiratory tract infection conditions including influenza / influenza-like illness

	,			
	Acute Rhinosinusitis	Acute Pharyngitis/ Tonsillitis	АОМ	Influenza/ILI
<u>Antibiotic</u>				
amoxicillin	0.33	0.16	0.46	0.19
amoxicillin with clavulanate	0.24	0.15	0.32	0.23
azithromycin	0.03	0.01	0.01	0.06
benzathine penicillin	0.00	0.00	0.00	0.00
cefaclor	0.02	0.01	0.03	0.02
cefalexin	0.12	0.10	0.09	0.11
ceftriaxone	0.00	0.00	0.00	0.00
cefuroxime	0.00	0.00	0.01	0.01
ciprofloxacin	0.00	0.00	0.01	0.00
clarithromycin	0.06	0.02	0.01	0.22
clindamycin	0.00	0.00	0.00	0.00
dicloxacillin	0.00	0.00	0.00	0.00
doxycyline	0.02	0.00	0.00	0.04
erythromycin	0.03	0.03	0.02	0.03
flucloxacillin	0.00	0.00	0.00	0.01
gentamycin	0.00	0.00	0.00	0.00
minocycline	0.00	0.00	0.00	0.00
moxifloxacin	0.00	0.00	0.00	0.00
nitrofurantoin	0.00	0.00	0.00	0.00
norfloxacin	0.00	0.00	0.00	0.00
phenoxymethylpenicillin	0.03	0.40	0.00	0.03
procaine penicillin	0.00	0.06	0.00	0.01
roxithromycin	0.10	0.05	0.01	0.04
tobramycin	0.00	0.00	0.00	0.00
trimethoprim	0.00	0.00	0.00	0.00
trimethoprim with sulfamethoxazole	0.01	0.00	0.02	0.00

Table E-8:Mean prescribing rates of individual antibiotics, by upper respiratory tract infection
condition, January 2012 to June 2017 inclusive.

E.2 Additional time series plots

E.2.1 All upper respiratory tract infection conditions excluding influenza / influenzalike illness



Figure E-1: Time Series Plot of Unnecessary Antibiotic Prescribing rates for all URTI diagnoses, Jan 2012-Jun 2017 inclusive, by month



Figure E-2: Time series plot of antibiotic prescribing rates for antibiotics for all initial presentations of upper respiratory tract infection, Jan 2012-Jun 2017 inclusive, by month





Figure E-3: Time series plot of second-line antibiotic prescribing rates for all initial presentations of upper respiratory tract infection, Jan 2012-Jun 2017 inclusive, by month



- 6 Figure E-4: Time series plot of antibiotic prescribing rates for antibiotics not recommended in
- the guidelines for all initial presentations of upper respiratory tract infection, Jan 2012-Jun 2017
- 8 inclusive, by month

1E.2.1By specific condition2E.2.1.1Acute rhinosinusitis



Figure E-5: Time series plot of antibiotic prescribing rates for initial presentations of acute rhinosinusitis, January 2012 to June 2017, inclusive, by month





9 Figure E-6: Time series plot of non-first-line antibiotic prescribing rates for initial presentations of 10 acute rhinosinusitis, January 2012 to June 2017, inclusive, by month



Figure E-7: Time series plot of second-line antibiotic prescribing rates for initial presentations of acute rhinosinusitis, January 2012 to June 2017, inclusive, by month

E.2.1.2 Acute pharyngitis / tonsillitis



- Figure E-8: Time series plot of unnecessary antibiotic prescribing rates for initial presentations
 of acute pharyngitis / tonsillitis, January 2012 to June 2017, inclusive, by month





Figure E-9: Time series plot of antibiotic prescribing rates for initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017, inclusive, by month



⁶

- 7 Figure E-10: Time series plot of amoxicillin and phenoxymethylpenicillin prescribing rates for
- 8 initial presentations of acute pharyngitis / tonsillitis, January 2012 to June 2017, inclusive, by
- 9 month



Figure E-11: Time series plot of non-first-line antibiotic prescribing rates for initial presentations of acute otitis media, January 2012 to June 2017, inclusive, by month



Figure E-12: Time series plot of second-line antibiotic prescribing rates for initial presentations of acute otitis media, January 2012 to June 2017, inclusive, by month

APPENDIX F – APPENDICES TO TRENDS IN PRESCRIBING FOR URINARY TRACT INFECTION CHAPTER (CHAPTER 7)

3 F.1 Tables

4

Table F-1: Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for patients with urinary tract infection

Linear Regression Model for:	Coef.	Std. Err.	t	P>t	[95% Conf.	Interval]	Num.	F(1, 64)	Prob > F	R-squared	Adj R-squared	Root MSE
antibiotic prescribing rate												
Month	0.000765	5.51E-05	13.87	0.0000	0.0006544	0.0008746	66	192.5	0.0000	0.7505	0.7466	0.00853
_cons	0.342587	0.03619	9.47	0.0000	0.2702891	0.4148852						
first-line agent												
Month	-0.00091	8.61E-05	-10.5	0.0000	-0.001079	-0.000735	66	111.07	0.0000	0.6344	0.6287	0.01332
_cons	0.995734	0.056537	17.61	0.0000	0.8827877	1.10868						
second-line agent												
Month	0.001555	0.000073	21.31	0.0000	0.0014094	0.001701	66	453.95	0.0000	0.8764	0.8745	0.0113
_cons	-0.61706	0.04794	-12.9	0.0000	-0.712835	-0.521293						
third-line agent												
Month	-0.0002	0.000116	-1.7	0.0940	-0.00043	0.0000346	66	2.89	0.0939	0.0432	0.0283	0.018
_cons	0.237212	0.076367	3.11	0.0030	0.084651	0.3897735						
not recommended agent												
Month	-0.00045	7.22E-05	-6.24	0.0000	-0.000595	-0.000306	66	38.9	0.0000	0.378	0.3683	0.01117
_cons	0.384118	0.047418	8.1	0.0000	0.289389	0.4788471						
non-first-line agent												
Month	0.000907	8.61E-05	10.54	0.0000	0.0007352	0.0010792	66	111.07	0.0000	0.6344	0.6287	0.01332
_cons	0.004266	0.056537	0.08	0.9400	-0.10868	0.1172121						
repeat(s) issued on script												
month	-0.00175	0.000078	-22.5	0.0000	-0.001907	-0.001596	66	504.65	0.0000	0.8875	0.8857	0.01206
_cons	1.44066	0.051199	28.14	0.0000	1.338379	1.54294						
private script												
month	-0.00084	5.62E-05	-14.9	0.0000	-0.000949	-0.000725	66	222.11	0.0000	0.7763	0.7728	0.00869
_cons	0.574207	0.036887	15.57	0.0000	0.5005167	0.6478969						
urine dipstick requested & performed												
month	0.000122	6.81E-05	1.78	0.0790	-1.46E-05	0.0002575	66	3.18	0.0792	0.0474	0.0325	0.01054
_cons	0.874982	0.044726	19.56	0.0000	0.7856307	0.9643334						
temperature recorded during consult												
month	0.001978	0.000138	14.35	0.0000	0.0017029	0.0022538	66	205.85	0.0000	0.7628	0.7591	0.02134
_cons	-1.14707	0.090562	-12.7	0.0000	-1.327988	-0.96615						

Table F-2: Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for women with initial presentations of urinary tract infection

Linear Regression Model for:	Coef.	Std. Err.	t	P>t	[95% Conf.	Interval]	Num. obs.	F(1, 64)	Prob > F	R-squared	Adj R-squared	Root MSE
antibiotic prescribing rate												
month	0.000753	5.81E-05	12.96	0.0000	0.000637	0.000869	66	167.9	0.0000	0.724	0.7197	0.00899
_cons	0.373134	0.038146	9.78	0.0000	0.296929	0.449339						
first-line agent												
month	-0.00102	0.000091	-11.16	0.0000	-0.0012	-0.00083	66	124.48	0.0000	0.6604	0.6551	0.01408
_cons	1.103023	0.05975	18.46	0.0000	0.983658	1.222387						
second-line agent												
month	0.001626	7.76E-05	20.96	0.0000	0.001471	0.001782	66	439.29	0.0000	0.8728	0.8709	0.01201
_cons	-0.6787	0.050966	-13.32	0.0000	-0.78052	-0.57689						
third-line agent												
month	-0.00022	0.000107	-2.06	0.0440	-0.00044	-6.48E-06	66	4.24	0.0437	0.0621	0.0474	0.01663
_cons	0.231226	0.070557	3.28	0.0020	0.090273	0.372178						
not recommended agent												
month	-0.00039	6.41E-05	-6.09	0.0000	-0.00052	-0.00026	66	37.04	0.0000	0.3666	0.3567	0.00993
_cons	0.344456	0.042124	8.18	0.0000	0.260305	0.428608						
non-first-line agent												
month	0.001015	0.000091	11.16	0.0000	0.000833	0.001197	66	124.48	0.0000	0.6604	0.6551	0.01408
_cons	-0.10302	0.05975	-1.72	0.0890	-0.22239	0.016342						
repeat(s) issued on script												
month	-0.00191	7.52E-05	-25.37	0.0000	-0.00206	-0.00176	66	643.52	0.0000	0.9095	0.9081	0.01164
_cons	1.521096	0.049386	30.8	0.0000	1.422437	1.619755						
private script												
month	-0.00088	5.72E-05	-15.4	0.0000	-0.001	-0.00077	66	237.09	0.0000	0.7874	0.7841	0.00886
_cons	0.603302	0.037587	16.05	0.0000	0.528214	0.678391						
urine dipstick requested &												
month	8.17E-05	7.94E-05	1.03	0.3070	-7.7E-05	0.00024	66	1.06	0.3070	0.0163	0.0009	0.01228
_cons	0.897667	0.052126	17.22	0.0000	0.793534	1.001799						
temperature recorded during												
month	0.001867	0.000127	14.74	0.0000	0.001614	0.00212	66	217.32	0.0000	0.7725	0.7689	0.0196
_cons	-1.08301	0.083169	-13.02	0.0000	-1.24916	-0.91686						

Table F-3: Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for men with initial presentations of urinary tract infection

Linear Regression Model for:	Coef.	Std. Err.	t	P>t	[95% Conf.	Interval]	Num. obs.	F(1, 64)	Prob > F	R-squared	Adj R-squared	Root MSE
antibiotic prescribing rate												
month	0.001107	0.0001	11.03	0.0000	0.000906	0.001307	66	121.66	0.0000	0.6553	0.6499	0.01553
_cons	0.049941	0.065884	0.76	0.4510	-0.08168	0.181559						
first line acoust												
first-line agent	0.00107	0 000225	6.05	0 0000	0 00262	0.00122	66	26.65	0 0000	0 2642	0 25/2	0.05024
cons	-0.00197	0.000525	-0.05	0.0000	-0.00202	2 040586	00	50.05	0.0000	0.5042	0.5542	0.05054
	1.022798	0.215050	7.0	0.0000	1.190011	2.049560						
second-line agent												
month	0.001421	0.000169	8.4	0.0000	0.001083	0.00176	66	70.48	0.0000	0.5241	0.5167	0.0262
_cons	-0.57416	0.111191	-5.16	0.0000	-0.79629	-0.35203						
third-line agent												
month	0.000643	0.00025	2.57	0.0120	0.000144	0.001143	66	6.62	0.0124	0.0937	0.0795	0.0387
cons	-0.18445	0.16422	-1.12	0.2660	-0.51252	0.143617		0.01	0.011	0.0007	0.0700	0.0007
not recommended agent												
month	-9.5E-05	0.000157	-0.61	0.5460	-0.00041	0.000218	66	0.37	0.5463	0.0057	-0.0098	0.02427
_cons	0.135809	0.102974	1.32	0.1920	-0.06991	0.341523						
non-first-line agent												
month	0.001969	0.000325	6.05	0.0000	0.00132	0.002619	66	36.65	0.0000	0.3642	0.3542	0.05034
cons	-0.6228	0.213636	-2.92	0.0050	-1.04959	-0.19601						
repeat(s) issued on script			<u> </u>									
month	-0.0012	0.000185	-6.47	0.0000	-0.00157	-0.00083	66	41.85	0.0000	0.3954	0.3859	0.02867
_cons	1.212897	0.121681	9.97	0.0000	0.969811	1.455984						
private script												
month	-0.00057	7.24E-05	-7.93	0.0000	-0.00072	-0.00043	66	62.87	0.0000	0.4955	0.4877	0.01121
_cons	0.409914	0.047566	8.62	0.0000	0.314889	0.504939						
uring directick requested & part	ormod											
month	0.000344	3 08F-05	8.64	0 0000	0 000264	0 000423	66	74 74	0 0000	0 5387	0 5315	0.00616
cons	0.739175	0.026128	28 29	0.0000	0.686979	0.791371	00	/4./4	0.0000	0.5587	0.5515	0.00010
	0.700170	0.020120	20.25	0.0000	0.000075	0.701071						
temperature recorded during c	onsult											
month	0.001759	0.000117	15.06	0.0000	0.001525	0.001992	66	226.85	0.0000	0.78	0.7765	0.01807
cons	-1.03954	0.07669	-13.56	0.0000	-1.19275	-0.88634						

 Table F-4:
 Summary of multiple, bivariable linear regression models / Chi-squared linear test for trend for children with initial presentations of urinary tract infection

Linear Regression Model for:	Coef.	Std. Err.	t	P>t	[95% Conf.	Interval]	Num. obs.	F(1, 64)	Prob > F	R-squared	Adj R-squared	Root MSE
antibiotic prescribing rate												
month	0.000506	0.000137	3.68	0.0000	0.000231	0.00078	66	13.57	0.0005	0.175	0.1621	0.02124
_cons	0.410088	0.090142	4.55	0.0000	0.230008	0.590168						
first-line agent												
month	0.000546	0.000142	3.84	0.0000	0.000262	0.00083	66	14.73	0.0003	0.1871	0.1744	0.02202
_cons	-0.18466	0.093466	-1.98	0.0530	-0.37138	0.00206						
second-line agent												
month	0.000975	0.00012	8.16	0.0000	0.000737	0.001214	66	66.63	0.0000	0.5101	0.5024	0.01849
_cons	-0.08013	0.07847	-1.0200	0.311	-0.2369	0.076627						
third-line agent												
month	-0.00027	0.000184	-1.44	0.1540	-0.00063	0.000102	66	2.08	0.1541	0.0315	0.0163	0.0285
_cons	0.334094	0.120941	2.76	0.0070	0.092487	0.575701						
not recommended agent												
month	-0.00126	0.000154	-8.16	0.0000	-0.00156	-0.00095	66	66.58	0.0000	0.5099	0.5022	0.02382
_cons	0.930701	0.101076	9.21	0.0000	0.728778	1.132624						
non-first-line agent												
month	-0.00055	0.000142	-3.84	0.0000	-0.00083	-0.00026	66	14.73	0.0003	0.1871	0.1744	0.02202
_cons	1.184661	0.093466	12.67	0.0000	0.997941	1.371381						
repeat(s) issued on script												
month	-0.00055	0.000209	-2.64	0.0100	-0.00097	-0.00013	66	6.98	0.0104	0.0983	0.0842	0.03232
_cons	0.708788	0.137154	5.17	0.0000	0.434792	0.982783						
private script												
month	-0.00074	9.14E-05	-8.08	0.0000	-0.00092	-0.00056	66	65.32	0.0000	0.5051	0.4974	0.01414
_cons	0.502395	0.060014	8.37	0.0000	0.382504	0.622286						
urine dipstick requested & perf	formed											
month	0.000305	5.61E-05	5.43	0.0000	0.000193	0.000416	66	29.5	0.0000	0.3155	0.3048	0.00867
_cons	0.769744	0.036812	20.91	0.0000	0.696203	0.843285						
temperature recorded during c	onsult											
month	0.002833	0.000276	10.28	0.0000	0.002282	0.003383	66	105.67	0.0000	0.6228	0.6169	0.04265
cons	-1.61954	0.180989	-8.95	0.0000	-1.98111	-1.25797						

2 Table F-5:

3

4

Frequency table of mean prescribing rates for individual antibiotics prescribed for initial presentations of urinary tract infection, January 2012 to June 2017 inclusive, by patient group

			5
	Women	Men	Children < 16yrs
Antibiotic Agent			6
amoxicillin	0.03	0.03	0. <u>0</u> 8
amoxicillin with clavulanate	0.06	0.12	0.16
azithromycin	0.00	0.00	0.00
benzathine penicillin	0.00	0.01	0.00
cefaclor	0.00	0.00	0. G 1
cefalexin	0.39	0.36	0.56
ceftriaxone	0.00	0.00	9.0 0
cefuroxime	0.00	0.00	0.00
ciprofloxacin	0.00	0.01	q.q 0
clarithromycin	0.00	0.00	0.00
clindamycin	0.00	0.00	d.2 0
dicloxacillin	0.00	0.00	0.00
doxycyline	0.00	0.00	d.g o
erythromycin	0.00	0.00	0.00
flucloxacillin	0.00	0.00	d.0 0
gentamycin	0.00	0.00	0.00
minocycline	0.00	0.00	d . go
moxifloxacin	0.00	0.00	0.00
nitrofurantoin	0.04	0.04	d.01
norfloxacin	0.02	0.08	q. g 0
phenoxymethylpenicillin	0.00	0.00	0.00
roxithromycin	0.00	0.00	G. 80
sodium fusidate	0.00	0.00	0.00
trimethoprim	0.44	0.33	q.g 9
trimethoprim with sulfamethoxazole	0.01	0.02	0.08

20

21 Table F-6:22

Frequency table of counts of antibiotic prescriptions for initial presentations of urinary tract infection per half-year, by patient group

Half-year	Women	Men	Children <16	Total
2012h1	232	21	21	274
2012h2	272	26	39	337
2013h1	242	26	31	299
2013h2	220	30	43	293
2014h1	230	33	36	299
2014h2	269	32	44	345
2015h1	303	42	33	378
2015h2	252	38	41	331
2016h1	304	33	41	378
2016h2	325	43	45	413
2017h1	401	57	46	504
total	13636	1631	2087	3851

 Table F-7:
 Frequency table of antibiotic prescribing rates per half-yearly period, by patient group

	2012h1	2012h2	2013h1	2013h2	2014h1	2014h2	2015h1	2015h2	2016h1	2016h2	2017h1
Women											
Prescribing Rate	0.85	0.85	0.85	0.85	0.85	0.87	0.89	0.89	0.88	0.88	0.88
First-line Prescribing Rate	0.48	0.47	0.45	0.43	0.44	0.44	0.42	0.42	0.44	0.42	0.40
Second-line Prescribing Rate	0.33	0.35	0.36	0.37	0.38	0.38	0.42	0.42	0.40	0.42	0.44
Third-line Prescribing Rate	0.07	0.07	0.09	0.12	0.11	0.10	0.09	0.08	0.08	0.07	0.07
Not Recommended Prescribing Rate	0.11	0.11	0.10	0.09	0.08	0.08	0.08	0.08	0.08	0.09	0.09
Non-first-line Prescribing Rate	0.52	0.53	0.55	0.57	0.56	0.56	0.58	0.58	0.56	0.58	0.60
Men											
Prescribing Rate	0.73	0.76	0.75	0.75	0.79	0.78	0.79	0.80	0.78	0.79	0.81
First-line Prescribing Rate	0.35	0.37	0.45	0.40	0.30	0.27	0.27	0.33	0.35	0.28	0.25
Second-line Prescribing Rate	0.34	0.35	0.32	0.31	0.33	0.35	0.40	0.39	0.35	0.40	0.42
Third-line Prescribing Rate	0.20	0.19	0.19	0.24	0.30	0.31	0.25	0.23	0.23	0.23	0.24
Not Recommended Prescribing Rate	0.12	0.09	0.04	0.05	0.07	0.07	0.08	0.05	0.06	0.09	0.09
Non-first-line Prescribing Rate	0.65	0.63	0.55	0.60	0.70	0.73	0.73	0.67	0.65	0.72	0.75
<u>Children</u>											
Prescribing Rate	0.70	0.71	0.75	0.76	0.74	0.74	0.75	0.77	0.77	0.75	0.72
First-line Prescribing Rate	0.13	0.17	0.18	0.17	0.20	0.17	0.15	0.18	0.20	0.20	0.16
Second-line Prescribing Rate	0.54	0.53	0.55	0.53	0.53	0.58	0.58	0.59	0.58	0.56	0.60
Third-line Prescribing Rate	0.16	0.14	0.14	0.20	0.19	0.18	0.18	0.13	0.13	0.17	0.14
Not Recommended Prescribing Rate	0.17	0.16	0.13	0.10	0.08	0.07	0.09	0.10	0.08	0.08	0.10
Non-first-line Prescribing Rate	0.87	0.83	0.82	0.83	0.80	0.83	0.85	0.82	0.80	0.80	0.84

1 F.2 Additional time series plots

F.2.1 All patients with initial presentations of urinary tract infection



Figure F-1: Time series plot of second-line antibiotic prescribing for all patients with initial presentations of urinary tract infection over time, by month



Figure F-2: Time series plot of antibiotic prescribing for all patients with initial presentations
 of urinary tract infection over time, by month

1 F.2.2 Women sixteen years and over with initial presentations of urinary tract

2 infection



Figure F-3: Time series plot of second-line antibiotic prescribing for women of sixteen years and over with initial presentation of urinary tract infection over time, by month



9 Figure F-4: Time series plot of second-line antibiotic prescribing for women of sixteen years 10 and over with initial presentations of urinary tract infection over time, by month



- 2 Figure F-5: Time series plot of prescribing for antibiotic agents not recommended in the
- 3 guidelines for women of sixteen years and over with initial presentations of urinary tract
- 4 infection over time, by month
- 5

F.2.3 Men sixteen years and over with initial presentations of urinary tract infection



8

9 Figure F-6: Time series plot of non-first-line antibiotic prescribing for all men of sixteen 10 years and over with initial presentation of urinary tract infection over time, by month





Figure F-7: Time series plot of second-line antibiotic prescribing for all men of sixteen years

and over with initial presentations of urinary tract infection over time, by month





F.2.4 Children under sixteen years of age with initial presentations of urinary 1 tract infection

2 3



4 5

6 7

Time series plot of non-first-line antibiotic prescribing for children under sixteen Figure F-9: years over time, by month



9 Figure F-10: Time series plot of second-line antibiotic prescribing for children under sixteen



11

APPENDIX G – APPENDIX TO THE DISCUSSION (CHAPTER 8): COMPARISON WITH QUALITY INDICATORS

3 4

5

G.1 Comparison with quality indicators for upper respiratory tract infection

The antibiotic prescribing in this dataset for initial presentations of URTI were compared against ESAC prescribing indicators and WHO AWaRe program targets (6,11). Antibiotic prescribing rates were 31%, 74%, 58% and 12% for acute rhinosinusitis, acute pharyngitis/tonsillitis, AOM and influenza/ILI, respectively. Of conditions excluding influenza/ILI, non-first-line prescribing occurred 68% of the time for rhinosinusitis, 47% of the time for pharyngitis/tonsillitis and 53% for AOM.

12

As detailed in **Table G-1**, for the strong majority of ESAC indicators published in 2011 (6), the proportions of patients treated with antibiotics for each and all condition(s) were higher than recommended. Non-first-line prescribing was substantially higher than recommended for all URTI conditions (6). Quinolone use was, however, consistently within the recommended low range (6).

18

These results suggest that there is room for improvement with respect to decisions whether to treat with antibiotics, and substantial room for improvement regarding the choice of antibiotic agent when prescribing. Performance on indicators regarding the use of recommended agents was poor, with the exception of quinolone use (6).

23

As seen in **Figure G-1**, 84% of antibiotic prescriptions for URTI in this dataset were on the 2019 WHO Access list, in excess of the 60% level (11). By URTI condition, this ranged from 65% for influenza/ILI to 90% for AOM. There were no instances of use for any antibiotics classified by WHO as 'Reserve' (11).

28

The WHO AWaRe program (current 2019) provides targets for at least 60% systemic antibacterials to be from their Access list (11). Performance against this indicator was well above the target for all conditions apart from influenza/ILI (11). The results compared against the WHO AWaRe program appear promising, however, they should be interpreted with a degree of caution (11). Note the WHO AWaRe program sets the same targets for use in countries with little pharmaceutical or medical legislation, regulation and oversight (11).

Table G-1: Compilation of indicators including the European Surveillance of Antibiotic Consumption indicators by Adriaenssens et al. (6) and relating to respiratory tract infection (6)

Prescribing Quality Measure / Indicator	Recommendation	Condition					
		Acute URTI (excluding influenza/ILI & AOM)	Acute Rhinosinusitis	Acute Pharyngitis / Tonsillitis	Acute Otitis Media (AOM)	Influenza/ ILI	
prescribing rate (%)		31%	31%	71%	58%	12%	
non-first-line antibiotic prescriptions (%)		68%	68%	47%	53%	n/a*	
ESAC indicator 2b: Percentage of patients older than 1 year with acute upper respiratory infection (ICPC-2-R: R74) prescribed antibacterials for systemic use (ATC: J01) (R74_J01_%)	0-30%	33%	33%	71%	59%	12%	
ESAC indicator 2b: =2a receiving the recommended antibacterials (ATC: J01CE) (R74_RECOM_%)	80-100%	32%	32%	53%	47%	n/a*	
ESAC Indicator 2c: 2c. =2a receiving quinolones (ATC: J01M) (R74_J01M_%)	0-5%	<<1%	<<1%	<<1%	1%	<1%	
ESAC Indicator 4a: Percentage of patients older than 1 year with acute tonsillitis (ICPC-2-R: R76) prescribed antibacterials for systemic use (ATC: J01)	0-20%			71%			
ESAC Indicator 4b: =4a receiving the recommended antibacterials (ATC: J01CE) (R76_RECOM_%)	80-100%			53%			
ESAC Indicator 4c:=4a receiving quinolones (ATC: J01M)	0-5%			<<1%			
ESAC Indicator 5a: 5a. Percentage of patients older than 18 years with acute/chronic sinusitis (ICPC-2-R: R75) prescribed antibacterials for systemic use (ATC: J01)+	0-20%		33%				
ESAC Indicator 5b: .=5a receiving the recommended antibacterials (ATC: J01CA or J01CE)+	80-100%		32%				
ESAC Indicator 5c: =5a receiving quinolones (ATC: J01M) (R75_J01M_%) +	0-5%		<<1%				
ESAC Indicator 6a: Percentage of patients older than 2 years with acute otitis media/myringitis (ICPC-2-R: H71) prescribed antibacterials for systemic use (ATC: J01)&	0-20%				59%		
ESAC Indicator 6b:= 6a receiving the recommended antibacterials (ATC: J01CA or J01CE)&	80-100%				56%		
ESAC Indicator 6c: =6a receiving quinolones (ATC: J01M)&	0-5%				<1%		
prescriptions on WHO AWaRe Access list (%)	>=60%	82%	76%	87%	90%	65%	

Note: + chronic sinusitis excluded from this dataset such that comparison must keep the difference in diagnostic criteria in mind.

Note: & myringitis excluded from acute otitis media diagnostic group .



Figure G-1: Comparison of the antibiotic prescribing for upper respiratory tract infection conditions in this dataset with the World Health Organization's 2019 Access, Watch and Reserve classifications (11)

G.2 Comparisons with quality indicators for urinary tract infection

1 2

The prescribing rate was 85% for all patients, with prescribing rates of 87% for women, and 3 78% for men and 75% for children, respectively. 57% of women, 68% men and 82% children 4 5 under sixteen received antibiotic prescriptions other than first-line. The prescribing rate was 87% for women at least 16 years of age and the same for women at least 18 years of age. 6 7 ESAC-net indicators presented by Adriaenssens et al. (6) recommend 80-100% for women at least 18 years with acute cystitis. However, as seen in Table G-2 below, only 43% of 8 9 women in this dataset (aged either sixteen years and over or eighteen years and over) received the recommended (first-line) antibiotic, whereas 80-100% is also the 10 recommended range for ESAC (6). Quinolone use among women with acute cystitis was 11 within ESAC range of under five percent, coming in at 2% for women of at least sixteen or 12 13 eighteen years of age in this dataset (6).

- 14
- 15
- 16

4

2 3

Table G-2: Compilation of indicators including the European Surveillance of Antibiotic Consumption indicators published by Adriaenssens et al. (6) relating to acute cystitis

Measure		
	Recommendation	Findings *
prescribing rate all patients (%)	n/a	85%
non-first-line antibiotic prescriptions all patients (%)	n/a	60%
Percentage of female patients older than 18 years with cystitis/ other urinary infection (ICPC-2-R: U71) prescribed antibacterials for systemic use (ATC: J01)	80-100%	87%
=3a. receiving the recommended antibacterials (ATC: J01XE or J01EA or J01XX)	80-100%	43%
=3a receiving quinolones (ATC: J01M)	0-5%	2%
prescriptions on WHO AWaRe Access list (%)	>=60%	97%

5

Note * same result confirmed regardless of age cut-off- for women 16 years and over and over 18 years.

- 6
- 7

UTI performed reasonably well against 2019-released WHO AWaRe program (11). Only 2% 8 9 antibiotics prescribed were on the Watchlist (11). However, these are all antibiotics that are Not Recommended in Australian guidelines at the time. So in another sense, improvements 10 can also be made. 11

12

These result from comparing quality indicators suggest that there is substantial room for 13 improvement with respect to the choice of antibiotic when prescribing, but that the decision 14 to prescribe is generally good. The data indicates reasonable quinolone prescribing within 15 the range of ESAC indicator 3c (6). Furthermore, WHO AWaRe program (11) (as at 2019 16 and closest to the study period) provides targets of at least 60% systemic antibacterials 17 prescribed to be from WHO's Access list, for which these WA-based GPs adhered to 97% 18 of the time (11). There were no instances of prescriptions from WHO's (2019) Reserve list 19 (11). 20

Figure G-2: Comparison of the antibiotic prescribing for urinary tract infection in this dataset with the World Health Organization's 2019 Access,
 Watch and Reserve classifications (11)



1 REFERENCES TO APPENDICES

- Antibiotic Expert Groups. Therapeutic guidelines: respiratory Version 4. West Melbourne:
 Therapeutic Guidelines Limited; 2009.
- Antibiotic Expert Groups. Therapeutic guidelines: antibiotic. Version 15. Melbourne: Therapeutic
 Guidelines Limited; 2014. ISBN: 9780992527211.
- Antibiotic Expert Groups. Therapeutic guidelines: antibiotic. Version 16. Melbourne: Therapeutic
 Guidelines Limited; 2019.
- Antibiotic Expert Groups. Therapeutic guidelines: antibiotic. Version 14. Melbourne: : Therapeutic Guidelines Limited.; 2010. <u>http://www.tg.org.au/</u>
- Adriaenssens N, Coenen S, on behalf of the European Surveillance of Antimicrobial Consumption
 Management Team. Disease-specific antibiotic prescribing quality indicators report [Internet].
 Antwerp, Belgium: University of Antwerp; 2010 [cited 2015 Jun 27]. 56.
- Adriaenssens N, Coenen S, Versporten A, Muller A, Vankerckhoven V, Goossens H, on behalf
 of the ESAC Project Group. European Surveillance of Antimicrobial Consumption (ESAC): quality
 appraisal of antibiotic use in Europe []. J Antimicrob Chemother. 2011;66(Suppl 6):vi71-vi77.
 doi:10.1093/jac/dkr459.
- Furopean Centre for Disease Prevention and Control. Surveillance of antimicrobial consumption in Europe 2011. ECDC, editor. Stockholm: European Centre for Disease Prevention and Control; 2014. ISBN: 978-92-9193-550-5.
- Coenen S, Ferech M, Haaijer-Ruskamp FM, Butler CC, Vander Stichele RH, Verheij TJM, Monnet
 DL, Little P, Goossens H, the ESAC Project Group. European Surveillance of Antimicrobial
 Consumption (ESAC): quality indicators for outpatient antibiotic use in Europe. Qual Safety Health
 Care. 2007;16(6):440-445. doi:10.1136/qshc.2006.021121.
- Hansen M, Bjerrum, L, , Gahrn-Hansen B, Jarbol, DE, . Quality indicators for diagnosis and treatment of respiratory tract infections in general practice: A modified Delphi study. Scand J Prim Health Care. 2010;28(1):4-11. doi:10.3109/02813431003602724.
- Le Maréchal M, Tebano G, Monnier AA, Aiaenssens N, Gyssens ICJ, Huttner B, Milanic R, Schouten JA, Benic MS, Versporten A, Vlahovic-Palcevski V, Zanichelli V, Wertheim HFL, Hulscher ME, Pulcini C. Quality indicators assessing antibiotic use in the outpatient setting: a systematic review followed by an international multidisciplinary consensus procedure. J Antimicrob Chemother. 2018;73(suppl_6):vi40-vi49. doi:10.1093/jac/dky117.
- 33 11. World Health Organization. WHO releases the 2019 AWaRe Classification Antibiotics Geneva: 34 World Health Organization: 2019 Feb 20. [2019 Oct 19]. Available from: https://www.who.int/medicines/news/2019/WHO releases2019AWaRe classification antibiotics 35 36 /en/
- Sharland M, Gandra S, Huttner B, Moja L, Pulcini C, Zeng M, Mendelson M, Cappello B, Cooke 37 12. G, Magrini N, Aziz Z, Cavalli F, De Vries E, Genazzani A, Imi M, Kearns G, Kokwaro G, Prutsky 38 39 GJ, Sarrafzadegan N, Sri Ranganathan S, Suleman F, Yoongthong W, Harbarth S, Loeb M, Mertz D, Tacconelli E, Villegas MV. Encouraging AWaRe-ness and discouraging inappropriate antibiotic 40 use-the new 2019 Essential Medicines List becomes a global antibiotic stewardship tool. The 41 2019/12/01/;19(12):1278-1280. 42 Lancet Infectious Diseases. 2019 43 doi:https://doi.org/10.1016/S1473-3099(19)30532-8.
- Sharland M, Pulcini C, Harbarth S, Zeng M, Gandra S, Mathur S, Magrini N. Classifying antibiotics
 in the WHO Essential Medicines List for optimal use—be AWaRe. The Lancet Infectious
 Diseases. 2018;18(1):18-20. doi:10.1016/S1473-3099(17)30724-7.
- 47 14. World Health Organization. 2021 AWaRe classification - WHO access, watch, reserve, 48 classification of antibiotics for evaluation and monitoring of use. Geneva: World Health 49 Organization: 2021 Sept 30. [2021 Sept 301. Available from: https://www.who.int/publications/i/item/2021-aware-classification 50
- 51

- Thilly N, Pereira O, Schouten J, Hulscher ME, Pulcini C. Proxy indicators to estimate 1 15. appropriateness of antibiotic prescriptions by general practitioners: a proof-of-concept cross-2 sectional study based on reimbursement data, north-eastern France 2017. Euro Surveill. 2020 3 Conflict of interest: None 4 Jul;25(27). eng. declared. doi:10.2807/1560-5 7917.Es.2020.25.27.1900468. Cited in: Pubmed; PMID 32672150.
- 6 16. StataCorp. Stata Statistical Software: Release 16. . College Station, TX: StataCorp LLC.; 2019.
- If. Little R. Statistical analysis with missing data / Roderick J. A. Little, Donald B. Rubin. 2nd ed.:
 Hoboken, New Jersey : John Wiley & Sons, Inc.; 2002. (Rubin DBa, editor.).
- 9 18. Josse J, Husson F. missMDA: A Package for Handling Missing Values in Multivariate Data
 10 Analysis. J Stat Soft. 2016 Apr 4;70(1):1 31. doi:10.18637/jss.v070.i01.
- Chen CC, Wu LC, Li CY, Liu CK, Woung LC, Ko MC. Non-adherence to antibiotic prescription guidelines in treating urinary tract infection of children: a population-based study in Taiwan [Research Support, Non-U.S. Gov't]. J Eval Clin Pract. 2011 Dec;17(6):1030-5. English. doi:http://dx.doi.org/10.1111/j.1365-2753.2010.01469.x. Cited in: Pubmed; PMID 20738469.
- Coco AS, Horst MA, Gambler AS. Trends in broad-spectrum antibiotic prescribing for children
 with acute otitis media in the United States, 1998–2004. BMC Pediatr. 2009;9:41-41.
 doi:10.1186/1471-2431-9-41.
- Steering Committee for the Review of Government Service Provision. Report on government
 services 2015: Indigenous compendium. . Canberra: Productivity Commission, Australian
 Government; 2015.
- Australian Institute of Health and Welfare. National best practice guidelines for collecting Indigenous status in health data sets [Internet]. Canberra: Australian Institute of Health and Welfare; 2010 [cited 2022 Nov 1]. Available from: <u>https://www.aihw.gov.au/reports/indigenous-</u> australians/national-guidelines-collecting-health-data-sets/notes
- 25 23. Copp HL, Shapiro DJ, Hersh AL. National ambulatory antibiotic prescribing patterns for pediatric
 26 urinary tract infection, 1998-2007. Pediatrics. 2011;127(6):1027. doi:10.1542/peds.2010-3465.
- 27 24. Kallen AJ, Welch HG, Sirovich BE. Current Antibiotic Therapy for Isolated Urinary Tract Infections
 28 in Women. Archives of internal medicine (1960). 2006;166(6):635-639.
 29 doi:10.1001/archinte.166.6.635.
- Martinez ME. The calendar of epidemics: Seasonal cycles of infectious diseases. PLoS Pathog.
 2018 Nov;14(11):e1007327. eng. The authors have declared that no competing interests exist.
 Epub 2018 Nov 8. doi:10.1371/journal.ppat.1007327. Cited in: Pubmed; PMID 30408114.
- Moriyama M, Hugentobler WJ, Iwasaki A. Seasonality of Respiratory Viral Infections. Annu Rev Virol. 2020;7(1):83-101. doi:10.1146/annurev-virology-012420-022445. Cited in: Pubmed; PMID 32196426.
- Durkin MJ, Jafarzadeh SR, Hsueh K, Sallah YH, Munshi KD, Henderson RR, Fraser VJ.
 Outpatient Antibiotic Prescription Trends in the United States: A National Cohort Study. Infect
 Control Hosp Epidemiol. 2018;39(5):584-589. Epub 2018 Mar 8. doi:10.1017/ice.2018.26.
- Sauerbrei W, Abrahamowicz M, Altman DG, Cessie S, Carpenter J. STRengthening Analytical
 Thinking for Observational Studies: the STRATOS initiative. Stat Med. 2014;33(30):5413-5432.
 doi:10.1002/sim.6265.
- 42 29. Ahmed H. Farewell D, Jones HM, Francis NA, Paranjothy S, Butler CC. Incidence and antibiotic 43 prescribing for clinically diagnosed urinary tract infection in older adults in UK primary care. 2004-2014. PLoS ONE. 2018 44 Jan:13(1) (no pagination). Enalish. doi:http://dx.doi.org/10.1371/journal.pone.0190521. Cited in: Pubmed; PMID 620083241. 45
- Gong Y, Li H, Yang H, Tan K, Liu W, Li X, Wu J, Zhang G, Yin X. Evaluation of the Quality of 46 30. Antibiotic Prescribing in Primary Care: A Multicenter Longitudinal Study From Shenzhen, China. 47 pagination). 48 Front Pharmacol. 2020 Feb 19:11 (no English. 49 doi:http://dx.doi.org/10.3389/fphar.2020.617260. Cited in: Pubmed; PMID 634383882.
- Andrews A, Budd EL, Hendrick A, Ashiru-Oredope D, Beech E, Hopkins S, Gerver S, MullerPebody B. Surveillance of antibacterial usage during the COVID-19 pandemic in England, 2020.
 Antibiotics. 2021 Jul;10(7) (no pagination). English.
 doi:<u>http://dx.doi.org/10.3390/antibiotics10070841</u>. Cited in: Pubmed; PMID 2007864100.

- 32. Gunnlaugsdottir MR, Linnet K, Jonsson JS, Blondal AB. Encouraging rational antibiotic prescribing behaviour in primary care - prescribing practice among children aged 0-4 years 2016-2018: an observational study. Scand J Prim Health Care. 2021 Sep 1;39(3):373-381. English. doi:http://dx.doi.org/10.1080/02813432.2021.1958506. Cited in: Pubmed; PMID 635812798.
- Mulder M, Baan E, Verbon A, Stricker B, Verhamme K. Trends of prescribing antimicrobial drugs for urinary tract infections in primary care in the netherlands: A population-based cohort study.
 BMJ Open. 2019 May 1;9(5) (no pagination). English. doi:<u>http://dx.doi.org/10.1136/bmjopen-</u>
 2018-027221. Cited in: Pubmed; PMID 627736217.
- 9 34. Piraux A, Faure S, Naber KG, Alidjanov JF, Ramond-Roquin A. Changes in the management of 10 urinary tract infections in women: impact of the new recommendations on antibiotic prescribing 11 behavior in France, between 2014 and 2019. BMC Health Serv Res. 2021 Jun 28;21(1):612.
 12 English. doi:<u>http://dx.doi.org/10.1186/s12913-021-06653-4</u>. Cited in: Pubmed; PMID 635479369.
- 35. World Health Organization. Anatomical Therapeutic Chemical classification system: Structure and principles [Internet]. Oslo, Norway: WHO Collaborating Centre for Drug Statistics Methodology; 2019. [updated 2018 Feb 15; cited 2019 Nov 1]. Available from: https://www.whocc.no/atc/structure_and_principles/
- 17
- 18
- 19